# THE GENETICS OF ALZHEIMER'S DISEASE, MODELLING DISABILITY AND ADVERSE SELECTION IN THE LONG-TERM CARE INSURANCE MARKET

By

Delme John Pritchard

SUBMITTED FOR THE DEGREE OF DOCTOR OF PHILOSOPHY AT HERIOT-WATT UNIVERSITY ON COMPLETION OF RESEARCH IN THE DEPARTMENT OF ACTUARIAL MATHEMATICS & STATISTICS MARCH 2002.

This copy of the thesis has been supplied on the condition that anyone who consults it is understood to recognise that the copyright rests with its author and that no quotation from the thesis and no information derived from it may be published without the prior written consent of the author or the university (as may be appropriate). I hereby declare that the work presented in this thesis was carried out by myself at Heriot-Watt University, Edinburgh, except where due acknowledgement is made, and has not been submitted for any other degree.

Delme John Pritchard (Candidate)

Professor Angus Macdonald (Supervisor)

Date

### Contents

A	Acknowledgements xxxi			
A	Abstract xxxii			
In	Introduction 1			
1	A S	Simple Model of Alzheimer's Disease and the APOE Gene	6	
	1.1	Introduction	6	
	1.2	Alzheimer's Disease and its Genetic Components	8	
		1.2.1 The Apolipoprotein E Gene	9	
		1.2.2 Other Genetic Factors of Alzheimer's Disease	11	
	1.3	A Model for Alzheimer's Disease	12	
		1.3.1 Statistical Framework	14	
	1.4	Estimation of Transition Intensities Not Depending on APOE Genotype	15	
		1.4.1 Baseline Mortality Tables	17	
		1.4.2 The Onset of Alzheimer's Disease in the Population	17	
		1.4.3 Time from Onset of Alzheimer's Disease to Institutionalisation		
		or Death	19	
		1.4.4 Mortality of Lives with Alzheimer's Disease	23	
	1.5	Estimation of Transition Intensities Depending on APOE Genotype .	26	
	1.6	Occupancy Probabilities	41	
	1.7	Summary and Discussion	51	
		1.7.1 Discussion $\ldots$	51	
<b>2</b>	The	e Potential for Adverse Selection in Long-Term Care Insurance		
	from	n Individuals Knowing their APOE Genotype	55	
	2.1	Introduction	55	
	2.2	Long-Term Care Insurance	57	
	2.3	Inclusion of Payment Streams in the Alzheimer's disease model	58	
	2.4	The Costs of Alzheimer's Disease in LTC Insurance	60	
		2.4.1 Long-Term Care Insurance Costs and APOE Genotype	63	
	2.5	Sensitivity Analysis	66	
	2.6	The Impact of Adverse Selection on LTC Insurance	74	
		2.6.1 An Anomaly? Adverse Selection for Males	77	
		2.6.2 The Cost of Adverse Selection based on Alzheimer's Disease		
		Alone	78	

		2.6.3 The Costs of Adverse Selection Based on Total LTC Insurance	~ 1
	2.7	The Cost of a Combined Pension and Long-Term Care Package	84 86
	2.0		90
3	Mo	delling Disability in Long-Term Care Insurance	92
	3.1	Introduction	92
	3.2	Activities of Daily Living (ADLs)	93
	3.3	A Model of Disability in Long-Term Care Insurance	94
	3.4	Overview of the U.S. National Long-Term Care Surveys	96
	3.5	Previous Research Using the NLTCS	101
	3.6	Anomalies in the Data	103
		3.6.1 Anomalies in the 1982 NLTCS	104
		3.6.2 Anomalies in the 1989 NLTCS	105
	3.7	Details of the 1982 and 1984 NLTCS	105
	3.8	Details of the 1984 and 1989 NLTCS	109
	3.9	Details of the 1989 and 1994 NLTCS	112
4	Est	imating the Transition Intensities in the Disability Model	114
	4.1	Introduction	114
	4.2	Calculation of Transition Probabilities	115
	4.3	Transformation of Transition Probabilities to Transition Intensities .	121
	4.4	Approximate (or Constrained to Real) Maximum Likelihood Esti- mates of the Transition Intensities	125
	4.5	Constrained (Positive) Maximum Likelihood Estimates of the Tran- sition Intensities	126
	4.6	Numerical Calculation of the Constrained (Real and Positive) Maxi- mum Likelihood Estimates	129
	4.7	Constrained Maximum Likelihood Estimates Compared with Uncon- strained Maximum Likelihood Estimates	130
_	C		
5	Cor	indence Intervals and Graduation of Transition Intensities in Disability Model	1 1/10
	5.1	Introduction	140
	5.2	Comparison of Two Methods for Estimating the Variance of the Tran-	1 1 1
	50	sition Intensities	141
	5.3	Comparison of Variance Estimates in Three Simple Models	149
	5.4	Disability Model	168
	5.5	Graduating the Transition Intensities Grouped in 10-year Age Bands	180
	5.6	Graduating the Transition Intensities Grouped in 5-year Age Bands $% \mathcal{A}$ .	188
6	Ove	erall Mortality in the Disability Model	196
	6.1	Introduction	196
	6.2	A Benchmark Force of Mortality at Older Ages	197
	6.3	Overall Model Mortality, Using the Fitted Parametric Transition In-	
		tensities from the 1982–84 NLTCS	200

	6.4	Overall Model Mortality, Using the Fitted Parametric Transition In- tensities from the 1984–89 NLTCS	208
	6.5	Overall Model Mortality, Using the Fitted Parametric Transition In-	910
	6.6	Comparison of Overall Mortality in the 1982–84, 1984–89 and 1989–	219
		94 Disability Models	229
7	Mod Insu 7.1 7.2 7.3 7.4 7.5	del Costs of Disability and Adverse Selection in Long-Term Care         trance Revisited         Introduction	e 234 236 245 253 261
8	Area	as for Further Research	266
Re	eferer	ices	269
A	Ove	rview of the 1982, 1984, 1989 and 1994 National Long-Tern	ı
	Care	e Surveys — Transitions between States	281
В	Nun	nbers of transitions between disability states of the 1982 and	ł
	1984	A NLTCS by gender and age group	285
С	Nun	nbers of transitions between disability states of the 1984 and	ł
	1989	O NLTCS by gender and age group	290
D	Nun	nbers of transitions between disability states of the 1989 and	ł
	1994	A NLTCS by gender and age group	295
Е	5-ye	ar transition probabilities between disability states calculated	ł
	fron	a the 1984, 1989 and 1994 National Long-Term Care Surveys	300
F	Max	timum likelihood estimates of the annual transition intensities	s
	betv	ween disability states calculated from the 1984, 1989 and 1994	1
	Nati	ional Long-Term Care Surveys	307
G	Con	strained maximum likelihood estimates of the annual transition	n
	inter	nsities between disability states calculated from the 1982, 1984	,
	1989	) and 1994 National Long-Term Care Surveys	314
Н	Log-	likelihood values for the maximum likelihood estimates, ad	-
	juste	ed maximum likelihood estimates and the constrained maxi	-
	mun	n likelihood estimates of the transition intensities calculated	1
	from	n the 1984–89 and 1989–94 NLTCS	330

- I 5-year transition probabilities between disability states calculated from the constrained transition intensities of the 1984, 1989 and 1994 National Long-Term Care Surveys 333
- J Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982, 1984, 1989 and 1994 National Long-Term Care Surveys 340
- K Tables of parameter values for the parametric transition intensities fitted to data grouped in 10-year age bands, calculated from the 1982, 1984, 1989 and 1994 NLTCS 356
- L Tables of parameter values for the parametric transition intensities fitted to data grouped in 5-year age bands, calculated from the 1982, 1984, 1989 and 1994 NLTCS 365
- M Graphs of the constrained (positive) MLEs of the transition intensities, 95% confidence intervals and parametric fits, using data in 5-year age bands from the 1982 and 1984 NLTCS 374
- N Graphs of the constrained (positive) MLEs of the transition intensities, 95% confidence intervals and parametric fits, using data in 5-year age bands from the 1984 and 1989 NLTCS
  381
- O Graphs of the constrained (positive) MLEs of the transition intensities, 95% confidence intervals and parametric fits, using data in 5-year age bands from the 1989 and 1994 NLTCS
  391

## List of Tables

1.1	Mean and median times to Institutionalisation (Inst'n) or First Event	
	for AD Patients.	21
1.2	Summary Statistics on Survival Times of AD Patients.	25
1.3	Summary Of Transition Intensities for the AD Model with Baseline	
	Mortality 100% (65%) of AM80 and AF80	26
1.4	Estimated Population Frequency of ApoE Genotypes	28
1.4	Estimated Population Frequency of ApoE Genotypes-continued	29
1.5	Aggregated Odds Ratios of AD for the ApoE $\varepsilon 4$ Allele	31
1.6	Odds Ratios of AD by Genotype and Age	32
1.7	Parameters for the Relative Risk of AD for Males, Females and in	
	Aggregate, by Genotype	36
1.8	$r_m$ for $m = 1, 0.5$ and 0.25	39
1.9	Frequencies of APOE genotypes among lives free of Alzheimer's dis-	
	ease at ages 65, 70 and 75, estimated by solving the Kolmogorov	
	equations forward from age 60	52
2.10	Mean, variance and skewness $(q = 1, 2 \text{ and } 3)$ of the present value	
	of AD claims costs, unit benefit increasing continuously ( $\delta_b = 0.05$ ),	
	using the aggregate incidence rate of AD	61
2.11	Expected present values (EPV) of unit benefit increasing continuously	
	$(\delta_b = 0.05)$ , depending on the incidence of AD after age 90	61
2.12	EPV of unit benefit increasing continuously ( $\delta_b = 0.05$ ), with gender-	
	specific mortality and incidence of AD.	62
2.13	EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) by genotype,	~
0.1.4	and averages over genotypes	65
2.14	Approximate underwriting ratings (equivalent to percentage extra	
	premiums) for the $\varepsilon 4/\varepsilon 4$ and female $\varepsilon 3/\varepsilon 4$ & $\varepsilon 2/\varepsilon 4$ genotypes, as-	
	suming total LTC insurance costs are twice AD-related costs. Unit	<b>C7</b>
0.15	benefit increasing continuously $(\delta_b = 0.05)$	67
2.15	EPV of benefits commencing 1 year before institutionalisation ( $w = 1$	
	years) as a % of EPV of benefits commencing on institutionalisation $(w = 0 \text{ wears})$ for unit barefit increasing continuously $(\delta = 0.05)$ by	
	$(w = 0 \text{ years})$ , for unit benefit increasing continuously $(b_b = 0.05)$ by	69
9 16	genotype and averages over genotypes. $\dots$	00
2.10	En V of benefits commencing 2 years before institutionalisation ( $w = 2$ years) as a $\%$ of EDV of benefits commonsing on institutionalisation	
	2  years as a 70 of ET V of benefits commencing on institutionalisation ( $w = 0 \text{ years}$ ) for unit benefit increasing continuously ( $\delta_{1} = 0.05$ ) by	
	$(w - 0 \text{ years})$ , for unit benefit increasing continuously $(v_b = 0.05)$ by	60
	genotype and averages over genotypes	09

2.1	7 EPVs of unit benefit increasing continuously ( $\delta_b = 0.025$ ) as a per- centage of EPV of unit benefit increasing continuously ( $\delta_b = 0.05$ )	
	by genotype and averages over genotypes for benefits commencing on institutionalisation.	70
2.1	8 EPVs of level unit benefit ( $\delta_b = 0$ ) as a percentage of EPV of unit benefit increasing continuously ( $\delta_b = 0.05$ ) by genotype, and averages	70
2.1	over genotypes for benefits commencing on institutionalisation 9 Sensitivity of the EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) to the mortality assumptions, for Females. Level of relative risk,	70
2.2	m = 1.00.	71
2.2	m = 0.25.	72
2.2	m = 1.00.	72
2.2	m = 0.25	73
$2.2^{-1}$	<ul><li>0.05) to the addition K to the force of mortality in the institution- alised state, for Females. Baseline mortality: 65% of AF80</li><li>4 Costs of adverse selection, for males, as a percentage of AD-related</li></ul>	74
	LTC insurance costs in the absence of adverse selection, with $\varepsilon 2/\varepsilon 4$ , $\varepsilon 3/\varepsilon 4$ and $\varepsilon 4/\varepsilon 4$ genotypes k times as likely to insure as low risk genotypes, for unit benefits increasing continuously ( $\delta_b = 0.05$ ) and	70
2.2	5 Costs of adverse selection as a percentage of AD-related LTC in- surance costs in the absence of adverse selection, with $\varepsilon 2/\varepsilon 4$ , $\varepsilon 3/\varepsilon 4$ and $\varepsilon 4/\varepsilon 4$ genotypes twice as likely to insure as low-risk genotypes $(k = 2)$ , for unit benefits increasing continuously ( $\delta_b = 0.05$ ) and	. 79
2.2	commencing w years before institutionalisation	80
2.2	commencing w years before institutionalisation	81
	$\varepsilon 4/\varepsilon 4$ genotypes 10 times as likely to insure as low-risk genotypes $(k = 10)$ , for unit benefits increasing continuously $(\delta_b = 0.05)$ and	01
2.2	8 Costs of adverse selection as a percentage of AD-related LTC insur- ance costs in the absence of adverse selection, with $\varepsilon 2/\varepsilon 4$ , $\varepsilon 3/\varepsilon 4$ and $\varepsilon 4/\varepsilon 4$ genetupes 100 times as likely to incurse a large selection.	01
	$\varepsilon_4/\varepsilon_4$ genotypes 100 times as likely to insure as low-risk genotypes $(k = 100)$ , for unit benefits increasing continuously $(\delta_b = 0.05)$ and commencing w years before institutionalisation.	82

2.29	Costs of adverse selection, for females, as a percentage of AD-related	
	LTC insurance costs in the absence of adverse selection, with $\varepsilon 2/\varepsilon 4$ ,	
	$\varepsilon_3/\varepsilon_4$ and $\varepsilon_4/\varepsilon_4$ genotypes k times as likely to insure as low risk	
	genotypes, for level unit benefits and benefits increasing continuously	
	$(\delta_b = 0 \text{ and } 0.025)$ and commencing on institutionalisation	
2.30	Costs of adverse selection as a percentage of total LTC insurance	
2.00	costs with $\varepsilon^{2/\epsilon 4}$ $\varepsilon^{3/\epsilon 4}$ and $\varepsilon^{4/\epsilon 4}$ genotypes k times as likely to	
	incure as low risk genetypes for bonefits increasing continuously ( $\delta_{\rm c}$ =	
	insure as low-fisk genotypes, for benefits increasing continuously $(b_b = 0.05)$ and commonoing on institutionalisation	
0.91	(0.05) and commencing on institutionalisation. $(5.002)$ with	
2.31	Comparison of the EPV of a pension of $\pm 3,254$ p.a. $(b_b = 0.03)$ with	
	the EPV of a package of a pension of £3,254 p.a. $(b_b = 0.03)$ and a	
	care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from	
	AD, for females. High level of relative risk $m = 1, \ldots, 88$	
2.32	Comparison of the EPV of a pension of $\pounds 3,254$ p.a. ( $\delta_b = 0.03$ ) with	
	the EPV of a package of a pension of $\pounds 3,254$ p.a. ( $\delta_b = 0.03$ ) and a	
	care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from	
	AD, for females. Low level of relative risk, $m = 0.25$	
2.33	Comparison of the EPV of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) with	
	the EPV of package of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) and a	
	care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from	
	AD, for males. High level of relative risk $m = 1, \ldots, 89$	
2.34	Comparison of the EPV of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) with	
	the EPV of package of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) and a	
	care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from	
	AD, for males. Low level of relative risk $m = 0.25$	,
3.35	Transitions between disability states in the 1982 and 1984 National	
	Long-Term Care Surveys, unadjusted for censored data	
3.36	Transitions between disability states in the 1982 and 1984 National	
	Long-Term Care Surveys, adjusted for censored data	
3.37	Transitions between disability states in the 1984 and 1989 National	
0.01	Long-Term Care Surveys unadjusted for censored data 110	
3 38	Transitions between disability states in the 1984 and 1989 National	
0.00	Long-Term Care Surveys adjusted for censored data 111	
3 39	Transitions between disability states in the 1989 and 1994 National	
0.05	Long Torm Caro Surveys, unadjusted for consored data 112	
3 40	Transitions between disability states in the 1080 and 1004 National	
0.40	Long Term Care Surveya adjusted for concerned date 112	
4 41	The 2 mean transition marked ilitical between disability states calor	
4.41	lated from the 1082 and 1084 National Lang Term Care Surrous as a	
	lated from the 1982 and 1984 National Long-Term Care Surveys as a	
4 40	percentage, for males using 5 year age groupings	
4.42	The 2-year transition probabilities between disability states calcu-	
	lated from the 1982 and 1984 National Long-Term Care Surveys as a	
	percentage, for females using 5 year age groupings	
4.43	The 2-year transition probabilities between disability states calcu-	
	lated from the 1982 and 1984 National Long-Term Care Surveys as a	
	percentage, for males and females using 5 year age groupings 118	

4.44	Summary of percentage change in disability for males and females	
	over the 1982, 1984, 1989 and 1994 NLTCS	120
4.45	The MLEs of the annual transition intensities between disability	
	states calculated from the 1982 and 1984 National Long-Term Care	
	Surveys, for males using 5 year age groupings	122
4.46	The MLEs of the annual transition intensities between disability	
	states calculated from the 1982 and 1984 National Long-Term Care	
	Surveys, for females using 5 year age groupings	123
4.47	The MLEs of the annual transition intensities between disability	
	states calculated from the 1982 and 1984 National Long-Term Care	
	Surveys, for males and females using 5 year age groupings	124
4.48	Approximate (constrained to real values, but not positive) MLEs of	
	the annual transition intensities between disability states in the 1984	
	and 1989 National Long-Term Care Surveys	127
4.49	Approximate (constrained to real but not positive values) MLEs of	
	the annual transition intensities between disability states in the 1989	
	and 1994 National Long-Term Care Surveys	128
4.50	The constrained MLEs of the annual transition intensities between	
	disability states calculated from the 1982 and 1984 National Long-	
	Term Care Surveys, males using 5 year age groupings	131
4.51	The constrained MLEs of the annual transition intensities between	
	disability states calculated from the 1982 and 1984 National Long-	
	Term Care Surveys, females using 5 year age groupings	132
4.52	The constrained MLEs of the annual transition intensities between	
	disability states calculated from the 1982 and 1984 National Long-	100
1 50	Term Care Surveys, for males and females using 5 year age groupings.	133
4.53	Comparison of log-likelihood values for the unconstrained MLEs (and	
	constrained (real) MLEs), adjusted MLEs and the constrained (posi-	
	tive) MLEs of the transition intensities calculated from the 1982-1984	195
4 5 4	The 2 mere transition much shifting realized from the construction	130
4.94	(negitive) MLEs of the transition intensition between disability states	
	(positive) MLES of the transition intensities, between disability states	
	calculated from the 1982 and 1984 National Long-Term Care Surveys	197
1 55	The 2 year transition probabilities, calculated from the constrained	194
4.00	(positive) MLEs of the transition intensition between disability states	
	(positive) MLES of the transition intensities, between disability states	
	calculated from the 1982 and 1984 National Long-Term Care Surveys	120
1 56	The 2 year transition probabilities calculated from the constrained	190
4.00	(positive) MLEs of the transition intensition between disability states	
	calculated from the 1082 and 1084 National Long Torm Caro Surrows	
	as a percentage for males and females using 5 year age groupings	120
5 57	Variance estimates of the unconstrained MI Es of the transition in	103
0.01	tensities using the asymptotic method for males and females ared	
	65-74 years in the 1082-84 NLTCS	1/1/
	05 + y = 05 + y = 05 + 000 = 000 = 000 = 000 = 0000 = 0000 = 0000 = 00000 = 00000 = 00000 = 00000 = 000000	144

5.58	Variance estimates of the unconstrained MLEs of the transition in- tensities using the information matrix for males and females aged	
	65 74 years in the 1082 84 NLTCS	144
5 50	Variance estimates of the unconstrained MLEs of the transition in-	. 144
0.05	tonsition using the asymptotic method as a percentage of these using	
	the information matrix for males and females ared 65.74 years in	
	the 1082 84 NUTCS	145
5 60	Comparison of methods for colculating the expected waiting time for	. 140
5.00	males and females aged 65.74 years in the 1082.84 NLTCS	146
5 61	Wariance estimates aged 05-74 years in the 1982-64 NLTCS.	. 140
0.01	variance estimates of the constrained (positive) MLES of the annual	
	1082 and 1084 National Long Term Care Surveys, for males using	
	1982 and 1984 National Long-Term Care Surveys, for males using	176
5 69	Variance estimates of the constrained (positive) MLEs of the appual	. 170
0.02	transition intensition between disability states calculated from the	
	1082 and 1084 National Lang Tang Can Survey for families using	
	5 year are grouping.	177
5 62	Variance estimates of the constrained (positive) MIEs of the en	. 177
5.05	variance estimates of the constrained (positive) MLES of the an-	
	the 1982 and 1984 National Long Term Caro Surveys, for malos and	
	females using 5 year age groupings	178
5.64	Mean ages of lives within each 10-year age hand by gender and in	. 110
0.01	aggregate in the 1982–1984 and 1989 NLTCS	181
5.65	Parameter values for the parametric transition intensities for males	. 101
0.00	and females grouped in 10-year age bands, calculated from the 1982	
	and 1984 NLTCS	. 182
5.66	Mean ages of lives within each 5-year age band by gender and in	
	aggregate, in the 1982, 1984 and 1989 NLTCS.	. 188
5.67	Parameter values for the parametric transition intensities for males	
	and females grouped in 5-year age bands, calculated from the 1982	
	and 1984 NLTCS	. 191
6.68	Parameter values for the logistic model fitted to data from 13 coun-	
	tries at ages 80–98 years for males and females separately over the	
	period 1980–90, from Thatcher, Kannisto & Vaupel (1998)	. 200
6.69	Model survival probabilities for lives in the healthy state at age 65	
	calculated from the $1982-84$ (10) models and the $1982-84$ (5) models,	
	for males, females and in aggregate (agg).	. 206
6.70	Model survival probabilities for lives in the healthy state at age 65	
	calculated from the $1984-89$ (10) models and the $1984-89$ (5) models,	
	for males, females and in aggregate (agg).	. 217
6.71	Model survival probabilities for lives in the healthy state at age 65	
	for the adjusted $1989-94$ (5) models and the adjusted $1989-94$ (10)	
~	models, for males, females and in aggregate (agg)	. 229
7.72	Mean, variance and skewness $(q = 1, 2 \text{ and } 3)$ of the present value	
	of disability claims costs for a life starting in each model state, unit	
	benefit increasing continuously ( $\delta_b = 0.05$ ), for males using the 1982–	0.00
	84 models	. 238

7.73 Mean, variance and skewness $(q = 1, 2 \text{ and } 3)$ of the present v disability claims costs for a life starting in each model state, un efft increasing continuously $(\delta_k = 0.05)$ for females using the 1	value of nit ben- 982–84
models.	239
7.74 Mean, variance and skewness $(q = 1, 2 \text{ and } 3)$ of the present of disability claims costs for a life starting in each model star benefit increasing continuously $(\delta_b = 0.05)$ , for males and	it value te, unit females
combined using the 1982–84 models	
males, females and combined using all disability models	241
7.76 Comparison of the EPV of disability claims costs (total LTC with the EPV of Alzheimer's disease (AD) claims costs fo starting in the healty state unit benefit increasing continuous	C  costs r a life v $(\delta_k =$
0.05), for females	242
7.77 Expected present value of disability claims costs, by time per- claiming state, for a life starting in the healty state at age ( benefit increasing continuously ( $\delta_{\rm e} = 0.05$ ) for males fema	iod and 50, unit les and
combined using the 1982-84 (5) models	· · · · · 246
7.78 EPV of disability claims costs with benefits increasing contin ( $\delta_{l} = 0.025$ ) as a percentage of EPV of disability claims cost	nuously ts_with
benefits increasing continuously ( $\delta_b = 0.05$ , baseline) for a life	e start-
ing in the healthy state for males, females and combined undisability models.	sing all 248
7.79 EPV of disability claims costs with level benefits as a percent EPV of disability claims costs with benefits increasing continues $(\delta_{1}, 0.05)$ here $\delta_{2}$ benefits at $\delta_{2}$ benefits at $\delta_{2}$ benefits at $\delta_{2}$ benefits at $\delta_{3}$ b	ntage of nuously
$(\delta_b = 0.05, \text{ baseline})$ for a life starting in the healthy state for females and combined using all disability models.	males, $\ldots 249$
7.80 EPV of disability claims costs with additional benefits of 1/2 is ing continuously ( $\delta_b = 0.05$ ) in state 3 as a percentage of 1 disability claims costs ( $\delta = 0.05$ ) with no benefits in state 3	ncreas- EPV of 6 (base-
line) for a life starting in the healthy state for males, fema	les and
combined using all disability models	250 efits in- EPV of
disability claims costs ( $\delta = 0.05$ ) with no benefits in state 3 (be for a life starting in the healthy state for males, females and co	aseline) mbined
using all disability models.	250
7.82 EPV of disability claims costs with mortality from all states a to be constant after age 90 as a percentage of EPV of disability	djusted 7 claims
costs with no mortality adjustments (baseline) for benefits incontinuously ( $\delta_{\rm e} = 0.05$ ) for a life starting in the baselither of	reasing
males, females and combined using all disability models. $\therefore$	$\ldots \ldots 252$

7.83	EPV of disability claims costs with mortality after age 90 from all	
	states adjusted to exponential (males — AM80, females and combined	
	— AF80) as a percentage of EPV of disability claims costs with no	
	mortality adjustments (baseline) for benefits increasing continuously	
	$(\delta_b = 0.05)$ for a life starting in the healthy state for males, females	
	and combined using all disability models.	. 253
7.84	EPV of benefits from Alzheimer's disease, with proportion of relative	
	risk, $m = 1.00, 0.5$ and 0.25 (from Table 2.13) as a percentage of EPV	
	of benefits from disability using the 1982–84 (5) model (from Table	
	7.75), with benefits increasing at $\delta_b = 0.05$ , for males, females and	
	combined	. 258
7.85	EPV of benefits from Alzheimer's disease, with proportion of relative	
	risk, $m = 1.00, 0.5$ and $0.25$ (from Table 2.13) as a range of percent-	
	ages of EPV of benefits from disability using all the disability models	
	(from Table 7.75), with benefits increasing at $\delta_b = 0.05$ , for males,	
	females and combined.	. 258
7.86	Costs of adverse selection as a percentage of total LTC insurance costs	
	(proportions from Table 7.85), with $\varepsilon 2/\varepsilon 4$ , $\varepsilon 3/\varepsilon 4$ and $\varepsilon 4/\varepsilon 4$ genotypes	
	k times as likely to insure as low-risk genotypes, for benefits increasing	
	continuously ( $\delta_b = 0.05$ ) and commencing on institutionalisation, for	
	males and females.	. 259
A.87	Key to classifications in each Nation Long-Term Care Survey Year.	. 282
A.88	Transitions between classifications in the 1982 and 1984 National	000
• • • •	Long-Term Care Surveys.	. 283
A.89	Transitions between classifications in the 1984 and 1989 National	000
1 00	Long-Term Care Surveys.	. 283
A.90	I ransitions between classifications in the 1989 and 1994 National	904
D 01	Transitions for formulas between disability states in the 1082 and 1084	. 204
D.91	National Long Term Care Survey using 10 year age grouping.	
	instead for consored data	286
R 02	Transitions for males between disability states in the 1982 and 1984	. 200
D.52	National Long-Term Care Surveys using 10 year age groupings ad-	
	iusted for censored data	287
B 93	Transitions for females between disability states in the 1982 and 1984	. 201
D.00	National Long-Term Care Surveys using 5 year age groupings, ad-	
	iusted for censored data.	. 288
B.94	Transitions for males between disability states in the 1982 and 1984	00
-	National Long-Term Care Surveys using 5 year age groupings, ad-	
	justed for censored data	. 289
C.95	Transitions for females between disability states in the 1984 and 1989	
	National Long-Term Care Surveys using 10 year age groupings, ad-	
	justed for censored data.	. 291
C.96	Transitions for males between disability states in the 1984 and 1989	
	National Long-Term Care Surveys using 10 year age groupings, ad-	
	justed for censored data.	. 292

C.97 Transitions for females between disability states in the 1984 and 1989	
National Long-Term Care Surveys using 5 year age groupings, ad-	
justed for censored data.	. 293
C.98 Transitions for males between disability states in the 1984 and 1989	
National Long-Term Care Surveys using 5 year age groupings, ad-	
iusted for censored data	294
D 99 Transitions for females between disability states in the 1989 and 1994	0 1
National Long-Term Care Surveys using 10 year age groupings ad-	
iusted for consored data	206
D 100 Transitions for males between disability states in the 1080 and 1004	. 250
National Long Torm Care Surveys using 10 year age groupings ad	
instead for congered date	207
Justed for censored data.	. 291
D.101 Transitions for females between disability states in the 1989 and	
1994 National Long-Term Care Surveys using 5 year age groupings,	200
adjusted for censored data.	. 298
D.102 Transitions for males between disability states in the 1989 and 1994	
National Long-Term Care Surveys using 5 year age groupings, ad-	
justed for censored data.	. 299
E.103 The 5-year transition probabilities between disability states calcu-	
lated from the 1984 and 1989 National Long-Term Care Surveys as a	
percentage, for males using 5 year age groupings	. 301
E.104 The 5-year transition probabilities between disability states calcu-	
lated from the 1984 and 1989 National Long-Term Care Surveys as a	
percentage, for females using 5 year age groupings. $\ldots \ldots \ldots \ldots$	. 302
E.105 The 5-year transition probabilities between disability states calcu-	
lated from the 1984 and 1989 National Long-Term Care Surveys as a	
percentage, for males and females using 5 year age groupings. $\ldots$	. 303
E.106 The 5-year transition probabilities between disability states calcu-	
lated from the 1989 and 1994 National Long-Term Care Surveys as a	
percentage, for males using 5 year age groupings	. 304
E.107 The 5-year transition probabilities between disability states calcu-	
lated from the 1989 and 1994 National Long-Term Care Surveys as a	
percentage, for females using 5 year age groupings.	. 305
E.108 The 5-year transition probabilities between disability states calcu-	
lated from the 1989 and 1994 National Long-Term Care Surveys as a	
percentage, for males and females using 5 year age groupings.	. 306
F.109 The MLEs of the annual transition intensities between disability	
states calculated from the 1984 and 1989 National Long-Term Care	
Surveys, for males using 5 year age groupings.	. 308
F.110 The MLEs of the annual transition intensities between disability	
states calculated from the 1984 and 1989 National Long-Term Care	
Surveys, for females using 5 year age groupings	309
F 111 The MLEs of the annual transition intensities between disability	
states calculated from the 1984 and 1989 National Long-Term Care	
Surveys for males and females using 5 year age groupings	310
Surveys, for mates and females using 6 year age groupings.	. 010

F.112 The MLEs of the annual transition intensities between disa	bility
states calculated from the 1989 and 1994 National Long-Term	Care
Surveys, for males using 5 year age groupings	311
F.113 The MLEs of the annual transition intensities between disa	bility
states calculated from the 1989 and 1994 National Long-Term	Care
Surveys, for females using 5 year age groupings	312
F.114 The MLEs of the annual transition intensities between disa	bility
states calculated from the 1989 and 1994 National Long-Term	Care
Surveys, for males and females using 5 year age groupings	313
G.115 The constrained (positive) MLEs of the annual transition inten	sities
between disability states calculated from the 1982 and 1984 Nat	lonal
Long-Term Care Surveys, for males and females using 10 yea	r age
C 116 The constrained MI Eq of the appual transition intensition bet	· · · · 310
disability states calculated from the 1084 and 1080 National I	Jween
Term Care Surveys for males and females using 10 year age gro	Jong- Junings 316
G 117 The constrained MLEs of the annual transition intensities bet	ween
disability states calculated from the 1989 and 1994 National I	Long-
Term Care Surveys, for males and females using 10 year age gro	oupings.317
G.118 The constrained (positive) MLEs of the annual transition inten	sities
between disability states calculated from the 1982 and 1984 Nat	tional
Long-Term Care Surveys, for females using 10 year age groupin	ıgs 318
G.119 The constrained MLEs of the annual transition intensities bet	ween
disability states calculated from the 1984 and 1989 National I	Long-
Term Care Surveys, for females using 10 year age groupings	319
G.120 The constrained MLEs of the annual transition intensities bet	Jween
disability states calculated from the 1989 and 1994 National I	Long-
Term Care Surveys, for females using 10 year age groupings.	320
G.121 The constrained (positive) MLEs of the annual transition inten	ISITIES
Long Term Care Surveys, for males using 10 year age grouping	
C = 122 The constrained MLEs of the annual transition intensities bet	5 021
disability states calculated from the 1984 and 1989 National I	Long-
Term Care Surveys, for males using 10 year age groupings.	
G.123 The constrained MLEs of the annual transition intensities bet	ween
disability states calculated from the 1989 and 1994 National I	Long-
Term Care Surveys, for males using 10 year age groupings	323
G.124 The constrained MLEs of the annual transition intensities bet	ween
disability states calculated from the 1984 and 1989 National I	Long-
Term Care Surveys, for males and females using 5 year age group	upings. 324
G.125 The constrained MLEs of the annual transition intensities bet	Jween
disability states calculated from the 1989 and 1994 National I	Long-
Term Care Surveys, for males and females using 5 year age group of the females and females using 5 year age group of the females and females using 5 year age group of the females and females are supported at the females at the females are supported at the females at the fema	upings. 325
G.126 The constrained MLEs of the annual transition intensities bet	Jween
disability states calculated from the 1984 and 1989 National I	Long-
Term Care Surveys, for females using 5 year age groupings	326

XV

G.127 The constrained MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-	
Term Care Surveys, for females using year age groupings,	27
G.128 The constrained MLEs of the annual transition intensities between	
disability states calculated from the 1984 and 1989 National Long-	
Term Care Surveys, for males using 5 year age groupings	28
G.129 The constrained MLEs of the annual transition intensities between	
disability states calculated from the 1989 and 1994 National Long-	
Term Care Surveys, for males using 5 year age groupings	29
H.130 Comparison of log-likelihood values for the unconstrained MLEs	
(and constrained (real) MLEs), adjusted MLEs and the constrained	
(positive) MLEs of the transition intensities calculated from the 1984-	
1989 NLTCS	31
H.131 Comparison of log-likelihood values for the unconstrained MLEs	
(and constrained (real) MLEs), adjusted MLEs and the constrained	
(positive) MLEs of the transition intensities calculated from the 1989-	
1994 NLTCS	32
I.132 The 5-year transition probabilities, calculated from the constrained	
(positive) MLEs of the transition intensities, between disability states	
calculated from the 1984 and 1989 National Long-Term Care Surveys	
as a percentage, for males using 5 year age groupings	34
1.133 The 5-year transition probabilities, calculated from the constrained	
(positive) MLEs of the transition intensities, between disability states	
calculated from the 1984 and 1989 National Long-Term Care Surveys	75
as a percentage, for females using 5 year age groupings	35
1.134 The 5-year transition probabilities, calculated from the constrained	
(positive) MLEs of the transition intensities, between disability states	
calculated from the 1984 and 1989 National Long-Term Care Surveys	26
L 125 The 5 year transition probabilities, calculated from the constrained	50
(positive) MI Es of the transition intensition between disability states	
(positive) will so in the transition intensities, between disability states	
as a percentage for males using 5 year age groupings	27
136 The 5-year transition probabilities calculated from the constrained	,,
(positive) MLEs of the transition intensities, between disability states	
calculated from the 1989 and 1994 National Long-Term Care Surveys	
as a percentage, for females using 5 year age groupings,	38
I.137 The 5-year transition probabilities, calculated from the constrained	
(positive) MLEs of the transition intensities, between disability states	
calculated from the 1989 and 1994 National Long-Term Care Surveys	
as a percentage, for males and females using 5 year age groupings 33	39
J.138 Variance estimates of the MLEs of the annual transition intensities	
between disability states calculated from the 1982 and 1984 National	
Long-Term Care Surveys, for males and females using 10 year age	
groupings	11

J.139 Variance estimates of the constrained (positive) MLEs of the an-	
nual transition intensities between disability states calculated from	
the 1984 and 1989 National Long-Term Care Surveys, for males and	
females using 10 year age groupings	2
J.140 Variance estimates of the constrained (positive) MLEs of the an-	
nual transition intensities between disability states calculated from	
the 1989 and 1994 National Long-Term Care Surveys, for males and	
females using 10 year age groupings.	3
J.141 Variance estimates of the MLEs of the annual transition intensities	
between disability states calculated from the 1982 and 1984 National	
Long-Term Care Surveys, for females using 10 year age groupings,, 344	Ł
J 142 Variance estimates of the constrained (positive) MLEs of the an-	-
nual transition intensities between disability states calculated from	
the 1984 and 1989 National Long-Term Care Surveys for females	
using 10 year age groupings	Ś
. I 143 Variance estimates of the constrained (positive) MLEs of the an-	
nual transition intensities between disability states calculated from	
the 1989 and 1994 National Long-Term Care Surveys for females	
using 10 year age groupings	š
I 144 Variance estimates of the MLEs of the annual transition intensities	,
between disability states calculated from the 1982 and 1984 National	
Long-Term Care Surveys, for males using 10 year age groupings 347	7
145 Variance estimates of the constrained (positive) MLEs of the an-	
nual transition intensities between disability states calculated from	
the 1084 and 1080 National Long Term Care Surveys, for males us	
ing 10 year age groupings	2
146 Variance estimates of the constrained (positive) MIEs of the an	,
nual transition intensition between disability states calculated from	
the 1080 and 1004 National Long Term Care Surveys, for males us	
ing 10 year ago groupings	)
1 147 Variance estimates of the constrained (positive) MIEs of the an	'
nual transition intensition between disability states calculated from	
the 1084 and 1080 National Long Term Care Surveys, for males and	
formalos using 5 year ago groupings	١
148 Variance estimates of the constrained (positive) MIEs of the an	,
nual transition intensition between disability states calculated from	
the 1080 and 1004 National Long Term Care Surveys, for males and	
females using 5 year ago groupings	
140 Variance estimates of the constrained (negitive) MIEs of the en	•
5.149 variance estimates of the constrained (positive) MLEs of the an-	
the 1084 and 1080 National Lang Term Care Surgeon for forcela	
the 1984 and 1989 National Long-Term Care Surveys, for females	,
using 5 year age groupings	1
J.150 variance estimates of the constrained (positive) MLEs of the an-	
the 1000 and 1004 National Lang The Constant for finite	
the 1989 and 1994 National Long-Term Care Surveys, for females	,
using 5 year age groupings	)

J.151 Variance estimates of the constrained (positive) MLEs of the an- nual transition intensities between disability states calculated from	
the 1984 and 1989 National Long-Term Care Surveys, for males us-	
ing 5 year age groupings	. 354
J.152 Variance estimates of the constrained (positive) MLEs of the an-	
nual transition intensities between disability states calculated from	
the 1989 and 1994 National Long-Term Care Surveys, for males us-	
ing 5 year age groupings	. 355
K.153 Parameter values for the parametric transition intensities for males,	
calculated from the 1982 and 1984 NLTCS, grouped in 10-year age	
bands	. 357
K.154 Parameter values for the parametric transition intensities for fe-	
males, calculated from the 1982 and 1984 NLTCS, grouped in 10-year	
age bands.	. 358
K.155 Parameter values for the parametric transition intensities for males,	
calculated from the 1984 and 1989 NLTCS, grouped in 10-year age	050
bands	. 359
K.156 Parameter values for the parametric transition intensities for fe-	
males, calculated from the 1984 and 1989 NLTCS, grouped in 10-year	260
Age Dallas	. 300
and females, calculated from the 1084 and 1080 NLTCS, grouped in	
10 year ago bands	361
K 158 Parameter values for the parametric transition intensities for males	. 501
calculated from the 1989 and 1994 NLTCS grouped in 10-year age	
hands	362
K 159 Parameter values for the parametric transition intensities for fe-	. 002
males, calculated from the 1989 and 1994 NLTCS, grouped in 10-year	
age bands	. 363
K.160 Parameter values for the parametric transition intensities for males	
and females, calculated from the 1989 and 1994 NLTCS, grouped in	
10-year age bands.	. 364
L.161 Parameter values for the parametric transition intensities for males,	
calculated from the 1982 and 1984 NLTCS, grouped in 5-year age	
bands	. 366
L.162 Parameter values for the parametric transition intensities for fe-	
males, calculated from the $1982$ and $1984$ NLTCS, grouped in 5-year	
age bands	. 367
L.163 Parameter values for the parametric transition intensities for males,	
calculated from the 1984 and 1989 NLTCS, grouped in 5-year age	
bands	. 368
L.164 Parameter values for the parametric transition intensities for fe-	
males, calculated from the 1984 and 1989 NLTCS, grouped in 5-year	0.00
age bands	. 369
and formales, coloulated from the 1084 and 1080 MLTCC, manual in	
5 year ago hands	370
$\sigma$ -year age ballus. $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	. 570

L.166 Parameter values for the parametric transition intensities for males,	
calculated from the 1989 and 1994 NLTCS, grouped in 5-year age	
bands	. 371
L.167 Parameter values for the parametric transition intensities for fe-	
males, calculated from the $1989$ and $1994$ NLTCS, grouped in 5-year	
age bands	. 372
L.168 Parameter values for the parametric transition intensities for males	
and females, calculated from the 1989 and 1994 NLTCS, grouped in	
5-year age bands	. 373

# List of Figures

1.1	A simple model of Alzheimer's disease in the $i$ th of $M$ subgroups,	
	each representing a different ApoE genotype. $x$ is the age at outset,	
	and $t$ the elapsed duration	13
1.2	Aggregate Incidence of Alzheimer's Disease: Point Estimates and 95%	
	Confidence Intervals. Source: Jorm & Jolley (1998)	19
1.3	Incidence of Alzheimer's Disease by Gender: Point Estimates from	
	Rocca et al. (1998). $\ldots$	20
1.4	Odds ratios (ORs) of AD relative to $\varepsilon_3/\varepsilon_3$ genotype for males and	
	females combined. Source: Farrer <i>et al.</i> (1997).	33
1.5	Odds ratios (ORs) of AD relative to $\varepsilon_3/\varepsilon_3$ genotype for $\varepsilon_3/\varepsilon_4$ and	
	$\varepsilon 4/\varepsilon 4$ genotypes. Source: Farrer <i>et al.</i> (1997).	34
1.6	Odds ratios (ORs) of AD relative to $\varepsilon_3/\varepsilon_3$ genotype for $\varepsilon_2/\varepsilon_2$ or	01
1.0	$\varepsilon^2/\varepsilon^3$ and $\varepsilon^2/\varepsilon^4$ genotypes Source: Farrer <i>et al</i> (1997)	34
17	Odds ratios (OBs) of AD relative to $\varepsilon^3/\varepsilon^3$ genotype from Farrer <i>et</i>	01
	al (1997) compared with OBs computed using modelled relative risk	
	functions	37
1.8	Modelled risk of AD relative to the $\varepsilon^3/\varepsilon^3$ genotype for $\varepsilon^4/\varepsilon^4$ and	0.
1.0	$\varepsilon_3/\varepsilon_4$ genotypes. Based on odds ratios from Farrer <i>et al.</i> (1997).	38
1.9	Modelled risk of AD, relative to the $\varepsilon_3/\varepsilon_3$ genotype, for $\varepsilon_2/\varepsilon_4$ and	00
1.0	$\varepsilon 2/\varepsilon 2 \& \varepsilon 2/\varepsilon 3$ genotypes. Based on odds ratios from Farrer <i>et al.</i>	
	(1997)	39
1.10	Comparison of the estimated population incidence of AD $\mu^{AD}_{AU}$ with	
	the aggregated incidence of AD for different levels of relative risk.	
	males and females combined.	40
1.11	Occupancy probabilities for females, healthy at age 60, high relative	-
	risks $(m = 1)$ .	43
1.12	Occupancy probabilities for females, healthy at age 60, low relative	
	risks $(m = 0.25)$ .	44
1.13	Occupancy probabilities for males, healthy at age 60, high relative	
	risks $(m = 1)$ .	45
1.14	Occupancy probabilities for males, healthy at age 60, low relative	
	risks $(m = 0.25)$ .	46
1.15	Prevalence rate of Alzheimer's disease for females healthy at age 60,	
	high relative risks $(m = 1)$	47
1.16	Prevalence rate of Alzheimer's disease for females healthy at age 60,	
	low relative risks $(m = 0.25)$	48

1.17	Prevalence rate of Alzheimer's disease for males healthy at age 60, high relative ricks $(m - 1)$	40
1.18	Prevalence rate of Alzheimer's disease for males healthy at age 60,	. 49
	low relative risks $(m = 0.25)$	. 50
3.19	A model of disability for the lifetime of an individual and long term	05
3.20	An overview of the 1982 1984 1989 and 1994 U.S. National Long-	. 95
5.20	Term Care Surveys.	. 100
5.21	Variance estimates of the transition intensities out of the 'Healthy'	
	state using the asymptotic method $(\bullet)$ , the information matrix $(\times)$	
	and the 'boot-strap' variance estimate (horizontal line) for 500 simu-	
	lations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females ared 65, 74 years in	
	the 1982–84 NLTCS.	. 149
5.22	Variance estimates of the transition intensities out of the 'IADL only'	. 110
	state using the asymptotic method $(\bullet)$ , the information matrix $(\times)$	
	and the 'boot-strap' variance estimate (horizontal line) for 500 simu-	
	lations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females ared 65, 74 years in	
	the 1982–84 NLTCS.	. 150
5.23	Variance estimates of the transition intensities out of the '1–2 ADLs'	
	state using the asymptotic method $(\bullet)$ , the information matrix $(\times)$	
	and the 'boot-strap' variance estimate (horizontal line) for 500 simu-	
	lations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females ared 65–74 years in	
	the 1982–84 NLTCS.	. 151
5.24	Variance estimates of the transition intensities out of the '3–4 ADLs'	-
	state using the asymptotic method $(\bullet)$ , the information matrix $(\times)$	
	and the 'boot-strap' variance estimate (horizontal line) for 500 simu-	
	lations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females ared 65, 74 years in	
	the 1982–84 NLTCS.	. 152
5.25	Variance estimates of the transition intensities out of the '5–6 ADLs'	-
	state using the asymptotic method $(\bullet)$ , the information matrix $(\times)$	
	and the 'boot-strap' variance estimate (horizontal line) for 500 sim-	
	ulations, based on the constrained (positive) MLEs of transition in- tonsitios (vortical line) for malos and fomalos agod 65–74 years in the	
	1982–84 NLTCS.	. 153
5.26	Variance estimates of the transition intensities out of the 'Institution-	
	alized' state using the asymptotic method $(\bullet)$ , the information matrix	
	$(\times)$ and the 'boot-strap' variance estimate (horizontal line) for 500	
	simulations, based on the constrained (positive) MLEs of transition intensities (vertical line) for males and females ared 65–74 years in	
	the 1982–84 NLTCS.	. 154
5.27	A simple 2-state model with one transition intensity.	. 155
5.28	Variance ratio, $R_{\hat{\mu}^{12}}(k)$ , for $0 \le k \le 1$ for the 2-state model with one	
	transition intensity	. 157

5.29	Expected variance ratio, $E[R'_{\hat{\mu}^{12}}(\mu^{12})]$ , for $0 \le \mu^{12} \le 1$ for the 2-state	150
F 20	model with one transition intensity.	. 159
0.3U 5.21	A simple 2-state model with two transition intensities	. 159
0.51	Variance ratios, $R_{\hat{\mu}^{ij}}(K)$ ( $0 \le K \le 10$ ) for $\mu^{-1}$ and $\mu^{-1}$ when $n_{01}^{-1} = \sum_{i=1}^{i} \frac{i}{2} \sum_{j=1}^{i} \frac{i}{2} \sum_{j=1}^$	1.01
r 90	$E[n_{01}]$ in the 2-state model with two transition intensities	. 101
5.32	variance estimates of the transition intensities for the 2-state model	
	with two transition intensities using the asymptotic method $(\bullet)$ , the information metric $(\chi)$ and the (heat streng) regions estimate (here	
	information matrix (x) and the boot-strap variance estimate (nor- izontal line) for 1,000 gipulations, with $\mu^{12} = 0.1$ , $\mu^{21} = 0.05$	
	$N^1 = 2,000 \text{ and } N^2 = 1,000 (K = 0.5)$	169
5 33	N = 2,000 and $N = 1,000$ ( $N = 0.5$ )	163
5.30	Variance ratios $R_{\text{scie}}(K)$ ( $0 \le K \le 10$ ) for $\mu^{12}$ $\mu^{13}$ and $\mu^{23}$ when	. 100
0.04	variance ratios, $R_{\mu}^{ij}(R)$ ( $0 \le R \le 10$ ) for $\mu$ , $\mu$ and $\mu$ when $n^{ij} = \mathbf{E}[n^{ij}]$ in the 3 state model	165
5 35	$N_{01} - E[n_{01}]$ in the 5-state model	. 100
0.00	using the asymptotic method $(\bullet)$ the information matrix $(\times)$ and the	
	'boot-strap' variance estimate (horizontal line) for 1 000 simulations	
	with $\mu^{12} = 0.1$ , $\mu^{13} = 0.025$ , $\mu^{23} = 0.05$ , $N^1 = 5,000$ and $N^2 = 2,500$	
	(K = 0.5)	. 166
5.36	Convergence of the 'boot-strap' variance estimates for the constrained	
	(positive) MLEs of the transition intensities out of the 'Healthy' state,	
	showing the maximum ( $\bullet$ ) and minimum ( $\times$ ) 'boot-strap' variance	
	estimate of 100 samples, for samples of size $n$ ( $n = 50, 100, 500,$	
	1,000, 2,000 and $5,000$ ) and where the horizontal line is the variance	
	estimate using an adjusted maximum likelihood approach, for males	
	and females aged 65–74 years in the 1982–84 NLTCS	. 169
5.37	Convergence of the 'boot-strap' variance estimates for the constrained	
	(positive) MLEs of the transition intensities out of the 'IADL only'	
	state, showing the maximum ( $\bullet$ ) and minimum ( $\times$ ) 'boot-strap' vari-	
	ance estimate of 100 samples, for samples of size $n$ ( $n = 50, 100, 500, 100, 2000$ and 5 000) and where the herizontal line is the variance	
	estimate using an adjusted maximum likelihood approach for males	
	and females aged 65–74 years in the 1982–84 NLTCS	170
5.38	Convergence of the 'boot-strap' variance estimates for the constrained	. 110
	(positive) MLEs of the transition intensities out of the '1–2 ADLs'	
	state, showing the maximum $(\bullet)$ and minimum $(\times)$ 'boot-strap' vari-	
	ance estimate of 100 samples, for samples of size $n$ ( $n = 50, 100, 500,$	
	1,000, 2,000 and $5,000$ ) and where the horizontal line is the variance	
	estimate using an adjusted maximum likelihood approach, for males	
	and females aged 65–74 years in the 1982–84 NLTCS	. 171
5.39	Convergence of the 'boot-strap' variance estimates for the constrained	
	(positive) MLEs of the transition intensities out of the '3–4 ADLs'	
	state, showing the maximum ( $\bullet$ ) and minimum ( $\times$ ) 'boot-strap' vari-	
	ance estimate of 100 samples, for samples of size $n$ ( $n = 50, 100, 500, 100, 2000, and 5000$ ) and submit the last stability is stability in the stability of	
	1,000, 2,000 and 5,000) and where the norizontal line is the variance	
	and formalos agod 65. 74 years in the 1082 84 NUTCS	179
	and remains aged $00-14$ years in the 1902-04 NLTOD	. 112

- 5.41 Convergence of the 'boot-strap' variance estimates for the constrained (positive) MLEs of the transition intensities out of the 'Institutionalized' state, showing the maximum (•) and minimum (×) 'boot-strap' variance estimate of 100 samples, for samples of size n (n = 50, 100, 500, 1,000, 2,000 and 5,000) and where the horizontal line is the variance estimate using an adjusted maximum likelihood approach, for males and females aged 65–74 years in the 1982–84 NLTCS. . . . . . . 174

5.43	Graphs of point estimates (CMLE), approximate 95% confidence in-	
	tervals and parametric fits of the transition intensities out of the	
	'IADL only' state for males and females grouped in 10-year age bands	
	in the 1982–84 NLTCS	. 185

5.44 Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.

5.45 Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.

5.46 Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.
186

5.49 Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males and females grouped in 5-year age bands in the 1982–84 NLTCS.

5.50	Graphs of point estimates (CMLE), approximate 95% confidence in- tervals and parametric fits of the transition intensities out of the	
	'IADL only' state for males and females grouped in 5-year age bands	
	in the 1982–84 NLTCS	193
5.51	Graphs of point estimates (CMLE), approximate 95% confidence in-	. 100
0.01	tervals and parametric fits of the transition intensities out of the ' $1-2$	
	ADLs' state for males and females grouped in 5-year age bands in the	
	1982–84 NLTCS	. 194
5.52	Graphs of point estimates (CMLE), approximate 95% confidence in-	
	tervals and parametric fits of the transition intensities out of the '3–4	
	ADLs' state for males and females grouped in 5-year age bands in the	
	1982–84 NLTCS.	. 194
5.53	Graphs of point estimates (CMLE), approximate 95% confidence in-	
	tervals and parametric fits of the transition intensities out of the '5–6 $$	
	ADLs' state for males and females grouped in 5-year age bands in the	
	1982–84 NLTCS	. 195
5.54	Graphs of point estimates (CMLE), approximate 95% confidence in-	
	tervals and parametric fits of the transition intensities out of the 'In-	
	stitutionalized' state for males and females grouped in 5-year age	
0 55	bands in the 1982–84 NLTCS. $\dots$ 1002 of (5)	. 195
6.55	Overall model mortality for the $1982-84$ (5) models and the $1982-84$	
	(10) models compared with a logistic model of mortality by Thatcher, $V_{\text{compised}} = V_{\text{compared}} (1008)$	901
6 56	Model accuracy probabilities conditional on being alive for the	. 201
0.00	Model occupancy probabilities, conditional on being anye, for the $1082, 84, (5)$ models (solid line) and the $1082, 84, (10)$ models (dotted	
	lino) for malos	203
6 57	The parametric force of mortality for the $1982-84$ (5) NLTCS models	. 203
0.01	(solid line) and 1982–84 (10) NLTCS models (dotted line) for males	203
6.58	Model occupancy probabilities conditional on being alive for the	200
0.00	1982–84 (5) models (solid line) and the 1982–84 (10) models (dotted	
	line), for females.	. 205
6.59	The parametric force of mortality for the 1982–84 (5) models (solid	
	line) and the 1982–84 (10) models (dotted line), for females	205
6.60	Model occupancy probabilities, conditional on being alive, for the	
	1982–84 (5) models (solid line) and the 1982–84 (10) models (dotted	
	line), for males and females combined.	. 207
6.61	The parametric force of mortality for the 1982–84 (5) models and the	
	1982-84 (10) models (dotted line), for males and females combined.	. 207
6.62	Overall model mortality for the 1984–89 (5) models and the 1984–89	
	(10) models compared with a logistic model of mortality by Thatcher,	200
0.00	Kannisto & Vaupel (1998). $\ldots$	. 209
6.63	Model occupancy probabilities, conditional on being alive, for the	
	1904–09 (3) models (solid line) and the 1984–89 (10) models (dotted line) for malor	910
6 64	The peremetric force of mortality for the 1084 80 (5) models (solid	. 210
0.04	line) and the $1084-80$ (10) models (dotted line) for males	911
	mic) and the 1304 03 (10) models (dotted life), for males	. 411

6.65	Model occupancy probabilities, conditional on being alive, for the $1984-89$ (5) models (solid line) and the $1984-89$ (10) models (dotted	
	line), for females.	212
6.66	The parametric force of mortality for the 1984–89 (5) models (solid	
	line) and the 1984–89 (10) models (dotted line), for females	213
6.67	Model occupancy probabilities, conditional on being alive, for the	
	1984-89 (5) models (solid line) and the $1984-89$ (10) models (dotted	
	line), for males and females combined.	213
6.68	The parametric force of mortality for the 1984–89 (5) models (solid	
	line) and the 1984–89 (10) models (dotted line), for males and females	914
6 60	Overall model mortality for the adjusted 1084 80 (5) models and	214
0.09	the adjusted 1984–89 (10) models compared with a logistic model of	
	mortality by Thatcher, Kannisto & Vaupel (1998).	215
6.70	Model occupancy probabilities, conditional on being alive, for the	
	adjusted 1984–89 (5) models (solid line) and the adjusted 1984–89	
	(10) models (dotted line), for males. $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	216
6.71	Model occupancy probabilities, conditional on being alive, for the	
	adjusted 1984–89 (5) models (solid line) and the adjusted 1984–89	
	(10) models (dotted line), for females	217
6.72	Model occupancy probabilities, conditional on being alive, for the	
	adjusted 1984–89 (5) models (solid line) and the adjusted 1984–89 (10) models (detted line) for models and females combined	910
6 73	(10) models (dotted line), for males and remains combined. $\ldots$ $\ldots$	219
0.75	(10) models compared with a logistic model of mortality by Thatcher	
	Kannisto & Vaupel (1998).	220
6.74	Model occupancy probabilities, conditional on being alive, for the	0
	1989–94 (5) models (solid line) and the 1989–94 (10) models (dotted	
	line), for males.	221
6.75	The parametric force of mortality for the 1989–94 (5) models (solid	
	line) and the 1989–94 (10) models (dotted line), for males. $\ldots$ .	221
6.76	Model occupancy probabilities, conditional on being alive, for the	
	1989-94 (5) models (solid line) and the $1989-94$ (10) models (dotted	000
6 77	The peremetric force of montality for the 1080 04 (5) models (colid	222
0.77	line) and $1989-94$ (10) models (dotted line) for females	222
6.78	Model occupancy probabilities conditional on being alive for the	
0.10	1989–94 (5) models (solid line) and the 1989–94 (10) models (dotted	
	line), for males and females combined.	224
6.79	The parametric force of mortality for the 1989–94 (5) models (solid	
	line) and the 1989–94 $(10)$ models (dotted line), for males and females	
	combined	224
6.80	Overall model mortality for the adjusted 1989–94 (5) models and	
	the adjusted 1989–94 (10) models compared with a logistic model of	005
	mortality by Thatcher, Kannisto & Vaupel (1998)	225

6.81 Model occupancy probabilities, conditional on being alive, for the adjusted 1989–94 (5) models (solid line) and the adjusted 1989–94	
(5) models (dotted line) for males	226
6.82 Model occupancy probabilities conditional on being alive for the	. 220
adjusted 1989–94 (5) models (solid line) and the adjusted 1989–94	
(10) models (dotted line) for females	<b>997</b>
6.83 Model occupancy probabilities conditional on being alive for the	. 221
adjusted 1080–04 (5) models (solid line) and the adjusted 1080–04	
(10) models (dotted line) for males and females combined	228
6.84 Overall mortality from the 1082–84 (10) the adjusted 1084–80 (10)	. 220
and the adjusted $1989-94$ (10) models compared with a logistic model	
of mortality by Thatcher Kannisto, & Vaunal (1008)	921
6.85 Overall mortality from the 1082 84 (5) the adjusted 1084 80 (5) and	. 201
the adjusted 1080 $04$ (5) models compared with a logistic model of	
mortality by Thatcher Kannisto & Vaupal (1008)	921
7.86 Comparison of the overall force of mortality in the disability models	. 201
to the overall force of mortality in the Alzheimer's disease model for	
fomalos	244
7.87 Comparison of the overall force of mortality in the disability models	. 444
to the overall force of mortality in the Alzheimer's disease model for	
males	255
7.88 Comparison of the overall force of mortality in the disability models	. 200
to the overall force of mortality in the Alzheimer's disease model for	
males and females combined	256
M 89Graphs of point estimates (CMLE) approximate 95% confidence in-	. 200
tervals and parametric fits of the transition intensities out of the	
'Healthy' state for males grouped in 5-year age bands in the 1982–84	
NLTCS	375
M 90 Graphs of point estimates (CMLE) approximate 95% confidence in-	. 010
tervals and parametric fits of the transition intensities out of the	
'IADL only' state for males grouped in 5-year age bands in the 1982–	
84 NLTCS.	. 375
M.91 Graphs of point estimates (CMLE), approximate 95% confidence in-	
tervals and parametric fits of the transition intensities out of the '1-2	
ADLs' state for males grouped in 5-year age bands in the 1982–84	
NLTCS.	. 376
M.92 Graphs of point estimates (CMLE), approximate 95% confidence in-	
tervals and parametric fits of the transition intensities out of the '3–4	
ADLs' state for males grouped in 5-year age bands in the 1982–84	
NLTCS.	. 376
M.93 Graphs of point estimates (CMLE), approximate 95% confidence in-	
tervals and parametric fits of the transition intensities out of the '5–6	
ADLs' state for males grouped in 5-year age bands in the 1982–84	
NLTCS	. 377

M.94Graphs of point estimates (CMLE), approximate $95\%$ confidence intervals and parametric fits of the transition intensities out of the 'In-	
stitutionalized' state for males grouped in 5-year age bands in the 1982–84 NLTCS	377
M.95 Graphs of point estimates (CMLE), approximate 95% confidence in-	
tervals and parametric fits of the transition intensities out of the 'Healthy' state for females grouped in 5-year age bands in the 1982–84	
NLTCS	. 378
M.96 Graphs of point estimates (CMLE), approximate 95% confidence in- tervals and parametric fits of the transition intensities out of the 'IADL only' state for females grouped in 5-year age bands in the	
1982–84 NLTCS	. 378
M.97 Graphs of point estimates (CMLE), approximate 95% confidence in- tervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for females grouped in 5-year age bands in the 1982–84	0.50
NLTCS. $\dots$ (CMLE) $\dots$ (CMLE)	. 379
M.98 Graphs of point estimates (CMLE), approximate 95% confidence in- tervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for females grouped in 5-year age bands in the 1982–84	
NLTCS	. 379
M.99 Graphs of point estimates (CMLE), approximate 95% confidence in- tervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for females grouped in 5-year age bands in the 1982–84	
NLTCS.	. 380
M.100Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for females grouped in 5-year age bands in	
the 1982–84 NLTCS	. 380
N.101 Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males grouped in 5-year age bands in the 1984–89	200
NL105	. 302
intervals and parametric fits of the transition intensities out of the	
'IADL only' state for males grouped in 5-year age bands in the 1984– 89 NLTCS	389
N 103 Graphs of point estimates (CMLE) approximate 95% confidence	. 002
intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males grouped in 5-year age bands in the 1984–	202
N 104 Graphs of point estimates (CMLE) approximate 95% confidence	00
intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males grouped in 5-year age bands in the 1984–	0.00
89 NLTCS	. 383

Ν	.105 Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the	
	(5-6 ADLs' state for males grouped in 5 year age bands in the 1084-	
	80 NLTCS	38/
N	106  Graphs of point estimates (CMLE) approximate $95%$ confidence	. 504
11	intervals and parametric fits of the transition intersities out of the	
	'Institutionalized' state for males grouped in 10 year age bands in	
	the 1084 80 NLTCS	38/
N	107 Craphs of point estimates (CMLF) approximate $05%$ confidence	. 504
IN	intervals and parametric fits of the transition intersition out of the	
	(Healthy' state for famalog grouped in 5 year age bands in the 1084	
	on NUTCO	<b>20</b> 5
N	108 Craphs of point estimates (CMLE) approximate $05%$ confidence	. 305
IN	intervals and parametric fits of the transition intervition out of the	
	(IADI only' state for formalis grouped in 5 year age bands in the	
	1004 on NUTCO	<b>20</b> 5
N	1964-69 NLICS	. 300
IN	intervals and parametric fits of the transition intervals out of the '1	
	2  ADL s' state for formalize grouped in 5 year are bands in the 1084 80	
	2 ADES state for remales grouped in 5-year age bands in the 1964–69 NUTCS	286
N	110 Craphs of point estimates (CMLF) approximate $05\%$ confidence	. 560
IN	intervals and parametric fits of the transition intensities out of the '3	
	4  ADL s state for formalog groupod in 5 year ago bands in the 1084–80	
	ADDS state for remains grouped in 5-year age bands in the 1964-69 NUTCS	386
N	111 Graphs of point estimates (CMLE) approximate 95% confidence	. 500
11	intervals and parametric fits of the transition intensities out of the '5-	
	6 ADLs' state for formales grouped in 5 year ago bands in the 1984–80	
	NLTCS	387
N	112 Graphs of point estimates (CMLE) approximate $95\%$ confidence	
11	intervals and parametric fits of the transition intensities out of the	
	'Institutionalized' state for females grouped in 5-year age hands in	
	the 1984–89 NLTCS	387
Ν	113 Graphs of point estimates (CMLE) approximate 95% confidence	
1,	intervals and parametric fits of the transition intensities out of the	
	'Healthy' state for males and females grouped in 5-year age bands in	
	the 1984–89 NLTCS	388
Ν	114 Graphs of point estimates (CMLE) approximate 95% confidence	
1,	intervals and parametric fits of the transition intensities out of the	
	'IADL only' state for males and females grouped in 5-year age bands	
	in the 1984–89 NLTCS.	. 388
Ν	.115 Graphs of point estimates (CMLE), approximate 95% confidence	
± ·	intervals and parametric fits of the transition intensities out of the	
	'1–2 ADLs' state for males and females grouped in 5-year age bands	
	in the 1984–89 NLTCS	. 389

N.116 Graphs of point estimates (CMLE), approximate $95\%$ confidence	
intervals and parametric fits of the transition intensities out of the	
'3–4 ADLs' state for males and females grouped in 5-year age bands	
in the 1984–89 NLTCS	. 389
N.117 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'5–6 ADLs' state for males and females grouped in 5-year age bands	
in the 1984–89 NLTCS	. 390
N.118 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'Institutionalized' state for males and females grouped in 5-year age	
bands in the 1984–89 NLTCS	. 390
O.119 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'Healthy' state for males grouped in 5-year age bands in the 1989–94	
NLTCS.	. 392
O.120 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'IADL only' state for males grouped in 5-year age bands in the 1989–	
94 NLTCS.	. 392
O.121 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'1–2 ADLs' state for males grouped in 5-year age bands in the 1989–	
94 NLTCS.	. 393
O.122 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'3–4 ADLs' state for males grouped in 5-year age bands in the 1989–	
94 NLTCS.	. 393
O.123 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'5–6 ADLs' state for males grouped in 5-year age bands in the 1989–	20.4
94 NLTCS. $(O, H, F)$	. 394
O.124 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'Institutionalized' state for males grouped in 5-year age bands in the	204
$1989-94 \text{ NLTCS} \dots \dots$	. 394
0.125 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'Healthy' state for females grouped in 5-year age bands in the 1989–	205
94 NLTCS. $(C)$ (C) (II I) $(C)$ (C) (II I) $(C)$	. 395
U.126 Graphs of point estimates (UMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
TADL only state for females grouped in 5-year age bands in the	0.0 <i>F</i>
1989–94 NLTCS	. 395

O.127 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the '1–	
2 ADLs' state for females grouped in 5-year age bands in the 1989–94	
NLTCS.	. 396
O.128 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the '3–	
4 ADLs' state for females grouped in 5-year age bands in the 1989–94	
NLTCS.	. 396
O.129 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the '5–	
6 ADLs' state for females grouped in 5-year age bands in the 1989–94	
NLTCS.	. 397
O.130 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'Institutionalized' state for females grouped in 5-year age bands in	
the 1989–94 NLTCS	. 397
O.131 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'Healthy' state for males and females grouped in 10-year age bands	
in the 1989–94 NLTCS	. 398
O.132 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'IADL only' state for males and females grouped in 5-year age bands	
in the 1989–94 NLTCS	. 398
O.133 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'1–2 ADLs' state for males and females grouped in 5-year age bands	
in the 1989–94 NLTCS	. 399
O.134 Graphs of point estimates (CMLE), approximate 95% confidence	
intervals and parametric fits of the transition intensities out of the	
'3–4 ADLs' state for males and females grouped in 5-year age bands	
in the 1989–94 NLTCS	. 399
O.135 Graphs of point estimates (CMLE), approximate $95\%$ confidence	
intervals and parametric fits of the transition intensities out of the	
'5–6 ADLs' state for males and females grouped in 5-year age bands	
in the 1989–94 NLTCS	. 400
O.136 Graphs of point estimates (CMLE), approximate $95\%$ confidence	
intervals and parametric fits of the transition intensities out of the	
'Institutionalized' state for males and females grouped in 5-year age	
bands in the 1989–94 NLTCS.	. 400

### Acknowledgements

I would first like to thank my supervisor, Professor Angus Macdonald, for introducing me to this subject area that I have found so exciting to work in, and then for encouraging and guiding me through my research. He has inspired me through his own continual enthusiasm for research. He has always found time, however busy, to discuss my work and is always willing to share his expertise.

It is to my loving parents, Gary and Yvonne, that I dedicate this thesis for all that they have given me — far too much to list here. They are always a source of strength and peace in my life. My wife, Neringa, I also thank for her love, patience and support.

The Standard Life Assurance Company generously sponsored me throughout my Ph.D., and I thank them for such a great opportunity. They also gave me the opportunity to share my research with many in the profession at one of their C.P.D. sessions, for which I am grateful.

The database on disability (National Long-Term Care Survey), on which a significant part of this thesis is based, was freely made available to me by Professor Kenneth Manton, who I would like to thank very much for providing me with such a valuable resource.

Dr Adrienne Cupples and Dr Robert Watson kindly answered my queries on their work, and provided me with extra information at critical times in my research, for which I am very grateful.

I would also like to thank Dr Iain Currie, Ross Ainslie, Dr Carol Brayne, Dr Peter Brett, Robert Plumb, Professor Andrew Read, Dr Virginia Warren and Professor Howard Waters for helpful discussions and all those in the department who took part in my seminars and gave me feedback on my research.

#### Abstract

The discovery in 1991 that a person's apolipoprotein E (APOE) genotype provides an indication of their predisposition to Alzheimer's disease (AD), which accounts for a significant proportion of long-term care costs, raises the question of adverse selection in the long-term care insurance market and about long-term care costs in general.

The aims of this thesis are: to develop multi-state models that help to quantify this potential for adverse selection; to parameterize such models using the available data; and to use these parameterized models to estimate the potential costs of adverse selection in the long-term care insurance market. This thesis is in two parts.

The first part (Chapters 1 and 2) concentrates on Alzheimer's disease (AD). In Chapter 1, I discuss the epidemiology and genetics of Alzheimer's disease and I propose a simple model for estimating the long-term care costs of Alzheimer's disease attributable to each of the APOE genotypes. I parameterize the model using data from the medical and epidemiolocal literature. Then in Chapter 2, the parameterized model is used to estimate the potential costs of adverse selection in the long-term care insurance market under the crude assumption that non-AD related care costs are a proportion of AD related care costs.

The second part (Chapters 3 to 7) concentrates on modelling the disability process in order to independently estimate the care costs arising from disability in long-term care insurance. In Chapter 3, I introduce a model of disability and discuss the datasets that I use to parameterize the model (the 1982, 1984, 1989 and 1994 National Long-Term Care Surveys) and discuss previous relevant research that have used these datasets. Then in Chapter 4, I use maximum likelihood theory to estimate the model parameters, and compare the methodology to that of the previous research, discussed in Section 3. In Chapter 5, I discuss two methods for calculating the variance of the estimated model parameters, and use these variance estimates as weights in graduating the model parameters. I compare aggregate mortality in the disability model with a benchmark force of mortality in Chapter 6, and make some adjustments to the graduated model parameters, to make aggregate mortality in the model more consistent with the benchmark force of mortality. Then in Chapter 7, the graduated and adjusted models are used to estimate the cost of disability in a long-term care insurance contract, and these combined with the results from Chapter 2 are used to revisit the potential costs of adverse selection in the long-term care insurance market.

### Introduction

Molecular genetics has far-reaching implications for all aspects of health economics, including the effectiveness, or even practicability, of insurance-based funding of all forms of care. This is a natural subject for quantitative modelling.

The insurance industry has begun to recognise the far-reaching possibilities that research into human genetics might hold for traditional insurance practice, particularly underwriting. The possibilities can be summed up, somewhat crudely, as follows:

- if applicants for insurance have better knowledge of their medical risks than insurers, because they know the results of genetic tests, they might select against the office; but
- 2. it is often deemed unfair to discriminate against individuals (for example, by charging different insurance premiums) on the basis of their genetic make-up, over which they have no control.

The U.K. Government has so far avoided legislating on this very sensitive issue of the use of genetic test information by insurers, but instead has set up the Genetics and Insurance Committee (GAIC), charged with the assessment of the likely relevance and reliability of genetic test information as it relates to different kinds of insurance. Such considerations require actuarial models of the insurance process, allowing for the effects of specific genes on mortality and morbidity.

Life insurance was the first type of insurance, in the U.K., to be considered, in which context it has been suggested that the overall costs of adverse selection might be limited (Macdonald, 1997, 1999; Pritchard, 1997). These models, proposed by Macdonald (1997, 1999), were based on parameter estimates intended to be extreme, with the aim of suggesting an upper bound on the cost of adverse selection, rather than on data-based statistical estimates. Tan (1997) applied a similar model to annuity business, with results that suggested higher costs. However, these papers acknowledge:

- 1. that different conclusions might hold in respect of other forms of insurance; and
- 2. the lack of sound epidemiological data in respect of any but a few genetic conditions.

The first study to look at specific genes is that treating breast and ovarian cancer and the BRCA1 and BRCA2 genes by Lemaire *et al.* (1999) and Subramanian & Lemaire (1999). Further research has since been done on this topic by Macdonald, Waters & Wekwete (2000), modelling breast and ovarian cancer, allowing for family history, as well as BRAC1 and BRAC2 genotype, with applications to critical illness insurance. Early-onset Alzheimer's disease and the Presenilin-1 gene have also been the subject of investigation (Gui & Macdonald, 2002).

At the time of writing, long-term care (LTC) insurance has yet to reach significant volumes in the U.K., but it is possible that it will figure in some way in any shift from public to private provision of care in old age. For that reason, it is timely to consider the problems of modelling LTC insurance. Further, one of the main reasons for requiring LTC insurance is dementia, of which Alzheimer's disease (AD) forms a significant proportion of cases. There is clear evidence that AD has a genetic component; at least one gene variant — that for the  $\varepsilon 4$  allele of the apolipoprotein E (APOE) gene — has been linked to earlier onset of AD in epidemiological studies. Thus, a study of LTC insurance is also timely from the point of view of human genetics.

In Chapter 1, I propose a continuous-time Markov model for AD, and estimate its transition intensities using published medical and epidemiological studies. To allow for genetic variation arising from the  $\varepsilon 4$  allele of the APOE gene, the intensity of onset of AD is modelled as a function of APOE genotype.

In Chapter 2, the model is used to estimate the costs of a single premium LTC contract for lives with each APOE genotype, and a sensitivity analysis is carried

out on the main model assumptions. These genotype specific costs are then used to estimate the potential costs of adverse selection, especially with respect to the size of the market. The model is also used to consider the costs of a comprehensive retirement package that provides both pension and LTC cover, and the effect this type of product may have on the potential for adverse selection. I provide a summary of the results and draw conclusions at the end of this chapter.

The main focus in the first two chapters is on modelling AD and genetic heterogeneity. The model described cannot be used for estimating any long-term care costs other than those arising from AD. The costs of adverse selection calculated are initially reported as percentages of AD-related long-term care costs, which are then converted into total long-term care costs using the simple assumption that AD-related long-term care costs are a fixed proportion of total long-term care costs. The other main cause of claiming in a long-term care contract is through disability, and the rest of the thesis is concerned with modelling the disability process, with the aim of estimating the costs of disability in long-term care insurance (independently of Alzheimer's disease) — these estimates can then be combined with the results from Chapter 2 to revisit the potential costs of adverse selection in the long-term care insurance market.

It is particularly timely to consider modelling disability in ageing populations as the question of how the costs of long-term care should be apportioned between public funds and individuals is being raised in the U.K. — a Royal Commission was set up at the end of 1997 to investigate this question. The Commission finished its report in early 1999 (Sutherland *et al.*, 1999), concluding that "private sector solutions do not and in the forseeable future, will not offer a universal solution". While this may seem to be bad news for the LTC insurance business, sources in the industry believe that the Report may actually have a positive effect as:

- the commission's recommendations about what the state would provide were not comprehensive, leaving room for insurance companies to make up any shortfall;
- 2. it does give greater clarity of what the state does and does not provide; and
#### 3. it increased levels of public awareness of LTC as an issue.

The first point would mean that some LTC costs would be covered by the state. LTC insurance could then be used to cover the shortfall between actual LTC costs and what the state provides. So, as LTC insurance would then only cover part of the LTC costs, the premiums would be reduced, making LTC insurance more affordable. This combination of factors may then lead to an increase in demand for some forms of long-term care insurance.

In Chapter 3, I discuss the disability process, propose a continuous-time Markov model for disability and give details of the datasets that I use to parameterize the disability model (the 1982, 1984, 1989 and 1994 National Long-Term Care Surveys from the U.S.A.) — I also discuss previous relevant research that has used the same datasets. The model of disability proposed in this chapter is very flexible and could be applied, using relevant data, to pricing many long-term care products. Another application of this model would be to investigate trends of disability within a population to aid in public sector planning.

In Chapter 4, I estimate the model parameters, namely the transition intensities. It is not possible to estimate them in the usual way as occurrence/exposure rates since the datasets do not provide enough detail — they only provide information at discrete points in time, whereas complete lifetime history data is needed to estimate them directly. Instead, I develop maximum likelihood estimates for this restricted data and conclude by comparing the methodology from this chapter to that of the previous research done, discussed in Chapter 3.

Then in Chapter 5, I compare two methods for calculating the variance of the estimated model parameters (one which is not valid given the partial data available, even though it can be estimated), which demonstrates the effect on the variance estimates of only having partial data. The valid variance estimates are used as weights in the graduation process, where I fit parametric functions to the point estimates of the transition intensities.

I compare overall mortality in the disability models to a benchmark force of mortality in Chapter 6, and adjust the graduated model parameters in some of the disability models to make overall mortality in the models more consistent with the benchmark force of mortality.

In Chapter 7, I use the graduated and adjusted models to estimate the costs of disability in a long-term care contract and carry out a sensitivity analysis on the main model assumptions. I then compare the overall forces of mortality in the Alzheimer's disease models with those in the disability models, to check for consistency. Using the costs of disability and the results from Chapter 2, I revisit the potential costs of adverse selection in the long-term care insurance market and provide a summary and conclusions.

Finally, in Chapter 8, I discuss areas for further research.

## Chapter 1

# A Simple Model of Alzheimer's Disease and the APOE Gene

### 1.1 Introduction

In this chapter I propose a simple Markov model for Alzheimer's disease (AD) and estimate its transition intensities from medical and epidemiological studies. Genetic variation arises because the  $\varepsilon 4$  allele of the Apolipoprotein E (APOE) gene is known to indicate a predisposition to earlier onset of AD.

In this chapter and the next I concentrate on estimating the costs that arise under a long-term care insurance contract in respect of Alzheimer's disease. In Chapter 2, in order to look at the potential costs of adverse selection arising from variants of the APOE gene, the cost of other events in the ageing process (mainly disability) that trigger benefits are very simply assumed to be a multiple of those costs arising from Alzheimer's disease.

In Section 1.2, I briefly describe AD and then summarise the evidence for a genetic component of AD. In Section 1.3 the model is specified, and the transition intensities are estimated in Sections 1.4 and 1.5. Then in Section 1.6, occupancy probabilities are calculated from the model, which are then converted to prevalence rates and gene frequencies at older ages. Finally, in Section 1.7 I provide a summary and discussion. The parameterized model is applied to the question of adverse selection in the long-term care insurance market in the next chapter.

The research described in this chapter and the next is joint work with my supervisor, Professor Angus Macdonald, and formed the basis of two published papers Macdonald & Pritchard (2000) and Macdonald & Pritchard (2001). Applications of the model also include the study of long-term care costs (Warren *et al.*, 1999). Most of the research presented here was done in 1999, and is based on research papers available at that time — there is no doubt, with the great speed at which genetic knowledge is advancing, that some of the information in this chapter will, by the time of writing this thesis, have been superceded.

The processes leading to LTC insurance claims are complex, when compared with other forms of insurance, and there are no insurance data to speak of; therefore it is necessary (even for insurance companies themselves) to rely on data collected and published for a variety of reasons, mostly in the medical literature. As a result, the model proposed here is far from definitive, however the process of extracting information from the medical literature and putting it to actuarial use is very instructive, and I suggest that any shortcomings of this model shed useful light on the problems that might be faced in the future.

In a long-term care contract, claims can arise for two reasons, either: on the failure of a given number of activities of daily living (see Section 3.2 for more detail); or on reaching a certain level of cognitive impairment, resulting in a need for continual care or supervision. The term 'cognitive impairment' covers AD, which accounts for by far the greater number of cases, and other forms of mental deterioration, chiefly vascular in origin (for example, arising from strokes). Assessment is liable to be imprecise, making it difficult to decide on an exact date of inception of cognitive impairment, if such a thing exists. Moreover, although AD is the commonest form of cognitive impairment, it is hard to diagnose with certainty except by post-mortem examination. These factors introduce considerable uncertainty into epidemiological studies of AD. Breteler *et al.* (1992) noted that:

- 1. AD itself can have a significant vascular component;
- 2. some of the neuropathological symptoms of AD can also be symptoms of vascular dementia; and

3. studies by Tierney *et al.* (1988) found that post-mortem examination confirmed only 64–86% of diagnoses of AD.

## 1.2 Alzheimer's Disease and its Genetic Components

AD is a disease of old age; it is rare below ages 60–70. These rare cases are called 'early-onset' AD, which should not be confused with early onset of AD within the usual age range. This research is concerned only with the latter.

Families with a history of AD are sometimes observed, but AD also occurs sporadically (that is, in the absence of a family history of AD) and it is always possible that a case of AD in an affected family is, in fact, sporadic. The differences between familial and sporadic AD are not clear, although the former may be marked by earlier onset and more rapid progression.

A very few families have several cases of early-onset AD in several generations, consistent with autosomal dominant transmission (Levy-Lahad & Bird, 1996), and three genes have been found. First was the gene encoding for amyloid precursor protein (APP), involved in the production of  $\beta$ -amyloid. It resides on chromosome 21, which is the chromosome affected in cases of Down syndrome, sufferers of which often develop AD in middle age. Several mutations have been found, but they are rare. Later, mutations in two genes labelled presenilin-1 (PS-1) and presenilin-2 (PS-2) were identified, which appeared to be associated with AD, though the mechanisms remain unclear.

Familial AD is not restricted to early-onset cases and family history remains an important risk factor for late-onset AD (Jarvik *et al.*, 1996). Susceptibility genes have been identified, of which the most studied is that which codes for apolipoprotein E.

For a recent survey of the genetic epidemiology of AD, see Slooter & van Duijn (1997); Breteler *et al.* (1992), reviews the position before much was known about the genetic component of AD.

The pathology of AD includes:

- senile plaques (deposits on the outside of neurones (brain cells), consisting largely of the protein β-amyloid);
- 2. neurofibrillary tangles (connections between neurones);
- 3. amyloid angiopathy (deposits of amyloid protein in the arteries of the brain)
- 4. loss of neurones; and
- 5. decreased activity of choline acetyltransferase (an enzyme).

Therefore, any gene whose expression leads to the production, or over-production, of substances associated with these changes is potentially a genetic marker for AD.

#### 1.2.1 The Apolipoprotein E Gene

The aim of this summary is to give an impression of the progress made in understanding genetic factors of AD, as well as some of the problems and (perhaps most important) the great speed at which human genetics is advancing.

Apolipoprotein E is found in senile plaques and neurofibrillary tangles in AD patients. It has also been studied because of its role in lipid metabolism. The gene that encodes it is on chromosome 19, which was linked to families with late-onset AD by Pericak-Vance *et al.* (1991), making it a clear candidate gene for familial AD. Strittmatter *et al.* (1993) confirmed this hypothesis, which was rapidly supported by many other studies. The basic facts are as follows:

- 1. The APOE gene has three common alleles  $\varepsilon 2$ ,  $\varepsilon 3$  and  $\varepsilon 4$  whose frequencies are roughly 0.09, 0.77 and 0.14 respectively.
- 2. Since each offspring receives one allele from each parent, there are six possible genotypes (ε2/ε2, ε2/ε3, ε2/ε4, ε3/ε3, ε3/ε4 and ε4/ε4). Offspring with two copies of the same allele are called homozygotes, while those with two different alleles are called heterozygotes.
- 3. The APOE ε4 allele increases the risk of AD in a dose related fashion, such that ε4 homozygotes (ε4/ε4) are at a greater risk then ε4 heterozygotes (ε2/ε4, ε3/ε4), who in turn are at greater risk than those without the ε4 allele

(Bickeböller et al., 1997; Corder et al., 1994; van Duijn et al., 1995; Farrer et al., 1997; Jarvik et al., 1996; Kuusisto et al., 1994; Lehtovirta et al., 1995; Mayeux et al., 1993; Myers et al., 1996; Poirier et al., 1993; Tsai et al., 1994). See Section 1.5 for risk estimates. The risk depends on age, being highest at ages 60–70, tapering off at older ages (Bickeböller et al., 1997; Corder et al., 1994; Farrer et al., 1997).

- 4. It is also possible that the ε4 allele is associated with earlier onset of AD (not to be confused with early-onset AD). The effect may be dose dependent (Farrer et al., 1997; Frisoni et al., 1995; Gomez-Isla et al., 1996); or not (Corder et al., 1995; Lehtovirta et al., 1995; Stern et al., 1997); or it may not exist at all (Liddell et al., 1994; Masullo et al., 1998; Norrman et al., 1995).
- 5. Investigations into the rate of mental decline of AD patients by genotype found no evidence for any difference (Basun et al., 1995; Gomez-Isla et al., 1996; Masullo et al., 1998; Norrman et al., 1995). There is conflicting evidence about mortality. It is possible that younger age at onset should imply longer survival times, because of the usual age-related mortality differentials, and therefore that the ε4 allele should be associated with longer life after onset of AD. While some studies support this (Corder et al., 1995; Gomez-Isla et al., 1996; Norrman et al., 1995), others have found no difference (Basun et al., 1995; Stern et al., 1997). If ε4 is associated with lighter mortality in AD patients then risk estimates from cross-sectional studies (the vast majority to date) should be interpreted with caution. An incidence study (Evans et al., 1997) confirmed ε4 to be a significant risk factor, but the estimated increased risk of onset was at the lower end of the reported range.
- 6. In contrast, the ε2 allele has been found to have a protective effect against late-onset AD (Corder et al., 1994; Farrer et al., 1997; Gomez-Isla et al., 1996; Jarvik et al., 1996; Lambert et al., 1998; Masullo et al., 1998). However, a study of early-onset AD patients (van Duijn et al., 1995), found a higher frequency of the ε2 allele, and an association of ε2 with a more aggressive form of AD, suggesting different rôles of APOE in early-onset and late-onset

AD. Findings relating to the  $\varepsilon 2$  allele are based on the  $\varepsilon 2/\varepsilon 3$  genotype, as  $\varepsilon 2$  homozygotes are rare. The risk attached to the  $\varepsilon 2/\varepsilon 4$  genotype is not clear, possibly because  $\varepsilon 2$  and  $\varepsilon 4$  have opposite effects (Jarvik *et al.*, 1996; Levy-Lahad *et al.*, 1996).

APOE  $\varepsilon 4$  is the most important genetic risk factor for AD identified yet. Though it is neither necessary nor sufficient to cause AD it does increase susceptibility. Approximately 26% of Caucasians carry at least one  $\varepsilon 4$  allele and it has been estimated that between 42% and 79% of AD cases are attributable to the associated excess risk (Nalbantoglu *et al.*, 1994).

### 1.2.2 Other Genetic Factors of Alzheimer's Disease

In 1997, a gene for the K-variant of butyrylcholinesterase (BCHE K), not a risk factor by itself, was found to act in synergy with APOE  $\varepsilon 4$ , such that carriers of both (an estimated 6% of Caucasians) were at over 30 times the risk of AD as a person with neither (Lehmann *et al.*, 1997). Subsequent studies (Brindle *et al.*, 1998; Singleton *et al.*, 1998) failed to reproduce the result. Although some explanations have been advanced, caution is advisable in using BCHE K as a risk factor for AD.

Payami *et al.* (1997) reported an association between AD and the A2 allele of the human leokocyte antigen (HLA); the HLA-A2 phenotype and APOE  $\varepsilon 4/\varepsilon 4$  genotype had similar and additive effects on reducing age at onset of AD, at ages below 60 and above 75. Further studies would be needed to confirm these findings.

Poduslo *et al.* (1998) found the apolipoprotein CI (apo CI) gene to be a risk factor for early-onset and late-onset AD, whether sporadic or familial. Apo CI A homozygotes had 4 to 5 times the odds of developing AD, heterozygotes about twice the risk. This was not unexpected, since Apo CI is closely linked to APOE and in linkage disequilibrium with APOE and AD. Linkage disequilibrium is the non-random assortment, in a population, of two genes on the same chromosome (the strength of the linkage is inversely proportional to the distance between them). It was thought that the association of AD with APOE may be more significant.

### **1.3** A Model for Alzheimer's Disease

I use a continuous-time multiple state model. In this section I will discuss the reasons for this choice and how it is used to represent a (genetically) heterogenous population and I discuss the statistical framework of such a model in Section 1.3.1. For more general comments on these models, see Macdonald (1996a) or Waters (1984). Lives with each APOE genotype are assumed to form a homogeneous population, suffering the different risks of AD discussed in Sections 1.2.1 and 1.5.

An important reason for using these models is that they allow the most complete representation of the underlying process. It is then necessary to estimate a large number of transition intensities, for which adequate data do not always exist, but it is better to obtain a clear picture of the data needed than to sweep the issue under the carpet by working with some less adequate model in the first place. In particular:

- if some simpler model is eventually recommended for use, because of missing data or for computational convenience, it is important to be able to assess its soundness in practice; and
- 2. if missing data become available later, for example, as the insured lives experience develops, it is a hindrance if too much has been invested in a model that cannot incorporate it.

Modern computing power is such that the computational demands of multiple state models (numerical integration of differential equations) can quite reasonably be met, for arbitrarily complex Markov models (Norberg, 1995) and for many semi-Markov models (Waters & Wilkie, 1987; Waters, 1991). The techniques can all be found in standard texts on numerical analysis, and no actuary should be prevented from choosing an adequate model by the need to use them.

Figure 1.1 shows a simple model of AD. Each genotype is represented by such a model; the transition intensities in each model will differ, representing the different genetic risks. x denotes the age at outset (for example, when insurance is purchased,) and t the elapsed duration. The choice of states is dictated entirely by the events that have been studied in the medical and epidemiological literature. For certain



Figure 1.1: A simple model of Alzheimer's disease in the *i*th of M subgroups, each representing a different ApoE genotype. x is the age at outset, and t the elapsed duration.

purposes, it would be desirable to model other events, such as the start of a longterm insurance claim. No data about that event are available; however a major event that has been studied is institutionalisation. Although becoming institutionalised need not coincide with the start of an insurance claim, it is the best available proxy.

Macdonald (1999) considered frailty models as an alternative to Markov models, for genetics and insurance applications. They offer the advantage of a simple model of the genetic variability, if that is justified by the circumstances. They may be especially useful for modelling multifactorial disorders, or genes with very many alleles or mutations, but for a single gene with just a few alleles it seems reasonable to model each separately. Other possible models (such as Cox-type models) might be useful for modelling individual transitions but do not lend themselves to the inclusion of payments contingent upon complicated life histories. AD alone does not account for all long-term care costs. Broadly speaking, the need for care arises because of cognitive disorder (including AD) or loss of ability to perform Activities of Daily Living (ADLs) such as dressing, washing and feeding (see Section 3.2 for more details). A comprehensive model of long-term care costs can be specified in terms of these causes, with AD included as a component, and the impact of the APOE gene on overall care costs can thereby be studied. However, incorporating AD explicitly in an expanded model will require data that describe, at the level of individual lives, the progress of AD and the loss of ADLs. I look at modelling the disability process in detail in Chapters 3 to 7, but even then there is no data on the loss of ADLs for a person suffering from AD — the work in these chapters is only able to look at overall care costs from disability.

#### **1.3.1** Statistical Framework

In this section I look at the assumptions underlying the model and the equations used to calculate the occupancy probabilities. I first introduce the notation used:

- 1. The population is divided into M subgroups, denoted i = 1, 2, ..., M. When considering the whole population, M = 1. The  $\varepsilon 2/\varepsilon 2$  and  $\varepsilon 2/\varepsilon 3$  genotypes are combined in the model, since the former is so scarce, so when considering APOE genotypes, each is a separate subgroup, and M = 5.
- 2. Each subgroup is represented by a different model, as in Figure 1.1;
- 3. Lives enter the model at age x, always in the starting state (state i1).
- 4. When M > 1, the transition intensity between states j and k in the *i*th subgroup is denoted  $\mu_{x+t}^{ijk}$  (when M = 1 the *i* is omitted from the superscript);

Assume that the time period of interest starts at time 0 and ends at time n and that there is a finite number of mutually exclusive states  $\mathbf{S} = \{1, 2, ..., S\}$ , 1 being the initial state at time 0.

Then, the lifetime of an individual is represented by a continuous time Markov process on the state space **S**. The transition intensities (ignoring genotypes, for brevity, so that M = 1),  $\mu_x^{jk}$   $(j \neq k$  and  $j, k \in \mathbf{S}$ ) are assumed to exist  $\forall x \in (0, n)$ and are such that:

- 1. The probability, at age x, of a life moving from state j to state k in the small time period  $\delta t$ ,  $P_{xx+\delta t}^{jk} = \mu_x^{jk} \, \delta t + o(\delta t)$ ;
- 2. the probability of two or more transitions in the small period of time  $\delta t$  is  $o(\delta t)$ ; and

3. the Markov assumption holds, that is the transition intensities depend only on the current state and time and not on any previous history.

The Kolmogorov forward equations are then,  $\forall j, k \in \mathbf{S}$ :

$$\frac{\partial}{\partial t} P_{x\,x+t}^{kl} = \sum_{l \neq k} P_{x\,x+t}^{jl} \mu_{x+t}^{lk} - P_{x\,x+t}^{jk} \mu_{x+t}^{kl} \tag{1.1}$$

These equations are easily solved recursively, to give the occupancy probabilities of the model, by standard numerical methods even with a large number of states — I used a 4th order Runge-Kutta algorithm (Press *et al.*, 1993) with fixed step size 0.0005 years. They are solved forwards, the boundary condition being that the matrix of transition probabilities is simply the identity matrix for t = 0. For the simple calculation of occupancy probabilities, the 4th order Runge-Kutta algorithm with fixed step size proved to be sufficiently fast, however, in Chapter 4, when calculating maximum likelihood estimates many hundreds of occupancy probabilities need to be calculated. In this case, the inefficiency of using a fixed step size in the Runge-Kutta Algorithm becomes apparent and I use much more efficient algorithm with an adaptive step size, which takes the largest possible step size while keeping within a fixed accuracy. This took considerably more effort to initially set-up than the algorithm with fixed step size and I would suggest:

- 1. where only single or a few calculations are required, the Runge-Kutta algorithm with fixed step size is sufficient; but
- 2. where many calculations are required (i.e. in the intermediate process of calculating maximum likelihood estimates), the extra effort required to set-up a Runge-Kutta algorithm with adaptable step size, is justifiable (or even necessary) in terms of computer run-time.

## 1.4 Estimation of Transition Intensities Not Depending on APOE Genotype

In this section I estimate the transition intensities for the events: onset of AD; institutionalisation; and death. All of these must be 'estimated' from results reported in the medical and epidemiological literature. It would be best to work with the original data, but these are almost never available. Reported results are not always ideal for the extraction of parameters for an actuarial model; often the age groups used are very wide, and different in different surveys; sometimes only graphs (such as Kaplan-Meier survival curves) are given.

A most important distinction must be drawn when estimating transition intensities from epidemiological studies (see, for example, Clayton & Hills (1993), Kahn & Sempos (1989), Lilienfeld & Hills (1993), Selvin (1996)):

- 1. *Prospective* studies, based on samples of the general population, ought to yield the most reliable estimates of population risk, but are expensive and timeconsuming. Moreover, they are rarely even begun until substantial evidence of an effect has been accumulated from other studies.
- 2. *Case-based* studies, based on affected persons (and controls) often yield relative risks greatly in excess of the true population risks, precisely because the subjects are affected or at risk. However, early studies into any medical condition are almost inevitably of this type.

Current knowledge of most genetic disorders is derived from case-based studies; this is certainly true of APOE and AD (see Section 1.2.1). It is very likely that estimates of risk conferred by APOE genotype will fall as more prospective studies are carried out, but this will take time.

The approach I adopt is as follows:

- 1. in Section 1.4.1, I state assumptions about the general level of mortality;
- 2. in Section 1.4.2, I estimate the aggregate incidence of AD, denoted  $\mu_{x+t}^{AD}$ , which has been investigated extensively;
- 3. in Section 1.4.3, I estimate the intensity of institutionalisation, following the onset of AD (that is,  $\mu_{x+t}^{i23}$ ) and the force of mortality following the onset of AD (that is,  $\mu_{x+t}^{i24}$ );
- 4. in Section 1.4.4, I estimate the force of mortality for lives institutionalised with AD (that is,  $\mu_{x+t}^{i34}$ ); and

5. in Section 1.5, I estimate the population frequencies of the APOE alleles and then estimate the incidence of AD for each genotype using odds ratios from the genetic studies: this gives estimates of  $\mu_{x+t}^{i12}$ .

### 1.4.1 Baseline Mortality Tables

For convenience, I choose parametric approximations to the AM80 and AF80 Ultimate mortality tables as bases for mortality assumptions; for use in the model they are adjusted in a variety of ways. Gompertz curves were fitted to  $\mu_{x+t}$  at ages 65–120, using log-linear least squares (see equation (1.2)):

$${}^{AM80}\mu_{x+t} = 0.000094116e^{0.084554(x+t)}$$

$${}^{AF80}\mu_{x+t} = 0.000025934e^{0.093605(x+t)}.$$
(1.2)

Experiments with the AM80 and AF80 tables themselves showed that the Gompertz approximations had a negligible effect in long-term care applications; I use them because they are sometimes useful in numerical work. For insurance use, some allowance must be made for future improvements in mortality. No experience is available to help, but following discussion with some actuaries experienced in pricing long-term care insurance, I choose 65% of these baseline tables as the aggregate mortality assumptions.

#### 1.4.2 The Onset of Alzheimer's Disease in the Population

AD has been the subject of some large-scale epidemiological studies, many of them pre-dating the discovery of the rôle of the APOE gene. Some of these report incidence rates, or 'occurrence/exposure' rates, which are exactly the estimates needed for transition intensities.

There is general agreement, in this literature, on the shape of the intensity  $\mu_{x+t}^{AD}$ in the age range 60–85 years; it is very low at ages 60–64 (about 0 to 0.002) and increases rapidly with age, approximately doubling every 5 years. Sayetta (1986) and Hebert *et al.* (1995) found that a Gompertz curve gave the best fit, despite trying a number of more complex models. A number of studies report the incidence of AD (that is, the intensity  $\mu_{x+t}^{AD}$ ) by age but not by genotype, including Copeland *et al.* (1992), Hagnell *et al.* (1992), Kokmen *et al.* (1993), Letenneur *et al.* (1994), Nilsson (1984), Ott *et al.* (1998), Rocca *et al.* (1998) and Rorsman *et al.* (1986). Of particular interest, however, is the recent meta-analysis of the incidence of AD by Jorm & Jolley (1998):

- 1. it draws on 23 studies world-wide, including 13 European studies;
- 2. the analysis is carried out separately for Europe, the U.S.A. and East Asia;
- 3. the incidence of AD is estimated by severity, categorised as Mild+ and Moderate+ AD, where Mild+ includes all cases classified as mild or worse; and
- 4. point estimates of  $\mu_{x+t}^{AD}$  were obtained for 5-year age groups from 65 to 95, and no *a priori* shape of  $\mu_{x+t}^{AD}$  was assumed.

I estimated  $\mu_{x+t}^{AD}$  from Jorm & Jolley (1998) using the figures from the European studies and for Mild+ AD. The estimates, 95% confidence limits and the log-linear least squares Gompertz fit:

$$\mu_{x+t}^{AD} = 1.31275 \times 10^{-7} e^{0.145961(x+t)} \tag{1.3}$$

are shown in Figure 1.2. It is clear that a Gompertz curve is a very good fit.

Data on the incidence of AD among the very elderly (> 90 years) are sparse, so estimates at these ages have wide confidence intervals and the trend is uncertain. The meta-analysis by Gao *et al.* (1998) found that the rate of increase in  $\mu_{x+t}^{AD}$  slowed down with age, but other studies found no evidence of this (Hebert *et al.*, 1995; Jorm & Jolley, 1998; Letenneur *et al.*, 1994). I simply extrapolated the Gompertz formula above to all ages; the effect of this assumption will depend on the particular application, or type of insurance, and this should be investigated when the model is used — in the next chapter where I apply the model, I first carry out a sensitivity analysis of this assumption.

Many studies have found men and women to be at the same risk of AD (Kokmen *et al.*, 1993; Nilsson, 1984; Ott *et al.*, 1998; Rocca *et al.*, 1998) and, when differences have been reported (Gao *et al.*, 1998; Jorm & Jolley, 1998; Letenneur *et al.*, 1994),



Figure 1.2: Aggregate Incidence of Alzheimer's Disease: Point Estimates and 95% Confidence Intervals. Source: Jorm & Jolley (1998).

women were found to be at greater risk only at very old ages. Figure 1.3 shows the following least-squares fits:

$${}^{male}\mu^{AD}_{x+t} = 1.60976 \times 10^{-7} e^{0.137301(x+t)}$$
(1.4)

$${}^{female}\mu^{AD}_{x+t} = 8.50561 \times 10^{-9} e^{0.172430(x+t)}$$
(1.5)

to the incidence rates found by Rocca *et al.* (1998), which were not found to be significantly different. Some experiments (described in Section 2.4) in applying the model to AD-related long-term care insurance costs using different rates of AD for men and women (equations (1.4) and (1.5)) showed that it made little difference, and in the rest of this chapter I have used the aggregate rate (equation (1.3)).

### 1.4.3 Time from Onset of Alzheimer's Disease to Institutionalisation or Death

The available data do not allow analysis of  $\mu_{x+t}^{i23}$ ,  $\mu_{x+t}^{i24}$  or  $\mu_{x+t}^{i34}$  by genotype.

Table 1.1 summarises the literature on time to the first of institutionalisation or death ('first event') for AD patients. Some studies give times from entry to the study



Figure 1.3: Incidence of Alzheimer's Disease by Gender: Point Estimates from Rocca *et al.*, (1998).

rather than from onset, which is usually not observed directly. A striking feature is that few lives die before becoming institutionalised. This may seem surprising as AD patients have generally been reported to suffer higher mortality than healthy lives (see Section 1.4.4). However, AD's debilitating effects are not sudden, and it may be expected that patients will be in receipt of informal care between onset and institutionalisation, which might lead  $\mu_{x+t}^{i24}$  to be relatively light.

I used the data from the study by Jost & Grossberg (1995). Although it is not the largest study, it does have advantages:

- it is a brain bank study, so all AD cases were confirmed by autopsy (the only reliable method of diagnosis);
- 2. there were no censored cases; and
- 3. the time from onset to institutionalisation is estimated.

Since genotypes cannot be distinguished here, I will just write  $\mu_{x+t}^{23}$  and  $\mu_{x+t}^{24}$  instead of  $\mu_{x+t}^{i23}$  and  $\mu_{x+t}^{i24}$ , respectively. Guided by these data, moment estimates of

Table 1.1: Mean and median times to Institutionalisation (Inst'n) or First Event for AD Patients.

				% for which
Age	e at	Time (	years) to	1st event
Onset	Entry	Inst'n	1st Event	is death
	$71.4^{(1)}$			7.1%
	$72.0^{(2)}$		$3.1^{(2)}$	13.1%
$75.1^{(3)}$		$4.3^{(1)}$		15.0%
	$79.4^{(1)}$	$2.5^{(2)(4)}$		10.0%
	Aga Onset 75.1 <sup>(3)</sup>	$\begin{array}{c} \text{Age at} \\ \text{Onset} & \text{Entry} \\ & 71.4^{(1)} \\ & 72.0^{(2)} \\ \\ 75.1^{(3)} & \\ & 79.4^{(1)} \end{array}$	$\begin{array}{cccc} Age \ at & Time (\\ Onset & Entry & Inst'n \\ & 71.4^{(1)} & \\ & 72.0^{(2)} & \\ 75.1^{(3)} & & 4.3^{(1)} \\ & & 79.4^{(1)} & 2.5^{(2)(4)} \end{array}$	$\begin{array}{cccc} Age \ at & Time \ (years) \ to \\ Onset & Entry & Inst'n & Ist \ Event \\ & 71.4^{(1)} & & & \\ & 72.0^{(2)} & & 3.1^{(2)} \\ 75.1^{(3)} & & 4.3^{(1)} & \\ & 79.4^{(1)} & 2.5^{(2)(4)} & & \end{array}$

(1) Mean.

(2) Median.

(3) Mean age at onset of AD, if institutionalised, estimated as (mean age at institutionalisation – mean time to institutionalisation).

 $\left(4\right)$  Median time from onset estimated as 5.6 years.

 $\mu_{x+t}^{23}$  (the force of institutionalisation) and  $\mu_{x+t}^{24}$  (the force of mortality of an AD patient prior to institutionalisation) can be derived. The usual indicator functions  $(I_j)$  and sample path functions  $(N_{jk})$  in respect of a single life (see Macdonald (1996b)) are defined by:

$$I_{j}(t) = \begin{cases} 1 & \text{if life is in state } j \text{ at time } t \\ 0 & \text{otherwise} \end{cases}$$
$$dN_{jk}(t) = \begin{cases} 1 & \text{if life transfers from state } j \text{ to state } k \text{ at time } t \\ 0 & \text{otherwise} \end{cases}$$
$$N_{jk}(T) = \int_{0}^{T} dN_{jk}(t) = \text{ No. of transfers from state } j \text{ to state } k$$

Also let  $P_{xy}^{ij}$  be the probability that a life in state *i* at age *x* is in state *j* at age *y*. Then equation (1.6) below is the mean age at onset of AD, given that the life was eventually institutionalised with AD (as in Jost & Grossberg (1995)):

$$E\left[x + \int_{x}^{\omega} I_{1}(t)dt \mid N_{23}(\omega - x) = 1 \text{ and } I_{1}(x) = 1\right] = x + \frac{\int_{x}^{\omega} (t - x)\mu_{t}^{12}P_{xt}^{11}\left\{\int_{t}^{\omega}\mu_{s}^{23}P_{ts}^{22}ds\right\}dt}{\int_{x}^{\omega}\mu_{t}^{12}P_{xt}^{11}\left\{\int_{t}^{\omega}\mu_{s}^{23}P_{ts}^{22}ds\right\}dt}; (1.6)$$

equation (1.7) is the mean time from onset of AD to institutionalisation:

$$\mathbb{E}\left[\int_{x}^{\omega} I_{2}(t)dt \mid N_{23}(\omega - x) = 1 \text{ and } I_{1}(x) = 1\right] = \frac{\int_{x}^{\omega} \mu_{t}^{12} P_{xt}^{11} \left\{\int_{t}^{\omega} (s - t) \mu_{s}^{23} P_{ts}^{22} ds\right\} dt}{\int_{x}^{\omega} \mu_{t}^{12} P_{xt}^{11} \left\{\int_{t}^{\omega} \mu_{s}^{23} P_{ts}^{22} ds\right\} dt}$$
(1.7)

and equation (1.8) is the probability that an AD patient dies before becoming institutionalised. The upper age bound, denoted  $\omega$ , is taken to be 120 years:

$$P[N_{24}(\omega - x) = 1 | N_{12}(\omega - x) = 1 \text{ and } I_1(x) = 1] = \frac{\int_x^{\omega} \mu_t^{12} P_{xt}^{11} \left\{ \int_t^{\omega} \mu_s^{24} P_{ts}^{22} ds \right\} dt}{\int_x^{\omega} \mu_t^{12} P_{xt}^{11} dt}$$
(1.8)

Setting equations (1.6), (1.7) and (1.8) equal to their estimated values from Table 1.1, gives 3 equations, which can be solved for at most 3 unknown parameters. The parametric forms I chose were as follows:

- 1.  $\mu_{x+t}^{12} = A + \mu_{x+t}^{AD}$ , where  $\mu_{x+t}^{AD}$  is given by equation (1.3). This Makeham term adjusts the incidence of AD to a level that gives the same mean age at onset (for AD patients who become institutionalised).
- 2.  $\mu_{x+t}^{23} = D$ . I felt that the data did not support anything more elaborate than a constant intensity.
- 3.  $\mu_{x+t}^{24} = P \mu_{x+t}^{14}$ . That is, the mortality of an AD patient before becoming institutionalised is a proportion of baseline mortality.
- 4.  $\mu_{x+t}^{14}$ , baseline mortality, was taken as AM80 mortality, using the Gompertz approximation given by equation (1.2). Although it is appropriate to allow for future improvements in mortality in applications, it is not appropriate to do so in estimation based on past data. The values of D and P do not depend strongly on the baseline mortality.

Solving these equations numerically yields the solutions:

$$A = 0.02025038$$
  $D = 0.18895779$   $P = 0.33502488.$ 

The Makeham term, A, is a nuisance parameter used to adjust the incidence of AD so that the mean age at onset in the model is the same as that in the data. Its only purpose here is to improve the estimation of the other terms, as the survival of a cohort of AD patients is strongly related to their mean age at onset. It does not furnish an estimate of the incidence of AD in the whole population, which was described in Section 1.4.2. The magnitude of D would give a mean time to institutionalisation of about 5.3 years if there were no mortality, so allowing for mortality this value would seem about right. Given the low proportion of AD lives that die before institutionalisation, P, representing 34% of baseline mortality is also as expected.

The transition intensities  $\mu_{x+t}^{23}$  and  $\mu_{x+t}^{24}$  are summarised in Table 1.3.

### 1.4.4 Mortality of Lives with Alzheimer's Disease

AD patients have been found to suffer higher mortality than the general population (Barclay *et al.*, 1985(b); Bonaiuto *et al.*, 1995; Bracco *et al.*, 1994; Burns *et al.*, 1991; van Dijk *et al.*, 1991; Evans *et al.*, 1991; Heyman *et al.*, 1996; Mölsä *et al.*, 1986; Treves *et al.*, 1986). However, there is little agreement on the magnitude of the increase, or its dependence on age at onset, duration since onset, sex, race, level of education, marital status, level of cognitive impairment, familial/non-familial AD and level of behavioural impairment. The main factors that need to be considered are:

The magnitude of the increase in mortality for AD lives. The mortality of lives with AD has been investigated using different methodologies. For example, Evans et al. (1991), estimated the relative risk of death for AD patients as 1.44 (95% confidence interval 1.05–1.96) times that of the unaffected. Others have suggested that AD has only a small impact on mortality: Barclay et al. (1985a) claimed that well-tended individuals may have life expectancy close to normal, and Sayetta et al. (1986) found that survival did not depend on disease acquisition.

- 2. The effect of age at onset on relative mortality. The mortality of patients with AD increases with age (Bonaiuto, et al., 1995; Burns et al., 1991). Most studies into survival times have found no relation between age at entry into the study and relative survival (Barclay et al., 1985b; Bracco et al., 1994; Heyman et al., 1996; Mölsä et al., 1986; Stern et al., 1995), except that Barclay et al. (1985b) found that younger lives had shorter relative survival times. Diesfeldt et al. (1986), investigating survival from onset of AD, found that AD patients with onset before age 76 had reduced survival times, but not those with later onset. Comparing the two methods of investigation, Walsh et al. (1990) found that older age at onset affected survival adversely, whereas older age at entry into the study did not; a possible explanation was that older patients have symptoms for a shorter time before presentation. Although no definitive relationship between age at onset and relative survival emerges, it is clear that:
  - (a) survival with AD depends on age; and
  - (b) if age at onset affects relative mortality, the relationship is only weak, but possibly stronger at younger ages.

In terms of the model in Figure 1.1, this suggests that mortality in state i3 (institutionalised from AD) could be modelled by the addition of a Makeham term to the normal force of mortality; the latter is age dependent, and the Makeham term will be less significant at older ages.

3. The effect of the duration of AD on relative survival. Perhaps surprisingly, the duration of AD has not been found to be associated with increased mortality (Barclay et al., 1985a; Bracco et al., 1994; Burns et al., 1991; Diesfeldt et al., 1986; Heyman et al., 1996; Sayetta et al., 1986; Walsh et al., 1990). That is, AD patients with long duration of symptoms do not suffer higher mortality than patients, of the same age, with short duration of symptoms. In terms of the model, this means that the mortality of lives in states 2 and 3 (onset of AD and institutionalised from AD) does not depend on the time spent in these states. This is especially convenient, as it allows us to work in a Markov framework.

Reference	Mean (Median) Age at Onset	Mean (Median) Survival Time	Addition to $\mu_{x+t}^{34}$
Barclay et al. (1985a)	(73.3)  yrs	(8.1) yrs	0.15829
Bracco et al. (1994)	(72.4) yrs	7.3  yrs	0.25259
Diesfeldt et al. (1986)	75.6 yrs	7.2  yrs	0.21056
Heyman et al. (1996)	(69.2)  yrs	(9.7) yrs	0.10993
Jost <i>et al.</i> (1995)	75.1  yrs	8.11 yrs	0.13345
Kokmen et al. (1988)	80.4  yrs	$6.2 \ \mathrm{yrs}$	0.26420
Treves $et al.$ (1986)	$73.9 \mathrm{\ yrs}$	(9.3) yrs	0.08135
Average			0.17291

Table 1.2: Summary Statistics on Survival Times of AD Patients.

4. The effect of gender on relative survival with AD. Many researchers have found that the differences in survival between men and women with AD can be explained by the usual mortality differential between men and women (Beard et al., 1994; Bonaiuto et al., 1995; Bracco et al., 1994; Burns et al., 1991; Heyman et al., 1996; Walsh et al., 1990), though Barclay et al. (1985a), did find greater differences. In terms of modelling, allowing for the normal differences in mortality between genders should be sufficient.

Table 1.2 summarises the literature on survival with AD. Since genotypes cannot be distinguished here, I will just write  $\mu_{x+t}^{34}$  instead of  $\mu_{x+t}^{i34}$ . As in the previous section, the mean age at onset (see equation (1.9)) and the mean survival time (see equation (1.10)) in the model of Figure 1.1 can be written down as:

$$\mathbb{E}\left[x + \int_{x}^{\omega} I_{1}(t)dt \mid N_{12}(\omega - x) = 1 \text{ and } I_{1}(x) = 1\right] = (1.9)$$

$$x + \frac{\int_{x}^{\omega} (t - x)\mu_{t}^{12}P_{xt}^{11}dt}{\int_{x}^{\omega} \mu_{t}^{12}P_{xt}^{11}dt}$$

$$E\left[\int_{x}^{\omega} I_{2}(t) + I_{3}(t)dt \mid N_{12}(\omega - x) = 1 \text{ and } I_{1}(x) = 1\right] =$$

$$\frac{\int_{x}^{\omega} \mu_{t}^{12} P_{xt}^{11} \left\{\int_{t}^{\omega} (s - t)(\mu_{s}^{23} + \mu_{s}^{24}) P_{ts}^{22} ds + \int_{t}^{\omega} \mu_{s}^{23} P_{ts}^{23} \int_{s}^{\omega} (r - s)\mu_{r}^{34} P_{sr}^{33} dr ds \right\} dt}{\int_{x}^{\omega} \mu_{t}^{12} P_{xt}^{11} dt}$$

$$(1.10)$$

Setting equations (1.9) and (1.10) equal to their estimated values in Table 1.2, and noting the estimates of  $\mu_{x+t}^{23}$  and  $\mu_{x+t}^{24}$  from the previous section, there are 2 Table 1.3: Summary Of Transition Intensities for the AD Model with Baseline Mortality 100% (65%) of AM80 and AF80.

	Parameter Values							
Transition			$B(\times$	$10^{-5}$ )	$C(\times$	$10^{-2}$ )		
Intensity	А	D	Male	Female	Male	Female		
$\mu_{x+t}^{24}$	0	$0.33502\ (0.21776)$	9.4116	2.5934	8.4554	9.3605		
$\mu_{x+t}^{23}$	0.18896	0.00						
$\mu_{x+t}^{34}$ Lower bound	0.08	$1.00 \ (0.65)$	9.4116	2.5934	8.4554	9.3605		
$\mu_{x+t}^{34}$ Mean	0.17291	1.00  (0.65)	9.4116	2.5934	8.4554	9.3605		
$\mu_{x+t}^{34}$ Upper bound	0.27	1.00  (0.65)	9.4116	2.5934	8.4554	9.3605		

equations, which can be solved for at most 2 unknown parameters. The parametric forms I used are as follows:

- 1.  $\mu_{x+t}^{12} = A + \mu_{x+t}^{AD}$ , where  $\mu_{x+t}^{AD}$  is given by equation (1.3). This is just the addition of a Makeham term to the force of incidence of AD, shifting the latter to a level that gives the estimated age at onset.
- 2.  $\mu_{x+t}^{34} = K + {}^{AM80}\mu_{x+t}$ . This is a Makeham term as discussed in (d) above.

The estimated values of K for each of the references cited are given in the last column of Table 1.2. They range from about 0.08 to 0.27, with an average of 0.173. The Makeham term A is, again, only included to improve the estimation of the other terms (see the end of the previous section).

For clarity, I summarise the transition intensities estimated here. They all have the form:

$$\mu_{x+t}^{ij} = A + D B e^{C(x+t)}$$

and the calculated values are given in Table 1.3. Three values are given for  $\mu_{x+t}^{34}$ , an upper bound, mean value and lower bound to enable a check of how sensitive the results are, in any particular investigation, to this term.

## 1.5 Estimation of Transition Intensities Depending on APOE Genotype

Table 1.4 shows the population frequencies of the APOE genotypes estimated in several studies.

Some features are clear: the  $\varepsilon 3/\varepsilon 4$  genotype is not uncommon (about 21%) while the  $\varepsilon 2/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes are quite uncommon (about 3% and 1% respectively). It might be expected that there would be lower proportions of 'dangerous' genotypes at older ages, because these lives suffer a higher rate of AD onset, but the two agerelated studies (Bickeböller *et al.* (1997) and Corder *at al.* (1995)) gave conflicting results. However, there is reasonable agreement on the gene frequencies at around ages 60–70, which is what is needed for modelling.

Farrer *et al.* (1997) is a meta-analysis, combining the results of 40 other studies, including 6,264 Caucasian subjects. As it is the largest study, and differentiates by ethnic group, gender and ascertainment methods, and as the APOE  $\varepsilon 4$  allele was found with the same frequency in respect of AD diagnosed at autopsy and clinically diagnosed probable AD, I use its estimated gene frequencies, namely:  $\varepsilon 2/\varepsilon 2$  0.008;  $\varepsilon 2/\varepsilon 3$  0.127;  $\varepsilon 2/\varepsilon 4$  0.026;  $\varepsilon 3/\varepsilon 3$  0.609;  $\varepsilon 3/\varepsilon 4$  0.213;  $\varepsilon 4/\varepsilon 4$  0.018. These sum to 1.001, because of roundings used in Farrer *et al.* (1997), but I leave this small discrepancy unadjusted.

In a heterogeneous population, it is often convenient to think of a given intensity in each sub-population as a multiple (not necessarily constant) of a 'baseline' intensity, either in one of the sub-populations or in an aggregated 'average' population. Similarly, if  $p_1$  and  $p_2$  are the probabilities of an event in populations 1 and 2 respectively, the relative risk in population 2 (with respect to population 1) is  $p_2/p_1$ . A related quantity is the odds ratio: the odds in populations 1 and 2, respectively, are  $p_1/(1-p_1)$  and  $p_2/(1-p_2)$ , and the odds ratio is:

$$\frac{p_2(1-p_1)}{p_1(1-p_2)}.$$
(1.12)

	Country /	No. of		Age	Alle	le Frequ	ency		G	enotype	Frequen	cy	
Reference	Ethnicity	Lives	Sex	Group	$\varepsilon 2$	$\varepsilon 3$	$\varepsilon 4$	$\varepsilon 2/\varepsilon 2$	$\varepsilon 2/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 4/\varepsilon 4$
Bickeböller et al. (1997)	France	1,030	M & F	All	0.085	0.770	0.145	0.01	0.12	0.03	0.59	0.24	0.01
		316	Μ	All	0.070	0.815	0.105	0.01	0.10	0.02	0.67	0.19	0.00
		40	Μ	< 60	0.050	0.800	0.150	0.00	0.08	0.02	0.62	0.28	0.00
		93	Μ	60 - 69	0.070	0.845	0.085	0.01	0.12	0.00	0.70	0.17	0.00
		80	Μ	70 - 79	0.060	0.840	0.090	0.00	0.10	0.02	0.71	0.16	0.00
		103	Μ	$\geq 80$	0.081	0.795	0.125	0.01	0.11	0.03	0.64	0.20	0.01
		714	$\mathbf{F}$	All	0.090	0.750	0.170	0.01	0.13	0.03	0.55	0.27	0.02
		47	$\mathbf{F}$	< 60	0.075	0.735	0.190	0.00	0.15	0.00	0.51	0.30	0.04
		75	$\mathbf{F}$	60 - 69	0.075	0.730	0.195	0.01	0.11	0.02	0.52	0.31	0.03
		143	$\mathbf{F}$	70 - 79	0.065	0.765	0.160	0.00	0.12	0.01	0.56	0.29	0.01
		449	F	$\geq 80$	0.095	0.750	0.165	0.01	0.13	0.04	0.56	0.25	0.02
Corder $et al.$ (1994)	U.S.A.	243	M & F	$\geq 60$	0.105	0.750	0.155	0.00	0.16	0.05	0.56	0.22	0.02
		111	Μ	$\geq 60$	0.155	0.760	0.145	0.00	0.26	0.05	0.53	0.20	0.02
		132	$\mathbf{F}$	$\geq 60$	0.085	0.755	0.160	0.01	0.12	0.05	0.58	0.23	0.02
Corder $et al.$ (1995)	U.S.A.	236	M & F	All	0.095	0.755	0.150	0.00	0.16	0.03	0.56	0.23	0.02
		60	M & F	60 - 66	0.067	0.800	0.133		0.07	0.07	0.68	0.17	0.02
		124	M & F	67 - 74	0.101	0.738	0.161		0.18	0.02	0.53	0.23	0.03
		52	M & F	$\geq 75$	0.115	0.740	0.144		0.23	0.00	0.48	0.29	0.00
van Duijn $et \ al. \ (1995)$	Netherlands	532	M & F	< 65	0.103	0.731	0.165	0.01	0.17	0.02	0.51	0.27	0.02
		228	Μ	< 65	0.105	0.705	0.190	0.01	0.16	0.03	0.48	0.29	0.03
		304	$\mathbf{F}$	< 65	0.100	0.745	0.155	0.01	0.17	0.01	0.53	0.26	0.02
Evans <i>et al.</i> (1997)	E. Boston	490	M & F	$\geq 65$	0.062	0.854	0.084	0.01	0.09	0.01	0.74	0.14	0.01
Farrer $et \ al. \ (1997)$	Caucasian	6,262	M & F	All	0.084	0.779	0.137	0.01	0.13	0.03	0.61	0.21	0.02
	Afr-Amer	240	M & F	All	0.083	0.727	0.19	0.01	0.13	0.02	0.50	0.32	0.02
	Hispanic	267	M & F	All	0.067	0.823	0.110	0.00	0.12	0.01	0.67	0.18	0.02
	Japanese	$1,\!977$	M & F	All	0.042	0.869	0.089	0.00	0.07	0.01	0.76	0.16	0.01

### Table 1.4: Estimated Population Frequency of ApoE Genotypes

	Country /	No. of		Age	Alle	le Frequ	ency		G	enotype	Frequen	cy	
Reference	Ethnicity	Lives	$\mathbf{Sex}$	Group	$\varepsilon 2$	$\varepsilon 3$	$\varepsilon 4$	$\varepsilon 2/\varepsilon 2$	$\varepsilon 2/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 4/\varepsilon 4$
Gomez-Isla et al. (1996)	U.S.A	129	M & F	42 - 102	0.063	0.810	0.130	0.00	0.11	0.02	0.64	0.23	0.01
Jarvik et al. (1996)	not given	310	M & F	48 - 98	0.098	0.750	0.132	0.01	0.12	0.05	0.59	0.20	0.01
		117	Μ	48 - 98	0.068	0.791	0.124	0.02	0.09	0.02	0.64	0.21	0.01
		193	$\mathbf{F}$	48 - 98	0.117	0.725	0.137	0.01	0.15	0.07	0.55	0.20	0.01
Lambert et al. (1998)	not given	308	M & F	not given	0.081	0.805	0.114	0.00	0.13	0.02	0.65	0.18	0.01
Lehtovirta et al. (1995)	Finland	55	M & F	not given	0.009	0.882	0.109	0.00	0.02	0.00	0.78	0.18	0.02
Levy-Lahad et al. (1996)	not given	304	M & F	not given	0.100	0.765	0.135	0.01	0.13	0.05	0.60	0.21	0.01
Liddel $et al.$ (1994)	U.K.	86	M & F	not given	0.110	0.766	0.123	0.01	0.17	0.03	0.60	0.17	0.03
Lopez <i>et al.</i> (1998)	Spain	45	M & F	not given	0.060	0.810	0.120	0.00	0.10	0.02	0.65	0.22	0.00
	U.S.A.	58	M & F	not given	0.045	0.760	0.195	0.00	0.09	0.00	0.55	0.33	0.03
Lucotte et al. (1997)	France	248	M & F	$\geq 65$	0.069	0.804	0.127	0.02	0.09	0.01	0.65	0.22	0.01
Nalbantoglu et al. (1994)	Canada	77	M & F	not given	0.050	0.900	0.050	0.00	0.10	0.00	0.82	0.06	0.01
		53	Μ	not given	0.060	0.930	0.020	0.00	0.11	0.00	0.85	0.04	0.00
		24	$\mathbf{F}$	not given	0.040	0.860	0.100	0.00	0.08	0.00	0.76	0.12	0.04
Poirier $et al.$ (1993)	Canada	77	M & F	not given	0.088	0.770	0.122	0.00	0.22	0.00	0.57	0.19	0.03
		29	Μ	not given	0.138	0.725	0.137	0.00	0.28	0.00	0.48	0.21	0.03
		45	$\mathbf{F}$	not given	0.088	0.799	0.113	0.00	0.18	0.00	0.62	0.18	0.02
Roses $et al.$ (1995)	not given	91	M & F	not given	0.104	0.731	0.165	0.01	0.11	0.08	0.57	0.21	0.02
Tsai <i>et al.</i> (1994)	U.S.A.	77	M & F	47 - 95	0.110	0.760	0.130	0.01	0.16	0.04	0.57	0.22	0.00

### Table 1.4: Estimated Population Frequency of ApoE Genotypes-continued

When intensities are small, the odds ratio is a good approximation to the relative risk. In many studies, the published results are either relative risks or odds ratios.

Few studies report prospectively the incidence of AD by genotype. Two that do are Evans *et al.*, (1997) and Slooter *et al.*, (1998). Both have quite small study populations, and neither provides age specific estimates of AD risk, so they are not appropriate for modelling purposes.

Table 1.5 gives the Odds Ratios (ORs) of AD and 95% confidence intervals from a number of genetic studies. The 'reference' populations (also shown in the table) were either the  $\varepsilon 3/\varepsilon 3$  genotype or the three non- $\varepsilon 4$  genotypes combined. The estimated ORs vary considerably across studies. For example, estimates of the OR for the  $\varepsilon 3/\varepsilon 4$  genotype (relative to the  $\varepsilon 3/\varepsilon 3$  genotype) range from 1.8% to 3.7%, and for the  $\varepsilon 4/\varepsilon 4$  genotype, from 6.2% to 30.7%. Some of the variation may be explained by the differences between the studies themselves. In particular, differences may arise from: the method of ascertainment of patients, the countries of study, the method of diagnosis of AD, the age structure of the samples, the reference/risk genotypes, and whether they are cross-sectional or prospective studies. I make the following observations:

- 1. The study by Lopez *et al.* (1998) suggests that the risk of AD associated with the APOE  $\varepsilon 4$  allele may be different in different countries.
- 2. In support of the above, Mayeux *et al.* (1993) found that the association between APOE and AD may depend on ethnic group and, in particular, may not be present in black populations.
- 3. Despite the differences between studies: the presence of one or two  $\varepsilon 4$  alleles is consistently reported to be significantly associated with AD; and homozygotes are generally reported to have higher risk of onset of AD than heterozygotes.
- 4. The weakest associations between APOE and AD were reported in the two prospective studies, those by Evans *et al.* (1997) and Slooter *et al.* (1998). This is as expected for the reasons given in Section 1.4.

For modelling purposes, the genetic risk of AD at different ages is important. Few studies have considered this; the odds ratios from two that have are given in

	Study	Reference		Odd	ls Ratio
Reference	$Type^{(1)}$	$Genotype^{(2)}$	$Genotype^{(2)}$	Mean	$95\%~{\rm CI}$
Evans <i>et al.</i> (1997)	Р	arepsilon 3/arepsilon 3	$\varepsilon 3/\varepsilon 4 \& \varepsilon 4/\varepsilon 4$	2.27	1.1 - 4.9
Frisoni et al. (1995)	$\mathbf{C}$	-/-	$\varepsilon 4/-$	6.6	2.2 - 19.5
			$\varepsilon 4/\varepsilon 4$	17.9	4.5 - 70.5
Jarvik et al. (1996)	С	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 3$	0.4	0.2 - 0.96
			$\varepsilon 2/\varepsilon 4$	1.4	0.6 - 3
			$\varepsilon 3/\varepsilon 4$	3.1	2 - 4.7
			$\varepsilon 4/\varepsilon 4$	30.7	7 - 131
Kuusisto et al. (1994)	Р	-/-	$\varepsilon 4/-$	2.7	1.4 - 5.2
			$\varepsilon 4/\varepsilon 4$	9.1	3.5 - 23.4
Lambert et al. (1998)	С	-/-	$\varepsilon 4/$ - & $\varepsilon 4/\varepsilon 4$	4.66	3.14 - 6.93
Lehtovirta et al. (1995)	$\mathbf{C}$	-/-	$\varepsilon 4/-$	5.1	2.4 - 11.1
			$\varepsilon 4/\varepsilon 4$	21.4	2.8 - 166.3
Liddell et al. (1994)	С	-/-	$\varepsilon 4/-$	2.2	1.1 - 4.7
		·	$\varepsilon 4/\varepsilon 4$	10.7	2.3 - 48.8
Lopez <i>et al.</i> (1998)	$\mathbf{C}$	-/-	$\varepsilon 4/$ - & $\varepsilon 4/\varepsilon 4$	$2.34^{(3)}$	1.03 - 5.55
•		,	$\varepsilon 4/-\& \varepsilon 4/\varepsilon 4$	$3.64^{(4)}$	1.78 - 7.69
Mayeux et al. (1993)	Р	-/-	$\varepsilon 4/-$	$4.2^{(5)}$	1.8 - 9.5
-		·	$\varepsilon 4/\varepsilon 4$	$17.9^{(5)}$	4.6 - 69.8
			$\varepsilon 4/-\& \varepsilon 4/\varepsilon 4$	$15.3^{(6)}$	3.0 - 78.1
			$\varepsilon 4/$ - & $\varepsilon 4/\varepsilon 4$	$0.7^{(7)}$	0.1 - 6.4
			$\varepsilon 4/$ - & $\varepsilon 4/\varepsilon 4$	$4.5^{(8)}$	0.7 - 27.7
Myers <i>et al.</i> (1996)	Р	arepsilon 3/arepsilon 3	$\varepsilon 3/\varepsilon 4$	3.7	1.9 - 7.5
- , ,		·	$\varepsilon 4/\varepsilon 4$	30.1	10.7 - 84.4
Nalbantoglu	А	-/-	$\varepsilon 4/-\& \varepsilon 4/\varepsilon 4$	15.5	6.2 - 38.5
et al. (1994)		,	, ,		
Slooter $et al.$ (1998)	Р	$\varepsilon 3/\varepsilon 3$	arepsilon 2/arepsilon 3	0.4	0.1 - 1.0
( )		,	$\varepsilon 2'/\varepsilon 4$	1.3	0.2 - 8.5
			$\varepsilon 3/\varepsilon 4$	1.8	1.0 - 3.1
			$\varepsilon 4/\varepsilon 4$	6.2	1.4 - 28.2
Tsai <i>et al.</i> (1994)	$\mathbf{C}$	-/-	$\varepsilon 4/$ - & $\varepsilon 4/\varepsilon 4$	4.6	1.9 - 12.3
× /		,	$\varepsilon 4/-$	3.6	1.5 - 9.8

Table 1.5: Aggregated Odds Ratios of AD for the ApoE  $\varepsilon$ 4 Allele.

(1) Study Type: C indicates clinic/hospital; P, population/community; and A, autopsy/brainbank.

- (2) Dash (-) represents  $\varepsilon 2$  or  $\varepsilon 3$  alleles.
- (3) Study population–Gerona, Spain.
- (4) Study population–Pittsburgh, USA.
- (5) Mixture of White, Black and Hispanic ethnic groups.
- (6) White ethnic group only.
- (7) Black ethnic group only.

(8) Hispanic ethnic group only.

AgeOdds RatioAgeOddsGroupGenotypeMean95% CIGroupGenotypeMean60-69 $\varepsilon 2/\varepsilon 3$ 0.30.0-2.360-66 $\varepsilon 2/\varepsilon 3$ 0.1 $\varepsilon 2/\varepsilon 4$ $\varepsilon 2/\varepsilon 4$ 1.2 $\varepsilon 3/\varepsilon 4$ 3.11.4-6.9 $\varepsilon 3/\varepsilon 4$ 11.1	Corder et al. (1994)			
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Ratio			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	95% CI			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-			
$\varepsilon 3/\varepsilon 4$ 3.1 1.4-6.9 $\varepsilon 3/\varepsilon 4$ 11.1	-			
	-			
$\varepsilon 4/\varepsilon 4$ 29.1 3.6-239.5 $\varepsilon 4/\varepsilon 4$ 123.8	-			
70-79 $\varepsilon 2/\varepsilon 3$ 0.4 0.1-2.3 67-74 $\varepsilon 2/\varepsilon 3$ 0.3	-			
$\varepsilon 2/\varepsilon 4$ $\varepsilon 2/\varepsilon 4$ 1.1	-			
arepsilon 3/arepsilon 4 3.2 1.5–6.6 $arepsilon 3/arepsilon 4$ 4.6	-			
$\varepsilon 4/\varepsilon 4$ $\varepsilon 4/\varepsilon 4$ 20.8	-			
$80+$ $\varepsilon 2/\varepsilon 3$ 0.3 0.0-2.6 75-92 $\varepsilon 2/\varepsilon 3$ 0.5	-			
$\varepsilon 2/\varepsilon 4$ $\varepsilon 2/\varepsilon 4$ 1.6	-			
$\varepsilon 3/\varepsilon 4$ 1.3 0.5–3.4 $\varepsilon 3/\varepsilon 4$ 3.2	-			
$\varepsilon 4/\varepsilon 4$ $\varepsilon 4/\varepsilon 4$ 10.0	-			
$60+ \varepsilon 2/\varepsilon 3 \qquad 0.4 \qquad 0.1-0.9 \qquad 60+ \varepsilon 2/\varepsilon 3 \qquad 0.3$	-			
arepsilon 2/arepsilon 4 1.6 0.5–5.5 $arepsilon 2/arepsilon 4$ 1.1	-			
$\varepsilon 3/ \varepsilon 4$ 2.2 1.5–3.5 $\varepsilon 3/ \varepsilon 4$ 4.4	-			
$\varepsilon 4/\varepsilon 4$ 11.2 4.0-31.6 $\varepsilon 4/\varepsilon 4$ 19.3	-			

Table 1.6: Odds Ratios of AD by Genotype and Age.

Table 1.6. Bickeböller *et al.* (1997) is based on hospital admissions, and Corder *et al.* (1994) on autopsy cases; both use  $\varepsilon 3/\varepsilon 3$  as the reference population. Although some ORs are missing, because of small sample sizes, the trends are fairly clear:

- 1. The odds of AD among the higher risk genotypes ( $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$ ) fall with age. This may be expected as higher risk genotypes will succumb to AD more rapidly, reducing the proportion of such genotypes within the population.
- 2. Conversely, the protection conferred by the lower risk genotype  $(\varepsilon 2/\varepsilon 3)$  seems to weaken with age, possibly as this genotype becomes more common in the remaining population.

These trends are supported by the meta-analysis by Farrer *et al.* (1997), and as it is from this study that I take estimates of the APOE genotype risks, I cite some relevant details:

1. The aggregate relative odds from Farrer *et al.* (1997) (relative to the  $\varepsilon 3/\varepsilon 3$  genotype) are shown in Figure 1.4.



Figure 1.4: Odds ratios (ORs) of AD relative to  $\varepsilon 3/\varepsilon 3$  genotype for males and females combined. Source: Farrer *et al.* (1997).

- 2. The genotype risks of AD were not significantly different in respect of Caucasian males and females, except in the case of the ε3/ε4 genotype, for which women had a significantly higher risk of AD. The relative odds of AD by APOE genotype for Caucasian men and women are shown in Figures 1.5 and 1.6. (The authors kindly provided me with the numerical values of the odds ratios; confidence intervals were not available.)
- The genotypes ε2/ε2 and ε2/ε3 were combined as there were very few ε2/ε2 genotypes, and the risks associated with the two genotypes appeared to be similar.
- 4. Note that the risks associated with the APOE  $\varepsilon 4$  allele were considerably higher than those found in the two population-based studies by Evans *et al.*, (1997) and Slooter *et al.*, (1998).

For use in the AD model, these odds ratios have to be converted into relative risks. More precisely, it is necessary to find a plausible set of age- and genotypedependent transition intensities that are consistent with the odds ratios and together



Figure 1.5: Odds ratios (ORs) of AD relative to  $\varepsilon 3/\varepsilon 3$  genotype for  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes. Source: Farrer *et al.* (1997).



Figure 1.6: Odds ratios (ORs) of AD relative to  $\varepsilon 3/\varepsilon 3$  genotype for  $\varepsilon 2/\varepsilon 2$  or  $\varepsilon 2/\varepsilon 3$  and  $\varepsilon 2/\varepsilon 4$  genotypes. Source: Farrer *et al.* (1997).

are consistent with the aggregate incidence of AD. There is no unique solution to this problem. The method I used was to model the incidence of AD for the ith genotype as:

$$\mu_{x+t}^{i12} = r_1 f_{x+t}^i \,\mu_{x+t}^{AD} \tag{1.13}$$

where:

- 1.  $\mu_{x+t}^{AD}$  is the aggregate incidence rate of AD (from Section 1.4.2);
- 2.  $f_{x+t}^i$  is a parametric function representing the risk relative to the aggregate incidence rate, where  $f_{x+t}^i = 1$  in the case of the  $\varepsilon 3/\varepsilon 3$  genotype; and
- 3.  $r_1$  is a constant chosen so that the aggregate incidence of AD based on the modelled intensities is consistent with the aggregate incidence  $\mu_{x+t}^{AD}$ .

I did this for for males and females separately and combined and only looked at ages 60 and over, in order to get a better fit in the age range of interest in applications. The form of the ORs, either rising to a peak and then falling, or gently declining, suggested a similar pattern of relative risks, and the following family of functions provided a satisfactory fit (note that constant relative risks, or proportional hazards, result in odds ratios with exponential growth):

$$f_{x+t}^{i} = Ee^{-F((x+t)-k_1)^2 - G((x+t)-k_2)} + H.$$
(1.14)

In actuarial notation this is a GM(1,3) function, familiar in the graduation of life tables (Forfar, McCutcheon & Wilkie, 1988), although as described below either For G is set to zero. This is flexible enough to give a good approximation to the ORs, and is also suitable for extrapolating beyond age 90. The fitting procedure was as follows:

- 1. by considering the form of the OR, set either F = 0 (giving an exponential function) or G = 0 (giving a bell-curve function), and set H equal to 0 or 1;
- 2. the best value of  $k_1$  or  $k_2$  was found, to the nearest integer, by inspection;
- 3. the resulting ORs were calculated from the model in Figure 1.1 using the intensities from previous sections; and

			Paran	neter Valu	es			
Gender	Genotype	Ε	$\mathbf{F}$	G	Η	$k_1$	$k_2$	$r_1$
Both	$\varepsilon 4/\varepsilon 4$	13.5	0.00529	0	1	60	_	0.93
	$\varepsilon 3/\varepsilon 4$	2.98	0.00312	0	1	62	_	
	$\varepsilon 2/\varepsilon 4$	2.87	0.00938	0	1	68	_	
	$\varepsilon 2/\varepsilon 2 \ \& \ \varepsilon 2/\varepsilon 3$	0.754	0	0.00859	0	—	60	
Female	$\varepsilon 4/\varepsilon 4$	10.4	0.00504	0	1	60	_	0.88
	$\varepsilon 3/\varepsilon 4$	3.68	0.00319	0	1	62	_	
	$\varepsilon 2/\varepsilon 4$	4.21	0.0102	0	1	68	_	
	$\varepsilon 2/\varepsilon 2 \ \& \ \varepsilon 2/\varepsilon 3$	0.675	0	0.00692	0	-	60	
Male	$\varepsilon 4/\varepsilon 4$	8.94	0.00656	0	1	60	_	1.27
	$\varepsilon 3/\varepsilon 4$	1.92	0.00103	0	0	51	_	
	$\varepsilon 2/\varepsilon 4$	1.42	0.00506	0	0	67	_	
	$\varepsilon 2/\varepsilon 2 \ \& \ \varepsilon 2/\varepsilon 3$	0.434	0	0.0160	0	-	60	

Table 1.7: Parameters for the Relative Risk of AD for Males, Females and in Aggregate, by Genotype.

4. the remaining coefficients (either E and F, or E and G) were fitted by least squares.

For the calculations in point 3 above the following parameters were used:

- 1.  $\mu_{x+t}^{14} = 0.65 \times {}^{AM80}\mu_{x+t}$  for males and  $\mu_{x+t}^{14} = 0.65 \times {}^{AF80}\mu_{x+t}$  for females and in aggregate, where  ${}^{AM80}\mu_{x+t}$  and  ${}^{AF80}\mu_{x+t}$  are given by equation (1.2).
- 2.  $\mu_{x+t}^{23} = 0.189$ , calculated in Section 1.4.3.
- 3.  $\mu_{x+t}^{24} = 0.335 \times \mu_{x+t}^{14}$ , calculated in Section 1.4.3.
- 4.  $\mu_{x+t}^{34} = 0.173 + \mu_{x+t}^{14}$ , the mean value calculated in Section 1.4.4.

The fitted parameters are given in Table 1.7. Figure 1.7 shows that the modelled ORs closely reproduce the estimates from Farrer *et al.* (1997) (see Figure 1.4; only the aggregate ORs are shown, and the modelled ORs start at age 61 because at age 60 all lives are assumed to be unaffected).

To determine the parameter  $r_1$ , the aggregate incidence of AD in the whole model was calculated, and this fitted to  $\mu_{x+t}^{AD}$  by least squares. If  ${}^{i}P_{xx+t}^{11}$  is the probability that a life with genotype *i*, healthy at age *x*, is unaffected by AD at age x + t, and if  $p_x^i$  is the population frequency of the *i*th genotype at age *x* then the aggregate



Figure 1.7: Odds ratios (ORs) of AD relative to  $\varepsilon 3/\varepsilon 3$  genotype from Farrer *et al.* (1997), compared with ORs computed using modelled relative risk functions.

incidence of AD is:

Aggregate incidence of AD at age 
$$(x+t) = \frac{r_1 \left\{ \sum_{i=1}^5 p_x^i \, {}^i P_{xx+t}^{11} \, f_{x+t}^i \right\} \, \mu_{x+t}^{AD}}{\sum_{i=1}^5 \, p_x^i \, {}^i P_{xx+t}^{11}}$$
(1.15)

I took x = 60, and for the  $p_x^i$  used the gene frequencies of the Caucasian control populations in Farrer *et al.* (1997), which were given in Table 1.4. The incidence of AD,  $\mu_{x+t}^{AD}$ , was taken as that estimated in equation (1.3) and the occupancy probabilities,  ${}^{i}P_{xx+t}^{11}$ , were calculated by solving Kolmogorov's forward equations numerically (see Section 2.3).

The values of  $r_1$  are given in Table 1.7. The adjustment to the overall level only had a marginal effect on the modelled ORs for the individual genotypes.

The relative risk functions for males and females are given in Figures 1.8 and 1.9. For females, the  $\varepsilon 4/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 2/\varepsilon 4$  genotypes are unambiguously high-risk; the relative risks exceed 1.0 at all ages. For males, only the  $\varepsilon 4/\varepsilon 4$  genotype confers higher risks at all ages. The  $\varepsilon 2$  allele appears to be protective, so the  $\varepsilon 2/\varepsilon 2$  and  $\varepsilon 2/\varepsilon 3$  genotypes are low-risk, while the  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 2/\varepsilon 4$  genotypes are initially at higher risk but are at lower risk from about age 75. The protection apparently given



Figure 1.8: Modelled risk of AD, relative to the  $\varepsilon 3/\varepsilon 3$  genotype, for  $\varepsilon 4/\varepsilon 4$  and  $\varepsilon 3/\varepsilon 4$  genotypes. Based on odds ratios from Farrer *et al.* (1997).

by the  $\varepsilon 2$  allele in males means that the  $\varepsilon 3/\varepsilon 3$  genotype confers slightly higher than average risk; this is why, in Table 1.7,  $r_1 > 1$  for males. It must be remembered that data in respect of males are relatively sparse, and data in respect of very old males even sparser, so these effects should be treated with caution; more confidence should be placed in the relative risks in respect of females.

These risk estimates probably overstate the true population risks, perhaps quite substantially, as they are from clinic- and autopsy-based studies, which investigate precisely the subjects that are affected or already known to be at risk. To allow for this possibility I will also consider models assuming that the true relative risks are a proportion m < 1 of those estimated above. This can be done by adjusting equation (1.13) so that for genotype *i*:

$$\mu_{x+t}^{i12} = r_m \left\{ \left( f_{x+t}^i - 1 \right) m + 1 \right\} \, \mu_{x+t}^{AD} \tag{1.16}$$

where  $f_{x+t}^i$  is as above, and  $r_m$  is chosen as above so that the aggregate incidence of AD in the model is consistent with  $\mu_{x+t}^{AD}$ . Values of  $r_{0.5}$  and  $r_{0.25}$  are shown in Table 1.8.



Figure 1.9: Modelled risk of AD, relative to the  $\varepsilon 3/\varepsilon 3$  genotype, for  $\varepsilon 2/\varepsilon 4$  and  $\varepsilon 2/\varepsilon 2$  &  $\varepsilon 2/\varepsilon 3$  genotypes. Based on odds ratios from Farrer *et al.* (1997).

Table 1.8:  $r_m$  for m = 1, 0.5 and 0.25.

Gender	$r_1$	$r_{0.5}$	$r_{0.25}$
Both	0.93	0.96	0.97
Female	0.88	0.94	0.97
Male	1.27	1.11	1.05


Figure 1.10: Comparison of the estimated population incidence of AD  $\mu_{x+t}^{AD}$  with the aggregated incidence of AD for different levels of relative risk, males and females combined.

Figure 1.10 shows that the aggregate incidence of AD in the genetic model for both sexes combined is quite close to  $\mu_{x+t}^{AD}$ . It also shows that increasing the level of relative risk tends to overstate the incidence of AD at younger ages, and to understate it at older ages; the reason is that higher relative risks deplete the highrisk groups more quickly, leaving a relatively healthier population at older ages.

Decreasing the level of relative risks for high-risk genotypes means increasing the relative risks for low-risk genotypes. Using a lower value of  $r_m$  will diminish any effects of the (possibly anomalous) feature, noted above, that the  $\varepsilon 3/\varepsilon 4$  genotype is low-risk for males.

I chose a simple model for the relative risks (equations (1.13) and (1.14)). Alternatives were considered, in particular cubic polynomials and Gamma functions, but these gave poorer fits, and were less suitable for extrapolation (cubics to older ages and Gamma functions to younger ages). Also, it is easily seen that if an OR is specified as a function of time, and the transition intensity in the reference population is given, the transition intensity in the second population is determined (as the solution to an ODE); Figure 1.7, therefore, gives good support for the choice of model. Further refinement seems somewhat spurious, given the type of data used, and in view of the major sensitivity analysis needed in respect of the dominant parameter m.

#### **1.6** Occupancy Probabilities

Figures 1.11 and 1.12 for females and Figures 1.13 and 1.14 for males, show occupancy probabilities (see Section 1.3.1 for more detail) up to age 90 for lives healthy at age 60, with high (m = 1) and low (m = 0.25) relative risks, respectively. Each shows:

- 1. Occupancy probabilities in respect of each genotype (with  $\varepsilon 2/\varepsilon 2$  and  $\varepsilon 2/\varepsilon 3$  combined).
- 2. Occupancy probabilities calculated by aggregating all the genotypes in the model. In the notation of equation (1.15) the probability of being in state j (j = 1, 2, 3, 4) at age 60 + t is  $\sum_{i=1}^{i=5} p_{60}^i i P_{60\,60+t}^{1j}$ , where the sum is over all genotypes. These are labelled 'Aggregated Genotypes'.
- 3. Occupancy probabilities based on the aggregate incidence of AD,  $\mu_{x+t}^{AD}$ . These are labelled 'Aggregate Model'.

With high relative risks (m = 1), the effect of the  $\varepsilon 4/\varepsilon 4$  allele is clear; AD cases rise to a peak in the early 70s, by which time over 10% of the original cohort are in one of the AD states, and then fall away. A similar but smaller effect can be seen for the  $\varepsilon 3/\varepsilon 4$  genotype. With low relative risks (m = 0.25) these features are all diminished; in particular the peaks in the early 70s disappear.

For males and females with low relative risks (Figures 1.12 and 1.14) the aggregated results from the genetic model are very close to the results from the aggregate (population) model. For females with high relative risks, the rate of onset of AD seems to be too low at younger ages and too high at older ages. Also of interest are prevalence rates, namely the proportion of those alive at every age who are in each of the three live states. Figures 1.15 and 1.16 show these for females (for m = 1 and m = 0.25, respectively), and Figures 1.17 and 1.18 show them for males (for m = 1 and m = 0.25, respectively), including, for convenience, the two AD states combined.

The most striking feature is the prevalence of AD in respect of the  $\varepsilon 4/\varepsilon 4$  genotype under high relative risks (Figure 1.15); it increases almost linearly. Again, for males and females the aggregated results from the genetic model are quite close to those from the aggregate model. Moreover, they fall within the range of prevalence rates actually observed. Breteler *et al.* (1992) cite the following rates: 47.2% at ages 85 and over (Evans *et al.*, (1989)); 31.7% at ages 85 and over (Pfeffer *et al.*, 1987); and 28.0% at ages 90 and over (O'Connor *et al.*, 1989); some other studies gave lower figures.

So far, it has been assumed that the gene frequencies given by Farrer *et al.* (1997) are appropriate for age 60. They will change with age, as higher-risk genotypes die more quickly. In order to consider entrants (to a study, or into insurance) at ages over 60, gene frequencies at older ages need to be estimated. Table 1.9 shows estimates of the gene frequencies in respect of lives unaffected by AD at ages 65, 70 and 75. Using the notation of equation (1.15), these are given by:

$$p_{60+t}^{i} = \frac{p_{60}^{i} P_{60\,60+t}^{11}}{\sum_{j=1}^{j=5} p_{60}^{j} P_{60\,60+t}^{11}}.$$
(1.17)

These are not the gene frequencies in respect of the whole population; lives alive but who have AD are omitted (as is appropriate for insurance applications). Nor are they the gene frequencies in respect of the healthy population; lives with disabilities other than AD are included.

Gene frequencies in the whole population at older ages can also be estimated, as:

$$\frac{p_{60}^{i}(^{i}P_{60\,60+t}^{11} + ^{i}P_{60\,60+t}^{12} + ^{i}P_{60\,60+t}^{13})}{\sum_{j=1}^{j=5} p_{60}^{j}(^{j}P_{60\,60+t}^{11} + ^{j}P_{60\,60+t}^{12} + ^{j}P_{60\,60+t}^{13})}$$
(1.18)

but these are not so relevant for insurance applications.



Figure 1.11: Occupancy probabilities for females, healthy at age 60, high relative risks (m = 1).



Figure 1.12: Occupancy probabilities for females, healthy at age 60, low relative risks (m = 0.25).



Figure 1.13: Occupancy probabilities for males, healthy at age 60, high relative risks (m = 1).



Figure 1.14: Occupancy probabilities for males, healthy at age 60, low relative risks (m = 0.25).



Figure 1.15: Prevalence rate of Alzheimer's disease for females healthy at age 60, high relative risks (m = 1).



Figure 1.16: Prevalence rate of Alzheimer's disease for females healthy at age 60, low relative risks (m = 0.25).



Figure 1.17: Prevalence rate of Alzheimer's disease for males healthy at age 60, high relative risks (m = 1).



Figure 1.18: Prevalence rate of Alzheimer's disease for males healthy at age 60, low relative risks (m = 0.25).

#### **1.7** Summary and Discussion

In this chapter I have specified and calibrated a simple continuous-time Markov model of AD allowing for the variability of the APOE gene, suitable for use in insurance and other applications — I will use it in the next chapter to investigate the potential costs of adverse selection in the long-term care insurance market. While much uncertainty remains, certain features of the model ought to be robust. Whatever reduction in relative risks was used, the genotype-specific incidence rates of AD were also adjusted so that the aggregated (population) incidence rates were close to those actually observed. The latter is one of the few reasonably reliable benchmarks available. Further, the model produces prevalence rates of AD that fall within the range of those observed in many studies.

As well as the intensities, APOE gene frequencies at ages up to 75 have been estimated, in respect of lives unaffected by AD at these ages. These are needed in (for example) insurance applications.

#### 1.7.1 Discussion

- 1. The model specification is dictated entirely by the events studied in the medical and epidemiological literature, and not by the events that might be of interest in any particular application. If it is the case that actuarial models might, in future, need to incorporate more medical detail, it would be very useful to try to collaborate with medical and other researchers.
- 2. The published conclusions of medical papers are usually in the form of summary statistics (means, medians, odds ratios and confidence intervals) and if age-related outcomes are shown they are usually in the form of graphs. These are not ideal for actuarial use. AD is a major condition, much studied, but some crude assumptions had to be made in order to calibrate the model from published data only. There must be many medical data sets that could furnish age-related estimates of incidence rates, if only they could be re-analysed. Again, closer collaboration between actuaries and other researchers would be valuable.

Table 1.9:	Frequencies of APOE ge	enotypes among l	ives free of Alzhei	mer's disease
at ages 65,	, 70 and 75, estimated by	solving the Kolm	nogorov equations	forward from
age 60.				

	Proportion		Gene	Frequenc	eies in AI	D-free pop	pulation
	of relative						$\varepsilon 2/\varepsilon 2$ &
Gender	risk, $m$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$
M & F	1.00	65	0.0168	0.2103	0.6114	0.0258	0.1357
		70	0.0151	0.2055	0.6168	0.0251	0.1374
		75	0.0133	0.1978	0.6246	0.0241	0.1403
	0.50	65	0.0174	0.2115	0.6100	0.0259	0.1353
		70	0.0164	0.2090	0.6128	0.0255	0.1362
		75	0.0154	0.2050	0.6170	0.0250	0.1377
	0.25	65	0.0177	0.2121	0.6092	0.0259	0.1351
		70	0.0172	0.2109	0.6106	0.0258	0.1355
		75	0.0166	0.2088	0.6128	0.0255	0.1363
$\mathbf{F}$	1.00	65	0.0171	0.2097	0.6116	0.0257	0.1358
		70	0.0159	0.2041	0.6176	0.0247	0.1377
		75	0.0145	0.1951	0.6263	0.0232	0.1409
	0.50	65	0.0175	0.2112	0.6101	0.0258	0.1354
		70	0.0168	0.2081	0.6133	0.0253	0.1364
		75	0.0160	0.2033	0.6181	0.0245	0.1381
	0.25	65	0.0177	0.2119	0.6093	0.0259	0.1351
		70	0.0174	0.2104	0.6110	0.0256	0.1357
		75	0.0169	0.2078	0.6135	0.0252	0.1365
Μ	1.00	65	0.0169	0.2120	0.6094	0.0260	0.1357
		70	0.0153	0.2110	0.6105	0.0258	0.1373
		75	0.0138	0.2097	0.6105	0.0257	0.1403
	0.50	65	0.0175	0.2125	0.6088	0.0260	0.1352
		70	0.0168	0.2120	0.6094	0.0259	0.1359
		75	0.0160	0.2115	0.6094	0.0258	0.1372
	0.25	65	0.0177	0.2126	0.6086	0.0260	0.1350
		70	0.0174	0.2124	0.6089	0.0259	0.1354
		75	0.0170	0.2122	0.6089	0.0259	0.1360

- 3. Another common type of medical statistic is prevalence rates. The difference between prevalence rates and incidence rates is exactly the difference between the Manchester Unity approach to modelling Permanent Health Insurance, and the multiple-state model approach. Prevalence rates are often easier to estimate, as they can be based on census-type surveys, but it would be helpful if the greater versatility of incidence rates (transition intensities) was more widely appreciated.
- 4. As a consequence of fitting the intensities using published summary statistics, it is impossible to estimate even crude confidence intervals for them. In any application, therefore, sensitivity analysis is especially important.
- 5. Several epidemiological authors have suggested the use of individual patient data rather than summary or published data, partly to avoid publication bias in meta-analyses. Useful references are Friedman, Furberg & DeMets (1998), Green, Benedetti & Crowley (1997) and Piantadosi (1997).
- 6. It is now about six years since the rôle of the APOE gene in AD was confirmed. Since then, the gene has been intensively studied, to the point that metaanalyses including thousands of lives have been published. Even so, little is known about population risk, and data are very scarce in places, so that:
  - (a) the relative risks were reduced in a rather arbitrary way to allow for the selectiveness of case-based studies; and
  - (b) the relative risks for males are suspect, with the  $\varepsilon 2$  allele apparently conferring such strong protection that carriers of the  $\varepsilon 4$  allele are not necessarily at higher risk overall.

If this is typical, it seems likely that the time between the identification of a gene disorder, and the assessment of its impact on insurance, will be of the order of ten years.

7. Despite the fact that AD is a much-studied condition, many studies reach conflicting conclusions. When setting up an actuarial model, it is necessary to consider the body of medical research in its entirety (hence the large number of references). Inevitably, some studies must be chosen as the basis of the model, but to confine one's attention only to these risks overlooking important features or sources of variation, and could impair the credibility of the results in the eyes of medical experts.

- 8. APOE is a relatively simple gene to consider, since it has only three relevant alleles, hence six genotypes. Other genes are more complex; for example, the BRCA1 gene (predisposing to breast and ovarian cancer) has several hundred known mutations, some of which have only been observed in a single family.
- 9. Human genetics is developing at a great speed. This work started in late 1997 and was finished in late 1999, in which time the volume of papers on AD and the APOE gene increased greatly, as the references show. It is very likely that assessments of the impact of specific genetic tests on insurance will have to be revisited quite frequently, if they are to remain credible.

Points 1 to 3 above suggest ways in which medical data might be made more useful for actuarial models, but there is no *a priori* reason why medical studies should be planned with that in mind. However, actuarial models derived from insurance practice, capable of dealing with fairly general payments while in different states or on transition between different states, could make a useful contribution to health economics, and it would be helpful to pursue collaborations from that point of view.

In the next chapter I apply the model specified and parameterized in this chapter to estimating the costs arising, in respect of AD, in a long-term care insurance contract and then to the question of the potential for adverse selection in the longterm care insurance market.

## Chapter 2

# The Potential for Adverse Selection in Long-Term Care Insurance from Individuals Knowing their APOE Genotype

#### 2.1 Introduction

The aims of this chapter are:

- to apply the model of Alzheimer's disease, discussed in the previous chapter, to a single premium LTC insurance contract, and to carry out sensitivity analyses;
- 2. to model the possible effect of adverse selection on LTC insurance, especially with respect to the size of the market; and
- 3. to consider the costs of a comprehensive retirement package providing both pension and LTC cover.

In Section 2.2, I describe LTC insurance contracts. Then in Section 2.3, I introduce payment streams to the Alzheimer's disease model (introduced in Section 1.3) and extend the statistical framework discussed in Section 1.3.1 to calculating moments of the present value of these payment streams. In Section 2.4, the basic costs of single premium LTC contracts covering AD alone are calculated, including:

- 1. the costs using the aggregate incidence of AD; this represents the pricing assumption that would be appropriate in the absence of any genetic markers for AD;
- 2. the costs in respect of each APOE genotype (Section 2.4.1) these allow underwriting ratings to be illustrated; and
- for comparison with (1) above, the average costs assuming no adverse selection (a weighted average of (2) above).

Then, in Section 2.5 the sensitivity of the results to the main model assumptions is considered. Section 2.6 is the main part of the paper; I estimate the possible costs of adverse selection if the results of genetic tests for the APOE gene are known to applicants but are not used in underwriting.

The model can be used to model benefits other than LTC. In Section 2.7, I consider the combined costs of LTC insurance and a pension. In particular, I look at whether or not there is any offset between the costs of the two benefits, so that the combined costs are more stable than either on its own. If this were so, there would be scope for including both in a comprehensive retirement package. Finally, I provide conclusions in Section 2.8.

It is important to emphasise that little reliance should be placed on the absolute costs shown here, in view of the unreliability or scarcity of much of the data, the use of studies of many different populations, the use of institutionalisation as a proxy for an insurance claim, and the early stage of research into the genetics of AD. These costs should not be used to price any genuine LTC contract. Nevertheless, the results do give useful quantitative information on the relative costs of adverse selection.

#### 2.2 Long-Term Care Insurance

Despite the low volume of business in force, there is great variety among LTC or related products. Dullaway & Elliott (1998) described the current market in the U.K., and the main data sources available for pricing and reserving. Another useful reference is Humble & Ryan (1998), which dealt with the closely related problems of Continuing Care Retirement Communities (CCRCs). Only the broad features of long-term care policies are described here, abstracted as the basis for the model; the model itself is flexible enough to be adapted to many product designs — the above references provide more specific details.

This research deals only with pre-funded LTC products, under which an ordinarily healthy person pays regular or single premiums in return for the entitlement to benefit when the need arises. Immediate care annuities, under which a person in immediate need of care buys an annuity to cover its cost, are not considered.

When a claim arises, the insurer will make regular payments to cover the cost of care, either at home or residential, up to some maximum amount per year (the annual sum assured). There is usually a limit to the annual sum assured that can be chosen by the policyholder, since the cost of reasonable care is bounded; this could affect the costs of adverse selection. The level of benefit payment sometimes depends on the severity of the claim (see Section 3.2) and usually is indexed.

Claims can arise:

- on failure of a given number of Activities of Daily Living (ADLs) see Section
   3.2 for more detail; or
- 2. on reaching a certain level of cognitive impairment (often because of AD), resulting in a need for continual care or supervision.

The term 'cognitive impairment' covers AD, which accounts for by far the majority of cases, and other forms of mental deterioration, chiefly vascular in origin (for example, arising from strokes).

LTC insurers will usually measure cognitive impairment using a form of minimental-state examination (see Dullaway & Elliott (1998) for examples) plus evidence from relevant health professionals. This is not a precise process; in particular it can be hard to distinguish AD from other dementias without a post-mortem examination.

# 2.3 Inclusion of Payment Streams in the Alzheimer's Disease Model

In this section I introduce payment streams to the model and extend the statistical framework introduced in Section 1.3.1 to calculating moments of the present value of these payment streams.

I use the same notation here as was introduced in Section 1.3.1, with the exception that I will ignore genotypes in the superscript of the transition intensities. I first look at how a claim will be represented in the Alzheimer's disease model, shown in Figure 1.1.

A major event in the progression of AD, studied by researchers, is institutionalisation. I use it as the best available proxy for the start of an insurance claim: It might be reasonable, because:

- 1. if full-time care is needed, a claim will almost certainly succeed; and
- 2. carers of AD patients with LTC cover may lack incentives to undertake care at home after a claim begins.

It should be noted that point 2 above could cause the insured experience to differ from that of the general population, reducing the amount of time for which an insured AD patient is cared for outside an institution.

Define the rate of benefit payable to a life age x + t in the *i*th subgroup while in the *j*th state of the model as  $b_t^{ij}$ . Then the LTC benefit in this model, arising from AD only, can be represented by  $b_t^{ij} = 1$  if j = 3, for all *i* (level benefit) or by  $b_t^{ij} = e^{\delta_b t}$  if j = 3, for all *i* (benefit escalating at rate  $\delta_b$  per year). The latter is the default, since this is a feature of most LTC policies (Dullaway & Elliot, 1998). I only use level benefits in some sensitivity tests. Norberg (1995) derived a system of differential equations for moments of prospective reserves in a very general Markov model, with both assurance-type and continuous annuity-type benefits, generalising Thiele's equations for expected values. I now summarise the results from his paper.

Using the notation introduced in Section 1.4.3, the payment function is assumed to be of the form:

$$dB_t = \sum_j I_j(t) \, db_t^j + \sum_{j \neq k} b_t^{jk} \, dN_{jk}(t)$$
(2.19)

where  $b_t^j$  and  $b_t^{jk}$  are payment functions specifying payment due during sojourns in state j at time t and upon transition from state j to state k at time t, respectively. It is assumed that jumps in the payment function,  $b_t^j$ , can only occur at a finite set of times,  $\mathbf{D} = \{t_0, t_1, \ldots, t_m\}$ .

The force of interest at time t is denoted  $\delta_t$  and the discount function at time t,  $v_t$  is defined by:

$$v_t = e^{-\int_0^t \delta_s ds} \tag{2.20}$$

and  $dv_t = -v_t \, \delta_t \, dt$ . The present value, at any time  $t \in [0, n]$ , of future benefits less premiums is then:

$$\frac{1}{v_t} \int_t^n v_\tau \, dB_\tau \tag{2.21}$$

where  $v_{\tau}/v_t$  is the value at time t of a unit due at time  $\tau$ . The qth moment of the present value in equation 2.21, conditional on the policy being in state j at time t:

$$V_t^{(q)j} = \mathbf{E}\left[\left(\frac{1}{v_t} \int_t^n v_\tau \, dB_\tau\right)^q \,\middle|\, I_j(t) = 1\right]$$
(2.22)

can be shown, under the above assumptions, to satisfy:

$$\frac{d}{dt}V_t^{(q)j} = \left(q\delta_t + \mu_t^{j}\right)V_t^{(q)j} - qb_t^j V_t^{(q-1)j} - \sum_{k\neq j}\mu_t^{jk}\sum_{r=0}^q \binom{q}{r}\left(\frac{q}{r}\right)\left(b_t^{jk}\right)^r V_t^{(q-r)k} \quad (2.23)$$

valid on  $(0, n) \setminus \mathbf{D}$  and subject to the boundary conditions:

$$V_{n-}^{(q)j} = \sum_{r=0}^{q} \begin{pmatrix} q \\ r \end{pmatrix} \left(\Delta b_n^j\right)^r V_n^{(q-r)j}$$
(2.24)

where  $\delta_t^j$ ,  $b_t^j$ ,  $b_t^{jk}$  and  $\mu_t^{jk}$  are assumed to be piecewise continuous and  $\mu_t^{j\cdot} = \sum_{k \neq j} \mu_t^{jk}$ . The central moments,  $m_t^{(q)j}$ , are then defined by:

$$m_t^{(q)j} = \begin{cases} V_t^{(1)j} & \text{if } q = 1\\ \sum_{p=1}^q \binom{q}{p} (-1)^{q-p} V_t^{(p)j} \left(V_t^{(1)j}\right)^{q-p} & \text{if } q > 1 \end{cases}$$
(2.25)

Using standard numerical methods, these equations are easily solved recursively, to give the qth moments of the payment streams in the model. I used the same numerical routine as for calculating the occupancy probabilities (see Section 1.3.1) — a 4th order Runge-Kutta algorithm (Press *et al.*, 1993) with step size 0.0005 years. They are solved backwards, the boundary condition being that reserves in all states are 0 at the terminal age (120 in this case).

## 2.4 The Costs of Alzheimer's Disease in LTC Insurance

I assume that the LTC contract has a single premium paid at outset, so the only policy cash-flows thereafter are the benefits. Benefits are payable continuously, only while institutionalised with AD. The quantum of benefit is £1 per annum at inception of the policy, increasing continuously at rate  $\delta_b$  per year. I use a force of interest  $\delta = 0.05$  throughout. The present value of the benefit is the random variable whose expected value and higher moments are obtained by solving Norberg's equations, the expected value being the relevant quantity for use in the traditional actuarial equation of value.

Table 2.10 gives the first three moments (mean, variance and skewness, denoted q = 1, 2 and 3 respectively) of the present value of AD claims payments, for lives entering at ages 60, 65, 70 and 75, under the following assumptions:

- 1. the rate of onset of AD is the aggregate rate,  $\mu_{x+t}^{AD}$ , men and women combined (equation (1.3));
- 2.  $\mu_{x+t}^{23}$  and  $\mu_{x+t}^{24}$  are as given in Table 1.3;
- 3.  $\mu_{x+t}^{34}$  is the mean value given in Table 1.3;

Table 2.10: Mean, variance and skewness (q = 1, 2 and 3) of the present value of AD claims costs, unit benefit increasing continuously ( $\delta_b = 0.05$ ), using the aggregate incidence rate of AD.

State at start			Entry	v Age	
of contract	q	60	65	70	75
	1	1.1534	1.1634	1.1739	1.1738
1	2	6.431	6.374	6.238	5.975
	3	52.312	50.591	47.668	43.158
	1	5.2367	4.9939	4.6694	4.2541
2	2	25.898	23.438	20.579	17.427
	3	223.272	185.912	148.209	112.619
	1	5.4984	5.3574	5.1578	4.8850
3	2	28.187	26.115	23.520	20.446
	3	259.646	222.571	182.189	141.315

Table 2.11: Expected present values (EPV) of unit benefit increasing continuously  $(\delta_b = 0.05)$ , depending on the incidence of AD after age 90.

				Overstat	tement of the
	$\mu_{x+t}^{12}$ a	after age	90	EPV of b	penefits under
	(A)	(B)	(C)	(A) con	npared with:
Age	$\operatorname{Gompertz}$	Level	Nil	(B)	(C)
60	1.1534	1.0584	0.8479	9.0%	36.0%
65	1.1634	1.0650	0.8468	9.2%	37.4%
70	1.1739	1.0674	0.8358	10.0%	40.5%
75	1.1738	1.0581	0.8016	10.9%	46.4%

- 4. the force of mortality for 'healthy' lives,  $\mu_{x+t}^{14}$ , is 65% of  ${}^{AF80}\mu_{x+t}$ , as the majority of elderly people will be women; and
- 5. benefits increase continuously, with force  $\delta_b = 0.05$ , representing indexation to earnings.

Table 2.10 also shows moments for lives starting in state 2 (that is, buying insurance just after the onset of AD) and in state 3 (that is, buying insurance just after institutionalisation). The main feature is that the expected present values (EPVs) are not strongly dependent on age at entry, especially for lives healthy at outset. The increasing incidence of AD with age appears to counter the effect of the shorter future lifetime.

In Section 1.4.2, it was noted that the data for rate of onset of AD did not extend beyond age 90, and that the Gompertz curve was extrapolated. This arguably

Table 2.12: EPV of unit benefit increasing continuously ( $\delta_b = 0.05$ ), with genderspecific mortality and incidence of AD.

		EPV of Benefits for						
		Entry at Age						
$\mu_{x+t}^{14}$	$\mu_{x+t}^{12}$	60	65	70	75			
$0.65^{AF80}\mu_{x+t}$	${\it Female}\mu^{AD}_{x+t}$	0.8268	0.8425	0.8631	0.8871			
$0.65 {}^{AM80}\!\mu_{x+t}$	${}^{Male}\!\mu^{AD}_{x+t}$	0.4370	0.4452	0.4531	0.4584			
$0.65^{AF80}\mu_{x+t}$	${}^{Male}\!\mu^{AD}_{x+t}$	0.8039	0.8089	0.8123	0.8107			
$0.65 {}^{AM80}\!\mu_{x+t}$	$Female \mu_{x+t}^{AD}$	0.4176	0.4345	0.4566	0.4830			

overstates the incidence of AD, although the evidence is not clear on this point. Column (B) in Table 2.11 shows by how much this would overstate the expected present value of the increasing benefit, if the true incidence rate levelled off at age 90 or was zero.  $\mu_{90+t}^{12} = \mu_{90}^{12}$  almost certainly understates the incidence of AD, so Table 2.11 shows that the Gompertz assumption has no great effect. The last column of the table shows that a significant proportion of the AD costs in the model arise from cases at very old ages; the fact that the rate of increase of the benefit is equal to the force of interest is part of the reason.

Next, I look at the different costs in respect of men and women. Two factors are relevant:

- 1. the baseline mortality  $(\mu_{x+t}^{14})$ , which I take to be 65% of AM80 ultimate and 65% of AF80 ultimate; and
- 2. the gender specific incidence of AD  $(\mu_{x+t}^{12})$ , which I take from equations 1.4 and 1.5.

The first two lines of Table 2.12 compare the EPVs of £1 p.a. sums assured at outset, increasing at  $\delta_b = 0.05$ , for females and males entering at different ages. As expected, the EPVs are higher for females. Taking these as a starting point, the third line shows the effect of changing only the rate of onset of AD, from female to male; the effect is not substantial, and this is why the aggregate rate of onset is used for all this work. The fourth line shows the effect of changing only the baseline mortality, from female to male; this reduces the costs very substantially. Females' longer lives mean that they are more likely to become institutionalised, and then live longer while institutionalised.

The sensitivity of the LTC costs to overall longevity is expected, because mortality while institutionalised is modelled by a constant addition to the assumed overall mortality. The same result will arise, in the model, if mortality at old ages continues to fall in the future as it has in the past. However, the extent to which such falls will be accompanied by a different pattern of infirmity cannot be predicted. Longer lifetimes could result from:

- people remaining healthy for longer, followed by a swifter decline (so-called 'compression of morbidity'); or
- 2. people being kept alive for longer while in decline.

These could have dramatically different implications for LTC costs. Here, it is worth noting that no attempt has been made to model changes in the pattern of infirmity.

## 2.4.1 Long-Term Care Insurance Costs and APOE Genotype

Recall that the incidence rates of AD were estimated to be consistent with 100%, 50% and 25% of the absolute relative risks implied by Farrer *et al.* (1997), to allow for the selective effect of using case-based data rather than population data (see Section 1.5). In the following, these three cases are represented by the parameter m = 1.0, 0.5 or 0.25.

Table 2.13 shows the EPV of the £1 p.a. benefit increasing continuously with force  $\delta_b = 0.05$  for each genotype, and averages weighted by gene frequencies, for males and females separately and in aggregate. The label *i* is used to indicate genotype: i = 1 represents  $\varepsilon 2/\varepsilon 2$  and  $\varepsilon 2/\varepsilon 3$ , i = 2 represents  $\varepsilon 3/\varepsilon 3$ , i = 3 represents  $\varepsilon 2/\varepsilon 4$ , i = 4 represents  $\varepsilon 3/\varepsilon 4$  and i = 5 represents  $\varepsilon 4/\varepsilon 4$ . Summation over *i* denotes summation over all genotypes. Thus, define  $p_x^i$  to be the population frequency of genotype *i* at age *x*, and  $\sum_i p_x^i = 1$ . Also, define  $C_x^i$  to be the EPV of the unit benefit, for a life with genotype *i* buying insurance at age *x*; these are the EPVs given in Table 2.13. The weighted average EPVs in Table 2.13 are then:

$$\sum_{i} p_x^i C_x^i. \tag{2.26}$$

I assume that the gene frequencies from Farrer *et al.*, (1997) (Section 1.5) apply at age 60, denoted  $p_{60}^i$ , and calculate the frequencies at higher ages by solving the Kolmogorov forward equations for  ${}^{i}P_{60\,60+t}^{11}$ , the probability that a healthy life age 60 with genotype *i* is still healthy at age 60 + *t*. Then, the  $p_{60+t}^i$  are given by Equation 1.17 and the average costs for entrants at age 60 + *t* is:

Average EPV at age 
$$60 + t = \frac{\sum_{i} p_{60}^{i} i P_{60\,60+t}^{11} C_{60+t}^{i}}{\sum_{i} p_{60}^{i} i P_{60\,60+t}^{11}}.$$
 (2.27)

The EPVs for males and females together are similar to those for females only. Males have significantly lower costs at all ages and levels of relative risk (values of m). This is largely because of the difference in baseline mortality.

The EPVs of the individual genotypes depend strongly on the level of relative risk, but the EPVs averaged over all the genotypes do not. This is because the modelled relative risks were adjusted to reproduce (approximately) the aggregate incidence of AD (see Section 1.5). Note some other consequences of adjusting the relative risks to reproduce the aggregate incidence of AD:

- 1. Increasing the level of relative risks for high-risk genotypes means decreasing the relative risks for low-risk genotypes. Since the  $\varepsilon 3/\varepsilon 4$  genotype is lowrisk for males (the relative risks fall below 1.0 eventually) the EPVs for this genotype are lower than those for the  $\varepsilon 3/\varepsilon 3$  genotype, especially for m = 1and for older entrants. It is doubtful that this feature is genuine, but it is not very great for m = 0.25, so I have not tried to eliminate it.
- 2. The EPVs of the high-risk genotypes decrease with age, while those of low-risk genotypes increase with age.

Table 2.13 also shows what might be the effect on underwriting, if LTC insurers were allowed to use APOE test results. Increased premiums might be indicated based on the excess of the costs for the  $\varepsilon 4/\varepsilon 4$  and  $\varepsilon 3/\varepsilon 4$  genotypes over total average costs. However, it is not correct to take the final column of Table 2.13 as the total

	Proportion		]	EPV of b	enefits fo	or genoty	ре	
	of relative						$\varepsilon 2/\varepsilon 2 \&$	Average
Gender	risk, $m$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV
M & F	1.00	60	2.3619	1.7056	1.1003	1.5955	0.7816	1.223
		65	2.2063	1.6784	1.1101	1.5780	0.7866	1.216
		70	1.9539	1.6180	1.1188	1.5025	0.7900	1.198
		75	1.6483	1.5184	1.1215	1.3662	0.7888	1.166
$\mathbf{F}$	1.00	60	2.0878	1.7565	1.0609	1.7065	0.7106	1.198
		65	1.9660	1.7233	1.0706	1.6799	0.7159	1.190
		70	1.7673	1.6521	1.0793	1.5734	0.7204	1.170
		75	1.5243	1.5370	1.0825	1.3873	0.7211	1.134
М	1.00	60	1.6233	0.6499	0.7728	0.4818	0.3001	0.691
		65	1.4866	0.6363	0.7897	0.4683	0.3042	0.695
		70	1.2788	0.6076	0.8065	0.4214	0.3075	0.693
		75	1.0652	0.5591	0.8183	0.3341	0.3086	0.683
M & F	0.50	60	1.8427	1.4510	1.1234	1.3874	0.9746	1.194
		65	1.7510	1.4402	1.1332	1.3823	0.9823	1.195
		70	1.6034	1.4104	1.1419	1.3449	0.9889	1.190
		75	1.4282	1.3562	1.1442	1.2722	0.9899	1.174
$\mathbf{F}$	0.50	60	1.6933	1.4958	1.1081	1.4649	0.9427	1.189
		65	1.6236	1.4808	1.1179	1.4537	0.9507	1.189
		70	1.5094	1.4433	1.1265	1.3966	0.9577	1.183
		75	1.3702	1.3782	1.1291	1.2932	0.9599	1.164
М	0.50	60	1.1156	0.6506	0.7006	0.5909	0.5068	0.669
		65	1.0530	0.6534	0.7164	0.5949	0.5177	0.679
		70	0.9571	0.6504	0.7326	0.5861	0.5290	0.687
		75	0.8603	0.6378	0.7449	0.5605	0.5379	0.691
M & F	0.25	60	1.5154	1.3012	1.1309	1.2669	1.0590	1.169
		65	1.4689	1.3002	1.1408	1.2690	1.0679	1.174
		70	1.3919	1.2885	1.1495	1.2536	1.0755	1.176
		75	1.2989	1.2611	1.1517	1.2170	1.0773	1.169
$\mathbf{F}$	0.25	60	1.4447	1.3362	1.1309	1.3189	1.0505	1.175
		65	1.4106	1.3327	1.1408	1.3174	1.0595	1.180
		70	1.3519	1.3164	1.1495	1.2908	1.0675	1.181
		75	1.2777	1.2823	1.1517	1.2368	1.0697	1.172
М	0.25	60	0.8778	0.6492	0.6723	0.6228	0.5828	0.658
		65	0.8536	0.6585	0.6877	0.6327	0.5960	0.671
		70	0.8134	0.6654	0.7035	0.6372	0.6096	0.683
		75	0.7720	0.6662	0.7159	0.6324	0.6206	0.691

Table 2.13: EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) by genotype, and averages over genotypes.

average LTC costs; it represents AD-related costs alone, and these are only about 25% - 33% of the total claim numbers (Watson, 1998) and perhaps 40% - 50% of the claim costs. For example, the percentage rating indicated for an  $\varepsilon 4/\varepsilon 4$  female aged 60, with m = 1, would not be given by:

$$\frac{2.0878 - 1.198}{1.198} = 0.742 \text{ or } 74\%$$

but would be closer to:

$$\frac{2.0878 - 1.198}{2 \times 1.198} = 0.371 \text{ or } 37\%$$

if total LTC costs were about twice AD-related LTC costs. Table 2.14 shows approximate underwriting ratings on that basis. Note that:

- 1. Ratings decrease with age, because the relative risks all decrease with age.
- 2. Ratings for  $\varepsilon 4/\varepsilon 4$  males exceed those for  $\varepsilon 4/\varepsilon 4$  females. This is because overall costs for males are significantly lower, but this in turn is because the  $\varepsilon 2/\varepsilon 4$  and  $\varepsilon 3/\varepsilon 4$  genotypes are in fact low-risk for males at most ages. It should not be considered that this is evidence that ratings for males should in fact be higher (see Section 2.6.1) and the following comments are based on females only.
- 3. If the odds ratios in Farrer *et al.* (1997) are about right (m = 1), then ratings could be up to +50% for  $\varepsilon 4/\varepsilon 4$  and +25% for  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 2/\varepsilon 4$ . Most insurers would probably charge extra premiums for these risks.
- 4. If, however, the relative risks in the population are much lower (m = 0.25), then ratings could be up to +15% for ε4/ε4 and +7% for ε3/ε4 and ε2/ε4. Most insurers would probably ignore the latter.

### 2.5 Sensitivity Analysis

In this section tests of sensitivity to key model assumptions are carried out. I consider:

Table 2.14: Approximate underwriting ratings (equivalent to percentage extra premiums) for the  $\varepsilon 4/\varepsilon 4$  and female  $\varepsilon 3/\varepsilon 4$  &  $\varepsilon 2/\varepsilon 4$  genotypes, assuming total LTC insurance costs are twice AD-related costs. Unit benefit increasing continuously ( $\delta_b = 0.05$ ).

Proportion	ortion Rating factor for sex and genotype									
of relative		Ma	le & Fen	nale		Female		Male		
risk, $m$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 4/\varepsilon 4$		
		%	%	%	%	%	%	%		
1.00	60	47	20	15	37	23	21	67		
	65	41	19	15	33	22	21	57		
	70	32	18	13	26	21	17	42		
	75	21	15	9	17	18	11	28		
0.50	60	27	11	8	21	13	12	33		
	65	22	10	8	18	12	11	28		
	70	19	9	7	14	11	9	20		
	75	11	8	4	9	9	6	12		
0.25	60	15	6	4	11	7	6	17		
	65	13	5	4	10	6	6	14		
	70	9	5	3	7	6	5	10		
	75	6	4	2	5	5	3	6		

1. the use of institutionalisation as a proxy for the start of a LTC claim;

- 2. the rate of benefit increases;
- 3. the level of baseline mortality;
- 4. mortality in state 2 (after onset of AD but before institutionalisation); and
- 5. the addition to baseline mortality in state 3 (lives institutionalised from AD).

For brevity, I look only at males and females separately, at entry ages 60 and 70, and at levels of relative risk m = 1.00 and m = 0.25.

Percentage changes to the EPVs in Table 2.13 are reported. The changes in the EPVs averaged over all genotypes indicate the overall uncertainty of AD costs in LTC insurance, while the relative changes in the EPVs of high-risk and low-risk genotypes indicate uncertainty about the potential for adverse selection.

It has been assumed that lives with AD do not claim LTC benefits before becoming institutionalised because, of all the events studied by epidemiologists, institutionalisation is the best proxy for the start of a claim. In practice, a claim might be

Table 2.15: EPV of benefits commencing 1 year before institutionalisation (w = 1 years) as a % of EPV of benefits commencing on institutionalisation (w = 0 years), for unit benefit increasing continuously ( $\delta_b = 0.05$ ) by genotype and averages over genotypes.

			EPV of benefits with $w = 1$ years as a %						
	Proportion		(	of EPV o	f benefits	s with $w$	= 0 years f	for	
	of relative						$\varepsilon 2/\varepsilon 2$ &	Average	
Gender	risk, $m$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV	
			%	%	%	%	%	%	
$\mathbf{F}$	1.00	60	123.96	125.54	129.17	125.28	130.33	127.82	
		70	126.03	126.88	129.91	126.72	131.13	128.94	
$\mathbf{F}$	0.25	60	126.66	127.54	128.90	127.46	129.18	128.51	
		70	128.18	128.57	129.64	128.55	129.93	129.36	
Μ	1.00	60	125.11	127.91	130.27	125.66	131.65	129.58	
		70	128.19	129.84	131.55	127.26	133.18	131.17	
Μ	0.25	60	128.43	130.34	130.73	130.33	131.09	130.63	
		70	130.77	131.81	132.00	131.80	132.39	131.98	

made sooner (later is less likely). The possibility that claims precede institutionalisation by w years, or start immediately upon entry if a life is institutionalised less than w years after entry, can be considered by paying a lump-sum benefit of:

$$b_{x+t}^{i23} = e^{\delta_b \max(t-w,0)} \int_{0}^{\min(t,w)} e^{\delta_b s + \delta(\min(t,w)-s)} ds$$
(2.28)

upon institutionalisation at age x + t.  $b_{x+t}^{i23}$  is the accumulated LTC benefit paid over w years preceding institutionalisation, or from outset. This method is only approximate:

- 1. it ignores the costs in respect of lives who die while in receipt of LTC benefit but before becoming institutionalised; and
- 2. it overstates the costs in respect of lives who become institutionalised within w years of the onset of AD

but these errors are unlikely to be significant. I consider w = 1 year in Table 2.15 and w = 2 years in Table 2.16.

The increases in costs range from 23% per additional year of benefit, in the case of high-risk genotypes, to 31% per additional year of benefit, in the case of low-risk

Table 2.16: EPV of benefits commencing 2 years before institutionalisation (w = 2 years) as a % of EPV of benefits commencing on institutionalisation (w = 0 years), for unit benefit increasing continuously ( $\delta_b = 0.05$ ) by genotype and averages over genotypes.

			EPV of benefits with $w = 2$ years as a %						
	Proportion		(	of EPV o	f benefits	s with $w$	= 0 years f	for	
	of relative						$\varepsilon 2/\varepsilon 2 \&$	Average	
Gender	risk, $m$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV	
			%	%	%	%	%	%	
$\mathbf{F}$	1.00	60	147.85	151.04	158.34	150.54	160.64	155.62	
		70	151.84	153.62	159.77	153.26	162.20	157.80	
$\mathbf{F}$	0.25	60	153.29	155.05	157.79	154.91	158.34	157.01	
		70	156.25	157.05	159.23	157.00	159.80	158.66	
М	1.00	60	150.11	155.76	160.52	151.28	163.27	159.12	
		70	156.04	159.50	163.00	154.25	166.26	162.22	
М	0.25	60	156.79	160.65	161.44	160.64	162.17	161.23	
		70	161.36	163.50	163.91	163.48	164.68	163.86	

genotypes, almost independent of age. Therefore, using institutionalisation as a proxy for the start of a claim is likely to understate costs quite significantly, but will not have much effect on underwriting, or the costs of adverse selection, unless the duration of an AD-related claim, prior to institutionalisation, is likely to be greater than that of other LTC insurance claims.

The force of benefit escalation of 0.05 represents indexation to earnings in a relatively low inflation environment, with force of interest  $\delta = 0.05$  also. Tables 2.17 and 2.18 give the EPVs of benefits increasing with force  $\delta_b = 0.025$ , and level benefits  $(\delta_b = 0.0)$ , respectively, as a percentage of the EPV of benefits with  $\delta_b = 0.05$ .

From Tables 2.18 and 2.17 it can be seen that:

- 1. claims inflation has a large impact on the EPV of benefits: with  $\delta_b = 0.025$ the EPVs of benefits are between 45% and 68% of those with  $\delta_b = 0.05$ , and with  $\delta_b = 0$ , between 22% and 49%.
- 2. claims inflation has a greater impact on low-risk genotypes ( $\varepsilon 3/\varepsilon 3$ ,  $\varepsilon 2/\varepsilon 2$  and  $\varepsilon 2/\varepsilon 3$ ) under all scenarios; the lower the claims inflation, the greater the difference between the EPVs of low-risk and high-risk groups, so the greater the scope for adverse selection.

			F	EPV of b	penefits v	with $\delta_b$ =	= 0.025 as	a %			
	Proportion		of EPV of benefits with $\delta_b = 0.05$ for								
	of relative						$\varepsilon 2/\varepsilon 2 \&$	Average			
Gender	risk, $m$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV			
$\mathbf{F}$	1.00	60	56.67	52.63	46.71	53.45	45.88	49.05			
		70	64.69	62.62	57.74	63.49	56.66	59.41			
F	0.25	60	51.22	49.18	46.98	49.47	46.74	47.65			
		70	60.65	59.76	58.04	60.02	57.75	58.51			
М	1.00	60	60.74	55.07	51.21	57.92	50.38	52.46			
		70	67.93	65.02	62.31	68.71	60.95	62.99			
М	0.25	60	54.80	51.48	50.75	51.55	50.45	50.98			
		70	63.82	62.23	61.83	62.35	61.46	61.92			

Table 2.17: EPVs of unit benefit increasing continuously ( $\delta_b = 0.025$ ) as a percentage of EPV of unit benefit increasing continuously ( $\delta_b = 0.05$ ) by genotype and averages over genotypes for benefits commencing on institutionalisation.

Table 2.18: EPVs of level unit benefit ( $\delta_b = 0$ ) as a percentage of EPV of unit benefit increasing continuously ( $\delta_b = 0.05$ ) by genotype, and averages over genotypes for benefits commencing on institutionalisation.

				EP	V of leve	el benefi <sup>.</sup>	ts as a $\%$	
	Proportion			of EPV	of benef	fits with	$\delta_b = 0.05$ d	for
	of relative						$\varepsilon 2/\varepsilon 2 \&$	Average
Gender	risk, $m$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV
			%	%	%	%	%	%
$\mathbf{F}$	1.00	60	33.99	29.28	23.00	30.19	22.28	25.51
		70	43.48	40.69	34.60	41.95	33.42	36.71
F	0.25	60	27.91	25.58	23.25	25.90	23.04	23.97
		70	38.28	37.09	34.94	37.47	34.61	35.54
Μ	1.00	60	38.69	31.89	27.53	34.90	26.82	28.96
		70	47.63	43.65	40.08	48.43	38.52	40.99
Μ	0.25	60	31.75	27.89	27.06	27.97	26.77	27.33
		70	42.15	40.04	39.49	40.22	39.05	39.62

Table 2.19: Sensitivity of the EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) to the mortality assumptions, for Females. Level of relative risk, m = 1.00.

			EF	V of be	nefits as	a $\%$ of	EPV of be	nefits		
Baseline	State 2		before adjusting mortality, for:							
Mortality	Mortality,						$\varepsilon 2/\varepsilon 2$ &	Average		
% AF80	% Baseline	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV		
			%	%	%	%	%	%		
100%	33.5%	60	77.78	72.65	62.32	73.70	59.98	66.26		
		70	73.80	71.03	62.53	71.85	59.98	65.34		
65%	65%	60	95.06	93.99	91.74	94.20	91.17	92.59		
		70	93.61	93.05	91.24	93.20	90.65	91.83		
65%	100%	60	90.56	88.60	84.65	89.00	83.73	86.16		
		70	87.89	86.88	83.76	87.17	82.80	84.79		
100%	100%	60	69.38	63.18	51.65	64.54	49.41	56.11		
		70	63.37	60.12	50.97	61.24	48.55	54.04		

The second point above arises because, when benefits increase at the force of interest, the effect of discounting is lost. The effect is much reduced if the level of relative risks is low (m = 0.25).

The major effect of the difference between male and female mortality has already been considered, in Section 2.4. Turning now to the mortality assumptions made in respect of lives with AD, I consider:

- 1.  $\mu_{x+t}^{14}$ , the mortality of lives without AD. Table 2.12 showed the effect of mortality using aggregate rates of AD incidence; I now look at individual genotypes and weighted average costs, with  $\mu_{x+t}^{14}$  given by 65% and 100% of AM80 or AF80 mortality, as appropriate.
- 2.  $\mu_{x+t}^{24}$ , the mortality of lives with AD prior to institutionalisation. In Section 1.4 this was estimated to be 33.5% of the baseline mortality. This low level could, however, be a feature of the data set used (from a brain-bank study, Jost & Grossberg (1995)) as persons from institutions might have been more likely to have been included. I look at the effects of taking  $\mu_{x+t}^{24}$  to be 65% or 100% of the baseline mortality.

The results are shown in Tables 2.19–2.22. As before, the EPVs are sensitive to the baseline mortality, increasing this to 100% of AM80 or AF80 decreases the EPVs

Table 2.20: Sensitivity of the EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) to the mortality assumptions, for Females. Level of relative risk m = 0.25.

		EPV of benefits as a % of EPV of benefits						nefits	
Baseline	State 2		before adjusting mortality, for:						
Mortality	Mortality,						$\varepsilon 2/\varepsilon 2$ &	Average	
% AF80	% Baseline	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV	
			%	%	%	%	%	%	
100%	33.5%	60	69.61	66.83	62.96	67.18	62.34	64.09	
		70	67.42	66.18	63.16	66.38	62.52	63.97	
65%	65%	60	93.31	92.73	91.88	92.80	91.74	92.13	
		70	92.28	92.03	91.39	92.06	91.24	91.55	
65%	100%	60	87.44	86.38	84.90	86.51	84.66	85.33	
		70	85.56	85.10	84.00	85.17	83.76	84.29	
100%	100%	60	59.92	56.62	52.30	57.08	51.69	53.58	
		70	56.28	54.85	51.60	55.15	50.97	52.48	

Table 2.21: Sensitivity of the EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) to the mortality assumptions, for Males. Level of relative risk, m = 1.00.

		EPV of benefits as a $\%$ of EPV of benefits							
Baseline	State 2	before adjusting mortality, for:							
Mortality	Mortality,						$\varepsilon 2/\varepsilon 2$ &	Average	
% AM80	% Baseline	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV	
			%	%	%	%	%	%	
100%	33.5%	60	73.69	64.85	57.81	70.86	55.39	59.98	
		70	68.40	63.41	58.27	70.55	55.10	59.50	
65%	65%	60	93.95	92.15	90.68	93.45	90.07	91.13	
		70	91.82	90.85	89.84	92.29	89.10	90.08	
65%	100%	60	88.50	85.34	82.81	87.52	81.85	83.59	
		70	84.69	83.03	81.33	85.44	80.17	81.73	
100%	100%	60	63.86	54.16	46.77	60.21	44.68	49.09	
		70	56.11	50.96	45.82	57.98	43.05	47.09	

Table 2.22: Sensitivity of the EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) to the mortality assumptions, for Males. Level of relative risk m = 0.25.

		EPV of benefits as a % of EPV of benefits					nefits	
Baseline	State 2	before adjusting mortality, for:						
Mortality	Mortality,						$\varepsilon 2/\varepsilon 2$ &	Average
% AM80	% Baseline	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV
			%	%	%	%	%	%
100%	33.5%	60	63.69	58.03	56.75	58.16	56.00	57.13
		70	60.87	57.93	57.21	58.10	56.38	57.36
65%	65%	60	91.88	90.70	90.44	90.72	90.26	90.52
		70	90.33	89.74	89.61	89.77	89.43	89.64
65%	100%	60	84.91	82.87	82.42	82.91	82.13	82.55
		70	82.18	81.19	80.96	81.24	80.66	81.00
100%	100%	60	53.16	47.13	45.76	47.29	45.07	46.18
		70	48.54	45.60	44.85	45.81	44.11	45.02

by about 35% - 45%. However, the use of 100% of AM80 or AF80 is mainly for illustration, as industry sources suggest that 65% of AM80 or AF80 is a reasonable assumption for LTC pricing. Of more potential concern is the assumption regarding mortality in state 2. Looking at the third scenario in each table, where mortality in state 2 is assumed to be the same as in state 1, the reductions in the EPVs are not large (about 15%) given the very large increase in  $\mu_{x+t}^{i24}$ .

The EPVs in respect of non- $\varepsilon 4$  genotypes are more sensitive to the mortality assumptions than those of the  $\varepsilon 4$  genotypes, as AD strikes later when mortality is higher. Therefore lighter baseline mortality would be expected to reduce the scope for adverse selection.

Finally, the mortality of institutionalised lives is considered. In Section 1.4,  $\mu_{x+t}^{34}$  was modelled by adding a constant K to the baseline force of mortality. The estimate of K was 0.173, but estimates from different studies ranged from 0.08 to 0.26. Again for comparison purposes I take values of K = 0.085 and 0.34, respectively about half and double the mean estimate.

Table 2.23 shows the results for females with baseline mortality of 65% of AF80; the results for males and for higher baseline mortality are very similar and I omit them. The EPVs of benefits are quite sensitive to the additional term K; doubling K reduces costs by about 35%–40%. As expected, the EPVs in respect of high-risk Table 2.23: Sensitivity of the EPVs of unit benefit increasing continuously ( $\delta_b = 0.05$ ) to the addition K to the force of mortality in the institutionalised state, for Females. Baseline mortality: 65% of AF80.

	Addition	EPV of benefits as a % of EPV of benefits						
Prop'n	to baseline				with $K$ :	$= 0.173, \pm$	for:	
of relative	mortality						$\varepsilon 2/\varepsilon 2$ &	Average
risk, $m$	in state $3$	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV
			%	%	%	%	%	%
1.00	0.085	60	153.82	148.87	140.86	149.95	139.58	144.00
		70	146.52	144.30	138.71	145.18	137.40	140.61
	0.34	60	59.31	60.91	64.03	60.58	64.72	62.83
		70	61.52	62.33	64.74	62.08	65.47	63.94
0.25	0.085	60	146.80	144.24	141.26	144.62	140.90	142.16
		70	142.01	141.04	139.06	141.30	138.71	139.61
	0.34	60	61.81	62.67	63.84	62.56	64.02	63.50
		70	63.35	63.71	64.57	63.65	64.75	64.34

genotypes are more sensitive to the assumed mortality of institutionalised lives, as these lives contract AD at younger ages, where K is relatively larger. A higher value of K, therefore, could increase the scope for adverse selection.

It is also worth noting that:

- 1. the change in the EPVs are roughly proportional to K; and
- 2. sensitivity to K is greater for younger ages at entry, again because K is relatively larger at younger ages.

# 2.6 The Impact of Adverse Selection on LTC Insurance

Adverse selection may arise in the following circumstances:

- insurers are not aware of, or are not allowed to use, any information regarding an applicant's APOE genotype;
- 2. a proportion of applicants are aware of their APOE genotype and do use this information in the process of deciding whether or not to purchase LTC insurance; and

3. applicants who know their APOE genotype have an impression of the risks of AD, whether right or wrong, that gives them an incentive to seek LTC insurance.

To model point 3 above, it is assumed that the presence of one  $\varepsilon 4$  allele is enough to lead any tested individual to regard themselves as at risk. Someone age 60, male or female, could reasonably infer this from the odds ratios in Farrer *et al.* (1997).

This model of adverse selection differs in four important respects from that used for life insurance by Macdonald (1997, 1999):

- 1. Only single premium contracts bought at a fixed age are considered, and it is assumed that any genetic testing has been carried out by then.
- 2. The sum assured applied for is fixed, and does not depend on the genetic test result. Macdonald (1997, 1999) found that large sums assured was the costliest aspect of adverse selection in life insurance, but LTC insurance is limited by the reasonable costs of care, and sums assured much in excess of needs are less likely to arise.
- 3. The impact of adverse selection in life insurance was found to be modest partly because it was measured against the background of a large, mature market. The LTC market is as yet undeveloped, so the impact of adverse selection will depend on the proportion of the population that, ultimately, is covered.
- 4. Macdonald (1997, 1999) made extreme assumptions in order to obtain an upper bound on the cost of adverse selection. This research is aiming to make realistic assumptions in order to obtain realistic estimates.

The weighted average EPVs in Table 2.13 (see equation (2.26)) give the net single premiums for a unit benefit, covering AD only, assuming that the proportion of each genotype in the insured population is the same as in the whole population. This is taken to be the premium charged to cover AD in the absence of adverse selection. The cost of adverse selection will be given as the percentage increase in these single premiums resulting from solving Thiele's equations with different proportions of high- and low-risk genotypes; in fact these will also be weighted averages of the
EPVs for each genotype in Table 2.13, with suitably chosen weights. The weights will depend on:

- 1. the prevalence of genetic testing for APOE (note that APOE is implicated in conditions other than AD, so testing could be carried out for other reasons);
- 2. the size of the LTC insurance market in the absence of adverse selection, which determines how many low-risk individuals join the risk pool; and
- 3. the probability that a person with one or two  $\varepsilon 4$  alleles will buy LTC insurance, which determines how many high-risk individuals join the risk pool.

Let  $\hat{p}_x^i$  be the frequency of genotype *i* among those buying insurance at age *x*. If  $\hat{p}_x^i \neq p_x^i$ , because of adverse selection, then the cost of adverse selection is:

$$\frac{\sum_{i} \hat{p}_{x}^{i} C_{x}^{i} - \sum_{i} p_{x}^{i} C_{x}^{i}}{\sum_{i} p_{x}^{i} C_{x}^{i}} \tag{2.29}$$

which I express as a percentage. As before, assume the gene frequencies from Farrer *et al.* (1997) apply to the whole population at age 60, and that for older ages  $p_{60+t}^i$  are found by solving the Kolmogorov forward equations (see equations (1.17) and (2.27)).

More precisely, the following steps are carried out:

- 1. Without adverse selection, the LTC market is a proportion z of the population. The same proportion z of lives of each genotype buy LTC insurance, and the proportion of purchasers at age x with genotype i is  $p_x^i$ .
- 2. With adverse selection, behaviour depends on genotype, and a proportion  $z^i$  of lives with genotype *i* purchase insurance.
- 3. Assume that lives with low-risk genotypes (i = 1 and i = 2) are just as likely as before to buy LTC insurance after a genetic test result:  $z^i = z$ .
- 4. A larger proportion of lives with high-risk genotypes (i = 3, i = 4 and i = 5)buy LTC insurance. Suppose that these lives are k times more likely to buy insurance:  $z^i = kz$  (up to a maximum of  $z^i = 1$ ).

5. The proportion of purchasers at age x with genotype i is then:

$$\hat{p}_x^i = \frac{z^i p_x^i}{\sum_{j=1}^{j=5} z^j p_x^j} \tag{2.30}$$

and the cost of adverse selection is given by equation (2.29).

At present, the LTC insurance market in the U.K. is tiny; well under 1% of the population is covered. In such a small market, there is scope for severe adverse selection, since the high-risk genotypes could be very much more likely to buy insurance (k could be very large). If the market were to attain large size, say 50% of the population (which would be large under a non-compulsory system) there could not be much adverse selection, since high-risk genotypes could not be more than twice as likely to buy insurance. The potential for adverse selection also depends on the prevalence of genetic testing for the APOE gene. It is not clear how common it might become, but note that APOE is implicated in coronary heart disease, in which context screening might be more likely.

It is also most important to realise that the costs of adverse selection calculated are in respect of AD-related claims only. Since these account for perhaps 25% - 33%of total claims (Watson, 1998) and about 40% - 50% of costs (Watson, personal communication), the costs of adverse selection, as a percentage of the total LTC insurance premiums charged in the absence of adverse selection, should be about 1/4 - 1/3 of those shown here. I will return to this point in Section 2.6.3.

#### 2.6.1 An Anomaly? Adverse Selection for Males

Before proceeding an apparent anomaly needs to be addressed. Using the fitted relative risks for men as estimated in Section 1.5, self-selection by those with one or two  $\varepsilon 4$  alleles is not generally adverse. Table 2.24 gives the costs of 'adverse' selection in percentage terms; they are negligible at age 60, and negative at higher ages. The reason lies in a feature of the odds ratios from Farrer *et al.* (1997), discussed in Setion 1.5. For men, the odds ratio for the  $\varepsilon 2/\varepsilon 4$  genotype peaks at only just over 1.0, at about age 70, then falls well below 1.0. The odds ratio for the  $\varepsilon 3/\varepsilon 4$  genotype peaks at less than 2.0 at a slightly earlier age, then also falls below 1.0. This means that:

- the ε3/ε3 genotype, which accounts for over 60% of the population at age 60, is high-risk at many ages, relative to the ε2/ε4 and ε3/ε4 genotypes;
- 2. at age 60, any increased costs caused by having a higher proportion of  $\varepsilon 4/\varepsilon 4$  genotypes in the insured population are just about balanced by the decreased costs caused by having the same higher proportion of  $\varepsilon 2/\varepsilon 4$  and  $\varepsilon 3/\varepsilon 4$  genotypes; and
- 3. at higher ages, at which the population frequency of  $\varepsilon 4/\varepsilon 4$  genotypes is reduced, a higher proportion of  $\varepsilon 2/\varepsilon 4$  and  $\varepsilon 3/\varepsilon 4$  genotypes in the insured population actually reduces costs.

The decision that must be made is whether to accept this as a genuine feature of the data, or to make adjustments so that the  $\varepsilon 4$  allele is never low-risk. The data upon which the odds ratios were based were much scarcer for males than for females, especially at older ages. Therefore, more reliance should be placed on the results for females, and the relative risks for the  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 2/\varepsilon 4$  genotypes for males adjusted to remove the apparent anomaly; this can be done by simply imposing a lower limit of 1.0. Define:

$$\bar{f}_{x+t}^i = \max(f_{x+t}^i, 1) \tag{2.31}$$

where  $f_{x+t}^i$  is defined in equation (1.14). Accordingly, new adjustment factors,  $\bar{r}_m$ , can be calculated to maintain a consistent overall incidence of AD (see Section 1.5 and Table 1.8). The adjusted values are:  $\bar{r}_1 = 1.12$  and  $\bar{r}_{0.25} = 1.02$ . These adjusted relative risk functions for males are used for the rest of this chapter.

## 2.6.2 The Cost of Adverse Selection based on Alzheimer's Disease Alone

All the costs in this section are expressed as percentages of AD care costs in the absence of adverse selection. These correspond directly to the increases in net single premiums, covering AD costs only, needed to absorb adverse selection.

Tables 2.25 to 2.28 give the percentage costs of adverse selection with high-risk lives 2, 4, 10 and 100 times more likely to buy LTC insurance; that is, for k = 2, 4, 10

Table 2.24: Costs of adverse selection, for males, as a percentage of AD-related LTC insurance costs in the absence of adverse selection, with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes k times as likely to insure as low risk genotypes, for unit benefits increasing continuously ( $\delta_b = 0.05$ ) and commencing w years before institutionalisation.

Increased						
likelihood of	Period of					
high risk	additional	Proportion	Co	st of adv	verse selec	tion
genotypes	benefits	of relative		at	age:	
insuring, $k$	w years	risk, $m$	60	65	70	75
			%	%	%	%
2	0	1.00	0.31	-0.56	-1.86	-3.48
		0.25	0.15	-0.04	-0.31	-0.63
	2	1.00	-0.29	-1.07	-2.25	-3.75
		0.25	0.03	-0.13	-0.37	-0.66
4	0	1.00	0.66	-1.20	-3.97	-7.47
		0.25	0.32	-0.08	-0.66	-1.34
	2	1.00	-0.61	-2.29	-4.82	-8.04
		0.25	0.06	-0.29	-0.80	-1.41
10	0	1.00	1.06	-1.92	-6.40	-12.08
		0.25	0.51	-0.13	-1.05	-2.15
	2	1.00	-0.99	-3.68	-7.76	-12.99
		0.25	0.10	-0.46	-1.28	-2.26
100	0	1.00	1.46	-2.66	-8.87	-16.79
		0.25	0.70	-0.18	-1.46	-2.97
	2	1.00	-1.36	-5.09	-10.75	-18.06
		0.25	0.14	-0.63	-1.76	-3.12

Table 2.25: Costs of adverse selection as a percentage of AD-related LTC insurance costs in the absence of adverse selection, with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes twice as likely to insure as low-risk genotypes (k = 2), for unit benefits increasing continuously ( $\delta_b = 0.05$ ) and commencing w years before institutionalisation.

	Period of					
	additional	Proportion	$\operatorname{Cost}$	of adv	erse sel	lection
	benefits	of relative		at	age:	
Gender	w years	risk, $m$	60	65	70	75
			%	%	%	%
$\mathbf{F}$	0	1.00	9.86	9.23	8.10	6.46
		0.25	2.92	2.70	2.32	1.80
Μ	0	1.00	4.08	3.52	2.80	2.22
		0.25	0.95	0.81	0.62	0.46
$\mathbf{F}$	1	1.00	9.28	8.70	7.64	6.10
		0.25	2.73	2.53	2.18	1.69
Μ	1	1.00	3.89	3.38	2.72	2.19
		0.25	0.90	0.77	0.59	0.45
$\mathbf{F}$	2	1.00	8.91	8.35	7.33	5.87
		0.25	2.61	2.42	2.08	1.62
Μ	2	1.00	3.78	3.29	2.66	2.17
		0.25	0.87	0.74	0.58	0.44

and 100 respectively. The market size (the proportion z of the population buying insurance) is not mentioned in the tables, because it does not affect the percentage costs for a given value of k; it only limits the value k can take. For example, the costs of adverse selection given in Table 2.25, where high-risk genotypes are twice as likely to insure as low-risk genotypes (k = 2), could represent any of the following possibilities:

- 1. a modest level of adverse selection given market penetration of 1%; or
- an upper limit for the costs of adverse selection given market penetration of 50%, and given that the prevalence of genetic testing in the population is 100%; or
- 3. an upper limit for the costs of adverse selection given market penetration of 10%, and given that the prevalence of genetic testing in the population is 20%.

From Tables 2.25 to 2.28 the greatest costs of adverse selection arise for younger ages at policy inception, under the assumption of high relative risk (m = 1) and

Table 2.26: Costs of adverse selection as a percentage of AD-related LTC insurance costs in the absence of adverse selection, with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes 4 times as likely to insure as low-risk genotypes (k = 4), for unit benefits increasing continuously ( $\delta_b = 0.05$ ) and commencing w years before institutionalisation.

	Period of					
	additional	Proportion	Cost	of adve	erse sele	$\operatorname{ction}$
	benefits	of relative		at a	age:	
Gender	w years	risk, $m$	60	65	70	75
			%	%	%	%
F	0	1.00	21.00	19.74	17.43	14.06
		0.25	6.22	5.76	4.96	3.86
Μ	0	1.00	8.68	7.50	5.98	4.77
		0.25	2.02	1.72	1.32	0.98
F	1	1.00	19.77	18.59	16.44	13.29
		0.25	5.81	5.39	4.65	3.63
Μ	1	1.00	8.30	7.21	5.81	4.69
		0.25	1.91	1.63	1.27	0.96
$\mathbf{F}$	2	1.00	18.97	17.85	15.79	12.78
		0.25	5.55	5.15	4.45	3.48
Μ	2	1.00	8.05	7.02	5.70	4.64
		0.25	1.84	1.58	1.23	0.94

Table 2.27: Costs of adverse selection as a percentage of AD-related LTC insurance costs in the absence of adverse selection, with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes 10 times as likely to insure as low-risk genotypes (k = 10), for unit benefits increasing continuously ( $\delta_b = 0.05$ ) and commencing w years before institutionalisation.

	Period of					
	additional	Proportion	Cost	of adve	erse sele	ction
	benefits	of relative		at a	age:	
Gender	w years	risk, $m$	60	65	70	75
			%	%	%	%
$\mathbf{F}$	0	1.00	33.69	31.80	28.32	23.14
		0.25	9.97	9.26	7.99	6.23
Μ	0	1.00	13.93	12.06	9.63	7.70
		0.25	3.25	2.76	2.12	1.58
$\mathbf{F}$	1	1.00	31.71	29.96	26.71	21.88
		0.25	9.32	8.66	7.49	5.87
Μ	1	1.00	13.31	11.59	9.36	7.58
		0.25	3.07	2.62	2.03	1.54
$\mathbf{F}$	2	1.00	30.43	28.75	25.65	21.04
		0.25	8.90	8.28	7.17	5.62
Μ	2	1.00	12.91	11.29	9.18	7.50
		0.25	2.96	2.53	1.98	1.51

Table 2.28: Costs of adverse selection as a percentage of AD-related LTC insurance costs in the absence of adverse selection, with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes 100 times as likely to insure as low-risk genotypes (k = 100), for unit benefits increasing continuously ( $\delta_b = 0.05$ ) and commencing w years before institutionalisation.

Period of					
additional	Proportion	Cost	of adve	erse sele	$\operatorname{ction}$
benefits	of relative		at a	age:	
w years	risk, $m$	60	65	70	75
		%	%	%	%
0	1.00	46.44	44.03	39.54	32.76
	0.25	13.75	12.77	11.05	8.65
0	1.00	19.20	16.65	13.34	10.69
	0.25	4.48	3.80	2.92	2.18
1	1.00	43.72	41.48	37.30	30.98
	0.25	12.85	11.95	10.36	8.14
1	1.00	18.34	16.01	12.96	10.52
	0.25	4.23	3.61	2.80	2.12
2	1.00	41.95	39.81	35.82	29.79
	0.25	12.27	11.42	9.92	7.81
2	1.00	17.80	15.59	12.70	10.41
	0.25	4.08	3.49	2.73	2.08
	Period of additional benefits w years 0 0 1 1 1 2 2 2	$\begin{array}{c cccc} {\rm Period \ of} & & \\ {\rm additional} & {\rm Proportion} & \\ {\rm of \ relative} & {\rm of \ relative} & \\ w \ years & {\rm risk}, \ m & \\ \end{array} \\ \hline 0 & 1.00 & \\ 0.25 & \\ 0 & 1.00 & \\ 0.25 & \\ 1 & 1.00 & \\ 0.25 & \\ 1 & 1.00 & \\ 0.25 & \\ 2 & 1.00 & \\ 0.25 & \\ 2 & 1.00 & \\ 0.25 & \\ \end{array} $	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

for benefits commencing on institutionalisation (w = 0). For k = 2, 4, 10 and 100 the greatest costs of adverse selection are 9.86%, 21.00%, 32.69% and 46.44% for females and 4.08%, 8.68%, 13.93% and 19.20% for males respectively. The costs in respect of males are less than half of those in respect of females in all cases. These drop very considerably, by about 2/3, if the population relative risks are much lower those reported in case-based studies (m = 0.25). Other features are:

- the costs of adverse selection are quite stable for ages 60–70 (decreasing slightly with age), but drop substantially at age 75 years;
- 2. the period of time spent claiming prior to institutionalisation (w) has very little effect and it is omitted from further analysis; and
- 3. as expected, the degree of adverse selection (equivalently, the size of the market) represented by k has a very large impact on the costs of adverse selection.

In Table 2.29, the costs of adverse selection are given for level benefits ( $\delta_b = 0$ ) and for benefits increasing at a reduced rate ( $\delta_b = 0.025$ ). For brevity, I only look at the

Table 2.29: Costs of adverse selection, for females, as a percentage of AD-related LTC insurance costs in the absence of adverse selection, with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes k times as likely to insure as low risk genotypes, for level unit benefits and benefits increasing continuously ( $\delta_b = 0$  and 0.025) and commencing on institutionalisation.

	Likelihood of					
	high risk	Proportion	Cost	of adve	erse sele	ction
Claims	genotypes	of relative		at a	age:	
inflation, $\delta_b$	insuring, $k$	risk, $m$	60	65	70	75
			%	%	%	%
0.025	2	1.00	12.32	11.34	9.70	7.51
		0.25	3.76	3.40	2.84	2.12
0.025	4	1.00	26.24	24.24	20.89	16.35
		0.25	8.00	7.26	6.06	4.55
0.025	10	1.00	42.10	39.05	33.93	26.91
		0.25	12.83	11.66	9.76	7.34
0.025	100	1.00	58.03	54.07	47.38	38.10
		0.25	17.69	16.09	13.50	10.20
0.000	2	1.00	14.90	13.52	11.34	8.57
		0.25	4.69	4.17	3.38	2.45
0.000	4	1.00	31.73	28.90	24.42	18.66
		0.25	9.98	8.88	7.23	5.25
0.000	10	1.00	50.90	46.56	39.67	30.71
		0.25	16.02	14.27	11.63	8.49
0.000	100	1.00	70.16	64.46	55.40	43.48
		0.25	22.08	19.69	16.09	11.78

costs for females. As expected from Section 2.5, a reduced rate of benefit escalation increases the percentage costs of adverse selection. This is because premature cases of AD cost relatively more if  $\delta_b < \delta$ . This increase in the percentage costs is roughly in proportion to the decrease in the rate of benefit escalation; at worst, it is about 5 percentage points per 1% decrease, but at the lower levels of relative risk (m = 0.25) less than 2 percentage points per 1% increase.

It should be remembered that these percentage costs are based on AD-related LTC insurance costs alone, and not on total LTC insurance costs. I defer any conclusions until total LTC costs are considered, in the next section.

## 2.6.3 The Costs of Adverse Selection Based on Total LTC Insurance Costs

The costs of adverse selection estimated so far are given as a percentage of AD costs only, so they overstate the true percentage costs as the LTC premiums actually charged will be higher to allow for other causes of claiming, such as loss of ADLs not linked to AD and other forms of cognitive impairment.

To estimate the total costs of LTC I take a simplistic approach and assume that the costs of AD make up a constant proportion of all costs. While this is unlikely to be true across ages, it does allow for estimates of the costs of adverse selection to be made in a way that at least allows for other causes of claiming.

Watson (1998) stated that 25% to 33% of all LTC insurance claims are attributable to AD. However, this refers to claim numbers and, based on LTC underwriting sources of four of the major underwriting companies at the time, AD was responsible for between 40% and 50% of all LTC claims expenses (Watson, personal correspondence). The reason AD patients have higher than average claim costs is because they have longer nursing home confinements, on average, than patients from other causes.

Using these figures total LTC insurance costs can be estimated as 2 to 2.5 times the cost of AD alone and, because of the methods used, this translates directly into the total percentage costs of adverse selection being 40% to 50% of the costs of adverse selection as a percentage of AD-related costs alone. Table 2.30 gives the range of costs of adverse selection as a percentage of total LTC costs for females, for the basic scenarios. Other costs can easily be calculated from the tables in the previous section.

From Table 2.30 I conclude the following:

- In a large LTC insurance market, the costs of adverse selection will not exceed 5%, and are probably negligible.
- 2. If the risks conferred by the  $\varepsilon 4$  allele in the whole population are much smaller than those observed to date (m = 0.25), then the same conclusion holds.
- 3. If the risks conferred by the  $\varepsilon 4$  allele in the whole population are comparable

Table 2.30: Costs of adverse selection as a percentage of total LTC insurance costs, with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes k times as likely to insure as low-risk genotypes, for benefits increasing continuously ( $\delta_b = 0.05$ ) and commencing on institutionalisation.

Likelihood of high risk Prop'n of Cost of adverse select:	ion
genotypes relative at age:	
Gender insuring, $k$ risk, $m$ 60 65 70	75
% % %	%
${ m F}$ 2 1.00 3.9 - 4.9 3.7 - 4.6 3.2 - 4.	1 2.6 - 3.2
$0.25 \qquad 1.2 - 1.5 \qquad 1.1 - 1.4 \qquad 0.9 - 1.$	2  0.7 - 0.9
M 2 $1.00  1.6 - 2.0  1.4 - 1.8  1.1 - 1.0$	.4  0.9 - 1.1
$0.25 \qquad 0.4 - 0.5 \qquad 0.3 - 0.4 \qquad 0.2 - 0.2$	$.3 \qquad 0.2 - 0.2$
$\mathbf{F}$ 4 1.00 8.4 - 10.5 7.9 - 9.9 7.0 - 8.	5.6 - 7.0
0.25 $2.5 - 3.1$ $2.3 - 2.9$ $2.0 - 2.5$	5    1.5 - 1.9
M 4 $1.00$ $3.5 - 4.3$ $3.0 - 3.8$ $2.4 - 3.3$	0 1.9 - 2.4
$0.25 \qquad 0.8-1.0 \qquad 0.7-0.9 \qquad 0.5-0.$	0.4 - 0.5
$ F \qquad 10 \qquad 1.00 \qquad 13.5 - 16.8  12.7 - 15.9  11.3 - 14 $	4.2  9.3 - 11.6
$0.25 \qquad 4.0 - 5.0 \qquad 3.7 - 4.6 \qquad 3.2 - 4.$	0 2.5 - 3.1
M 10 $1.00  5.6 - 7.0  4.8 - 6.0  3.9 - 4.8$	$.8 \qquad 3.1 - 3.9$
$0.25 \qquad 1.3 - 1.6 \qquad 1.1 - 1.4 \qquad 0.8 - 1.$	1  0.6 - 0.8
$ F \qquad 100 \qquad 1.00 \qquad 18.6 - 23.2  17.6 - 22.0  15.8 - 19 $	9.8  13.1 - 16.4
$0.25 \qquad 5.5-6.9 \qquad 5.1-6.4 \qquad 4.4-5.$	$.5 \qquad 3.5 - 4.3$
M 100 $1.00$ $7.7 - 9.6$ $6.7 - 8.3$ $5.3 - 6.7$	4.3 - 5.3
$0.25 \qquad 1.8-2.2 \qquad 1.5-1.9 \qquad 1.2-1.$	$.5 \qquad 0.9 - 1.1$

with those observed to date (m = 1) then costs of adverse selection begin to exceed 10% when high-risk genotypes are about 4 times more likely to buy insurance (k = 4). This would be feasible, though extreme, if about 25% or fewer of the population had LTC insurance; in a larger market, k would necessarily be less than 4.

- 4. With the same level of  $\varepsilon 4$  risks in a smaller LTC insurance market, say less than 10% of the population, the cost of adverse selection could reach or exceed 20%. This would probably be regarded as a significant problem by most observers. From Table 2.14, it can be seen that extra single premiums of between +20% and +40% would then be needed (for females).
- 5. These levels of adverse selection assume that the whole population is tested for the APOE gene. This is unlikely, although as previously noted the gene is also a risk factor for coronary heart disease. Lower rates of genetic testing would cut the costs of adverse selection. If a proportion y of all genotypes were tested, the effect would be as if high-risk lives were ky times more likely to buy insurance. However, it is possible that higher proportions of lives with an  $\varepsilon 4$  allele would be tested, for example the relatives of someone known to carry the allele might be more likely to be tested. This assumption is conservative.
- 6. It has been assumed that benefits increase at rate  $\delta_b = 0.05$ , which is also the force of interest  $\delta$ . This eliminates the effect of the  $\varepsilon 4$  allele in advancing the age of onset of AD; the benefit escalation cancels out the discounting. Lower rates of benefit escalation would increase the cost of adverse selection.
- No conclusion is drawn in respect of males, because of the apparently anomalous relative risks discussed in Section 2.6.1.

## 2.7 The Cost of a Combined Pension and Long-Term Care Package

LTC insurance avoids the need to use savings to pay for care, for example by selling a house. It is needed most by those whose pensions are inadequate to pay for care. Warren *et al.* (1999) compared the single premium costs of a pension and a LTC policy. Here I consider something different; under the assumption that any pension will be used to pay for care as needed, then the ideal, and cheapest, form of LTC insurance would be a top-up benefit, to pay for the excess of care costs over a pension.

In 1996–97, the average retirement pension in the U.K. was £3,254, which is certainly not sufficient to meet care costs. From discussions with LTC pricing actuaries, I took a typical sum assured under a LTC policy to be £9,600. Tables 2.31 to 2.34 show, for females and males, with high (m = 1) and low (m = 0.25) relative risks:

- 1. The cost at ages 60–75 of a pension starting at £3,254 per year, payable continuously and increasing continuously at rate  $\delta_p = 0.03$  per year. These costs are found by using the model with the pension benefit payable while in any of the three live states.
- 2. the costs at ages 60–75 of a combined retirement package, providing the pension above throughout life, increased to a care benefit payable while institutionalised with AD of £9,600 per year at outset, increasing at rate  $\delta_b = 0.05$  per year. I assume that care costs are linked to earnings inflation while pensions are linked to price inflation.
- 3. The additional cost of the LTC insurance 'top-up' as a percentage of the cost of the pension alone.

The cost of a comprehensive LTC 'top-up' would be rather higher than shown in these tables, for two reasons:

- the LTC benefits in these tables are payable only in the event of AD; as before, this might account for 40% - 50% of the total;
- 2. the use of institutionalisation as a proxy for the start of a claim further underestimates the cost; see Tables 2.15 and 2.16.

Allowing for these together, the percentage extra costs might be three to four times those shown.

Table 2.31: Comparison of the EPV of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) with the EPV of a package of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) and a care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from AD, for females. High level of relative risk m = 1.

Expected Present Value

						$\varepsilon 2/\varepsilon 2$ &	Average
Scheme	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV
	60	60,812	$63,\!087$	65,774	$62,\!956$	66,314	$65,\!178$
Pension	65	$53,\!571$	$55,\!213$	$57,\!665$	$54,\!967$	58,219	57,086
only	70	46,759	$47,\!583$	$49,\!547$	47,508	50,086	49,126
	75	40,191	$40,\!349$	$41,\!663$	$40,\!597$	42,139	$41,\!427$
Pension	60	$76,\!562$	$76,\!545$	74,088	$75,\!989$	71,901	$74,\!484$
with LTC	65	68,150	$68,\!155$	$65,\!881$	67,527	63,734	66,147
benefit	70	$59,\!680$	59,758	$57,\!651$	59,065	$55,\!518$	$57,\!855$
while inst'd	75	$51,\!197$	$51,\!478$	$49,\!614$	$50,\!647$	47,460	49,721
	60	25.90~%	21.33~%	12.64~%	20.70~%	8.43~%	14.28~%
Percentage	65	27.21~%	23.44~%	14.25~%	22.85~%	9.47~%	15.87~%
increase	70	27.63~%	25.59~%	16.36~%	24.33~%	10.85~%	17.77~%
	75	27.38~%	27.58~%	19.09~%	24.76~%	12.63~%	20.02~%

Table 2.32: Comparison of the EPV of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) with the EPV of a package of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) and a care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from AD, for females. Low level of relative risk, m = 0.25.

				r			
						$\varepsilon 2/\varepsilon 2 \&$	Average
Scheme	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV
	60	64,146	64,854	$65,\!633$	64,816	65,781	65,505
Pension	65	$56,\!305$	$56,\!813$	$57,\!523$	56,742	$57,\!675$	$57,\!351$
only	70	48,595	48,844	49,412	48,826	49,559	49,283
	75	$41,\!123$	41,168	$41,\!545$	41,244	$41,\!677$	$41,\!470$
Pension	60	$75,\!276$	$75,\!229$	74,488	$75,\!045$	74,013	74,685
with LTC	65	$66,\!970$	66,949	66,269	66,747	$65,\!806$	66,375
benefit	70	$58,\!634$	$58,\!653$	58,033	$58,\!435$	$57,\!575$	58,122
while inst'd	75	$50,\!444$	$50,\!531$	49,995	$50,\!280$	49,535	$50,\!058$
	60	17.35~%	16.00~%	13.49~%	15.78~%	12.51~%	14.01~%
Percentage	65	18.94~%	17.84~%	15.20~%	17.63~%	14.10~%	15.73~%
increase	70	20.66~%	20.08~%	17.45~%	19.68~%	16.17~%	17.94~%
	75	22.67~%	22.74~%	20.34~%	21.91~%	18.85~%	20.71~%

Table 2.33: Comparison of the EPV of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) with the EPV of package of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) and a care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from AD, for males. High level of relative risk m = 1.

Expected Present Value  $\varepsilon^{2/\varepsilon^{2}} \& A$ 

						$\varepsilon 2/\varepsilon 2 \&$	Average
Scheme	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	EPV
	60	$52,\!884$	$55,\!675$	$55,\!897$	55,738	56,361	$55,\!910$
Pension	65	$45,\!897$	47,989	48,134	47,996	48,578	48,122
only	70	$39,\!376$	$40,\!576$	$40,\!635$	40,573	41,010	$40,\!653$
	75	$33,\!138$	$33,\!622$	$33,\!625$	$33,\!624$	$33,\!861$	$33,\!650$
Pension	60	63,943	61,476	$61,\!335$	61,463	$58,\!434$	$61,\!085$
with LTC	65	$55,\!861$	$53,\!683$	$53,\!578$	$53,\!681$	$50,\!638$	$53,\!242$
benefit	70	47,841	$46,\!135$	46,082	46,136	$43,\!051$	45,705
while inst'd	75	40,118	39,048	39,045	39,045	$35,\!869$	$38,\!615$
	60	20.91~%	10.42~%	9.73~%	10.27~%	3.68~%	9.26~%
Percentage	65	21.71~%	11.87~%	11.31~%	11.85~%	4.24~%	10.64~%
increase	70	21.50~%	13.70~%	13.40~%	13.71~%	4.98~%	12.43~%
	75	21.06~%	16.14~%	16.12~%	16.12~%	5.93~%	14.75~%

Table 2.34: Comparison of the EPV of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) with the EPV of package of a pension of £3,254 p.a. ( $\delta_b = 0.03$ ) and a care benefit of £9,600 p.a. ( $\delta_b = 0.05$ ) while institutionalised from AD, for males. Low level of relative risk m = 0.25.

			Ε	xpected P	resent Val	ue	
						$\varepsilon 2/\varepsilon 2$ &	Average
Scheme	Age	$\varepsilon 4/\varepsilon 4$	$\varepsilon 3/\varepsilon 4$	$\varepsilon 3/\varepsilon 3$	$\varepsilon 2/\varepsilon 4$	$\varepsilon 2/\varepsilon 3$	$\mathrm{EPV}$
	60	55,223	$55,\!915$	55,966	$55,\!930$	56,077	56,012
Pension	65	$47,\!653$	48,167	48,201	48,169	48,309	$48,\!198$
only	70	40,389	$40,\!679$	40,693	$40,\!678$	40,787	$40,\!697$
	75	$33,\!551$	$33,\!665$	$33,\!665$	$33,\!665$	33,731	$33,\!672$
Pension	60	61,749	$61,\!077$	61,044	61,074	60,480	61,049
with LTC	65	$53,\!886$	$53,\!310$	$53,\!286$	53,310	52,717	$53,\!225$
benefit	70	46,231	45,797	45,784	45,797	45,201	45,716
while inst'd	75	39,003	38,740	38,739	38,739	38,132	$38,\!661$
	60	11.82~%	9.23~%	9.07~%	9.20~%	7.85~%	8.99~%
Percentage	65	13.08~%	10.68~%	10.55~%	10.67~%	9.13~%	10.43~%
increase	70	14.46~%	12.58~%	12.51~%	12.58~%	10.82~%	12.33~%
	75	16.25~%	15.08~%	15.07~%	15.07~%	13.05~%	14.82~%

The level of relative risks affects the cost for each genotype, but not the average cost. More important, the cost is much less dependent on the APOE genotype than LTC insurance alone; pension costs are relatively insensitive to genotype, but what effect there is offsets the genetic risk under the LTC part of the contract, and reduces the differences in cost. Allowing for non-AD LTC claims as well will reduce the dependence even further. Even with a high level of relative risks (m = 1), the difference between the cost in respect of  $\varepsilon 4/\varepsilon 4$  lives and the average cost is negligible, and so is the scope for adverse selection.

I have picked a particular combination of pension and LTC benefits, and the effect of combining them will clearly vary with the circumstances. However, it is clear that adverse selection can be much reduced if the same insurer underwrites both pension and LTC top-up.

### 2.8 Conclusions

- 1. The cost of AD-related LTC insurance depends strongly on the relative risks of the APOE genotypes. If these are consistent with observations to date, the costs for the  $\varepsilon 4/\varepsilon 4$  genotype could be about 3 times those for the  $\varepsilon 2/\varepsilon 2$  and  $\varepsilon 2/\varepsilon 3$  genotypes. These are based on studies that selected at-risk subjects; the relative risks in the population could be lower. If they were 25% of those above, the above difference in cost would fall to about 1.5 times.
- 2. At low levels of relative risk, underwriting ratings, for females, could be up to +10% (ε4/ε4 genotype) or +7% (ε3/ε4 and ε2/ε4 genotypes). At high levels of relative risk these could increase to about +40% and +25% respectively. The model suggests higher ratings for male ε4/ε4 lives, and none at all for male ε3/ε4 lives, but these may be unreliable.
- 3. The costs are very sensitive to the background level of mortality, and to mortality while institutionalised. The latter is quite uncertain, and the different studies have given very different estimates; using these could change costs by almost  $\pm 40\%$ . However, costs are not very sensitive to mortality after the onset of AD but before institutionalisation, which is also subject to great

uncertainty.

- 4. The model almost certainly underestimates AD-related costs, because institutionalisation is used as a proxy for the start of a claim. This affects the level of costs but not the cost of adverse selection. On the other hand, the use of a Gompertz formula for the rate of onset of AD at ages over 90 probably overstates costs.
- 5. The costs of adverse selection for males appear to be negative, because the  $\varepsilon 2$  allele confers such protection that the  $\varepsilon 3/\varepsilon 3$  genotype is high-risk at many ages. For the study of adverse selection only, I adjusted the relative risks so that the  $\varepsilon 4$  allele never conferred lower risk, but no reliance is placed on these results.
- 6. The cost of adverse selection is only likely to be significant if:
  - (a) the level of relative risks in the population is as high as observed to date;
  - (b) the LTC insurance market is small;
  - (c) carriers of the  $\varepsilon 4$  allele are very much more likely to buy insurance (more than 4 times more likely); and
  - (d) a high proportion of the population is tested for the APOE gene.

In the worst-case scenario, adverse selection would increase costs, and single premiums, for females by 18%–23% at age 60, and 13%–16% at age 75. This is likely to be a substantial overestimate.

7. Adverse selection might sometimes be reduced by combining pension and 'topup' LTC benefits into a comprehensive retirement package.

## Chapter 3

# Modelling Disability in Long-Term Care Insurance

### **3.1** Introduction

In Chapters 1 and 2 the main emphasis was on estimating the costs that arise under a long-term care insurance contract in respect of Alzheimer's disease. In these chapters, in order to look at the potential costs of adverse selection arising from variants of the APOE gene, the cost of other events in the ageing process (mainly disability) that trigger benefits were very simply assumed to be a multiple of those costs arising from Alzheimer's disease (see Chapter 2 for more detail).

The research in the next five chapters is motivated by wanting independently (from Alzheimer's disease) to estimate the costs of disability arising under a longterm care contract, providing a better basis for estimating the potential for adverse selection.

The aim of this chapter is to describe a continuous-time Markov model of the disability process and to describe the data I will use to estimate the model parameters, which are from the 1982, 1984, 1989 and 1994 National Long-Term Care Surveys (NLTCS) in the U.S.A.. Maximum likelihood estimates of the model parameters are discussed and obtained in Chapter 4. Then in Chapter 5, the methodology for calculating variance estimates of the transition intensities is discussed, and these variance estimates are used as weights in graduating the transition intensities. In Chapter 6, aggregate mortality in the disability model is investigated and adjustments are made to some of the disability models. Finally, in Chapter 7, the parameterized models are used to calculate the expected present value (EPV) of model long-term care benefits in respect of disability, which together with the estimated costs of Alzheimer's disease (in Chapter 2) are used to revisit the potential costs of adverse selection in the long-term care insurance market arising from people's knowledge that they are a carrier of a high risk allele of the APOE gene (and therefore at higher risk of Alzheimer's disease).

In this chapter I concentrate on the disability process of ageing only. For a detailed discussion of long-term care insurance products in general see Section 2.2. The model presented here does not aim to be product specific, rather it is based on the underlying process, so it can be modified to many long-term care product designs. In Section 3.2 disability is defined in terms of Activities of Daily Living (ADLs), and a model of disability is proposed in Section 3.3. I then provide an overview of the four National Long-Term Care Surveys (NLTCS) in Section 3.4. Then, before looking at the data in more detail, previous research, relevant to this work, that has used these data sets is discussed in Section 3.5. I deal with some anomalies found in the data in Section 3.6, before providing a detailed discussion of each pair of consecutive surveys (1982–84, 1984–89 and 1989–94), as they are used, in Sections 3.7 - 3.9.

I have presented some of the research in this chapter to the RSC2001 conference in March 2001.

## 3.2 Activities of Daily Living (ADLs)

A typical set of ADLs is that used as a benchmark by the Association of British Insurers: Washing; Dressing; Mobility; Toiletting; Feeding; and Transferring. A typical LTC policy would pay benefits upon failure of a given number of ADLs, often 3 or 4. Sometimes a reduced benefit might be paid on failure of a smaller number of ADLs.

Failure of ADLs is not irreversible; the studies by Manton (1988) and Manton,

Corder & Stallard (1993) show quite high rates of recovery. Evidently, LTC insurance shares many of the features of Permanent Health Insurance (PHI) both in the difficulty of claims underwriting, and in statistical analysis.

An insurance policy that pays claims depending on a complex series of events is conveniently represented by a multiple state model, which is the approach adopted in Section 3.3. The order in which ADLs fail is then of potential importance, leading to the unpleasant possibility that, with 6 ADLs, 6! = 720 states might be needed to represent all possibilities, even ignoring recoveries. Some studies have suggested that ADLs typically fail in a given order. Dullaway & Elliott (1998) stated, without mentioning a source, that ADLs usually fail in the order given above, and that recoveries are typically in reverse order. Dunlop, Hughes & Manheim (1997) suggest the following orders, based on a 1984–90 study with 5,151 subjects:

- 1. Mobility; Washing; Transferring; Dressing; Toiletting; Feeding; or
- 2. Mobility; Washing; Transferring; Toiletting; Dressing; Feeding.

These two lists agree on the first three ADLs to fail, and therefore about the events that would trigger a claim under a typical LTC policy. It is therefore a reasonable simplification to consider only the number of failing ADLs, and to ignore the order of failure.

## 3.3 A Model of Disability in Long-Term Care Insurance

Figure 3.19 shows a simple continuous time, discrete state model of disability, where disability is defined in terms of loss of ADLs (see Section 3.2). The reasons for grouping disability states in this way are discussed in Section 3.7. The states can easily be expanded using the same methodology if the data are available. Disability is a reversible process (see Section 4.2) and is modelled as such, though this does require the estimation of a large number of transition intensities. However, given sufficient data this is just a matter of 'number crunching'. Cognitive impairment is also covered by LTC insurance (discussed in Chapter 1), however only claims arising

from disability will be considered in this chapter. This will include a proportion of claims whose underlying cause is cognitive impairment — those lives impaired in a sufficient number of ADLs caused by cognitive impairment for them to be eligible for benefits, before being eligible from cognitive impairment itself.



Figure 3.19: A model of disability for the lifetime of an individual and long term care insurance.

A comprehensive model of LTC costs would need to include all forms of cognitive impairment as well as disability. However, this would require data at the individual level, of progression through ADLs and all forms of cognitive impairment. Until such data becomes available (from demographic studies or as the experience of insured lives increases), such detailed models cannot be utilized. A major cause of claiming in LTC insurance is onset of Alzheimer's disease — this was discussed in detail in Chapters 1 and 2.

## 3.4 Overview of the U.S. National Long-Term Care Surveys

The primary source of useful data on Activities of Daily Living (ADLs) is the series of 1982, 1984, 1989 and 1994 U.S. National Long-Term Care Surveys (NLTCS) based at Duke University. The 'raw' data sets were kindly made available to me by Professor Kenneth Manton at Duke University. The data, with separate keys, are in four main files:

- 1. an analysis file which provides summary data from all four surveys, though not in enough detail for the purposes of this analysis;
- 2. a file of raw data for the 1982 and 1984 surveys combined (the merging of these two years seems to have resulted in a loss of detailed information about the status of lives in 1984); and
- 3. two separate 'raw data' files for the 1989 and 1994 surveys.

The first five digits of each of these files is a unique identifier number that allows individual lives to be traced though all four data sets. All lives are accounted for in the analysis file in all survey years with lives not in a particular survey year classified as 'Not in Survey Year'. However, in the 'raw' data files, only those lives taking part in that particular survey are included.

These surveys are based on samples of Medicare enrollees aged over 65 years, representing more than 97% of the U.S. population (Manton, 1988). Medicare is a health insurance program for:

- 1. people aged 65 years or older;
- 2. some people with disability under age 65 years; and
- 3. people with end stage renal disease (permanent kidney failure requiring dialysis or a transplant).

Medicare has two parts: Part A: Hospital Insurance — most people pay for this through medicare taxes while they are working, though people not otherwise eligible can pay a monthly premium for cover; and Part B: Medical Insurance — for which most people pay a monthly premium. The sample for the NLTCS were taken from people covered under Part A.

In 1982, in order to identify approximately 6,000 chronically disabled persons, over 35,000 records were screened, also identifying over 1,900 institutionalised lives. Of these sampled lives, 25,541 showed no sign of disability. In 1984 and subsequent years, only a subset of 12,100 of these 25,541 lives were used. The other 13,441 lives were screened out of the surveys and there is no information about them. The cut-off date used for this screen was 1 April 1982, so that people were categorised as of this date. The 1984 NLTCS was designed to track changes in functional level and returns to the community, as were the 1989 and 1994 surveys. They are longitudinal in the sense that they follow the same cohort of lives, but they only report 'snapshot' views of the cohort at fixed times — they do not continuously track the cohort, but individual lives are traceable between each pair of surveys. Therefore, complete life histories are not available.

All four surveys had a two stage design — first a community screener questionnaire was administered to the selected lives to determine if they had any physical disability. Then, in general, if they showed any signs of disability a community detail questionnaire was undertaken at a later date to clarify the type and level of disability as well as many other personal factors. In 1982, if no disability was identified during the screener questionnaire the person was not interviewed further. Once a life has been given a detailed interview (including an institutional questionnaire), they were given detailed interviews in all later surveys until death. The specifics of each individual survey will be discussed further in Sections 3.7 - 3.9.

The 1984, 1989 and 1994 surveys incorporated aged-in samples to maintain adequate coverage of the U.S. population aged over 65 years. For example, the sample used in 1982 who were over 65 years old would be over 67 years old in 1984, so in order to maintain a sample of over-65 year old lives in 1984 a new sample, the aged-in population, of people aged 65-67 year old in 1984 was combined with the

97

original sample.

The same measures of disability were used in all years and were defined by the inability to perform one or more of 8 Independent Activities of Daily Living (IADLs light housework; laundry; meal preparation; grocery shopping; getting around outside; getting to places outside within walking distance; money management; using the telephone) or one or more of 6 ADLs (eating; getting in and out of bed; getting around inside; dressing; bathing; getting to the bathroom or using the toilet) without using personal assistance or special equipment. To be classified as chronically disabled in respect of any of the ADLs or IADLs the person would have to be unable to (or be expected to be unable to) perform the activity for more than 90 days because of disability or by reason of health. This may be taken to be a proxy for a deferred period under a LTC policy (the period during which the claimant must continuously meet the claims criteria before payments commence). In the U.K. the most common deferred period is 3 months, or approximately 90 days (Dullaway & Elliot, 1998).

Figure 3.20 is an overview of the four surveys, with the total number of people in each classification given and, where the figures are in brackets, the number of these lives from the aged-in population that are joining the survey for the first time. Transitions between all the classifications in Figure 3.20 are given in Appendix A. The surveys include a total of 35,848 lives, though not all are used in any one survey. The following classifications have been included to account for all lives in all survey years:

- 1. 'Aged-in population' these are lives that are not over age 65 years in the current survey year, but will be included in future survey years; and
- 2. 'Not in Survey Year' these are lives that have previously died, as well as other lives that have, somewhat ambiguously, been left out of the survey year. (I assume that there is no systematic reason why these lives have been left out.)

The large number of lives in the classification 'Not in Survey Year' in the 1989 (9,524 lives) and 1994 (12,730 lives) NLTCS that are not accounted for (i.e. not

classified as dead in a previous survey) raises the question: Is there any evidence in the data to suggest what may have happened to these lives? It will become apparent in Section 4.2, when looking at overall trends in the data, that the number of deaths recorded between the 1984 and 1989 NLTCS and between the 1989 and 1994 NLTCS is substantially understated when compared with the number between the 1982 and 1984 NLTCS, which suggests that a substantial proportion of the lives unaccounted for, were dead. There is further support for this in Chapter 6, when comparing overall mortality in the parameterized disability models with a benchmark force of mortality: for the models parameterized using the 1982 and 1984 NLTCS, overall model mortality is comparable to the benchmark; whereas the overall forces of mortality for the models parameterized using the 1984–89 NLTCS and the 1989–94 NLTCS are substantially below the benchmark (see Chapter 6 for more details).

The 'Non-response dead' classification has been used in Figure 3.20 in two slightly different ways. In 1982, the first survey, people who were alive as of the cut-off date (1 April 1982), but died before the screener questionnaire took place were classified as 'Community screener only — Non-response dead' (classification  $\mathbf{B}$ .). It is not known if these lives had any disability or not. Lives that were given the screener in 1982, but died before answering the detailed questionnaire are classified as 'Community detail questionnaire - Non-response dead' (classification  $\mathbf{E}$ .). It is known that these lives may have had some form of disability (as they had been chosen for the detailed interview) though they need not necessarily have been disabled in accordance with the definitions used. (Some lives that completed a detailed questionnaire were found to be non-disabled.)

In 1984, a 'deceased' questionnaire was administered to the next of kin, so lives that died between 1982 and 1984 were classified as 'dead', and there is no 'Nonresponse dead' classification. However, in 1989 and 1994, there was no such survey and lives that died between surveys were classified as 'Non-response dead' in the questionnaire in which they were found to have died. So, for example, between 1984 and 1989, 728 (529 + 131 + 68) lives died, of which 49 (38 + 11 + 0) lives were from the aged-in population.

#### 1982 NLTCS

#### 1984 NLTCS

#### 1989 NLTCS

#### 1994 NLTCS

Community Screener Only	Community Screener Only	Community Screener Only	Community Screener Only
A. Questionnaire complete: 11,570	A.Questionnaire complete: 13,786(4,263)	<b>A.</b> Questionnaire complete: $10,330(4,193)$	A.Questionnaire complete: 10,474(3,750)
<b>B.</b> Non-response dead: 215		<b>B.</b> Non-response dead: $529(38)$	B.Non-response dead: 726 (172)
C. Non-response other: 315	Non-response other: see $\mathbf{C}$ .	<b>C.</b> Non-response other: $390(101)$	<b>C.</b> Non-response other: $851$ (272)
Community detail questionnaire	Community detail questionnaire	Community detail questionnaire	Community detail questionnaire
<b>D.</b> Questionnaire complete: 6,088	<b>B.</b> Questionnaire complete: 5,934 (440)	<b>D.</b> Questionnaire complete: $4,463$ (467)	<b>D.</b> Questionnaire complete: 5,089 (999)
E. Non-response dead: 67	C. Non-response other: 689 (118)	<b>E.</b> Non-response dead: 131 (11)	<b>E.</b> Non-response dead: 81 (20)
<b>F.</b> Non-response other: 238	(Screener - 359, Detail - 330)	<b>F.</b> Non-response other: 283 (40)	<b>F.</b> Non-response other: $573$ (128)
Institutionalized	Institutionalized	Institutionalized	Institutionalized
<b>G.</b> Inst'd before 1/4/1982: 1.708	<b>D.</b> Survey complete: 1,690 (39)	<b>G.</b> Survey complete: 1,354 (55)	<b>G.</b> Survey complete: 1,330 (190)
		<b>H.</b> Non-response dead: $68(0)$	H. Non-response dead: 32 (6)
<b>H.</b> Inst'd after 1/4/1982: 284	<b>E.</b> Non-response other: $83(0)$	I. Non-response other: 17 (2)	I. Non-response other: 15 (3)
1984 Aged-in population	Dead	1994 Aged in population	Not in Survey Year
<b>I.</b> 4,916	TE 2.210	<b>J.</b> 65–74 yrs: 5,000	<b>J.</b> 16,677
	<b>F</b> . 3,219	<b>K.</b> 95+ yrs: 540	(previously died $-3,947$ , other $-12,730$ )
1989 Aged-in population	1989 Aged-in population	Not in Survey Year	
		<b>L.</b> 12,743	
<b>J.</b> 4,907	<b>G.</b> 4,907	(previously died $-3,219$ , other $-9,524$ )	
1994 Aged-in population	1994 Aged-in population		
<b>K.</b> 65–74 yrs: 5,000	<b>H.</b> 65–74 yrs: 5,000		
<b>L.</b> 95+ yrs: 540	<b>I.</b> 95+ yrs: 540		

Figure 3.20: An overview of the 1982, 1984, 1989 and 1994 U.S. National Long-Term Care Surveys.

The '1984 aged-in' group (4,916 lives) are lives that were aged 63–65 years in 1982, and were added to the survey in 1984, when they were aged 65–67 years to maintain coverage of the over 65 years old population. Similarly, the '1989 aged-in' group (4,907 lives) were added to the survey in 1989 when the lives were aged 65-70 years old. In 1994, two groups of lives were added to the survey — the usual aged in sample of lives aged 65–70 years in 1994 (5,000 lives), as well as a group of lives then aged over 95 years (540 lives) to supplement the sample of old age lives, which diminished over time because of mortality. (Strictly speaking these are not aged-in lives, but a supplementary sample to ensure adequate coverage of the entire over 65 years old population.) However, neither of these groups are used in this work as either their status is unknown in 1989 (for the 95+ years old group) or they are too young to be included in the 1989 survey (the group aged 65-74 year olds in 1994). I include them here for completeness only.

In 1982, lives that were found to be in institutions were split into those that had been institutionalised before or after the cut-off date of 1 April 1982. This is because these two groups represent different types of institutionalised lives. Those institutionalised after 1 April 1982 include lives that had been admitted for acute medical reasons who may quickly recover (for example, lives admitted from physical injury) or die relatively soon (for example, lives admitted towards the end of a fatal disease). Those admitted before 1 April 1982, represent the 'stayers', the group left after lives with acute medical conditions have either stabilised in their condition or left the institutions. This is discussed in more detail in Section 3.7.

These data have been used before by other researchers to investigate trends of disability. Previous research, relevant to this work, is summarized in the next section and the differences between their work and that presented here are highlighted before the data are discussed in more detail in Section 3.6.

## 3.5 Previous Research Using the NLTCS

These data have already been used for research into trends of disability. Papers by Manton (1988) and Manton, Corder & Stallard (1993) are of particular interest as they look at transition probabilities between disability levels. In both of these studies disability was grouped into 6 categories: Healthy, loss of IADLs only, loss of 1–2 ADLs, loss of 3–4 ADLs, loss of 5–6 ADLs and Institutionalised (Inst'd).

Manton (1988) studied changes in functional level between 1982 and 1984 producing 2-year adjusted (for mortality and institutionalisation) and unadjusted transition probabilities between disability levels, stratified by gender and age (grouped into: 65–74 years, 75–84 years and 85+ years as of 1 April 1982). His aim was to look at trends of disability in the general U.S. population using longitudinal data.

Manton, Corder & Stallard (1993) studied the secular trends in changes in functional level between 1982, 1984 and 1989, providing 2-year and 5-year transition probabilities for changes in functional level, stratified by age (as above), but not by gender. This paper looked at changes in trends of disability over time by comparing transition probabilities between 1982–1984 and 1984–1989. I look at it in more detail as it raises some interesting methodological questions relevant to this research.

After adjusting for censored data, Manton (1988) and Manton, Corder & Stallard (1993) calculated maximum likelihood estimates of the *t*-year probabilities of a person aged *x* moving from state *i* to state j ( $P_{x\ x+t}^{i\ j}$ ) using equation (3.32)

$$P_{x\ x+t}^{i\ j} = \frac{n_{x\ x+t}^{i\ j}}{\sum_{j} n_{x\ x+t}^{i\ j}} \tag{3.32}$$

where, for example in the case of the 1982 and 1984 NLTCS,  $n_x^{ij}_{x+t}$  are the number of people in state *i*, aged *x* in 1982 and in state *j*, aged x+2 in 1984. More specifically, lives are grouped by their age in 1982 and this cohort of lives is then followed to 1984, when they are 2 years older.

To compare transition probabilities between the two time periods 1982–1984 and 1984–1989, it is necessary to transform the 5-year transition probabilities (from 1984–1989) into 2-year transition probabilities, or vice versa. Assuming constant transition intensities, 2-year transition probabilities can be calculated by raising the matrix of 5-year transition probabilities to the power 2/5 (or 5-year transition probabilities, by raising the matrix of 2-year transition probabilities to the power 5/2). However, some of the 2-year transition probabilities calculated in this way are

negative. A simple solution to this is to set any negative probabilities to zero, as in Manton, Corder & Stallard (1993). However, this presents more difficulty if it is the transition intensities that are of interest because:

- in a multiple state model in which all non-dead states commute, if one transition probability between two non-dead states is zero, then all transition probabilities are necessarily zero; and
- 2. it implies that at least some of the transition intensities calculated from these data will be negative or complex, as a set of positive transition intensities cannot produce negative probabilities.

This problem arises because not all discrete-time homogeneous Markov chains have transition probabilities consistent with continuous-time Markov processes. In fact, it would be slightly surprising if real data led to a set of transition probabilities that were consistent in this way.

The difference between this work and the research described above is that I aim to:

- model the underlying disability process by estimating the transition intensities of disability, which are needed for actuarial applications (discussed in more detail in Chapter 1); and
- 2. set out a consistent methodology for dealing with such data, that would be of use for modelling purposes and for the calculation of transition probabilities.

Before looking at the data in more detail, some anomalies in the data are highlighted and discussed in the next section.

### **3.6** Anomalies in the Data

As is clear from the tables of transitions between the classifications of Figure 3.20 (given in Appendix A), some of the data are misclassified in some years. Not all of the lives that are classified as 'Community screener only — Non-response dead' in 1982 and 1989 are classified as dead in the 1984 and 1994 surveys, respectively. I look at these two years next.

#### 3.6.1 Anomalies in the 1982 NLTCS

There are 12 lives classified as 'Community screener only — non-response dead' in 1982 that are not classified as 'Dead' in 1984: 6 lives as 'Community screener only — complete'; 4 lives as 'Community detail questionnaire — non-response other'; and 2 lives as 'Institutionalised — questionnaire complete'.

To aid in reclassifying these lives I assume that if a given life has completed a questionnaire (or at least, that there are entries in the data file for the individual in question) then the person is alive in that survey year. So, of the 8 lives that have been classified as having completed a questionnaire in 1984 all 8 have data entries in the detailed surveys in 1984 (6 have answers to the community detail questionnaire and 2 have answers to the institutionalised questionnaire). As there is no information about these lives in 1982 (as they did not even answer the screener interview) I reclassify these 8 lives as 'Community screener only — non-response other' in 1982. This adjustment only affects the 1982 survey.

The four lives classified as 'Community detail questionnaire — non-response other' in 1984 are more difficult to reclassify as there is no information about them in 1984. Futhermore, they are classified as 'Not in survey year' in 1989 and 1994, providing no more information about them. As it is not clear whether these lives were misclassified in 1982 or 1984 and as they are effectively censored observations in all four survey years, providing little extra information, it seems reasonable to exclude them from all surveys.

This adjustment affects all survey years in the following way:

- 1. in 1982, 4 lives are removed from the 'Community screener only non-response dead' classification;
- in 1984, 4 lives are removed from the 'Community detail non-response other' classification;
- 3. in 1989 and 1994, 4 lives are removed from the 'Not in survey year' classification; and
- 4. the total number of lives in each survey year is reduced to 35,844.

I now look at the anomalies in the 1989 NLTCS, which are dealt with in a similar fashion.

#### 3.6.2 Anomalies in the 1989 NLTCS

In the 1989 survey there are 44 lives classified as 'Community screener only — nonresponse dead' that are not classified as dead (i.e. in the classification 'Non-response dead' in the community screener, the community detail or the institutionalised questionnaire) or as 'Not in survey year' in 1994.

Of these 44 lives, 25 answered questionnaires (19 lives have data entries for the screener interview and 6 have data entries for the institutionalised questionnaire). Using the same reasoning as in Section 3.6.1, I reclassify these lives as 'Community screener only — non-response other' in 1989. Seven of these lives are part of the '1989 aged-in population', which has to be allowed for when looking at the surveys in more detail, as in Section 3.8. This adjustment only affects the 1989 survey.

There are 16 non-responders in 1994 (15 from the screener only group and 1 from the community detail group) from the 19 lives not yet accounted for. The other 3 lives, although classified as having completed the screener questionnaire, have no such data entries and so I treat them as if they were non-responders. It seems reasonable to assume (or without loss of information) that these 19 lives did actually die prior to the 1989 survey, are correctly classified in 1989 and are, accordingly, taken out of the 1994 survey (reclassified as 'Not in survey year' in 1994). This adjustment only affects the 1994 survey.

These adjustments were made before looking in more detail at the individual surveys as some of these adjustments affect several survey years and once the adjustments are made to the raw data, they can easily be incorporated in any further analysis.

## 3.7 Details of the 1982 and 1984 NLTCS

Table 3.35 shows the 1982 and 1984 NLTCS in more detail, after all the adjustments in Section 3.6 have been carried out and some the classifications merged. I first Table 3.35: Transitions between disability states in the 1982 and 1984 National Long-Term Care Surveys, unadjusted for censored data.

		$1984  { m Status}$						
1982 Status		IADL	1 - 2	3 - 4	5 - 6			
	Healthy	only	ADLs	ADLs	ADLs	$Inst'd^{(1)}$	Dead	$N-R^{(2)}$
Healthy	9866	470	391	116	111	166	773	213
IADL only	274	541	335	80	80	100	261	58
1-2 ADLs	115	220	660	232	135	151	389	76
3–4 ADLs	16	32	145	194	161	81	197	25
5–6 ADLs	16	35	65	89	298	96	363	32
Inst'd								
before $1/4/82$	12	10	9	15	14	953	688	7
1982 detail								
non-responders	20	12	23	16	21	44	123	46
Inst'd								
after $1/4/82$	8	9	10	6	7	118	122	4
$\mathrm{N-R}^{(3)}$	121	9	7	4	7	25	247	106
Total	10448	1338	1645	752	834	1734	3163	567

(1) Including institutionalized non-responders.

(2) These are non-responders of the screener and the detail questionnaire.

(3) These are non-responders of the screener only, including lives that died after 1 April 1982 but before the screener took place.

clarify what the states/classifications (I refer to lives as being in states from now on as the level of disability of lives is introduced) used in Table 3.35 are in more detail, then discuss how the censored data are dealt with, and finally provide a table of data after adjusting for censored lives.

All lives that completed a screener only are healthy lives (as they were not chosen for a detailed questionnaire), lives that completed a detailed questionnaire are classified in the analysis data file as: Healthy (not disabled), IADL only, 1–2 ADLs, 3–4 ADLs or 5–6 ADLs. So the 'healthy' state consists of some lives that completed the detail questionnaire and all of those that completed the screener only. I grouped disability into the same 6 categories as did Manton (1998) and Manton, Corder & Stallard (1993) (see Section 3.5). I chose the same set of groupings because:

1. grouping disability in this way had desirable features — this grouping was predictive of mortality and disability according to Manton (1988);

- 2. the groups were not so small as to reduce their credibility, and gave a reasonable number of groups to model; and
- 3. the raw data were already classified into these groupings, making data extraction easier.

There are many other ways and instruments to classify disability; however the use of ADLs is consistent with methods used by insurance companies (see Section 3.2). For more information on the classification of disability see Manton (1988).

For more specific modelling it may be necessary to split some of these states further, providing the data are available. For example, to model a product that pays out half the sum assured on the loss of 2 ADLs and the full sum assured after the loss of three or more ADLs, the 1–2 ADL group would need to be split into two separate groups. However, this grouping is sufficient for this research — to model contracts that pay out the full sum assured on the loss of 3 or more ADLs.

Looking at the 1982 survey first, for the same reasons as in Manton (1988), two groups of lives were classified separately, as their experiences were unlike those of any other groups:

- non-responders in 1982 had a poor disability experience and higher mortality which can be explained by disability or poor health being a reason for nonresponse; and
- 2. people institutionalised after 1 April 1982, but before the actual survey took place, had very high mortality rates and high recovery rates, which may be expected from admissions to institutions for acute medical reasons (as discussed in Section 3.4).

For these reasons, these two groups of lives are left out of any further analysis in the 1982 and 1984 surveys (though they are included in the later surveys). Lives in the 1984 aged-in group are also excluded here as they are too young (<65 years old) and their status is unknown in 1982.

In 1982, the categories 'Screener only — non-response other' and 'Screener only — non-response dead' have been merged, as all lives in this second classification died after 1 April 1982 but before the survey took place, so to leave them out would be to understate mortality. As these lives did not answer a questionnaire it is not known whether they were disabled or not, thus it seems reasonable to distribute them *prorata* to states covered by the questionnaires (excluding the lives institutionalised after 1 April 1982), using as weights both lives that were only given a screener and those selected for a detail questionnaire (as they could have been either).

In 1984 it is not possible to distinguish between non-responders to the screener only and non-responders to the detail questionnaire (as the data are not available). So these two groups of censored lives are dealt with together from necessity. They are distributed to states in 1984 in the same way as non-responders to the screener only were dealt with in 1982. This is because this joint group contains lives that did not answer the screener and thus could be in either group (screener only or community detail). A better way to distribute lives that were censored on the detail questionnaire (and thus had completed the screener questionnaire), were it possible to distinguish between these two groups, would be to distribute them prorata over disability states using as weights only those lives that answered the detail questionnaire (i.e. excluding the non-disabled lives that completed the screener only). Then censored lives that only completed the screener could be distributed as the 1982 censored lives were.

In redistributing the non-responders in 1982 and 1984 all the data were used (i.e. over both genders and all ages) to calculate the proportions used to allocate the lives. This assumes that non-responders do not differ in their level of disability with respect to age or gender. An alternative method would be only to use lives within the age group of the censored lives in the re-allocation. However, as it is not clear whether non-responders would differ in their level of disability by age or gender, using all the data provides a more robust method as more lives are used.

The data after the above adjustments for censored lives are summarized, for all age groups and both genders, in Table 3.36. The same data stratified by gender and age group (see below) are given in Appendix B. The age groupings I chose to look at are:

1. 10-year age bands (65-74, 75-84 and 85+ years) — to allow comparison with

Table 3.36: Transitions between disability states in the 1982 and 1984 National Long-Term Care Surveys, adjusted for censored data.

	1984 Status								
1982 Status		IADL	1 - 2	3 - 4	5 - 6				
	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead		
Healthy	10129.12	499.07	423.91	131.46	129.78	210.18	941.67		
IADL only	326.96	547.18	342.23	83.34	83.90	108.55	282.39		
1-2 ADLs	180.54	227.73	669.11	236.20	139.86	161.48	411.95		
3–4 ADLs	39.07	34.68	148.14	195.45	162.70	84.73	206.54		
5-6 ADLs	43.16	38.21	68.80	90.75	300.02	100.33	372.09		
$Inst'd^{(1)}$	28.12	11.65	10.78	15.85	15.10	955.85	703.37		
Total	10746.97	1358.52	1662.97	753.05	831.36	1621.12	2918.00		

(1) These are lives that were institutionalized before 1 April 1982.

the work done by the previous researchers; and

2. 5-year age bands (65–69, 70–74, 75–79, 80–84 and 85+ years) — to get more point estimates, which may help highlight any patterns in the data with age.

### 3.8 Details of the 1984 and 1989 NLTCS

Table 3.37 shows the 1984 and 1989 NLTCS in more detail, after the adjustments for anomolous data have been made and some of the classifications merged/renamed. This pair of surveys includes all the lives in the 1982 survey that survived to 1984, including those lives that were excluded from the 1982 and 1984 analysis (the 1982 detail non-responders and those persons institutionalised after 1 April 1982), as well as the 1984 aged-in population (4,916 lives) that survived to 1984 (4,860 lives).

In both survey years the institutionalised state includes lives that did not respond to the institutional questionnaire, as they were identified as living in an institution. There is only one institutionalised state in 1984 as the survey does not provide enough information to classify them as institutionalised before or after the cut-off date of 1 April 1984. It may then be that lives institutionalised in 1984 have higher mortality and recovery rates than those in 1982, as these were exactly the features of the lives left out of the 1982 institutionalised population — lives institutionalised after the cut-off date, but before the survey took place (see Section 3.7 for more detail). Table 3.37: Transitions between disability states in the 1984 and 1989 National Long-Term Care Surveys, unadjusted for censored data.

		1989 Status							
1984 Status		IADL	1 - 2	3 - 4	5-6			N-I	$R^{(2)}$
	Healthy	only	ADLs	ADLs	ADLs	$Inst'd^{(1)}$	Dead	А	В
Healthy	6581	416	465	194	159	336	271	171	109
IADL only	157	219	228	91	78	120	79	21	37
1-2 ADLs	85	91	350	182	102	175	78	28	40
3–4 ADLs	10	17	77	122	70	86	54	17	13
5–6 ADLs	10	12	43	46	117	73	38	6	4
$Inst'd^{(1)}$	4	4	7	3	4	492	112	10	6
N-R other <sup><math>(3)</math></sup>	106	18	24	26	15	32	29	54	34
Total	6953	777	1194	664	545	1314	661	307	243

(1) Including institutionalized non-responders.

(2) A: Non-response other to the screener questionnaire, B: Non-response other to the detail questionnaire.

(3) These are non-responders of the screener and the detail questionnaire.

The dead state in 1989 includes the total number of lives classified as non-response dead in any questionnaire in 1989. I aggregated these lives as the survey in which a life is found to be deceased does not provide any further useful information.

In the 1982 survey, lives that did not respond to the detail questionnaire were left in a separate classification and excluded from the analysis due to their very high mortality (see Section 3.7 for more detail). In 1984, it is not possible to distinguish (the data are not available) between non-responders of the screener only and those of the detail questionnaire. So it is not possible to see if this is also the case with the 1984 detail non-responders — both of these groups of lives have to be dealt with together. However, in 1989 it is possible to assign these non-responders to either the screener only or the detail questionnaire, and thus when redistributing them, to use slightly different weights to allow for the survey to which they did not respond. Even so, it is not possible to look at the mortality experience of the 1989 detail non-responders as they are all left out of the 1994 survey (classified as 'Not in survey year'). They may have been left out of the 1994 survey exactly because they were detail non-responders in 1989, although there is no further evidence to support this.

Non-responders to the screener and detail questionnaire in 1984 are dealt with

Table 3.38: Transitions between disability states in the 1984 and 1989 National Long-Term Care Surveys, adjusted for censored data.

	1989 Status								
1984 Status		IADL	1 - 2	3 - 4	5-6				
	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead		
Healthy	6805.15	461.29	532.09	239.96	192.39	413.65	292.06		
IADL only	187.48	228.42	242.22	99.78	84.77	136.04	81.31		
1-2 ADLs	121.44	101.6	366.01	191.86	109.62	193.05	80.54		
3-4 ADLs	27.49	21.16	83.28	125.87	72.99	93.08	54.99		
5-6 ADLs	18.31	13.86	45.77	47.86	118.37	76.19	38.78		
Inst'd	17.91	7.04	11.52	6.05	6.23	497.21	113.33		
Total	7177.78	833.37	1280.89	711.38	584.36	1409.21	661.00		

in the same way as those in 1982 — they are allocated pro-rata to the other states in 1984 using as weights all the other lives (i.e. including non-disabled lives that only took the screener). This may understate levels of disability among these lives, as some of them were scheduled for the detailed questionnaire, an indication that they may have been disabled, but they are allocated using all lives — including the non-disabled lives that only took the screener.

In 1989, the censored lives are classified by the survey to which they did not respond. The non-responders to the screener only were distributed in the same way as the 1984 non-responders, as they may or may not have been asked to take a detail questionnaire. For the detail non-responders, it is known that they were asked to take a detailed questionnaire, so when allocating these lives pro-rata to states in 1989, the non-disabled screener only group is excluded before allocating these lives.

For the same reasons as in Section 3.7, when calculating the proportions for allocating non-responders to disability states all the data were used (over all ages and both genders).

The data after the above adjustments for censored lives are summarized, for all age groups and both genders, in Table 3.38. The same data stratified by gender and age group (10-year and 5-year age bands, with an upper age group of lives aged over 85 years) are given in Appendix C.
Table 3.39: Transitions between disability states in the 1989 and 1994 National Long-Term Care Surveys, unadjusted for censored data.

				$1994 \mathrm{St}$	tatus				
1989 Status		IADL	1 - 2	3 - 4	5-6			N-I	$A^{(2)}$
	Healthy	only	ADLs	ADLs	ADLs	$Inst'd^{(1)}$	Dead	А	В
Healthy	5189	349	462	180	163	370	272	246	236
IADL only	96	110	113	61	41	64	46	38	24
1-2 ADLs	66	59	181	138	78	136	69	42	31
3–4 ADLs	14	13	41	70	58	88	48	22	19
5–6 ADLs	4	3	11	19	70	47	31	15	8
$Inst'd^{(1)}$	1	2	5	3	2	291	55	21	15
N-R other <sup><math>(3)</math></sup>	109	11	9	12	13	32	11	46	6
Total	5479	547	822	483	425	1028	532	430	339

(1) Including institutionalized non-responders.

(2) A: Non-response other to the screener questionnaire, B: Non-response other to the detail questionnaire.

(3) These are non-responders of the screener only.

#### 3.9 Details of the 1989 and 1994 NLTCS

Table 3.39 shows the 1989 and 1994 NLTCS in more detail after the adjustments for anomolous data have been made and some of the classifications merged/renamed. This pair of surveys includes all lives that are in the 1984 survey, that survived to 1989 and are not classified as 'Not in survey year' in either 1989 or 1994. Also included in this pair of surveys is the 1989 aged-in population (4,907 lives) that survived to 1989 (4,858 lives). The details of the institutionalised state (in 1989 and 1994) and the dead state (in 1994) are the same as in Section 3.8.

An additional feature of this pair of surveys is that a supplementary sample of 922 lives that would otherwise have been scheduled for the screener interview only in 1994 were also given a detailed questionnaire to improve the description of nondisabled persons (because of other factors of interest in the survey, such as use of Medicare-funded health services). For the purpose of redistributing censored cases these 922 lives are classified as screener only, as they would not have been asked to take a detailed questionnaire as part of the normal screening procedure.

As already observed in Section 3.8 all non-responders of the detailed questionnaire are excluded from the 1994 survey (classified as 'not in survey year') and so are left out of this analysis. The reason why they have been excluded from the 1994 survey Table 3.40: Transitions between disability states in the 1989 and 1994 National Long-Term Care Surveys, adjusted for censored data.

			1	994 Statı	ıs		
1989 Status		IADL	1 - 2	3 - 4	5-6		
	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	5491.93	407.93	545.39	233.61	211.85	488.53	280.59
IADL only	131.82	116.54	122.49	66.88	46.29	76.81	46.62
1-2 ADLs	108.36	67.07	192.65	145.27	84.55	151.87	69.84
3–4 ADLs	35.68	17.49	47.56	74.03	61.60	96.73	48.36
5-6 ADLs	17.27	5.31	14.37	21.07	71.86	51.50	31.20
Inst'd	21.69	5.92	10.67	6.53	5.17	298.69	55.39
Total	5806.75	620.26	933.13	547.39	481.32	1164.13	532.00

is not clear, whether by sampling or selection. This may have been a fairly disabled group of lives, as disability/illness is often a reason for non-response (see Section 3.7) and it is unfortunate that their experience is lost.

This leaves only the non-responders of the screen in 1989, which I redistribute in the same way as the non-responders in 1984 — they are allocated pro-rata over all disability states using all lives as weights. Including in the weights lives that only answered the screener is justified here as these lives may have been either screener only or have been disabled enough to be asked to complete a detailed questionnaire.

In 1994, the non-responders are classified by the survey to which they did not respond — exactly the same as the 1989 non-responders in the 1984–89 analysis. So, these are dealt with in exactly the same way (see Section 3.8 for details).

Again, when calculating the proportions used to allocate the non-responders to disability states all the data were used (all ages and both genders) for the same reasons as in Section 3.7.

The data after the above adjustments for censored lives are summarized, for all age groups and both genders, in Table 3.40. The same data stratified by gender and age group (10-year and 5-year age bands, with an upper age group of lives aged over 85 years) are given in Appendix D.

With the data classified in to disability states and censored cases dealt with, in the next chapter they are used to estimate the transition intensities defining the disability model (illustrated in Figure 3.19).

#### Chapter 4

# Estimating the Transition Intensities in the Disability Model

#### 4.1 Introduction

The aim of this chapter is to estimate the transition intensities in the disability model using the data described in the previous chapter. However, the data do not provide sufficient information to estimate the transition intensities directly. This is because the usual maximum likelihood estimators (occurrence–exposure rates) require information at the individual level of the numbers and times of transitions between states (Waters 1984). The NLTCS do not provide this much detail (see Section 3.4), they only provide information at discrete points in time and no information about what happened in between.

As in Section 3.5, I calculate transition probabilities between disability states (in Section 4.2) and show in Section 4.3, that these are not consistent with a set of positive transition intensities (as some of the implied intensities are negative). To avoid negative transition intensities, the maximum likelihood estimates have to be constrained to lie in the feasible region of the parameter space, but the unconstrained estimates are still a useful starting point for the fitting process. In Section 4.5 I describe a method for estimating constrained (non-negative) maximum likelihood estimate them. In Chapter 5, I find approximate confidence intervals and graduate the estimated

intensities.

In the previous section, the data was split into 10-year age bands (to allow comparison with previous research) and 5-year age bands (to give more data points). Looking forward to section 5.5 (graduation of the transition intensities) it becomes clear that having more data points is very beneficial (it allows a more robust graduation procedure to be used) and so I concentrate the results on the data grouped into 5-year age bands. I initially used both sets of age groupings for all of the following work, but I report only the main results for the data grouped into 10-year age bands (final estimates of the transition intensities). Also, for brevity, in this chapter, I only use data from the 1982–84 NLTCS to illustrate the methods. The corresponding results for the 1984–89, 1989–94 surveys are given in the specified Appendices. Comments on these results are given in the text, where they differ from those illustrated.

I now introduce some notation. The theoretical (or population) transition intensity between states *i* and *j* (the states are numbered as in Figure 3.19) for a person aged x+t I denote as  $\mu_{x+t}^{ij}$ , the maximum likelihood estimate as  $\hat{\mu}_{x+t}^{ij}$ , the constrained maximum likelihood estimate as  $\overline{\mu}_{x+t}^{ij}$  and the graduated transition intensity as  $\hat{\mu}_{x+t}^{ij}$ .

#### 4.2 Calculation of Transition Probabilities

The MLEs of the transition probabilities for men, women and in aggregate in 5-year age bands, for the 1982–84 NLTCS, calculated using equation (3.32), are given in Tables 4.41, 4.42 and 4.43 respectively. Corresponding transition probabilities for the 1984–89 and 1989–94 NLTCS are given in Appendix E. It is noticeable that the probability of recovery from disability can be fairly high, especially for younger age groups, justifying the use of a disability-recovery model.

It is worth noting a few features of mortality, from Tables 4.41, 4.42 and 4.43, which are also generally true for the 1984–89 and 1989–94 NLTCS in Appendix E (however I focus on mortality in the 1982–84 NLTCS, since it becomes clear in Sections 6.4 and 6.5 that mortality in the 1984–89 NLTCS and the 1989–94 NLTCS is anomalous, see below):

Table 4.41: The 2-year transition probabilities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys as a percentage, for males using 5 year age groupings.

				198	34 Status	3		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	89.04	2.17	1.47	0.65	0.61	0.63	5.43
	70 - 74	81.22	2.69	2.27	0.78	0.94	0.98	11.11
	75 - 79	75.99	4.37	3.17	0.85	1.34	1.47	12.81
	80 - 84	66.02	6.54	5.98	1.04	1.74	2.94	15.73
	85 +	46.16	9.48	10.46	3.42	1.84	4.39	24.25
IADL only	65 - 69	30.39	30.89	13.35	1.39	6.70	1.62	15.68
	70 - 74	30.11	31.85	11.30	1.50	2.10	3.02	20.11
	75 - 79	16.46	30.67	11.74	4.96	8.60	7.03	20.55
	80 - 84	9.94	28.30	17.24	6.17	9.84	7.57	20.95
	85 +	2.15	33.64	13.35	7.34	4.42	6.18	32.93
12  ADLs	65 - 69	14.52	8.63	31.01	11.69	10.17	3.44	20.54
	70 - 74	9.13	10.73	29.86	16.20	5.09	6.58	22.41
	75 - 79	7.54	13.50	26.86	7.15	10.37	0.28	34.30
	80 - 84	4.33	5.65	30.79	9.99	5.50	5.68	38.07
	85 +	5.22	4.06	18.34	9.98	13.52	15.76	33.13
3-4 ADLs	65 - 69	10.74	1.96	20.52	15.63	15.66	5.22	30.27
	70 - 74	11.71	1.90	16.23	27.34	15.94	0.68	26.21
	75 - 79	4.89	2.05	7.30	20.89	14.07	10.81	39.98
	80 - 84	0.65	8.63	11.36	14.17	19.84	8.64	36.72
	85 +	0.39	3.81	3.89	11.32	20.73	13.37	46.49
5-6 ADLs	65 - 69	6.13	5.21	7.72	10.90	27.39	6.57	36.08
	70 - 74	7.84	4.13	11.08	9.03	29.75	5.22	32.96
	75 - 79	3.60	4.09	2.86	6.56	31.03	9.32	42.53
	80 - 84	2.85	5.77	5.80	9.61	23.10	7.18	45.68
	85 +	0.41	0.04	4.20	6.14	24.48	6.31	58.41
Inst'd	65 - 69	6.94	1.79	0.37	1.79	0.23	68.12	20.74
	70 - 74	4.02	0.15	4.55	0.18	3.00	41.10	47.00
	75 - 79	2.65	2.21	1.20	1.11	0.14	49.12	43.56
	80 - 84	0.43	0.09	0.02	1.18	1.19	44.74	52.35
	85 +	0.25	0.03	0.08	0.01	0.02	37.96	61.66

Table 4.42: The 2-year transition probabilities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys as a percentage, for females using 5 year age groupings.

				198	34 Status	3		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	90.78	2.59	2.27	0.54	0.52	0.35	2.94
	70 - 74	85.58	4.27	2.51	1.08	0.93	1.20	4.43
	75 - 79	77.25	6.10	4.70	1.60	1.17	2.26	6.92
	80 - 84	64.77	6.34	7.71	2.49	1.66	5.35	11.68
	85 +	47.82	8.29	11.35	2.20	4.11	9.96	16.27
IADL only	65 - 69	30.93	32.02	17.51	6.97	2.43	2.63	7.52
	70 - 74	22.61	32.73	21.54	4.26	3.23	6.01	9.62
	75 - 79	14.69	30.55	23.49	5.04	3.09	7.88	15.25
	80 - 84	9.19	30.15	28.39	3.84	4.35	8.19	15.90
	85 +	2.24	26.68	24.29	7.31	7.37	11.42	20.70
1-2 ADLs	65 - 69	17.23	16.17	38.76	9.82	3.71	2.66	11.65
	70 - 74	11.35	14.79	37.25	9.93	6.12	7.13	13.41
	75 - 79	9.89	11.89	38.15	11.66	5.19	8.50	14.73
	80 - 84	6.20	12.20	32.31	12.30	5.80	11.21	19.97
	85 +	3.10	8.00	31.96	14.62	8.94	13.01	20.38
3-4 ADLs	65 - 69	4.79	6.54	33.43	28.16	11.53	4.42	11.14
	70 - 74	6.19	4.25	21.35	29.69	20.25	6.21	12.06
	75 - 79	3.27	4.14	23.12	20.54	17.57	11.10	20.26
	80 - 84	2.85	3.69	15.65	23.94	22.29	13.06	18.52
	85 +	1.27	3.79	6.73	21.17	24.12	17.13	25.79
5-6 ADLs	65 - 69	7.90	6.32	9.76	7.22	38.49	5.15	25.15
	70 - 74	4.51	5.96	6.89	11.39	27.30	10.67	33.28
	75 - 79	4.19	3.62	6.25	8.50	36.98	12.99	27.47
	80 - 84	4.27	3.73	7.09	10.84	28.86	13.58	31.63
	85 +	1.57	0.72	5.49	7.99	26.92	13.51	43.80
Inst'd	65 - 69	8.04	0.29	2.67	2.71	0.03	75.61	10.65
	70 - 74	2.32	1.11	0.13	1.90	1.97	59.34	33.24
	75 - 79	1.29	1.14	0.71	0.56	2.13	60.67	33.49
	80 - 84	1.07	0.71	0.72	0.66	0.36	62.70	33.78
	85 +	0.51	0.38	0.05	0.86	0.71	51.81	45.68

Table 4.43: The 2-year transition probabilities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys as a percentage, for males and females using 5 year age groupings.

				198	84 Status	5		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65–69	90.01	2.41	1.92	0.59	0.56	0.47	4.04
	70 - 74	83.72	3.60	2.41	0.95	0.93	1.11	7.28
	75 - 79	76.74	5.40	4.08	1.30	1.24	1.94	9.30
	80 - 84	65.24	6.42	7.07	1.95	1.69	4.46	13.18
	85 +	47.30	8.67	11.07	2.58	3.40	8.20	18.78
IADL only	65 - 69	30.69	31.53	15.70	4.55	4.28	2.19	11.06
	70 - 74	25.31	32.42	17.86	3.27	2.82	4.94	13.39
	75 - 79	15.40	30.60	18.75	5.01	5.31	7.54	17.39
	80 - 84	9.40	29.64	25.34	4.47	5.85	8.02	17.28
	85 +	2.21	28.72	21.09	7.32	6.50	9.88	24.28
1-2 ADLs	65 - 69	16.25	13.44	35.96	10.50	6.05	2.94	14.87
	70 - 74	10.49	13.22	34.40	12.35	5.72	6.92	16.89
	75 - 79	9.20	12.36	34.83	10.34	6.71	6.08	20.48
	80 - 84	5.68	10.39	31.89	11.66	5.72	9.68	24.99
	85 +	3.69	6.91	28.19	13.33	10.21	13.77	23.91
3–4 ADLs	65 - 69	7.19	4.69	28.22	23.10	13.19	4.74	18.85
	70 - 74	8.40	3.31	19.30	28.75	18.52	4.00	17.72
	75 - 79	3.76	3.51	18.31	20.65	16.51	11.01	26.25
	80 - 84	2.34	4.83	14.66	21.69	21.72	12.03	22.73
	85 +	1.03	3.79	5.94	18.43	23.17	16.08	31.56
5-6 ADLs	65 - 69	7.02	5.77	8.75	9.05	32.98	5.86	30.57
	70 - 74	6.13	5.07	8.92	10.24	28.49	8.02	33.12
	75 - 79	3.96	3.81	4.91	7.73	34.63	11.54	33.41
	80 - 84	3.73	4.50	6.60	10.38	26.68	11.16	36.93
	85 +	1.33	0.58	5.22	7.61	26.42	12.04	46.80
Inst'd	65 - 69	7.54	0.97	1.63	2.30	0.12	72.22	15.22
	70 - 74	2.97	0.75	1.81	1.25	2.36	52.39	38.48
	75 - 79	1.74	1.50	0.87	0.74	1.48	56.86	36.81
	80 - 84	0.93	0.58	0.57	0.77	0.54	58.88	37.74
	85 +	0.46	0.31	0.05	0.71	0.58	49.25	48.64

- 1. mortality generally increases with disability level (that is, people with greater disability have a higher probability of death), the most notable exceptions are from relatively low mortality in the institutionalised state;
- mortality increases with age, with only a few exceptions for which mortality is usually very similar between age groups; and
- also as expected, mortality is greater for men than for women for almost all levels of disability and age groups, the only exception is for the 5–6 ADLs state, age group 70–74 years.

The first point above supports the inclusion of 'IADL only' as a separate state in the process of disability as it is predictive of mortality, as noted by Manton (1988). Also, it may be expected that the institutionalised state would have relatively low mortality, since lives admitted to an institution after a given cut-off date were excluded from the analysis, removing lives that may recover of die relatively quickly (see Section 3.7 for more detail).

From the tables in Appendix E, mortality in the 1984–89 and 1989–94 surveys is roughly comparable. However, 2–year probabilities of death in the 1982–84 surveys is roughly double that of the 5-year probabilities of death in the 1984–89 and 1989– 94 surveys, even though it is based on a shorter time period (2 years compared with 5 years). This does raise some concerns about the data. All lives in the 1982 survey are accounted for in the 1984 survey as a deceased questionnaire was administered in 1984 (see Section 3.7). However, in the 1989 and 1994 surveys this was not the case, and a new classification was introduced — 'Not in survey year'. It may be because of this change in accounting for all lives between surveys that deceased lives in 1989 and 1994 were more likely to be left out of the survey, thus reducing the overall mortality in the 1984–89 and 1989–94 surveys. However, as there is no information as to why certain lives were left out of some of the surveys it is difficult to be certain. I look at the overall mortality in more detail in Sections 6.3 to 6.5.

Table 4.44 summarises the trends of disability and recovery — it gives the overall percentage change in disability status between surveys. In this table 'less disabled'

Change in	% change between surveys							
disability	1982 - 84		1984	4 - 89	1989 - 94			
	$\mathbf{F}$	Μ	$\mathbf{F}$	М	F	Μ		
Less disabled	6.9	5.6	5.9	5.3	5.8	3.9		
Same disability	63.9	65.1	62.0	68.5	58.9	67.4		
More disabled	16.1	12.0	27.3	20.4	30.4	22.8		
Died	13.1	17.3	4.9	5.8	4.9	5.9		

Table 4.44: Summary of percentage change in disability for males and females over the 1982, 1984, 1989 and 1994 NLTCS.

covers those lives that moved to a less disabled state between surveys; 'same disability' covers those lives that were in the same state of disability in consecutive surveys; 'more disabled' covers those lives that moved to a more disabled state (other than dead) between surveys; and 'died' covers all lives that died between the surveys. This table needs to be interpreted carefully as the surveys are over different time periods, the 1982–84 survey is over 2 years while the 1984–89 and 1989–1994 surveys are over 5-year periods and the age structure of the population could be quite different over different time periods. Even so, some observations are worth noting:

- perhaps surprisingly, females consistently have a slightly higher probability of recovery from disability than males;
- 2. males have a higher tendency to stay in the same disability state;
- 3. females have a considerably higher probability of becoming more disabled than males; and
- 4. as expected, males have a higher probability of death than females.

As already mentioned above, an apparent anomaly in these surveys is that the 5-year probabilities of dying over the periods 1984–89 and 1989–94 are considerably less (1/2 - 1/3) than the 2-year probability dying in the period 1982–84. The other main difference between the surveys is that the probabilities of becoming more disabled in the 5-year periods 1984–89 and 1989–94 are almost double those of becoming more disabled in the 2-year period 1982–84. This may be expected though, as although the lives cover the same age range at the start of any period (>65 years), at the end of the 2-year period (1982–84) they are then aged over 67 years, whereas they are

aged over 70 years at the end of the 5-year periods (1984–89 and 1989–94). Thus, at the end of a 5-year period the population will be older and they would be expected to be more disabled than after a 2-year period.

### 4.3 Transformation of Transition Probabilities to Transition Intensities

By assuming the transition intensities are constant for each age group in the NLTCS data, the matrix of intensities can easily be computed from the matrix of two-year transition probabilities, calculated in Section 4.2. If P(t) is the matrix of transition probabilities over t years, and Q is the matrix of constant transition intensities (in rates per annum), then:

$$P(t) = \exp(Qt). \tag{4.33}$$

The usual problem is to calculate P(t) from Q; here it is the other way round. The method given in Section 6.4.2 of Kulkarni (1995) is easily inverted once it is noticed that P(t) and Qt have the same eigenvectors, and the eigenvalues of P(t) are the exponentiated eigenvalues of Qt.

Tables 4.45, 4.46 and 4.47 give the annual transition intensities, calculated from the 1982–84 NLTCS grouped in 5-year age bands, for men, women and in aggregate, respectively. Corresponding transition intensities are given for the 1984–89 and 1989–94 surveys (5-year age bands) in Appendix F. As predicted in Section 3.5, a number of the transition intensities are negative, which have no meaning in models of physical processes. I include the estimates here, which I will refer to as 'unconstrained MLEs', where they exist, because:

- 1. they are a good starting point for a search for the constrained MLEs (see Section 4.6); and
- 2. they are used to compare methods of calculating confidence intervals for the transition intensities in Chapter 5.

For some categories (in 1984–89: males aged 65–69 years, 80–84 years and 85+ years; in 1989–94: males and females aged 85+ years; females aged 85+ years; and

Table 4.45: The MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males using 5 year age groupings.

1982				1	984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0192	0.0080	0.0064	0.0015	0.0034	0.0247
	70 - 74		0.0232	0.0174	0.0028	0.0071	0.0061	0.0540
	75 - 79		0.0391	0.0270	0.0034	0.0053	0.0089	0.0602
	80 - 84		0.0721	0.0493	-0.0065	0.0106	0.0189	0.0713
	85 +		0.1133	0.1504	0.0091	-0.0146	0.0150	0.1253
IADL	65 - 69	0.2665		0.2656	-0.0984	0.1132	0.0072	0.0893
only	70 - 74	0.2873		0.1964	-0.0419	0.0326	0.0207	0.1233
	75 - 79	0.1561		0.2030	0.0589	0.1012	0.0768	0.0613
	80 - 84	0.1007		0.2873	0.0279	0.1756	0.0710	0.0252
	85 +	0.0087		0.2754	0.1117	-0.0413	0.0089	0.2066
1 - 2	65 - 69	0.0849	0.1588		0.3335	0.0848	0.0190	0.0904
ADLs	70 - 74	0.0150	0.1997		0.3432	-0.0132	0.1019	0.1004
	75 - 79	0.0424	0.2469		0.1248	0.1307	-0.0475	0.2381
	80 - 84	0.0466	0.0572		0.2919	-0.0449	0.0479	0.2640
	85 +	0.0895	0.0477		0.3514	0.1591	0.2143	0.0715
3 - 4	65 - 69	0.0982	-0.1033	0.6488		0.4301	0.0481	0.2269
ADLs	70 - 74	0.1082	-0.0407	0.2888		0.3098	-0.0372	0.1612
	75 - 79	0.0433	-0.0223	0.1556		0.2716	0.1455	0.2552
	80 - 84	-0.0417	0.1885	0.2328		0.7223	0.1048	0.1530
	85 +	0.0001	0.1134	0.0255		0.7467	0.2733	0.1615
5-6	65 - 69	0.0116	0.1030	-0.0195	0.3437		0.0640	0.2655
ADLs	70 - 74	0.0472	0.0407	0.1539	0.1245		0.0626	0.2259
	75 - 79	0.0208	0.0593	0.0197	0.1262		0.1020	0.2985
	80 - 84	0.0328	0.0713	0.0336	0.3402		0.0807	0.3702
	85 +	-0.0016	-0.0199	0.1075	0.1872		0.0505	0.5036
Inst'd	65 - 69	0.0393	0.0209	-0.0132	0.0347	-0.0076		0.1190
	70 - 74	0.0334	-0.0130	0.0685	-0.0248	0.0470		0.3423
	75 - 79	0.0174	0.0262	0.0093	0.0145	-0.0060		0.2973
	80 - 84	0.0043	-0.0031	-0.0043	0.0244	0.0084		0.3752
	85 +	0.0030	-0.0001	0.0009	-0.0002	0.0001		0.4809

Table 4.46: The MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for females using 5 year age groupings.

1982				1	984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0204	0.0146	0.0012	0.0033	0.0014	0.0132
	70 - 74		0.0377	0.0102	0.0068	0.0052	0.0054	0.0192
	75 - 79		0.0584	0.0233	0.0108	0.0054	0.0100	0.0292
	80 - 84		0.0619	0.0563	0.0174	0.0060	0.0312	0.0525
	85 +		0.1046	0.1072	-0.0137	0.0423	0.0738	0.0614
IADL	65 - 69	0.2622		0.2233	0.0884	0.0117	0.0189	0.0368
only	70 - 74	0.1939		0.3393	0.0124	0.0221	0.0454	0.0377
	75 - 79	0.1261		0.3783	0.0075	0.0238	0.0617	0.0825
	80 - 84	0.0801		0.5520	-0.0880	0.0850	0.0406	0.0570
	85 +	0.0143		0.4446	0.0057	0.0924	0.0842	0.0870
1 - 2	65 - 69	0.0955	0.2555		0.1439	0.0246	0.0157	0.0734
ADLs	70 - 74	0.0600	0.2345		0.1535	0.0415	0.0587	0.0714
	75 - 79	0.0741	0.1921		0.2419	0.0048	0.0602	0.0657
	80 - 84	0.0527	0.2293		0.2822	-0.0256	0.0955	0.1228
	85 +	0.0355	0.1253		0.3034	0.0037	0.0967	0.0913
3 - 4	65 - 69	-0.0187	-0.0462	0.5734		0.1706	0.0323	0.0348
ADLs	70 - 74	0.0331	-0.0493	0.3634		0.3879	0.0158	-0.0056
	75 - 79	-0.0061	-0.0292	0.4904		0.3481	0.0901	0.1033
	80 - 84	0.0021	-0.0215	0.3142		0.5072	0.0882	0.0303
	85 +	0.0055	0.0784	0.0574		0.5702	0.1708	0.0312
5-6	65 - 69	0.0390	0.0705	0.0692	0.0954		0.0417	0.1812
ADLs	70 - 74	0.0180	0.0976	0.0156	0.2165		0.1221	0.2715
	75 - 79	0.0298	0.0450	0.0133	0.1636		0.1211	0.1743
	80 - 84	0.0400	0.0494	0.0405	0.2291		0.1338	0.2339
	85 +	0.0174	-0.0140	0.0921	0.1598		0.1497	0.3416
Inst'd	65 - 69	0.0475	-0.0020	0.0129	0.0280	-0.0042		0.0587
	70 - 74	0.0138	0.0122	-0.0097	0.0208	0.0178		0.2089
	75 - 79	0.0075	0.0116	0.0013	0.0035	0.0214		0.2084
	80 - 84	0.0076	0.0070	0.0027	0.0076	0.0007		0.2096
	85 +	0.0050	0.0040	-0.0030	0.0132	0.0034		0.3094

Table 4.47: The MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1982				19	84 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0198	0.0119	0.0027	0.0029	0.0022	0.0184
	70 - 74		0.0315	0.0132	0.0056	0.0059	0.0057	0.0338
	75 - 79		0.0507	0.0254	0.0072	0.0055	0.0092	0.0420
	80 - 84		0.0647	0.0523	0.0106	0.0077	0.0263	0.0599
	85 +		0.1072	0.1157	-0.0071	0.0277	0.0582	0.0802
IADL	65 - 69	0.2607		0.2321	0.0339	0.0440	0.0138	0.0562
only	70 - 74	0.2274		0.2931	-0.0139	0.0297	0.0376	0.0640
	75 - 79	0.1391		0.3007	0.0314	0.0518	0.0670	0.0795
	80 - 84	0.0857		0.4693	-0.0542	0.1085	0.0493	0.0461
	85 +	0.0122		0.3906	0.0350	0.0503	0.0635	0.1223
1 - 2	65 - 69	0.0943	0.2214		0.1966	0.0466	0.0179	0.0822
ADLs	70 - 74	0.0457	0.2217		0.2213	0.0192	0.0710	0.0880
	75 - 79	0.0635	0.2058		0.2034	0.0398	0.0320	0.1090
	80 - 84	0.0494	0.1804		0.2711	-0.0286	0.0816	0.1657
	85 +	0.0471	0.1040		0.3084	0.0410	0.1180	0.0951
3 - 4	65 - 69	0.0161	-0.0583	0.5779		0.2342	0.0373	0.0967
ADLs	70 - 74	0.0600	-0.0416	0.3288		0.3539	0.0013	0.0577
	75 - 79	0.0119	-0.0313	0.3876		0.3170	0.1132	0.1394
	80 - 84	-0.0035	0.0162	0.2897		0.5482	0.0962	0.0387
	85 +	0.0032	0.0851	0.0521		0.5996	0.1823	0.0774
5-6	65 - 69	0.0325	0.0788	0.0418	0.1637		0.0519	0.2283
ADLs	70 - 74	0.0312	0.0672	0.0797	0.1783		0.0936	0.2504
	75 - 79	0.0264	0.0510	0.0139	0.1490		0.1131	0.2227
	80 - 84	0.0354	0.0654	0.0328	0.2464		0.1137	0.2897
	85 +	0.0138	-0.0149	0.0957	0.1670		0.1330	0.3682
Inst'd	65 - 69	0.0441	0.0081	0.0030	0.0277	-0.0047		0.0856
	70 - 74	0.0192	0.0034	0.0154	0.0088	0.0270		0.2533
	75 - 79	0.0106	0.0161	0.0028	0.0071	0.0138		0.2355
	80 - 84	0.0069	0.0055	0.0017	0.0099	0.0022		0.2407
	85 +	0.0047	0.0031	-0.0023	0.0117	0.0028		0.3368

males aged 85+ years), these estimates are not even real (the matrix of transition intensities is complex). In Section 4.4, I calculate approximate MLEs of the transition intensities in these categories by maximizing the log-likelihood over the real plane. It is interesting to note that this problem mostly arises at ages over 85 years, where the assumption of constant transition intensities may be least realistic.

## 4.4 Approximate (or Constrained to Real) Maximum Likelihood Estimates of the Transition Intensities

In the previous section, for some of the categories, transformation of the transition probabilities resulted in complex transition intensities (with the complex component being significantly non-zero). The aim of this section is, for these categories, to calculate sets of transition intensities that lie in the real plane, consistent with the maximum likelihood approach. The approach I adopt is to express transition probabilities  $P_{rr+t}^{ij}$  as functions of the transition intensities,  $P_{rr+t}^{ij} = P_{rr+t}^{ij} (\mu_r^{12}, \ldots, \mu_r^{nn-1})$ , and then maximise the likelihood function,  $L_r$ , for each age group r, where:

$$L_r \propto \prod_{all \, i,j} \left[ P_{r \, r+t}^{i \, j} \left( \mu_r^{1 \, 2}, \dots, \mu_r^{n \, n-1} \right) \right]^{n_{r \, r+t}^{i \, j}}$$
(4.34)

which is equivalent to maximising the log-likelihood function,  $l_r$ :

$$l_r \propto \sum_{all\,i,j} \left( n_{r\,r+t}^{i\,j} \,\log\left[ P_{r\,r+t}^{i\,j} \left( \mu_r^{1\,2}, \dots, \mu_r^{n\,n-1} \right) \right] \right) \tag{4.35}$$

The transition intensities,  $\mu_r^{ij}$ , I assume are constant for each age group. Given a set of transition intensities, transition probabilities can be calculated from them by solving Kolmogorov's forward equations. These can then be used to evaluate the log-likelihood function. The set of transition intensities that maximises the loglikelihood over the real region (positive and negative) of the parameter space will then be the closest set consistent with maximum likelihood approach. Although these are not the actual maximum likelihood estimates, they are the closest real equivalent (as some of the first derivatives of the likelihood function will be nonzero). These estimates, for all those groups whose transition intensities could not be calculated using the method described in Section 4.3, are given in Tables 4.48 and 4.49 for the 1984–89 and 1989–94 surveys, respectively. I refer to these estimates as 'constrained (real) MLEs'. For those groups whose transition intensities *could* be calculated using the method described in Section 4.3, this method would give identical results. The numerical method I used for this maximization process is discussed in Section 4.6.

### 4.5 Constrained (Positive) Maximum Likelihood Estimates of the Transition Intensities

The problem is now to estimate a set of transition intensities consistent with a maximum likelihood approach while ensuring that all intensities are non-negative. This can be done by constraining the transition intensities to lie in the non-negative region of the parameter space while maximising the log-likelihood by introducing a penalty function:

$$\max_{\mu_{r}^{ij}i\neq j} \left\{ \sum_{all\,i,j} \left\{ n_{r\,r+t}^{i\,j} \log \left[ P_{r\,r+t}^{i\,j} \left( \mu_{r}^{1\,2}, \dots, \mu_{r}^{n\,n-1} \right) \right] \right\} + F\left( \mu_{r}^{1\,2}, \dots, \mu_{r}^{n\,n-1} \right) \right\}$$
(4.36)

where:

$$F\left(\mu_r^{1\,2},\ldots,\mu_r^{n\,n-1}\right) = \begin{cases} 0 & \text{if } \mu_r^{i\,j} \ge 0 \ i \neq j \\ -K + \sum_{i \neq j} \min\left(0,\mu_r^{i\,j}\right) & \text{otherwise} \end{cases}$$

for each age group r. The penalty function  $F(\mu_r^{1\,2},\ldots,\mu_r^{n\,n-1})$  ensures that all transition intensities are kept positive during a computational maximization scheme — if all intensities are non-negative, F is 0; if one or more of the intensities are negative, then F has a large negative value that decreases as the intensities move towards the positive region (which ensures the gradient is in the correct direction). The constant K is chosen with respect to the size of the log-likelihood to ensure that the 'cost' of a negative intensity is sufficiently large for the intensity to be kept non-negative — it does not punish a zero transition intensity, as this is a reasonable estimate (unlike a zero transition probability). The transition intensities calculated in this way I refer to as 'constrained (positive) MLEs'. It is now only a matter of choosing a suitable maximization algorithm.

Table 4.48: Approximate (constrained to real values, but not positive) MLEs of the annual transition intensities between disability states in the 1984 and 1989 National Long-Term Care Surveys.

1984			-	1989 Status	5		
Status		IADL	1 - 2	3 - 4	5-6		
	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy		0.0127	0.0122	0.0009	0.0015	0.0017	0.0043
IADL only	0.1501		0.1860	-0.0036	0.0146	0.0222	0.0207
1-2 ADLs	0.0056	0.1731		0.1602	0.0387	0.0107	0.0008
3-4 ADLs	0.0394	-0.0015	0.2190		0.1297	0.0160	0.0168
5-6 ADLs	0.0167	0.0180	0.0534	0.0663		0.0500	0.0186
inst'd	0.0024	0.0155	-0.0004	0.0011	0.0004		0.0197
Healthy		0.0385	0.0312	0.0172	0.0037	0.0124	0.0172
IADL only	0.0476		0.0484	0.0624	0.1670	0.0168	0.0410
1-2 ADLs	0.0519	0.0929		-0.0130	0.0097	0.0915	0.0078
3-4 ADLs	-0.0078	-0.0088	0.0873		0.1883	0.2523	-0.0009
5-6 ADLs	-0.0104	-0.0087	0.0697	0.4707		-0.0168	-0.0175
inst'd	0.0171	-0.0002	-0.0022	0.0231	-0.0038		0.1031
Healthy		0.1052	0.0118	-0.0048	0.0020	0.0310	0.0184
IADL only	0.0587		0.2818	0.1817	0.0513	0.0387	0.0520
1-2 ADLs	-0.0087	0.2350		0.1037	0.1358	0.0168	0.0299
3-4 ADLs	-0.0003	0.0006	0.0000		0.0021	0.1947	0.0997
5-6 ADLs	-0.0007	0.0004	0.0000	0.0638		0.1140	0.0331
inst'd	0.0177	-0.0017	0.0040	0.0014	0.0010		0.0420
	1984 Status Healthy IADL only I–2 ADLs 3–4 ADLs 5–6 ADLs inst'd Healthy IADL only 1–2 ADLs 3–4 ADLs 5–6 ADLs inst'd Healthy IADL only 1–2 ADLs 3–4 ADLs 3–4 ADLs 3–4 ADLs 5–6 ADLs inst'd	1984   Status   Healthy   IADL only 0.1501   1–2 ADLs 0.0056   3–4 ADLs 0.0394   5–6 ADLs 0.0167   inst'd 0.0024   Healthy 0.00476   1–2 ADLs 0.0519   3–4 ADLs 0.0519   3–4 ADLs -0.0078   5–6 ADLs -0.0104   inst'd 0.0171   Healthy 1   JADL only 0.0587   1–2 ADLs -0.0087   JADL only 0.0587   1–2 ADLs -0.0087   3–4 ADLs -0.0003   5–6 ADLs -0.0003   5–6 ADLs -0.0003   5–6 ADLs -0.0007   inst'd 0.0177	$\begin{array}{llllllllllllllllllllllllllllllllllll$	1984IADL $1-2$ StatusIADL $1-2$ HealthyonlyADLsHealthy $0.0127$ $0.0122$ IADL only $0.1501$ $0.0127$ IADL only $0.1501$ $0.1860$ 1-2 ADLs $0.0056$ $0.1731$ $3-4$ ADLs $0.0394$ $-0.0015$ $5-6$ ADLs $0.0167$ $0.0180$ $0.5-6$ ADLs $0.0167$ $0.0180$ $0.024$ $0.0155$ $-0.0004$ Healthy $0.0385$ $0.0312$ IADL only $0.0476$ $0.0484$ $1-2$ ADLs $0.0519$ $0.0929$ $3-4$ ADLs $-0.0078$ $-0.0088$ $0.0873$ $5-6$ ADLs $-0.0104$ $-0.0087$ $0.0697$ inst'd $0.0171$ $-0.0002$ $-0.0022$ Healthy $0.0587$ $0.2818$ $1-2$ ADLs $-0.0087$ $0.2350$ $3-4$ ADLs $-0.0003$ $0.0006$ $0.0000$ $5-6$ ADLs $-0.0007$ $0.0044$	19841989 StatusStatusIADL $1-2$ $3-4$ HealthyonlyADLsADLsHealthy $0.0127$ $0.0122$ $0.0009$ IADL only $0.1501$ $0.1860$ $-0.0036$ $1-2$ ADLs $0.0056$ $0.1731$ $0.1602$ $3-4$ ADLs $0.0394$ $-0.0015$ $0.2190$ $5-6$ ADLs $0.0167$ $0.0180$ $0.0534$ $0.0663$ inst'd $0.0024$ $0.0155$ $-0.0004$ $0.0011$ Healthy $0.0385$ $0.0312$ $0.0172$ IADL only $0.0476$ $0.0484$ $0.0624$ $1-2$ ADLs $0.0519$ $0.0929$ $-0.0130$ $3-4$ ADLs $-0.0078$ $-0.0088$ $0.0873$ $5-6$ ADLs $-0.0104$ $-0.0087$ $0.0697$ $0.4707$ inst'd $0.0171$ $-0.0022$ $0.0231$ Healthy $0.1052$ $0.0118$ $-0.0048$ IADL only $0.0587$ $0.2818$ $0.1817$ $1-2$ ADLs $-0.0087$ $0.2350$ $0.1037$ $3-4$ ADLs $-0.0003$ $0.0006$ $0.0000$ $5-6$ ADLs $-0.0007$ $0.0004$ $0.0000$ $5-6$ ADLs $-0.0007$ $0.0004$ $0.0000$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

Table 4.49: Approximate (constrained to real but not positive values) MLEs of the annual transition intensities between disability states in the 1989 and 1994 National Long-Term Care Surveys.

	1989		1994 Status							
Group	Status		IADL	1 - 2	3 - 4	5 - 6				
		Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead		
Males	Healthy		0.0092	0.0525	0.0100	0.0296	0.0595	0.0187		
aged	IADL only	0.2609		-3.2488	1.9198	-0.0350	1.2899	-0.3004		
over	1-2 ADLs	0.0316	0.0166		-0.1336	0.0648	0.0171	0.0551		
85 years	3-4 ADLs	-0.1499	-0.3665	2.5812		0.2076	-0.9818	0.2731		
	5-6 ADLs	-0.0123	-0.0091	0.0095	-0.0111		0.2329	0.1139		
	Inst'd	0.0233	0.0107	0.0102	0.0099	0.0023		0.0749		
Females	Healthy		0.0367	0.0969	-0.0388	0.0581	0.0625	0.0102		
aged	IADL only	-0.0260		0.0528	0.5240	-0.0600	-0.0332	-0.0420		
over	1-2 ADLs	-0.0173	0.1430		0.2788	0.0199	0.1178	0.0053		
85 years	3-4 ADLs	0.1259	-0.0902	0.3304		0.2056	0.1750	0.1131		
	5-6 ADLs	0.0392	0.0125	0.0119	0.0869		0.0875	0.0336		
	Inst'd	0.0124	0.0020	0.0021	0.0013	0.0001		0.0298		
Males	Healthy		0.0279	0.0967	-0.0474	0.0513	0.0594	0.0183		
and	IADL only	-0.0434		0.0398	0.6372	-0.0585	-0.0674	-0.0488		
Females	1-2 ADLs	-0.0080	0.1263		0.3126	0.0082	0.1364	0.0010		
aged	3-4 ADLs	0.1286	-0.0653	0.4683		0.2380	0.1422	0.1287		
over	5-6 ADLs	0.0312	0.0114	-0.0009	0.0912		0.1042	0.0408		
85 years	Inst'd	0.0135	0.0036	0.0033	0.0013	0.0007		0.0352		

## 4.6 Numerical Calculation of the Constrained (Real and Positive) Maximum Likelihood Estimates

I now discuss the numerical methods used to calculate the constrained (positive) MLEs and the constrained (real) MLEs, all of which can be found in Press *et al.* (1993).

The first step is: given a set of transition intensities, to calculate the corresponding matrix of 2 or 5-year transition probabilities. This is done by solving Kolmogorov's forward equations — a set of  $n^2$  simultaneous differential equations, where n is the number of states, in this case 7 (for more detail see Section 2.3). The method I chose was a 4th order Runge-Kutta algorithm with adaptable step size. Although initially more effort to set up, it is much more efficient (in terms of computer run-time) than an equivalent algorithm with constant step size. I found the extra effort worthwhile, as the number of calculations required in the maximization routine is considerable (> 1,000).

The log-likelihood function (equation (4.35)), can now be evaluated using these transition probabilities. This process can be thought of as one function evaluation during the maximization process.

The maximization method I chose was a form of conjugate gradient method in multidimensions, called the Polak-Ribiere method. I chose this method because:

- 1. other methods, not involving derivative information, took a very long time to converge to a maximum; and
- 2. although it requires derivative information, it makes very efficient use of this information, converging rapidly.

The basic routine is to start at an initial point  $P_0$  (in our case a matrix of transition intensities) and move from point  $P_i$  to point  $P_{i+1}$  by maximizing along the uphill gradient from point  $P_i$ , with the restriction that the new gradient is conjugate to the old gradient. The method for choosing a new direction was modified by Polak & Ribiere, making the process more efficient near the maximum.

To calculate the constrained (positive) MLEs I used as a starting point the MLE transition intensities calculated in Section 4.3, where they exist (for some groups, the transition intensity matrix was complex). These contain negative transition intensities, but the penalty function F, (in equation (4.36)) ensures that these quickly become positive.

To calculate the constrained (real) MLEs, (for the groups where the MLEs of the transition intensities are complex) I use a simple starting matrix, with every annual transition intensity set to an arbitrary value of 0.01 — from which the maximization routine still converged but took considerably longer to do so than from the MLEs of the transition intensities. It is then possible to use these constrained (real) MLEs, as a starting point to calculate the corresponding constrained (positive) MLEs.

# 4.7 Constrained Maximum Likelihood Estimates Compared with Unconstrained Maximum Likelihood Estimates

I now compare the constrained MLEs, calculated using the above routines with the transition intensities calculated in Section 4.3 (transformed from the maximum likelihood estimates of the transition probabilities), which I will simply refer to as the MLEs for convenience (I include under this heading those approximate MLEs calculated in Section 4.5). I also compare 2 and 5-year transition probabilities calculated from these constrained MLEs with the original 2 and 5-year transition probabilities calculated directly from the data (given in Section 4.2). For males, females and in aggregate the constrained MLEs of the annual transition intensities calculated from the 1982–84 NLTCS in 5-year age bands are given in Tables 4.50, 4.51 and 4.52, respectively (these are directly comparable to Tables 4.45, 4.46 and 4.47). The constrained MLEs in 10-year age bands for the 1982–84 NLTCS and in 5 and 10-year age bands for the 1984–89 and 1989–94 NLTCS are given in Appendix G.

It is noticeable that for the majority of transition intensities there is very little

Table 4.50: The constrained MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, males using 5 year age groupings.

1982				19	84 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0188	0.0090	0.0051	0.0021	0.0034	0.0247
	70 - 74		0.0231	0.0176	0.0026	0.0072	0.0061	0.0540
	75 - 79		0.0393	0.0268	0.0036	0.0054	0.0085	0.0602
	80 - 84		0.0713	0.0467	0.0000	0.0065	0.0187	0.0718
	85 +		0.1145	0.1448	0.0000	0.0000	0.0168	0.1218
IADL	65 - 69	0.2608		0.1946	0.0000	0.0733	0.0065	0.0855
only	70 - 74	0.2820		0.1567	0.0000	0.0197	0.0263	0.1206
	75 - 79	0.1552		0.1974	0.0644	0.1016	0.0604	0.0647
	80 - 84	0.0986		0.2917	0.0225	0.1758	0.0713	0.0255
	85 +	0.0087		0.2708	0.0724	0.0000	0.0145	0.1971
1 - 2	65 - 69	0.0933	0.1197		0.2108	0.1310	0.0200	0.0959
ADLs	70 - 74	0.0255	0.1698		0.2763	0.0123	0.0824	0.1078
	75 - 79	0.0445	0.2204		0.1022	0.1177	0.0000	0.2310
	80 - 84	0.0384	0.0667		0.2356	0.0000	0.0485	0.2579
	85 +	0.0882	0.0429		0.4066	0.0925	0.2068	0.0866
3 - 4	65 - 69	0.0759	0.0000	0.4438		0.3163	0.0467	0.2183
ADLs	70 - 74	0.0949	0.0000	0.2107		0.2723	0.0000	0.1508
	75 - 79	0.0407	0.0000	0.1376		0.2611	0.1145	0.2629
	80 - 84	0.0000	0.1393	0.1863		0.5751	0.1038	0.1734
	85 +	0.0000	0.0859	0.0431		0.7191	0.2653	0.1722
5-6	65 - 69	0.0193	0.0703	0.0447	0.2497		0.0644	0.2680
ADLs	70 - 74	0.0501	0.0307	0.1692	0.1114		0.0511	0.2286
	75 - 79	0.0214	0.0552	0.0221	0.1251		0.0897	0.2994
	80 - 84	0.0201	0.0849	0.0429	0.2904		0.0808	0.3640
	85 +	0.0000	0.0000	0.0918	0.1486		0.0576	0.4924
Inst'd	65 - 69	0.0400	0.0154	0.0000	0.0180	0.0000		0.1192
	70 - 74	0.0292	0.0000	0.0419	0.0000	0.0368		0.3390
	75 - 79	0.0175	0.0236	0.0089	0.0111	0.0000		0.2944
	80 - 84	0.0032	0.0000	0.0000	0.0151	0.0112		0.3746
	85 +	0.0029	0.0000	0.0008	0.0000	0.0001		0.4809

Table 4.51: The constrained MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, females using 5 year age groupings.

1982				19	84 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0205	0.0144	0.0013	0.0033	0.0014	0.0132
	70 - 74		0.0376	0.0104	0.0068	0.0053	0.0054	0.0192
	75 - 79		0.0583	0.0235	0.0107	0.0055	0.0101	0.0292
	80 - 84		0.0613	0.0581	0.0155	0.0071	0.0312	0.0523
	85 +		0.1052	0.0985	0.0000	0.0352	0.0731	0.0626
IADL	65 - 69	0.2624		0.2231	0.0874	0.0123	0.0189	0.0366
only	70 - 74	0.1929		0.3320	0.0142	0.0214	0.0454	0.0380
	75 - 79	0.1265		0.3727	0.0096	0.0231	0.0616	0.0824
	80 - 84	0.0818		0.4742	0.0000	0.0398	0.0420	0.0644
	85 +	0.0143		0.4450	0.0000	0.0950	0.0844	0.0866
1 - 2	65 - 69	0.0888	0.2385		0.1386	0.0262	0.0159	0.0726
ADLs	70 - 74	0.0620	0.2185		0.1491	0.0443	0.0584	0.0704
	75 - 79	0.0720	0.1810		0.2350	0.0076	0.0604	0.0658
	80 - 84	0.0514	0.2105		0.2147	0.0106	0.0947	0.1173
	85 +	0.0352	0.1232		0.2886	0.0112	0.0976	0.0902
3 - 4	65 - 69	0.0000	0.0000	0.4933		0.1579	0.0316	0.0386
ADLs	70 - 74	0.0273	0.0000	0.3115		0.3688	0.0172	0.0000
	75 - 79	0.0000	0.0000	0.4447		0.3375	0.0897	0.1033
	80 - 84	0.0028	0.0000	0.2701		0.4703	0.0889	0.0354
	85 +	0.0065	0.0662	0.0570		0.5585	0.1695	0.0334
5-6	65 - 69	0.0363	0.0647	0.0771	0.0926		0.0418	0.1805
ADLs	70 - 74	0.0207	0.0789	0.0286	0.2117		0.1216	0.2689
	75 - 79	0.0283	0.0388	0.0217	0.1589		0.1212	0.1742
	80 - 84	0.0399	0.0417	0.0540	0.2112		0.1334	0.2320
	85 +	0.0166	0.0000	0.0817	0.1506		0.1500	0.3401
Inst'd	65 - 69	0.0460	0.0000	0.0124	0.0239	0.0000		0.0582
	70 - 74	0.0140	0.0075	0.0000	0.0149	0.0192		0.2082
	75 - 79	0.0075	0.0115	0.0014	0.0035	0.0213		0.2084
	80 - 84	0.0076	0.0068	0.0032	0.0071	0.0011		0.2096
	85 +	0.0047	0.0026	0.0000	0.0118	0.0034		0.3093

Table 4.52: The constrained MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1982				19	84 Statu	S		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0198	0.0119	0.0028	0.0029	0.0023	0.0184
	70 - 74		0.0314	0.0134	0.0054	0.0060	0.0057	0.0338
	75 - 79		0.0507	0.0254	0.0072	0.0056	0.0092	0.0420
	80 - 84		0.0644	0.0533	0.0094	0.0084	0.0263	0.0598
	85 +		0.1069	0.1115	0.0000	0.0243	0.0579	0.0807
IADL	65 - 69	0.2598		0.2285	0.0347	0.0435	0.0138	0.0562
only	70 - 74	0.2257		0.2769	0.0000	0.0241	0.0384	0.0645
	75 - 79	0.1388		0.2978	0.0321	0.0515	0.0668	0.0795
	80 - 84	0.0871		0.4307	0.0000	0.0757	0.0493	0.0525
	85 +	0.0122		0.3886	0.0323	0.0517	0.0637	0.1220
1 - 2	65 - 69	0.0977	0.1991		0.1893	0.0500	0.0179	0.0819
ADLs	70 - 74	0.0490	0.2032		0.2038	0.0269	0.0699	0.0873
	75 - 79	0.0645	0.1940		0.2001	0.0413	0.0327	0.1089
	80 - 84	0.0477	0.1749		0.2275	0.0000	0.0818	0.1605
	85 +	0.0469	0.1026		0.2980	0.0463	0.1186	0.0945
3 - 4	65 - 69	0.0083	0.0000	0.5143		0.2174	0.0374	0.0989
ADLs	70 - 74	0.0529	0.0000	0.2829		0.3386	0.0032	0.0595
	75 - 79	0.0095	0.0000	0.3554		0.3102	0.1114	0.1401
	80 - 84	0.0000	0.0197	0.2696		0.5076	0.0963	0.0454
	85 +	0.0042	0.0692	0.0563		0.5877	0.1808	0.0798
5-6	65 - 69	0.0342	0.0676	0.0523	0.1586		0.0519	0.2276
ADLs	70 - 74	0.0337	0.0535	0.0923	0.1715		0.0930	0.2497
	75 - 79	0.0271	0.0442	0.0197	0.1467		0.1135	0.2225
	80 - 84	0.0339	0.0638	0.0408	0.2263		0.1135	0.2867
	85 +	0.0130	0.0000	0.0847	0.1570		0.1337	0.3663
Inst'd	65 - 69	0.0442	0.0065	0.0052	0.0228	0.0000		0.0851
	70 - 74	0.0191	0.0036	0.0153	0.0088	0.0269		0.2533
	75 - 79	0.0107	0.0158	0.0030	0.0070	0.0138		0.2355
	80 - 84	0.0069	0.0055	0.0019	0.0094	0.0025		0.2406
	85 +	0.0044	0.0022	0.0000	0.0107	0.0027		0.3368

difference between the two methods of estimation. It is known that there will be no negative transition intensities using the contrained MLE method. However, in adjusting these to non-negative values, it may be expected that other transition intensities will change to compensate, especially transition intensities complementary to those that are negative (acting between the same states but in the opposite direction, for example  $\mu_{x+t}^{ji}$  is complementary to  $\mu_{x+t}^{ij}$ ). I make the following observations:

- almost all negative transition intensities in the MLE routine, were estimated as zero using the constrained MLE routine, however there are exceptions (for example, the transition intensity from 1–2 ADLs to 5–6 ADLs for females aged 80–84 years);
- the differences between transition intensities complementary to negative MLE transition intensities, are not much larger than the differences in general, which may be surprising; and
- 3. there are significant differences between constrained (positive) and unconstrained MLEs of some intensities that are not complementary to negative MLE transition intensities (for example, for males aged 65–75, the transition between 1–2 ADLs and Healthy).

The first point above suggests that it may be more efficient to calculate the maximum likelihood estimates of the transition probabilities (from data), transform them to transition intensities and then set any negative transition intensities to zero, rather than using a constrained maximum likelihood approach. I would argue against this method as:

- not all of the constrained MLEs of negative transition intensities were zero all intensities should have the possibility to take a positive value;
- 2. there is no intuitive reason why they should be set to zero, unless the physical process being modelled dictates this which is not the case here;
- 3. quite a few transition intensities changed significantly, compensating in some way for the transition intensities that were forced to be non-negative; and

Table 4.53: Comparison of log-likelihood values for the unconstrained MLEs (and constrained (real) MLEs), adjusted MLEs and the constrained (positive) MLEs of the transition intensities calculated from the 1982-1984 NLTCS.

Gender	Age	Log-	likelihood va	alue for:
	group	Original	Adjusted	Constrained
		MLE	MLE	MLE
M & F	65 - 69	-4186.71	-4188.62	-4187.52
	70 - 74	-4780.57	-4782.02	-4781.23
	65 - 74	-9036.11	-9038.72	-9037.15
	75 - 79	-4580.38	-4580.93	-4580.62
	80 - 84	-3576.82	-3580.53	-3577.39
	75 - 84	-8202.97	-8203.06	-8203.00
	85 +	-3283.32	-3284.97	-3284.39
$\mathbf{F}$	65 - 69	-2279.62	-2281.16	-2280.45
	70 - 74	-2718.12	-2719.78	-2718.94
	65 - 74	-5044.61	-5045.82	-5045.07
	75 - 79	-2863.02	-2863.43	-2863.17
	80 - 84	-2478.92	-2484.38	-2480.03
	75 - 84	-5376.17	-5378.18	-5376.72
	85 +	-2435.97	-2438.00	-2437.24
Μ	65 - 69	-1863.88	-1869.68	-1865.89
	70 - 74	-1999.99	-2004.51	-2002.36
	65 - 74	-3907.77	-3914.12	-3910.72
	75 - 79	-1660.43	-1664.38	-1663.37
	80 - 84	-1066.74	-1068.54	-1067.42
	75 - 84	-2752.60	-2752.62	-2752.61
	85 +	-813.83	-815.24	-814.45

4. this transformation of transition probabilities to transition intensities does not alway produce a set of real transition intensities from which to start.

Table 4.53 provides further support for not simply setting negative intensities to zero. The table shows for the transition intensities calculated from the 1982–84 NLTCS the log-likelihood (equation (4.35)) for: the unconstrained MLEs (and the constrained (real) MLEs) — highest value by definition; the unconstrained MLEs (and the constrained (real) MLEs) with all negative transition intensities set to zero (I refer to these as adjusted MLEs); and the constrained (positive) MLEs. The same likelihood values for the 1984–89 and 1989–94 NLTCS are given in Appendix H. The constrained (positive) MLEs, as well as always existing, are always better and in some cases substantially better than the adjusted MLEs, as well as making sense intuitively. Further comparisons can be made by solving the Kolmogorov equations to calculate 2 or 5-year transition probabilities from the constrained (positive) MLEs which are then directly comparable to the 2 or 5-year transition probabilities calculated from the data (given in Tables 4.41 - 4.43). These transition probabilities calculated from the constrained (positive) MLEs of the transition intensities for the 1982–84 NLTCS grouped in 5-year age bands for men, women and in aggregate are given in Tables 4.54, 4.55 and 4.56, respectively. The same transition probabilities in 5-year age bands for the 1984–89 and 1989–94 NLTCS are given in Appendix I. These can be thought of as the matrices of transition probabilities closest (in the sense of likelihood) to the original transition probabilities, that are consistent with a continuous-time Markov chain with constant (positive and real) transition intensities. This means that they can also be transformed into non-negative transition probabilities over any time span, using the methods in Section 3.5.

As expected there are no '0 transition probabilities' in these estimates (even though some of the transition intensities are zero) and overall there is not much difference between the probabilities using these two methods of estimation. It is not estimation of the transition probabilities that is of primary interest though — the main aim is the estimation of the underlying transition intensities, which can then be applied to the problem of estimating the cost of disability in LTC insurance.

Looking back at Table 3.36 it is noticeable that the group in 1982 with the least exposure is the 3–4 ADL group. It is then not surprising that differences in the transition probabilities out of the 3–4 ADL state are generally the greatest. This is a function of the maximum likelihood technique as the log-likelihood function is a weighted sum of log-probabilities, the weight being the number of transitions (see equation (4.35)) — the fewer the number of transitions, the less effect the probability has on the overall log-likelihood value, which gives that probability greater freedom to adjust to other effects. Another interesting observation is that the force of mortality from all states is almost unaffected, even from the 3–4 ADL state.

In the next chapter, I estimate variances for the constrained (positive) MLEs, and use them to graduate these point estimates using parametric methods, to get smoothed, time-continuous transition intensities, better for use in applications.

Table 4.54: The 2-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys as a percentage, for males using 5 year age groupings.

				198	84 Status	3		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	89.05	2.17	1.47	0.64	0.61	0.63	5.43
	70 - 74	81.22	2.69	2.28	0.77	0.94	0.98	11.11
	75 - 79	76.00	4.37	3.17	0.85	1.34	1.47	12.81
	80 - 84	66.04	6.54	5.88	1.18	1.70	2.94	15.72
	85 +	46.20	9.49	10.26	3.20	2.24	4.39	24.23
IADL only	65 - 69	30.27	31.16	12.30	2.97	6.06	1.60	15.64
	70 - 74	30.03	32.08	10.09	2.70	2.02	3.03	20.05
	75 - 79	16.51	31.15	11.72	5.12	8.66	6.25	20.59
	80 - 84	9.71	28.44	17.56	5.73	10.01	7.59	20.96
	85 +	2.13	33.75	13.38	6.67	5.04	6.16	32.87
1-2 ADLs	65 - 69	14.61	8.02	32.54	10.02	10.71	3.47	20.63
	70 - 74	9.20	10.33	31.49	14.95	5.19	6.29	22.55
	75 - 79	7.48	12.55	27.33	6.32	9.42	2.70	34.20
	80 - 84	4.23	5.64	31.09	9.47	5.81	5.68	38.08
	85 +	5.19	3.92	18.51	10.98	12.36	15.80	33.23
3-4 ADLs	65 - 69	10.61	3.71	18.73	17.09	14.57	5.16	30.13
	70 - 74	11.49	2.94	14.01	28.29	15.07	2.21	25.99
	75 - 79	4.89	2.67	6.97	21.94	14.08	9.27	40.19
	80 - 84	2.17	7.69	10.38	15.81	18.55	8.62	36.78
	85 +	0.45	3.40	3.87	11.30	21.00	13.40	46.58
5-6 ADLs	65 - 69	6.15	4.83	7.97	10.04	28.24	6.60	36.17
	70 - 74	7.88	3.82	11.67	8.66	30.24	4.69	33.05
	75 - 79	3.62	4.01	2.89	6.72	31.82	8.28	42.66
	80 - 84	2.54	5.87	5.80	9.53	23.37	7.19	45.70
	85 +	0.47	0.79	3.96	5.41	24.91	6.28	58.19
Inst'd	65 - 69	6.91	1.58	0.80	1.30	0.51	68.18	20.72
	70 - 74	3.94	0.67	3.43	1.05	2.75	41.37	46.79
	75 - 79	2.63	2.05	1.12	0.94	0.51	49.42	43.33
	80 - 84	0.39	0.23	0.26	0.92	1.10	44.79	52.31
	85 +	0.25	0.03	0.07	0.02	0.02	37.96	61.66

Table 4.55: The 2-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys as a percentage, for females using 5 year age groupings.

				198	34 Status	3		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	90.78	2.59	2.27	0.55	0.52	0.35	2.94
	70 - 74	85.59	4.27	2.50	1.08	0.93	1.20	4.43
	75 - 79	77.25	6.10	4.70	1.60	1.17	2.26	6.92
	80 - 84	64.78	6.34	7.72	2.48	1.66	5.35	11.68
	85 +	47.90	8.34	10.84	2.69	4.03	9.94	16.26
IADL only	65 - 69	30.96	32.06	17.45	6.97	2.43	2.62	7.51
	70 - 74	22.60	32.86	21.44	4.26	3.22	6.01	9.61
	75 - 79	14.69	30.60	23.45	5.04	3.09	7.88	15.25
	80 - 84	9.15	30.43	26.73	5.57	4.11	8.16	15.86
	85 +	2.24	26.78	24.53	6.89	7.44	11.42	20.71
1-2 ADLs	65 - 69	16.61	15.77	39.72	9.80	3.73	2.68	11.69
	70 - 74	11.37	14.36	37.65	9.91	6.15	7.14	13.41
	75 - 79	9.77	11.65	38.49	11.64	5.21	8.51	14.74
	80 - 84	6.22	12.06	33.41	11.04	5.98	11.25	20.03
	85 +	3.11	7.83	32.37	14.28	8.97	13.03	20.40
3-4 ADLs	65 - 69	6.48	7.64	30.67	28.83	10.95	4.34	11.10
	70 - 74	6.13	5.77	19.86	30.17	19.74	6.18	12.15
	75 - 79	3.65	4.81	22.13	20.75	17.36	11.06	20.23
	80 - 84	2.83	4.04	15.16	24.60	21.94	12.99	18.45
	85 +	1.29	3.61	6.42	21.61	24.17	17.12	25.78
5-6 ADLs	65 - 69	7.64	6.20	9.92	7.18	38.74	5.17	25.16
	70 - 74	4.53	5.52	6.93	11.42	27.64	10.70	33.25
	75 - 79	4.09	3.48	6.35	8.46	37.13	13.01	27.48
	80 - 84	4.28	3.60	7.21	10.58	29.09	13.59	31.65
	85 +	1.51	1.21	5.26	7.64	27.08	13.50	43.80
Inst'd	65 - 69	7.92	0.54	2.42	2.41	0.41	75.67	10.64
	70 - 74	2.31	0.90	0.70	1.55	1.95	59.37	33.21
	75 - 79	1.29	1.14	0.72	0.56	2.13	60.67	33.49
	80 - 84	1.07	0.71	0.72	0.65	0.36	62.70	33.78
	85 +	0.49	0.28	0.22	0.83	0.67	51.83	45.67

Table 4.56: The 2-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys as a percentage, for males and females using 5 year age groupings.

				198	34 Status	3		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	90.01	2.41	1.92	0.59	0.56	0.47	4.04
	70 - 74	83.73	3.60	2.41	0.95	0.93	1.11	7.28
	75 - 79	76.74	5.40	4.08	1.30	1.24	1.94	9.30
	80 - 84	65.24	6.41	7.08	1.95	1.69	4.46	13.18
	85 +	47.35	8.67	10.83	2.81	3.37	8.20	18.78
IADL only	65 - 69	30.67	31.61	15.67	4.55	4.27	2.19	11.05
	70 - 74	25.27	32.53	17.48	3.64	2.77	4.93	13.38
	75 - 79	15.39	30.65	18.73	5.01	5.31	7.53	17.38
	80 - 84	9.38	29.79	24.44	5.51	5.62	8.00	17.25
	85 +	2.21	28.85	21.15	7.10	6.53	9.88	24.27
12  ADLs	65 - 69	16.29	12.84	36.49	10.47	6.08	2.95	14.89
	70 - 74	10.52	12.85	35.00	11.99	5.78	6.94	16.92
	75 - 79	9.22	12.06	35.08	10.33	6.73	6.09	20.50
	80 - 84	5.67	10.36	32.47	10.87	5.92	9.69	25.03
	85 +	3.70	6.76	28.44	13.17	10.23	13.78	23.93
3-4 ADLs	65 - 69	7.15	6.27	26.77	23.56	12.68	4.73	18.83
	70 - 74	8.31	4.64	18.09	29.12	18.18	3.98	17.67
	75 - 79	3.73	4.37	17.57	20.78	16.35	10.97	26.23
	80 - 84	2.46	4.81	14.48	22.31	21.26	12.01	22.68
	85 +	1.04	3.53	5.76	18.78	23.22	16.09	31.58
5-6 ADLs	65 - 69	7.03	5.54	8.85	9.01	33.15	5.86	30.57
	70 - 74	6.15	4.71	9.13	10.16	28.68	8.03	33.15
	75 - 79	3.97	3.64	4.98	7.71	34.73	11.56	33.42
	80 - 84	3.68	4.50	6.66	10.10	26.94	11.17	36.94
	85 +	1.27	1.13	4.99	7.28	26.54	12.02	46.77
Inst'd	65 - 69	7.53	0.94	1.59	2.01	0.48	72.24	15.21
	70 - 74	2.97	0.75	1.81	1.24	2.36	52.40	38.48
	75 - 79	1.74	1.48	0.88	0.74	1.48	56.87	36.81
	80 - 84	0.93	0.58	0.57	0.77	0.53	58.88	37.74
	85 +	0.44	0.25	0.18	0.69	0.55	49.26	48.63

### Chapter 5

# Confidence Intervals and Graduation of Transition Intensities in the Disability Model

#### 5.1 Introduction

The main aim of this chapter is to graduate the transition intensities that were calculated in the previous chapter. I first, however, calculate variance estimates of the transition intensities. These are useful for two reasons:

- 1. to use as weights in the graduation process; and
- 2. to estimate confidence intervals for the transition intensities, against which the graduated transition intensities can be compared.

In Section 5.2, I look at two different methods for calculating variance estimates of the transition intensities, both of which use asymptotic maximum likelihood theory:

- a standard method, which I will refer to as the asymptotic method, for which the likelihood function factorises to provide concise variance estimates — which is only valid given complete life-history data (i.e. the timing of each transition between states); and
- 2. a method which requires explicit calculation of the information matrix, which I will refer to as the information matrix method – which is valid given only

the partial data available from the NLTCS (i.e. the number of lives in each state at the start and end of the survey period).

The reason for looking at the first method is that this theory provides a very concise variance estimate, one which can even be calculated using the partial data available — this is, however, only an estimate of an estimate. The interest lies in comparing estimates from these two methods, the difference between them is an indication of the amount of 'information' lost by having access only to partial data, as the NLTCS provides, rather than complete life-history data.

I investigate these differences further, in Section 5.3, by comparing these variance estimates in three simple models. Then in Section 5.4, I describe a method for estimating the variance of the constrained (positive) MLEs of the transition intensities, and provide tables of these estimates. I use these as weights in Sections 5.5 and 5.6 where I graduate the constrained MLEs using parametric methods, to get smoothed, time-continuous transition intensities, more reasonable for use in applications. At the end of each of these graduation sections I provide graphs of the graduated transition intensities and confidence intervals, show that the functions do provide a good fit and give a table of parameter values for the functional forms of the transition intensities.

For brevity, in this chapter as in the previous one, I only use data from the 1982– 84 NLTCS to illustrate the methods and concentrate on the data grouped by 5-year age bands (65–69 years, 70–74 years, 75–80 years and 80–84 years and 85+ years ). The corresponding results for the 1984–89, 1989–94 surveys and for data grouped into 10-year age bands are given in the specified Appendices. Comments on these results are given in the text, where they differ from those illustrated.

### 5.2 Comparison of Two Methods for Estimating the Variance of the Transition Intensities

In this section I illustrate the results using one specific data set, that for males and females aged 65–74 years in the 1982–84 NLTCS, though the results presented here hold for all the other data sets. I first look at a standard method for estimating the

variance of the transition intensities, which can be used when complete life histories are known.

Given the life histories of each individual life in the NLTCS (i.e. the timing of each transition between states), the calculation of variance estimates of the transition intensities would be straightforward. As summarised by Macdonald (1996), for age group r over period t, let:

$$^{k}W_{rr+t}^{i} =$$
waiting time in state *i* for *k*th life (5.37)

$$^{k}N_{r\,r+t}^{i\,j}$$
 = number of transitions from state *i* to state *j* by *k*th life (5.38)

Let  $W_{rr+t}^i = \sum_k {}^k W_{rr+t}^i$  and  $N_{rr+t}^{ij} = \sum_k {}^k N_{rr+t}^{ij}$ , and using lower case symbols for observed samples, then the likelihood function for age group r,  $L_r$  can be shown to be:

$$L_r\left(\mu_r^{1\,2}\dots\mu_r^{6\,7}\right) = \prod_{i=1}^7 \left[ e^{-\mu_r^{i\,\cdot}w_{r\,r+t}^i} \prod_{\substack{j=1\\j\neq i}}^7 \left(\mu_r^{i\,j}\right)^{n_{r\,r+t}^{i\,j}} \right]$$
(5.39)

where  $\mu_r^{i} = \sum_{j \neq i} \mu_r^{ij}$ . From which it can be shown that the  $\hat{\mu}_r^{ij}$  are asymptotically independent with distribution:

$$\hat{\mu}_r^{ij} - \mu_r^{ij} \stackrel{asy.}{\sim} \mathcal{N}\left(0, \frac{\mu_r^{ij}}{\mathcal{E}\left[W_{r\,r+t}^i\right]}\right)$$
(5.40)

So, given the life history of each individual,  $\mu_r^{ij}/E\left[W_{rr+t}^i\right]$  is an asymptotically unbiased estimate of the variance of  $\hat{\mu}_r^{ij}$ , which can be estimated by using the MLEs as approximations of the actual population transition intensities, and the actual waiting times as approximations for the expected waiting times. Even with the incomplete data provided by the the NLTCS, this variance estimate can be approximated — by using the unconstrained MLEs of the transition intensities (where they are positive) as approximations to the actual population transition intensities, and the expected waiting times can be approximated using the census method:

$$\mathbf{E}\left[W_{r\,r+t}^{i}\right] \approx t \times \left(\frac{\mathbf{n}_{r\,r+t}^{i \cdot} + \mathbf{n}_{r\,r+t}^{\cdot \, j}}{2}\right) \tag{5.41}$$

where  $n_{rr+t}^{i}$  and  $n_{rr+t}^{j}$  are the number of lives in state *i* for age group *r* and in state *j* in age group r + t, respectively. These variance estimates for the data set of males and females aged 65–74 years in the 1982–84 NLTCS are given in Table 5.57 — it should be noted that these are approximate estimates as if the life history of each individual were given (which is not the case in the NLTCS). I now look at a method for estimating the variance estimates which does not assume that complete life histories are known.

This standard method from MLE theory uses the information matrix (for those groups for which there exist unconstrained MLEs of the transition intensities), see Morgan (2000) for example. Although the method is straightforward, it is notationally difficult to represent, partly due to the parameters (transition intensities) themselves being in matrix notation. I will write, for convenience, the 36 non-zero transition intensities in the model as  $\mu_r = (\mu_r^1 \dots \mu_r^{36})$ , where r denotes age group. The elements of the information matrix  $I(\mu_r)$  can now easily be defined:

$$\left[I\left(\mu_{r}\right)\right]_{ij} = -\mathbf{E}\left[\frac{\partial^{2}l_{r}}{\partial\mu_{r}^{i}\partial\mu_{r}^{j}}\right] \text{ for } i, j = 1, 2, \dots, 36$$

and where  $l_r$  is the log-likelihood function for age group r, defined in equation (4.35). We then have, from Morgan (2000), that if  $\hat{\mu}_r^i$  are any unbiased estimates of the  $\mu_r^i$ :

$$\operatorname{Var}(\hat{\mu}_{r}^{i}) \geq \left[I\left(\mu_{r}\right)^{-1}\right]_{i\,i}$$

and that for large samples that this is a good approximation to the variance or more specifically that:

$$\hat{\mu}_{r}^{i} - \mu_{r}^{i} \sim \mathcal{N}\left(0, \left[I\left(\mu_{r}\right)^{-1}\right]_{i\,i}\right) \text{ for } i = 1, 2, \dots, 36$$

 $I(\mu_r)$  can be approximated by substituting the maximum likelihood estimates of the transition intensities  $(\hat{\mu}_{x+t}^i)$  in place of the population (theoretical) transition intensities  $(\mu_r^i)$ . The elements of  $I(\mu_r)$  can then be calculated by numerically differentiating the log-likelihood function (see Press *et al.* (1988) for more detail).

When estimating the variances of the transition intensities it is necessary to use the unconstrained MLEs, rather than the constrained (positive or real) MLEs (the set of first derivatives of the log-likelihood will be non-zero for the constrained MLEs). This is not a problem for the data set that I have chosen here, as the unconstrained MLEs are real. The variance estimates for the data set of males and females aged 65–74 years in the 1982–84 NLTCS are given in Table 5.58 — these are directly comparable to the variance estimates calculated using the asymptotic method. Table 5.57: Variance estimates of the unconstrained MLEs of the transition intensities using the asymptotic method, for males and females aged 65–74 years in the 1982–84 NLTCS.

1984				1989 Status			
Status		IADL	1 - 2	3 - 4	5 - 6		
	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy		$1.55\times10^{-6}$	$7.81\times10^{-7}$	$2.46\times 10^{-7}$	$2.63\times 10^{-7}$	$2.27\times 10^{-7}$	$1.57 \times 10^{-6}$
IADL only	$1.65 \times 10^{-4}$		$1.82 \times 10^{-4}$	$4.78\times10^{-6}$	$2.44\times10^{-5}$	$1.78 \times 10^{-5}$	$4.20 \times 10^{-5}$
1-2 ADLs	$4.58 \times 10^{-5}$	$1.49 \times 10^{-4}$		$1.41 \times 10^{-4}$	$2.27 \times 10^{-5}$	$2.94 \times 10^{-5}$	$5.85 \times 10^{-5}$
3–4 ADLs	$6.49\times10^{-5}$	_ (1)	$6.80  imes 10^{-4}$		$4.57\times 10^{-4}$	$2.78  imes 10^{-5}$	$1.14 \times 10^{-4}$
5-6 ADLS	$4.52\times 10^{-5}$	$1.06  imes 10^{-4}$	$8.75  imes 10^{-5}$	$2.50  imes 10^{-4}$		$1.05  imes 10^{-4}$	$3.51 \times 10^{-4}$
Inst'd	$4.48\times10^{-5}$	$8.00 \times 10^{-6}$	$1.48 \times 10^{-5}$	$2.57 \times 10^{-5}$	$1.53 \times 10^{-5}$		$2.52\times10^{-4}$

(1) Undefined, since the unconstrained MLE is negative.

Table 5.58: Variance estimates of the unconstrained MLEs of the transition intensities using the information matrix, for males and females aged 65–74 years in the 1982–84 NLTCS.

1984				1989 Status			
Status		IADL	1 - 2	3 - 4	5 - 6		
	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy		$3.95\times10^{-6}$	$3.41 \times 10^{-6}$	$1.47\times 10^{-6}$	$1.02\times 10^{-6}$	$4.67\times 10^{-7}$	$2.00 \times 10^{-6}$
IADL only	$3.62 \times 10^{-4}$		$1.06 \times 10^{-3}$	$3.69 \times 10^{-4}$	$2.00 \times 10^{-4}$	$7.10 \times 10^{-5}$	$1.38 \times 10^{-4}$
1-2 ADLs	$2.21\times 10^{-4}$	$8.57  imes 10^{-4}$		$1.23 \times 10^{-3}$	$4.81\times 10^{-4}$	$1.15  imes 10^{-4}$	$2.12  imes 10^{-4}$
3-4 ADLs	$3.80  imes 10^{-4}$	$1.15  imes 10^{-3}$	$5.78  imes 10^{-3}$		$3.27 \times 10^{-3}$	$3.12 \times 10^{-4}$	$7.60  imes 10^{-4}$
5-6 ADLS	$1.95  imes 10^{-4}$	$6.00  imes 10^{-4}$	$1.31 \times 10^{-3}$	$1.64  imes 10^{-3}$		$2.88\times10^{-4}$	$7.42  imes 10^{-4}$
Inst'd	$7.51 \times 10^{-5}$	$4.19 \times 10^{-5}$	$9.96 \times 10^{-5}$	$1.13 \times 10^{-4}$	$6.92 \times 10^{-5}$		$3.62 \times 10^{-4}$

Table 5.59: Variance estimates of the unconstrained MLEs of the transition intensities using the asymptotic method as a percentage of those using the information matrix, for males and females aged 65–74 years in the 1982–84 NLTCS.

1984			198	89 Status	3		
Status		IADL	1 - 2	3 - 4	5-6		
	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
	%	%	%	%	%	%	%
Healthy		39.25	22.90	16.71	25.81	48.72	78.29
IADL only	45.45		17.20	1.29	12.24	25.01	30.47
1-2 ADLs	20.73	17.37		11.49	4.72	25.50	27.62
3–4 ADLs	17.10	_ (1)	11.75		13.99	8.91	14.95
5–6 ADLs	23.18	17.62	6.68	15.24		36.60	47.24
Inst'd	59.64	19.09	14.90	22.73	22.11		69.61

(1) Undefined, since the unconstrained MLE is negative.

Table 5.59 illustrates the differences between these two variance estimates — it gives the variance estimates calculated using the asymptotic method (which can be thought of as an estimate of a variance estimate) as a percentage of those calculated using the information matrix. The variance estimates calculated using the asymptotic method are between 79.3% and 1.3% of those calculated using the information The fact that the variance estimates using the asymptotic method are matrix. smaller than those calculated using the information matrix is not surprising, since the asymptotic method assumes that more information is available — complete life histories of all the individuals (aggregated), whereas the information matrix method only uses the information of which state an individual is in at the start and end of the survey period. Given that the variance is a measure of the uncertainty of a point estimate of a transition intensity, it would be expected that using more (relevant) information would provide a more certain estimate and, correspondingly, a smaller variance. This heuristic argument may not be the only reason why the variance estimates differ, and I first investigate the asymptotic estimate further, before looking at simulations of both methods.

Another possible reason for these differences may lie in using the census method to approximate the expected waiting time in the asymptotic method — the census method will only give a reasonable approximation if the change in exposure in a given state is roughly linear over time. Table 5.60: Comparison of methods for calculating the expected waiting time for males and females aged 65–74 years in the 1982–84 NLTCS.

		Expected	d waiting	time for s	state:	
Method used to		IADL	1 - 2	3 - 4	5-6	
Calculate $\mathbf{E}\left[W_{r}^{i}\right]$	Healthy	only	ADLs	ADLs	ADLs	Inst'd
Census Method	15931.50	1468.81	1489.13	650.96	688.71	689.89
Expectations	15931.60	1444.07	1495.43	654.58	679.68	699.05

As a check of how good the census method approximation is, it can be compared to the actual expected waiting time (using the unconstrained MLEs as approximations for the population transition intensities), which can easily be calculated using Norberg's equations (Norberg, (1995)). For example, in the notation of Section 2.3, by setting the force of interest,  $\delta$ , equal to 0%, and the payment function  $b^i = 1$ (with  $b^j = 0$  for  $i \neq j$ ), the solutions to Norberg's equations, using the unconstrained MLEs of the transition intensities, will give (an approximation to),  $E[W_r^i(j)]$ , the expected waiting time in state *i*, given that a life started in state *j* for age group *r*. The expected waiting time in state *i* is then simply the weighted sum of these conditional waiting times — the weights being the probability of starting in a given state:

$$\mathbf{E}\left[W_{r}^{i}\right] = \frac{\sum_{j=1}^{7} \mathbf{n}_{r\,r+t}^{j\,\cdot} \times \mathbf{E}\left[W_{r}^{i}(j)\right]}{\sum_{j=1}^{7} \mathbf{n}_{r\,r+t}^{j\,\cdot}}$$
(5.42)

Table 5.60 compares the census method approximations to the expected waiting times and the expected waiting times, calculated as described above. The census method does give very good approximations to the expected waiting times, with a maximum difference of 1.71% — clearly not enough to account for the differences in the variance calculations. Furthermore, the census method does not always over-estimate the expected waiting times (for example in the 1–2 ADLs state), which would need to be the case to account for the underestimated variances.

To investigate further this discrepancy between variance estimates, it is possible to run simulations from a given set of transition intensities, to see how these two variance estimates behave — and the simulations themselves will provide a third method to estimate the variances of the transition intensities, as further comparison. For a given matrix of transition intensities, the corresponding matrix of transition probabilities,  $\hat{P}_{rr+t}^{ij}$ , can be calculated by solving Kolmogorov's equations. Then for each starting state i (i = 1, 2, ..., 7), the number of lives in state i in age group r who are in state j after time t,  $N_{rr+t}^{ij}$ , are multinomially distributed with:

$$P\left[N_{rr+t}^{ij} = n_{rr+t}^{ij} \text{ for } 1 \le j \le 7\right] = \frac{n_{rr+t}^{i}}{\prod_{j=1}^{7} n_{rr+t}^{ij}!} \prod_{j=1}^{7} \left(\hat{P}_{rr+t}^{ij}\right)^{n_{rr+t}^{ij}}$$
(5.43)

where  $n_{rr+t}^{i}$  are the number of lives starting in state *i* in age group *r* and for any collection  $n_{rr+t}^{i1}$ ,  $n_{rr+t}^{i2}$ ,  $\dots$ ,  $n_{rr+t}^{i7}$  of non-negative integers with sum  $n_{rr+t}^{i}$ . Then given a simulated matrix of numbers of transitions between states, the corresponding matrix of transition probabilities can be calculated (as in Section 4.2), which can then be transformed, using the method described in Section 4.3, to a matrix of transition intensities. These can then be used to calculate variance estimates of the transition intensities using both of the methods described earlier in this Section. Also, given sufficient simulations (investigated further in Section 5.4), these simulated transition intensities themselves can be used to calculate a 'boot-strap' estimate of the variance of the transition intensities.

Figures 5.21 to 5.26 show the results for 500 simulations based on the probabilities calculated from the annual constrained (positive) MLEs of transition intensities for males and females aged 65–74 years in the 1982–84 NLTCS, for starting states 'Healthy' through to 'Institutionalized', respectively. It should be noted that the 'bootstrap' variance estimates are not accurate in these figures as they are only based on 500 simulations, and it takes more that 5,000 simulations before the 'bootstrap' variance estimates converge (see Section 5.4). I include them here only as an indication of the variance of the simulated samples. I used the constrained (positive) MLEs rather than the unconstrained MLEs to avoid any problems associated with negative transition intensities and because none of the 'true' population transition intensities would be negative.

A few points are worth noting about these simulations:

the asymptotic variance estimates change almost linearly with varying transition intensity, which is not surprising, as the variance is estimated as μ<sup>ij</sup>/<sub>E[W<sup>i</sup><sub>rr+t</sub>]</sub>
 — also suggesting that the expected waiting time for any given state remains
 relatively constant with varying intensities;
- 2. the variance estimates calculated using the information matrix are much more scattered for any given value of  $\mu_r^{ij}$ , showing dependency on other transition intensities, which is expected as they are not independent, though overall they do exhibit a similar trend — the variance estimates increase as the transition intensities increase in magnitude;
- even though the 'population' transition intensities are all non-negative (as the constrained (positive) MLEs were used), from random fluctuation quite a number of the unconstrained MLEs based on the simulated samples contained negative transition intensities;
- 4. the MLEs of the transition intensities for 2 of the 500 simulated samples were complex, showing that, although not common, random deviation in data sets generated by non-negative 'population' transition intensities, can lead to MLEs that are complex; and
- 5. even though the 'bootstrap' variance estimates have not yet converged, they do give very good agreement with the variance estimates calculated using the information matrix, which is expected since they are based on the same information.

Given the data that are available, the variance estimates calculated using the information matrix are approximate lower bounds (they would be lower bounds if the population transition intensities,  $\mu_{x+t}^{jk}$ , were used in the calculations). The asymptotic method, even though it can be approximated using the same data as the information matrix method, assumes that more information is known than the NLTCS provides and so the method is not valid. The differences in the variance estimates between the two methods gives an indication of how much information is lost by only knowing which states individual lives start and end in and not their life history. (By life history I mean the knowledge of the states each life moved to and the timing of these transitions.) In the disability model (7 states and 36 non-zero transition intensities), having only the limited information means that the variance estimates of the transition intensities are up to 77 times greater than if the life history of each individual was known (or equivalently, the confidence intervals of



Figure 5.21: Variance estimates of the transition intensities out of the 'Healthy' state using the asymptotic method ( $\bullet$ ), the information matrix ( $\times$ ) and the 'boot-strap' variance estimate (horizontal line) for 500 simulations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females aged 65–74 years in the 1982–84 NLTCS.

the point estimates are up to 8.8 times larger) — a substantial loss of information. In the next section I will look at some simpler models to compare these variance estimates further.

## 5.3 Comparison of Variance Estimates in Three Simple Models

In this section I look at three very simple Markov models, two 2-state models and a 3-state model each with known constant transition intensities and compare variance



Figure 5.22: Variance estimates of the transition intensities out of the 'IADL only' state using the asymptotic method ( $\bullet$ ), the information matrix ( $\times$ ) and the 'boot-strap' variance estimate (horizontal line) for 500 simulations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.23: Variance estimates of the transition intensities out of the '1–2 ADLs' state using the asymptotic method (•), the information matrix (×) and the 'boot-strap' variance estimate (horizontal line) for 500 simulations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.24: Variance estimates of the transition intensities out of the '3–4 ADLs' state using the asymptotic method ( $\bullet$ ), the information matrix ( $\times$ ) and the 'boot-strap' variance estimate (horizontal line) for 500 simulations, based on the annual constrained (positive) MLEs of transition intensities (vertical line) for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.25: Variance estimates of the transition intensities out of the '5–6 ADLs' state using the asymptotic method ( $\bullet$ ), the information matrix ( $\times$ ) and the 'boot-strap' variance estimate (horizontal line) for 500 simulations, based on the constrained (positive) MLEs of transition intensities (vertical line) for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.26: Variance estimates of the transition intensities out of the 'Institutionalized' state using the asymptotic method ( $\bullet$ ), the information matrix ( $\times$ ) and the 'boot-strap' variance estimate (horizontal line) for 500 simulations, based on the constrained (positive) MLEs of transition intensities (vertical line) for males and females aged 65–74 years in the 1982–84 NLTCS.

estimates using the asymptotic method and information matrix method, described in the previous section. For the first and simplest model (illustrated in Figure 5.27) I calculate the variance of the transition intensity algebraically using both of the methods described in the previous section and compare them by looking at the ratio of the variance estimate using the asymptotic method to that using the information matrix. For the next two models (illustrated in Figures 5.30 and 5.33), the algebra becomes unwieldly, but, for constant transition intensities it is possible, with the help of Maple software, to look at the ratios of the variance estimates with variable numbers of people in each starting state. For these two models I also look at 1,000 simulated samples to see how the variance estimates behave, and compare the average ratio of the variance estimates from these simulated samples with the theoretical ratios.



Figure 5.27: A simple 2-state model with one transition intensity.

The first model is illustrated in Figure 5.27. It is a simple model of mortality, with no competing risks. For brevity I will write the one year probability of staying in state 1 (or one year probability of survival) as  $P_{01}^{11}$ , the one year probability of moving to state 2 (or annual probability of dying) as  $P_{01}^{12}$  and the constant transition intensity from state 1 to state 2 (or force of mortality) as  $\mu^{12}$ . It is easily shown that:

$$P_{01}^{11} = e^{-\mu^{12}}$$
$$P_{01}^{12} = 1 - e^{-\mu^{12}}$$

The log-likelihood, l, can then be written as:

$$l \propto n_{01}^{11} \log \left( \mathbf{P}_{01}^{11} \right) + n_{01}^{12} \log \left( \mathbf{P}_{01}^{12} \right)$$
$$= -n_{01}^{11} \mu^{12} + n_{01}^{12} \log \left( 1 - e^{-\mu^{12}} \right)$$

where  $n_{rr+t}^{ij}$  is defined in the previous section. By setting the first partial derivative:

$$\frac{\partial l}{\partial \mu^{12}} = -n_{01}^{11} + n_{01}^{12} \frac{\mathrm{e}^{-\mu^{12}}}{1 - \mathrm{e}^{-\mu^{12}}}$$
(5.44)

to zero, we obtain:

$$\hat{\mu}^{12} = \log\left(\frac{n_{01}^{11} + n_{01}^{12}}{n_{01}^{11}}\right) \tag{5.45}$$

and from the second partial derivative:

$$\frac{\partial^2 \mu^{12}}{\partial (\mu^{12})^2} = -n_{01}^{12} \frac{\mathrm{e}^{-\mu^{12}}}{\left(1 - \mathrm{e}^{-\mu^{12}}\right)^2} \tag{5.46}$$

we get that:

$$\operatorname{Var}_{I}\left(\hat{\mu}^{1\,2}\right) \geq \frac{\left(1 - \mathrm{e}^{-\mu^{1\,2}}\right)^{2}}{n_{0\,1}^{1\,2}\,\mathrm{e}^{-\mu^{1\,2}}} \tag{5.47}$$

which can be approximated by substituting  $\hat{\mu}_{01}^{12}$  (from equation (5.45)) for  $\mu^{12}$ , which gives:

$$\operatorname{Var}_{I}\left(\hat{\mu}^{1\,2}\right) \geq \frac{n_{0\,1}^{1\,2}}{n_{0\,1}^{1\,1}\left(\mu^{1\,1}+\mu^{1\,2}\right)} \tag{5.48}$$

If we now look at the asymptotic method, we are given that  $\frac{\hat{\mu}^{12}}{\mathbf{E}[W_{01}^1]}$ , is an asymptotically unbiased estimate of the variance of  $\hat{\mu}_r^{12}$ . Using the census method to calculate the central exposed to risk, we have:

$$\mathbf{E}\left[W_{01}^{1}\right] \approx n_{01}^{11} + \frac{n_{01}^{12}}{2} \tag{5.49}$$

and combining this with equation (5.45) we have:

$$\operatorname{Var}_{A}\left(\hat{\mu}^{1\,2}\right) \approx \frac{\log\left(\frac{n_{0\,1}^{1\,1} + n_{0\,1}^{0\,1}}{n_{0\,1}^{1\,1}}\right)}{n_{0\,1}^{1\,1} + \frac{n_{0\,1}^{1\,2}}{2}} \tag{5.50}$$

The variance ratio,  $R_{\hat{\mu}^{ij}}$ , which I define as  $\operatorname{Var}_A(\hat{\mu}^{ij}) / \operatorname{Var}_I(\hat{\mu}^{12})$  can then be written as:

$$R_{\hat{\mu}^{1\,2}}(k) = \frac{1+k}{k(1+\frac{k}{2})}\log\left(1+k\right) \tag{5.51}$$

where  $k = n_{01}^{12}/n_{01}^{11}$ . It is worth noting here that this variance ratio only depends on the ratio  $n_{01}^{12}/n_{01}^{11}$  and not on the number of lives under observation,  $n_{01}^{12} + n_{01}^{11}$ and is thus independent of sample size. Figure 5.28 shows  $R_{\hat{\mu}^{12}}(k)$  for  $0 \le k \le 1$ — it is always less than unity and decreases with increasing k (which is equivalent to increasing  $n_{01}^{12}$ ). For small k it is very close to unity, suggesting that for small



Figure 5.28: Variance ratio,  $R_{\hat{\mu}^{12}}(k)$ , for  $0 \leq k \leq 1$  for the 2-state model with one transition intensity.

 $\mu^{12}$  (and thus small k), the asymptotic method and the information matrix method provide very similar variance estimates in this 2-state model. This result is not surprising considering the following points:

- 1. the only extra information that is assumed in using the asymptotic method is the precise timing of any transitions, since there is no uncertainty as to how many transitions took place (once a life moves to state 2 they stay there); and
- 2. the fewer transitions that occur (the smaller  $\mu^{12}$  is), the less impact the precise timings of the transitions have on the estimation of the expected waiting times, and thus on the variance estimates as observed by Sverdrup (1965).

It is also possible to look at the effect of using the census approximation to estimate the expected waiting time — by using expectations we have that:

$$\mathbb{E} \left[ W_{01}^{1} \right] = \left( n_{01}^{11} + n_{01}^{12} \right) \int_{0}^{1} e^{-\mu^{12}t} dt$$
$$= \left( n_{01}^{11} + n_{01}^{12} \right) \frac{1 - e^{-\mu^{12}}}{\mu^{12}}$$

which gives a variance estimate of:

$$\operatorname{Var}_{A'}\left(\hat{\mu}^{1\,2}\right) \approx \frac{\left(\mu^{1\,2}\right)^2}{\left(n_{0\,1}^{1\,1} + n_{0\,1}^{1\,2}\right)\left(1 - \mathrm{e}^{-\mu^{1\,2}}\right)} \tag{5.52}$$

and a variance ratio,  $R'_{\hat{\mu}^{12}}$ , of:

$$R_{\hat{\mu}^{12}}^{\prime}\left(\mu^{12}\right) = n_{01}^{12} \left[\frac{\left(\mu^{12}\right)^2 e^{-\mu^{12}}}{\left(n_{01}^{11} + n_{01}^{12}\right)\left(1 - e^{-\mu^{12}}\right)^3}\right]$$
(5.53)

As all the terms in the square brackets in equation (5.53) are constant and by noting that:

we have that:

$$\mathbf{E}\left[R_{\hat{\mu}^{1\,2}}^{\prime}\left(\mu^{1\,2}\right)\right] = \left(\frac{\mu^{1\,2}}{\mathrm{e}^{\frac{\mu^{1\,2}}{2}} - \mathrm{e}^{-\frac{\mu^{1\,2}}{2}}}\right)^{2} \tag{5.54}$$

This expected variance ratio is also independent of sample size. It only depends on the transition intensity  $\mu^{12}$  and not on the number of lives under observation. Figure 5.29 shows  $E[R'_{\hat{\mu}^{12}}(\mu^{12})]$  for  $0 \le \mu^{12} \le 1$  — it is always less than unity and decreases with increasing  $\mu^{12}$  (which is equivalent to increasing  $n_{01}^{12}$ ). Figures 5.28 and 5.29 are comparable if we note that:

$$k = \frac{n_{01}^{12}}{n_{01}^{11}}$$
  

$$\approx \frac{(n_{01}^{11} + n_{01}^{12}) P_{01}^{12}}{(n_{01}^{11} + n_{01}^{12}) P_{01}^{11}}$$
  

$$= e^{\mu^{12}} - 1$$
  

$$= \mu^{12} + O((\mu^{12})^2)$$

which shows that the asymptotic method using the census approximation to the expected waiting time slightly underestimates the variance. This is because the census approximation slightly overestimates the expected waiting time, and so underestimates the variance, in the case of this simple two state model.

I now look at the effects on these variance estimates when there are two transition intensities in the two state model. The second model is illustrated in Figure 5.30. I



Figure 5.29: Expected variance ratio,  $E[R'_{\hat{\mu}^{12}}(\mu^{12})]$ , for  $0 \leq \mu^{12} \leq 1$  for the 2-state model with one transition intensity.

State 1	$\mu^{21}$	State 2
	$\mu^{12}$	

Figure 5.30: A simple 2-state model with two transition intensities.

will use the same notation as for the previous model. It is easily shown that:

$$P_{01}^{11} = \frac{1}{\mu^{12} + \mu^{21}} \left\{ \mu^{21} + \mu^{12} e^{-(\mu^{12} + \mu^{21})} \right\}$$

$$P_{01}^{12} = \frac{\mu^{12}}{\mu^{12} + \mu^{21}} \left\{ 1 - e^{-(\mu^{12} + \mu^{21})} \right\}$$

$$P_{01}^{22} = \frac{1}{\mu^{12} + \mu^{21}} \left\{ \mu^{12} + \mu^{21} e^{-(\mu^{12} + \mu^{21})} \right\}$$

$$P_{01}^{21} = \frac{\mu^{21}}{\mu^{12} + \mu^{21}} \left\{ 1 - e^{-(\mu^{12} + \mu^{21})} \right\}$$

and the asymptotic variance estimates,  $V_A(\mu^{12})$  and  $V_A(\mu^{21})$ , using the census

method to approximate the expected waiting times, are:

$$V_A(\mu^{12}) = \frac{\mu^{12}}{n_{01}^{11} + \frac{n_{01}^{12} + n_{01}^{21}}{2}}$$
$$V_A(\mu^{21}) = \frac{\mu^{21}}{n_{01}^{22} + \frac{n_{01}^{12} + n_{01}^{21}}{2}}$$

To simplify the following equations, I now fix the transition intensities as  $\mu^{12} = 0.1$  and  $\mu^{21} = 0.05$ . Then using Maple software, the variance ratios  $R_{\hat{\mu}^{12}}(\cdot)$  and  $R_{\hat{\mu}^{21}}(\cdot)$  are:

$$\begin{aligned} R_{\hat{\mu}^{12}}(\cdot) &= 1.028 \left[ n_{01}^{11} \left( -2.137 n_{01}^{11} - 3.982 n_{01}^{22} + 27.73 n_{01}^{12} - 55.47 n_{01}^{22} \right) \\ &+ n_{01}^{22} \left( -1.871 n_{01}^{22} - 25.73 n_{01}^{12} - 51.47 n_{01}^{21} \right) \\ &+ n_{01}^{12} \left( -66.98 n_{01}^{12} + 3.215 \times 10^5 n_{01}^{21} \right) - 267.91 \left( n_{01}^{21} \right)^2 \right] \\ &\div \left[ \left( 2n_{01}^{11} + n_{01}^{02} + n_{01}^{21} \right) \\ &\times \left( 1.461 n_{01}^{11} - 1.290 n_{01}^{22} - 3.441 n_{01}^{12} + 1.652 \times 10^4 n_{01}^{21} \right) \right] \\ R_{\hat{\mu}^{21}}(\cdot) &= 2.055 \left[ n_{01}^{11} \left( 2.137 n_{01}^{11} + 3.982 n_{01}^{22} - 27.73 n_{01}^{12} + 55.47 n_{01}^{22} \right) \\ &+ n_{01}^{22} \left( 1.871 n_{01}^{22} + 25.73 n_{01}^{12} - 51.47 n_{01}^{21} \right) \\ &+ n_{01}^{12} \left( 66.98 n_{01}^{12} - 3.215 \times 10^5 n_{01}^{21} \right) + 267.91 \left( n_{01}^{21} \right)^2 \right] \\ &\div \left[ \left( 2n_{01}^{11} + n_{01}^{12} + n_{01}^{21} \right) \\ &\times \left( 2.814 n_{01}^{11} - 2.677 n_{01}^{22} - 1.652 \times 10^4 n_{01}^{12} + 13.77 n_{01}^{21} \right) \right] \end{aligned}$$

By substituting the mean values for  $n_{01}^{ij}$  into the above equations they can be simplified further. Let  $n_{01}^{i} = n_{01}^{11} + n_{01}^{12}$  and  $n_{01}^{2i} = n_{01}^{22} + n_{01}^{21}$ , then  $\mathbf{E}[n_{01}^{ij}] = n_{01}^{i} \times \mathbf{P}_{01}^{ij}$  and:

$$\overline{R}_{\hat{\mu}^{12}}(K) = \frac{7.081 \times 10^5 K}{(6.705 + 5.110 \times 10^4 K) (143.0 + 3.482K)}$$
  
$$\overline{R}_{\hat{\mu}^{21}}(K) = \frac{5.311 \times 10^5 K}{(7.659 \times 10^3 + 9.568K) (3.482 + 73.26K)}$$

where  $K = n_{01}^{2}/n_{01}^{1}$ . These are again independent of sample size, and only depend on the ratio of the numbers of lives initially under observation in each starting state. These variance ratios are illustrated in Figure 5.31. It is noticeable that they are always less than unity and that with increasing K (increasing proportion of lives starting in state 2) the variance ratio,  $\overline{R}_{\hat{\mu}^{12}}(K)$  tends downwards away from unity,



Figure 5.31: Variance ratios,  $R_{\hat{\mu}^{ij}}(K)$   $(0 \le K \le 10)$  for  $\mu^{12}$  and  $\mu^{21}$  when  $n_{01}^{ij} = E[n_{01}^{ij}]$  in the 2-state model with two transition intensities.

while  $\overline{R}_{\hat{\mu}^{21}}(K)$  tends towards unity. For the specific case that  $K = \frac{1}{2}$  (as I chose for the following simulations) the values of the variance ratios are:

$$\overline{R}_{\hat{\mu}^{12}}(0.5) = 0.9548 \tag{5.55}$$

$$\overline{R}_{\hat{\mu}^{21}}(0.5) = 0.8638 \tag{5.56}$$

The results from 1,000 simulations (using the same method as described in the previous section), with  $\mu^{12} = 0.1$ ,  $\mu^{21} = 0.05$ ,  $n_{01}^{1.} = 2,000$  and  $n_{01}^{2.} = 1,000$  (K = 0.5) are illustrated in Figure 5.32. The 'bootstrap' variance estimates in these graphs may not be accurate as they are only based on 1,000 simulations — I have included them for illustration only. The mean value of the estimated transition intensities for these 1,000 simulations are  $\overline{\mu}^{12} = 0.1006$  and  $\overline{\mu}^{21} = 0.0496$ . From the graphs in Figure 5.32 the difference between the two variance estimates seems fairly constant, so it is then not surprising that the mean of the variance ratios in the simulations for  $\mu^{12}$  and  $\mu^{21}$ , which are 0.9550 and 0.8632, respectively — are in very close agreement with the variance ratios calcuated in equations 5.55 and 5.56,



Figure 5.32: Variance estimates of the transition intensities for the 2-state model with two transition intensities using the asymptotic method (•), the information matrix (×) and the 'boot-strap' variance estimate (horizontal line) for 1,000 simulations, with  $\mu^{12} = 0.1$ ,  $\mu^{21} = 0.05$ ,  $N^1 = 2,000$  and  $N^2 = 1,000$  (K = 0.5).

even though they are slightly different measures.

It is very noticeable that the variance ratio has become considerably lower with the addition of just one more transition intensity in the two state model. In the first model with  $\mu^{12} = 0.1$  the variance ratio is 0.9983, whereas for the second model, depending on the transition intensity and ratio of lives in each starting state, it is between 0.86 and 0.95 — quite a large reduction.

This is not surprising, considering the information that is lost with only the knowledge of the states that each life starts and ends in (as assumed in the information matrix method) rather than the life histories (as assumed in the asymptotic method) — not only is the timing of each transition not known (as in the 2-state model with one transition intensity), but in this model the number of transitions between states is also unknown (since knowing the state a life starts and ends in does not tell you how many times they have travelled between states). So the difference in information used by the asymptotic method and the information matrix method

is greater than in the previous model, which is clearly shown by lower values of the variance ratio.



Figure 5.33: A simple 3-state model with three transition intensities.

The final model that I investigate, a 3-state model with three transition intensities, is illustrated in Figure 5.33. I use the same methodology and notation in this model as the previous one. The transition probabilities can be written as:

$$\begin{aligned} \mathbf{P}_{01}^{11} &= \mathbf{e}^{-\left(\mu^{12}+\mu^{13}\right)} \\ \mathbf{P}_{01}^{12} &= \frac{\mu^{12}}{\mu^{12}+\mu^{13}-\mu^{23}} \left[ \mathbf{e}^{-\mu^{23}} - \mathbf{e}^{-\left(\mu^{12}+\mu^{13}\right)} \right] \\ \mathbf{P}_{01}^{13} &= 1 - \frac{1}{\mu^{12}+\mu^{13}-\mu^{23}} \left[ \mu^{12} \mathbf{e}^{-\mu^{23}} + \left(\mu^{13}-\mu^{23}\right) \mathbf{e}^{-\left(\mu^{12}+\mu^{13}\right)} \right] \\ \mathbf{P}_{01}^{22} &= \mathbf{e}^{-\mu^{23}} \\ \mathbf{P}_{01}^{23} &= 1 - \mathbf{e}^{-\mu^{23}} \end{aligned}$$

and the asymptotic variance estimates,  $V_A(\mu^{12})$ ,  $V_A(\mu^{13})$  and  $V_A(\mu^{23})$ , using the census method to approximate the expected waiting times are:

$$V_A(\mu^{12}) = \frac{\mu^{12}}{n_{01}^{11} + \frac{n_{01}^{12} + n_{01}^{13}}{2}}$$
$$V_A(\mu^{13}) = \frac{\mu^{13}}{n_{01}^{11} + \frac{n_{01}^{12} + n_{01}^{13}}{2}}$$
$$V_A(\mu^{23}) = \frac{\mu^{23}}{n_{01}^{22} + \frac{n_{01}^{12} + n_{01}^{23}}{2}}$$

To simplify the following equations, I now fix the transition intensities as  $\mu^{12} = 0.1$ ,  $\mu^{13} = 0.025$  and  $\mu^{23} = 0.05$ . Then using Maple software, the variance ratios

 $R_{\hat{\mu}^{12}}(\cdot), R_{\hat{\mu}^{13}}(\cdot)$  and  $R_{\hat{\mu}^{23}}(\cdot)$  are:

$$\begin{split} R_{\hat{\mu}^{12}}(\cdot) &= 2.000 \left[ \left( n_{01}^{12} \right)^2 \left( 20.89 n_{01}^{12} - 1.214 \times 10^{10} n_{01}^{12} - 3.332 \times 10^9 n_{01}^{23} \right) \\ &+ \left( n_{01}^{13} \right)^2 \left( 1.669 \times 10^{11} n_{01}^{12} - 4.391 \times 10^{12} n_{01}^{13} - 1.939 \times 10^{12} n_{01}^{23} \right) \\ &+ 5.284 \times 10^{13} n_{01}^{12} n_{01}^{23} n_{01}^{23} \right] \div \left[ \left( 2n_{01}^{11} + n_{01}^{12} + n_{01}^{13} \right) \\ &\times \left( 2.089 \left( n_{01}^{12} \right)^2 - 1.214 \times 10^9 n_{01}^{12} n_{01}^{13} - 3.332 \times 10^8 n_{01}^{12} n_{01}^{23} \right) \right] \\ &+ 1.629 \times 10^{10} \left( n_{011}^{13} \right)^2 + 5.288 \times 10^{12} n_{01}^{13} n_{01}^{23} \right) \right] \\ R_{\hat{\mu}^{13}}(\cdot) &= 5 \times 10^{-7} \left[ \left( n_{01}^{12} \right)^2 \left( 20.89 n_{01}^{12} - 1.214 \times 10^{10} n_{01}^{12} - 3.332 \times 10^9 n_{01}^{23} \right) \\ &+ \left( n_{011}^{03} \right)^2 \left( 1.669 \times 10^{11} n_{01}^{12} - 4.391 \times 10^{12} n_{01}^{13} - 1.939 \times 10^{12} n_{01}^{23} \right) \\ &+ 5.284 \times 10^{13} n_{01}^{12} n_{01}^{23} n_{01}^{23} \right] \div \left[ \left( 2n_{01}^{11} + n_{01}^{12} + n_{01}^{13} \right) \\ &\times \left( -8.331 \left( n_{01}^{12} \right)^2 + 4.446 \times 10^2 n_{01}^{12} n_{01}^{13} + 3.996 \times 10^4 n_{01}^{12} n_{01}^{23} \right) \right] \\ R_{\hat{\mu}^{23}}(\cdot) &= 5 \times 10^{-7} \left[ \left( n_{01}^{12} \right)^2 \left( 20.89 n_{01}^{12} - 1.214 \times 10^{10} n_{01}^{12} - 3.332 \times 10^9 n_{01}^{23} \right) \\ &+ \left( n_{01}^{13} \right)^2 \left( 1.669 \times 10^{11} n_{01}^{12} - 4.391 \times 10^{12} n_{01}^{13} \right) \\ &+ \left( n_{01}^{12} n_{01}^{23} \right)^2 + 4.446 \times 10^2 n_{01}^{13} n_{01}^{23} \right) \right] \\ R_{\hat{\mu}^{23}}(\cdot) &= 5 \times 10^{-7} \left[ \left( n_{01}^{12} \right)^2 \left( 20.89 n_{01}^{12} - 1.214 \times 10^{10} n_{01}^{12} - 3.332 \times 10^9 n_{01}^{23} \right) \\ &+ \left( n_{01}^{13} \right)^2 \left( 1.669 \times 10^{11} n_{01}^{22} - 4.391 \times 10^{12} n_{01}^{13} - 1.939 \times 10^{12} n_{01}^{23} \right) \\ &+ \left( n_{01}^{13} \right)^2 \left( 1.669 \times 10^{11} n_{01}^{22} - 4.391 \times 10^{12} n_{01}^{13} - 1.939 \times 10^{12} n_{01}^{23} \right) \\ &+ 5.284 \times 10^{13} n_{01}^{12} n_{01}^{23} n_{01}^{23} \right] \div \left[ \left( 2n_{01}^{11} + n_{01}^{12} + n_{01}^{13} \right) \\ &\times \left( -4.165 \left( n_{01}^{12} \right)^2 + 6.607 \times 10^4 n_{01}^{12} n_{01}^{13} + 2.424 \times 10^2 \left( n_{01}^{13} \right)^2 \right) \right] \end{aligned}$$

By substituting the mean values for  $n_{01}^{ij}$  into the above equations they can be simplified further. Let  $n_{01}^{1.} = \sum_{j} n_{01}^{1j}$  and  $n_{01}^{2.} = \sum_{j} n_{01}^{2j}$ , then  $E[n_{01}^{ij}] = n_{01}^{i.} \times P_{01}^{ij}$  and:

$$\overline{R}_{\hat{\mu}^{12}}(K) = \frac{6480.57K - 1.912 \times 10^{-7}}{6668.23K + 8.015}$$
  

$$\overline{R}_{\hat{\mu}^{13}}(K) = \frac{810.07K - 2.390 \times 10^{-8}}{894.18K + 4.008}$$
  

$$\overline{R}_{\hat{\mu}^{43}}(K) = \frac{195.04K - 5.755 \times 10^{-9}}{195.12K + 9.164}$$

where  $K = n_{01}^{2}/n_{01}^{1}$ . These are again independent of sample size, and only depend on the ratio of the numbers of lives initially under observation in each starting state. These variance ratios are illustrated in Figure 5.34. It is noticeable that they are always less than unity and that with increasing K (increasing proportion of lives starting in state 2) all the variance ratios increase, but with an upper limit of less than unity. For the specific case that  $K = \frac{1}{2}$  (as I chose for the following simulations)



Figure 5.34: Variance ratios,  $R_{\hat{\mu}^{ij}}(K)$   $(0 \leq K \leq 10)$  for  $\mu^{12}$ ,  $\mu^{13}$  and  $\mu^{23}$  when  $n_{01}^{ij} = \mathbb{E}[n_{01}^{ij}]$  in the 3-state model.

the values of the variance ratios are:

$$\overline{R}_{\hat{\mu}^{12}}(0.5) = 0.9695 \tag{5.57}$$

$$\overline{R}_{\hat{\mu}^{13}}(0.5) = 0.8979 \tag{5.58}$$

$$\overline{R}_{\hat{\mu}^{23}}(0.5) = 0.9138 \tag{5.59}$$

The results from 1,000 simulations (using the same method as described in the previous section), with  $\mu^{12} = 0.1$ ,  $\mu^{13} = 0.025$ ,  $\mu^{23} = 0.05$ ,  $N^1 = 5,000$  and  $N^2 = 2,500$  (K = 0.5) are illustrated in Figure 5.35. The 'bootstrap' variance estimates in these graphs may not be accurate as they are only based on 1,000 simulations — I have included them for illustration only. The mean value of the transition intensities for these 1,000 simulations are  $\overline{\mu}^{12} = 0.1003$ ,  $\overline{\mu}^{13} = 0.0249$  and  $\overline{\mu}^{23} = 0.0499$ . From the graphs in Figure 5.32 the difference between the two variance estimates seems fairly constant, so it is then not surprising that the mean of the variance ratios in the simulations for  $\mu^{12}$ ,  $\mu^{13}$  and  $\mu^{23}$ , which are 0.9696, 0.8968 and 0.9136, respectively — are in very close agreement with the variance ratios given



Figure 5.35: Variance estimates of the transition intensities for the 3-state model using the asymptotic method (•), the information matrix (×) and the 'boot-strap' variance estimate (horizontal line) for 1,000 simulations, with  $\mu^{12} = 0.1$ ,  $\mu^{13} = 0.025$ ,  $\mu^{23} = 0.05$ ,  $N^1 = 5,000$  and  $N^2 = 2,500$  (K = 0.5).

in equations 5.57 to 5.59, even though they are slightly different measures.

The variance ratios in this 3-state model (0.90 to 0.97) are comparable to the variance ratios in the 2-state model with two transition intensities (0.86 to 0.95). While they are considerably less than unity, they have not become yet smaller with the introduction of an extra state and transition intensity. This may be because, while the model has become more complex, the number of transitions a life can make is limited to a maximum of two and in some cases, from the knowledge of the state in which a life started and finished, the exact number of transitions that a life has made is known (i.e. if a life starts in state 1 and ends in state 2, then that life has only made one transition in that time).

Care is required in interpreting the results presented in this section, as the variance ratios depend not only on the model, the magnitude of the transition intensities and the numbers of lives starting in each state, but also on the numbers of transitions between states. However, these results have shown that variance estimates of transition intensities can be substantially increased, even in simple models, if the only data available are the states in which a life is at the start and end of a survey period, rather than their complete life histories.

In summary, in this section I have compared, in three simple models, two methods of calculating variance estimates of transition intensities, one which uses complete life history data (i.e. the knowledge of the states each life moved to and the timing of these transitions), the asymptotic method, and one which uses only partial data (i.e. the knowledge of which state a life starts and ends in), the information matrix method. For the simplest model, a 2-state model with one transition intensity, the difference between the variance estimates of the two methods is negligible, especially when the underlying transition intensity is small in magnitude. The reason for this may be because the only difference in the information used by the two methods is small — the timing of any transition. However, for the slightly more complex models, the difference between variance estimates increases rapidly, with variance estimates from the asymptotic method up to 14% smaller than those using the information matrix method. This is representative of the information lost in only knowing the starting and ending states that a life is in, rather than their complete life history (i.e. the timing, and number of transitions between states). Given the complexity of the disability model (7 states with 36 transition intensities), it seems unreasonable to use the asymptotic variance estimator, given that the NLTCS only provides partial information about the lives in the survey. The figures in Table 5.59 indicate that, in the disability model, if complete life histories had been available, and thus the asymptotic method would have been valid then the variance estimates would be reduced by up to 98%, compared with those using the information matrix with partial data.

Applied to data collection, this is strong motivation, for any model more complex than the basic mortality model, to collect complete life history data and not just data on the states in which lives are at fixed points in time. The cost of using partial data is an increase in the variance estimates of the transition intensities — an increase which, even for simple models, can be substantial.

In the next section I look at methods for estimating the variance of the constrained (positive) MLEs of the transition intensities in the disability model.

## 5.4 Calculation of Variance Estimates for the Transition Intensities in the Disability Model

The aim of this section is to estimate the variance of the constrained (positive) MLEs of the transition intensities in the disability model. The maximum likelihood method used in the previous section cannot be used to calculate variance estimates of the constrained (positive) MLEs of the transition intensities, since, for these estimates, the set of first derivatives of the log-likelihood function will be non-zero. (The unconstrained MLEs maximise the log-likelihood, whereas the constrained (positive) MLEs maximise the log-likelihood in the non-negative region of the parameter space). I first look at calculating 'boot-strap' variance estimates (explained below), however, the amount of computer time required to calculate these estimates is prohibitive. Using the same methodology as the 'boot-strap' method, it is possible to adjust the maximum likelihood technique to calculate equivalent variance estimates in a much more efficient manner. The variance estimates from the 'boot-strap' method are then used to check the reasonableness of those calculated using the adjusted maximum likelihood method — and it is illustrated that the former do converge (given sufficient simulations) to the latter.

The first method is to calculate 'boot-strap' variance estimates. That is, run a large number of simulations using the method described in Section 5.2, and from these, estimate the corresponding transition intensities. The variance of these simulated transition intensities are estimates of the variance of the original transition intensities. However, it is necessary to find how many simulations are required for these 'boot-strap' variance estimates to converge.

I initially ran 10,000 simulations based on the constrained (positive) MLEs of the transition intensities for the data set of males and females aged 65–74 in the 1982–84 NLTCS. I then found the maximum and minimum variance estimate of 100



Figure 5.36: Convergence of the 'boot-strap' variance estimates for the constrained (positive) MLEs of the transition intensities out of the 'Healthy' state, showing the maximum (•) and minimum (×) 'boot-strap' variance estimate of 100 samples, for samples of size n (n = 50, 100, 500, 1,000, 2,000 and 5,000) and where the horizontal line is the variance estimate using an adjusted maximum likelihood approach, for males and females aged 65–74 years in the 1982–84 NLTCS.

samples, each based on n (n = 50, 100, 500, 1,000, 2,000 and 5,000) simulations from the initial 10,000 simulations (the n simulations from 10,000 were chosen at random without replacement). The reason for selecting from only 10,000 simulations is lack of computer time — each simulation took about 30 seconds to run, with an additional 10 seconds to calculate the covariance matrix using the information matrix. The convergence of the variance estimates is illustrated in Figures 5.36 to 5.41 — even using 5,000 simulations there is still some uncertainty. I will now discuss how the information matrix can be used to calculate the variance estimates much more efficiently (in terms of computer time).



Figure 5.37: Convergence of the 'boot-strap' variance estimates for the constrained (positive) MLEs of the transition intensities out of the 'IADL only' state, showing the maximum (•) and minimum (×) 'boot-strap' variance estimate of 100 samples, for samples of size n (n = 50, 100, 500, 1,000, 2,000 and 5,000) and where the horizontal line is the variance estimate using an adjusted maximum likelihood approach, for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.38: Convergence of the 'boot-strap' variance estimates for the constrained (positive) MLEs of the transition intensities out of the '1–2 ADLs' state, showing the maximum (•) and minimum (×) 'boot-strap' variance estimate of 100 samples, for samples of size n (n = 50, 100, 500, 1,000, 2,000 and 5,000) and where the horizontal line is the variance estimate using an adjusted maximum likelihood approach, for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.39: Convergence of the 'boot-strap' variance estimates for the constrained (positive) MLEs of the transition intensities out of the '3–4 ADLs' state, showing the maximum (•) and minimum (×) 'boot-strap' variance estimate of 100 samples, for samples of size n (n = 50, 100, 500, 1,000, 2,000 and 5,000) and where the horizontal line is the variance estimate using an adjusted maximum likelihood approach, for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.40: Convergence of the 'boot-strap' variance estimates for the constrained (positive) MLEs of the transition intensities out of the '5–6 ADLs' state, showing the maximum (•) and minimum (×) 'boot-strap' variance estimate of 100 samples, for samples of size n (n = 50, 100, 500, 1,000, 2,000 and 5,000) and where the horizontal line is the variance estimate using an adjusted maximum likelihood approach, for males and females aged 65–74 years in the 1982–84 NLTCS.



Figure 5.41: Convergence of the 'boot-strap' variance estimates for the constrained (positive) MLEs of the transition intensities out of the 'Institutionalized' state, showing the maximum (•) and minimum (×) 'boot-strap' variance estimate of 100 samples, for samples of size n (n = 50, 100, 500, 1,000, 2,000 and 5,000) and where the horizontal line is the variance estimate using an adjusted maximum likelihood approach, for males and females aged 65–74 years in the 1982–84 NLTCS.

It is not possible to use the information matrix method as it stands as the constrained (positive) MLEs will result in non-zero first derivatives of the log-likelihood function. However, this method can be adapted, to calculate variance estimates consistent with these 'boot-strap' estimates. The 'boot-strap' method assumes that the estimated transition intensities (constrained (positive) MLEs in this case) are the true population transition intensities. From these the transition probabilities are calculated, which are then used as the probabilities in the multinomial distribution for the purposes of sampling (see Section 5.2). The resulting variance estimates are those of the constrained (positive) MLEs of the transition intensities, under the assumption that the constrained (positive) MLEs of the transition intensities are the true population transition intensities. Under the same assumption, variance estimates using the information matrix can be calculated as follows:

- 1. as before, from the constrained (positive) MLEs of the transition intensities, calculate the corresponding transition probabilities;
- 2. use these transition probabilities to calculate the expected number of lives that move between states, by multiplying the number of lives starting in a given state from the data, by the corresponding transition probability; then
- 3. as the constrained (positive) MLEs of the transition intensities maximise the likelihood function using these numbers of transitions, the information matrix method can be used to estimate the variance (as the first derivatives of the log-likelihood will now be zero).

Or more specifically, let  $n_{rr+t}^{ij}$  be the actual number of lives in state *i* and age group *r* at the start of the survey period who are in state *j* after *t* years and let  $n_r^{i\cdot} = \sum_{j=1}^7 n_{rr+t}^{ij}$  be the number of lives in age group *r* starting in state *i*. Then the expected number of lives, consistent with the transition probabilities calculated from the constrained (positive) MLEs of the transition intensities  $(P_{rr+t}^{ij}(\overline{\mu}_r^{1\,2},\ldots,\overline{\mu}_r^{n\,n-1}))$ that moved from state *i* to state *j* in age group *r* over *t* years,  $\overline{n}_{rr+t}^{ij}$ , can be calculated as:

$$\overline{n}_{r\,r+t}^{i\,j} = n_r^{i\,\cdot} \mathbf{P}_{r\,r+t}^{i\,j} \left( \overline{\mu}_r^{1\,2}, \dots, \overline{\mu}_r^{n\,n-1} \right) \tag{5.60}$$

Table 5.61: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males using 5 year age groupings.

1982					1984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		$1.06 \times 10^{-5}$	$9.01 \times 10^{-6}$	$7.32 \times 10^{-6}$	$4.09 \times 10^{-6}$	$1.34 \times 10^{-6}$	$7.83 \times 10^{-6}$
Ū	70 - 74		$1.96 \times 10^{-5}$	$1.94 \times 10^{-5}$	$8.29 \times 10^{-6}$	$7.37 \times 10^{-6}$	$5.24 \times 10^{-6}$	$2.38 \times 10^{-5}$
	75 - 79		$6.00 \times 10^{-5}$	$5.02 \times 10^{-5}$	$1.65 \times 10^{-5}$	$1.90 \times 10^{-5}$	$1.14 \times 10^{-5}$	$4.82 \times 10^{-5}$
	80 - 84		$2.62{\times}10^{-4}$	$2.40 \times 10^{-4}$	$1.53{ imes}10^{-4}$	$1.46 { imes} 10^{-4}$	$6.72 \times 10^{-5}$	$1.63 { imes} 10^{-4}$
IADL only	65 - 69	$1.75{ imes}10^{-3}$		$3.37{ imes}10^{-3}$	$2.12 \times 10^{-3}$	$1.74 \times 10^{-3}$	$1.32{\times}10^{-4}$	$8.23 \times 10^{-4}$
	70 - 74	$2.09 \times 10^{-3}$		$2.58{ imes}10^{-3}$	$9.74 \times 10^{-4}$	$4.99 { imes} 10^{-4}$	$4.42 \times 10^{-4}$	$1.06 \times 10^{-3}$
	75 - 79	$1.21 \times 10^{-3}$		$3.41 \times 10^{-3}$	$1.77 \times 10^{-3}$	$2.04 \times 10^{-3}$	$7.62 \times 10^{-4}$	$1.32 \times 10^{-3}$
	80 - 84	$1.82 \times 10^{-3}$		$1.01 \times 10^{-2}$	$1.11 \times 10^{-2}$	$1.14 \times 10^{-2}$	$2.27 \times 10^{-3}$	$3.48 \times 10^{-3}$
1-2 ADLs	65 - 69	$1.33 \times 10^{-3}$	$2.58 \times 10^{-3}$		$1.06 \times 10^{-2}$	$5.18 \times 10^{-3}$	$4.26 \times 10^{-4}$	$1.74 \times 10^{-3}$
	70 - 74	$9.50 \times 10^{-4}$	$2.68 \times 10^{-3}$		$6.23 \times 10^{-3}$	$2.28 \times 10^{-3}$	$1.09 \times 10^{-3}$	$1.57 \times 10^{-3}$
	75 - 79	$8.76 \times 10^{-4}$	$4.70 \times 10^{-3}$		$3.30 \times 10^{-3}$	$3.34 \times 10^{-3}$	$5.70 \times 10^{-4}$	$2.83 \times 10^{-3}$
	80 - 84	$5.92 \times 10^{-4}$	$2.83 \times 10^{-3}$		$1.62 \times 10^{-2}$	$1.02 \times 10^{-2}$	$1.44 \times 10^{-3}$	$3.80 \times 10^{-3}$
3-4 ADLs	65 - 69	$3.73 \times 10^{-3}$	$6.05 \times 10^{-3}$	$4.07 \times 10^{-2}$		$3.09 \times 10^{-2}$	$2.34 \times 10^{-3}$	$9.05 \times 10^{-3}$
	70 - 74	$2.24 \times 10^{-3}$	$2.09 \times 10^{-3}$	$1.08 \times 10^{-2}$		$1.08 \times 10^{-2}$	$1.02 \times 10^{-3}$	$3.97 \times 10^{-3}$
	75 - 79	$1.37 \times 10^{-3}$	$2.89 \times 10^{-3}$	$7.69 \times 10^{-3}$		$1.31 \times 10^{-2}$	$4.35 \times 10^{-3}$	$8.57 \times 10^{-3}$
	80 - 84	$2.41 \times 10^{-3}$	$2.76 \times 10^{-2}$	$3.76 \times 10^{-2}$		$1.61 \times 10^{-1}$	$1.44 \times 10^{-2}$	$2.84 \times 10^{-2}$
5-6 ADLs	65 - 69	$1.00 \times 10^{-3}$	$2.54 \times 10^{-3}$	$7.41 \times 10^{-3}$	$1.70 \times 10^{-2}$		$1.22 \times 10^{-3}$	$4.42 \times 10^{-3}$
	70 - 74	$9.30 \times 10^{-4}$	$1.38 \times 10^{-3}$	$4.94 \times 10^{-3}$	$4.25 \times 10^{-3}$		$1.11 \times 10^{-3}$	$2.85 \times 10^{-3}$
	75 - 79	$5.48 \times 10^{-4}$	$1.84 \times 10^{-3}$	$1.64 \times 10^{-3}$	$4.43 \times 10^{-3}$		$1.98 \times 10^{-3}$	$4.65 \times 10^{-3}$
	80 - 84	$8.44 \times 10^{-4}$	$6.77 \times 10^{-3}$	$7.47 \times 10^{-3}$	$4.06 \times 10^{-2}$		$3.89 \times 10^{-3}$	$9.89 \times 10^{-3}$
Inst'd	65 - 69	$4.66 \times 10^{-4}$	$3.07 \times 10^{-4}$	$2.30 \times 10^{-4}$	$5.67 \times 10^{-4}$	$1.44 \times 10^{-4}$		$1.20 \times 10^{-3}$
	70 - 74	$4.41 \times 10^{-4}$	$2.35 \times 10^{-4}$	$1.28 \times 10^{-3}$	$5.08 \times 10^{-4}$	$9.36 \times 10^{-4}$		$4.11 \times 10^{-3}$
	75 - 79	$1.91 \times 10^{-4}$	$4.08 \times 10^{-4}$	$2.63 \times 10^{-4}$	$2.50 \times 10^{-4}$	$1.09 \times 10^{-4}$		$2.29 \times 10^{-3}$
	80 - 84	$4.05 \times 10^{-5}$	$6.68 \times 10^{-5}$	$7.70 \times 10^{-5}$	$6.10 \times 10^{-4}$	$5.24 \times 10^{-4}$		$3.45 \times 10^{-3}$

Table 5.62: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for females using 5 year age groupings.

1982					1984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		$1.07 \times 10^{-5}$	$9.20 \times 10^{-6}$	$2.87 \times 10^{-6}$	$1.60 \times 10^{-6}$	$5.33 \times 10^{-7}$	$3.28 \times 10^{-6}$
Ŭ	70 - 74		$2.46 \times 10^{-5}$	$1.65 \times 10^{-5}$	$7.40 \times 10^{-6}$	$6.59 \times 10^{-6}$	$3.32 \times 10^{-6}$	$6.94 \times 10^{-6}$
	75 - 79		$6.20 \times 10^{-5}$	$5.51 \times 10^{-5}$	$2.90 \times 10^{-5}$	$1.04 \times 10^{-5}$	$1.00 \times 10^{-5}$	$1.76 \times 10^{-5}$
	80 - 84		$1.54 \times 10^{-4}$	$2.11{ imes}10^{-4}$	$8.47 \times 10^{-5}$	$4.93 { imes} 10^{-5}$	$4.87 \times 10^{-5}$	$6.59 \times 10^{-5}$
IADL only	65 - 69	$1.42 \times 10^{-3}$		$3.28 \times 10^{-3}$	$1.54 \times 10^{-3}$	$3.36{ imes}10^{-4}$	$1.41 { imes} 10^{-4}$	$3.02 \times 10^{-4}$
	70 - 74	$9.12 \times 10^{-4}$		$3.59{ imes}10^{-3}$	$9.84 \times 10^{-4}$	$7.50 \times 10^{-4}$	$3.82{ imes}10^{-4}$	$3.60 \times 10^{-4}$
	75 - 79	$7.78 \times 10^{-4}$		$5.17 \times 10^{-3}$	$2.38 \times 10^{-3}$	$6.13 \times 10^{-4}$	$5.62 \times 10^{-4}$	$5.98 \times 10^{-4}$
	80-84	$7.30 \times 10^{-4}$		$8.72 \times 10^{-3}$	$2.91 \times 10^{-3}$	$1.68 \times 10^{-3}$	$7.37 \times 10^{-4}$	$8.35 \times 10^{-4}$
1-2 ADLs	65 - 69	$7.79 \times 10^{-4}$	$2.78 \times 10^{-3}$		$2.03 \times 10^{-3}$	$4.77 \times 10^{-4}$	$1.35 \times 10^{-4}$	$4.17 \times 10^{-4}$
	70 - 74	$5.01 \times 10^{-4}$	$2.35 \times 10^{-3}$		$2.04 \times 10^{-3}$	$1.45 \times 10^{-3}$	$4.50 \times 10^{-4}$	$4.96 \times 10^{-4}$
	75 - 79	$4.15 \times 10^{-4}$	$1.91 \times 10^{-3}$		$4.22 \times 10^{-3}$	$1.06 \times 10^{-3}$	$5.09 \times 10^{-4}$	$4.90 \times 10^{-4}$
	80 - 84	$3.89 \times 10^{-4}$	$2.66 \times 10^{-3}$		$3.96 \times 10^{-3}$	$2.30 \times 10^{-3}$	$7.49 \times 10^{-4}$	$7.85 \times 10^{-4}$
3-4 ADLs	65 - 69	$1.05 \times 10^{-3}$	$4.90 \times 10^{-3}$	$1.64 \times 10^{-2}$		$3.83 \times 10^{-3}$	$6.24 \times 10^{-4}$	$1.27 \times 10^{-3}$
	70 - 74	$8.22 \times 10^{-4}$	$3.47 \times 10^{-3}$	$1.02 \times 10^{-2}$		$1.34 \times 10^{-2}$	$1.37 \times 10^{-3}$	$1.88 \times 10^{-3}$
	75 - 79	$7.02 \times 10^{-4}$	$3.90 \times 10^{-3}$	$1.77 \times 10^{-2}$		$1.00 \times 10^{-2}$	$2.54 \times 10^{-3}$	$2.52 \times 10^{-3}$
	80 - 84	$6.45 \times 10^{-4}$	$3.41 \times 10^{-3}$	$1.23 \times 10^{-2}$		$1.95 \times 10^{-2}$	$2.87 \times 10^{-3}$	$2.80 \times 10^{-3}$
5-6 ADLs	65 - 69	$7.15 \times 10^{-4}$	$2.01 \times 10^{-3}$	$3.32 \times 10^{-3}$	$2.65 \times 10^{-3}$		$5.25 \times 10^{-4}$	$1.88 \times 10^{-3}$
	70 - 74	$5.69 \times 10^{-4}$	$2.60 \times 10^{-3}$	$3.83 \times 10^{-3}$	$6.80 \times 10^{-3}$		$1.92 \times 10^{-3}$	$3.28 \times 10^{-3}$
	75 - 79	$3.47 \times 10^{-4}$	$1.03 \times 10^{-3}$	$2.43 \times 10^{-3}$	$4.66 \times 10^{-3}$		$1.43 \times 10^{-3}$	$1.70 \times 10^{-3}$
	80 - 84	$5.83 \times 10^{-4}$	$1.69 \times 10^{-3}$	$4.03 \times 10^{-3}$	$7.40 \times 10^{-3}$	-	$1.95 \times 10^{-3}$	$2.64 \times 10^{-3}$
Inst'd	65 - 69	$3.86 \times 10^{-4}$	$9.39 \times 10^{-5}$	$3.94 \times 10^{-4}$	$4.17 \times 10^{-4}$	$5.28 \times 10^{-5}$		$4.58 \times 10^{-4}$
	70 - 74	$1.06 \times 10^{-4}$	$1.28 \times 10^{-4}$	$1.17 \times 10^{-4}$	$2.61 \times 10^{-4}$	$3.35 \times 10^{-4}$		$1.26 \times 10^{-3}$
	75 - 79	$3.72 \times 10^{-5}$	$9.38 \times 10^{-5}$	$7.16 \times 10^{-5}$	$8.22 \times 10^{-5}$	$1.41 \times 10^{-4}$		$7.11 \times 10^{-4}$
	80 - 84	$2.17 \times 10^{-5}$	$3.72 \times 10^{-5}$	$4.45 \times 10^{-5}$	$4.60 \times 10^{-5}$	$2.42 \times 10^{-5}$		$4.30 \times 10^{-4}$

Table 5.63: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1982					1984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		$5.42 \times 10^{-6}$	$4.87 \times 10^{-6}$	$2.13 \times 10^{-6}$	$1.20 \times 10^{-6}$	$4.18 \times 10^{-7}$	$2.53 \times 10^{-6}$
Ū.	70 - 74		$1.16 \times 10^{-5}$	$8.90 \times 10^{-6}$	$3.92 \times 10^{-6}$	$3.46 \times 10^{-6}$	$1.98 \times 10^{-6}$	$6.50 \times 10^{-6}$
	75 - 79		$3.16 \times 10^{-5}$	$2.61 \times 10^{-5}$	$1.25 \times 10^{-5}$	$6.65 \times 10^{-6}$	$5.44 \times 10^{-6}$	$1.39 \times 10^{-5}$
	80 - 84		$9.57{ imes}10^{-5}$	$1.20{ imes}10^{-4}$	$4.97{ imes}10^{-5}$	$3.50 \times 10^{-5}$	$2.73{ imes}10^{-5}$	$4.74 \times 10^{-5}$
IADL only	65 - 69	$7.95{ imes}10^{-4}$		$1.87{ imes}10^{-3}$	$8.38 { imes} 10^{-4}$	$3.98{\times}10^{-4}$	$7.10 \times 10^{-5}$	$2.53{ imes}10^{-4}$
	70 - 74	$6.47{ imes}10^{-4}$		$1.84 \times 10^{-3}$	$5.59 \times 10^{-4}$	$3.49 \times 10^{-4}$	$2.21 \times 10^{-4}$	$2.87 \times 10^{-4}$
	75 - 79	$4.74 \times 10^{-4}$		$2.27{ imes}10^{-3}$	$1.07 \times 10^{-3}$	$5.24 \times 10^{-4}$	$3.32 \times 10^{-4}$	$4.12 \times 10^{-4}$
	80 - 84	$5.19 \times 10^{-4}$		$5.26 \times 10^{-3}$	$2.28 \times 10^{-3}$	$1.66 \times 10^{-3}$	$5.31 \times 10^{-4}$	$7.01 \times 10^{-4}$
1-2 ADLs	65 - 69	$5.24 \times 10^{-4}$	$1.59 \times 10^{-3}$		$2.36 \times 10^{-3}$	$7.44 \times 10^{-4}$	$1.14 \times 10^{-4}$	$3.96 \times 10^{-4}$
	70 - 74	$3.32 \times 10^{-4}$	$1.34 \times 10^{-3}$		$1.78 \times 10^{-3}$	$9.48 \times 10^{-4}$	$3.33{ imes}10^{-4}$	$4.07 \times 10^{-4}$
	75 - 79	$2.82 \times 10^{-4}$	$1.33 \times 10^{-3}$		$2.15 \times 10^{-3}$	$7.64 \times 10^{-4}$	$2.87{ imes}10^{-4}$	$4.68 \times 10^{-4}$
	80 - 84	$2.48 \times 10^{-4}$	$1.53 \times 10^{-3}$		$3.34 \times 10^{-3}$	$2.06 \times 10^{-3}$	$5.03 \times 10^{-4}$	$7.29 \times 10^{-4}$
3-4 ADLs	65 - 69	$8.57 \times 10^{-4}$	$3.20 \times 10^{-3}$	$1.43 \times 10^{-2}$		$4.81 \times 10^{-3}$	$5.65 \times 10^{-4}$	$1.63 \times 10^{-3}$
	70 - 74	$6.65 \times 10^{-4}$	$1.57 \times 10^{-3}$	$5.84 \times 10^{-3}$		$6.60 \times 10^{-3}$	$6.16 \times 10^{-4}$	$1.29 \times 10^{-3}$
	75 - 79	$4.29 \times 10^{-4}$	$2.09 \times 10^{-3}$	$8.04 \times 10^{-3}$		$5.87 \times 10^{-3}$	$1.69 \times 10^{-3}$	$2.09 \times 10^{-3}$
	80 - 84	$4.73 \times 10^{-4}$	$3.17 \times 10^{-3}$	$1.00 \times 10^{-2}$		$1.94 \times 10^{-2}$	$2.38 \times 10^{-3}$	$3.10 \times 10^{-3}$
5-6 ADLs	65 - 69	$4.08 \times 10^{-4}$	$1.14 \times 10^{-3}$	$2.48 \times 10^{-3}$	$3.12 \times 10^{-3}$		$3.85 \times 10^{-4}$	$1.40 \times 10^{-3}$
	70 - 74	$3.74 \times 10^{-4}$	$9.98 \times 10^{-4}$	$2.19 \times 10^{-3}$	$2.84 \times 10^{-3}$		$7.89 \times 10^{-4}$	$1.54 \times 10^{-3}$
	75 - 79	$2.16 \times 10^{-4}$	$6.62 \times 10^{-4}$	$1.14 \times 10^{-3}$	$2.42 \times 10^{-3}$		$8.76 \times 10^{-4}$	$1.32 \times 10^{-3}$
	80 - 84	$3.45 \times 10^{-4}$	$1.42 \times 10^{-3}$	$2.72 \times 10^{-3}$	$6.10 \times 10^{-3}$	-	$1.20 \times 10^{-3}$	$2.19 \times 10^{-3}$
Inst'd	65 - 69	$2.13 \times 10^{-4}$	$8.37 \times 10^{-5}$	$1.78 \times 10^{-4}$	$2.62 \times 10^{-4}$	$4.42 \times 10^{-5}$		$3.76 \times 10^{-4}$
	70 - 74	$9.75 \times 10^{-5}$	$7.73 \times 10^{-5}$	$1.93 \times 10^{-4}$	$1.57 \times 10^{-4}$	$2.65 \times 10^{-4}$		$1.02 \times 10^{-3}$
	75 - 79	$3.61 \times 10^{-5}$	$8.65 \times 10^{-5}$	$5.93 \times 10^{-5}$	$6.94 \times 10^{-5}$	$7.47 \times 10^{-5}$		$5.58 \times 10^{-4}$
	80 - 84	$1.58 \times 10^{-5}$	$2.51 \times 10^{-5}$	$2.97 \times 10^{-5}$	$5.17 \times 10^{-5}$	$3.31 \times 10^{-5}$		$4.03 \times 10^{-4}$

The adjusted log-likelihood,  $\overline{l}_r$ , where:

$$\overline{l}_r \propto \sum_{all\,i,j} \left( \overline{n}_{r\,r+t}^{i\,j} \log \left[ P_{r\,r+t}^{i\,j} \left( \overline{\mu}_r^{1\,2}, \dots, \overline{\mu}_r^{n\,n-1} \right) \right] \right) \tag{5.61}$$

is then (by definition) maximised by the constrained (positive) MLEs of the transition intensities, and so can be used to calculate variance estimates of these transition intensities using the information matrix, as in Section 5.2. These estimates are illustrated in Figures 5.36 to 5.41 as horizontal lines, to which the 'boot-strap' estimates converge. Thus, by adjusting the numbers of lives moving between states to be consistent with the transition probabilities calculated from the constrained (positive) MLEs of the transition intensities, the information matrix can be used to calculate, in a much more efficient way than the 'boot-strap' method, variance estimates of the constrained (positive) MLEs of the transition intensities. These estimates for male, females and in aggregate for the 1982–1984 NLTCS are given in 5-year age bands in Tables 5.61 to 5.63, respectively (except for the 85+ age group which is included in the tables for data grouped in 10-year age bands). The same variance estimates in 10-year age bands for the 1982–84 NLTCS and in 5 and 10-year age bands for the 1984–89 and 1989–94 NLTCS are given in Appendix J.

These variance estimates are useful for the two reasons:

- 1. they provide a set of weights to use in the graduation process (see Sections 5.5 and 5.6); and
- 2. the 95% confidence intervals for the MLEs (for  $\hat{\mu}_{x+t}^{ij}$ ) can be calculated and used to check if the graduated transition intensities lie within these limits.

There are a few abnormally large (> 1) variance estimates for males aged over 85 years in the 1989–94 NLTCS (for example, the variance of the transition intensity from '3–4 ADLs' to '1–2 ADLs' is 24.8). These large variance estimates only occur for transition intensities out of two states — 'IADL only' and '3–4 ADLs', suggesting that it may be caused by a small exposed to risk for these states. From Table D.100, the exposed to risk for the 'IADL only' state is 69.55 life years (over 5 years) and for the '3–4 ADLs' state it is only 45.7 life years (over 5 years), compared to, for example, 110.55 life years (over 5 years) for the '1–2 ADLs state'. A second factor

is that the number of observed transitions for those transition intensities with large variance, is relatively high (for example, there were 3.15 observed transitions from '3–4 ADLs' to '1–2 ADLs', when the 5-year exposed to risk is only 45.7). These two factors in combination, coupled with uncertainty in the other parameters, give rise to very high variance estimates. With such uncertainty, caution is advised when estimating or extrapolating from this dataset.

In the next two sections, I graduate the constrained (positive) MLEs of the transition intensities, in 10 and 5-year age bands, respectively.

## 5.5 Graduating the Transition Intensities Grouped in 10-year Age Bands

I chose a very simple graduation procedure. With the data grouped in 10-year age bands (65–74 years, 75–84 years and 85+ years) there are only 3 data points and so the maximum number of parameters that can be used for the fit is also 3. In the fitting procedure, for simplicity, I assume that the transition intensity for an age group is actually a point estimate of the transition intensity for the mid-point of that age group plus half of the duration between surveys or age 90 years plus half of the duration between surveys for the 85+ years group (i.e. for the 1982–84 NLTCS,  $\bar{\mu}_{71}^{ij} = \bar{\mu}_{65-74}^{ij}, \bar{\mu}_{81}^{ij} = \bar{\mu}_{75-84}^{ij}$  and  $\bar{\mu}_{91}^{ij} = \bar{\mu}_{85+}^{ij}$ ). From Table 5.64, which gives the mean ages for each 10-year age band at the start of each survey period, this assumption seems reasonable, even though, as expected, the mean ages are slightly less than the mid-points for most age-groups.

The parametric form I chose was a Makeham curve, which has 3 parameters and, when the form of the data allow, can be fitted exactly to the data points (to fit a Makeham curve it is necessary that  $\overline{\mu}_{65-74}^{ij} - \overline{\mu}_{75-84}^{ij} < \overline{\mu}_{75-84}^{ij} - \overline{\mu}_{85+}^{ij}$  and that  $\overline{\mu}_{65-74}^{ij} \neq \overline{\mu}_{75-84}^{ij} \neq \overline{\mu}_{85+}^{ij}$ ). Where it was not possible to fit a Makeham curve, a straight line was fitted using weighted least squares, the weights  $(w_{x+t}^{ij})$  being the inverse of the variance (i.e.  $w_{x+t}^{ij} = 1/\text{Var}[\overline{\mu}_{x+t}^{ij}]$ ). The only exception to this procedure was if the Makeham fit had a very high exponential parameter (i.e. very steep gradient), in which case a linear fit was chosen — however, this was only the

Survey	Gender	Mean age for age group:				
year		65-74  yrs	75–84 yrs	85+ yrs		
		years	years	years		
1982	М	69.56	79.02	88.51		
	F	69.77	79.45	88.87		
	M & F	69.68	79.31	88.79		
1984	М	70.09	78.69	87.64		
	F	70.38	79.15	88.08		
	M & F	70.27	79.01	88.00		
1989	М	69.88	78.31	87.14		
	F	70.26	78.52	87.54		
	M & F	70.13	78.46	87.49		

Table 5.64: Mean ages of lives within each 10-year age band by gender and in aggregate, in the 1982, 1984 and 1989 NLTCS.

case for very few transition intensities (in the 1982–84 NLTCS from '3–4 ADLs' to 'IADL only', for females and in aggregate; in the 1984–89 NLTCS from '3–4 ADLs' to 'Healthy' and from 'Institutionalized' to 'IADL only' for males; and in the 1989– 94 NLTCS from '3–4 ADLs' to 'IADL only' for females and in aggregate). The form of the graduated transition intensities, for  $65 \le x + t \le 120$ , is then:

$$\mathring{\mu}_{x+t}^{ij} = \begin{cases} A_{ij} + B_{ij} e^{C_{ij}((x-(70+M))+t)} & \text{if } (\overline{\mu}_{70+M}^{ij} - \overline{\mu}_{80+M}^{ij}) < (\overline{\mu}_{80+M}^{ij} - \overline{\mu}_{90+M}^{ij}) , \\ & \overline{\mu}_{70+M}^{ij} \neq \overline{\mu}_{80+M}^{ij} \neq \overline{\mu}_{90+M}^{ij} \text{ and } |C_{ij}| < 0.5 \\ & A_{ij} + D_{ij} (x+t) & \text{otherwise} \end{cases}$$

and with a lower bound of zero on all intensities at all ages and where M = 1 for the 1982–84 NLTCS and M = 2.5 for the 1984–89 and 1989–94 NLTCS. The values for the parameters A, B, C and D are given in Table 5.65 for males and females together. The same tables for males and females separately in the 1982–84 NLTCS and for males and females separately and aggregated for the 1984–89 and 1989–94 NLTCS are given in Appendix K.

The reasons for choosing a Makeham curve where possible and a linear form otherwise are:

- 1. for modelling mortality a Makeham curve is a reasonable assumption, given the number of data points;
- 2. increasing disability is a predictor of mortality and as such can be seen to be

From	То	Parameter					
State	State	А	В	$\mathbf{C}$	D		
Healthy	IADL only	$-1.84 \times 10^{-2}$	$4.31 \times 10^{-2}$	$5.34 \times 10^{-2}$	-		
v	1–2 ADLs	$4.67 \times 10^{-3}$	$7.78 \times 10^{-3}$	$1.31 \times 10^{-1}$	-		
	3–4 ADLs	$-2.04 \times 10^{-2}$	-	-	$3.44 \times 10^{-4}$		
	5–6 ADLs	$3.94 \times 10^{-3}$	$2.48 \times 10^{-4}$	$2.20 \times 10^{-1}$	-		
	Inst'd	$-1.10 \times 10^{-4}$	$3.73 \times 10^{-3}$	$1.37{ imes}10^{-1}$	-		
	Dead	$-2.57{ imes}10^{-2}$	$5.07 \times 10^{-2}$	$3.71 \times 10^{-2}$	-		
IADL only	Healthy	$-6.12 \times 10^{-1}$	$8.53 \times 10^{-1}$	$-1.56 \times 10^{-2}$	-		
v	1-2 $ADLs$	$-2.72 \times 10^{-1}$	-	-	$7.61 \times 10^{-3}$		
	3–4 ADLs	$-1.78 \times 10^{-2}$	-	-	$3.41 \times 10^{-4}$		
	5–6 ADLs	$-1.05 \times 10^{-1}$	-	-	$2.01 \times 10^{-3}$		
	Inst'd	$-1.59 \times 10^{-1}$	-	-	$2.62 \times 10^{-3}$		
	Dead	$6.07 \times 10^{-2}$	$9.21 \times 10^{-4}$	$2.10 \times 10^{-1}$	-		
1-2 ADLs	Healthy	$3.40 \times 10^{-2}$	$3.74 \times 10^{-2}$	$-5.32 \times 10^{-2}$	-		
	IADL only	$5.61 \times 10^{-1}$	-	-	$-4.88 \times 10^{-3}$		
	3–4 ADLs	$1.88 \times 10^{-1}$	$1.61 \times 10^{-2}$	$9.59 \times 10^{-2}$	-		
	5–6 ADLs	$7.32 \times 10^{-2}$	-	-	$-5.68 \times 10^{-4}$		
	Inst'd	$4.09 \times 10^{-2}$	$2.64 \times 10^{-3}$	$1.69 \times 10^{-1}$	-		
	Dead	$-1.90 \times 10^{-2}$	-	-	$1.60 \times 10^{-3}$		
3–4 ADLs	Healthy	$4.14 \times 10^{-3}$	$3.07 \times 10^{-2}$	$-3.09 \times 10^{-1}$	-		
	IADL only	$-2.44 \times 10^{-1}$	-	-	$3.29 \times 10^{-3}$		
	1–2 ADLs	$1.66{ imes}10^{0}$	-	-	$-1.73 \times 10^{-2}$		
	5–6 ADLs	$5.30 \times 10^{-2}$	$2.34 \times 10^{-1}$	$4.13 \times 10^{-2}$	-		
	Inst'd	$-5.66{ imes}10^{-1}$	-	-	$8.25 \times 10^{-3}$		
	Dead	$1.69 \times 10^{-2}$	-	-	$8.65 \times 10^{-4}$		
5–6 ADLs	Healthy	$1.11 \times 10^{-1}$	-	-	$-1.05 \times 10^{-3}$		
	IADL only	$3.01 \times 10^{-1}$	-	-	$-3.25 \times 10^{-3}$		
	1-2 ADLs	$7.68 \times 10^{-3}$	-	-	$6.11 \times 10^{-4}$		
	3–4 ADLs	$1.83 \times 10^{-1}$	-	-	$-1.35 \times 10^{-4}$		
	Inst'd	$-1.65{ imes}10^{-1}$	-	-	$3.36 \times 10^{-3}$		
	Dead	$2.40 \times 10^{-1}$	$9.76 \times 10^{-4}$	$2.43 \times 10^{-1}$	-		
Inst'd	Healthy	$3.57 \times 10^{-3}$	$2.74 \times 10^{-2}$	$-1.75 \times 10^{-1}$	-		
	IADL only	$2.96 \times 10^{-2}$	-	-	$-2.95 \times 10^{-4}$		
	1-2 ADLs	$-5.61 \times 10^{-4}$	$1.13 \times 10^{-2}$	$-1.50{ imes}10^{-1}$	-		
	3–4 ADLs	$2.59 \times 10^{-2}$	-	-	$-1.83 \times 10^{-4}$		
	5–6 ADLs	$4.12 \times 10^{-2}$	-	-	$-4.21 \times 10^{-4}$		
	Dead	$4.90 \times 10^{-2}$	$1.25 \times 10^{-1}$	$4.19 \times 10^{-2}$	_		

Table 5.65: Parameter values for the parametric transition intensities for males and females grouped in 10-year age bands, calculated from the 1982 and 1984 NLTCS.

part of the ageing process, so movement through stages of disability may then behave in a similar fashion; and

3. recovery from disability decreases with age and a negative exponential may be appropriate.

In the next section, when there are more data points (because of smaller age groups) I look at the trends more closely, however in this case I felt the data did not justify a more complex procedure.

Figures 5.42 to 5.47 give graphs of the transition intensities, for males and females, out of states 1–6, respectively, for the 1982–84 NLTCS. They show the point estimates, in 10-year age bands (constrained (positive) MLEs), the confidence intervals using the variance estimates from the previous section and the parametric form of the transition intensities. While all graphs in a given figure use the same scale, it is worth noting that different figures use different scales and that the scales were chosen such that the confidence intervals could be shown on the graphs, with the exception of when the confidence intervals were very large requiring very large scales which would obscure the patterns of the point estimates — in these cases the scale is limited to -0.2 to 0.5 (or the point estimate of the largest transition intensity plus 0.05 of its standard deviation, if the largest point estimate is greater than 0.5). All the Makeham curves pass though the confidence intervals (as all the point estimates lie inside the confidence intervals). It is also noticeable that almost all of the linear fits also pass through the confidence intervals (in fact only 13 of the linear fits do not pass through the confidence intervals of the transition intensities, which is less than 5% of all the parametric fits).

For brevity, I exclude similar graphs for all of the other datasets, since there is no new information in them — they just provide a visual summary of information given elsewhere.

In general the transition intensities are as expected: the force of disability and mortality increases with increasing age; and the force of recovery decreases with increasing age. There are, however, some notable exceptions (e.g. from the healthy state to 3–4 ADLs), all of these involve small transition intensities though, calculated from relatively few data (they are all linear fits, as well, suggesting no clear pattern


Figure 5.42: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.

between the transition intensities).

Figure 5.48, which compares a Makeham fit to a weighted least squares fit for the transition intensity 'Healthy' to '1–2 ADLs' for males in the 1982–84 NLTCS, illustrates a potential problem with fitting a Makeham curve to three data points — it gives equal weight (necessarily, as it must pass through all three points) to all three points, irrespective of the variance of each point estimate. This means that, in some cases, the size of the exponential parameter in a Makeham fit is most sensitive to the estimate that has the largest variance — that for the oldest age group (which generally has the largest variance as it generally has the scarcest data). While this instability in the fitting procedure is worth noting, with so few data points, it is difficult to overcome. However, in the next section, with the data grouped into 5-year age bands (providing 5 data points), the fitting procedure can be made more robust.



Figure 5.43: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.



Figure 5.44: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.



Figure 5.45: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.



Figure 5.46: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.



Figure 5.47: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for males and females grouped in 10-year age bands in the 1982–84 NLTCS.



Figure 5.48: Comparison of a Makeham fit with a weighted linear least squares fit for the transition intensity 'Healthy' to '1–2 ADLs' for males grouped in 10-year age bands in the 1982–84 NLTCS.

Survey	Gender	Mean age for age group				
year		6569  yrs	$7074~\mathrm{yrs}$	75–79 yrs	8084  yrs	85 + yrs
		years	years	years	years	years
1982	М	67.46	72.37	77.33	82.29	88.51
	F	67.49	72.45	77.43	82.34	88.87
	M & F	67.48	72.42	77.39	82.32	88.79
1984	М	67.72	72.28	77.20	82.31	87.64
	F	67.76	72.43	77.37	82.28	88.08
	M & F	67.74	72.37	77.32	82.29	88.00
1989	М	67.68	72.19	77.10	82.25	87.14
	F	67.73	72.38	77.22	82.07	87.54
	M & F	67.72	72.32	77.19	82.11	87.49

Table 5.66: Mean ages of lives within each 5-year age band by gender and in aggregate, in the 1982, 1984 and 1989 NLTCS.

## 5.6 Graduating the Transition Intensities Grouped in 5-year Age Bands

I chose a similar graduation procedure to the previous section. With the data grouped in 5-year age bands (65–69 years, 70–74 years, 75–79 year, 80–84 and 85+ years) there are 5 data points. I assume, again, that the transition intensity for an age group is actually a point estimate of the transition intensity for the mid-point of that age group plus half of the duration between surveys or age 90 years plus half of the duration between surveys for the 85+ years group (i.e. for the 1982–84 NLTCS,  $\overline{\mu}_{68.5}^{ij} = \overline{\mu}_{65-69}^{ij}, \overline{\mu}_{73.5}^{ij} = \overline{\mu}_{70-74}^{ij}, \overline{\mu}_{78.5}^{ij} = \overline{\mu}_{75-79}^{ij}, \overline{\mu}_{83.5}^{ij} = \overline{\mu}_{80-84}^{ij}$  and  $\overline{\mu}_{91}^{ij} = \overline{\mu}_{85+}^{ij}$ ). From Table 5.66, which gives the mean ages for each 5-year age band at the start of each survey period, this assumption seems reasonable, even though, as expected, the mean ages are slightly less than the mid-points for most age-groups.

For the graduation, I chose the same two parametric forms as in the previous section — either a Makeham curve or a straight line, which from the shape of the data, (illustrated in Figures 5.49 to 5.54 and in Appendices M to O) seem to provide adequate flexibility. It would be possible to fit a more complex parametric form (more parameters), but this would be at the cost of the smoothness of the fit, and with only five data points, I feel this is not justified.

The method I use is, for each transition intensity, to fit both a staight line and a

Makeham curve using weighted least squares, the weights  $(w_{x+t}^{ij})$  being the inverse of the variance (i.e.  $w_{x+t}^{ij} = 1/\text{Var}[\overline{\mu}_{x+t}^{ij}]$ ). Then I chose the parametric form which provides the best fit (in terms of the smallest sum of weighted squared residuals). The form of the graduated transition intensities, for  $65 \leq x + t \leq 120$ , is then, depending on which provides a better fit, either:

$$\overset{\circ}{\mu}_{x+t}^{ij} = A_{ij} + B_{ij} e^{C_{ij}((x - (67.5 + M)) + t)}$$

or:

$$\mathring{\mu}_{x+t}^{i\,j} = A_{i\,j} + D_{i\,j}\,(x+t)$$

and with a lower bound of zero on all intensities at all ages and where M = 1 for the 1982–84 NLTCS and M = 2.5 for the 1984–89 and 1989–94 NLTCS. There was no explicit limit placed on the  $C_{ij}$  here, since this fitting procedure did not produce any unreasonably large estimates of  $C_{ij}$  (discussed in more detail later in this section). The values for the parameters A, B, C and D are given in Table 5.67 for males and females together. The same tables for males and females separately in the 1982–84 NLTCS and for males and females separately and aggregated for the 1984–89 and 1989–94 NLTCS are given in Appendix L.

Figures 5.49 to 5.54 give graphs of the transition intensities, for males and females, out of states 1–6, respectively, for the 1982–84 NLTCS. They show the point estimates, in 5-year age bands (constrained (positive) MLEs), the confidence intervals using the variance estimates from the previous section and the parametric form of the transition intensities. The same graphs for males and females separately for the 1982–84 NLTCS are given in Appendix M and those for males and females separately and aggregated for the 1984–89 and 1989–94 NLTCS are given in Appendices N and O, respectively. While all graphs in a given figure use the same scale, it is worth noting that different figures use different scales and that the scales were choosen such that the confidence intervals could be shown on the graphs, with the exception of when the confidence intervals were very large requiring very large scales which would obscure the patterns of the point estimates — in these cases the scale is limited to -0.2 to 0.5 (or the point estimate of the largest transition intensity plus 0.05 of its standard deviation, if the largest point estimate is greater than 0.5). In general, the parametric fits are similar to those in the previous section when using only 3 data points. The main differences are:

- in the choice a linear fit over a Makeham fit in some cases in the previous section if a Makeham curve could be fitted it was, whereas with 5 data points a linear fit was chosen over a Makeham fit if it provided a better fit (for example for males and females in the 1982–84 NLTCS: a linear fit was chosen over a Makeham fit (which was used with 3 data points) for the transition intensities 'Healthy' to 'Dead', 'IADL only' to 'Dead', '1–2 ADLs' to 'Inst'd', '3–4 ADLs' to 'Healthy', etc.);
- 2. where a Makeham fit was chosen using 3 and 5 data points, the exponential parameter of the fit using 5 data points is generally less (sometimes substantially less) than that using 3 data points (for example for males and females in the 1982–84 NLTCS: 'Healthy' to '5–6 ADLs', 'IADL only' to 'Dead', '1–2 ADLs' to 'Inst'd', '5–6 ADLs' to 'Dead' and 'Inst'd' to '1–2 ADLs').

In the previous section, with three data points, there were problems of sensitivity with the Makeham fits — as all data points had equal weighting (as the Makeham fit had to pass through all three points, irrespective of the size of the variance estimates). In some cases this led to high exponential paramaters  $(C_{ij})$ , for the Makeham fit — which I then, subjectively, rejected in favor of a linear fit. This problem does not arise in this fitting procedure as the Makeham fit is now a weighted fit, and so parameters with high variance are given less weight accordingly. In fact, the largest positive exponential parameter is 0.174 (for males and females from 'Healthy' to '1–2 ADLs' in the 1982–84 NLTCS) and the largest negative exponential parameter is 0.277 (for males from '3–4 ADLs' to '1–2 ADLs' in the 1989–94 NLTCS). This provides support for limiting the exponential parameter in the Makeham fits in the previous section.

When analysing age-dependent data like that from the NLTCS, there is always the trade-off between:

1. having age bands large enough to ensure that there are enough observations in each age band for the estimates to be useful (too few observations lead to

From	То	Parameter				
State	State	А	В	$\mathbf{C}$	D	
Healthy	IADL only	$-3.22 \times 10^{-2}$	$5.19 \times 10^{-2}$	$4.35 \times 10^{-2}$	-	
· ·	1-2 ADLs	$9.58 \times 10^{-3}$	$2.11 \times 10^{-3}$	$1.74 \times 10^{-1}$	-	
	3–4 ADLs	$-2.34 \times 10^{-2}$	-	-	$3.85 \times 10^{-4}$	
	5–6 ADLs	$-1.37{ imes}10^{-4}$	$3.16 \times 10^{-3}$	$8.01 \times 10^{-2}$	-	
	Inst'd	$-9.05 \times 10^{-4}$	$3.15 \times 10^{-3}$	$1.32 \times 10^{-1}$	-	
	Dead	$-1.62{ imes}10^{-1}$	-	-	$2.64 \times 10^{-3}$	
IADL only	Healthy	$1.04 \times 10^{0}$	-	-	$-1.13 \times 10^{-2}$	
v	1-2 ADLs	$-3.38{ imes}10^{-1}$	-	-	$8.32 \times 10^{-3}$	
	3–4 ADLs	$2.94 \times 10^{-2}$	-	-	$-1.59 \times 10^{-4}$	
	5–6 ADLs	$-9.89 \times 10^{-2}$	$1.33 \times 10^{-1}$	$8.16 \times 10^{-3}$	-	
	Inst'd	$-1.81 \times 10^{-1}$	-	-	$2.90 \times 10^{-3}$	
	Dead	$-3.19 \times 10^{-2}$	$8.80 \times 10^{-2}$	$1.60 \times 10^{-2}$	-	
1-2 ADLs	Healthy	$1.74 \times 10^{-1}$	-	-	$-1.45 \times 10^{-3}$	
	IADL only	$5.45 \times 10^{-1}$	-	-	$-4.71 \times 10^{-3}$	
	3–4 ADLs	$1.85 \times 10^{-1}$	$5.62 \times 10^{-3}$	$1.33 \times 10^{-1}$	-	
	5–6 ADLs	$-6.10 \times 10^{-2}$	$1.04 \times 10^{-1}$	$-1.11 \times 10^{-2}$	-	
	Inst'd	$-5.61 \times 10^{-2}$	$7.72 \times 10^{-2}$	$3.48 \times 10^{-2}$	-	
	Dead	$-4.68 \times 10^{-2}$	-	-	$1.93 \times 10^{-3}$	
3–4 ADLs	Healthy	$1.03 \times 10^{-1}$	-	-	$-1.11 \times 10^{-3}$	
	IADL only	$-4.26 \times 10^{-3}$	$2.14 \times 10^{-3}$	$1.48 \times 10^{-1}$	-	
	1–2 ADLs	$1.61{ imes}10^{0}$	-	-	$-1.69{ imes}10^{-2}$	
	5-6 ADLs	$1.64 \times 10^{-2}$	$2.13 \times 10^{-1}$	$4.51 \times 10^{-2}$	-	
	Inst'd	$-9.20 \times 10^{-2}$	$1.09 \times 10^{-1}$	$3.52 \times 10^{-2}$	-	
	Dead	$1.27 \times 10^{-1}$	-	-	$-5.50 \times 10^{-4}$	
5–6 ADLs	Healthy	$1.06 \times 10^{-1}$	-	-	$-9.93 \times 10^{-4}$	
	IADL only	$2.85 \times 10^{-1}$	-	-	$-3.08 \times 10^{-3}$	
	1-2 ADLs	$-1.81 \times 10^{-1}$	$2.23 \times 10^{-1}$	$4.62 \times 10^{-3}$	-	
	3–4 ADLs	$1.40 \times 10^{-1}$	-	-	$3.16 \times 10^{-4}$	
	Inst'd	$-2.00 \times 10^{-1}$	-	-	$3.80 \times 10^{-3}$	
	Dead	$1.76 \times 10^{-1}$	$4.53 \times 10^{-2}$	$5.28 \times 10^{-2}$	-	
Inst'd	Healthy	$2.39 \times 10^{-3}$	$2.84 \times 10^{-2}$	$-1.19 \times 10^{-1}$	-	
	IADL only	$2.89 \times 10^{-2}$	-	-	$-2.90 \times 10^{-4}$	
	1–2 ADLs	$-3.10 \times 10^{-2}$	$3.89 \times 10^{-2}$	$-1.02 \times 10^{-2}$	-	
	3–4 ADLs	$-1.94{ imes}10^{-1}$	$2.05 \times 10^{-1}$	$-3.68 \times 10^{-4}$	-	
	5-6 ADLs	$9.87{ imes}10^{-3}$	-	-	$-6.85{ imes}10^{-5}$	
	Dead	$-5.71 \times 10^{-1}$	-	-	$9.98 \times 10^{-3}$	

Table 5.67: Parameter values for the parametric transition intensities for males and females grouped in 5-year age bands, calculated from the 1982 and 1984 NLTCS.

estimates with very large confidence intervals); and

2. having enough separate groups (i.e. smaller age bands) to ensure that the trends of the transition intensities with age are clear.

The main reason for grouping the data into 3 age bands (65-74, 75-84 and 85+years), as in the previous section, is to allow comparison with previous research (see Section 3.5), since these are the groupings that were used there. However, while using 3 age bands ensures greater numbers of observations in each age band, this is at the cost of having only three data points with which to graduate the estimates — which caused the fitting procedure to be unstable in some cases as the trend of the transition intensities with age was not clear. With 5 data points, this instability in the fitting procedure does not occur, as there are enough estimates to use a more robust method (weighted least squares) when fitting a Makeham curve. Also, using 5 age bands does not, in general, worsen the reliability of the estimates of the transition intensities by much (as indicated by slightly larger confidence intervals). In fact, this problem of scarce data is only apparent for the oldest age band, 85+years, in some cases (for example, in the 1989–94 NLTCS, for males, the variance estimates of the transition intensities: IADL only to 3-4 ADLs and 3-4 ADLs to 1-2 ADLs) and this age group is the same, whether the data is grouped into 3 or 5 age bands. Given that using 5 age bands has the advantage of then being able to use a more stable fitting procedure for graduating the transition intensities, and that this does not worsen the estimates by much, I will place more reliance on the graduated transition intensities that used data grouped into 5 age bands. The graduated transition intensities using data grouped into 3 age bands, will still be useful for two reasons:

- 1. for comparison with previous research; and
- 2. for comparing with those fitted to the data in 5 age bands.

In the next chapter, I look at the overall forces of mortality in the models and compare them to mortality from other investigations.



Figure 5.49: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males and females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure 5.50: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for males and females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure 5.51: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males and females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure 5.52: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males and females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure 5.53: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males and females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure 5.54: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for males and females grouped in 5-year age bands in the 1982–84 NLTCS.

#### Chapter 6

# Overall Mortality in the Disability Model

#### 6.1 Introduction

The aim of this chapter is to compare implied aggregate (across disability states), or overall, mortality from the parameterized disability models to that of standard mortality tables, or other mortality investigations.

To avoid confusion, I will refer to the mortality of males and females combined as aggregate mortality and mortality aggregated across all disability states for a given model as *overall* mortality.

Overall mortality at any given age in the disability model is the weighted sum of the force of mortality in each non-dead state, where the weights are the probability of a life being in any given state. Assuming that a life is healthy at age 65 (a reasonable assumption for an insured life), the probability of the life being in state j at age 65 + t is then simply  $P_{6565+t}^{1j}$ , the probability that the life moves from state 1 (Healthy) to state j in time t. The overall force of mortality,  $\mu_{65+t}$ , at any age 65 + t can then be calculated as:

$$\mu_{65+t} = \frac{\sum_{i=1}^{6} \mu_{65+t}^{i\,7} \mathbf{P}_{65\,65+t}^{1\,i}}{\sum_{i=1}^{6} \mathbf{P}_{65\,65+t}^{1\,i}} \tag{6.62}$$

where  $P_{6565+t}^{1i}$  can be calculated by solving Kolmogorov's equations, for any given set of transition intensities and  $\mu_{65+t}^{i7}$  is the force of mortality in state *i* at age 65 + t. The motivation for comparing overall model mortality with a benchmark force of mortality is two-fold:

- the force of mortality has been subject to large scale investigations, which provide benchmarks to compare the overall force of mortality from the disability model to see if it produces reasonable estimates; and
- 2. if necessary, the transition intensities in the model can be adjusted to be consistent with the findings from these large scale investigations, to make the model more reasonable for use in application (for example, in estimating the cost of disability in a long-term care insurance contract).

First, in Section 6.2, I summarise an investigation into mortality at old ages and extract a benchmark model of mortality from it. Then in Section 6.3, I compare this benchmark mortality with overall model mortality for the disability models parameterized from the 1982–84 NLTCS (which I refer to as the 1982–84 models). In Sections 6.4 and 6.5, I compare this benchmark force of mortality with overall model mortality for the disability models parameterized using the 1984–89 NLTCS and 1989–94 NLTCS, respectively — which leads to adjustments being made to mortality in these models (the models after these adjustments have been made, I refer to as the adjusted 1984–89 and the adjusted 1989–94 models). Finally, in Section 6.6, I compare overall mortality from the (adjusted) disability models across all three survey periods and comment on their application to estimating costs of disability in a long-term care insurance contract.

### 6.2 A Benchmark Force of Mortality at Older Ages

The monograph by Thatcher, Kannisto, & Vaupel (1998) provides a comprehensive discussion and analysis of the force of mortality at ages 80–120. In summary:

 they use data from 13 countries — Austria, Denmark, England and Wales, Finland, France, Germany (West), Iceland, Italy, Japan, the Netherlands, Sweden and Switzerland;

- 2. these include almost 40 million lives that survived to age 80, over 120,000 that reached age 100 and covers over 32 million deaths at age 80 and over;
- 3. the period covered is 1960–1990, and the analysis is done in 10-year age periods and cohorts (1960–70, 1970–80 and 1980–90); and
- 4. the analysis was done by gender for each country separately and all countries together;

The approach taken was, after a comprehensive literature review, to choose 6 models for comparison — 4 explanatory models and 2 descriptive, as follows:

1. Gompertz's law of mortality:

$$\mu_{x+t} = a \,\mathrm{e}^{b(x+t)} \tag{6.63}$$

This was chosen over a Makeham fit as the difference between them was negligible — the constant term in the Makeham fit was very small, especially at older ages where the exponential term dominates.

2. the logistic model:

$$\mu_{x+t} = c + \frac{a e^{b(x+t)}}{1 + \sigma^2 \frac{a}{b} (e^{b(x+t)} - 1)}$$
(6.64)

Of which the Makeham fit is a special case — when  $\sigma = 0$ .

3. the Kannisto model (another special case of the logistic model):

$$\mu_{x+t} = \frac{a e^{b(x+t)}}{1 + a e^{b(x+t)}} \tag{6.65}$$

4. the Weibull model:

$$\mu_{x+t} = a \, (x+t)^b \tag{6.66}$$

5. the Heligman & Pollard model, which describes the form for the one year probability of dying between ages x + t and x + t + 1,  $q_{x+t}$ :

$$q_{x+t} = \frac{a e^{b(x+t)}}{1 + a e^{b(x+t)}}$$
(6.67)

6. and the quadratic model

$$\mu_{x+t} = e^{a+b(x+t)+c(x+t)^2} \quad \text{with } c < 0 \tag{6.68}$$

These models were fitted to the data using the method of maximum likelihood at ages 80–98 years (85–98 years for the quadratic model) and then extrapolated up to age 120 years. The extrapolated values for each model were then compared to the actual values estimated from the data (with the data from all countries combined), by looking at the value of the likelihood function for each fit and by comparing values of the chi-squared statistic. For three of the models the conclusions are clear, the Gompertz, Weibull and Heligman & Pollard models always provided a worse (by overstating mortality at the oldest ages), and generally far worse, fit than the logistic, Kannisto and quadratic models.

The usefulness of this monograph is that it gives a benchmark force of mortality with which to compare overall mortality from the disability model. In this respect, all three of the models that provide superior fits (the logistic, Kannisto and quadratic models) would be suitable, as they are the same general shape (in the age range 85– 120) and are almost at the same level. Another consideration is extrapolation of the fitted mortality down to age 65 (since the disability model is parameterized over the age range 65–120 years) — for which the quadratic model is not suitable. A disadvantage of the quadratic model is that it is only a descriptive model and while it provides a good fit in the age range 85–120 years, it cannot be sensibly extrapolated to younger or older ages.

Of the remaining two models, in some cases the Kannisto model provided a better fit than the logistic model, which, as the authors pointed out, is anomalous, because the Kannisto model is a special case of the logistic model and so cannot provide a better fit with both models fitted to an entire data set. The problem here lies in fitting the models to one data set (80–98 years) and then comparing them on another set (98–120 years).

For the purposes of comparison with overall mortality in the disability model, the fits from either the Kannisto model or the logistic model would be adequate, as the differences between them are small. I therefore chose the logistic model, since it is more general. Table 6.68 gives the parameter values for the logistic model fitted to Table 6.68: Parameter values for the logistic model fitted to data from 13 countries at ages 80–98 years for males and females separately over the period 1980–90, from Thatcher, Kannisto & Vaupel (1998).

Parameter	Gender				
	males	females			
a	$2.99258 \times 10^{-5}$	$1.46725 \times 10^{-5}$			
b	$1.02220 \times 10^{-1}$	$1.08791 \times 10^{-1}$			
c	$9.93451 \times 10^{-6}$	$9.99586 \times 10^{-6}$			
$\sigma$	$2.89834 \times 10^{-1}$	$3.47119 \times 10^{-3}$			

the data for all countries combined at ages 80–98 for males and females separately, in the period 1980–90 (as most of the NLTCS took place during this period), taken from Thatcher, Kannisto & Vaupel (1998). These fits are given in Figure 6.55 for comparison with the overall mortality from the disability model — unfortunately, for males and females combined no logistic fit was given and I use the logistic fit for females (as the majority of the population is female at older ages due to the usual mortality differential) for comparison with overall mortality for males and females combined in the disability model.

# 6.3 Overall Model Mortality, Using the Fitted Parametric Transition Intensities from the 1982–84 NLTCS

In this section, I compare the overall force of mortality in the disability models using the parametric transition intensities fitted to the 1982–84 NLTCS (which I will refer to as the 1982–84 models) with the benchmark force of mortality described in the previous section. I then investigate the shape of overall model mortality by looking at its separate components and finally discuss whether adjustments should be made to the 1982–84 models. For each pair of NLTCS and for each gender classification, two sets of parametric transition intensities were fitted: those with the data grouped into 10-year age bands; and those with the data grouped into 5-year age bands. For brevity and to avoid confusion I will refer to the former as the '1982–84 (10) models'



Figure 6.55: Overall model mortality for the 1982–84 (5) models and the 1982–84 (10) models compared with a logistic model of mortality by Thatcher, Kannisto & Vaupel (1998).

and the latter as the '1982–84 (5)' models, for the 1982–84 NLTCS.

In Figure 6.55, overall mortality in the 1982–84 (10) models and the 1982–84 (5) models, for males, females and in aggregate, are compared to those of the logistic model (see previous section for more detail). In general, overall mortality in the disability model is in reasonable agreement with that of the logistic model, though a few noticeable features are:

- overall mortality in the disability models is in very good agreement with the logistic model up to ages 90–95 years (which is the oldest age group for which there was a point estimate when estimating the parametric transition intensities);
- 2. after age 95 years overall mortality in the disability models falls below that of the logistic model;
- the 1982–84 (10) models always produce greater overall mortality than the 1982–84 (5) models;

- 4. overall mortality in the 1982–84 (10) model, for males and females combined, is the closest to that of the logistic model they are in very close agreement up to just below age 110 years (however, it should be noted that the logistic fit is that for females, so it would be expected that it should be below that for males and females combined); and
- 5. as expected, overall model mortality for males is always above that for females, though overall mortality in the 1982–84 (5) model for males and females combined seems anomalous it is below that for females (this is possible because the parametric transition intensities were fitted separately for each gender classification, so that overall model mortality for males and females combined is not a weighted sum of that for males and females taken separately).

The form of overall model mortality can be investigated by looking at its separate components — the occupancy probabilities of each state (conditional on being alive) and the force of mortality for each non-dead state. For males, these are illustrated in Figures 6.56 and 6.57, for females in Figures 6.58 and 6.59 and for males and females combined in Figures 6.60 and 6.61.

Looking at the overall model mortality for males first, it is clear (from Figures 6.56 and 6.57) why overall mortality is higher for the 1982–84 (10) model, since the parametric force of mortality is higher for each individual state (except the IADL only state, which has little influence, as it has low occupancy probabilities). It can also be seen that overall model mortality levels off at older ages because of the increasing occupancy probability of the 1–2 ADLs state, which has a fairly low and linearly increasing force of mortality.

For females, the reason why overall mortality is higher for the 1982–84 (10) model is not so clear, though after age 90 for the three states with the highest occupancy probabilities (1–2 ADLs state, 5–6 ADLs state and the Institutionalised state), the parametric force of mortality in the 1982–84 (10) model is greater than, or equal to, that of the 1982–84 (5) model. The reason why it levels off at older ages differs from that in the case of males:

1. for the 1982–84 (10) model, it is because of the increasing proportion of lives



Figure 6.56: Model occupancy probabilities, conditional on being alive, for the 1982–84 (5) models (solid line) and the 1982–84 (10) models (dotted line), for males.



Figure 6.57: The parametric force of mortality for the 1982–84 (5) NLTCS models (solid line) and 1982–84 (10) NLTCS models (dotted line), for males.

in the 1–2 ADLs state, the 3–4 ADLs state and the 5–6 ADLs state (all of which have low and linearly increasing fitted forces of mortality); but

 for the 1982–84 (5) model, the main influence is from the high proportion of lives in the institutionalised state (which has a linearly increasing force of mortality).

For males and females combined, the reason why overall mortality is higher for the 1982–84 (10) model is similar to that in the case of males. Overall mortality for the 1982–84 (5) model is the same shape as that for males and females separately it levels off at older ages (mainly due to the high occupancy of the institutionalised state at older ages, which has a linearly increasing force of mortality). However, overall mortality for the 1982–84 (10) model continues to increase exponentially with age, even at the oldest ages. This is because, even though the force of mortality in the institutionalised state increases exponentially with age, the occupancy probability (conditional on being alive) of this state is almost 1 by age 100, and remains at this level. This must be because of very high rates of institutionalizion (directly or indirectly) from all the other non-dead states at older ages and low rates of recovery.

Having compared the overall force of mortality in the 1982–84 models with a benchmark force of mortality, the question is whether or not mortality in the model should be adjusted in some manner, to be more consistent with the benchmark force of mortality. In adjusting model mortality a number of factors need to be considered:

- 1. whether mortality should be uniformly adjusted across all disability groups and if not, in what manner should they be adjusted;
- 2. whether the choice between a linear fit and an exponential fit for the parametric transition intensities should be considered;
- 3. the nature of the adjustment, for example whether the force of mortality in each disability state should be adjusted by a constant amount, a muliplicative factor or an increase/decrease in the exponential factor;
- 4. how appropriate is the benchmark force of mortality (for example, the USA is not one of the 13 countries upon which it is based); and



Figure 6.58: Model occupancy probabilities, conditional on being alive, for the 1982–84 (5) models (solid line) and the 1982–84 (10) models (dotted line), for females.



Figure 6.59: The parametric force of mortality for the 1982–84 (5) models (solid line) and the 1982–84 (10) models (dotted line), for females.

Table 6.69: Model survival probabilities for lives in the healthy state at age 65 calculated from the 1982–84 (10) models and the 1982–84 (5) models, for males, females and in aggregate (agg).

Survival probability from age 65 for the adjusted:							
	$1982-84 (10) \mod 100$			1982-84~(5) model			
Age	males	females	$\operatorname{agg}$	males	females	$\operatorname{agg}$	
	%	%	%	%	%	%	
70	83.010	94.007	89.114	87.277	94.578	90.915	
75	62.866	81.566	73.128	66.596	84.233	75.137	
80	42.112	63.636	54.029	43.983	68.115	55.754	
85	23.518	42.752	34.397	24.447	47.662	36.092	
90	9.946	23.144	17.411	10.869	27.256	19.411	
95	2.739	9.161	5.921	3.545	11.925	8.038	
100	0.409	2.391	0.992	0.760	3.728	2.332	
105	0.031	0.396	0.074	0.099	0.794	0.444	
110	0.001	0.043	0.003	0.008	0.114	0.056	

5. whether a factor of mortality improvement with time should be incorporated into the adjustments (the NLTCS is over the period 1982–84, while the data used for the logistic fit is over the period 1980–1990).

and most of these choices would necessarily be arbitrary (for example, due to lack of information about the force of mortality by disability status).

Another important consideration is the application of this model. I will use this parameterized model to look at the costs of disability in a long-term care contract, in which the most important age range is at younger ages (65–90 years), especially when the costs are discounted over time. Given that overall model mortality at younger ages is reasonably consistent with the benchmark force of mortality (so any adjustments made would need to be aimed primarily at older ages), together with the consideration of having to make some arbitrary decisions when adjusting mortality, I would argue against making any mortality adjustments here. However, for use in an insurance application, careful consideration would need to be made as to how mortality could be adjusted appropriately.

With no adustments made, a common feature of overall model mortality, irrespective of the method and set of data used to estimate the transition intensities, is that overall model mortality is much closer to the logistic fit at younger ages (65–90



Figure 6.60: Model occupancy probabilities, conditional on being alive, for the 1982–84 (5) models (solid line) and the 1982–84 (10) models (dotted line), for males and females combined.



Figure 6.61: The parametric force of mortality for the 1982–84 (5) models and the 1982–84 (10) models (dotted line), for males and females combined.

years) than at older ages (90 + years). This is not surprising though, considering that the oldest data point used in any of the fits is age 91 years and that after this age, all of the parametric transition intensities are extrapolations of the fits before this age — which may not be a reasonable assumption. Furthermore, the graphs of occupancy probabilities (Figures 6.56, 6.58 and 6.60) may be misleading since these occupancy probabilities have been adjusted for deaths (to aid in illustrating the components of overall model mortality). Table 6.69 gives the survival probabilities (as percentages) for each set of parametric transition intensities, assuming that a person is alive and in the healthy state at age 65. For the 1982–84 models, the probability that a person who was alive and in the healthy state at age 65 is still alive at age 100 is between 0.41% and 3.73% — a small percentage, which decreases to less than 0.12% by age 110. So, overall mortality in the model is determined by small occupancy probabilities at ages over 100 and must then be sensitive to small changes in the occupancy probabilities and thus the transition intensities which are, at these ages, extrapolations. Given these uncertainties in the parameter estimates at older ages, when I estimate the costs of disability in long-term care insurance in Section 7, I look at the sensitivity of these costs to arbitrary adjustments in the force of mortality at older ages (95 + years), as well as isolating the model costs attributable to disabled lives over age 95.

# 6.4 Overall Model Mortality, Using the Fitted Parametric Transition Intensities from the 1984–89 NLTCS

In this section I compare the overall force of mortality in the disability models using the parametric transition intensities fitted to the 1984–89 NLTCS (which I will refer to, using the notation of the previous section, as the 1984–89 (10) models and the 1984–89 (5) models, depending on the age bands used to group the original data) with the same benchmark force of mortality as the previous section. Figure 6.62 shows the overall model force of mortality in the disability model (as defined in



Figure 6.62: Overall model mortality for the 1984–89 (5) models and the 1984–89 (10) models compared with a logistic model of mortality by Thatcher, Kannisto & Vaupel (1998).

Equation 6.62) for males, females and in aggregate compared with a logistic model of mortality (described in Section 6.3). It is clear that overall mortality in the 1984– 89 models is substantially lighter than those given by the logistic model in all cases — though overall model mortality for females is considerably higher than that for males and females together or males separately. This is not altogether surprising, since it was noted in Section 4.2 that an apparent anomaly in the NLTCS surveys was that the 5-year probabilities of dying over the periods 1984–89 and 1989–94 were considerably less (1/2 - 1/3) than the 2-year probability of dying in the period 1982–84 — which in turn, may be a consequence of the classification 'Not in survey year' being used in the 1989 and 1994 NLTCS (and not in the 1984 NLTCS).

The reason that overall model mortality is so low becomes clear when it is decomposed into its separate components — the occupancy probabilities of each state (conditional on being alive) and the force of mortality for each non-dead state. For males, these are illustrated in Figures 6.63 and 6.64, for females in Figures 6.65 and 6.66 and for males and females combined in Figures 6.67 and 6.68. At younger



Figure 6.63: Model occupancy probabilities, conditional on being alive, for the 1984–89 (5) models (solid line) and the 1984–89 (10) models (dotted line), for males.

ages (< 90 years), overall model mortality is heavily influenced by the mortality from the healthy state (as the occupancy probability, conditional on being alive, of the healthy state starts at 1 at age 65 and drops to about 1/2 by age 85–90 years), which, for all three gender classifications, are very low and remain very low, compared with, for example, mortality from the healthy state in the previous section. At older ages (> 90 years), the dominent influence on overall mortality is (in general) the mortality of the institutionalised state, which for males and females combined and males alone, is very low and linearly increasing. For females, mortality in the institutionalised state is exponentially increasing, and the effect of this is clear in the exponentially increasing overall mortality for females — although it is still substantially below that of the logistic model.

With such low overall model mortality, some adjustments need to be made. The options for adjusting mortality in the models are either:

1. to increase mortality in each state in some way, such that overall mortality in the model is consistent with mortality from the logistic model (in the sense of least squares, or some similar criterion); or



Figure 6.64: The parametric force of mortality for the 1984–89 (5) models (solid line) and the 1984–89 (10) models (dotted line), for males.

2. to use the mortality transition intensities from the 1982–84 NLTCS.

The advantage of the first method above is that overall mortality in the model could be adjusted to be arbitarily close to that of the logistic fit. The disadvantage is that a number of choices need to be made about how mortality in each state is adjusted (as discussed in Section 6.3), some of which will necessarily be arbitrary (for example, whether mortality should be uniformly adjusted by disability level). The main advantage of using the mortality estimates from the 1982–84 NLTCS is that these are based on actual experience, and it may be expected that mortality experience by disability level would not change much between the periods 1982–84 and 1984–89. Further support for using the mortality estimates from the 1982–84 NLTCS is gained by comparing the occupancy probabilities, conditional on being alive, of the disability models from the 1982–84 and 1984–89 NLTCS, as they show the same patterns across survey periods:

1. the occupancy probabilities for the healthy state start at 1 (by assumption) and decrease to about 0.5 by age 85–90 years, then rapidly drop to zero;



Figure 6.65: Model occupancy probabilities, conditional on being alive, for the 1984–89 (5) models (solid line) and the 1984–89 (10) models (dotted line), for females.

- the occupancy probabilities for the disability states: IADL only, 1–2 ADLs,
  3–4 ADLs and 5–6 ADLs, follow the same trend they start at zero at age
  65, slowly increase up to ages 90–100 years, then decrease back to zero;
- 3. the occupancy probabilities for the institutionalised state start at zero, then steadily increase to 1 by age 120 years.

Although not all of the occupancy probabilities follow exactly the trends described above, the majority do, which suggests that overall mortality from the disability model using the mortality estimates from the 1982–84 NLTCS and otherwise parameterized using the 1984–89 NLTCS should be reasonably consistent with the logistic model (since overall model mortality using the 1982–84 NLTCS was).

This is confirmed in Figure 6.69, which compares overall mortality for the adjusted parametric transition intensities fitted to the 1984–89 NLTCS, with the mortality transition intensities from the 1982–84 NLTCS, (which I will refer to as the adjusted 1984–89 (5) models and the adjusted 1984–89 (10) models), with that of the logistic model. At younger ages, overall mortality for all the adjusted 1984–89



Figure 6.66: The parametric force of mortality for the 1984–89 (5) models (solid line) and the 1984–89 (10) models (dotted line), for females.



Figure 6.67: Model occupancy probabilities, conditional on being alive, for the 1984–89 (5) models (solid line) and the 1984–89 (10) models (dotted line), for males and females combined.



Figure 6.68: The parametric force of mortality for the 1984–89 (5) models (solid line) and the 1984–89 (10) models (dotted line), for males and females combined.

models is very close to that of the logistic model, but by age 90, they drop below those of the logistic model and, in general, the difference increases with increasing age.

The reason why overall mortality of the adjusted 1984–89 models is lower than that of the 1982–84 models (even though they both have the same mortality transition intensities) can be investigated by looking at the separate components of overall mortality — the occupancy probabilities of each state (conditional on being alive) and the force of mortality for each non-dead state. The force of mortality from each non-dead state was given in the previous section (as they are taken from the 1982–84 NLTCS), and the occupancy probabilities are given in Figures 6.70, 6.71 and 6.72, for males, females and in aggregate, respectively.

At younger ages, for both the 1982–84 models and the adjusted 1984–89 models, the state with the largest occupancy probability is the same, that for the Healthy state — it starts at one at age 65 and deceases to about 1/2 by age 90. So it is not surprising that, at younger ages, overall mortality in the adjusted 1984–89 models is very similar to that of the 1982–84 models. However, at older ages the patterns of



Figure 6.69: Overall model mortality for the adjusted 1984–89 (5) models and the adjusted 1984–89 (10) models compared with a logistic model of mortality by Thatcher, Kannisto & Vaupel (1998).

the occupancy probabilities are different in respect of the 1982–84 models and the adjusted 1984–89 models:

- the occupancy probabilities of the institutionalised state in the 1982–84 models, generally dominate (apart from for males, where the occupancy probability of the 1–2 ADL state dominates) at older ages (and often increases to almost 1 by age 120); whereas
- 2. for the adjusted 1984–89 models there is a strong tendency for the occupancy probabilities of the 3–4 ADLs state to monotonically increase with age, remaining at a similar level to those of the institutionalised state.

Now, looking at the parametric force of mortality from the 3–4 ADLs state in the 1982–84 models (Figures 6.57, 6.59 and 6.61), which are small and linearly increase slowly (and for one data set even decrease) with age, it is clear why the overall forces of mortality in the 1984–89 models drop below those of the 1982–84 models (and thus below that of the logistic model). So even using the same transition intensities



Figure 6.70: Model occupancy probabilities, conditional on being alive, for the adjusted 1984–89 (5) models (solid line) and the adjusted 1984–89 (10) models (dotted line), for males.

for the force of mortality from each non-dead state, there is no guarantee that the overall force of mortality of two models otherwise parameterized differently, will be similar. This also suggests that an important factor of overall model mortality is the force of disability and recovery within the model and not just the force of mortality from each state, especially at older ages — another factor to take into account when adjusting the level of overall model mortality.

Table 6.70 gives the survival probabilities (as percentages) for each of the adjusted 1984–89 models, assuming that a person is alive and in the healthy state at age 65. They are slightly higher than for the 1982–84 models — at age 65, they are about 1% higher, and the difference increases to about 3% by age 90 (apart from the adjusted 1984–89 (5) model for females, where the difference is almost 7%). The differences then decrease with age, which is expected since they are tending towards 0%. So, while the survival probabilities are higher for the adjusted 1984–89 models than the 1982–84 models, the probability of survival to 100 years is still low (less than 8%) and to 110 years it is less than 0.29%.



Figure 6.71: Model occupancy probabilities, conditional on being alive, for the adjusted 1984–89 (5) models (solid line) and the adjusted 1984–89 (10) models (dotted line), for females.

Table 6.70: Model survival probabilities for lives in the healthy state at age 65 calculated from the 1984–89 (10) models and the 1984–89 (5) models, for males, females and in aggregate (agg).

Survival probability from age 65 for the adjusted:							
	$1984-89 (10) \mod 100$			1984-89~(5) model			
Age	males	females	agg	males	females	agg	
	%	%	%	%	%	%	
70	84.117	94.164	89.679	87.948	94.742	91.442	
75	65.401	81.990	74.389	68.266	85.219	76.521	
80	45.366	64.740	55.947	46.287	70.884	57.916	
85	26.860	45.161	37.028	26.939	52.815	38.926	
90	12.844	26.733	20.439	13.176	33.952	22.630	
95	4.648	12.569	8.432	5.266	18.034	10.981	
100	1.195	4.248	2.274	1.651	7.564	4.259	
105	0.210	0.974	0.398	0.386	2.418	1.266	
110	0.026	0.164	0.048	0.064	0.587	0.283	

Should further adjustments be made to the 1984–89 models, to make overall model mortality more consistent with that from the logistic model? I would argue 'no' for the following reasons:

- as discussed in the previous section, many arbitrary choices would need to be made about how to adjust mortality;
- as discussed earlier in this section, the transition intensities between disability states themselves can have a large impact on overall mortality (especially at older ages), and so should also be considered when adjusting overall model mortality;
- 3. furthermore, mortality improvement (decreasing force of mortality) over time is a well documented observation (Macdonald *et al.* (1998)), and so it would be expected that overall mortality in the 1984–89 NLTCS should be lighter than in the 1982–84 NLTCS, on average — so some mortality improvement should be allowed for;
- 4. at younger ages, the overall force of mortality is reasonably consistent with that from the logistic model, and at older ages the transition intensities will be subject to sensitivity analyses (as described in the previous section).

So, given the adjustments already made to the 1984–89 models, and the points discussed above, I think that further adjustments to mortality (or otherwise) in the models are not justified.

In this section I have analysed and made some major adjustments to the 1984–89 models (replacing all the mortality transition intensities with those from the 1982–84 NLTCS). The adjustments made, improved the overall force of mortality for all of these models (closer to mortality from the logistic model), and while I have argued that this is a reasonable approach to use, there are many other ways that mortality could have been adjusted. With such major adjustments made, it would be difficult to justify attributing much weight to the absolute value of any application of these models, however, they will be very useful in providing estimates to compare with estimates from the 1982–84 models. Again, I give the same caution as for the 1982–84 models — for use in an insurance application, careful consideration would need



Figure 6.72: Model occupancy probabilities, conditional on being alive, for the adjusted 1984–89 (5) models (solid line) and the adjusted 1984–89 (10) models (dotted line), for males and females combined.

to be made as to how models could be adjusted appropriately.

# 6.5 Overall Model Mortality, Using the Fitted Parametric Transition Intensities from the 1989–94 NLTCS

In this section I compare the overall force of mortality in the disability models using the parametric transition intensities fitted to the 1989–94 NLTCS (which I will refer to, using the notation of the Section 6.3, as the 1989–94 (10) models and the 1989–94 (5) models, depending on the age bands used to group the original data) with the same benchmark force of mortality as in Section 6.3. Figure 6.73 shows the overall model force of mortality in the 1989–94 models (as defined in Equation 6.62) for males, females and combined compared with a logistic model of mortality (described in Section 6.3).


Figure 6.73: Overall model mortality for the 1989–94 (5) models and the 1989–94 (10) models compared with a logistic model of mortality by Thatcher, Kannisto & Vaupel (1998).

The overall forces of model mortality in the 1989–94 models (except for the anomalous 1989–94 (10) model for males) are very similar those of the 1984–89 models, and the general comments given in the previous section, on why they are substantially lighter than those given by the logistic model, apply equally.

The reason that overall model mortality is so low becomes clear when it is decomposed it into its separate components — the occupancy probabilities of each state (conditional on being alive) and the force of mortality for each non-dead state. For males, these are illustrated in Figures 6.74 and 6.75, for females in Figures 6.76 and 6.77 and for males and females combined in Figures 6.78 and 6.79. For all of the sets of parametric transition intensities (except those for the 1989–94 (10) model for males) the reason why overall mortality is so low is clear — the force of mortality from every state is very low, so overall mortality must also be low. Even though the forces of mortality are very low, it is worth noting that the patterns of the occupancy probabilities are very similar to those of the 1982–84 models.

For the 1989–94 (10) model for males, overall mortality behaves unexpectedly:



Figure 6.74: Model occupancy probabilities, conditional on being alive, for the 1989–94 (5) models (solid line) and the 1989–94 (10) models (dotted line), for males.



Figure 6.75: The parametric force of mortality for the 1989–94 (5) models (solid line) and the 1989–94 (10) models (dotted line), for males.



Figure 6.76: Model occupancy probabilities, conditional on being alive, for the 1989–94 (5) models (solid line) and the 1989–94 (10) models (dotted line), for females.



Figure 6.77: The parametric force of mortality for the 1989–94 (5) models (solid line) and 1989–94 (10) models (dotted line), for females.

it remains low until about age 90, then it increases exponentially until about age 102, when it decreases rapidly for a few years and then remains almost constant up to age 120. Most of this can be explained by looking at the occupancy probability (conditional on being alive) and mortality of the instutionalized state. The occupancy probability of the institutionalised state starts at zero at age 65 and exponentially increases until age 90 and the force of mortality in this state is very low until age 90, when it starts to exponentially increase (which is a large element of overall mortality, since the occupancy probability is quite high at these ages), causing overall mortality to increase exponentially also. The force of mortality of the institutionalised state continues to increase rapidly, exceeding 1.0 by age 100, thus increasing the probability of dying from this state (relative to other states), and reducing its occupancy probability (conditional on being alive). Also, at about age 100, the occupancy probability of the IADL only state starts to increase rapidly, which has a very low force of mortality that linearly decreases with age (by age 100, it is zero and remains zero) — these effects combined act to reduce overall mortality. By about age 110, the only other two states that contribute significantly to overall mortality are the 1–2 ADLs state and the 3–4 ADLs state, both of which have fairly low and linearly increasing mortality, the combined effect of which is a levelling out of overall mortality.

It is clear, as with the 1984–89 models, that some adjustment to the forces of mortality in the 1989–94 models is necessary, to make overall model mortality more consistent with the benchmark mortality — that from the logistic model. I will adjust the models in the same way and for the same reasons as the 1984–89 models were adjusted, by substituting the mortality transition intensities from the 1982–84 models (see previous section for more detail). Figure 6.80 compares overall mortality for the adjusted parametric transition intensities fitted to the 1989–94 NLTCS, with the mortality transition intensities from the 1982–84 NLTCS, (which I will refer to as the adjusted 1989–94 (5) models and the adjusted 1989–94 (10) models), with that of the logistic model.

At younger ages, overall mortality for all the adjusted 1989–94 models is very close to that of the logistic model, but by age 90, it drops below those of the logistic



Figure 6.78: Model occupancy probabilities, conditional on being alive, for the 1989–94 (5) models (solid line) and the 1989–94 (10) models (dotted line), for males and females combined.



Figure 6.79: The parametric force of mortality for the 1989–94 (5) models (solid line) and the 1989–94 (10) models (dotted line), for males and females combined.



Figure 6.80: Overall model mortality for the adjusted 1989–94 (5) models and the adjusted 1989–94 (10) models compared with a logistic model of mortality by Thatcher, Kannisto & Vaupel (1998).

model and, in general, the difference between the two increases with increasing age. The overall mortality for the adjusted 1989–94 models lies, in general, between those of the 1982–84 models and the adjusted 1984–89 models (see next section for comparison of overall mortality in the 1982–84, 1984–89 and 1989–94 models) and is very similar in shape to the 1982–84 models — except for the adjusted 1989–94 (10) model for males, which partly retains the shape of overall mortality from the unadjusted 1989–94 model.

The occupancy probabilities of each state (conditional on being alive) for the adjusted 1989–94 models are given in Figures 6.81, 6.82 and 6.83, for males, females and in aggregate, respectively. From Figures 6.82 and 6.83 the reason why overall model mortality, for the female and aggregate adjusted 1989–94 models, is similar to the 1982–84 models is clear – the occupancy probabilities are very similar (and the force of mortality from each state is identical). For males, the occupancy probabilities of the 1982–84 models and the adjusted 1989–94 models are similar at younger ages (65–90 years), but are very different for older ages (90–120 years). For



Figure 6.81: Model occupancy probabilities, conditional on being alive, for the adjusted 1989–94 (5) models (solid line) and the adjusted 1989–94 (5) models (dotted line), for males.

the 1982–84 males models, the occupancy probabilities for the 1–2 ADLs state and the institutionalised state dominate by age 100 and continue to increase with age, whereas:

- for the adjusted 1989–94 (10) model, by age 100 the occupancy probabilities for the IADL only state, the 3–4 ADLs state and the institutionalised state dominate, but only those for the IADL only state continue to increase with age — the other two decrease thereafter; and
- 2. for the adjusted 1989–94 (5) model, by age 100 the occupancy probabilities are spread out fairly evenly among all states, but only those for the 3–4 ADLs state and the institutionalised state continue to increase with age, clearly dominating by age 110.

The first point above explains why overall mortality for the adjusted 1989–94 (10) model for males levels off so suddenly at about age 100 — by this age the occupancy probability of the IADL state dominates and continues to increase with age, giving



Figure 6.82: Model occupancy probabilities, conditional on being alive, for the adjusted 1989–94 (5) models (solid line) and the adjusted 1989–94 (10) models (dotted line), for females.

increasing weight to the force of mortality from this state, which is very low and linearly increases (with a low gradient) with age, the effect of which is a levelling-off of overall mortality. The second point above indicates why overall mortality for the adjusted 1989–94 (5) model is below that of the 1982–84 model, since the force of mortality for the 3–4 ADLs state (which has a high occupancy probability at older ages in the adjusted 1989–94 model) is below that of the 1–2 ADL state (which has a high occupancy probability at older ages in the 1982–84 model) and increases at a slower rate — and the other state with high occupancy probability at older ages is the same for both models, the institutionalised state.

Table 6.71 gives the survival probabilities (as percentages) for each of the adjusted 1989–94 models, assuming that a person is alive and in the healthy state at age 65. They are very similar to those of the 1982–84 models (see Tables 6.69) — at all ages and for all models (except that for females with transition intensities fitted to the data in 5-year age bands) the survival probabilities do not differ by more than 2% (even for the male models) from the 1982–84 models and in general the



Figure 6.83: Model occupancy probabilities, conditional on being alive, for the adjusted 1989–94 (5) models (solid line) and the adjusted 1989–94 (10) models (dotted line), for males and females combined.

survival probabilities are slightly higher than for the 1982–84 models. Given that overall mortality for the 1989–94 models is closer to those given by the logistic fit than overall mortality in the 1984–89 models and that survival to older ages is low (less than 7% by age 100 and less than 0.4% by age 110), I argue as for the 1984–89 models — that further adjustments to mortality (or otherwise) in the models are not justified (see previous section). However, when I estimate the costs of disability in long-term care insurance in Section 7, I will look at the sensitivity of these costs to arbitrary adjustments in the force of mortality at older ages (95+ years), as well as isolating the model costs attributable to disabled lives over age 95.

In this section I have analysed and made some major adjustments to the 1989–94 models (replacing all the mortality transition intensities with those from the 1982–84 NLTCS). These improved the overall force of mortality for all of these models (making it closer to mortality from the logistic model), and while I have argued that this is a reasonable approach to use, there are many other ways that mortality could have been adjusted. With such major adjustments made, it would be difficult

Table 6.71: Model survival probabilities for lives in the healthy state at age 65 for the adjusted 1989–94 (5) models and the adjusted 1989–94 (10) models, for males, females and in aggregate (agg).

	Surviv	al probabi	ility from	n age 65 f	for the ad	justed:			
	1989-	–94 (10) n	nodel	1989	1989-94~(5) model				
Age	males	females	agg	males	females	agg			
	%	%	%	%	%	%			
70	83.455	93.485	88.897	87.716	94.316	90.772			
75	63.815	80.425	72.639	67.285	83.887	74.715			
80	43.384	62.626	53.741	44.695	68.691	55.342			
85	25.007	42.763	34.832	25.311	50.147	36.218			
90	11.338	24.372	18.590	11.925	31.458	20.342			
95	3.540	10.684	7.124	4.518	16.233	9.384			
100	0.638	3.231	1.479	1.319	6.519	3.350			
105	0.082	0.607	0.140	0.283	1.892	0.862			
110	0.011	0.056	0.007	0.043	0.362	0.148			

to justify attributing much weight to the absolute value of any application of these models, however, they will be very useful in providing estimates to compare with estimates from the 1982–84 models. Again, I give the same caution as for the 1982–84 and 1984–89 models — for use in an insurance application, careful consideration would need to be made as to how models could be adjusted appropriately.

## 6.6 Comparison of Overall Mortality in the 1982– 84, 1984–89 and 1989–94 Disability Models

The aim of this section is first to compare overall mortality from the 1982–84, the adjusted 1984–89 and the adjusted 1989–94 models and then to discuss their application to estimating the costs of disability in a long-term care insurance contract.

Figures 6.84 and 6.85 compare the overall forces of mortality of the 1982–84, the adjusted 1984–89 and the adjusted 1989–94 models, with the transition intensities fitted to the data in 10-year age bands and 5-year age bands, respectively. The force of mortality from the logistic model is also shown for comparison. A few points are worth noting:

1. for all of the disability models, overall mortality is lighter than that given by

the logistic model, even for males and females combined (since the logistic fit here is for females, that for males and females would give a higher force of mortality);

- the overall forces of mortality for all models are very close, to each other and to that of the logistic model, up to (at least) ages 90–95 years, after which they tend to diverge;
- 3. in general, the overall forces of mortality of the 1982–84 models are the highest, followed by those of the adjusted 1989–94 models and then those of the adjusted 1984–89 models (the notable exception to this is the overall force of mortality for the adjusted 1989–94 (5) model for males and females combined, which is exceptionally high);
- 4. there is less difference, in general, between the overall force of mortality for the models fitted to the data in 5-year age bands across survey periods, than for those fitted to the data in 10-year age bands.

The first point above suggests that any costs of disability estimated using these models may be overstated, as mortality is understated and lives, in the models, will be living longer than expected, increasing the estimates of the costs of disability — this can, however, be seen to be an implicit allowance for improving mortality over time. This will especially be the case at older ages, where the differences between overall mortality in the models and that of the logistic model are the greatest. This, together with the second observation above, suggests that more reliance should be given to applications of the model to younger ages (< 90 years) and less to older ages. For example, when estimating the costs of disability, sensitivity of the costs to the mortality assumptions at older ages could be carried out in a similar manner to that in the Alzheimer's model (see Section 2.5):

- 1. to estimate an upper bound on the costs, assume that after age 95 mortality remains constant at the level it was at age 95, in all states; and
- 2. to estimate a lower bound on the costs, assume that after age 95, mortality continues to increase exponentially, in all states;



Figure 6.84: Overall mortality from the 1982–84 (10), the adjusted 1984–89 (10) and the adjusted 1989–94 (10) models compared with a logistic model of mortality by Thatcher, Kannisto, & Vaupel, (1998).



Figure 6.85: Overall mortality from the 1982–84 (5), the adjusted 1984–89 (5) and the adjusted 1989–94 (5) models compared with a logistic model of mortality by Thatcher, Kannisto, & Vaupel (1998).

In Section 6.3, I looked at the overall force of mortality in the 1982–84 models and compared them to a logistic model of mortality that was parameterized using a very large dataset. I then argued that, while further adjustments could be made to mortality in the disability models (to make them closer to that of the logistic model), it would be difficult to justify as a number of choices would need to be made, some of which would necessarily be arbitrary. Rather than making adjustments to the models, I suggested that sensitivity analyses should be carried out for any application of the models — especially for any application of the model to older ages (> 90 years). The justification for this is two-fold:

- the parametric transition intensities were fitted to point estimates that had been assigned to single ages, the oldest of which is just over 90 years (91 years for the 1982–84 models and 92.5 years for the 1984–89 and 1989–94 models), so after this age the transition intensities are extrapolations; and
- at younger ages (< 95 years), overall mortality in the models is fairly consistent with that of the benchmark force of mortality (from the logistic model), but it becomes less consistent at older ages.

Then in Sections 6.4 and 6.5, I looked at the overall force of mortality in the 1984–89 and 1989–94 models, respectively, and compared them to the same logistic model of mortality as in Section 6.3. For both sets of models, overall mortality was very low, which may be due to the addition of an extra classification in the original data all lives in the 1982 and 1984 NLTCS were accounted for, but in the 1989 and 1994 NLTCS a new classification, 'Not in Survey Year' was introduced, leaving 9,524 lives unaccounted for in the 1989 NLTCS and 12,730 lives unaccounted for in the 1994 NLTCS. In answer to the question posed in Section 3.4, 'Is there any evidence in the data to suggest what may have happened to these lives?', I would argue that there is strong evidence that suggests a substantial proportion of these lives unaccounted for, were unaccounted for because of death — which would justify adjusting mortality in these models. I outlined various methods for adjusting mortality in the models to make them more consistent with the benchmark force of mortality (from the logistic model) and argued the case that the forces of mortality from each non-dead state in the models should be replaced with those from the 1982–84 models, which I call the adjusted 1984–89 and the adjusted 1989–94 models. Overall mortality in these adjusted models is much more consistent with the logistic model and I argued that further adjustments were not justified, but rather sensitivity analyses should be carried out on any application of the models.

In the next chapter, I will use these parameterized models to estimate the costs of disability in a long-term care contract. Of the three sets of models (the 1982–84, the adjusted 1984–89 and the adjusted 1989–94), I place most reliance on the results of the 1982–84 models, even though it is oldest data set since:

- 1. it is based on the largest data set (see Section 3.4) and all lives were accounted for in this survey period;
- 2. overall mortality from these models is fairly consistent with the benchmark force of mortality; and
- no adjustments were made to these models (unlike the adjusted 1984–89 and the adjusted 1989–94 models)

For these reasons, when estimating costs in the next chapter, I will focus on the results from the 1982–84 models. I will also use the adjusted 1984–89 and adjusted 1989–94 models, but the results from these will mainly be for comparison with the results from the 1982–84 models — they will also be useful in estimating ranges of feasible costs.

### Chapter 7

# Model Costs of Disability and Adverse Selection in Long-Term Care Insurance Revisited

#### 7.1 Introduction

In Chapter 2, in order to look at the potential costs of adverse selection arising from variants of the APOE gene, the cost of other events in the ageing process (mainly disability) that trigger benefits were very simply assumed to be a multiple of those costs arising from Alzheimer's disease.

The aim of this chapter is to use the model described in Chapter 3, and parameterized in Chapters 4 to 6 to estimate (independently from, but including Alzheimer's disease) the costs of disability in long-term care insurance. These estimates can then be combined with estimates of the genotype-specific costs of Alzheimer's disease, given in Chapter 2, to estimate the potential costs of adverse selection without having to make such a simplifying assumption. This will provide a better basis for estimating the potential for adverse selection, as well as providing an independent check on how reasonable the assumptions were in Chapter 2.

In Section 7.2, I use the disability model to estimate the costs of disability in a long-term care contract and look at trends of the costs by gender and age at entry — also providing a first check of the assumptions made in Chapter 2. Then in Section 7.3, I look at the sensitivity of the costs of disability to some of the model assumptions. Using the genotype-specific costs of Alzheimer's disease (from Chapter 2), in Section 7.4, I look at the potential costs of adverse selection arising from variants of the APOE gene. Finally, I provide conclusions in Section 7.5.

The notation introduced in the previous section I will continue to use (slightly adjusted for brevity) in this section and it is as follows: the disability models using the parametric transition intensities fitted to the 1982–84 NLTCS, I will refer to as the 1982–84 models; those using the parametric transition intensities fitted to the 1984–89 NLTCS and 1989–94 NLTCS, adjusted for mortality (see Section 6.4 and 6.5), I will simply refer to as the 1984–89 models and 1989–94 models, respectively. For each pair of NLTCS and for each gender classification, two sets of parametric transition intensities were fitted: those with the data grouped into 10-year age bands; and those with the data grouped into 5-year age bands. For brevity and to avoid confusion I will refer to the former as the '1982–84 (10) models' and the latter as the '1982–84 (5)' models, for the 1982–84 NLTCS, and I will use similar notation for the 1984–89 models and the 1989–94 models.

Of the three sets of models (the 1982–84, the 1984–89 and the 1989–94 models), I will place most reliance on the results of the 1982–84 models, even though it is from the oldest data set since:

- 1. it is based on the largest data set (see Section 3.4) and all lives were accounted for in this survey period;
- 2. overall mortality from these models is fairly consistent with the benchmark force of mortality (see Section 6.3); and
- no adjustments were made to these models (unlike the 1984–89 and the 1989– 94 models)

Furthermore, using 10 year age groups (providing 3 data points for graduation) caused the fitting procedure to be unstable in some cases as the trend of the transition intensities with age was not clear (see Section 5.5). When using 5 year age bands (providing 5 data points for graduation), this instability in the fitting procedure did not occur, as there were enough estimates to use a more robust method (weighted least squares) when fitting a Makeham curve. Given that the process of graduating the transition intensities with data in 5 year age groups used a more stable fitting procedure, I will place more reliance on the graduated transition intensities that used data grouped into 5 year age bands.

For the reasons outline above, the models that I will concentrate on in this chapter are the 1982–84 (5) models — I will also use the other models, but the results from these will mainly be for comparison with the results from the 1982–84 (5) models and to help estimate ranges of feasible costs.

## 7.2 The Costs of Disability in Long-Term Care Insurance

The disability model was specified in Chapter 3 and parameterized in Chapters 4 to 6 and is shown in Figure 3.19. (The states in the model I denote as follows: Healthy — state 1, IADL only — state 2, 1–2 ADLs — state 3, 3–4 ADLs — state 4, 5–6 ADLs — state 5, instutionalized — state 6 and dead — state 7.) The notation I use is defined in Sections 1.3.1 and 2.3.

I assume that the LTC contract has a single premium paid at outset, so the only policy cash-flows thereafter are the benefits. Benefits are payable continuously, while in a claiming state — which, as discussed in Section 3.2, are the states that represent the loss of 3 or more ADLs (states 4, 5 and 6).

The quantum of benefit is £1 per annum at inception of the policy, increasing continuously at rate  $\delta_b$  per year. This is represented in the model by  $b_t^j = 1$  if j = 4, 5or 6 (level benefit) or by  $b_t^j = e^{\delta_b t}$  if j = 4, 5 or 6 (benefit escalating at rate  $\delta_b$  per year). The latter is the default, since this is a feature of most LTC policies (Dullaway & Elliot, 1998). I only use level benefits in some sensitivity tests in the next section — in this section I assume that  $\delta_b = 0.05$ , representing indexation to earnings. I use a force of interest  $\delta = 0.05$  throughout. The present value of the benefit is the random variable whose expected value and higher moments are obtained by solving Norberg's equations (see Section 2.3), the expected value being the relevant quantity for use in the traditional actuarial equation of value. I look at policies with inception ages of 60, 65, 70 and 75. Note that the data used to parameterize the disability models started from age 65, so using the parametric transition intensities back to age 60 is extrapolation — I did this to allow comparison with the Alzheimer's disease model which was parameterized from age 60.

Tables 7.72 to 7.74 give the first 3 moments of the present value of benefits using the 1982–84 models for lives entering at ages 60, 65, 70 and 75 in each state in the model, for males, females and both combined, respectively. General trends accross all three tables are:

- 1. the EPV of benefits increases with lives entering with increasing disability level, even for the non-claiming disability states (states 2 and 3), indicating that disability may be a predictor of further disability;
- the increase in the EPV of benefits is the greatest between lives starting in state 3 and those starting in state 4, which is as expected since lives starting in state 4 immediately start claiming benefits;
- the variance of the present value of benefits also tends to increase with lives entering with increasing disability level, indicating greater uncertainty for the costs of these lives;
- 4. the variance of the present value of benefits generally decreases for lives starting at older ages, which is also as expected since these lives have a shorter expected future life time and so there is less uncertainty of their costs;
- 5. the variance and skewness of the present value of benefits are much greater for females than for males, which may be from a combination of females' longer lives and their tendency to suffer prolonged disability (as opposed to males' quick decline).

It is worth noting here that the models for males and females combined were parameterized using aggregate (across genders) data and are not an aggregation of the separate models for males and females — so there is no guarantee that the EPV of benefits for males and females combined need be consistent with those for males and females calculated separately. Even so, there is almost complete consistency

Table 7.72: Mean, variance and skewness (q = 1, 2 and 3) of the present value of disability claims costs for a life starting in each model state, unit benefit increasing continuously  $(\delta_b = 0.05)$ , for males using the 1982–84 models.

				Stat	te at star	t of cont	ract	
	Entry			IADL	1 - 2	3 - 4	5 - 6	
Model	age	q	Healthy	only	ADLs	ADLs	ADLs	Inst'd
1982 - 84(5)	60	1	1.2062	1.3187	1.8603	2.3550	3.2400	7.0200
		2	5.492	6.221	8.238	8.347	11.801	23.512
		3	36.415	44.099	58.941	58.094	83.316	107.723
	65	1	1.1267	1.3340	1.8979	2.6738	3.1980	5.2636
		2	5.032	6.114	8.050	8.683	10.431	16.960
		3	32.060	40.618	51.805	52.972	62.795	80.390
	70	1	1.0726	1.3992	1.8939	2.9764	3.1409	4.1032
		2	4.545	6.027	7.625	8.962	9.406	12.037
		3	27.083	36.638	44.326	47.896	49.891	55.573
	75	1	1.0364	1.4598	1.8007	3.1012	3.0060	3.2959
		2	4.060	5.668	6.750	8.412	8.225	8.585
		3	22.165	30.637	35.433	38.952	38.886	37.139
1982–84 (10)	60	1	1.0627	1.2248	1.9148	2.4599	2.8785	3.0011
		2	4.630	5.383	7.373	7.553	8.472	9.201
		3	29.591	35.499	46.272	45.545	49.936	54.876
	65	1	1.0691	1.3166	1.9907	2.7708	3.0161	3.5940
		2	4.536	5.698	7.811	8.370	9.036	11.414
		3	27.821	36.216	47.247	47.665	51.192	64.168
	70	1	1.0624	1.4041	1.9316	2.9328	3.0124	3.5538
		2	4.274	5.739	7.327	8.356	8.637	10.536
		3	24.550	33.718	40.954	42.988	44.704	52.869
	75	1	1.0270	1.4639	1.7813	2.9449	2.8841	3.1837
		2	3.812	5.408	6.342	7.686	7.618	8.449
		3	19.991	28.275	32.184	34.870	35.208	37.957

between EPV of the benefits between gender classification, for all starting states, in that the EPV of benefits is greatest for females, then for males and females combined, and the lowest, that for males (the only exception being for the 1982–94 (10) model for males, where the EPV of benefits for lives entering at age 60 in the 3–4 ADL state is greater than that for males and females combined).

In Table 7.75, the EPV of benefits for lives entering at ages 60, 65, 70 and 75 in the healthy state only (which is of primary interest) are given for all models and gender classifications. It is again noticeable that there is consistency of the EPV of benefits between gender classifications, as discussed above. In fact, for any given starting age, across all 6 models:

				$\operatorname{Sta}$	ate at star	t of contra	act	
	Entry			IADL	1 - 2	3 - 4	5-6	
Model	age	q	Healthy	only	ADLs	ADLs	ADLs	Inst'd
1982 - 84(5)	60	1	2.5600	3.2462	3.1062	4.6016	4.8299	10.2584
		2	13.647	17.632	17.716	21.081	23.172	41.379
		3	99.436	139.430	144.930	165.939	195.143	214.538
	65	1	2.5982	3.1725	3.2607	4.9122	4.8583	7.9417
		2	13.535	17.382	18.200	21.582	22.549	32.683
		3	95.913	131.516	139.762	154.083	171.723	182.072
	70	1	2.6052	3.0869	3.3515	5.0287	4.6986	6.2800
		2	12.931	15.928	17.226	19.940	20.056	24.717
		3	86.859	110.550	118.982	125.560	136.238	141.989
	75	1	2.5433	2.9698	3.3136	4.9506	4.4228	5.0890
		2	11.749	13.971	15.262	17.367	17.102	18.495
		3	73.392	87.994	94.143	97.225	104.500	104.726
1982 - 84 (10)	60	1	2.6288	3.0560	3.0353	4.5973	4.6616	5.2376
		2	13.433	16.134	16.608	19.544	20.504	24.788
		3	94.979	122.832	130.840	144.753	160.218	199.394
	65	1	2.5676	3.0796	3.1872	4.8115	4.6264	5.7377
		2	12.959	16.244	17.112	19.878	20.370	26.004
		3	89.492	120.051	128.784	138.173	151.875	185.646
	70	1	2.5258	3.0485	3.2652	4.9041	4.4776	5.4624
		2	12.234	15.291	16.376	18.811	18.727	22.542
		3	80.134	104.476	112.049	117.788	127.589	143.519
	75	1	2.4749	2.9406	3.2335	4.8132	4.2283	4.8681
		2	11.159	13.535	14.635	16.575	16.229	18.035
		3	67.387	83.363	88.875	91.993	99.241	104.133

Table 7.73: Mean, variance and skewness (q = 1, 2 and 3) of the present value of disability claims costs for a life starting in each model state, unit benefit increasing continuously  $(\delta_b = 0.05)$ , for females using the 1982–84 models.

				$\operatorname{St}$	ate at star	rt of contr	act	
	Entry			IADL	1 - 2	3 - 4	5-6	
Model	age	q	Healthy	only	ADLs	ADLs	ADLs	Inst'd
1982 - 84(5)	60	1	1.9986	2.1463	2.5246	3.6596	3.8504	7.2711
		2	10.399	11.496	13.155	14.874	16.289	31.180
		3	76.511	90.090	104.908	114.024	131.132	196.274
	65	1	1.9526	2.2783	2.6692	3.9193	3.9298	6.2260
		2	10.018	12.084	13.734	15.608	16.240	25.009
		3	71.879	91.554	103.526	110.932	120.269	150.968
	70	1	1.9397	2.3823	2.7396	4.0996	3.8970	5.2410
		2	9.517	11.878	13.248	15.209	15.117	19.492
		3	64.646	82.764	90.761	96.534	100.925	113.449
	75	1	1.9451	2.4165	2.7322	4.1575	3.7682	4.4207
		2	8.885	10.983	12.095	13.928	13.465	15.031
		3	55.846	68.967	74.400	78.203	80.904	83.364
1982–84 (10)	60	1	1.8667	1.8998	2.2284	2.7237	3.2374	4.2109
		2	9.456	9.907	11.184	11.097	12.305	17.620
		3	68.072	75.578	86.939	81.742	93.720	135.053
	65	1	1.8944	2.1407	2.5580	3.5677	3.5823	4.7174
		2	9.385	11.020	12.723	13.804	14.126	19.385
		3	65.285	81.676	94.184	96.547	103.448	135.465
	70	1	1.9136	2.3247	2.7160	4.0141	3.7462	4.6496
		2	9.032	11.314	12.843	14.579	14.341	17.720
		3	59.015	77.226	86.424	91.227	95.330	110.654
	75	1	1.9099	2.4049	2.7251	4.1289	3.7444	4.2679
		2	8.331	10.608	11.780	13.590	13.225	14.714
		3	49.692	64.108	69.966	73.837	77.154	82.350

Table 7.74: Mean, variance and skewness (q = 1, 2 and 3) of the present value of disability claims costs for a life starting in each model state, unit benefit increasing continuously  $(\delta_b = 0.05)$ , for males and females combined using the 1982–84 models.

Table 7.75: Expected present value of disability claims costs for a life starting in the healty state, unit benefit increasing continuously ( $\delta_b = 0.05$ ), for males, females and combined using all disability models.

				EPV	using:		
	Entry	1982 - 84	models	1984 - 89	models	1989-94	models
Gender	age	(5)	(10)	(5)	(10)	(5)	(10)
M & F	60	1.9986	1.8667	1.7891	1.8678	1.9441	2.0039
	65	1.9526	1.8944	1.7549	1.8488	1.8330	1.9404
	70	1.9397	1.9136	1.7524	1.8569	1.7744	1.8975
	75	1.9451	1.9099	1.7708	1.8579	1.7439	1.8603
$\mathbf{F}$	60	2.5600	2.6288	2.7357	2.8789	2.7668	2.9927
	65	2.5982	2.5676	2.6377	2.6784	2.6478	2.7604
	70	2.6052	2.5258	2.5612	2.5365	2.5159	2.5894
	75	2.5433	2.4749	2.4661	2.4407	2.3770	2.4677
Μ	60	1.2062	1.0627	0.9624	0.8855	1.0680	1.0593
	65	1.1267	1.0691	0.8824	0.9339	1.0059	1.0500
	70	1.0726	1.0624	0.8848	1.0005	0.9577	1.0381
	75	1.0364	1.0270	0.9080	1.0415	0.9285	1.0189

- the lowest EPV of benefits for females is higher than the highest for males and females combined; and
- 2. the lowest EPV of benefits for males and females combined is higher than the highest for males alone.

Given that the three sets of models (1982–84, 1984–89 and 1989–94) were parameterized using different data sets (with the exception of the forces of mortality in the 1984–89 and 1989–94 models, which use the forces of mortality from the 1982–84 models), the estimates of the EPV of benefits are reasonably consistent between the models. For example, for males and females combined, the greatest difference between the maximum and minimum estimate of the EPV of benefits (as a percentage of the mid-point), for all starting ages is 11.33% — for females it is 15.59% and for males it is 30.66%.

There is no clear pattern of the EPV of benefits with starting age. For many cases, the EPV of benefits is almost constant across starting ages. This may be caused by, with increasing starting age, shorter expected future lifetime reducing expected costs, being compensated for by shorter expected time to disability increasing costs.

It was assumed in Chapter 2 (Section 2.6.3) that the costs arising from

Table 7.76: Comparison of the EPV of disability claims costs (total LTC costs) with the EPV of Alzheimer's disease (AD) claims costs for a life starting in the healty state, unit benefit increasing continuously ( $\delta_b = 0.05$ ), for females.

		EPV of:		
Entry	AD claims	Total L <sub>1</sub>	TC costs	AD claims costs as a
age	$costs^{(1)}$	$\operatorname{Highest}^{(2)}$	$Lowest^{(2)}$	% of total LTC costs
60	1.1534	2.9927	2.5600	38.54 - 45.05
65	1.1634	2.7604	2.5676	42.15 - 45.31
70	1.1739	2.6052	2.5159	45.06 - 46.66
75	1.1738	2.5433	2.3770	46.15 - 49.38

(1) From Table 2.10.

(2) Highest and lowest estimates from the 6 disability models in Table 7.75.

Alzheimer's disease (AD) were between 40% and 50% of total long-term care costs. The values in Table 7.75 are estimates of total long-term care costs and can be compared with the costs of AD, given in Table 2.10, to check the reasonableness of this assumption. Table 7.76 compares the costs arising from AD with total longterm care costs for females, since the AD claims costs in Table 2.10 were calculated using female mortality (Makeham approximation to 65% AF80, since the majority of elderly people will be women) and aggregate incidence of AD.

Table 7.76 gives strong support of the assumption made in Chapter 2 that the costs arising from Alzheimer's disease (AD) are indeed between 40% and 50% of total long-term care costs — the costs arising from the model of Alzheimer's disease as a percentage of those arising from the disability models are between 38.54% and 49.38%. However, it is not clear whether the two models are comparable in terms of the mortality assumptions. Figure 7.86 shows that overall mortality (as defined in Section 6.1) in the Alzheimer's disease model is fairly consistent with overall mortality in the disability models. This figure shows:

- overall mortality for the 1982–84(10) model for females (which, for females, has the highest overall mortality of the 6 models);
- 2. overall mortality for the 1984–89 (5) model for females (which, for females, has the lowest overall morality of the 6 models); and
- overall mortality for the Alzheimer's disease model using 65% AF80 as baseline mortality (see Section 2.4 for more detail).

Overall mortality in the Alzheimer's disease model is very close to that of the 1984– 89 (5) model (representing a lower bound on overall mortality in the disability models for females) until just after age 100. Overall mortality in the disability models then starts to level off with age, whereas in the Alzheimer's disease model it continue to grow exponentially, becoming greater than overall mortality in the 1982– 84 (10) model (representing an upper bound on mortality in the disability models for females) by just after age 110. Baseline mortality in the Alzheimer's disease model was taken as a Makeham approximation to 65% AF80, and this results in overall mortality in the model that is nearly exponential — which, as illustrated in Section 6.2, may not be appropriate at older ages. Overall, however, it seems reasonable to compare the EPV of benefits from the Alzheimer's disease model with those from the disability models.

Care is required in interpreting the effect on the EPV of benefits from any of the disability models from their overall force of mortality. It may be expected that a model with a high overall force of mortality would give lower estimates of the EPV of benefits (using the argument that if people are expected to die sooner then the costs will be lower). However, this argument does not always hold — the EPVs of benefits from the 1984–89 (5) model for females, which has the highest overall mortality, are not the lowest for entry at any age and, moreover, for entry at age 75, it gives a higher estimate than does the 1982–84 (10) model, which has the lowest overall force of mortality. Though, of course, adjusting the mortality in any given model will affect the EPVs of benefits in the way described above.

The EPV of benefits given in Table 7.75 can be analysed in more detail, by looking at the EPV of benefits attributable to a given time period of the contract and specific claiming state. For example, using the 1982–84 (5) model, the EPV of benefits for a females aged 60 is 2.5600, but how much of this is from claiming in the 3–4 ADL state in the time period 0 to 5? (i.e. between the ages 60 and 65). This can easily be calculated in the model by setting:

$$b_{60+t}^{i} = \begin{cases} e^{0.05t} & \text{if } i = 4 \text{ and } 0 \le t \le 5\\ 0 & \text{otherwise} \end{cases}$$
(7.69)

This can be done for the other claiming states and for any given time period in a



Figure 7.86: Comparison of the overall force of mortality in the disability models to the overall force of mortality in the Alzheimer's disease model, for females.

similar manner. For illustration, in Table 7.77, I look at the EPVs of benefits at age 60 from the 1982–84(5) model for males and females combined and separately, in 5-year time periods (upto age 90 and for 90+ years) and for each of the claiming states — the total (across all claiming states) for each time period and the total (across all time periods) for each claiming state are also given. It is worth noting some of the general trends from this table:

- over all time periods, the EPV of the benefits paid for the institutionalised state is greater than for the 5–6 ADLs state, which in turn, is greater than for the 3–4 ADLs state for all gender classifications;
- 2. for early time periods (0–5 years and 5–10 years), the EPV of benefits is greatest while in the 3–4 ADLs state or the 5–6 ADLs state, while after this period, the EPV of benefits is greatest for the institutionalised state, possibly as disability becomes more likely with age (and all lives are assumed to be healthy at age 60); and
- 3. the EPV of benefits increases with time periods that cover older ages up to

a maximum and then decrease — for males this maximum is the time period 15–20 years, whereas for females and both genders combined it is the time period 20–25 years (it may be expected that the EPV of benefits for females continues to increase at older ages than males, as disability becomes more likely with age, and females are expected to live longer than males)

So, although the total EPVs of benefits vary greatly by gender classification (for females, it is over double that of males), the patterns by claiming state and time period show consistency between gender classifications.

In this section I have used the disability model to estimate the costs of disability in a long-term care contract and have looked at trends of the costs by gender, age at entry, claiming state and by time period in the contract. I have also used these costs of disability to check the crude assumption made in Chapter 2, that the costs of Alzheimer's disease make up 40–50% of total long-term care costs and showed that the two models are comparable and that the disability model produces results that are in good agreement with this assumption. In the next section I look at the sensitivity of these costs of disability to some of the model assumptions.

#### 7.3 Sensitivity Analysis

In this section I look at the sensitivity of the model costs of disability to some of the key assumptions, namely:

- 1. the rate of benefit increase I have so far assumed that benefits increase at a rate of  $\delta_b = 0.05$ , which I compare with  $\delta_b = 0.025$  and level benefits;
- 2. the effect of benefits commencing at a lesser disability level; and
- 3. the mortality assumptions at older ages (> 90 years).

I assume throughout that the discount rate is  $\delta = 0.05$ . The effect of the above changes are reported as percentage changes to the EPVs in Tables 7.75, which I refer to as the baseline EPVs. For the baseline EPVs:

1. the discount rate,  $\delta = 0.05$ ;

Table 7.77: Expected present value of disability claims costs, by time period and claiming state, for a life starting in the healty state at age 60, unit benefit increasing continuously ( $\delta_b = 0.05$ ), for males, females and combined using the 1982-84 (5) models.

		EP	V of claim	ns from sta	ate:
	Time	3 - 4	5-6		All 3
Gender	period	ADLs	ADLs	Inst'd	States
M & F	0 - 5	0.01561	0.01720	0.00558	0.03838
	5 - 10	0.05313	0.04911	0.03787	0.14011
	10 - 15	0.08339	0.07773	0.10240	0.26352
	15 - 20	0.10122	0.09924	0.17764	0.37810
	20 - 25	0.10001	0.10529	0.22520	0.43050
	25 - 30	0.07807	0.08985	0.21242	0.38033
	30 +	0.06041	0.08527	0.22197	0.36765
	All	0.49183	0.52368	0.98307	1.9986
$\mathbf{F}$	0 - 5	0.00526	0.01719	0.00088	0.02334
	5 - 10	0.03247	0.03901	0.01913	0.09061
	10 - 15	0.08203	0.07024	0.10138	0.25364
	15 - 20	0.12512	0.10886	0.23305	0.46703
	20 - 25	0.13687	0.13184	0.33364	0.60235
	25 - 30	0.10870	0.11920	0.32915	0.55706
	30 +	0.08522	0.11375	0.36695	0.56592
	All	0.57566	0.60010	1.38418	2.5599
Μ	0 - 5	0.02915	0.01367	0.01069	0.05351
	5 - 10	0.05176	0.04928	0.04917	0.15022
	10 - 15	0.06286	0.07358	0.09125	0.22769
	15 - 20	0.06618	0.08308	0.11098	0.26025
	20 - 25	0.05844	0.07591	0.10144	0.23579
	25 - 30	0.04057	0.05408	0.07132	0.16596
	30 +	0.02589	0.03640	0.05047	0.11276
	All	0.33484	0.38601	0.48532	1.2062

- 2. unit benefits at policy inception increase continuously at a rate of  $\delta_b = 0.05$ , payable while in states 4, 5 and 6 (the 3–4 ADLs, 5–6 ADLs and institutionalised state, respectively); and
- 3. the parameters of the models are as given in Chapters 5 and 6 (the 1982–84 models and the adjusted 1984–89 and 1989–94 models, as then described).

It is a common policy feature for benefits to increase at some rate, as discussed in Section 2.2. I chose a rate of 0.05, to represent indexation to earnings and to be consistent with the assumption made in Chapter 2, when estimating the costs of Alzheimer's disease in a LTC contract. In Tables 7.78 and 7.79 I look at the EPVs of benefits increasing continuously at  $\delta_b = 0.025$  and level benefits ( $\delta_b = 0.0$ ), respectively, as a percentage of the EPVs of benefits with  $\delta_b = 0.05$ . From these tables:

- 1. it is clear that the rate of benefit escalation has a large impact on the EPV of benefits: with  $\delta_b = 0.025$ , the EPVs of benefits are between 55% and 78% of those with  $\delta_b = 0.05$ , and with  $\delta_b = 0.0$ , they are between 32% and 62%;
- 2. the effect of the rate of benefit escalation is much greater at younger entry ages: with  $\delta_b = 0.025$  compared to  $\delta_b = 0.05$ , for every 5-year period that entry into insurance is delayed, the difference in the EPV of benefits is reduced by approximately 5% (this effect is slightly greater with  $\delta_b = 0.0$  compared with  $\delta_b = 0.05$ );
- 3. the rate of benefit escalation has a greater impact on EPV of benefits for females than for males (between 5% and 10% difference).

The third point above may be expected, considering the timing of benefit payments given in Table 7.77 — the benefit payments to females peak at older ages than for males, which are then discounted over a longer time period, producing a greater effect for a given difference between discount rate and rate of benefit escalation. The effect on males and females combined for a change in the rate of benefit escalation is closer to that of females than males (as expected, since females made up the majority of the aggregate data).

Table 7.78: EPV of disability claims costs with benefits increasing continuously ( $\delta_b = 0.025$ ) as a percentage of EPV of disability claims costs with benefits increasing continuously ( $\delta_b = 0.05$ , baseline) for a life starting in the healthy state for males, females and combined using all disability models.

		$\mathbf{E}$	PV of ben	efits wit	h $\delta_b = 0.02$	25 as a $\%$	% of
			EPV of	benefits	with $\delta_b =$	0.05 for:	
	Entry	1982 - 8	4 models	1984 - 8	9 models	1989 - 9	4 models
Gender	age	(5)	(10)	(5)	(10)	(5)	(10)
		%	%	%	%	%	%
M & F	60	59.05	59.51	55.93	57.02	59.36	59.85
	65	64.82	65.14	61.69	61.83	64.21	64.23
	70	70.01	70.34	66.82	66.89	68.84	68.75
	75	74.63	75.11	71.32	71.56	73.17	73.32
$\mathbf{F}$	60	56.42	57.37	55.81	56.72	57.68	58.73
	65	62.92	63.25	60.67	61.11	62.65	62.97
	70	68.92	68.84	65.66	65.74	67.38	67.54
	75	73.91	73.79	70.21	70.19	71.82	72.15
Μ	60	64.25	65.36	60.82	58.67	62.26	63.65
	65	69.43	69.60	64.83	63.58	67.75	67.54
	70	73.91	73.89	69.89	69.08	72.05	71.45
	75	77.79	77.79	74.21	73.79	75.55	75.21

Adjusting the rate of benefit escalation has a similar effect on the EPV of benefits in both the disability model and the Alzheimer's disease model (see Section 2.5), though overall the EPV of benefits in the Alzheimer's disease model seem slightly more sensitive.

It has been assumed so far that benefits are payable while in states 4, 5 and 6 (3–4 ADLs, 5–6 ADLs and institutionalised). For some LTC contracts there is the option of benefits commencing earlier than this — for example on the failure of 2 ADLs, half or all of the annual sum assured may be payable. To look at the extra cost of a proportion of benefits commencing on the loss of exactly 2 ADLs, the disability model would need to be redefined (with separate states for loss of 1 ADL and loss of 2 ADLs) and then parameterized again. Instead, I look the the extra costs of benefits commencing in state 3 (1–2 ADLs), which can easily be estimated by setting  $b_{x+t}^3 = 1/2$  or 1 reflecting a contract that pays half or all of the annual sum assured, respectively, on the loss of 1 or 2 ADLS. Tables 7.80 and 7.81 give the EPV of disability claims costs with additional benefits of 1/2 and 1 increasing continuously ( $\delta_b = 0.05$ ) in state 3, respectively, as a percentage of EPV of disability claims costs

Table 7.79: EPV of disability claims costs with level benefits as a percentage of EPV of disability claims costs with benefits increasing continuously ( $\delta_b = 0.05$ , baseline) for a life starting in the healthy state for males, females and combined using all disability models.

		I	EPV of lev	vel benef	its as a $\%$	of EPV	of
			bene	efits with	$\delta_b = 0.05$	for:	
	Entry	1982 - 8	4 models	1984 - 8	9 models	1989 - 9	4 models
Gender	age	(5)	(10)	(5)	(10)	(5)	(10)
		%	%	%	%	%	%
M & F	60	36.46	37.05	32.93	34.35	37.11	37.76
	65	43.57	43.95	39.77	39.89	42.95	42.92
	70	50.41	50.78	46.27	46.24	48.91	48.67
	75	56.86	57.45	52.30	52.50	54.85	54.91
$\mathbf{F}$	60	33.13	34.42	33.00	34.16	35.04	36.41
	65	40.98	41.47	38.55	39.12	40.95	41.33
	70	48.85	48.74	44.72	44.80	46.97	47.07
	75	55.80	55.59	50.73	50.63	52.98	53.30
Μ	60	43.02	44.66	38.94	36.16	40.54	42.67
	65	49.77	50.05	43.68	42.00	47.67	47.44
	70	55.94	55.91	50.39	49.22	53.49	52.55
	75	61.57	61.54	56.44	55.77	58.41	57.75

with no benefits in state 3 (baseline). It is easily seen that the percentage increases in Table 7.81 (for unit benefits payable in state 3) are twice those in Table 7.80 (for 1/2 benefits payable in state 3). I include them both for illustration.

The increases in costs range from 136% to 174% for an extra unit benefit payable while in state 3. For females and both genders combined the extra costs are roughly 50%, slightly more at younger entry ages and less at older entry ages (except for females in the 1982–84 (10) model), though the difference between them is always less than 10%. For males, the increase in costs is slightly greater, between 60% and 70% (except in the 1989–94 (10) model) and there is no clear trend by entry age.

In Section 6.6, when looking at overall mortality in the disability models, it was noted that the oldest data point used for any of the models was just over 90 years and that after this age the transition intensities were extrapolations. Also, overall mortality in the disability models were very consistent with the benchmark force of mortality used up to about the same age (see Chapter 6 for more detail), after which they became less consistent. With this motivation, I aim to look at the sensitivity of the EPVs of benefits to the force of mortality after age 90. Table 7.80: EPV of disability claims costs with additional benefits of 1/2 increasing continuously ( $\delta_b = 0.05$ ) in state 3 as a percentage of EPV of disability claims costs ( $\delta = 0.05$ ) with no benefits in state 3 (baseline) for a life starting in the healthy state for males, females and combined using all disability models.

		EPV of	benefits	with $1/2$	benefits :	in state 3	as a $\%$
		of EPV	V of bene	fits with :	no benefi	ts in state	e 3 for:
	Entry	1982 - 84	models	1984 - 89	models	1989-94	models
Gender	age	(5)	(10)	(5)	(10)	(5)	(10)
		%	%	%	%	%	%
M & F	60	128.17	127.99	129.40	127.50	131.63	129.23
	65	126.29	126.96	128.45	126.49	130.87	128.38
	70	124.51	125.77	126.95	125.87	130.08	127.13
	75	122.94	124.55	125.01	124.12	129.02	125.42
$\mathbf{F}$	60	126.30	127.28	126.34	125.74	130.23	127.24
	65	125.56	126.27	126.89	125.33	129.99	126.48
	70	123.82	124.88	127.41	125.41	129.52	125.25
	75	121.59	123.31	126.63	124.35	128.55	123.35
Μ	60	133.45	133.35	137.43	129.80	134.59	125.40
	65	131.67	132.08	131.26	126.81	132.65	127.58
	70	131.69	131.89	124.60	122.43	131.73	128.42
	75	133.27	133.61	119.25	118.24	131.41	127.21

Table 7.81: EPV of disability claims costs with the addition of unit benefits increasing continuously ( $\delta_b = 0.05$ ) in state 3 as a percentage of EPV of disability claims costs ( $\delta = 0.05$ ) with no benefits in state 3 (baseline) for a life starting in the healthy state for males, females and combined using all disability models.

		EPV of	benefits	with unit	benefits	in state 3	B as a $%$
		of EPV	V of bene	fits with	no benefi <sup>.</sup>	ts in state	e 3 for:
	Entry	1982 - 84	models	1984 - 89	) models	1989-94	l models
Gender	age	(5)	(10)	(5)	(10)	(5)	(10)
		%	%	%	%	%	%
M & F	60	156.35	155.98	158.80	155.00	163.26	158.47
	65	152.58	153.91	156.89	152.98	161.74	156.75
	70	149.02	151.55	153.90	151.74	160.16	154.27
	75	145.87	149.11	150.02	148.25	158.04	150.84
$\mathbf{F}$	60	152.60	154.56	152.69	151.48	160.47	154.47
	65	151.13	152.54	153.79	150.66	159.99	152.96
	70	147.65	149.75	154.83	150.82	159.03	150.51
	75	143.17	146.63	153.25	148.70	157.11	146.70
Μ	60	166.91	166.71	174.87	159.61	169.18	150.79
	65	163.34	164.17	162.52	153.61	165.31	155.15
	70	163.38	163.78	149.19	144.87	163.46	156.84
	75	166.55	167.21	138.49	136.47	162.82	154.42

For a lower bound on the force of mortality, an obvious choice is to assume that the force of mortality from each state after age 90 remains at the same level as it was at age 90. To ensure that mortality is not actually being increased in any state using this assumption (for example, if mortality in a state is actually decreasing), after age 90 I take the minimum of the force of mortality at age 90 and the original force of mortality. (This modification only affected the mortality in 3 models, all for males and females combined — the 1982–84 (5) model, the 1984–89 (5) model and the 1989–94 (5) model, all of which have, necessarily identical, slightly decreasing mortality in the 3–4 ADL state.) Or more specifically, if  $\mu_{x+t}^{i7}$  is the force of mortality in state *i* for a person aged x + t then define a lower bound on the forces of mortality after age 90,  ${}^{l}\mu_{x+t}^{i7}$ , as:

$${}^{l}\mu_{x+t}^{i7} = \begin{cases} \mu_{x+t}^{i7} & \text{if } x+t \le 90\\ \min\left(\mu_{90}^{i7}, \, \mu_{x+t}^{i7}\right) & \text{otherwise} \end{cases}$$
(7.70)

Upper bounds on the forces of mortality can be defined in a similar way — by assuming that the force of mortality after age 90 continues to increase exponentially. For males I assume that the forces of mortality increase at least as fast as does the Makeham fit to AM80 mortality,  $^{AM80}\mu_{x+t}$ , given in equation 1.2. For females and both genders combined, I assume the forces of mortality increase at least as fast as fast as does the Makeham fit to AF80 mortality,  $^{AF80}\mu_{x+t}$ , also given in equation 1.2. For example, for males define an upper bound on the force of mortality after age 90,  $^{u}\mu_{x+t}^{i7}$ , as:

$${}^{u}\!\mu_{x+t}^{i7} = \begin{cases} \mu_{x+t}^{i7} & \text{if } x+t \le 90\\ \max\left(\left[\mu_{90}^{i7} - {}^{AM80}\mu_{90} + {}^{AM80}\mu_{x+t}\right], \, \mu_{x+t}^{i7}\right) & \text{otherwise} \end{cases}$$
(7.71)

For females and for both genders combined, an upper bound can be defined by replacing AM80 mortality with AF80 mortality in the above equation.

In a few of the disability models, the force of mortality in some states increased faster than the Makeham fit to AM80 or AF80 (for males and both genders combined in the 1982–84 (10) model, the 1984–89 (10) model and the 1989–94 (10) model; and for females in the 1984–89 (10) model). Though it is noticeable that this only occurred in the models that were parameterized to data grouped in 10-year age

Table 7.82: EPV of disability claims costs with mortality from all states adjusted to be constant after age 90 as a percentage of EPV of disability claims costs with no mortality adjustments (baseline) for benefits increasing continuously ( $\delta_b = 0.05$ ) for a life starting in the healthy state for males, females and combined using all disability models.

		EPV of	f benefits	with level	l mortality	v (> 90  yr)	rs) as a
		% of	EPV of b	enefits wi	th origina	l mortalit	y for:
	Entry	1982 - 84	models	1984 - 89	models	1989 - 94	models
Gender	age	(5)	(10)	(5)	(10)	(5)	(10)
		%	%	%	%	%	%
M & F	60	102.24	104.03	103.84	106.30	103.00	105.46
	65	102.43	104.38	104.11	107.00	103.38	106.27
	70	102.86	105.15	104.74	108.08	104.08	107.58
	75	103.77	106.80	106.03	110.20	105.40	109.91
$\mathbf{F}$	60	103.33	103.43	105.23	105.60	104.46	104.54
	65	103.46	103.66	105.77	106.34	105.01	105.25
	70	103.89	104.17	106.63	107.42	105.98	106.29
	75	105.00	105.33	108.34	109.33	107.73	108.07
Μ	60	101.27	101.62	102.45	103.94	102.13	102.91
	65	101.47	101.94	102.89	104.41	102.40	103.47
	70	101.89	102.50	103.46	105.09	103.05	104.39
	75	102.77	103.65	104.61	106.56	104.40	106.14

bands. This is not surprising since it was noted in Section 5.5 that some of the Makeham fits using these age groupings were unstable and, in particular, produced unusually high estimates of the exponential parameter.

Tables 7.82 and 7.83 give the EPV of benefits using the lower and upper bound forces of mortality, as defined above, respectively, as a percentage of the EPV of benefits using the original forces of mortality. From these Tables:

- using the lower bound forces of mortality, the EPV of benefits are increased by between 1% and 10% and the increase is greater at older entry ages (102% to 110% at entry age 75) and less at younger entry ages (101% to 106% at entry age 60);
- using the upper bound forces of mortality, the EPV of benefits are reduced by between 1% and 10% and the reduction is greater at older entry ages (90% to 99% at entry age 75) and less at younger entry ages (94% to 99% at entry age 60);

Table 7.83: EPV of disability claims costs with mortality after age 90 from all states adjusted to exponential (males — AM80, females and combined — AF80) as a percentage of EPV of disability claims costs with no mortality adjustments (baseline) for benefits increasing continuously ( $\delta_b = 0.05$ ) for a life starting in the healthy state for males, females and combined using all disability models.

		EPV of benefits with exponential mortality $(> 90 \text{ yrs})$					
		as a % of EPV of benefits with original mortality for:					
	Entry	1982-84  models		1984-89  models		1989-94  models	
Gender	age	(5)	(10)	(5)	(10)	(5)	(10)
		%	%	%	%	%	%
M & F	60	98.12	99.20	94.60	96.53	96.99	98.36
	65	97.97	99.13	94.23	96.15	96.61	98.12
	70	97.60	98.98	93.33	95.55	95.91	97.72
	75	96.82	98.65	91.49	94.35	94.57	97.01
$\mathbf{F}$	60	97.62	98.06	94.08	95.09	95.91	97.25
	65	97.53	97.93	93.46	94.44	95.41	96.82
	70	97.22	97.63	92.48	93.47	94.51	96.18
	75	96.41	96.95	90.54	91.73	92.89	95.07
Μ	60	99.15	99.41	96.26	96.07	97.39	98.37
	65	99.01	99.30	95.58	95.60	97.06	98.05
	70	98.73	99.09	94.71	94.92	96.26	97.53
	75	98.13	98.67	92.94	93.45	94.60	96.55

It is expected that the percentage changes to the EPV of benefits will be largest for the oldest entry ages, since the changes to the force of mortality are at ages over 90, which affect the benefits at ages over 90, which, in turn, make up a greater proportion of costs for a contract starting at age 75, as opposed to one starting at age 60 (which also includes the EPV of benefits payable in the period 60 to 75 years). This effect is also compounded by discounting over a longer period for contracts starting at age 60. Overall, it seems that the mortality assumptions over age 90 years have a relatively small effect on the EPV of benefits — even for contracts starting at older ages.

## 7.4 Impact of Adverse Selection on Long-Term Care Insurance Revisited

In this section, the costs of disability from the disability models are combined with the costs of Alzheimer's disease (from Section 2.6.2) to revisit the potential costs of adverse selection in LTC insurance. I use the same definitions and terminology as introduced in Chapters 1 and 2 — for example, by adverse selection here I am referring to lives that know they have ApoE genotypes that predispose them to Alzheimer's disease (the  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes) being more likely to take out long-term care insurance than lives with other genotypes.

In Chapter 2, the costs of adverse selection were first calculated based on the costs of Alzheimer's disease alone (i.e. the extra costs from adverse selection were given as a percentage of the baseline costs of AD only). The 'baseline' assumptions were:

- mortality: 65% AM80 for males and 65% AF80 for females and both genders combined;
- 2. incidence of AD: genotype-specific aggregate (across genders) incidence rate of AD (defined in equations 1.3 and 1.13) used for all gender classifications;
- 3. level of relative risk, m = 1, 0.5, 0.25;
- 4. rate of benefit escalation,  $\delta_b = 0.05$ ;
- 5. rate of discount,  $\delta = 0.05$ ; and
- 6. claiming commences w = 0 years before institutionalization.

Adverse selection was then modelled by assuming that lives with high-risk genotypes were k (k = 2, 4, 10 and 100) times more likely to insure than lives with other genotypes (see Section 2.6 for more detail) and the associated costs were given as a percentage of the baseline costs of AD. Then to allow for other costs of claiming in a LTC contract, it was very simply assumed that AD was responsible for between 40% and 50% of all long-term care costs (for all age groups and gender classifications). This translated directly into the total percentage costs of adverse selection being 40–50% of the costs of adverse selection as a percentage of AD-related costs alone. With estimates of total long-term care costs from the disability models (the EPV of benefits, given in the previous two sections), it is now possible to estimate the proportion of total long-term care costs attributable to Alzheimer's disease by gender and starting age. These proportions can then be applied to the costs of adverse



Figure 7.87: Comparison of the overall force of mortality in the disability models to the overall force of mortality in the Alzheimer's disease model, for males.

selection (as a percentage of AD-related costs) to give estimates of total percentage costs of adverse selection.

As lives with Alzheimer's disease were not excluded nor identifiable in the NLTCS, I assume that the EPV of benefits calculated from the disability models are estimates of total long-term care costs (i.e. they include the costs of claiming from lives with Alzheimer's disease, as these lives in the survey would become disabled from Alzheimer's disease). It is now a simple matter to calculate the proportion of costs in a long-term care contract attributable to Alzheimer's disease — by dividing the aggregated (by genotype) costs of AD given in Table 2.13 by the total long-term care costs given in Table 7.75. However, before comparing the EPV of benefits between the two models, I look at overall mortality in both models to check for consistency. For females, this was done in Section 7.2 and for males and both genders combined the forces of mortality are compared in Figures 7.87 and 7.88, respectively. These figures show:

1. the highest overall mortality for the 6 disability models (for males and both genders combined this is from the 1982–84 (10) model);


Figure 7.88: Comparison of the overall force of mortality in the disability models to the overall force of mortality in the Alzheimer's disease model, for males and females combined.

- 2. the lowest overall mortality for the 6 disability models (for males and both genders combined this is from the 1984–89 (5) model); and
- overall mortality for the Alzheimer's disease model with baseline mortality as 65% AM80 for males and as 65% AF80 for both genders combined (see Section 2.4 for more detail).

For both of these gender classifications, the trends are very similar to those for females. However, for these two gender classifications, the difference is greater at younger ages (60 – 90 years) between the overall forces of mortality in the AD models and that from the disability model with the lowest mortality. The trends at older ages are similar for all three models, at about age 100 (age 95 for males), the forces of mortality in the AD models increase above the force of mortality from the 1984– 89 (5) models (representing a lower bound), which are then starting to level off. After this, overall mortality in the AD models continue to increase exponentially: for males, going above the upper limit of mortality in the disability models (from the 1982–84 (10) model) by just after age 110; and for both genders combined, at a similar rate but at a smaller magnitude to the upper limit of mortality in the disability models (from the 1982–84 (10) model).

From the previous section and Section 2.5, it was seen that the EPVs of benefits were not very sensitive, in either the AD model nor the disability model, to the mortality assumptions at older ages (> 90 years) — however, the difference in mortality in younger ages may be significant, and it is noticeable that, for males and both genders combined, the forces of mortality at younger ages in the AD models are below those in the disability models. This may result in relatively higher estimates of EPVs of benefits in the AD models compared to the disability models, which, in turn, could cause estimates of the proportion of total LTC cost attributable to AD to be artificially high.

Table 7.84 gives these proportions using the EPV of benefits from the 1982–84 (5) model and in Table 7.85 ranges of proportions are calculated using the EPV of benefits from all of the disability models. The figures in Table 7.85 for females differ slightly from those given in Table 7.76, since different estimates of the costs of AD are used — in the former the costs of AD are calculated from aggregating the genotype specific costs, whereas in the latter the aggregate costs of AD are calculated directly.

From these tables, for females the proportion of total LTC costs attributable to AD is between 39.25% and 49.31%. As noted in Section 7.2, the assumption used in Chapter 2 that the costs of AD make up between 40% and 50% of total LTC costs is strongly supported for females. The figures from the 1982–84 (5) model are almost constant, at about 45%, for all ages at entry into insurance and for all levels of relative risk. Looking at Table 7.85 using all disability models, there is a slight trend for the range of proportions to increase with age at entry: at entry age 60 the range is 39.26%–46.80%; at entry age 65 it is 42.75%–46.31%; at entry age 70 it is 44.91%–47.02%; and at entry age 75 it is 44.59%–49.31%.

For males, the proportion of total LTC costs are much higher than for females — from the 1982–84 (5) model, the proportions are about 55% at entry age 60, increasing to about 65% at entry age 75. The ranges for the proportions are much wider for males as well (ranges of 20% in some cases), which is a direct result of the Table 7.84: EPV of benefits from Alzheimer's disease, with proportion of relative risk, m = 1.00, 0.5 and 0.25 (from Table 2.13) as a percentage of EPV of benefits from disability using the 1982–84 (5) model (from Table 7.75), with benefits increasing at  $\delta_b = 0.05$ , for males, females and combined.

		AD costs a	as a $\%$ of dis	sability costs
	Entry	with prop	portion of re	elative risk:
Gender	age	m = 1.00	m = 0.50	m = 0.025
		%	%	%
M & F	60	61.19	59.74	58.49
	65	62.28	61.20	60.12
	70	61.76	61.35	60.63
	75	59.95	60.36	60.10
$\mathbf{F}$	60	46.80	46.45	45.90
	65	45.80	45.76	45.42
	70	44.91	45.41	45.33
	75	44.59	45.77	46.08
Μ	60	57.29	55.46	54.55
	65	61.68	60.26	59.55
	70	64.61	64.05	63.68
	75	65.90	66.67	66.67

Table 7.85: EPV of benefits from Alzheimer's disease, with proportion of relative risk, m = 1.00, 0.5 and 0.25 (from Table 2.13) as a range of percentages of EPV of benefits from disability using all the disability models (from Table 7.75), with benefits increasing at  $\delta_b = 0.05$ , for males, females and combined.

		AD costs	as a $\%$ of disab	ility costs
	Entry	with pro	oportion of relat	ive risk:
Gender	age	m = 1.00	m = 0.50	m = 0.025
		% - %	% - %	% - %
M & F	60	61.03 - 68.36	59.58 - 66.74	58.34 - 65.34
	65	62.28 - 69.29	61.20 - 68.10	60.12 - 66.90
	70	61.76 - 68.36	61.35 - 67.91	60.63 - 67.11
	75	59.95 - 66.86	60.36 - 67.32	60.10 - 67.03
$\mathbf{F}$	60	40.03 - 46.80	39.73 - 46.45	39.26 - 45.90
	65	43.11 - 46.35	43.07 - 46.31	42.75 - 45.96
	70	44.91 - 46.50	45.41 - 47.02	45.33 - 46.94
	75	44.59 - 47.71	45.77 - 48.97	46.08 - 49.31
Μ	60	57.29 - 78.04	55.46 - 75.55	54.55 - 74.31
	65	61.68 - 78.76	60.26 - 76.95	59.55 - 76.04
	70	64.61 - 78.32	64.05 - 77.64	63.68 - 77.19
	75	65.58 - 75.22	66.35 - 76.10	66.35 - 76.10

Table 7.86: Costs of adverse selection as a percentage of total LTC insurance costs (proportions from Table 7.85), with  $\varepsilon 2/\varepsilon 4$ ,  $\varepsilon 3/\varepsilon 4$  and  $\varepsilon 4/\varepsilon 4$  genotypes k times as likely to insure as low-risk genotypes, for benefits increasing continuously ( $\delta_b = 0.05$ ) and commencing on institutionalisation, for males and females.

	Likelihood					
	of high risk	Prop'n of		Cost of adve	erse selection	
	genotypes	relative		at a	age:	
Gender	insuring, $k$	risk, $m$	60	65	70	75
			%	%	%	%
$\mathbf{F}$	2	1.00	3.9 - 4.6	4.0 - 4.3	3.6 - 3.8	2.9 - 3.1
		0.25	1.1 - 1.3	1.2-1.2	1.1 - 1.1	0.8 - 0.9
Μ	2	1.00	2.3 - 3.2	2.2-2.8	1.8-2.2	1.5 - 1.7
		0.25	0.5 - 0.7	0.5 - 0.6	0.4 - 0.5	0.3 - 0.4
$\mathbf{F}$	4	1.00	8.4 - 9.8	8.5 - 9.1	7.8 - 8.1	6.3 - 6.7
		0.25	2.4-2.9	2.5-2.6	2.2-2.3	1.8 - 1.9
Μ	4	1.00	5.0 - 6.8	4.6 - 5.9	3.9 - 4.7	3.1 - 3.6
		0.25	1.1 - 1.5	1.0 - 1.3	0.8 - 1.0	0.7 - 0.7
$\mathbf{F}$	10	1.00	13.5 - 15.8	13.7 - 14.7	12.7 - 13.2	10.3 - 11.0
		0.25	3.9 - 4.6	4.0 - 4.3	3.6 - 3.8	2.9 - 3.1
Μ	10	1.00	8.0 - 10.9	7.4-9.5	6.2-7.5	5.0 - 5.8
		0.25	1.8-2.4	1.6-2.1	1.3-1.6	1.0 - 1.2
$\mathbf{F}$	100	1.00	18.6 - 21.7	19.0 - 20.4	17.8 - 18.4	14.6 - 15.6
		0.25	5.4 - 6.3	5.5-5.9	5.0 - 5.2	4.0 - 4.3
Μ	100	1.00	11.0 - 15.0	10.3 - 13.1	8.6 - 10.4	7.0 - 8.0
		0.25	2.4 - 3.3	2.3-2.9	1.9-2.3	1.4 - 1.7

wider spread of cost estimates of disability for males (which in turn may be a result of there being less data for males when parameterizing the original disability models). This implies that, for males, the assumption that AD costs constitute 40%–50% of total LTC costs is an underestimate — the figures in Table 7.85 suggest that the lower bound on the proportion of total LTC costs attributable to AD should be about 55% at entry age 60, increasing to just below 65% at entry age 75 and that an upper bound of just below 80% would be appropriate for all ages at entry. This means that the upper bound on the costs of adverse selection as a percentage of total LTC costs could have been underestimated by as much as 30% in Chapter 2.

The proportions for both genders combined are, at first, quite surprising. They are at a similar level to those for males, though there does not seem to be any trend by age at entry, and the ranges are much smaller — over all ages at entry and models the range of proportions is 58.34%–69.29%. It would be expected that the

proportions would lie between those for males separately and females separately, as do the estimates of total LTC costs (EPV of benefits in the disability models). This anomaly comes from the fact that the estimates of the costs of AD for males and females combined were very similar to those for females alone, which is a consequence of how the AD model for males and females combined was parameterized. While the relative risks used were for both genders combined, the baseline mortality was assumed to be 65% of a Makeham fit to AF80 (the same assumption that was used for the female AD model) — the justification for this was that the majority of older people will be females, due to the usual difference in mortality between genders. I have given the proportions for both genders combined for illustration, but, for the above reason, I will place more reliance on the results for males and females separately.

These proportions can be applied to the costs of adverse selection as percentages of AD-related costs given in Section 2.4 to give new estimates of the total percentage costs of adverse selection. For illustration, Table 7.86 gives ranges of total percentage costs using the proportions from Table 7.85 and the costs of adverse selection from Section 2.4. Comparing this table to the original estimates of ranges of total percentage costs given in Table 2.30:

- 1. for females there is very little difference between the two and, in fact, the revised ranges of costs all lie within those originally calculated; and
- 2. for males, the revised costs are substantially higher, the new lower bounds always exceed the upper bounds from the original estimates however the costs for males are still only about 2/3 of those for females.

While I have included the revised costs of adverse selection for males, it is worth noting that the relative risks used for males were found to be anomalous in Section 2.6.1 (in that 'high risk' genotypes seemed to confer protection in some cases), and they were adjusted to ensure that 'high risk' genotypes did actually confer risk however this was a subjective adjustment. This does not affect the validity of the estimates of the proprotion of total LTC costs attributable to AD for males though, and these could be used in future to estimate total percentage costs of AD for men (given a consistent set of relative risks for males). Table 7.86 gives the range of costs of adverse selection for the baseline models. However, other costs can easily be calculated using other tables. For example, for females taking out insurance at age 70 and under the following assumptions:

- claiming from Alzheimer's disease commences 2 years before institutionalisation;
- 2. the extra costs of Alzheimer's disease from the above assumption do not increase total long-term care costs (which may be a reasonable assumption, since lives suffering from Alzheimer's disease would start claiming before institutionalisation in the disability model);
- there is an additional benefit of 1/2 in state 3 (1–2 ADL state) in the disability model;
- 4. there is an increased likelihood of k = 10 of high-risk genotypes insuring;
- 5. the proportion of relative risk m = 1.00;
- 6. all benefits increase continuously at a rate  $\delta_b = 0.05$ ; and
- 7. the discount rate  $\delta = 0.05$ .

an upper bound on the cost of adverse selection can be calculated as (from Tables 2.27, 7.80 and 7.85):

$$\frac{0.2565 \times 0.4650}{1.2382} = 9.63\% \tag{7.72}$$

and a lower bound as:

$$\frac{0.2565 \times 0.4491}{1.2952} = 8.89\% \tag{7.73}$$

#### 7.5 Summary and Conclusions

In Chapter 2, in order to look at the potential costs of adverse selection arising from variants of the APOE gene, the cost of other events in the ageing process (mainly disability) that trigger benefits were very simply assumed to be a multiple of those costs arising from Alzheimer's disease.

In this chapter I have used the disability model described in Chapter 3, and parameterized in Chapters 4 to 6 to estimate (independently from Alzheimer's disease) the costs of disability (EPV of benefits) in long-term care insurance.

The sensitivity analysis, on the key model assumptions, indicated that: the EPVs of benefits are sensitive to the rate of benefit escalation and to the addition of extra benefits at lesser disability levels; and that they are not very sensitive to mortality assumptions after age 90 — this was investigated since the overall forces of mortality in the disability models were very consistent with the benchmark forces of mortality up to about age 90, but became less consistent at older ages (see Section 6.6).

I then compared overall mortality in the AD model with that in the disability model to check for consistency. It was clear that: at younger ages (60–90 years), for females, mortality between the two models was very similar; but for males and both genders combined, the overall force of mortality in the AD model was clearly below that of the disability models. At older ages, the differences between the forces of mortality of the two models were similar for all gender classifications. Overall:

- 1. for females, mortality between the two models seemed comparable; but
- 2. for males and both genders combined, lighter mortality at younger ages in the AD model may cause the estimates of AD-related costs to increase relative to total LTC costs, artificially increasing estimates of the proportion of total LTC costs attributable to AD.

The proportions of total LTC costs attributable to AD were then estimated for each gender classification and starting age (60, 65, 70 and 75), using the EPV of benefits from the AD model and from the disability model (assumed to be estimates of total LTC costs), using the baseline assumptions for both models, to ensure consistency. In Chapter 2, it was assumed that the costs of AD made up 40%–50% of total LTC costs for all genders classifications and starting ages. In this chapter the range of estimated proportions was estimated as:

1. for females, between 39.26% and 49.31%, with a slight tendancy for the proportion to increase with increasing starting age — though overall, this is in strong agreement with the assumption made in Chapter 2;

- for males, between 54.55% and 78.76%, with a tendacy for the proportions to increase with starting age, with a fairly stable upper bound of about 80% for all starting ages; and
- 3. for both genders combined, between 58.34% and 69.29%, with no clear trend by starting age. For the 1982–84 (5) model, the proportions were about 60%.

For illustration, I applied these proportions to the costs of adverse selection as percentages of AD-related costs (from Section 2.4), to give ranges of estimates of total percentage costs of adverse selection, for a given set of assumptions about the level of adverse selection. Also, for example, I illustrated how a range of total percentage costs of adverse selection could be estimated, for other types of contracts/assumptions, using the results from the sensitivity analyses of the AD model and the disability model.

The results differ for each gender classifications and I conclude for each separately. For females:

- 1. When parameterizing both models, there was either more data for females (disability model), or more reliable data (AD model), suggesting that the results for females will be the most reliable of the three gender classifications.
- 2. Furthermore, the overall forces of model mortality for females in the AD model and the disability models were very similar (and the closest of the 3 genders classifications) at younger ages (60–90 years), suggesting that it would be reasonable to compare results from the AD model and the disability model.
- 3. The proportions estimated using the EPV of benefits from the AD model and the disability model were in almost total agreement with the assumption made in Chapter 2 — providing very strong support to the results and conclusions of that chapter (see Section 2.8 for conclusions).

#### For males:

1. In Chapter 2, the costs of adverse selection for males appeared to be negative (the  $\varepsilon 2$  allele confered such protection that the  $\varepsilon 3/\varepsilon 3$  genotype was high-risk at many ages). For the study of adverse selection in this chapter, the relative risks were adjusted so that the  $\varepsilon 4$  allele never conferred lower risk, but it was noted that no reliance should be placed on these results.

- 2. Even so, this does not affect the overall estimates of the costs of AD for males, since the genotype-specific incidence rates of AD were adjusted so that the overall incidence of AD was consistent with the benchmark incidence of AD suggesting that it is still reasonable to use these overall cost estimates of AD to calculate the proportion of total LTC costs attributable to AD.
- 3. The overall force of mortality in the AD model was found to be slightly lighter than the overall forces of mortality in the disability models at younger ages (60–95 years). After age 95, mortality between the two models was roughly comparable. This suggests that the costs of AD may be slightly inflated in comparison to the total LTC costs estimated from the disability models.
- 4. Even after allowing for the above point, it still seems very likely that the proportion of total LTC costs attributable to AD for males is substantially greater than that assumed in Chapter 2 (40%–50%). For example, assuming that total LTC costs had been underestimated by 10% (from mortality differences), the estimated range of proportions would still be 45% 70%.
- 5. The main findings for males are: estimates of total LTC costs; and the suggestion that the proportion of total LTC costs attributable to AD is likely to be in excess of 50%. Application of the proportions found in this chapter resulted in estimates of total percentage costs of adverse selection for males to be about 2/3 of those for females though no reliance can be placed on these results because of the anomalous relative risks.

And for both genders combined:

1. It is not really necessary to look at both genders combined, since the results for males and females separately can be combined to get the same result (using the appropriate proportions). The main advantage of looking at both genders combined is that there will be more data for parameterizing models, especially important when there is a scarcity of data.

- 2. In the Alzheimer's disease model baseline mortality was taken to be 65% AF80 (the same as for females) the only difference between the model for females and both genders combined was the relative risks for AD (the main focus in this chapter). However, as the relative risks were adjusted to be consistent with a benchmark incidence of AD (for both genders), the aggregate (over genotypes) costs of AD were very similar for females and both genders combined.
- 3. In the disability model, the results for both genders combined were very consistent with the results for males and females separately (even though the models were parameterized separately) they lay in between the results for males and females, closer to the results for females (as expected, since there were more females in the original data sets).
- 4. It would then be expected that the estimated proportions of AD-related LTC costs would be overestimated (as the costs of AD would be overestimated).
- 5. This was confirmed when comparing the overall force of mortality in the AD model with the overall forces of mortality in the disability models the force of mortality in the AD model was lighter than that for the disability model with the lightest force of mortality, at younger ages (60–100 years). The difference was greater than that reported above for males.
- 6. Given this inconsistency between models it is very difficult to draw any firm conclusions — combining the results for males and females separately, and the proportions calculated for both genders combined, it may be that the proportion of total LTC costs attributable to AD for both genders is slighter greater than that assumed in Chapter 2 (40%–50%).

### Chapter 8

### Areas for Further Research

I will first discuss areas of further research for the two models that I have proposed in this thesis separately (the Alzheimer's disease model, described in Chapter 1, and the disability model, described in Chapter 3) and then discuss areas common to both. For the Alzheimer's disease model:

- 1. There was no single study that would allow all the intensities in the model to be estimated simultaneously. The estimation was based on a number of different studies, some quite small, of different populations, with different research protocols and methods of analysis, and very likely different definitions of 'onset of AD' and 'institutionalization'. Use of a complete data set of lives with Alzheimer's disease would remove many uncertainties in the estimation process as well as allowing confidence intervals to be estimated and, depending on the data, claiming could be represented using definitions closer to those used in LTC insurance (i.e. on reaching a certain level of cognitive impairment, or loss of ADLs).
- 2. The relative risks of the APOE genotypes were based on case-based studies, not prospective population studies, and the risks associated with the  $\varepsilon 4$  allele are almost certain to be lower than those estimated to date. I was unable to do more than to show what effect this might have. It would be very useful to update the relative risks of AD by APOE genotype as more prospective population studies are done. Ideally, the incidence of AD by APOE genotype could be directly estimated, rather than having to adjust an aggregate (by

genotype) incidence rate — though this will depend on the data and research that become available.

3. In the AD model, all of the transition intensities other than the incidence of AD were assumed to be same for each APOE genotype (the data was not available to assume otherwise). As more genetic studies are undertaken, it may be possible to allow (or at least check) for the dependency of the other transition intensities on APOE genotype (i.e. overall mortality may be different between different genotypes).

In general, with the great speed at which research on human genetics is advancing, revisiting the model and updating it in the light of new knowledge will be necessary to keep it credible. While some of the refinements described above will depend on the research that is done in the future and the data gathered, others could be approached given access to present data sets. For example, access to the CERAD (Consortium to Establish a Registry for Alzheimer's Disease) data set would enable research to be done on issues raised in the first point above.

For the disability model:

- The data set I used to parameterise the disability model has been used by previous researchers to investigate trends of disability (using different methodology). The disability model I have described here could be put to similar uses and the results compared with their results.
- 2. I have applied the disability model in this thesis to estimating the costs of disability in a LTC insurance contract. However, it could easily be applied to estimating health care costs from disability in general, or for projecting disability levels.
- 3. The data set used, the NLTCS, is from the U.S.A. and experience in the U.K. may be different. Similar data from the U.K. could be analysed using the methodology set out in this thesis, and the trends of disability compared between countries, as well as then having a model more relevant to the U.K.. Of course, longitudinal data would be ideal, but is rarely available since collection of such data is very costly and time consuming.

Areas in general and those that affect both models:

- 1. It would be ideal to be able to combine disability and Alzheimer's disease in a single model that describes both processes. However, this would require data that describe at an individual level progress through ADLs and onset and progress of Alzheimer's disease.
- 2. Throughout this thesis, I assumed a constant discount rate of 0.05. Incorporation of a financial model of interest rates would be particularly useful for modelling products that have index linked benefits/premiums.

#### References

- BARCLAY, L.L., ZEMCOV, A. BLASS, J.P. & MCDOWELL F.H. (1985a). Factors associated with duration of survival in Alzheimer's disease. *Biological Psychiatry*, 20, 86–93.
- BARCLAY, L.L., ZEMCOV, A., BLASS, J.P. & SANSONE, J. (1985b). Survival in Alzheimer's disease and vascular dementias. *Neurology*, 35, 834–840.
- BASUN, H., GRUT, M., WINBLAD, B. & LANNFELDT, L. (1995). Apolipoprotein ε4 allele and disease progression in patients with late onset Alzhiemer's disease. Neuroscience Letters, 183, 32–34.
- BEARD, C.M., KOKMEN, E., O'BRIEN, P.C. & KURLAND, L.T. (1994). Are patients with Alzheimer's disease surviving longer in recent years? *Neurology*, 44, 1869–1871.
- BERG, L., MILLER, J.P., STORANDT, M., DUCHEK, J., MORRIS, J.C., RUBIN,
  E.H., BURKE, W.J. & COBEN, L.A. (1988). Mild senile dementia of the
  Alzheimer type: 2. Longitudinal assessment. Annals of Neurology, 23, 477–484.
- BICKEBÖLLER, H., CAMPION, D., BRICE, A. AMOUYEL, P., HANNEQUIN, D., DIDIERJEAN, O., PENET, C., MARTIN, C., PÉREZ-TUR, J., MICHON, A., DUBOIS, B., LEDOZE, F., THOMAS-ANTERION, C., PASQUIER, F., PUEL, M., DEMONET, J-F., MOREAUD, O., BABRON, M-C., MEULIEN, D., GUEZ, D., CHARTIER-HARLIN, M-C., FREBOURG, T., AGID, Y., MARTINEZ, M. & CLERGET-DARPOUX, F. (1997). Apolipoprotein E and Alzheimer's disease: genotype-specific risks by age and sex. *American Journal of Human Genetics*, 60, 439–446.
- BONAIUTO, S., MELE, M., GALLUZO, L. & GIANNANDREA, E. (1995). Survival and dementia: A 7-year follow up of an Italian population. Archives of Gerontology and Geriatrics, 20, 105–113.
- BRACCO, L., GALLATO, R., GRIGOLETTO, F., LIPPI, A., LEPORE, V., BINO, G., LAZZARO, M.P., CARELLA, F., PICCOLO, T., POZZILLI, C., GIOMETTO,

B. & AMADUCCI, L. (1994). Factors affecting course and survival in Alzheimer's disease. *Archives of Neurology*, **51**, 1213–1219.

- BRETELER, M.B.B., CLAUS, J.J., VAN DUIJN, C.M., LAUNER, L.J. & HOF-MAN, A. (1992). Epidemiology of Alzheimer's disease. *Epidemiologic Reviews*, 14,59–82.
- BRINDLE, N., SONG, Y., ROGAEVA, E., PREMKUMAR, S., LEVESQUE, G., YU. G., IKEDA, M., NISHIMURA, M., PATERSON, A., SORBI, S., DUARA, R., FARRER, L. & ST GEORGE-HYSLOP, P. (1998). Analysis of the butyrycholinesterase gene and nearby chromosome 3 markers in Alzheimer's disease. *Human Molecular Genetics*, 7, 933–935.
- BURNS, A., LEWIS, G., JACOBY, R. & LEVY, R. (1991). Factors affecting survival in Alzheimer's disease. *Psychological Medicine*, 21, 363–370.
- CLAYTON, D. & HILLS, M. (1993). *Statistical methods in epidemiology*. Oxford University Press.
- CONTE, S.D. & DE BOOR, C. (1972). *Elementary numerical analysis*. McGraw-Hill, New York.
- COPELAND, J.R.M., DAVIDSON, I.A., DEWEY, M.E., GILMORE, C., LARKIN, B.A., MCWILLIAM, C., SAUNDERS, P.A., SCOTT, A., SHARMA, V. & SUL-LIVAN, C. (1992). Alzheimer's disease, other dementias, depression and pseudodementia: prevalence, incidence and three-year outcome in Liverpool. *British Journal of Psychiatry*, **161**, 230–239.
- CORDER, E.H., SAUNDERS, A.M., STRITTMATTER, W.J., SCHMECHEL, D.E., GASKELL, P.C., RIMMLER, J.B., LOCKE, P.A., CONNEALLY, P.M., SCH-MADER, K.E., SMALL, G.W., ROSES, A.D., HAINES, J.L. & PERICAK-VANCE, M.A. (1994). Protective effect of apolipoprotein E type 2 allele for late onset Alzheimer's disease. *Nature Genetics*, 7, 180–184.
- CORDER, E.H., SAUNDERS, A.M., STRITTMATTER, W.J., SCHMECHEL, D.E., GASKELL, P.C., RIMMLER, J.B., LOCKE, P.A., CONNEALLY, P.M., SCH-MADER, K.E., TANZI, R.E., GUSELLA, J.F., SMALL, G.W., ROSES, A.D., PERICAK-VANCE, M.A. & HAINES, J.L. (1995). Apolipoprotein E, survival in

Alzheimer's disease and the competing risks of death and Alzheimer's disease. Neurology, 45, 1323–1328.

- DIESFELDT, H.F.A, VAN HOUTE, L.R. & MOERKENS, R.M. (1986). Duration of survival in senile dementia. Acta Psychiatr Scandinavica, **73**, 366-371.
- VAN DIJK, P.T.M., DIPPEL, D.W.J. & HABBEMA, J. DIK F. (1991). Survival of patients with dementia. *Journal of the American Geriatric Society*, **39**, 603–610.
- VAN DUIJN, C.M., DE KNIFF, P., WEHNERT, A., DE VOECHT, J., BRON-ZOVA, J.B., HAVEKES, L.M., HOFMAN, A. & VAN BROECKHOVEN, C. (1995). The apolipoprotein E  $\varepsilon 4$  allele is associated with an increased risk of early-onset Alzheimer's disease and a reduced survival. *Annals of Neurology*, **37**, 605–610.
- DULLAWAY, D. & ELLIOTT, S. (1998). Long-term care insurance: A guide to product design and pricing. Presented to the Staple Inn Actuarial Society, 10 March 1998.
- DUNLOP, D.D., HUGHES, S.L. & MANHEIM, L.M. (1997). Disability in activities of daily living: patterns of change and a hierarchy of disability.*American Journal* of Public Health, 87, 378–383.
- EVANS, D.A., FUNKENSTEIN, H., ALBERT, M.S., SCHERR, P.A., COOK, N.R., CHOWN, M.J., HEBERT, L.E., HENNEKENS, C.H. & TAYLOR, J.O. (1989).
  Prevalence of Alzheimer's disease in a community population of older persons: Higher than previously reported. *Journal of the American Medical Association*, 262, 2551–2556.
- EVANS, D.A., SMITH, L.A., SCHERR, P.A., ALBERT, M.S., FUNKENSTEIN, H.H. & HEBERT, L.E. (1991). Risk of death from Alzheimer's disease in a community population of older persons. *American Journal of Epidemiology*, 134, 403–412.
- EVANS, D.A., BECKETT, L.A., FIELD, T.S., FENG, L., ALBERT, M.S., BENNETT, D.A., TYCKO, B. & MAYEUX R. (1997). Apolipoprotein E ε4 and incidence of Alzheimer's disease in a community population of elder persons. Journal of the American Medical Association, 277, 822–824.
- FARRER, L.A., CUPPLES, L.A., HAINES, J.L., HYMAN, B., KUKULL, W.A., MAYEUX, R., MYERS, R.H., PERICAK-VANCE, M.A., RISCH, N., VAN

DUIJN, C.M. & APOE AND ALZHEIMER DISEASE META ANALYSIS CON-SORTIUM (1997). Effects of age, gender and ethnicity on the association between apolipoprotein E genotype and Alzheimer's disease. *Journal of the American Medical Association*, **278**, 1349–1356.

- FORFAR, D.O., MCCUTCHEON, J.J. & WILKIE, A.D. (1988). On graduation by mathematical formula. J.I.A. 115, 1 – 149 and (with discussion) T.F.A. 41, 97 – 269.
- FRIEDMAN, L.M., FURBERG, C.D. & DEMETS, D.L. (1998). Fundamentals of clinical trials. Springer-Verlag, New York.
- FRISONI, G.B., GOVONI, S., GEROLDI, C., BIANCHETTI, A., CALABRES, L., FRANCESCHINI, G. & TRABUCCHI. M. (1995). Gene dose of the ε4 allele of apolipoprotein E and disease progression in sporodic late onset Alzheimer's disease. Annals of Neurology, 37, 596–604.
- GAO, S., HENDRIE, H.C., HALL, K.S. & HUI, S. (1998). The relationship between, age, sex and the incidence of dementia and Alzheimer's disease. Archives of General Psychiatry, 55, 809–815.
- GOMEZ-ISLA, T., WEST, H.L., REBECK, G.W., HARR, S.D., GOWDON, J.H., LOCASCIO, J.J., PERLS, T.T., LEPSITZ. L.A. & HYMAN, B.T. (1996). Clinical and pathological correlates of apolipoprotein E ε4 in Alzheimer's disease. Annals of Neurology, **39**, 62–70.
- GREEN, S., BENEDETTI, J. & CROWLEY, J. (1997). *Clinical trials in oncology*. Chapman and Hall, London.
- GUI, E.H. & MACDONALD, A.S. (2001) A Nelson-Aalen estimate of the incidence rates of early-onset Alzheimer's disease associated with the presenilin-1 gene. To appear in *ASTIN Bulletin*.
- HAGNELL, O., OJESJO, L. & RORSMAN, B. (1992). Incidence of dementia in the Lundby study. *Neuroepidemiology*, **11**, 61–66.
- HEBERT, L.E., SCHERR, P.A., BECKETT, L.A., ALBERT, M.S., PILGRIM, D.M., CHOWN, M.J., FUNKENSTEIN, H.H. & EVANS, D.A. (1995). Age specific incidence of Alzheimer's disease in a community population. *Journal of* the American Medical Association, 273, 1354–1359.

- HEYMAN, A., PETERSON, B., FILLENBAUM, G. & PIEPER, C. (1996). The consortium to establish a registry for Alzheimer's disease (CERAD) Part XIV: Demographic and clinical predictors of survival in patients with Alzheimer's disease. *Neurology*, 46, 656–660.
- HEYMAN, A., PETERSON, B., FILLENBAUM, G. & PIEPER, C. (1997). Predictors of time to institutionalization of patients with Alzheimer's disease: the CERAD experience part XVII. *Neurology*, 48, 1304–1309.
- HUMBLE, R.A. & RYAN, D.G. (1998). Continuing care retirement communities
   attractive to members, but what about sponsors? (with discussion). B.A.J., 4,547–614.
- JARVIK, G.P., LARSON, E.B., GODDARD, K., KUKULL, W.A., SCHELLENBERG, G.D. & WIJSMAN, E.M. (1996). Influence of apolipoprotein E genotype on the transmission of Alzheimer's disease in a community-based sample. American Journal of Human Genetics, 58, 191–200.
- JORM, A.F. & JOLLEY, D. (1998). The incidence of dementia a meta analysis. Neurology, 51, 728–733.
- JOST, B.C. & GROSSBERG, G.T. (1995). The natural history of Alzhiemer's disease: a brain bank study. Journal of the American Geriatrics Society, 43, 1248–1255.
- KAHN, H. & SEMPOS, C.T. (1989). Statistical methods in epidemiology. Oxford University Press.
- KOKMEN, E., CHANDRA, V. & SCHOENBERG, B. (1988). Trends in incidence of dementing illness in Rochester, Minnesota, in three quinquennial periods, 1960– 1974. Neurology, 38, 975–980.
- KOKMEN, E., BEARD, R.N., O'BRIEN, P.C., OFFORD, M.S. & KURLAND, L.T. (1993). Is the incidence of dementing illness changing? *Neurology*, **43**, 1887–1892.
- KULKARNI, V.G. (1995). Modeling and analysis of stochasic systems. Chapman and Hall.
- KUUSISTO, J., KOIVISTO, K., MYKKÄNEN, L., HELKALA, E-L., VANHANEN, M., HÄNNINEN, T., PYÖRÄLÄ, K., KESÄNIEMI, Y.A., RIEKKINON, P.&

LAAKSO, K. (1994). Association of apolipoprotein E phenotypes with late onset Alzheimer's disease: population based study. *British Medical Journal*, **309**, 636–638.

- LAMBERT, J-C., PASQUIER, F., COTTEL, D., FRIGARD, B., AMOUYEL, P. & CHARTIER-HARLIN, M-C. (1998). A new polymorphism in the APOE promoter associated with risk of developing Alzheimer's disease, *Human Molecular Genetics*, 7, 533–540.
- LEHMANN, D.J., JOHNSTON, C. & SMITH A.D. (1997). Synergy between the genes for butyrylcholinesterase K variant and apolipoprotein E4 in late-onset confirmed Alzheimer's disease. *Human Molecular Genetics*, 6, 1933–1936.
- LEHTOVIRTA, M., HELISALMI, S., MANNERMAA, A., SOININEN, H., KOIVISTO, K., RYYNÄNEN, M. & RIEKKINEN SR. P. (1995). Apolipoprotein E polymorphism and Alzheimer's disease in Eastern Finland. *Neuroscience Letters*, 185, 13–15.
- LEMAIRE, J., SUBRAMANIAN, K., ARMSTRONG, K. & ASCH, D.A. (1999). Pricing term insurance in the presence of a family history of breast or ovarian cancer. Unpublished manuscript.
- LETENNEUR, L., COMMENGES, D., DARTIGUES, J.F. & BARBERGERGATEAU, P. (1994). Incidence of dementia and Alzheimer's disease in elderly community residents of south-western France. *International Journal of Epidemiology*, 23, 1256–1261.
- LEVY-LAHAD, E. & BIRD, T.D. (1996). Genetic factors in Alzheimer's disease: a review of recent advances. *Annals of Neurology*, **40**, 829–840.
- LIDDELL, M., WILLIAMS, J., BAYER, A., KAISER, F. & OWEN, M. (1994). Confirmation of association between the ε4 allele of apolipoprotein E and Alzheimer's disease. Journal of Medical Genetics, **31**, 197–200.
- LILIENFELD, D.E. & STOLLEY, P.D. (1994). Foundations of epidemiology. Oxford University Press.
- LOPEZ, O.L., LOPEZ-POUSA, S., KAMBOH, M.I., ADROER, R., OLIVA, R., LOZANO-GALLEGO, M., BECKER, T.B. & DEKOSKY, S.T. (1998).

Apolipoprotein E polymorphism in Alzheimer's disease: a comparative study of two research populations from Spain and the United States. *European Neurology*, **39**, 229–233.

- LUCOTTE, G., DAVID, F., LEGRAND, C., TURPIN, J-C. & CLAVEL, C. (1997). Predictive value of apolipoprotein E ε4 allele genotyping in Alzheimer's disease. Alzheimer's Research, 3, 7–9.
- MACDONALD, A.S. (1996a). An actuarial survey of statistical models for decrement and transition data I: Multiple state, poisson and binomial models. *British Actuarial Journal*, 2,129–155.
- MACDONALD, A.S. (1996b). An actuarial survey of statistical models for decrement and transition data III: Counting process models. *British Actuarial Journal*, 2,703–726.
- MACDONALD, A.S. (1997). How will improved forecasts of individual lifetimes affect underwriting?. *Philosophical Transactions of the Royal Society* B, **352**,1067– 1075, and (with discussion) *British Actuarial Journal*, **3**, 1009–1025 and 1044– 1058.
- MACDONALD, A.S., CAIRNS, A.J.G., GWILT, P.L. & MILLER, K.A. (1998). An international comparison of recent trends in population mortality. *British Actuarial Journal*, 4, 3–141.
- MACDONALD, A.S. (1999). Modeling the impact of genetics on insurance. North American Actuarial Journal, **3:1**,83–101.
- MACDONALD, A.S. & PRITCHARD, D.J. (2000). A mathematical model of Alzheimer's disease and the ApoE gene. *Astin Bulletin*, **30**, 69–110.
- MACDONALD, A.S. & PRITCHARD, D.J. (2001). Genetics, Alzheimer's disease and long-term care insurance. North American Actuarial Journal, 5, 54–78.
- MACDONALD, A.S., WATERS, H.R. & WEKWETE, C.T. (2000). The genetics of breast of ovarian cancer I: a model of family history. To appear in *Scandinavian Actuarial Journal*.
- MACDONALD, A.S., WATERS, H.R. & WEKWETE, C.T. (2000). The genetics of breast of ovarian cancer II: a model of critical illness insurance. To appear in

Scandinavian Actuarial Journal.

- MASSULLO, C., DANIELE, A., SERIPA, D., FILIPPINI, V., GRAVINA, C., CAR-BONE, G., GAINOTTI, G. & FAZIO, V.M. (1998). Apolipoprotein E genotype in sporadic early- and late-onset Alzheimer's disease. *Dementia and Geriatric Cognitive Disorders*, 9, 121–125.
- MANTON, K.G. (1988). A longitudinal study of functional change and mortality in the United States. *Journal of Gerontology: SOCIAL SCIENCES*, **43**, S153–S161.
- MANTON, K.G., CORDER, L.S. & STALLARD, E. (1993). Estimates of change in chronic disability and institutional incidence and prevalence rates in the U.S. elderly population from the 1982, 1984 and 1989 National Long Term Care Survey. *Journal of Gerontology: SOCIAL SCIENCES*, 48,S153–S166.
- MAYEUX, R., STERN, Y., OTTMAN, R., TATEMICHI, T.K., TANG, M-X., MAESTRE, G., NGAI, C., TYCKO, B. & GINSBERG, H. (1993). The apolipoprotein ε4 allele in patients with Alzheimer's disease. Annals of Neurology, 34, 752–754.
- MÖLSÄ, P.K., MARTTILA, R.J. & RINNES, U.K. (1986). Survival and cause of death in Alzheimer's disease and multi-infarct dementia. Acta Neurologica Scandinavica, 74, 103–107.
- MORGAN, B.J.T. (2000). Applied stochastic modelling. Arnold.
- MYERS, R.H., SCHAEFFER, E.J., WILSON, P.W.F., D'AGOSTINO, R. OR-DOVAS, J.M., ESPINO, A., WHITE, R.F., KNOEFEL, J.E., COBB, J.L., MC-CNULTY, K.A., BEISER, A. & WOLF, P.A. (1996). Apolipoprotein E \$\varepsilon4\$ association with dementia in a population based study: the Framlingham study. *Neurology*, 46, 673–677.
- NALBANTOGLU, J., GILFIX, B.M., BERTRAND, P., ROBITAILLE, Y., GAUTHIER, S., ROSENBLATT, D.S. & POIRIER, J. (1994). Predictive value of apolipoprotein E genotyping in Alzheimer's disease: results of an autopsy series and an analysis of several combined studies. *Annals of Neurology*, **36**, 889–895.
- NILSSON, L.V. (1984). Incidence of severe dementia in an urban sample followed from 70 to 79 years of age. Acta Psychiatrica Scandinavica, 70, 478–486.

- NORBERG, R. (1995) Differential equations for moments of present values in life insurance. *Insurance: Mathematics & Economics*, **17**, 171–180.
- NORRMAN, J., BROOKES, A.J., YATES, C. & CLAIR, D.S. (1995). Apolipoprotein E genotype and its effect on duration and severity of early and late onset Alzheimer's disease. *British Journal of Psychiatry*, **167**, 533–536.
- O'CONNOR, D.W., POLLITT, P.A., HYDE, J.B. et al. (1989). The prevalence of dementia as measured by the Cambridge Mental Disorders of the Elderly Examination. Acta Psychiatrica Scandinavica, 79, 190–198.
- OTT, A., BRETELER, M.M.B., VAN HARSKAMP, F., STIJNEN, T. & HOFMAN, A. (1998). Incidence and risk of dementia. *American Journal of Epidemiology*, 147, 574–580.
- PAYAMI, H., SCHELLENBERG, G.D., ZAREPARSI, S., KAYE, J., SEXTON, G.J., HEAD, M.A., MATSUYAMA, S.S., JARVIK, L.F., MILLER, B., MCMANUS, D.Q., BIRD, T.D., KATZMAN, R., HESTON, L., NORMAN, D. & SMALL, G.W. (1997). Evidence for association of HLA-A2 allele with onset age of Alzheimer's disease. *Neurology*, 48, 512–518.
- PERICAK-VANCE, M.A., BEBOUT, J.L., GASKELL, P.C., YAMAOKA, L.H., HUNG, W.Y., ALBERTS, M.J., WALKER, A.P., BARTLETT, R.J., HAYNES, C.A., WELSH, K.A., EARL, N.L., HEYMAN, A., CLARK, C.M. & ROSES, A.D. (1991). Linkage studies in familial Alzheimer's disease: evidence for chromosome 19 linkage. American Journal of Human Genetics, 48,1034–1050.
- PFEFFER, R.I., AFIFI, A.A. & CHANCE J.M. (1987). Prevalence of Alzheimer's disease in a retirement community. *American Journal of Epidemiology*, **125**, 420–436.
- PIANTADOSI, S. (1997). *Clinical trials: a methodologic perspective*. John Wiley and Sons, New York.
- PODUSLO, S.E., NEAL, M., HERRING, K. & SHELLY, J. (1998). The apolipoprotein CI A allele as a risk factor for Alzheimer's disease. *Neurochemical Research*, 23, 361–367.
- POIRIER, J., DAVIGNON, J., BOUTHILLIER, D., KOGAN, S., BERTRAND, P. & GAUTHIER, S. (1993). Apolipoprotein E polymorphism and Alzheimer's disease.

Lancet, **342**, 697–699.

- POLAK, E. (1971). Computational methods in optimization. New York: Academic Press.
- PRESS, W.H., TEUKOLSKY, S.A., VETTERLING, W.T. & FLANNERY, B.P. (1993) Numerical recipes in C: The art of scientific computing. *Cambridge University Press*, 2nd edition.
- PRITCHARD, D.J. (1997). Life assurance: financial implications of a change in insuring behaviour resulting from individuals' increased knowledge of their genetic predispositions. M.Sc. dissertation, Heriot-Watt University, Edinburgh.
- ROCCA, W.A., CHA, R.H., WARING, S.C. & KOKMEN, E. (1998). Incidence of dementia and Alzheimer's disease: a reanalysis of data from Rochester, Minnesota, 1975–1984. American Journal of Epidemiology, 148, 51–62.
- RORSMAN, B., HAGNELL, O. & LANKE, J. (1986). Prevalence and incidence of senile and multi-infarct dementia in the Lundby study: a comparison between the time periods 1947–1957 and 1957–1972. Neuropsychobiology, 15, 122–129.
- ROSES, A.D. (1995). Apolipoprotein E genotyping in the differential diagnosis, not prediction, of Alzheimer's disease. Annals of Neurology, 38, 6–14.
- SAYETTA, R.B. (1986). Rates of senile dementia—Alzheimer's type in the Baltimore longitudinal study. *Journal of Chronic Disabilities*, **39**, 271–286.
- SELVIN, S. (1996). Statistical analysis of epidemiologic data. Oxford University Press.
- SEVERSON, M.A., SMITH, G.E., TANGALOS, E.G., PETERSON, R.C. KOKMEN, E., IVNIK, R.J., ATKINSON, E.J. & KURLAND, L.T. (1994). Patterns and predictors of institutionalization in community-based dementia patients. *Journal* of the American Geriatrics Society, 42, 181–185.
- SINGLETON, A., SMITH, G., GIBSON, A.M., WOODWARD, R., PERRY, R.H., INCE, P.G., EDWARDSON, J.A. & MORRIS, C.M. (1998). No association between the K variant of the butyrylcholinesterase gene and pathologically confirmed Alzheimer's disease. *Human Molecular Genetics*, 7, 937–939.

- SLOOTER, A.J.C. & VAN DUIJN, C.M. (1997). Genetic epidemiology of Alzheimer disease. *Epidemiologic Reviews*, **19**,107–119.
- SLOOTER, A.J.C., CRUTS, M., KALMIJN, S., HOFMAN, A., BRETELER, M.M.B., VAN BROECKHOVEN, C. & VAN DUIJN, C.M. (1998). Risk estimates of dementia by apolipoprotein E genotype from a population-based incidence study: the Rotterdam study. Archives of Neurology, 55,964–968.
- STERN, G., TANG, M.X., DENARO, J. & MAYEUX, R. (1995). Increased risk of mortality in Alzheimer's disease patients with more advanced educational and occupational attainment. *Annals of Neurology*, **37**, 590–595.
- STERN, Y., BRANDT, J., ALBERT, M., JACOBS, D.M., LIU, X., BELL, K., MARDER, K., SANO, M., ALBERT, S., CASTENADA, C.D-C., BYLSMA, F., TYCKO, B. & MAYEUX, R. (1997). The absence of an apolipoprotein ε4 allele is associated with a more aggressive form of Alzheimer's disease. Annals of Neurology, 41, 615–620.
- STRACHAN, T. & READ, A.P. (1996). Human Molecular Genetics. BIOS Scientific Publishers, Oxford.
- STRITTMATTER, W.J., SAUNDERS, A.M., SCHMECHEL, PERICAK-VANCE, M., ENGHILD, J., SALVESEN, G.S. & ROSES, A.D. (1993). Apolipoprotein E: High-avidity binding to β-amyloid and increased frequency of type-4 allele in late-onset Alzheimer's disease. Proceedings of the National Academy of Sciences of the United States of America, 90, 1977–1981.
- SUBRAMANIAN, K., LEMAIRE, J., HERSHEY, J.C., PAULY, M.V., ARMSTRONG, K. & ASCH, D.A. (2000). Estimating adverse selection costs from genetic testing for breast and ovarian cancer: The case of life insurance. *Journal of Risk and Insurance*, 66, 531–550.
- SUTHERLAND, S. (CHAIRMAN) (1999). With respect to old age: long-term care rights and responsibilities: a report by the royal commission on long-term care. Stationary Office.
- SVERDRUP, E. (1965), Estimates and test procedures in connection with stochastic models for death, recoveries and transfers between states of health.*Skandinavisk Aktuaritidskriff*, 48, 184–211.

- TAN, K.W. (1997). The financial impact of genetic testing on annuities. M.Sc. dissertation, Heriot-Watt University, Edinburgh.
- THATCHER, A.R., KANNISTO, V. & VAUPEL, J.W. (1998). The force of mortality at ages 80 to 120. *Odense University Press*.
- TIERNEY, M.C., FISHER, R.H., LEWIS, A.J., ZORZITTO, M.L., SNOW, W.G., REID, D.W. & NIEUWSTRATEN, P. (1988). The NINCDS–ADRDA work group criteria for the clinical diagnosis of probable Alzheimer's disease: a clinicopathologic study of 57 cases. *Neurology*, 38,359–364.
- TREVES, T., KORCZYN, A.D., ZILBER, N., KAHANA, E., LEIBOWITZ, Y., AL-TER, M. & SCHOENBERG, B.S. (1986). Presenile dementia in Israel. Archives of Neurology, 43, 26–29.
- TSAI, M-S., TANGALOS, E.G., PETERSEN, R.C., SMITH, G.E., SCHAID, D.J., KOKMEN, E., IVNIK, R.J. & THIBODEAU, S.N. (1994). Apolipoprotein E: risk factor for Alzheimer's disease. *American Journal of Human Genetics*, 54, 643-649.
- WALSH, J.S., WELCH, H.G. & LARSON, E.B. (1990). Survival of outpatients with Alzheimer-type dementia. Annals of Internal Medicine, **113**, 429–434.
- WARREN, V., BRETT, P., BRAYNE, C., MACDONALD, A.S., PLUMB, R.H. & READ, A.P. (1999). Genetic tests and future need for long-term care in the UK: report of a Work Group of the Continuing Care Conference Genetic Tests and Long-term Care Study Group. Continuing Care Conference, London.
- WATERS, H.R. (1984). An approach to the study of multiple state models. *Journal* of the Institute of Actuaries, **111**,363–374.
- WATERS, H.R. & WILKIE, A.D. (1987). A short note on the construction of life tables and multiple decrement tables. *Journal of the Institute of Actuaries*,, 114,569–580.
- WATERS, H.R. (1991). Computational procedures for the model. Continuous Mortality Investigation Bureau Report, 12,79–96.
- WATSON, R. (1998). Alzheimer's disease and genetic testing. On The Risk, 14:2,57–62.

### Appendix A

# Overview of the 1982, 1984, 1989 and 1994 National Long-Term Care Surveys — transitions between states

The tables in this appendix summarise the number of transitions between the classification of Figure 3.20. The first table, Table A.87 is the key for the following 3 tables — it defines what classifications the letters A–K represent in each survey year. Take, for example, the classification '1994 aged-in population', then from Table A.87:

- in the 1982 survey, the letter K is used to represent '1994 aged-in population' (lives aged 65–70 years in 1994 and lives aged over 95 years in 1994) in Table A.88;
- in 1984, the letter H is used (in Tables A.88 and A.89) for the same group of lives;
- 3. in 1989, for the 1994 aged-in lives, J1 and J2 are used (in Table A.90) to represent lives aged 65–70 years and those over 95 years in 1994, respectively, and J is used to represent both (in Table A.89); and

			Classif in surv	fication ev vear:	
Classi	fication	1982	1984	1989	1994
Screener only	complete	А	А	А	А
	non-response dead	В		В	В
	non-response other	$\mathbf{C}$	$C^{(1)}$	$\mathbf{C}$	$\mathbf{C}$
Community detail	complete	D	В	D	D
	non-response dead	$\mathbf{E}$		$\mathbf{E}$	$\mathbf{E}$
	non-response other	F	$C^{(1)}$	$\mathbf{F}$	$\mathbf{F}$
Institutionalized	before 1 April 1982	G			
	after 1 April 1982	Η			
	complete		D	G	G
	non-response dead			Η	Η
	non-response other		$\mathbf{E}$	Ι	Ι
Dead			F		
1984 aged-in popula	ation	Ι			
1989 aged-in popula	ation	J	G		
1994 aged-in popula	ation 65-70 yrs:	Κ	Η	$J1^{(2)}$	
1994 aged-in popula	Κ	Η	$J2^{(2)}$		
Not in Survey Year				Κ	J

Table A.87: Key to classifications in each Nation Long-Term Care Survey Year.

(1) Non-responders to the screener questionnaire and the community detail questionnaire in 1984 are grouped together in a single classification due to data not being available in 1982.

(2) Classification J is used in Table A.89 to represent both J1 and J2 in 1989.

4. in 1994, this group of lives is part of the survey and are included in the other states.

1982											
$\mathrm{Status}^{(1)}$		$1984 \text{ Status}^{(1)}$									
	А	В	$\mathbf{C}$	D	$\mathbf{E}$	$\mathbf{F}$	G	Η	Total		
А	9404	1091	194	152	6	723	0	0	11570		
В	6	0	4	2	0	203	0	0	215		
$\mathbf{C}$	113	29	106	20	3	44	0	0	315		
D	0	4182	210	414	22	1260	0	0	6088		
Ε	0	0	0	0	0	67	0	0	67		
F	0	92	46	41	3	56	0	0	238		
G	0	60	7	908	45	688	0	0	1708		
Η	0	40	4	114	4	122	0	0	284		
Ι	4263	440	118	39	0	56	0	0	4916		
J	0	0	0	0	0	0	4907	0	4907		
Κ	0	0	0	0	0	0	0	5540	5540		
Total	13786	5934	689	1690	83	3219	4907	5540	35848		

Table A.88: Transitions between classifications in the 1982 and 1984 National Long-Term Care Surveys.

(1) See Table A.87 for description of classifications.

Table A.89: Transitions between classifications in the 1984 and 1989 National Long-Term Care Surveys.

1984												
$\mathrm{Status}^{(1)}$						1989 S	$tatus^{(1)}$	)				
	А	В	$\mathbf{C}$	D	$\mathbf{E}$	$\mathbf{F}$	G	Η	Ι	J	Κ	Total
А	6038	195	139	1105	47	77	277	7	5	0	5896	13786
В	8	191	89	2771	60	126	503	35	5	0	2146	5934
$\mathbf{C}$	90	25	51	99	4	34	32	3	4	0	351	689
D	1	73	9	19	9	3	476	23	1	0	1073	1690
Ε	0	7	1	2	0	3	11	0	0	0	58	83
$\mathbf{F}$	0	0	0	0	0	0	0	0	0	0	3219	3219
G	4193	38	101	467	11	40	55	0	2	0	0	4907
Η	0	0	0	0	0	0	0	0	0	5540	0	5540
Total	10330	529	390	4463	131	283	1354	68	17	5540	12743	35848

(1) See Table A.87 for description of classifications.

Table A.90: Transitions between classifications in the 1989 and 1994 National Long-Term Care Surveys.

1989											
$\mathrm{Status}^{(1)}$				1994 S	tatus	(1)					
	А	В	$\mathbf{C}$	D	$\mathbf{E}$	$\mathbf{F}$	G	Η	Ι	J	Total
А	4315	202	211	1474	23	195	304	8	8	3590	10330
В	13	18	15	9	0	1	6	2	0	465	529
$\mathbf{C}$	88	10	46	47	1	6	26	0	0	166	390
D	0	202	152	1800	23	123	392	8	1	1762	4463
$\mathbf{E}$	0	0	0	0	0	0	0	0	0	131	131
$\mathbf{F}$	0	0	0	0	0	0	0	0	0	283	283
G	0	45	21	13	3	15	290	7	1	959	1354
Η	0	0	0	0	0	0	0	0	0	68	68
Ι	0	0	0	0	0	0	0	0	0	17	17
J1	3692	57	202	872	11	114	51	0	1	0	5000
J2	58	115	70	127	9	14	139	6	2	0	540
Κ	2308	77	134	747	11	105	122	1	2	9236	12743
Total	10474	726	851	5089	81	573	1330	32	15	16677	35848

(1) See Table A.87 for description of classifications.

## Appendix B

# Numbers of transitions between disability states of the 1982 and 1984 NLTCS by gender and age group

The tables in this appendix give the number of transitions between disability states for the 1982–84 NLTCS for separate genders. This data has been adjusted for censored data (see Section 3.7), which is why the numbers of transition are not integer numbers of lives. Tables B.91 and B.92 give the data split into 10 year age groups (65–74 years, 75–84 years and 85+ years) for females and males, respectively. Tables B.93 and B.94 give the data split into 5 year age groups (65–69 years, 70–74 years, 75–79 years, 80–84 years and 85+ years) for females and males, respectively. The same data over all ages (>65 years) are given in Table 3.36.

Table B.91: Transitions for females between disability states in the 1982 and 1984 National Long-Term Care Surveys using 10 year age groupings, adjusted for censored data.

				19	84 Statu	S		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74	4140.92	155.90	111.18	36.52	32.94	33.90	168.27
	75 - 84	1587.47	134.50	124.87	41.47	29.08	72.36	186.36
	85 +	193.85	33.63	46.00	8.90	16.68	40.39	65.95
	All	5922.24	324.03	282.05	86.89	78.70	146.65	420.58
IADL only	65 - 74	130.90	161.26	98.24	27.18	14.29	22.41	43.20
	75 - 84	56.67	141.95	120.51	20.94	17.21	37.52	72.72
	85 +	3.71	44.18	40.23	12.10	12.20	18.91	34.29
	All	191.28	347.39	258.98	60.22	43.70	78.84	150.21
1-2 ADLs	65 - 74	68.91	75.38	185.35	48.25	24.36	24.55	61.44
	75 - 84	48.26	71.79	210.54	71.39	32.72	58.52	103.00
	85 +	9.22	23.75	94.93	43.42	26.55	38.64	60.53
	All	126.39	170.92	490.82	163.06	83.63	121.71	224.97
3-4 ADLs	65 - 74	11.14	10.77	54.71	58.40	32.43	10.80	23.43
	75 - 84	7.69	9.83	49.07	55.40	49.52	30.08	48.66
	85 +	1.75	5.21	9.25	29.11	33.17	23.55	35.47
	All	20.58	25.81	113.03	142.91	115.12	64.43	107.56
5-6 ADLs	65 - 74	11.66	11.84	15.80	18.44	62.47	15.88	57.36
	75 - 84	10.24	8.90	16.16	23.44	79.56	32.16	71.58
	85 +	2.99	1.37	10.45	15.20	51.26	25.73	83.40
	All	24.89	22.11	42.41	57.08	193.29	73.77	212.34
Inst'd	65 - 74	8.77	1.47	2.19	4.21	2.22	124.37	45.30
	75 - 84	5.89	4.47	3.66	3.16	5.30	315.71	171.67
	85 +	3.04	2.24	0.29	5.13	4.21	307.78	271.36
	All	17.70	8.18	6.14	12.50	11.73	747.86	488.33

Table B.92: Transitions for males between disability states in the 1982 and 1984 National Long-Term Care Surveys using 10 year age groupings, adjusted for censored data.

				19	84 Statu	s		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74	3097.98	86.51	65.45	25.42	27.05	28.25	283.24
	75 - 84	1022.67	70.83	56.86	12.75	20.61	27.09	192.56
	85 +	86.24	17.71	19.55	6.39	3.43	8.20	45.29
	All	4206.89	175.05	141.86	44.56	51.09	63.54	521.09
IADL only	65 - 74	98.37	101.92	40.22	4.68	14.65	7.44	57.85
	75 - 84	35.84	74.81	33.89	13.40	22.53	18.03	51.76
	85 +	1.47	23.05	9.15	5.03	3.03	4.23	22.56
	All	135.68	199.78	83.26	23.11	40.21	29.70	132.17
1-2 ADLs	65 - 74	33.77	28.69	88.94	41.57	21.51	15.19	63.20
	75 - 84	14.44	23.50	68.50	20.23	19.34	6.67	86.11
	85 +	5.93	4.61	20.85	11.34	15.38	17.91	37.67
	All	54.14	56.80	178.29	73.14	56.23	39.77	186.98
3–4 ADLs	65 - 74	15.23	2.62	24.79	29.42	21.41	3.88	38.16
	75 - 84	3.06	4.24	8.24	17.11	15.16	9.31	36.12
	85 +	0.21	2.02	2.07	6.01	11.01	7.10	24.70
	All	18.50	8.88	35.10	52.54	47.58	20.29	98.98
5-6 ADLs	65 - 74	13.15	8.61	17.79	18.41	53.41	10.87	64.06
	75 - 84	4.92	7.47	6.54	12.24	41.30	12.59	67.02
	85 +	0.20	0.02	2.06	3.01	12.01	3.10	28.66
	All	18.27	16.10	26.39	33.66	106.72	26.56	159.74
Inst'd	65 - 74	7.17	1.25	3.36	1.26	2.21	71.54	45.48
	75 - 84	2.91	2.20	1.17	2.08	1.15	85.29	86.42
	85 +	0.34	0.03	0.11	0.02	0.02	51.17	83.12
	All	10.42	3.48	4.64	3.36	3.38	208.00	215.02

Table B.93: Transitions for females between disability states in the 1982 and 1984 National Long-Term Care Surveys using 5 year age groupings, adjusted for censored data.

				19	984 Statu	S		
1982 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	2375.92	67.90	59.50	14.26	13.74	9.05	76.97
	70 - 74	1765.01	88.00	51.69	22.26	19.20	24.85	91.29
	75 - 79	1101.46	86.92	67.02	22.80	16.66	32.24	98.71
	80-84	486.01	47.58	57.85	18.67	12.42	40.12	87.66
	85 +	193.85	33.63	46.00	8.90	16.68	40.39	65.95
	All	5922.25	324.03	282.06	86.89	78.70	146.65	420.58
IADL only	65 - 69	68.46	70.88	38.76	15.43	5.38	5.82	16.65
-	70 - 74	62.43	90.38	59.48	11.75	8.91	16.60	26.56
	75 - 79	36.59	76.10	58.50	12.56	7.70	19.63	37.99
	80 - 84	20.08	65.86	62.01	8.38	9.50	17.90	34.73
	85 +	3.71	44.18	40.23	12.10	12.20	18.91	34.29
	All	191.27	347.40	258.98	60.22	43.69	78.86	150.22
1-2 ADLs	65 - 69	39.58	37.14	89.06	22.57	8.53	6.11	26.77
	70 - 74	29.33	38.24	96.29	25.68	15.83	18.43	34.67
	75 - 79	30.31	36.43	116.92	35.74	15.91	26.05	45.14
	80 - 84	17.95	35.36	93.62	35.65	16.81	32.48	57.86
	85 +	9.22	23.75	94.93	43.42	26.55	38.64	60.53
	All	126.39	170.92	490.82	163.06	83.63	121.71	224.97
3–4 ADLs	65 - 69	4.62	6.30	32.23	27.14	11.11	4.26	10.73
	70 - 74	6.52	4.47	22.48	31.26	21.32	6.54	12.70
	75 - 79	4.34	5.50	30.68	27.25	23.32	14.73	26.89
	80 - 84	3.35	4.34	18.40	28.14	26.20	15.35	21.77
	85 +	1.75	5.21	9.25	29.12	33.17	23.55	35.47
	All	20.58	25.82	113.04	142.91	115.12	64.43	107.56
5-6 ADLs	65 - 69	6.82	5.46	8.42	6.23	33.21	4.44	21.70
	70 - 74	4.84	6.38	7.38	12.21	29.26	11.44	35.66
	75 - 79	5.02	4.33	7.47	10.16	44.22	15.53	32.85
	80 - 84	5.23	4.57	8.69	13.28	35.34	16.63	38.74
	85 +	2.99	1.37	10.45	15.20	51.26	25.73	83.40
	All	24.90	22.11	42.41	57.08	193.29	73.77	212.35
Inst'd	65 - 69	6.18	0.23	2.05	2.08	0.02	58.11	8.18
	70 - 74	2.59	1.24	0.14	2.13	2.19	66.26	37.12
	75 - 79	2.52	2.23	1.40	1.09	4.17	118.51	65.43
	80-84	3.37	2.23	2.26	2.06	1.13	197.19	106.24
	85 +	3.04	2.24	0.29	5.13	4.21	307.78	271.36
	All	17.70	8.17	6.14	12.49	11.72	747.85	488.33

Table B.94: Transitions for males between disability states in the 1982 and 1984 National Long-Term Care Surveys using 5 year age groupings, adjusted for censored data.

				19	84 Statu	S		
1982 Status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	1853.76	45.23	30.64	13.47	12.66	13.18	112.97
-	70 - 74	1244.21	41.28	34.80	11.95	14.39	15.07	170.27
	75 - 79	732.68	42.10	30.58	8.17	12.97	14.17	123.49
	80-84	289.98	28.73	26.29	4.58	7.64	12.92	69.07
	85 +	86.24	17.71	19.55	6.39	3.43	8.20	45.29
	All	4206.87	175.05	141.86	44.56	51.09	63.54	521.09
IADL only	65 - 69	51.64	52.49	22.68	2.36	11.38	2.75	26.65
Ŭ	70 - 74	46.73	49.44	17.54	2.33	3.27	4.69	31.20
	75 - 79	27.70	51.62	19.77	8.35	14.47	11.83	34.60
	80 - 84	8.14	23.19	14.12	5.06	8.06	6.21	17.16
	85 +	1.47	23.05	9.15	5.03	3.03	4.23	22.56
	All	135.68	199.79	83.26	23.13	40.21	29.71	132.17
1-2 ADLs	65 - 69	18.92	11.25	40.40	15.23	13.25	4.48	26.77
	70 - 74	14.85	17.44	48.54	26.34	8.27	10.70	36.44
	75 - 79	9.63	17.23	34.28	9.13	13.23	0.36	43.79
	80 - 84	4.81	6.28	34.22	11.10	6.11	6.31	42.32
	85 +	5.93	4.61	20.85	11.34	15.38	17.91	37.67
	All	54.14	56.81	178.29	73.14	56.24	39.76	186.99
3–4 ADLs	65 - 69	7.00	1.28	13.38	10.19	10.21	3.41	19.73
	70 - 74	8.23	1.34	11.41	19.23	11.21	0.48	18.43
	75 - 79	2.83	1.19	4.23	12.10	8.15	6.26	23.16
	80 - 84	0.23	3.05	4.01	5.01	7.01	3.05	12.97
	85 +	0.21	2.02	2.07	6.01	11.01	7.10	24.70
	All	18.50	8.88	35.10	52.54	47.59	20.30	98.99
5–6 ADLs	65 - 69	5.21	4.44	6.57	9.28	23.30	5.59	30.70
	70 - 74	7.93	4.18	11.22	9.13	30.11	5.28	33.36
	75 - 79	2.81	3.19	2.23	5.10	24.15	7.26	33.10
	80 - 84	2.11	4.29	4.31	7.14	17.15	5.33	33.92
	85 +	0.20	0.02	2.06	3.01	12.01	3.10	28.66
	All	18.26	16.12	26.39	33.66	106.72	26.56	159.74
Inst'd	65 - 69	4.41	1.14	0.23	1.14	0.15	43.29	13.18
	70 - 74	2.76	0.11	3.13	0.12	2.06	28.25	32.30
	75 - 79	2.55	2.13	1.15	1.07	0.14	47.21	41.87
	80-84	0.37	0.08	0.02	1.01	1.01	38.08	44.56
	85 +	0.34	0.03	0.11	0.02	0.02	51.17	83.12
	All	10.43	3.49	4.64	3.36	3.38	208.00	215.03

## Appendix C

# Numbers of transitions between disability states of the 1984 and 1989 NLTCS by gender and age group

The tables in this appendix give the number of transitions between disability states for the 1984–89 NLTCS for separate genders. This data has been adjusted for censored data (see Section 3.8), which is why the numbers of transition are not integer numbers of lives. Tables C.95 and C.96 give the data split into 10 year age groups (65–74 years, 75–84 years and 85+ years) for females and males, respectively. Tables C.97 and C.98 give the data split into 5 year age groups (65–69 years, 70–74 years, 75–79 years, 80–84 years and 85+ years) for females and males, respectively. The same data over all ages (>65 years) are given in Table 3.38.

Table C.95: Transitions for females between disability states in the 1984 and 1989 National Long-Term Care Surveys using 10 year age groupings, adjusted for censored data.

				19	89 Statu	s		
1984 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74	2662.36	146.43	157.67	61.05	51.95	82.55	47.63
	75 - 84	1220.66	119.35	191.79	81.05	64.35	145.00	73.63
	85 +	108.62	19.10	27.38	21.17	16.83	59.09	20.63
	All	3991.64	284.88	376.84	163.27	133.13	286.64	141.89
IADL only	65 - 74	88.12	76.71	90.55	22.33	15.69	24.29	12.40
	75 - 84	31.47	68.44	74.90	34.47	22.82	48.85	19.40
	85 +	2.95	16.62	22.91	7.69	14.62	25.47	17.40
	All	122.54	161.77	188.36	64.49	53.13	98.61	49.20
1-2 ADLs	65 - 74	47.71	28.55	145.74	53.38	22.89	31.33	12.44
	75 - 84	27.85	28.92	105.18	69.75	35.86	86.31	20.44
	85 +	5.20	6.13	31.69	26.14	16.99	42.37	21.44
	All	80.76	63.60	282.61	149.27	75.74	160.01	54.32
3-4 ADLs	65 - 74	11.04	10.67	31.50	49.49	23.21	14.82	11.17
	75 - 84	5.63	4.05	16.67	39.06	18.78	34.65	15.17
	85 +	2.77	1.50	5.75	12.49	9.43	17.02	12.17
	All	19.44	16.22	53.92	101.04	51.42	66.49	38.51
5-6 ADLs	65 - 74	6.32	4.62	18.89	20.59	26.46	18.05	4.13
	75 - 84	6.27	2.45	8.74	8.52	32.35	17.67	14.13
	85 +	0.91	0.28	2.42	3.29	19.26	19.61	11.13
	All	13.50	7.35	30.05	32.40	78.07	55.33	29.39
Inst'd	65 - 74	4.23	0.64	3.89	1.66	2.50	92.09	10.23
	75 - 84	5.17	2.83	5.35	0.94	0.64	175.23	28.23
	85 +	2.92	0.69	1.03	0.68	2.58	129.40	42.23
	All	12.32	4.16	10.27	3.28	5.72	396.72	80.69
Table C.96: Transitions for males between disability states in the 1984 and 1989 National Long-Term Care Surveys using 10 year age groupings, adjusted for censored data.

				198	89 Statu	s		
1984 Status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74	2034.79	99.74	87.26	32.47	23.77	46.45	60.63
	75 - 84	731.01	65.01	60.97	38.94	31.30	66.01	76.63
	85 +	47.71	11.65	7.01	5.28	4.18	14.56	12.90
	All	2813.51	176.40	155.24	76.69	59.25	127.02	150.16
IADL only	65 - 74	45.00	43.48	33.87	12.06	10.81	16.28	12.40
	75 - 84	17.75	20.89	16.58	16.92	17.57	13.54	12.40
	85 +	2.19	2.27	3.41	6.30	3.27	7.61	7.32
	All	64.94	66.64	53.86	35.28	31.65	37.43	32.12
1-2 ADLs	65 - 74	27.75	23.97	50.58	27.45	18.13	12.09	8.44
	75 - 84	12.46	10.98	28.73	10.01	10.62	13.68	11.44
	85 +	0.47	3.06	4.09	5.14	5.13	7.27	6.35
	All	40.68	38.01	83.40	42.60	33.88	33.04	26.23
3-4 ADLs	65 - 74	7.14	4.56	20.69	16.39	11.30	9.85	5.17
	75 - 84	0.73	0.36	8.64	6.38	10.23	12.62	8.17
	85 +	0.18	0.02	0.03	2.05	0.05	4.11	3.14
	All	8.05	4.94	29.36	24.82	21.58	26.58	16.48
5-6 ADLs	65 - 74	4.18	4.29	9.31	7.18	25.13	8.42	4.13
	75 - 84	0.49	2.20	6.38	7.23	11.12	8.35	3.13
	85 +	0.14	0.02	0.03	1.04	4.04	4.08	2.11
	All	4.81	6.51	15.72	15.45	40.29	20.85	9.37
Inst'd	65 - 74	1.95	2.56	0.63	1.36	0.26	46.82	7.23
	75 - 84	2.17	0.16	0.36	1.23	0.08	33.29	20.23
	85 +	1.47	0.17	0.25	0.18	0.16	20.37	5.18
	All	5.59	2.89	1.24	2.77	0.50	100.48	32.64

Table C.97: Transitions for females between disability states in the 1984 and 1989 National Long-Term Care Surveys using 5 year age groupings, adjusted for censored data.

				19	989 Statu	S		
1984 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	1350.66	56.48	52.30	19.52	21.39	30.37	19.73
-	70 - 74	1311.70	89.95	105.38	41.52	30.56	52.19	27.90
	75 - 79	894.18	79.04	113.64	45.11	34.24	76.16	40.18
	80 - 84	326.47	40.31	78.15	35.94	30.10	68.83	33.45
	85 +	108.62	19.10	27.38	21.17	16.83	59.09	20.63
	All	3991.63	284.88	376.85	163.26	133.12	286.64	141.89
IADL only	65 - 69	49.34	34.31	45.32	9.97	6.74	10.88	4.08
Ŭ	70 - 74	38.78	42.40	45.23	12.36	8.95	13.41	8.32
	75 - 79	22.24	39.52	42.65	16.41	12.32	24.27	8.24
	80 - 84	9.23	28.92	32.25	18.06	10.50	24.58	11.16
	85 +	2.95	16.62	22.91	7.69	14.62	25.47	17.40
	All	122.54	161.77	188.36	64.49	53.13	98.61	49.20
1-2 ADLs	65 - 69	33.85	14.25	56.66	21.08	9.01	16.08	5.09
	70 - 74	13.86	14.30	89.09	32.30	13.87	15.25	7.35
	75 - 79	13.37	13.94	59.32	35.77	23.63	30.94	10.26
	80 - 84	14.48	14.98	45.86	33.98	12.23	55.37	10.18
	85 +	5.20	6.13	31.69	26.14	16.99	42.37	21.44
	All	80.76	63.60	282.62	149.27	75.73	160.01	54.32
3–4 ADLs	65 - 69	4.75	8.84	20.19	19.71	13.64	7.40	5.03
	70 - 74	6.29	1.83	11.31	29.78	9.57	7.42	6.14
	75 - 79	3.70	2.42	8.78	20.41	8.41	16.59	6.10
	80 - 84	1.93	1.62	7.89	18.65	10.37	18.06	9.07
	85 +	2.77	1.50	5.75	12.49	9.43	17.02	12.17
	All	19.44	16.21	53.92	101.04	51.42	66.49	38.51
5-6 ADLs	65 - 69	5.43	1.31	12.40	10.27	13.26	6.52	3.03
	70 - 74	0.89	3.30	6.49	10.32	13.20	11.53	1.11
	75 - 79	1.68	2.17	7.37	4.19	14.21	9.19	7.08
	80 - 84	4.59	0.28	1.37	4.33	18.15	8.48	7.05
	85 +	0.91	0.28	2.42	3.29	19.26	19.61	11.13
	All	13.50	7.34	30.05	32.40	78.08	55.33	29.40
Inst'd	65 - 69	2.36	0.31	1.33	1.27	1.29	31.50	4.05
	70 - 74	1.87	0.34	2.56	0.39	1.22	60.59	6.18
	75 - 79	2.17	0.30	4.64	0.33	0.35	77.33	11.14
	80 - 84	3.00	2.53	0.71	0.61	0.29	97.90	17.09
	85 +	2.92	0.69	1.03	0.68	2.58	129.40	42.23
	All	12.32	4.17	10.27	3.28	5.73	396.72	80.69

Table C.98: Transitions for males between disability states in the 1984 and 1989 National Long-Term Care Surveys using 5 year age groupings, adjusted for censored data.

				193	89 Statu	s		
1984 Status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	1080.98	45.91	49.76	15.29	13.73	15.06	29.90
-	70 - 74	953.81	53.82	37.50	17.18	10.04	31.38	30.73
	75 - 79	558.93	43.33	34.78	23.14	19.63	40.67	47.45
	80 - 84	172.08	21.68	26.20	15.80	11.67	25.34	29.18
	85 +	47.71	11.65	7.01	5.28	4.18	14.56	12.90
	All	2813.51	176.39	155.25	76.69	59.25	127.01	150.16
IADL only	65 - 69	33.15	20.82	20.34	4.69	5.57	6.54	6.32
	70 - 74	11.85	22.66	13.53	7.37	5.24	9.74	6.08
	75 - 79	13.52	12.78	10.45	7.90	10.55	5.41	6.16
	80 - 84	4.23	8.11	6.13	9.03	7.02	8.13	6.24
	85 +	2.19	2.27	3.41	6.30	3.27	7.61	7.32
	All	64.94	66.64	53.86	35.29	31.65	37.43	32.12
1-2 ADLs	65 - 69	12.28	15.83	21.37	14.70	9.58	5.58	3.35
	70 - 74	15.47	8.14	29.22	12.75	8.55	6.51	5.09
	75 - 79	7.44	6.64	17.26	8.80	8.45	6.17	7.18
	80 - 84	5.02	4.34	11.47	1.21	2.17	7.51	4.26
	85 +	0.47	3.06	4.09	5.14	5.13	7.27	6.35
	All	40.68	38.01	83.41	42.60	33.88	33.04	26.23
3–4 ADLs	65 - 69	4.48	1.40	9.65	6.34	6.28	2.75	2.14
	70 - 74	2.66	3.16	11.03	10.05	5.02	7.11	3.03
	75 - 79	0.47	0.17	6.36	4.24	7.12	5.32	5.07
	80 - 84	0.25	0.19	2.28	2.13	3.11	7.30	3.10
	85 +	0.18	0.02	0.03	2.05	0.05	4.11	3.14
	All	8.04	4.94	29.35	24.81	21.58	26.59	16.48
5–6 ADLs	65 - 69	2.44	2.03	3.08	3.03	10.02	4.11	2.11
	70 - 74	1.74	2.26	6.24	4.16	15.11	4.31	2.03
	75 - 79	0.41	2.16	4.33	3.22	10.11	5.30	3.05
	80 - 84	0.08	0.04	2.04	4.01	1.01	3.04	0.08
	85 +	0.14	0.02	0.03	1.04	4.04	4.08	2.11
	All	4.81	6.51	15.72	15.46	40.29	20.84	9.38
Inst'd	65 - 69	0.91	1.20	0.36	0.17	0.14	28.43	3.18
	70 - 74	1.04	1.36	0.27	1.19	0.12	18.39	4.05
	75 - 79	1.04	0.10	0.29	0.22	0.07	22.21	12.09
	80 - 84	1.13	0.06	0.07	1.01	0.01	11.08	8.14
	85 +	1.47	0.17	0.25	0.18	0.16	20.37	5.18
	All	5.59	2.89	1.24	2.77	0.50	100.48	32.64

## Appendix D

# Numbers of transitions between disability states of the 1989 and 1994 NLTCS by gender and age group

The tables in this appendix give the number of transitions between disability states for the 1989–94 NLTCS for separate genders. This data has been adjusted for censored data (see Section 3.9), which is why the numbers of transition are not integer numbers of lives. Tables D.99 and D.100 give the data split into 10 year age groups (65–74 years, 75–84 years and 85+ years) for females and males, respectively. Tables D.101 and D.102 give the data split into 5 year age groups (65–69 years, 70–74 years, 75–79 years, 80–84 years and 85+ years) for females and males, respectively. The same data over all ages (>65 years) are given in Table 3.40.

Table D.99: Transitions for females between disability states in the 1989 and 1994 National Long-Term Care Surveys using 10 year age groupings, adjusted for censored data.

		1994 Status							
1989 Status	Age		IADL	1 - 2	3 - 4	5-6			
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead	
Healthy	65 - 74	1984.58	104.13	158.11	51.75	42.10	81.23	39.00	
	75 - 84	1152.14	130.17	186.76	89.88	67.87	194.10	79.34	
	85 +	95.52	18.24	30.91	17.40	28.05	70.48	19.12	
	All	3232.24	252.54	375.78	159.04	138.02	345.81	137.47	
IADL only	65 - 74	45.95	36.45	29.03	16.19	9.06	7.45	5.00	
	75 - 84	35.54	31.36	47.38	18.99	17.88	33.48	19.17	
	85 +	5.34	9.78	13.18	13.80	7.71	17.84	6.23	
	All	86.83	77.59	89.59	48.98	34.65	58.77	30.39	
1-2 ADLs	65 - 74	36.27	21.85	55.56	40.51	20.34	12.09	8.00	
	75 - 84	35.99	23.34	72.78	63.82	34.65	60.33	27.23	
	85 +	6.48	8.15	19.74	17.16	12.03	42.67	15.31	
	All	78.75	53.33	148.09	121.49	67.02	115.09	50.54	
3–4 ADLs	65 - 74	9.61	4.52	10.20	22.28	9.13	16.68	6.00	
	75 - 84	7.83	6.39	18.99	32.17	22.11	35.64	10.10	
	85 +	7.58	1.67	9.01	4.65	10.58	27.47	17.13	
	All	25.02	12.57	38.20	59.11	41.82	79.78	33.23	
5-6 ADLs	65 - 74	4.15	1.55	3.79	6.46	19.41	5.95	6.00	
	75 - 84	4.03	1.69	3.98	9.58	21.55	20.31	11.05	
	85 +	3.42	0.82	2.24	2.75	11.66	11.64	6.07	
	All	11.61	4.06	10.00	18.79	52.62	37.90	23.13	
Inst'd	65 - 74	6.11	0.97	3.36	0.80	0.71	46.65	6.00	
	75 - 84	4.96	1.73	2.98	2.59	2.60	90.41	19.11	
	85 +	4.25	0.89	1.36	0.86	0.76	100.91	17.14	
	All	15.32	3.59	7.70	4.25	4.07	237.97	42.25	

Table D.100: Transitions for males between disability states in the 1989 and 1994 National Long-Term Care Surveys using 10 year age groupings, adjusted for censored data.

				199	94 Statu	s		
1989 Status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74	1490.55	74.51	79.05	33.55	30.03	44.28	57.78
	75 - 84	711.60	77.67	76.93	37.95	35.09	70.84	65.56
	85 +	57.55	3.21	13.62	3.07	8.71	27.60	19.78
	All	2259.69	155.39	169.61	74.57	73.83	142.72	143.12
IADL only	65 - 74	32.41	16.11	13.66	7.15	2.87	7.13	6.06
	75 - 84	11.84	20.75	12.05	7.67	4.65	10.57	7.11
	85 +	0.75	2.09	7.19	3.08	4.13	0.34	3.06
	All	45.00	38.95	32.90	17.90	11.64	18.04	16.23
1-2 ADLs	65 - 74	20.49	8.72	28.06	12.87	5.57	9.43	4.08
	75 - 84	7.53	4.72	12.97	9.64	10.64	22.57	12.15
	85 +	1.60	0.30	3.53	1.26	1.31	4.79	3.08
	All	29.62	13.74	44.56	23.78	17.52	36.79	19.31
3–4 ADLs	65 - 74	7.27	2.66	5.98	8.68	12.51	5.26	5.03
	75 - 84	2.69	1.18	0.23	6.16	6.18	10.44	7.07
	85 +	0.70	1.08	3.15	0.07	1.09	1.24	3.03
	All	10.66	4.92	9.36	14.91	19.79	16.95	15.13
5-6 ADLs	65 - 74	2.10	1.09	3.13	0.14	8.08	5.20	2.02
	75 - 84	3.53	0.15	1.20	2.13	10.14	6.33	4.04
	85 +	0.04	0.01	0.03	0.01	1.02	2.07	2.02
	All	5.67	1.25	4.36	2.28	19.24	13.60	8.07
Inst'd	65 - 74	3.61	1.45	0.66	0.51	0.35	22.88	3.04
	75 - 84	1.53	0.39	1.53	1.34	0.34	26.83	4.07
	85 +	1.23	0.49	0.78	0.43	0.41	11.02	6.04
	All	6.36	2.33	2.97	2.28	1.11	60.72	13.14

Table D.101: Transitions for females between disability states in the 1989 and 1994 National Long-Term Care Surveys using 5 year age groupings, adjusted for censored data.

	1994 Status								
1989 Status	Age		IADL	1 - 2	3 - 4	5-6			
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead	
Healthy	65 - 69	961.09	41.79	61.56	19.09	20.45	26.76	21.00	
Ŭ	70 - 74	1023.49	62.34	96.55	32.66	21.65	54.48	18.00	
	75 - 79	777.76	68.77	109.23	38.77	36.65	95.15	31.78	
	80 - 84	374.37	61.41	77.54	51.11	31.22	98.95	47.56	
	85 +	95.52	18.24	30.91	17.40	28.05	70.48	19.12	
	All	3232.24	252.54	375.78	159.04	138.02	345.81	137.47	
IADL only	65 - 69	29.66	16.60	6.97	8.53	7.46	2.12	0.00	
c c	70 - 74	16.29	19.85	22.06	7.66	1.59	5.33	5.00	
	75 - 79	23.81	20.18	23.68	10.05	10.99	10.27	8.06	
	80 - 84	11.73	11.18	23.70	8.94	6.89	23.21	11.11	
	85 +	5.34	9.78	13.18	13.80	7.71	17.84	6.23	
	All	86.83	77.59	89.59	48.98	34.65	58.77	30.39	
1-2 ADLs	65 - 69	15.11	5.27	18.49	12.24	6.21	2.51	2.00	
	70 - 74	21.16	16.57	37.07	28.27	14.13	9.58	6.00	
	75 - 79	26.53	12.34	41.88	43.20	17.14	17.59	13.08	
	80 - 84	9.46	10.99	30.91	20.63	17.51	42.74	14.15	
	85 +	6.48	8.15	19.74	17.16	12.03	42.67	15.31	
	All	78.75	53.33	148.09	121.49	67.02	115.09	50.54	
3–4 ADLs	65 - 69	3.22	3.42	1.67	7.37	2.32	7.78	2.00	
	70 - 74	6.39	1.10	8.53	14.91	6.81	8.89	4.00	
	75 - 79	4.88	4.54	10.75	17.48	12.46	13.04	6.03	
	80 - 84	2.96	1.85	8.24	14.69	9.64	22.59	4.07	
	85 +	7.58	1.67	9.01	4.65	10.58	27.47	17.13	
	All	25.02	12.57	38.20	59.11	41.82	79.78	33.23	
5-6 ADLs	65 - 69	1.18	0.20	0.33	2.18	8.16	4.38	2.00	
	70 - 74	2.98	1.35	3.46	4.28	11.25	1.57	4.00	
	75 - 79	2.40	0.37	2.52	4.33	10.31	6.70	3.02	
	80 - 84	1.63	1.32	1.46	5.25	11.24	13.61	8.04	
	85 +	3.42	0.82	2.24	2.75	11.66	11.64	6.07	
	All	11.61	4.06	10.00	18.79	52.62	37.90	23.13	
Inst'd	65 - 69	2.17	0.27	1.45	0.24	0.21	11.51	1.00	
	70 - 74	3.94	0.70	1.91	0.56	0.50	35.14	5.00	
	75 - 79	3.12	0.35	1.45	1.31	0.32	41.69	12.04	
	80 - 84	1.84	1.38	1.53	1.28	2.28	48.73	7.07	
	85 +	4.25	0.89	1.36	0.86	0.76	100.91	17.14	
	All	15.32	3.59	7.70	4.25	4.07	237.97	42.25	

Table D.102: Transitions for males between disability states in the 1989 and 1994 National Long-Term Care Surveys using 5 year age groupings, adjusted for censored data.

	1994 Status								
1989 Status	Age		IADL	1 - 2	3 - 4	5 - 6			
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead	
Healthy	65 - 69	745.64	26.24	30.27	11.57	12.19	17.92	15.00	
-	70 - 74	744.91	48.27	48.78	21.98	17.83	26.36	42.78	
	75 - 79	489.08	55.57	47.42	24.92	22.74	36.20	32.78	
	80 - 84	222.51	22.10	29.51	13.04	12.34	34.65	32.78	
	85 +	57.55	3.21	13.62	3.07	8.71	27.60	19.78	
	All	2259.69	155.39	169.61	74.57	73.83	142.72	143.12	
IADL only	65 - 69	11.64	6.27	3.32	3.24	0.16	1.45	3.00	
Ŭ	70 - 74	20.77	9.84	10.34	3.91	2.71	5.68	3.06	
	75 - 79	10.29	10.55	6.84	3.54	2.43	5.14	5.06	
	80 - 84	1.55	10.20	5.21	4.12	2.22	5.43	2.06	
	85 +	0.75	2.09	7.19	3.08	4.13	0.34	3.06	
	All	45.00	38.95	32.90	17.90	11.64	18.04	16.23	
1-2 ADLs	65 - 69	8.01	1.28	10.31	3.25	1.15	3.45	2.00	
	70 - 74	12.48	7.45	17.75	9.62	4.42	5.98	2.08	
	75 - 79	5.67	2.36	10.55	5.39	8.28	14.82	3.08	
	80-84	1.86	2.36	2.42	4.25	2.37	7.75	9.08	
	85 +	1.60	0.30	3.53	1.26	1.31	4.79	3.08	
	All	29.62	13.74	44.56	23.78	17.52	36.79	19.31	
3–4 ADLs	65 - 69	5.71	1.50	4.71	4.44	5.36	1.90	1.00	
	70 - 74	1.55	1.16	1.27	4.24	7.15	3.36	4.03	
	75 - 79	1.48	1.15	0.22	3.16	3.11	7.34	1.03	
	80 - 84	1.22	0.04	0.00	3.00	3.07	3.10	6.03	
	85 +	0.70	1.08	3.15	0.07	1.09	1.24	3.03	
	All	10.66	4.92	9.36	14.91	19.79	16.95	15.13	
5-6 ADLs	65 - 69	0.89	0.08	2.10	0.08	3.05	2.14	1.00	
	70 - 74	1.21	1.01	1.03	0.06	5.03	3.06	1.02	
	75 - 79	2.79	0.07	0.10	1.08	3.05	4.16	3.02	
	80 - 84	0.74	0.08	1.10	1.06	7.08	2.17	1.02	
	85 +	0.04	0.01	0.03	0.01	1.02	2.07	2.02	
	All	5.67	1.25	4.36	2.28	19.24	13.60	8.07	
Inst'd	65 - 69	0.53	1.04	0.01	0.04	0.01	10.05	1.00	
	70 - 74	3.08	0.41	0.65	0.46	0.35	12.83	2.04	
	75 - 79	0.50	0.15	1.22	0.17	0.11	12.35	2.04	
	80 - 84	1.03	0.24	0.31	1.18	0.23	14.48	2.04	
	85 +	1.23	0.49	0.78	0.43	0.41	11.02	6.04	
	All	6.36	2.33	2.97	2.28	1.11	60.72	13.14	

## Appendix E

# 5-year transition probabilities between disability states calculated from the 1984, 1989 and 1994 National Long-Term Care Surveys

The tables in this appendix give the transition probabilities between disability states as a percentage for men, women and in aggregate, calculated using Equation 3.32 and based on data grouped in 5-year age bands. They are given for the 1984–89 NLTCS in Tables E.103 to E.105 and for the 1989–94 NLTCS in Tables E.106 to E.108. Table E.103: The 5-year transition probabilities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys as a percentage, for males using 5 year age groupings.

				198	39 Status	3		
1984 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	86.43	3.67	3.98	1.22	1.10	1.20	2.39
	70 - 74	84.08	4.74	3.31	1.51	0.88	2.77	2.71
	75 - 79	72.78	5.64	4.53	3.01	2.56	5.30	6.18
	80 - 84	56.99	7.18	8.68	5.23	3.86	8.39	9.66
	85 +	46.19	11.28	6.79	5.11	4.05	14.09	12.49
IADL only	65 - 69	34.02	21.37	20.88	4.81	5.72	6.71	6.49
	70 - 74	15.50	29.63	17.69	9.64	6.85	12.74	7.95
	75 - 79	20.25	19.15	15.65	11.83	15.80	8.10	9.23
	80 - 84	8.66	16.59	12.53	18.46	14.36	16.63	12.76
	85 +	6.77	7.00	10.54	19.48	10.09	23.51	22.61
1-2 ADLs	65 - 69	14.85	19.15	25.84	17.78	11.59	6.74	4.05
	70 - 74	18.05	9.50	34.08	14.87	9.97	7.59	5.93
	75 - 79	12.01	10.73	27.86	14.21	13.64	9.97	11.58
	80 - 84	13.96	12.05	31.88	3.36	6.04	20.86	11.85
	85 +	1.50	9.70	12.98	16.30	16.28	23.08	20.16
3-4 ADLs	65 - 69	13.56	4.24	29.22	19.18	19.01	8.32	6.47
	70 - 74	6.32	7.51	26.23	23.90	11.92	16.89	7.21
	75 - 79	1.65	0.59	22.13	14.76	24.75	18.50	17.63
	80 - 84	1.39	1.05	12.41	11.61	16.93	39.74	16.88
	85 +	1.91	0.23	0.36	21.42	0.52	42.84	32.72
5-6 ADLs	65 - 69	9.10	7.58	11.48	11.29	37.37	15.32	7.86
	70 - 74	4.85	6.31	17.40	11.60	42.16	12.03	5.66
	75 - 79	1.42	7.57	15.15	11.26	35.37	18.55	10.68
	80 - 84	0.77	0.36	19.83	38.92	9.78	29.56	0.78
	85 +	1.26	0.15	0.24	9.09	35.24	35.63	18.39
Inst'd	65 - 69	2.64	3.50	1.04	0.49	0.41	82.67	9.25
	70 - 74	3.94	5.15	1.03	4.52	0.47	69.59	15.31
	75 - 79	2.89	0.27	0.80	0.61	0.19	61.67	33.57
	80 - 84	5.28	0.29	0.34	4.72	0.06	51.49	37.83
	85 +	5.28	0.59	0.92	0.66	0.58	73.31	18.65

Table E.104: The 5-year transition probabilities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys as a percentage, for females using 5 year age groupings.

				198	39 Status	3		
1984 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	87.11	3.64	3.37	1.26	1.38	1.96	1.27
	70 - 74	79.06	5.42	6.35	2.50	1.84	3.15	1.68
	75 - 79	69.72	6.16	8.86	3.52	2.67	5.94	3.13
	80 - 84	53.24	6.57	12.74	5.86	4.91	11.22	5.45
	85 +	39.81	7.00	10.04	7.76	6.17	21.66	7.56
IADL only	65 - 69	30.71	21.36	28.21	6.21	4.20	6.77	2.54
	70 - 74	22.89	25.02	26.69	7.29	5.28	7.92	4.91
	75 - 79	13.43	23.86	25.75	9.91	7.44	14.65	4.97
	80 - 84	6.85	21.47	23.94	13.41	7.80	18.25	8.28
	85 +	2.74	15.44	21.28	7.15	13.58	23.66	16.16
1-2 ADLs	65 - 69	21.70	9.13	36.31	13.51	5.78	10.31	3.26
	70 - 74	7.45	7.69	47.89	17.36	7.46	8.20	3.95
	75 - 79	7.14	7.44	31.68	19.11	12.62	16.52	5.48
	80 - 84	7.74	8.01	24.52	18.17	6.53	29.60	5.44
	85 +	3.47	4.09	21.13	17.43	11.33	28.26	14.30
3-4 ADLs	65 - 69	5.97	11.11	25.37	24.78	17.14	9.30	6.33
	70 - 74	8.70	2.53	15.64	41.16	13.22	10.26	8.48
	75 - 79	5.57	3.65	13.23	30.73	12.66	24.98	9.19
	80 - 84	2.85	2.40	11.67	27.59	15.35	26.72	13.42
	85 +	4.54	2.45	9.41	20.43	15.42	27.85	19.91
5-6 ADLs	65 - 69	10.40	2.52	23.74	19.67	25.39	12.48	5.80
	70 - 74	1.90	7.05	13.86	22.03	28.18	24.61	2.36
	75 - 79	3.67	4.74	16.06	9.13	30.95	20.03	15.42
	80 - 84	10.37	0.63	3.10	9.79	41.01	19.16	15.94
	85 +	1.59	0.49	4.25	5.79	33.84	34.47	19.57
Inst'd	65 - 69	5.60	0.73	3.16	3.02	3.06	74.82	9.61
	70 - 74	2.55	0.46	3.50	0.53	1.66	82.84	8.45
	75 - 79	2.25	0.31	4.82	0.34	0.37	80.34	11.57
	80 - 84	2.46	2.07	0.58	0.50	0.23	80.16	13.99
	85 +	1.63	0.38	0.58	0.38	1.44	72.08	23.52

Table E.105: The 5-year transition probabilities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys as a percentage, for males and females using 5 year age groupings.

	1989 Status								
1984 Status	Age		IADL	1 - 2	3 - 4	5-6			
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead	
Healthy	65 - 69	86.81	3.66	3.64	1.24	1.25	1.62	1.77	
-	70 - 74	81.09	5.15	5.11	2.10	1.45	2.99	2.10	
	75 - 79	70.87	5.97	7.24	3.33	2.63	5.70	4.27	
	80 - 84	54.47	6.77	11.40	5.65	4.56	10.29	6.84	
	85 +	41.57	8.18	9.14	7.03	5.59	19.58	8.92	
IADL only	65 - 69	31.96	21.36	25.44	5.68	4.77	6.75	4.03	
	70 - 74	20.59	26.46	23.89	8.02	5.77	9.42	5.85	
	75 - 79	15.39	22.51	22.85	10.46	9.84	12.77	6.20	
	80 - 84	7.33	20.17	20.90	14.76	9.54	17.82	9.48	
	85 +	3.67	13.49	18.80	10.00	12.77	23.62	17.65	
1-2 ADLs	65 - 69	19.33	12.60	32.69	14.99	7.79	9.07	3.54	
	70 - 74	10.79	8.26	43.53	16.58	8.25	8.01	4.58	
	75 - 79	8.35	8.26	30.73	17.89	12.88	14.89	7.00	
	80 - 84	8.75	8.66	25.70	15.78	6.46	28.19	6.47	
	85 +	3.12	5.06	19.72	17.23	12.19	27.36	15.32	
3–4 ADLs	65 - 69	8.20	9.09	26.50	23.14	17.69	9.01	6.37	
	70 - 74	7.83	4.36	19.53	34.82	12.75	12.70	8.02	
	75 - 79	4.38	2.72	15.92	25.91	16.31	23.02	11.74	
	80 - 84	2.54	2.11	11.83	24.17	15.69	29.50	14.16	
	85 +	4.18	2.15	8.18	20.57	13.40	29.88	21.64	
5–6 ADLs	65 - 69	9.96	4.23	19.58	16.83	29.46	13.45	6.49	
	70 - 74	3.18	6.73	15.39	17.50	34.24	19.16	3.79	
	75 - 79	2.81	5.82	15.71	9.95	32.65	19.46	13.60	
	80 - 84	8.55	0.58	6.26	15.29	35.11	21.12	13.08	
	85 +	1.54	0.43	3.57	6.34	34.08	34.67	19.37	
Inst'd	65 - 69	4.27	1.97	2.21	1.88	1.87	78.35	9.45	
	70 - 74	2.92	1.70	2.84	1.59	1.35	79.33	10.27	
	75 - 79	2.42	0.30	3.72	0.41	0.32	75.26	17.56	
	80 - 84	2.88	1.80	0.55	1.13	0.21	75.87	17.56	
	85 +	2.12	0.41	0.62	0.41	1.32	72.24	22.87	

Table E.106: The 5-year transition probabilities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys as a percentage, for males using 5 year age groupings.

				199	94 Status	5		
1989 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	86.82	3.05	3.53	1.35	1.42	2.09	1.75
	70 - 74	78.34	5.08	5.13	2.31	1.88	2.77	4.50
	75 - 79	69.01	7.84	6.69	3.52	3.21	5.11	4.63
	80 - 84	60.64	6.02	8.04	3.55	3.36	9.44	8.93
	85 +	43.09	2.41	10.20	2.30	6.53	20.67	14.81
IADL only	65 - 69	40.03	21.55	11.41	11.15	0.56	4.98	10.32
	70 - 74	36.89	17.48	18.36	6.95	4.80	10.09	5.43
	75 - 79	23.46	24.07	15.60	8.08	5.53	11.73	11.53
	80 - 84	5.03	33.12	16.93	13.39	7.21	17.64	6.68
	85 +	3.64	10.13	34.86	14.92	20.00	1.63	14.81
1-2 ADLs	65 - 69	27.20	4.34	34.99	11.05	3.92	11.72	6.79
	70 - 74	20.88	12.46	29.70	16.10	7.39	10.01	3.47
	75 - 79	11.31	4.71	21.03	10.75	16.51	29.56	6.13
	80 - 84	6.17	7.83	8.06	14.12	7.87	25.76	30.18
	85 +	10.07	1.89	22.26	7.96	8.24	30.18	19.40
3-4 ADLs	65 - 69	23.20	6.09	19.12	18.04	21.76	7.72	4.06
	70 - 74	6.83	5.08	5.59	18.61	31.42	14.75	17.71
	75 - 79	8.44	6.56	1.28	18.08	17.79	41.95	5.90
	80 - 84	7.40	0.21	0.02	18.23	18.63	18.85	36.65
	85 +	6.74	10.40	30.40	0.66	10.54	12.00	29.25
5-6 ADLs	65 - 69	9.54	0.90	22.48	0.81	32.66	22.91	10.71
	70 - 74	9.71	8.14	8.33	0.51	40.48	24.62	8.20
	75 - 79	19.54	0.48	0.72	7.55	21.39	29.17	21.15
	80 - 84	5.61	0.61	8.27	7.97	53.47	16.39	7.68
	85 +	0.78	0.17	0.60	0.15	19.70	39.80	38.80
Inst'd	65 - 69	4.15	8.22	0.08	0.32	0.04	79.29	7.89
	70 - 74	15.54	2.06	3.30	2.34	1.76	64.73	10.27
	75 - 79	3.01	0.90	7.41	1.00	0.69	74.69	12.31
	80 - 84	5.30	1.22	1.57	6.04	1.16	74.28	10.44
	85 +	6.02	2.41	3.82	2.11	2.02	54.02	29.59

Table E.107: The 5-year transition probabilities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys as a percentage, for females using 5 year age groupings.

				199	94 Status	3		
1989 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	83.45	3.63	5.35	1.66	1.78	2.32	1.82
	70 - 74	78.18	4.76	7.37	2.50	1.65	4.16	1.37
	75 - 79	67.16	5.94	9.43	3.35	3.16	8.22	2.74
	80 - 84	50.44	8.27	10.45	6.89	4.21	13.33	6.41
	85 +	34.15	6.52	11.05	6.22	10.03	25.20	6.84
IADL only	65 - 69	41.57	23.27	9.77	11.95	10.46	2.97	0.00
	70 - 74	20.94	25.52	28.36	9.85	2.05	6.85	6.43
	75 - 79	22.24	18.86	22.12	9.39	10.27	9.59	7.53
	80 - 84	12.12	11.55	24.49	9.24	7.12	23.99	11.48
	85 +	7.23	13.24	17.84	18.68	10.44	24.15	8.43
12  ADLs	65 - 69	24.44	8.53	29.90	19.79	10.04	4.06	3.23
	70 - 74	15.94	12.48	27.92	21.29	10.64	7.22	4.52
	75 - 79	15.45	7.19	24.38	25.15	9.98	10.24	7.61
	80 - 84	6.46	7.51	21.11	14.09	11.96	29.20	9.67
	85 +	5.33	6.70	16.24	14.12	9.90	35.11	12.59
3-4 ADLs	65 - 69	11.59	12.31	6.02	26.52	8.36	28.00	7.20
	70 - 74	12.62	2.17	16.85	29.46	13.45	17.56	7.90
	75 - 79	7.05	6.56	15.54	25.27	18.01	18.85	8.72
	80 - 84	4.62	2.89	12.86	22.94	15.06	35.28	6.35
	85 +	9.70	2.14	11.54	5.96	13.55	35.18	21.94
5-6 ADLs	65 - 69	6.39	1.11	1.78	11.83	44.26	23.77	10.85
	70 - 74	10.31	4.67	11.97	14.82	38.95	5.44	13.85
	75 - 79	8.10	1.24	8.50	14.59	34.77	22.62	10.18
	80 - 84	3.83	3.11	3.44	12.35	26.42	31.97	18.88
	85 +	8.85	2.11	5.79	7.13	30.22	30.16	15.73
Inst'd	65 - 69	12.88	1.62	8.60	1.42	1.24	68.30	5.94
	70 - 74	8.25	1.46	4.01	1.17	1.04	73.59	10.47
	75 - 79	5.18	0.57	2.41	2.18	0.53	69.16	19.97
	80 - 84	2.86	2.16	2.38	1.99	3.56	76.01	11.03
	85 +	3.37	0.71	1.08	0.68	0.60	79.98	13.59

Table E.108: The 5-year transition probabilities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys as a percentage, for males and females using 5 year age groupings.

				199	94 Status	3		
1989 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	84.89	3.38	4.57	1.53	1.62	2.22	1.79
-	70 - 74	78.24	4.89	6.43	2.42	1.75	3.58	2.69
	75 - 79	67.86	6.66	8.39	3.41	3.18	7.04	3.46
	80 - 84	53.82	7.53	9.65	5.78	3.93	12.05	7.24
	85 +	37.04	5.19	10.78	4.95	8.90	23.73	9.41
IADL only	65 - 69	41.13	22.77	10.25	11.72	7.59	3.55	2.99
	70 - 74	27.64	22.14	24.16	8.63	3.21	8.21	6.01
	75 - 79	22.60	20.37	20.23	9.01	8.89	10.21	8.69
	80 - 84	10.41	16.76	22.67	10.24	7.14	22.46	10.32
	85 +	6.44	12.56	21.56	17.86	12.53	19.23	9.82
1-2 ADLs	65 - 69	25.33	7.18	31.54	16.97	8.07	6.53	4.38
	70 - 74	17.47	12.47	28.47	19.68	9.63	8.08	4.19
	75 - 79	14.51	6.63	23.63	21.90	11.46	14.61	7.28
	80 - 84	6.41	7.57	18.89	14.10	11.26	28.61	13.16
	85 +	5.88	6.15	16.94	13.41	9.71	34.54	13.38
3–4 ADLs	65 - 69	17.05	9.39	12.17	22.54	14.66	18.47	5.72
	70 - 74	10.83	3.07	13.35	26.09	19.02	16.69	10.94
	75 - 79	7.33	6.56	12.66	23.82	17.97	23.52	8.15
	80 - 84	5.19	2.34	10.24	21.98	15.79	31.92	12.55
	85 +	9.35	3.10	13.75	5.34	13.20	32.46	22.79
5–6 ADLs	65 - 69	7.45	1.04	8.74	8.12	40.36	23.48	10.80
	70 - 74	10.13	5.72	10.88	10.52	39.41	11.20	12.15
	75 - 79	11.82	0.99	5.97	12.31	30.42	24.74	13.74
	80 - 84	4.26	2.52	4.58	11.31	32.84	28.27	16.22
	85 +	7.89	1.88	5.18	6.30	28.97	31.31	18.47
Inst'd	65 - 69	9.13	4.45	4.94	0.95	0.73	73.02	6.77
	70 - 74	10.39	1.64	3.80	1.52	1.25	70.99	10.41
	75 - 79	4.71	0.64	3.49	1.92	0.56	70.35	18.32
	80 - 84	3.43	1.94	2.19	2.94	3.00	75.61	10.89
	85 +	3.74	0.95	1.46	0.88	0.80	76.37	15.81

## Appendix F

Maximum likelihood estimates of the annual transition intensities between disability states calculated from the 1984, 1989 and 1994 National Long-Term Care Surveys

The tables in this appendix give the maximum likelihood estimates of the annual transition intensities between disability states for men, women and in aggregate (as described in Section 4.2), based on data grouped in 5-year age bands. They are given for the 1984–89 NLTCS in Tables F.109 – F.111 and for the 1989–94 NLTCS in Tables F.112 to F.114. These estimates are the result of direct transformations of the corresponding transition probabilities and so there is no constraint on the transition intensities to be positive.

For some categories these estimates do not lie in the real plane (the matrix of transition intensities is complex) and they are marked '-' in these tables. However, in Section 4.5, I calculate approximate MLEs for these categores, which are given in Tables 4.48 and 4.49, respectively.

Table F.109: The MLEs of the annual transition intensities between disability st	ates
calculated from the 1984 and 1989 National Long-Term Care Surveys, for males u	ising
5 year age groupings.	

1984				198	39 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		-	-	-	-	-	-
Ū	70 - 74		0.0173	0.0079	0.0021	0.0008	0.0050	0.0045
	75 - 79		0.0275	0.0119	0.0072	0.0008	0.0127	0.0095
	80 - 84		-	-	-	-	-	-
	85 +		-	-	-	-	-	-
IADL	65 - 69	-		-	-	-	-	-
only	70 - 74	0.0458		0.0985	0.0438	0.0223	0.0426	0.0180
	75 - 79	0.0986		0.0704	0.1170	0.0607	0.0028	0.0127
	80 - 84	-		-	-	-	-	-
	85 +	-		-	-	-	-	-
1 - 2	65 - 69	-	-		-	-	-	-
ADLs	70 - 74	0.0651	0.0527		0.1168	0.0406	0.0065	0.0118
	75 - 79	0.0351	0.1452		0.1677	0.0004	0.0169	0.0182
	80 - 84	-	-		-	-	-	-
	85 +	-	-		-	-	-	-
3 - 4	65 - 69	-	-	-		-	-	-
ADLs	70 - 74	-0.0074	0.0237	0.2147		0.0608	0.0731	0.0131
	75 - 79	0.0088	-0.1668	0.3077		0.3036	0.0778	0.0533
	80 - 84	-	-	-		-	-	-
	85 +	-	-	-		-	-	-
5-6	65 - 69	-	-	-	-		-	-
ADLs	70 - 74	0.0021	0.0199	0.0728	0.0549		0.0320	0.0089
	75 - 79	-0.0143	0.0620	0.0636	0.0825		0.0664	0.0078
	80 - 84	-	-	-	-		-	-
	85 +	-	-	-	-		-	-
Inst'd	65 - 69	-	-	-	-	-		-
	70 - 74	0.0088	0.0217	-0.0096	0.0227	-0.0018		0.0353
	75 - 79	0.0083	0.0007	0.0023	0.0032	-0.0011		0.0841
	80 - 84	-	-	-	-	-		-
	85 +	-	-	-	-	-		-

Table F.110: The MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for females using 5 year age groupings.

1984				1	989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0154	0.0043	0.0024	0.0039	0.0035	0.0021
	70 - 74		0.0221	0.0134	0.0041	0.0044	0.0052	0.0025
	75 - 79		0.0265	0.0266	0.0049	0.0043	0.0095	0.0050
	80 - 84		0.0298	0.0596	0.0054	0.0146	0.0157	0.0093
	85 +		0.0508	0.0383	0.0367	0.0093	0.0528	0.0027
IADL	65 - 69	0.1032		0.2245	-0.0172	0.0262	0.0117	0.0046
only	70 - 74	0.0954		0.1559	0.0040	0.0227	0.0192	0.0133
	75 - 79	0.0541		0.1998	0.0087	0.0159	0.0371	0.0081
	80 - 84	0.0169		0.2380	0.0300	0.0327	0.0182	0.0208
	85 +	0.0113		0.2705	-0.0575	0.0926	0.0501	0.0389
1 - 2	65 - 69	0.0709	0.0473		0.1089	0.0003	0.0334	0.0041
ADLs	70 - 74	0.0126	0.0429		0.0766	0.0251	0.0153	0.0055
	75 - 79	0.0195	0.0508		0.1335	0.0637	0.0297	0.0029
	80 - 84	0.0390	0.0750		0.1593	0.0031	0.1017	-0.0055
	85 +	0.0099	0.0399		0.2009	0.0333	0.0863	0.0146
3 - 4	65 - 69	-0.0297	0.1137	0.1007		0.1697	0.0176	0.0175
ADLs	70 - 74	0.0296	-0.0039	0.0663		0.0806	0.0181	0.0245
	75 - 79	0.0163	0.0128	0.0631		0.0766	0.0823	0.0171
	80 - 84	-0.0062	0.0007	0.1070		0.0965	0.0813	0.0331
	85 +	0.0272	0.0160	0.0837		0.1121	0.0875	0.0501
5 - 6	65 - 69	0.0288	-0.0487	0.1527	0.1602		0.0400	0.0125
ADLs	70 - 74	-0.0150	0.0522	0.0384	0.1328		0.0911	-0.0058
	75 - 79	0.0055	0.0224	0.0944	0.0312		0.0622	0.0456
	80 - 84	0.0441	-0.0042	-0.0002	0.0602		0.0532	0.0392
	85 +	0.0034	-0.0017	0.0266	0.0352		0.1266	0.0412
Inst'd	65 - 69	0.0123	0.0003	0.0060	0.0094	0.0108		0.0214
	70 - 74	0.0060	-0.0002	0.0106	-0.0018	0.0062		0.0184
	75 - 79	0.0053	-0.0012	0.0205	-0.0041	-0.0011		0.0256
	80 - 84	0.0073	0.0094	-0.0031	0.0012	-0.0003		0.0307
	85 +	0.0059	0.0012	0.0011	0.0003	0.0051		0.0546

Table F.111: The MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1984				1	989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0149	0.0066	0.0026	0.0028	0.0026	0.0031
	70 - 74		0.0202	0.0108	0.0037	0.0027	0.0052	0.0033
	75 - 79		0.0264	0.0206	0.0058	0.0036	0.0102	0.0071
	80 - 84		0.0324	0.0494	0.0069	0.0131	0.0151	0.0125
	85 +		0.0639	0.0302	0.0275	0.0078	0.0453	0.0073
IADL	65 - 69	0.1198		0.2537	-0.0459	0.0312	0.0147	0.0111
only	70 - 74	0.0767		0.1412	0.0150	0.0212	0.0273	0.0147
	75 - 79	0.0646		0.1786	0.0237	0.0376	0.0294	0.0084
	80 - 84	0.0218		0.1879	0.0738	0.0432	0.0184	0.0213
	85 +	0.0194		0.2671	-0.0076	0.0753	0.0559	0.0445
1 - 2	65 - 69	0.0545	0.1042		0.1551	0.0050	0.0276	0.0030
ADLs	70 - 74	0.0265	0.0449		0.0854	0.0287	0.0137	0.0071
	75 - 79	0.0252	0.0664		0.1414	0.0515	0.0270	0.0068
	80 - 84	0.0428	0.0824		0.1328	0.0015	0.0938	-0.0028
	85 +	0.0071	0.0655		0.1974	0.0499	0.0730	0.0163
3 - 4	65 - 69	-0.0119	0.0469	0.1998		0.1633	0.0144	0.0172
ADLs	70 - 74	0.0205	0.0089	0.0965		0.0718	0.0344	0.0216
	75 - 79	0.0118	-0.0065	0.1034		0.1109	0.0814	0.0241
	80 - 84	-0.0101	-0.0028	0.1110		0.1223	0.1048	0.0354
	85 +	0.0236	0.0095	0.0802		0.0936	0.1071	0.0571
5-6	65 - 69	0.0245	-0.0143	0.1004	0.1270		0.0450	0.0151
ADLs	70 - 74	-0.0040	0.0352	0.0509	0.0957		0.0614	0.0017
	75 - 79	-0.0009	0.0320	0.0864	0.0421		0.0611	0.0340
	80 - 84	0.0379	-0.0082	0.0162	0.1134		0.0526	0.0277
	85 +	0.0025	-0.0027	0.0218	0.0419		0.1259	0.0400
Inst'd	65 - 69	0.0080	0.0077	0.0017	0.0069	0.0052		0.0207
	70 - 74	0.0058	0.0057	0.0068	0.0034	0.0036		0.0225
	75 - 79	0.0059	-0.0013	0.0169	-0.0030	-0.0006		0.0401
	80 - 84	0.0088	0.0084	-0.0027	0.0040	-0.0012		0.0395
	85 +	0.0076	0.0009	0.0015	0.0002	0.0045		0.0529

Table F.112: The MLEs of the annual transition intensities between disability states	3
calculated from the 1989 and 1994 National Long-Term Care Surveys, for males using	r
5 year age groupings.	

1989				-	1994 Statu	5		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0126	0.0093	0.0019	0.0047	0.0031	0.0022
Ū	70 - 74		0.0233	0.0140	0.0051	0.0030	0.0040	0.0087
	75 - 79		0.0362	0.0241	0.0088	0.0047	0.0034	0.0073
	80 - 84		0.0195	0.0547	-0.0047	0.0083	0.0142	0.0122
	85 +		-	-	-	-	-	-
IADL	65 - 69	0.1516		0.0698	0.1034	-0.0429	0.0182	0.0381
only	70 - 74	0.1770		0.1822	0.0070	0.0213	0.0388	0.0138
	75 - 79	0.1082		0.1421	0.0483	-0.0262	0.0066	0.0415
	80 - 84	0.0072		0.2319	0.0244	0.0139	0.0131	-0.0334
	85 +	-		-	-	-	-	-
1 - 2	65 - 69	0.0818	0.0118		0.0913	-0.0107	0.0422	0.0185
ADLs	70 - 74	0.0456	0.1214		0.1403	-0.0266	0.0267	-0.0098
	75 - 79	0.0203	0.0294		0.0871	0.1252	0.0765	-0.0048
	80 - 84	0.0220	0.1018		0.2213	0.0078	0.1396	0.1094
	85 +	-	-		-	-	-	-
3 - 4	65 - 69	0.0651	0.0609	0.0791		0.1897	-0.0084	-0.0043
ADLs	70 - 74	0.0105	0.0088	0.0144		0.2246	0.0245	0.0595
	75 - 79	-0.0158	0.0802	-0.0405		0.2224	0.1767	-0.0273
	80 - 84	0.0355	0.0072	-0.0620		0.1193	0.0928	0.1556
	85 +	-	-	-		-	-	-
5 - 6	65 - 69	0.0122	-0.0107	0.1431	-0.0267		0.0746	0.0268
ADLs	70 - 74	0.0038	0.0537	0.0286	-0.0153		0.0900	0.0190
	75 - 79	0.1038	-0.0251	-0.0116	0.0883		0.1158	0.0775
	80 - 84	0.0135	-0.0103	0.0821	0.0198		0.0320	-0.0023
	85 +	-	-	-	-		-	-
Inst'd	65 - 69	0.0009	0.0392	-0.0050	-0.0060	0.0032		0.0153
	70 - 74	0.0408	0.0052	0.0107	0.0079	0.0023		0.0233
	75 - 79	0.0069	0.0010	0.0355	-0.0014	-0.0067		0.0286
	80 - 84	0.0138	0.0025	0.0102	0.0273	-0.0029		0.0155
	85 +	-	-	-	-	-		-

Table F.113: The MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for females using 5 year age groupings.

1989					1994 Status	5		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0145	0.0189	-0.0016	0.0026	0.0061	0.0036
-	70 - 74		0.0157	0.0254	0.0001	0.0030	0.0093	0.0013
	75 - 79		0.0297	0.0376	-0.0058	0.0076	0.0201	0.0016
	80 - 84		0.0716	0.0101	0.0306	0.0083	0.0196	0.0096
	85 +		-	-	-	-	-	-
IADL	65 - 69	0.1732		0.0644	0.0693	0.0519	-0.0188	-0.0096
only	70 - 74	0.0712		0.3037	-0.0360	-0.0248	0.0250	0.0221
	75 - 79	0.0854		0.2496	-0.0876	0.0813	0.0261	0.0154
	80 - 84	0.0911		0.5310	-0.0467	-0.0218	0.0618	0.0367
	85 +	-		-	-	-	-	-
1 - 2	65 - 69	0.0792	0.0299		0.1365	0.0389	-0.0276	0.0028
ADLs	70 - 74	0.0401	0.1391		0.1993	0.0516	-0.0028	-0.0004
	75 - 79	0.0722	0.0326		0.2993	-0.0129	0.0027	0.0170
	80 - 84	0.0152	0.1445		0.1481	0.0818	0.0836	0.0157
	85 +	-	-		-	-	-	-
3 - 4	65 - 69	-0.0028	0.1077	0.0120		0.0354	0.1285	0.0209
ADLs	70 - 74	0.0354	-0.0358	0.1598		0.0716	0.0761	0.0174
	75 - 79	-0.0046	0.0624	0.1230		0.1361	0.0662	0.0129
	80 - 84	0.0169	-0.0383	0.1836		0.1079	0.1202	-0.0037
	85 +	-	-	-		-	-	-
5-6	65 - 69	0.0153	-0.0128	-0.0022	0.0741		0.0683	0.0263
ADLs	70 - 74	0.0225	0.0193	0.0456	0.0770		0.0056	0.0387
	75 - 79	0.0247	-0.0123	0.0391	0.0912		0.0784	0.0191
	80 - 84	0.0094	0.0470	-0.0523	0.1235		0.1128	0.0605
	85 +	-	-	-	-		-	-
Inst'd	65 - 69	0.0275	0.0037	0.0358	-0.0056	0.0006		0.0137
	70 - 74	0.0200	0.0017	0.0167	-0.0016	0.0019		0.0238
	75 - 79	0.0141	-0.0001	0.0090	0.0062	-0.0006		0.0472
	80 - 84	0.0071	0.0118	0.0023	0.0043	0.0135		0.0229
	85 +	-	-	-	-	-		-

Table F.114: The MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1989					1994 Statı	18		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0134	0.0156	-0.0008	0.0036	0.0046	0.0031
Ū	70 - 74		0.0188	0.0197	0.0022	0.0037	0.0072	0.0044
	75 - 79		0.0328	0.0301	0.0004	0.0066	0.0146	0.0038
	80 - 84		0.0442	0.0337	0.0170	0.0064	0.0190	0.0118
	85 +		-	-	-	-	-	-
IADL	65 - 69	0.1650		0.0522	0.0915	0.0259	-0.0081	0.0042
only	70 - 74	0.1109		0.2462	-0.0217	-0.0003	0.0312	0.0192
	75 - 79	0.0900		0.2047	-0.0355	0.0522	0.0210	0.0203
	80 - 84	0.0524		0.3368	0.0018	0.0025	0.0426	0.0150
	85 +	-		-	-	-	-	-
1 - 2	65 - 69	0.0756	0.0288		0.1367	0.0171	0.0015	0.0086
ADLs	70 - 74	0.0410	0.1295		0.1818	0.0166	0.0031	-0.0027
	75 - 79	0.0610	0.0296		0.2348	0.0179	0.0200	0.0141
	80 - 84	0.0206	0.1071		0.1526	0.0590	0.0964	0.0327
	85 +	-	-		-	-	-	-
3 - 4	65 - 69	0.0297	0.0819	0.0739		0.0951	0.0724	0.0107
ADLs	70 - 74	0.0273	-0.0186	0.1040		0.1256	0.0674	0.0283
	75 - 79	-0.0051	0.0645	0.0903		0.1442	0.0827	0.0050
	80 - 84	0.0182	-0.0153	0.1265		0.1140	0.1124	0.0242
	85 +	-	-	-		-	-	-
5-6	65 - 69	0.0128	-0.0125	0.0394	0.0468		0.0789	0.0281
ADLs	70 - 74	0.0195	0.0268	0.0454	0.0511		0.0317	0.0313
	75 - 79	0.0470	-0.0150	0.0225	0.0930		0.0888	0.0347
	80 - 84	0.0121	0.0156	0.0035	0.0866		0.0902	0.0426
	85 +	-	-	-	-		-	-
Inst'd	65 - 69	0.0151	0.0197	0.0181	-0.0051	0.0005		0.0151
	70 - 74	0.0257	0.0029	0.0140	0.0011	0.0028		0.0238
	75 - 79	0.0119	0.0001	0.0157	0.0027	-0.0007		0.0429
	80-84	0.0092	0.0085	0.0029	0.0104	0.0086		0.0229
	85 +	-	-	-	-	-		-

## Appendix G

Constrained maximum likelihood estimates of the annual transition intensities between disability states calculated from the 1982, 1984, 1989 and 1994 National Long-Term Care Surveys

The tables in this appendix give the constrained maximum likelihood estimates of the annual transition intensities between disability states for men, women and in aggregate (as described in Section 4.5). They are given for all pairs of surveys in 10year age bands (65–74 years, 75–84 years and 85+ years) in Tables G.115 to G.123 and in 5-year age bands (65–69 years, 70–74 years, 75–79 years, 80–84 years and 85+ years) in Tables G.124 to G.129 (except for those from the 1982–84 NLTCS, which are given in Tables 4.50 to 4.52).

Table G.115: The constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males and females using 10 year age groupings.

1982				19	84 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0247	0.0125	0.0039	0.0042	0.0036	0.0250
	75 - 84		0.0551	0.0335	0.0084	0.0062	0.0146	0.0477
	85 +		0.1069	0.1115	0.0000	0.0243	0.0579	0.0807
IADL	65 - 74	0.2405		0.2619	0.0085	0.0353	0.0262	0.0617
only	75 - 84	0.1175		0.3687	0.0000	0.0686	0.0603	0.0683
	85 +	0.0122		0.3886	0.0323	0.0517	0.0637	0.1220
1 - 2	65 - 74	0.0714	0.2019		0.2043	0.0365	0.0436	0.0869
ADLs	75 - 84	0.0560	0.1907		0.2303	0.0164	0.0553	0.1329
	85 +	0.0469	0.1026		0.2980	0.0463	0.1186	0.0945
3-4	65 - 74	0.0348	0.0000	0.3907		0.2869	0.0189	0.0754
ADLs	75 - 84	0.0056	0.0000	0.3389		0.4066	0.1036	0.0972
	85 +	0.0042	0.0692	0.0563		0.5877	0.1808	0.0798
5-6	65 - 74	0.0332	0.0603	0.0718	0.1672		0.0723	0.2409
ADLs	75 - 84	0.0292	0.0555	0.0238	0.1897		0.1132	0.2511
	85 +	0.0130	0.0000	0.0847	0.1570		0.1337	0.3663
Inst'd	65 - 74	0.0310	0.0051	0.0108	0.0175	0.0107		0.1736
	75 - 84	0.0084	0.0100	0.0020	0.0086	0.0076		0.2384
	85 +	0.0044	0.0022	0.0000	0.0107	0.0027		0.3368

Table G.116: The constrained MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males and females using 10 year age groupings.

1984				19	89 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0175	0.0089	0.0031	0.0028	0.0039	0.0032
	75 - 84		0.0281	0.0283	0.0061	0.0062	0.0116	0.0086
	85 +		0.0638	0.0307	0.0269	0.0080	0.0453	0.0073
IADL	65 - 74	0.1000		0.1765	0.0000	0.0205	0.0225	0.0126
only	75 - 84	0.0471		0.1810	0.0453	0.0404	0.0248	0.0140
	85 +	0.0191		0.2603	0.0000	0.0740	0.0555	0.0440
1 - 2	65 - 74	0.0377	0.0663		0.1038	0.0236	0.0187	0.0053
ADLs	75 - 84	0.0304	0.0727		0.1354	0.0282	0.0575	0.0028
	85 +	0.0073	0.0646		0.1927	0.0506	0.0733	0.0166
3 - 4	65 - 74	0.0106	0.0245	0.1331		0.1037	0.0261	0.0197
ADLs	75 - 84	0.0033	0.0000	0.0978		0.1148	0.0924	0.0289
	85 +	0.0237	0.0075	0.0812		0.0933	0.1069	0.0571
5-6	65 - 74	0.0081	0.0178	0.0686	0.1089		0.0539	0.0078
ADLs	75 - 84	0.0143	0.0122	0.0611	0.0701		0.0555	0.0318
	85 +	0.0022	0.0000	0.0191	0.0422		0.1258	0.0398
Inst'd	65 - 74	0.0067	0.0064	0.0052	0.0045	0.0044		0.0217
	75 - 84	0.0068	0.0034	0.0071	0.0005	0.0000		0.0399
	85 +	0.0076	0.0008	0.0016	0.0002	0.0045		0.0529

Table G.117: The constrained MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males and females using 10 year age groupings.

1989				19	94 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0158	0.0181	0.0006	0.0036	0.0059	0.0038
	75 - 84		0.0359	0.0325	0.0056	0.0064	0.0160	0.0067
	85 +		0.0340	0.0645	0.0000	0.0417	0.0578	0.0139
IADL only	65 - 74	0.1333		0.1449	0.0342	0.0099	0.0143	0.0109
	75 - 84	0.0752		0.2319	0.0000	0.0274	0.0345	0.0194
	85 +	0.0058		0.2640	0.1718	0.0314	0.0000	0.0000
1-2 ADLs	65 - 74	0.0501	0.0819		0.1553	0.0171	0.0049	0.0033
	75 - 84	0.0419	0.0610		0.1859	0.0390	0.0505	0.0195
	85 +	0.0000	0.0887		0.3329	0.0000	0.1105	0.0000
3-4 ADLs	65 - 74	0.0313	0.0265	0.0795		0.1139	0.0664	0.0202
	75 - 84	0.0080	0.0305	0.0965		0.1287	0.0966	0.0146
	85 +	0.0906	0.0000	0.2898		0.2054	0.1367	0.1034
5-6 ADLs	65 - 74	0.0151	0.0102	0.0473	0.0464		0.0521	0.0299
	75 - 84	0.0268	0.0000	0.0152	0.0876		0.0906	0.0390
	85 +	0.0368	0.0046	0.0261	0.0533		0.1048	0.0445
Inst'd	65 - 74	0.0225	0.0083	0.0144	0.0000	0.0022		0.0211
	75 - 84	0.0102	0.0039	0.0098	0.0070	0.0039		0.0324
	85 +	0.0133	0.0036	0.0032	0.0016	0.0006		0.0350

Table G.118: The constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for females using 10 year age groupings.

1982				19	84 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0279	0.0126	0.0036	0.0043	0.0030	0.0159
	75 - 84		0.0595	0.0339	0.0124	0.0059	0.0170	0.0369
	85 +		0.1052	0.0985	0.0000	0.0352	0.0731	0.0626
IADL	65 - 74	0.2212		0.2877	0.0462	0.0155	0.0344	0.0386
only	75 - 84	0.1048		0.4225	0.0000	0.0333	0.0530	0.0747
	85 +	0.0143		0.4450	0.0000	0.0950	0.0844	0.0866
1 - 2	65 - 74	0.0797	0.2286		0.1476	0.0350	0.0371	0.0733
ADLs	75 - 84	0.0627	0.1945		0.2287	0.0067	0.0765	0.0905
	85 +	0.0352	0.1232		0.2886	0.0112	0.0976	0.0902
3 - 4	65 - 74	0.0037	0.0000	0.4167		0.2620	0.0265	0.0194
ADLs	75 - 84	0.0000	0.0000	0.3665		0.3997	0.0881	0.0697
	85 +	0.0065	0.0662	0.0570		0.5585	0.1695	0.0334
5-6	65 - 74	0.0307	0.0719	0.0474	0.1513		0.0791	0.2269
ADLs	75 - 84	0.0334	0.0403	0.0333	0.1856		0.1271	0.2013
	85 +	0.0166	0.0000	0.0817	0.1506		0.1500	0.3401
Inst'd	65 - 74	0.0286	0.0052	0.0013	0.0241	0.0070		0.1437
	75 - 84	0.0073	0.0086	0.0023	0.0057	0.0094		0.2090
	85 +	0.0047	0.0026	0.0000	0.0118	0.0034		0.3093

Table G.119: The constrained MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for females using 10 year age groupings.

1984				19	89 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0187	0.0091	0.0033	0.0042	0.0043	0.0023
	75 - 84		0.0275	0.0357	0.0052	0.0073	0.0114	0.0061
	85 +		0.0503	0.0412	0.0331	0.0101	0.0528	0.0029
IADL	65 - 74	0.1024		0.1810	0.0000	0.0210	0.0168	0.0090
only	75 - 84	0.0393		0.2156	0.0177	0.0243	0.0285	0.0134
	85 +	0.0090		0.2246	0.0000	0.0801	0.0498	0.0358
1 - 2	65 - 74	0.0361	0.0464		0.0873	0.0172	0.0229	0.0048
ADLs	75 - 84	0.0248	0.0628		0.1422	0.0332	0.0637	0.0000
	85 +	0.0113	0.0376		0.1693	0.0400	0.0872	0.0164
3 - 4	65 - 74	0.0080	0.0389	0.0883		0.1139	0.0169	0.0212
ADLs	75 - 84	0.0073	0.0073	0.0801		0.0852	0.0832	0.0243
	85 +	0.0265	0.0152	0.0771		0.1084	0.0866	0.0491
5-6	65 - 74	0.0089	0.0099	0.0845	0.1435		0.0668	0.0036
ADLs	75 - 84	0.0233	0.0072	0.0493	0.0444		0.0547	0.0425
	85 +	0.0033	0.0000	0.0259	0.0329		0.1266	0.0412
Inst'd	65 - 74	0.0079	0.0003	0.0095	0.0013	0.0082		0.0195
	75 - 84	0.0058	0.0045	0.0069	0.0000	0.0000		0.0285
	85 +	0.0059	0.0011	0.0012	0.0003	0.0050		0.0546

Table G.120: The constrained MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for females using 10 year age groupings.

1989				19	94 Statu	S		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0148	0.0222	0.0000	0.0028	0.0073	0.0023
	75 - 84		0.0384	0.0357	0.0057	0.0064	0.0201	0.0047
	85 +		0.0461	0.0694	0.0000	0.0509	0.0597	0.0061
IADL only	65 - 74	0.1179		0.1519	0.0307	0.0149	0.0069	0.0058
	75 - 84	0.0855		0.2600	0.0000	0.0360	0.0414	0.0214
	85 +	0.0161		0.2375	0.2106	0.0017	0.0000	0.0000
1-2 ADLs	65 - 74	0.0479	0.0843		0.1414	0.0413	0.0000	0.0038
	75 - 84	0.0452	0.0687		0.1867	0.0361	0.0393	0.0178
	85 +	0.0000	0.1002		0.3196	0.0000	0.1021	0.0000
3–4 ADLs	65 - 74	0.0262	0.0258	0.0784		0.0619	0.0804	0.0166
	75 - 84	0.0052	0.0300	0.1221		0.1241	0.0889	0.0025
	85 +	0.0789	0.0000	0.1895		0.2017	0.1776	0.0946
5–6 ADLs	65 - 74	0.0172	0.0078	0.0291	0.0732		0.0317	0.0336
	75 - 84	0.0176	0.0056	0.0084	0.1026		0.0994	0.0415
	85 +	0.0469	0.0000	0.0423	0.0514		0.0896	0.0379
Inst'd	65 - 74	0.0220	0.0022	0.0196	0.0000	0.0014		0.0211
	75 - 84	0.0103	0.0051	0.0067	0.0052	0.0060		0.0348
	85 +	0.0123	0.0018	0.0025	0.0010	0.0002		0.0299

Table G.121: The constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males using 10 year age groupings.

1982				19	84 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0206	0.0129	0.0036	0.0043	0.0045	0.0369
	75 - 84		0.0484	0.0339	0.0014	0.0064	0.0118	0.0636
	85 +		0.1145	0.1448	0.0000	0.0000	0.0168	0.1218
IADL	65 - 74	0.2707		0.1737	0.0000	0.0520	0.0137	0.1039
	75 - 84	0.1402		0.2263	0.0534	0.1225	0.0733	0.0496
	85 +	0.0087		0.2708	0.0724	0.0000	0.0145	0.1971
1 - 2	65 - 74	0.0552	0.1497		0.2621	0.0553	0.0552	0.1069
ADLs	75 - 84	0.0409	0.1626		0.1713	0.0760	0.0000	0.2465
	85 +	0.0882	0.0429		0.4066	0.0925	0.2068	0.0866
3-4	65 - 74	0.0890	0.0000	0.3155		0.3106	0.0047	0.1762
ADLs	75 - 84	0.0187	0.0383	0.1789		0.3831	0.1284	0.2251
	85 +	0.0000	0.0859	0.0431		0.7191	0.2653	0.1722
5-6	65 - 74	0.0338	0.0493	0.1169	0.1699		0.0650	0.2475
ADLs	75 - 84	0.0202	0.0730	0.0342	0.1854		0.0921	0.3240
	85 +	0.0000	0.0000	0.0918	0.1486		0.0576	0.4924
Inst'd	65 - 74	0.0355	0.0059	0.0265	0.0040	0.0172		0.2211
	75 - 84	0.0115	0.0142	0.0026	0.0176	0.0013		0.3330
	85 +	0.0029	0.0000	0.0008	0.0000	0.0001		0.4809

Table G.122: The constrained MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males using 10 year age groupings.

1984				19	89 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0155	0.0094	0.0020	0.0011	0.0032	0.0044
	75 - 84		0.0298	0.0178	0.0098	0.0006	0.0129	0.0112
	85 +		0.0965	0.0157	0.0000	0.0021	0.0299	0.0182
IADL	65 - 74	0.0954		0.1592	0.0026	0.0198	0.0331	0.0193
only	75 - 84	0.0769		0.0697	0.1579	0.0620	0.0000	0.0198
	85 +	0.0491		0.2636	0.1636	0.0509	0.0435	0.0525
1 - 2	65 - 74	0.0437	0.1119		0.1606	0.0385	0.0049	0.0064
ADLs	75 - 84	0.0365	0.1020		0.0550	0.0339	0.0394	0.0193
	85 +	0.0000	0.2103		0.1114	0.1367	0.0139	0.0300
3 - 4	65 - 74	0.0124	0.0000	0.2486		0.0893	0.0497	0.0168
ADLs	75 - 84	0.0000	0.0000	0.0883		0.3225	0.1644	0.0354
	85 +	0.0000	0.0000	0.0001		0.0022	0.1953	0.0998
5-6	65 - 74	0.0080	0.0208	0.0613	0.0615		0.0391	0.0130
ADLs	75 - 84	0.0000	0.0000	0.1087	0.2476		0.0343	0.0000
	85 +	0.0000	0.0000	0.0000	0.0644		0.1134	0.0331
Inst'd	65 - 74	0.0048	0.0156	0.0000	0.0073	0.0000		0.0261
	75 - 84	0.0104	0.0000	0.0005	0.0096	0.0000		0.0901
	85 +	0.0168	0.0000	0.0033	0.0010	0.0010		0.0420

Table G.123: The constrained MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males using 10 year age groupings.

1989				19	94 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		0.0176	0.0120	0.0037	0.0036	0.0037	0.0056
	75 - 84		0.0305	0.0313	0.0056	0.0053	0.0077	0.0094
	85 +		0.0131	0.0575	0.0000	0.0213	0.0677	0.0200
IADL only	65 - 74	0.1633		0.1311	0.0434	0.0000	0.0278	0.0202
	75 - 84	0.0598		0.1515	0.0453	0.0000	0.0114	0.0114
	85 +	0.0000		0.0000	0.5805	0.0000	0.0000	0.0000
1-2 ADLs	65 - 74	0.0568	0.0741		0.1064	0.0000	0.0329	0.0028
	75 - 84	0.0286	0.0472		0.1294	0.0670	0.1024	0.0332
	85 +	0.0538	0.0248		0.0899	0.2037	0.0000	0.0134
3–4 ADLs	65 - 74	0.0417	0.0294	0.0605		0.1819	0.0115	0.0301
	75 - 84	0.0196	0.0190	0.0000		0.1381	0.1299	0.0671
	85 +	0.0000	0.1643	0.7154		0.0000	0.0000	0.1001
5-6 ADLs	65 - 74	0.0121	0.0148	0.0670	0.0000		0.0838	0.0218
	75 - 84	0.0455	0.0000	0.0150	0.0493		0.0714	0.0332
	85 +	0.0000	0.0000	0.0000	0.0000		0.2885	0.1228
Inst'd	65 - 74	0.0225	0.0221	0.0017	0.0037	0.0018		0.0206
	75 - 84	0.0103	0.0017	0.0175	0.0120	0.0000		0.0231
	85 +	0.0221	0.0138	0.0033	0.0082	0.0055		0.0741

Table G.124: The constrained MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1984				19	89 Statu	S		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0148	0.0072	0.0019	0.0031	0.0026	0.0031
	70 - 74		0.0202	0.0108	0.0037	0.0027	0.0052	0.0033
	75 - 79		0.0264	0.0206	0.0058	0.0036	0.0102	0.0071
	80 - 84		0.0323	0.0489	0.0073	0.0129	0.0151	0.0124
	85 +		0.0638	0.0307	0.0269	0.0080	0.0453	0.0073
IADL	65 - 69	0.1233		0.2151	0.0000	0.0137	0.0159	0.0104
only	70 - 74	0.0762		0.1410	0.0151	0.0212	0.0273	0.0146
	75 - 79	0.0644		0.1756	0.0255	0.0375	0.0292	0.0083
	80 - 84	0.0224		0.1826	0.0767	0.0417	0.0192	0.0203
	85 +	0.0191		0.2603	0.0000	0.0740	0.0555	0.0440
1 - 2	65 - 69	0.0487	0.1033		0.1154	0.0205	0.0265	0.0036
ADLs	70 - 74	0.0263	0.0450		0.0854	0.0287	0.0137	0.0071
	75 - 79	0.0255	0.0621		0.1364	0.0514	0.0276	0.0070
	80 - 84	0.0391	0.0775		0.1258	0.0048	0.0922	0.0000
	85 +	0.0073	0.0646		0.1927	0.0506	0.0733	0.0166
3 - 4	65 - 69	0.0000	0.0315	0.1864		0.1474	0.0155	0.0169
ADLs	70 - 74	0.0190	0.0098	0.0966		0.0714	0.0344	0.0215
	75 - 79	0.0109	0.0000	0.0966		0.1093	0.0810	0.0240
	80 - 84	0.0000	0.0000	0.0969		0.1131	0.1050	0.0328
	85 +	0.0237	0.0075	0.0812		0.0933	0.1069	0.0571
5-6	65 - 69	0.0197	0.0000	0.0896	0.1246		0.0448	0.0149
ADLs	70 - 74	0.0000	0.0321	0.0505	0.0941		0.0612	0.0018
	75 - 79	0.0000	0.0291	0.0876	0.0417		0.0610	0.0340
	80 - 84	0.0311	0.0000	0.0131	0.1093		0.0525	0.0279
	85 +	0.0022	0.0000	0.0191	0.0422		0.1258	0.0398
Inst'd	65 - 69	0.0079	0.0075	0.0023	0.0062	0.0054		0.0207
	70 - 74	0.0057	0.0057	0.0068	0.0034	0.0036		0.0225
	75 - 79	0.0058	0.0000	0.0124	0.0000	0.0000		0.0399
	80 - 84	0.0080	0.0065	0.0000	0.0025	0.0000		0.0391
	85 +	0.0076	0.0008	0.0016	0.0002	0.0045		0.0529

Table G.125: The constrained MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1989				19	94 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0131	0.0153	0.0000	0.0035	0.0043	0.0030
	70 - 74		0.0188	0.0198	0.0021	0.0037	0.0072	0.0044
	75 - 79		0.0325	0.0306	0.0000	0.0066	0.0147	0.0038
	80 - 84		0.0440	0.0341	0.0168	0.0064	0.0190	0.0118
	85 +		0.0340	0.0645	0.0000	0.0417	0.0578	0.0139
IADL only	65 - 69	0.1629		0.0550	0.0782	0.0247	0.0000	0.0042
	70 - 74	0.1092		0.2143	0.0000	0.0000	0.0284	0.0170
	75 - 79	0.0937		0.1812	0.0000	0.0412	0.0183	0.0210
	80 - 84	0.0518		0.3241	0.0056	0.0030	0.0431	0.0152
	85 +	0.0058		0.2640	0.1718	0.0314	0.0000	0.0000
1-2 ADLs	65 - 69	0.0753	0.0302		0.1279	0.0187	0.0030	0.0086
	70 - 74	0.0431	0.1117		0.1626	0.0164	0.0056	0.0000
	75 - 79	0.0575	0.0344		0.2081	0.0265	0.0223	0.0134
	80 - 84	0.0215	0.0959		0.1476	0.0585	0.0959	0.0325
	85 +	0.0000	0.0887		0.3329	0.0000	0.1105	0.0000
3-4 ADLs	65 - 69	0.0323	0.0723	0.0711		0.0938	0.0645	0.0107
	70 - 74	0.0247	0.0000	0.0803		0.1248	0.0656	0.0257
	75 - 79	0.0000	0.0519	0.0885		0.1356	0.0811	0.0058
	80 - 84	0.0168	0.0000	0.1058		0.1140	0.1126	0.0246
	85 +	0.0906	0.0000	0.2898		0.2054	0.1367	0.1034
5-6 ADLs	65 - 69	0.0097	0.0000	0.0384	0.0372		0.0762	0.0281
	70 - 74	0.0204	0.0202	0.0517	0.0487		0.0320	0.0314
	75 - 79	0.0416	0.0000	0.0187	0.0801		0.0896	0.0340
	80 - 84	0.0125	0.0109	0.0096	0.0846		0.0901	0.0425
	85 +	0.0368	0.0046	0.0261	0.0533		0.1048	0.0445
Inst'd	65 - 69	0.0159	0.0161	0.0159	0.0000	0.0000		0.0150
	70 - 74	0.0257	0.0029	0.0139	0.0013	0.0028		0.0237
	75 - 79	0.0119	0.0000	0.0155	0.0024	0.0000		0.0428
	80-84	0.0093	0.0079	0.0036	0.0102	0.0086		0.0229
	85 +	0.0133	0.0036	0.0032	0.0016	0.0006		0.0350

Table G.126: The constrained MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for females using 5 year age groupings.

1984				19	89 Statu	s		
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0148	0.0049	0.0022	0.0040	0.0035	0.0021
	70 - 74		0.0220	0.0135	0.0041	0.0043	0.0052	0.0025
	75 - 79		0.0265	0.0266	0.0049	0.0043	0.0095	0.0050
	80 - 84		0.0296	0.0590	0.0061	0.0144	0.0159	0.0090
	85 +		0.0503	0.0412	0.0331	0.0101	0.0528	0.0029
IADL	65 - 69	0.1008		0.2046	0.0000	0.0178	0.0122	0.0043
only	70 - 74	0.0939		0.1539	0.0056	0.0220	0.0194	0.0130
	75 - 79	0.0540		0.1973	0.0105	0.0163	0.0368	0.0080
	80 - 84	0.0183		0.2267	0.0364	0.0314	0.0201	0.0183
	85 +	0.0090		0.2246	0.0000	0.0801	0.0498	0.0358
1 - 2	65 - 69	0.0637	0.0521		0.0949	0.0065	0.0330	0.0043
ADLs	70 - 74	0.0124	0.0432		0.0754	0.0253	0.0154	0.0055
	75 - 79	0.0196	0.0494		0.1286	0.0624	0.0305	0.0032
	80 - 84	0.0360	0.0732		0.1504	0.0044	0.0990	0.0000
	85 +	0.0113	0.0376		0.1693	0.0400	0.0872	0.0164
3 - 4	65 - 69	0.0000	0.0658	0.1018		0.1551	0.0179	0.0170
ADLs	70 - 74	0.0244	0.0000	0.0644		0.0773	0.0188	0.0223
	75 - 79	0.0162	0.0127	0.0632		0.0764	0.0820	0.0170
	80 - 84	0.0000	0.0000	0.0983		0.0934	0.0828	0.0281
	85 +	0.0265	0.0152	0.0771		0.1084	0.0866	0.0491
5-6	65 - 69	0.0099	0.0000	0.1357	0.1330		0.0402	0.0128
ADLs	70 - 74	0.0000	0.0375	0.0433	0.1155		0.0871	0.0000
	75 - 79	0.0055	0.0226	0.0935	0.0316		0.0620	0.0455
	80 - 84	0.0397	0.0000	0.0000	0.0576		0.0526	0.0399
	85 +	0.0033	0.0000	0.0259	0.0329		0.1266	0.0412
Inst'd	65 - 69	0.0121	0.0000	0.0063	0.0095	0.0106		0.0214
	70 - 74	0.0054	0.0002	0.0097	0.0000	0.0051		0.0177
	75 - 79	0.0052	0.0000	0.0147	0.0000	0.0000		0.0254
	80 - 84	0.0067	0.0075	0.0000	0.0003	0.0000		0.0300
	85 +	0.0059	0.0011	0.0012	0.0003	0.0050		0.0546

Table G.127: The constrained MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for females using year age groupings.

1989				19	94 Statu	S		
status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0133	0.0189	0.0000	0.0028	0.0048	0.0034
	70 - 74		0.0161	0.0248	0.0003	0.0031	0.0093	0.0013
	75 - 79		0.0281	0.0344	0.0000	0.0061	0.0197	0.0017
	80 - 84		0.0646	0.0221	0.0262	0.0067	0.0199	0.0097
	85 +		0.0461	0.0694	0.0000	0.0509	0.0597	0.0061
IADL only	65 - 69	0.1616		0.0619	0.0489	0.0415	0.0000	0.0000
	70 - 74	0.0694		0.2160	0.0000	0.0000	0.0200	0.0184
	75 - 79	0.0910		0.1856	0.0000	0.0539	0.0196	0.0159
	80 - 84	0.0835		0.4002	0.0000	0.0000	0.0578	0.0350
	85 +	0.0161		0.2375	0.2106	0.0017	0.0000	0.0000
1-2 ADLs	65 - 69	0.0722	0.0445		0.1051	0.0350	0.0000	0.0031
	70 - 74	0.0424	0.0999		0.1669	0.0312	0.0020	0.0030
	75 - 79	0.0629	0.0434		0.1990	0.0161	0.0103	0.0166
	80 - 84	0.0187	0.1087		0.1248	0.0726	0.0858	0.0159
	85 +	0.0000	0.1002		0.3196	0.0000	0.1021	0.0000
3–4 ADLs	65 - 69	0.0087	0.0813	0.0167		0.0385	0.0863	0.0149
	70 - 74	0.0327	0.0000	0.1006		0.0808	0.0725	0.0153
	75 - 79	0.0019	0.0481	0.0945		0.1142	0.0615	0.0130
	80 - 84	0.0132	0.0000	0.1192		0.1070	0.1181	0.0000
	85 +	0.0789	0.0000	0.1895		0.2017	0.1776	0.0946
5-6 ADLs	65 - 69	0.0115	0.0000	0.0000	0.0613		0.0583	0.0206
	70 - 74	0.0230	0.0104	0.0564	0.0719		0.0062	0.0386
	75 - 79	0.0213	0.0000	0.0424	0.0660		0.0793	0.0189
	80 - 84	0.0120	0.0156	0.0000	0.0992		0.1130	0.0579
	85 +	0.0469	0.0000	0.0423	0.0514		0.0896	0.0379
Inst'd	65 - 69	0.0275	0.0015	0.0319	0.0000	0.0006		0.0136
	70 - 74	0.0200	0.0021	0.0148	0.0000	0.0020		0.0237
	75 - 79	0.0139	0.0000	0.0094	0.0050	0.0000		0.0473
	80 - 84	0.0071	0.0120	0.0023	0.0043	0.0132		0.0226
	85 +	0.0123	0.0018	0.0025	0.0010	0.0002		0.0299
Table G.128: The constrained MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males using 5 year age groupings.

1984				19	89 Statu	S		
status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0127	0.0122	0.0008	0.0015	0.0017	0.0043
	70 - 74		0.0173	0.0079	0.0021	0.0008	0.0050	0.0045
	75 - 79		0.0269	0.0123	0.0071	0.0008	0.0126	0.0095
	80 - 84		0.0384	0.0312	0.0184	0.0024	0.0125	0.0171
	85 +		0.0965	0.0157	0.0000	0.0021	0.0299	0.0182
IADL	65 - 69	0.1501		0.1837	0.0000	0.0135	0.0223	0.0207
only	70 - 74	0.0459		0.0951	0.0460	0.0217	0.0422	0.0180
	75 - 79	0.0981		0.0909	0.1050	0.0678	0.0034	0.0130
	80 - 84	0.0427		0.0563	0.0321	0.2079	0.0195	0.0360
	85 +	0.0491		0.2636	0.1636	0.0509	0.0435	0.0525
1 - 2	65 - 69	0.0063	0.1717		0.1586	0.0399	0.0107	0.0009
ADLs	70 - 74	0.0618	0.0522		0.1107	0.0407	0.0072	0.0117
	75 - 79	0.0308	0.0971		0.1402	0.0161	0.0180	0.0189
	80 - 84	0.0496	0.0934		0.0000	0.0000	0.0878	0.0087
	85 +	0.0000	0.2103		0.1114	0.1367	0.0139	0.0300
3 - 4	65 - 69	0.0385	0.0000	0.2251		0.1287	0.0160	0.0169
ADLs	70 - 74	0.0000	0.0250	0.1914		0.0596	0.0712	0.0133
	75 - 79	0.0000	0.0000	0.1254		0.2545	0.0760	0.0512
	80 - 84	0.0000	0.0000	0.0774		0.2134	0.2334	0.0000
	85 +	0.0000	0.0000	0.0001		0.0022	0.1953	0.0998
5-6	65 - 69	0.0163	0.0197	0.0485	0.0699		0.0501	0.0185
ADLs	70 - 74	0.0016	0.0200	0.0740	0.0540		0.0320	0.0089
	75 - 79	0.0000	0.0116	0.1012	0.0717		0.0645	0.0083
	80 - 84	0.0000	0.0000	0.0592	0.5061		0.0000	0.0000
	85 +	0.0000	0.0000	0.0000	0.0644		0.1134	0.0331
Inst'd	65 - 69	0.0025	0.0152	0.0000	0.0009	0.0005		0.0197
	70 - 74	0.0077	0.0185	0.0000	0.0145	0.0000		0.0352
	75 - 79	0.0075	0.0000	0.0032	0.0019	0.0000		0.0841
	80-84	0.0144	0.0000	0.0000	0.0192	0.0000		0.1000
	85 +	0.0168	0.0000	0.0033	0.0010	0.0010		0.0420

Table G.129: The constrained MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males using 5 year age groupings.

1989		1994 Status						
status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		0.0121	0.0096	0.0027	0.0038	0.0033	0.0023
	70 - 74		0.0232	0.0140	0.0052	0.0030	0.0040	0.0086
	75 - 79		0.0358	0.0242	0.0090	0.0045	0.0035	0.0072
	80 - 84		0.0208	0.0486	0.0000	0.0073	0.0144	0.0130
	85 +		0.0131	0.0575	0.0000	0.0213	0.0677	0.0200
IADL only	65 - 69	0.1486		0.0597	0.0631	0.0000	0.0119	0.0348
	70 - 74	0.1767		0.1782	0.0211	0.0084	0.0409	0.0092
	75 - 79	0.1019		0.1263	0.0371	0.0000	0.0081	0.0353
	80 - 84	0.0094		0.1755	0.0132	0.0192	0.0235	0.0000
	85 +	0.0000		0.0000	0.5805	0.0000	0.0000	0.0000
1-2 ADLs	65 - 69	0.0827	0.0143		0.0779	0.0000	0.0403	0.0175
	70 - 74	0.0461	0.1142		0.1089	0.0000	0.0224	0.0000
	75 - 79	0.0257	0.0262		0.0948	0.1084	0.0784	0.0000
	80 - 84	0.0226	0.0793		0.2070	0.0118	0.1323	0.0743
	85 +	0.0538	0.0248		0.0899	0.2037	0.0000	0.0134
3-4 ADLs	65 - 69	0.0661	0.0462	0.0986		0.1436	0.0000	0.0000
	70 - 74	0.0092	0.0126	0.0184		0.1992	0.0306	0.0493
	75 - 79	0.0029	0.0423	0.0000		0.1610	0.1603	0.0000
	80 - 84	0.0288	0.0000	0.0000		0.1031	0.0800	0.1464
	85 +	0.0000	0.1643	0.7154		0.0000	0.0000	0.1001
5-6 ADLs	65 - 69	0.0091	0.0000	0.1129	0.0000		0.0710	0.0253
	70 - 74	0.0043	0.0535	0.0200	0.0000		0.0873	0.0181
	75 - 79	0.0831	0.0000	0.0000	0.0617		0.1098	0.0599
	80 - 84	0.0144	0.0000	0.0437	0.0290		0.0373	0.0032
	85 +	0.0000	0.0000	0.0000	0.0000		0.2885	0.1228
Inst'd	65 - 69	0.0023	0.0280	0.0000	0.0000	0.0000		0.0159
	70 - 74	0.0408	0.0051	0.0111	0.0067	0.0027		0.0228
	75 - 79	0.0066	0.0018	0.0224	0.0000	0.0000		0.0250
	80-84	0.0145	0.0027	0.0050	0.0252	0.0000		0.0167
	85 +	0.0221	0.0138	0.0033	0.0082	0.0055		0.0741

## Appendix H

Log-likelihood values for the maximum likelihood estimates, adjusted maximum likelihood estimates and the constrained maximum likelihood estimates of the transition intensities calculated from the 1984–89 and 1989–94 NLTCS

Tables H.130 and H.131 show for the transition intensities between disability states calculated from the 1984–89 and 1989–94 NLTCS, respectively, the log-likelihood (Equation 4.35) for: the unconstrained maximum likelihood estimates (transformed from transition probabilities in Section 4.3) and the constrained (real) MLEs (I refer to these as original MLEs); the unconstrained MLEs (and constrined (real) MLEs) with all negative transition intensities set to zero (I refer to these as adjusted MLEs); and for the constrained (positive) MLEs (given in Section 4.7). The same likelihood Table H.130: Comparison of log-likelihood values for the unconstrained MLEs (and constrained (real) MLEs), adjusted MLEs and the constrained (positive) MLEs of the transition intensities calculated from the 1984-1989 NLTCS.

Gende	r Age	Log-	likelihood va	alue for:
	group	Original	Adjusted	Constrained
		MLE	MLE	MLE
Males	65 - 69	-2975.92	-2980.42	-2976.64
and	70 - 74	-3551.98	-3552.12	-3552.07
Female	es 65–74	-6566.07	-6566.08	-6566.07
	75 - 79	-3561.64	-3562.47	-3562.05
	80 - 84	-2446.15	-2447.63	-2446.87
	75 - 84	-6069.67	-6069.88	-6069.77
	85 +	-1592.23	-1592.30	-1592.26
Female	es 65–69	-1736.24	-1740.28	-1737.64
	70 - 74	-2266.43	-2267.63	-2267.12
	65 - 74	-4043.75	-4043.78	-4043.75
	75 - 79	-2345.92	-2346.80	-2346.44
	80 - 84	-1762.77	-1763.70	-1763.20
	75 - 84	-4153.31	-4153.51	-4153.44
	85 +	-1245.85	-1247.47	-1246.27
Males	65 - 69	-1223.94	-1224.02	-1224.01
	70 - 74	-1255.02	-1255.53	-1255.26
	65 - 74	-2493.78	-2496.47	-2494.29
	75 - 79	-1185.60	-1192.50	-1187.13
	80 - 84	-658.30	-659.70	-659.38
	75 - 84	-1867.18	-1886.68	-1869.33
	85 +	-332.13	-332.41	-332.34

values for the 1982–84 NLTCS are given in Table 4.53.

Gender	Age	Log-	likelihood va	alue for:
	group	Original	Adjusted	Constrained
		MLE	MLE	MLE
Males	65 - 69	-1862.61	-1862.50	-1862.03
and	70 - 74	-2872.68	-2873.63	-2872.91
Females	65 - 74	-4765.33	-4765.33	-4765.33
	75 - 79	-3200.60	-3202.39	-3201.06
	80 - 84	-2522.97	-2523.29	-2523.05
	75 - 84	-5784.29	-5785.67	-5784.45
	85 +	-1462.64	-1475.80	-1464.12
Females	65 - 69	-1148.09	-1154.55	-1151.00
	70 - 74	-1710.73	-1714.46	-1711.80
	65 - 74	-2887.95	-2888.85	-2888.20
	75 - 79	-2119.99	-2129.32	-2121.20
	80 - 84	-1830.69	-1833.06	-1830.99
	75 - 84	-4009.23	-4017.13	-4009.95
	85 +	-1121.22	-1128.02	-1122.36
Males	65 - 69	-693.96	-696.37	-695.12
	70 - 74	-1136.63	-1138.28	-1137.11
	65 - 74	-1853.20	-1854.26	-1853.78
	75 - 79	-1056.70	-1059.11	-1057.53
	80 - 84	-663.75	-666.37	-664.89
	75 - 84	-1745.57	-1746.96	-1746.21
	85 +	-319.13	-360.93	-321.90

Table H.131: Comparison of log-likelihood values for the unconstrained MLEs (and constrained (real) MLEs), adjusted MLEs and the constrained (positive) MLEs of the transition intensities calculated from the 1989-1994 NLTCS.

## Appendix I

5-year transition probabilities between disability states calculated from the constrained transition intensities of the 1984, 1989 and 1994 National Long-Term Care Surveys

The tables in this appendix give the 5-year transition probabilities between disability states calculated from the constrained (positive) MLEs of the transition intensities, based on data grouped in 5-year age bands (see Section 4.5) as a percentage for men, women and in aggregate, calculated using Equation 3.32. They are given for the 1984–89 NLTCS in Tables I.132 to I.134 and for the 1989–94 NLTCS in Tables I.135 to I.137.

Table I.132: The 5-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys as a percentage, for males using 5 year age groupings.

				198	39 Status	5		
1984 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	86.43	3.67	3.99	1.22	1.10	1.20	2.39
	70 - 74	84.08	4.74	3.30	1.51	0.88	2.77	2.71
	75 - 79	72.79	5.64	4.53	3.01	2.55	5.29	6.18
	80 - 84	57.03	7.17	8.66	5.22	3.86	8.39	9.66
	85 +	46.46	10.39	6.63	5.99	3.97	14.07	12.48
IADL only	65 - 69	33.94	22.07	18.73	6.58	5.53	6.69	6.47
	70 - 74	15.48	29.78	17.49	9.75	6.80	12.75	7.96
	75 - 79	20.19	19.28	15.57	11.84	15.80	8.09	9.22
	80 - 84	8.71	16.54	12.49	17.24	15.67	16.60	12.75
	85 +	5.77	9.64	10.60	16.62	10.58	23.95	22.84
1-2 ADLs	65 - 69	14.98	17.12	29.73	15.36	11.92	6.81	4.08
	70 - 74	17.73	9.52	34.44	14.79	9.98	7.60	5.94
	75 - 79	10.80	10.48	29.19	14.19	13.63	10.06	11.65
	80 - 84	12.89	11.08	33.78	4.69	4.88	21.02	11.66
	85 +	2.58	8.31	13.54	16.48	16.27	22.66	20.16
3-4 ADLs	65 - 69	13.31	9.38	22.75	21.56	18.47	8.12	6.41
	70 - 74	7.13	7.43	25.00	24.52	11.82	16.87	7.22
	75 - 79	3.03	4.00	16.65	15.71	24.84	18.22	17.56
	80 - 84	3.39	2.17	11.11	20.44	11.58	37.25	14.06
	85 +	1.79	0.25	0.32	22.84	0.50	42.20	32.09
5-6 ADLs	65 - 69	9.14	6.55	12.52	10.93	37.60	15.41	7.87
	70 - 74	4.79	6.31	17.50	11.55	42.17	12.03	5.66
	75 - 79	3.21	4.59	16.95	10.91	35.40	18.30	10.65
	80 - 84	2.92	2.38	13.25	26.47	17.46	29.35	8.17
	85 +	1.36	0.18	0.24	9.26	35.02	35.56	18.38
Inst'd	65 - 69	2.63	3.30	1.26	0.45	0.41	82.69	9.25
	70 - 74	3.83	4.57	2.29	3.45	0.85	69.74	15.27
	75 - 79	2.61	0.25	0.83	0.49	0.31	61.91	33.60
	80 - 84	3.98	0.36	0.93	2.98	1.03	53.52	37.20
	85 +	4.93	0.81	0.87	0.65	0.59	73.49	18.67

Table I.133: The 5-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys as a percentage, for females using 5 year age groupings.

				198	39 Status	3		
1984 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	87.11	3.64	3.37	1.26	1.38	1.96	1.28
	70 - 74	79.06	5.42	6.35	2.50	1.84	3.15	1.68
	75 - 79	69.72	6.16	8.86	3.51	2.67	5.94	3.13
	80 - 84	53.28	6.56	12.72	5.87	4.91	11.23	5.44
	85 +	39.82	7.00	10.04	7.75	6.17	21.66	7.56
IADL only	65 - 69	29.78	22.81	27.43	6.61	4.09	6.73	2.55
	70 - 74	22.76	25.22	26.58	7.35	5.28	7.93	4.88
	75 - 79	13.43	23.88	25.72	9.92	7.44	14.65	4.97
	80 - 84	6.88	21.76	23.57	13.55	7.80	18.31	8.13
	85 +	2.70	15.71	19.56	8.93	13.37	23.61	16.12
1-2 ADLs	65 - 69	21.21	8.98	37.22	13.08	5.86	10.36	3.27
	70 - 74	7.40	7.73	48.02	17.27	7.49	8.19	3.91
	75 - 79	7.16	7.37	32.13	18.79	12.50	16.55	5.49
	80 - 84	7.59	7.91	24.82	17.86	6.44	29.22	6.16
	85 +	3.50	4.02	22.14	16.23	11.43	28.34	14.34
3-4 ADLs	65 - 69	8.95	9.18	23.48	26.00	16.93	9.15	6.31
	70 - 74	8.06	2.66	15.66	42.02	13.18	10.29	8.13
	75 - 79	5.55	3.66	13.12	30.88	12.68	24.93	9.17
	80 - 84	3.55	2.31	11.31	28.17	15.25	26.99	12.42
	85 +	4.52	2.42	9.18	20.99	15.32	27.74	19.83
5-6 ADLs	65 - 69	9.02	5.18	23.80	17.83	25.82	12.57	5.79
	70 - 74	4.25	5.75	13.64	20.23	28.85	23.85	3.43
	75 - 79	3.66	4.75	15.96	9.17	31.02	20.02	15.41
	80 - 84	9.59	1.08	3.17	9.65	41.43	19.07	16.01
	85 +	1.56	0.64	4.30	5.44	34.02	34.47	19.57
Inst'd	65 - 69	5.53	0.72	3.16	3.02	3.06	74.89	9.62
	70 - 74	2.41	0.47	3.31	0.89	1.50	83.20	8.21
	75 - 79	2.20	0.47	3.77	1.01	0.61	80.39	11.55
	80 - 84	2.34	1.75	0.98	0.46	0.27	80.53	13.67
	85 +	1.63	0.37	0.58	0.37	1.44	72.08	23.52

Table I.134: The 5-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys as a percentage, for males and females using 5 year age groupings.

				198	39 Status	3		
1984 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	86.81	3.65	3.65	1.24	1.25	1.62	1.77
	70 - 74	81.10	5.15	5.11	2.10	1.45	2.99	2.10
	75 - 79	70.87	5.96	7.24	3.33	2.63	5.70	4.27
	80 - 84	54.51	6.76	11.39	5.66	4.56	10.29	6.84
	85 +	41.57	8.18	9.14	7.03	5.59	19.58	8.92
IADL only	65 - 69	31.87	21.59	24.40	6.81	4.58	6.73	4.03
	70 - 74	20.51	26.51	23.91	8.03	5.77	9.42	5.86
	75 - 79	15.37	22.55	22.81	10.47	9.84	12.76	6.19
	80 - 84	7.35	20.38	20.68	14.89	9.46	17.85	9.40
	85 +	3.67	13.54	18.60	10.19	12.75	23.62	17.65
1-2 ADLs	65 - 69	19.10	12.58	33.91	13.71	8.04	9.11	3.55
	70 - 74	10.74	8.27	43.56	16.58	8.25	8.01	4.58
	75 - 79	8.36	8.11	31.13	17.68	12.80	14.92	7.01
	80 - 84	8.56	8.42	26.09	15.61	6.46	28.01	6.85
	85 +	3.13	5.02	19.86	17.09	12.20	27.37	15.32
3-4 ADLs	65 - 69	9.15	8.34	25.90	23.96	17.33	8.97	6.35
	70 - 74	7.62	4.41	19.59	34.91	12.74	12.71	8.02
	75 - 79	4.34	3.14	15.52	26.07	16.22	23.00	11.72
	80 - 84	3.55	2.35	11.10	24.65	15.19	29.43	13.73
	85 +	4.20	2.07	8.20	20.64	13.39	29.87	21.63
5-6 ADLs	65 - 69	9.58	5.01	19.27	16.43	29.76	13.44	6.50
	70 - 74	3.98	6.37	15.15	17.28	34.35	19.08	3.79
	75 - 79	2.93	5.56	15.73	9.94	32.79	19.44	13.60
	80 - 84	7.59	1.40	5.85	15.26	35.76	21.04	13.10
	85 +	1.49	0.63	3.45	6.30	34.11	34.66	19.37
Inst'd	65 - 69	4.26	1.95	2.22	1.87	1.87	78.38	9.45
	70 - 74	2.87	1.71	2.85	1.59	1.34	79.36	10.28
	75 - 79	2.37	0.46	3.01	0.83	0.50	75.30	17.54
	80-84	2.71	1.50	0.88	0.95	0.41	76.17	17.38
	85 +	2.13	0.40	0.62	0.41	1.32	72.24	22.87

Table I.135: The 5-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys as a percentage, for males using 5 year age groupings.

				199	94 Status	3		
1989 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	86.84	3.05	3.52	1.35	1.41	2.09	1.75
	70 - 74	78.34	5.08	5.13	2.31	1.87	2.77	4.49
	75 - 79	69.04	7.82	6.67	3.52	3.20	5.11	4.63
	80 - 84	60.69	6.01	7.79	3.81	3.35	9.41	8.93
	85 +	43.11	2.40	10.19	2.31	6.49	20.69	14.80
IADL only	65 - 69	39.40	22.65	11.32	7.96	3.40	5.00	10.27
	70 - 74	37.01	17.57	18.35	7.14	4.64	10.17	5.11
	75 - 79	23.47	24.59	14.73	7.61	6.45	11.76	11.39
	80 - 84	4.93	33.87	15.68	10.26	7.23	17.30	10.73
	85 +	4.85	12.04	32.42	13.74	12.60	7.46	16.89
1-2 ADLs	65 - 69	27.36	4.14	35.69	10.16	4.12	11.78	6.75
	70 - 74	20.65	12.29	30.24	13.91	8.43	9.77	4.71
	75 - 79	11.30	4.64	21.62	11.06	15.45	29.39	6.54
	80 - 84	6.30	7.25	10.55	14.63	7.95	26.69	26.62
	85 +	8.83	3.46	24.39	5.06	17.63	20.17	20.45
3-4 ADLs	65 - 69	23.60	5.92	19.02	20.73	18.59	7.82	4.32
	70 - 74	6.98	5.21	5.54	21.74	29.58	15.22	15.73
	75 - 79	8.26	4.87	3.32	19.21	16.14	39.83	8.37
	80 - 84	6.53	0.65	1.87	19.06	16.79	18.47	36.62
	85 +	6.64	5.56	28.33	7.72	15.30	12.66	23.79
5-6 ADLs	65 - 69	9.20	2.05	19.26	3.19	34.42	21.70	10.18
	70 - 74	9.54	7.89	7.03	2.09	41.07	24.04	8.33
	75 - 79	17.88	2.13	2.39	6.63	23.84	28.75	18.38
	80 - 84	5.68	1.46	5.87	7.82	55.23	16.45	7.48
	85 +	2.65	1.13	1.45	0.98	13.64	41.54	38.60
Inst'd	65 - 69	4.07	6.23	1.03	0.84	0.22	79.72	7.90
	70 - 74	15.58	2.07	3.40	2.14	1.74	65.02	10.05
	75 - 79	3.11	0.90	4.93	1.04	1.30	77.44	11.28
	80 - 84	5.40	1.10	1.39	5.43	1.65	74.70	10.35
	85 +	5.58	2.13	3.56	1.97	2.02	55.20	29.54

Table I.136: The 5-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys as a percentage, for females using 5 year age groupings.

				199	94 Status	3		
1989 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	83.50	3.61	5.34	1.67	1.77	2.31	1.80
	70 - 74	78.18	4.76	7.38	2.49	1.65	4.16	1.38
	75 - 79	67.22	5.90	9.14	3.67	3.12	8.20	2.74
	80 - 84	50.45	8.27	10.45	6.88	4.20	13.33	6.41
	85 +	34.22	6.34	10.86	6.58	9.99	25.18	6.84
IADL only	65 - 69	40.15	24.40	9.86	10.01	9.16	4.63	1.79
	70 - 74	20.80	26.27	25.38	10.44	3.91	6.81	6.39
	75 - 79	21.96	19.86	20.50	10.70	10.00	9.50	7.48
	80 - 84	12.05	12.00	23.26	9.71	7.60	23.90	11.48
	85 +	6.89	14.12	18.57	15.63	10.08	24.45	10.26
1-2 ADLs	65 - 69	24.02	9.11	30.34	17.51	9.53	6.62	2.87
	70 - 74	16.05	11.89	29.73	21.08	9.44	7.26	4.55
	75 - 79	15.50	7.31	26.74	22.13	10.28	10.34	7.70
	80 - 84	6.50	7.23	21.89	13.77	11.70	29.30	9.60
	85 +	6.07	6.32	17.18	13.40	9.94	34.91	12.17
3-4 ADLs	65 - 69	12.10	11.27	5.95	33.00	9.39	22.19	6.10
	70 - 74	12.42	3.84	14.88	29.67	13.86	17.47	7.86
	75 - 79	7.00	6.32	14.46	27.50	17.45	18.66	8.61
	80 - 84	4.56	3.37	11.89	23.57	14.69	34.94	6.98
	85 +	8.71	2.92	9.38	8.70	14.02	35.43	20.84
5-6 ADLs	65 - 69	6.43	2.29	2.06	12.06	48.43	20.00	8.73
	70 - 74	10.30	4.11	12.34	14.67	39.38	5.41	13.78
	75 - 79	7.93	2.18	8.97	12.18	35.90	22.64	10.20
	80 - 84	3.85	2.47	4.65	11.62	26.94	32.05	18.42
	85 +	9.05	1.95	6.87	5.92	30.15	30.17	15.89
Inst'd	65 - 69	12.48	1.44	7.77	1.93	1.15	69.27	5.96
	70 - 74	8.27	1.37	3.94	1.34	1.00	73.61	10.47
	75 - 79	5.16	0.56	2.53	1.91	0.61	69.19	20.03
	80 - 84	2.87	2.21	2.31	2.01	3.56	76.16	10.88
	85 +	3.34	0.69	1.10	0.66	0.60	79.97	13.64

Table I.137: The 5-year transition probabilities, calculated from the constrained (positive) MLEs of the transition intensities, between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys as a percentage, for males and females using 5 year age groupings.

				199	94 Status	3		
1989 Status	Age		IADL	1 - 2	3 - 4	5-6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69	84.89	3.37	4.54	1.56	1.62	2.22	1.79
	70 - 74	78.25	4.89	6.43	2.42	1.75	3.58	2.69
	75 - 79	67.87	6.65	8.39	3.42	3.18	7.03	3.46
	80 - 84	53.82	7.53	9.65	5.78	3.93	12.05	7.24
	85 +	37.08	5.11	10.44	5.40	8.85	23.72	9.41
IADL only	65 - 69	40.98	23.12	10.41	10.79	7.20	4.51	2.98
	70 - 74	27.58	22.46	23.10	9.20	3.52	8.21	5.93
	75 - 79	22.57	20.54	19.16	10.14	8.72	10.19	8.67
	80 - 84	10.40	16.84	22.61	10.25	7.14	22.45	10.32
	85 +	6.12	13.67	20.99	14.24	11.57	22.67	10.74
1-2 ADLs	65 - 69	25.39	7.11	31.97	16.57	8.17	6.40	4.39
	70 - 74	17.50	12.19	29.23	19.27	9.35	8.07	4.39
	75 - 79	14.51	6.59	24.33	21.09	11.54	14.64	7.30
	80 - 84	6.43	7.36	19.05	14.08	11.28	28.63	13.17
	85 +	6.30	5.89	18.92	12.45	10.04	33.10	13.30
3-4 ADLs	65 - 69	17.10	9.04	12.07	23.61	14.96	17.49	5.73
	70 - 74	10.75	3.89	12.54	26.42	19.11	16.71	10.58
	75 - 79	7.52	6.18	12.36	24.63	17.72	23.45	8.13
	80 - 84	5.11	2.92	9.79	22.08	15.72	31.83	12.54
	85 +	9.12	3.27	11.76	8.66	13.64	31.43	22.12
5-6 ADLs	65 - 69	7.36	2.34	8.56	7.43	40.96	22.54	10.81
	70 - 74	10.16	5.19	11.22	10.49	39.61	11.21	12.13
	75 - 79	11.32	2.18	5.93	11.12	30.93	24.76	13.74
	80 - 84	4.28	2.35	4.67	11.28	32.89	28.30	16.23
	85 +	7.96	1.86	6.06	5.07	29.43	31.14	18.47
Inst'd	65 - 69	9.12	3.87	4.54	1.59	0.71	73.43	6.74
	70 - 74	10.40	1.59	3.85	1.51	1.24	70.99	10.42
	75 - 79	4.70	0.60	3.50	1.81	0.70	70.37	18.31
	80-84	3.44	1.88	2.21	2.94	3.00	75.64	10.89
	85 +	3.70	0.95	1.48	0.84	0.80	76.47	15.77

### Appendix J

Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1982, 1984, 1989 and 1994 National Long-Term Care Surveys

The tables in this appendix give the variance estiamtes of the constrained (positive) MLEs of the annual transition intensities between disability states for men, women and in aggregate (as described in Section 5.4). They are given for all pairs of surveys in 10-year age bands (65–74 years, 75–84 years and 85+ years) in Tables J.138 – J.146 and in 5-year age bands (65–69 years, 70–74 years, 75–79 years and 80–84 years) in Tables J.147 – J.152 (except for those from the 1982–84 NLTCS, which are given in Tables 5.61 – 5.63).

Table J.138: Variance estimates of the MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males and females using 10 year age groupings.

1982					1984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$3.88 \times 10^{-6}$	$3.24 \times 10^{-6}$	$1.42 \times 10^{-6}$	$1.00 \times 10^{-6}$	$4.67 \times 10^{-7}$	$2.00 \times 10^{-6}$
	75 - 84		$2.46 \times 10^{-5}$	$2.50 \times 10^{-5}$	$1.12 \times 10^{-5}$	$6.58 \times 10^{-6}$	$5.22 \times 10^{-6}$	$1.12 \times 10^{-5}$
	85 +		$3.48 \times 10^{-4}$	$5.09 \times 10^{-4}$	$2.55 \times 10^{-4}$	$2.11 \times 10^{-4}$	$1.72 \times 10^{-4}$	$1.97 \times 10^{-4}$
IADL only	65 - 74	$3.56 \times 10^{-4}$		$9.63 \times 10^{-4}$	$3.40 \times 10^{-4}$	$1.91{\times}10^{-4}$	$6.99 \times 10^{-5}$	$1.36 \times 10^{-4}$
	75 - 84	$2.55{ imes}10^{-4}$		$1.83 \times 10^{-3}$	$8.04 \times 10^{-4}$	$4.66 \times 10^{-4}$	$2.06 \times 10^{-4}$	$2.63 \times 10^{-4}$
	85 +	$2.23 \times 10^{-4}$		$4.80 \times 10^{-3}$	$3.12 \times 10^{-3}$	$2.21 \times 10^{-3}$	$9.70 \times 10^{-4}$	$1.11 \times 10^{-3}$
1-2 ADLs	65 - 74	$2.09 \times 10^{-4}$	$7.43 \times 10^{-4}$		$1.08 \times 10^{-3}$	$4.37 \times 10^{-4}$	$1.10 \times 10^{-4}$	$2.02 \times 10^{-4}$
	75 - 84	$1.37{ imes}10^{-4}$	$7.67{ imes}10^{-4}$		$1.55{ imes}10^{-3}$	$6.65 { imes} 10^{-4}$	$2.00 \times 10^{-4}$	$2.98 \times 10^{-4}$
	85 +	$2.15{\times}10^{-4}$	$8.45 \times 10^{-4}$		$3.87 \times 10^{-3}$	$3.02 \times 10^{-3}$	$9.17{ imes}10^{-4}$	$8.47 \times 10^{-4}$
3–4 ADLs	65 - 74	$3.63 \times 10^{-4}$	$1.13 \times 10^{-3}$	$4.67 \times 10^{-3}$		$2.97 \times 10^{-3}$	$2.95 \times 10^{-4}$	$7.22 \times 10^{-4}$
	75 - 84	$2.27 \times 10^{-4}$	$1.32 \times 10^{-3}$	$5.11 \times 10^{-3}$		$5.52 \times 10^{-3}$	$1.02 \times 10^{-3}$	$1.26 \times 10^{-3}$
	85 +	$2.38 \times 10^{-4}$	$1.52 \times 10^{-3}$	$3.22 \times 10^{-3}$		$1.74 \times 10^{-2}$	$3.43 \times 10^{-3}$	$3.90 \times 10^{-3}$
5–6 ADLs	65 - 74	$1.92 \times 10^{-4}$	$5.34 \times 10^{-4}$	$1.18 \times 10^{-3}$	$1.53 \times 10^{-3}$		$2.85 \times 10^{-4}$	$7.35 \times 10^{-4}$
	75 - 84	$1.32 \times 10^{-4}$	$4.91 \times 10^{-4}$	$9.48 \times 10^{-4}$	$2.03 \times 10^{-3}$		$5.05 \times 10^{-4}$	$8.32 \times 10^{-4}$
	85 +	$1.35 \times 10^{-4}$	$2.72 \times 10^{-4}$	$1.22 \times 10^{-3}$	$3.30 \times 10^{-3}$		$1.28 \times 10^{-3}$	$2.24 \times 10^{-3}$
Inst'd	65 - 74	$7.50 \times 10^{-5}$	$4.04 \times 10^{-5}$	$9.30 \times 10^{-5}$	$1.09 \times 10^{-4}$	$6.82 \times 10^{-5}$		$3.61 \times 10^{-4}$
	75 - 84	$1.14 \times 10^{-5}$	$2.40 \times 10^{-5}$	$2.15 \times 10^{-5}$	$3.06 \times 10^{-5}$	$2.45 \times 10^{-5}$		$2.34 \times 10^{-4}$
	85 +	$6.68 \times 10^{-6}$	$6.91 \times 10^{-6}$	$6.36{ imes}10^{-6}$	$3.30 \times 10^{-5}$	$2.43 \times 10^{-5}$		$3.41 \times 10^{-4}$

Table J.139: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males and females using 10 year age groupings.

1984					1989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$2.35 \times 10^{-6}$	$2.18 \times 10^{-6}$	$1.00 \times 10^{-6}$	$5.78 \times 10^{-7}$	$3.08 \times 10^{-7}$	$1.88 \times 10^{-7}$
	75 - 84		$9.56 \times 10^{-6}$	$1.38 \times 10^{-5}$	$7.44 \times 10^{-6}$	$3.51 \times 10^{-6}$	$2.66 \times 10^{-6}$	$1.23 \times 10^{-6}$
	85 +		$2.59 \times 10^{-4}$	$3.32{\times}10^{-4}$	$2.02 \times 10^{-4}$	$7.22 \times 10^{-5}$	$9.13 \times 10^{-5}$	$2.98 \times 10^{-5}$
IADL only	65 - 74	$1.38{ imes}10^{-4}$		$6.46{ imes}10^{-4}$	$2.91{\times}10^{-4}$	$1.32{\times}10^{-4}$	$4.43 \times 10^{-5}$	$1.88 \times 10^{-5}$
	75 - 84	$1.07{ imes}10^{-4}$		$1.09 \times 10^{-3}$	$6.56{ imes}10^{-4}$	$2.79{ imes}10^{-4}$	$1.32{ imes}10^{-4}$	$4.11 \times 10^{-5}$
	85 +	$2.80 \times 10^{-4}$		$8.27 \times 10^{-3}$	$4.72 \times 10^{-3}$	$1.56 \times 10^{-3}$	$1.07 \times 10^{-3}$	$4.29 \times 10^{-4}$
1-2 ADLs	65 - 74	$6.12 \times 10^{-5}$	$2.28 \times 10^{-4}$		$4.00 \times 10^{-4}$	$1.55 \times 10^{-4}$	$3.67 \times 10^{-5}$	$1.32 \times 10^{-5}$
	75 - 84	$6.58 \times 10^{-5}$	$3.34{ imes}10^{-4}$		$7.93{ imes}10^{-4}$	$2.90 \times 10^{-4}$	$1.59{ imes}10^{-4}$	$3.71 \times 10^{-5}$
	85 +	$2.04 \times 10^{-4}$	$1.46 \times 10^{-3}$		$4.48 \times 10^{-3}$	$1.18 \times 10^{-3}$	$9.14 \times 10^{-4}$	$3.17 \times 10^{-4}$
3–4 ADLs	65 - 74	$9.90 \times 10^{-5}$	$4.37 \times 10^{-4}$	$1.18 \times 10^{-3}$		$7.36 \times 10^{-4}$	$1.28 \times 10^{-4}$	$5.68 \times 10^{-5}$
	75 - 84	$7.43 \times 10^{-5}$	$3.07 \times 10^{-4}$	$1.33 \times 10^{-3}$		$9.41 \times 10^{-4}$	$4.34 \times 10^{-4}$	$1.45 \times 10^{-4}$
	85 +	$3.93{ imes}10^{-4}$	$9.77 \times 10^{-4}$	$3.48 \times 10^{-3}$		$1.90 \times 10^{-3}$	$1.59 \times 10^{-3}$	$6.80 \times 10^{-4}$
5–6 ADLs	65 - 74	$8.52 \times 10^{-5}$	$3.49 \times 10^{-4}$	$8.93 \times 10^{-4}$	$1.12 \times 10^{-3}$		$1.97{\times}10^{-4}$	$4.65 \times 10^{-5}$
	75 - 84	$9.09 \times 10^{-5}$	$2.62 \times 10^{-4}$	$8.61 \times 10^{-4}$	$9.13 \times 10^{-4}$		$3.03 \times 10^{-4}$	$1.35 \times 10^{-4}$
	85 +	$7.62 \times 10^{-5}$	$1.32 \times 10^{-4}$	$5.97{\times}10^{-4}$	$9.15 \times 10^{-4}$		$9.68 \times 10^{-4}$	$3.53 \times 10^{-4}$
Inst'd	65 - 74	$1.32 \times 10^{-5}$	$2.50 \times 10^{-5}$	$3.02 \times 10^{-5}$	$2.43 \times 10^{-5}$	$1.71 \times 10^{-5}$		$2.88 \times 10^{-5}$
	75 - 84	$8.21 \times 10^{-6}$	$1.07 \times 10^{-5}$	$2.02 \times 10^{-5}$	$8.20 \times 10^{-6}$	$2.68 \times 10^{-6}$		$3.40 \times 10^{-5}$
	85 +	$1.42 \times 10^{-5}$	$8.59 \times 10^{-6}$	$1.27{\times}10^{-5}$	$7.72 \times 10^{-6}$	$1.14 \times 10^{-5}$		$6.21 \times 10^{-5}$

Table J.140: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males and females using 10 year age groupings.

1989					1994 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$3.98 \times 10^{-6}$	$5.07 \times 10^{-6}$	$3.00 \times 10^{-6}$	$9.93 \times 10^{-7}$	$6.82 \times 10^{-7}$	$3.30 \times 10^{-7}$
	75 - 84		$1.68 \times 10^{-5}$	$2.89 \times 10^{-5}$	$1.54 \times 10^{-5}$	$5.58 \times 10^{-6}$	$4.31 \times 10^{-6}$	$1.50 \times 10^{-6}$
	85 +		$2.43 \times 10^{-4}$	$1.41 \times 10^{-3}$	$2.71 \times 10^{-3}$	$3.06 \times 10^{-4}$	$1.33 \times 10^{-4}$	$5.23 \times 10^{-5}$
IADL only	65 - 74	$4.29 \times 10^{-4}$		$1.31 \times 10^{-3}$	$8.98 { imes} 10^{-4}$	$2.06 \times 10^{-4}$	$9.34{ imes}10^{-5}$	$4.02 \times 10^{-5}$
	75 - 84	$3.09 \times 10^{-4}$		$2.97{ imes}10^{-3}$	$1.74{ imes}10^{-3}$	$5.21 \times 10^{-4}$	$2.62{\times}10^{-4}$	$8.51 \times 10^{-5}$
	85 +	$1.68 \times 10^{-3}$		$4.51 \times 10^{-2}$	$8.65 \times 10^{-2}$	$7.16 \times 10^{-3}$	$2.17 \times 10^{-3}$	$1.06 \times 10^{-3}$
1-2 ADLs	65 - 74	$2.27{ imes}10^{-4}$	$6.78 \times 10^{-4}$		$1.38 \times 10^{-3}$	$3.60 \times 10^{-4}$	$1.22 \times 10^{-4}$	$3.68 \times 10^{-5}$
	75 - 84	$1.66{ imes}10^{-4}$	$6.43 { imes} 10^{-4}$		$2.21 \times 10^{-3}$	$7.03 { imes} 10^{-4}$	$3.03{ imes}10^{-4}$	$7.54 \times 10^{-5}$
	85 +	$3.97{ imes}10^{-3}$	$1.06 \times 10^{-2}$		$2.96 \times 10^{-1}$	$2.18{ imes}10^{-2}$	$3.31{ imes}10^{-3}$	$2.79 \times 10^{-3}$
3–4 ADLs	65 - 74	$3.28 \times 10^{-4}$	$7.78 \times 10^{-4}$	$1.60 \times 10^{-3}$		$1.12 \times 10^{-3}$	$4.50 \times 10^{-4}$	$1.42 \times 10^{-4}$
	75 - 84	$1.99 \times 10^{-4}$	$7.10 \times 10^{-4}$	$2.27 \times 10^{-3}$		$1.52 \times 10^{-3}$	$6.69 \times 10^{-4}$	$1.55 \times 10^{-4}$
	85 +	$1.03 \times 10^{-2}$	$2.14 \times 10^{-2}$	$2.86 \times 10^{-1}$		$5.32 \times 10^{-2}$	$8.10 \times 10^{-3}$	$6.98 \times 10^{-3}$
5–6 ADLs	65 - 74	$2.07 \times 10^{-4}$	$3.99{ imes}10^{-4}$	$9.24 \times 10^{-4}$	$1.01 \times 10^{-3}$		$3.79 { imes} 10^{-4}$	$1.76 \times 10^{-4}$
	75 - 84	$2.03 \times 10^{-4}$	$2.49 \times 10^{-4}$	$8.35 \times 10^{-4}$	$1.58 \times 10^{-3}$		$6.00 \times 10^{-4}$	$2.10 \times 10^{-4}$
	85 +	$9.69 \times 10^{-4}$	$9.02 \times 10^{-4}$	$5.85 \times 10^{-3}$	$1.18 \times 10^{-2}$		$1.59 \times 10^{-3}$	$6.76 \times 10^{-4}$
Inst'd	65 - 74	$7.86 \times 10^{-5}$	$7.58 \times 10^{-5}$	$1.15 \times 10^{-4}$	$4.91 \times 10^{-5}$	$1.88 \times 10^{-5}$		$5.38 \times 10^{-5}$
	75 - 84	$2.48 \times 10^{-5}$	$2.84 \times 10^{-5}$	$6.88 { imes} 10^{-5}$	$5.68 \times 10^{-5}$	$2.53 \times 10^{-5}$		$5.06 \times 10^{-5}$
	85 +	$3.87{ imes}10^{-5}$	$2.86{ imes}10^{-5}$	$6.83{ imes}10^{-5}$	$8.61 \times 10^{-5}$	$1.40 \times 10^{-5}$		$5.81 \times 10^{-5}$

Table J.141: Variance estimates of the MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for females using 10 year age groupings.

1982					1984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$7.99 \times 10^{-6}$	$6.07 \times 10^{-6}$	$2.26 \times 10^{-6}$	$1.58 \times 10^{-6}$	$7.46 \times 10^{-7}$	$2.34{\times}10^{-6}$
	75 - 84		$4.43 \times 10^{-5}$	$4.71 \times 10^{-5}$	$2.17 \times 10^{-5}$	$9.83 \times 10^{-6}$	$9.59 \times 10^{-6}$	$1.49 \times 10^{-5}$
	85 +		$5.35 \times 10^{-4}$	$6.77 \times 10^{-4}$	$2.99 \times 10^{-4}$	$3.20 \times 10^{-4}$	$2.79 \times 10^{-4}$	$2.49 \times 10^{-4}$
IADL only	65 - 74	$5.61{\times}10^{-4}$		$1.77{ imes}10^{-3}$	$5.98{ imes}10^{-4}$	$2.51 \times 10^{-4}$	$1.32{\times}10^{-4}$	$1.66{ imes}10^{-4}$
	75 - 84	$3.77 \times 10^{-4}$		$3.35 \times 10^{-3}$	$1.32 \times 10^{-3}$	$5.05 \times 10^{-4}$	$3.16 \times 10^{-4}$	$3.46 \times 10^{-4}$
	85 +	$3.45 \times 10^{-4}$		$8.18 \times 10^{-3}$	$4.51 \times 10^{-3}$	$3.78 \times 10^{-3}$	$1.65 \times 10^{-3}$	$1.54 \times 10^{-3}$
1-2 ADLs	65 - 74	$3.21{\times}10^{-4}$	$1.31 \times 10^{-3}$		$1.05 \times 10^{-3}$	$4.47{ imes}10^{-4}$	$1.46{ imes}10^{-4}$	$2.29{\times}10^{-4}$
	75 - 84	$2.00 \times 10^{-4}$	$1.12 \times 10^{-3}$		$2.09 \times 10^{-3}$	$7.84 \times 10^{-4}$	$3.08 \times 10^{-4}$	$3.08 \times 10^{-4}$
	85 +	$2.27{\times}10^{-4}$	$1.38 \times 10^{-3}$		$4.33 \times 10^{-3}$	$3.35{\times}10^{-3}$	$1.04 \times 10^{-3}$	$9.00 \times 10^{-4}$
3–4 ADLs	65 - 74	$4.14 \times 10^{-4}$	$2.11 \times 10^{-3}$	$7.00 \times 10^{-3}$		$3.81 \times 10^{-3}$	$4.77 \times 10^{-4}$	$7.49{\times}10^{-4}$
	75 - 84	$3.12 \times 10^{-4}$	$1.85 \times 10^{-3}$	$7.63 \times 10^{-3}$		$6.94 \times 10^{-3}$	$1.34 \times 10^{-3}$	$1.30 \times 10^{-3}$
	85 +	$3.36 \times 10^{-4}$	$2.08 \times 10^{-3}$	$3.74 \times 10^{-3}$		$1.93 { imes} 10^{-2}$	$4.09 \times 10^{-3}$	$3.98 \times 10^{-3}$
5–6 ADLs	65 - 74	$3.18{ imes}10^{-4}$	$1.12 \times 10^{-3}$	$1.83 \times 10^{-3}$	$2.19 \times 10^{-3}$		$5.51{\times}10^{-4}$	$1.24 \times 10^{-3}$
	75 - 84	$2.16 \times 10^{-4}$	$6.46 \times 10^{-4}$	$1.56 \times 10^{-3}$	$2.96 \times 10^{-3}$		$8.22 \times 10^{-4}$	$1.04 \times 10^{-3}$
	85 +	$1.92 \times 10^{-4}$	$4.00 \times 10^{-4}$	$1.40 \times 10^{-3}$	$3.51 \times 10^{-3}$		$1.65 \times 10^{-3}$	$2.55{\times}10^{-3}$
Inst'd	65 - 74	$1.09 { imes} 10^{-4}$	$5.92 \times 10^{-5}$	$1.04 \times 10^{-4}$	$1.90 { imes} 10^{-4}$	$8.65 \times 10^{-5}$		$4.84 \times 10^{-4}$
	75 - 84	$1.31 \times 10^{-5}$	$2.77{ imes}10^{-5}$	$2.71 \times 10^{-5}$	$3.02 \times 10^{-5}$	$3.09 \times 10^{-5}$		$2.68 { imes} 10^{-4}$
	85 +	$8.51 \times 10^{-6}$	$1.01 \times 10^{-5}$	$8.55 \times 10^{-6}$	$4.03 \times 10^{-5}$	$3.24 \times 10^{-5}$		$3.74{ imes}10^{-4}$

Table J.142: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for females using 10 year age groupings.

1984					1989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$4.18 \times 10^{-6}$	$3.51 \times 10^{-6}$	$1.79 \times 10^{-6}$	$1.53 \times 10^{-6}$	$6.03 \times 10^{-7}$	$2.57 \times 10^{-7}$
	75 - 84		$1.43 \times 10^{-5}$	$2.58 \times 10^{-5}$	$1.05 \times 10^{-5}$	$4.84 \times 10^{-6}$	$4.31 \times 10^{-6}$	$1.53 \times 10^{-6}$
	85 +		$2.28 \times 10^{-4}$	$3.46 \times 10^{-4}$	$2.63 \times 10^{-4}$	$1.08 \times 10^{-4}$	$1.39 { imes} 10^{-4}$	$3.61 \times 10^{-5}$
IADL only	65 - 74	$2.12{\times}10^{-4}$		$7.88 { imes} 10^{-4}$	$3.64 \times 10^{-4}$	$2.38{\times}10^{-4}$	$6.26 \times 10^{-5}$	$2.19 \times 10^{-5}$
-	75 - 84	$1.29{ imes}10^{-4}$		$1.67{ imes}10^{-3}$	$6.98 { imes} 10^{-4}$	$2.43 \times 10^{-4}$	$1.75{ imes}10^{-4}$	$4.61 \times 10^{-5}$
	85 +	$2.29 \times 10^{-4}$		$5.65 \times 10^{-3}$	$3.34 \times 10^{-3}$	$1.61 \times 10^{-3}$	$1.09 \times 10^{-3}$	$4.06 \times 10^{-4}$
1-2 ADLs	65 - 74	$6.94 \times 10^{-5}$	$2.00 \times 10^{-4}$		$3.70 \times 10^{-4}$	$1.98 \times 10^{-4}$	$4.63 \times 10^{-5}$	$1.44 \times 10^{-5}$
	75 - 84	$7.58 \times 10^{-5}$	$3.58{ imes}10^{-4}$		$8.39 \times 10^{-4}$	$2.81 \times 10^{-4}$	$2.00 \times 10^{-4}$	$4.06 \times 10^{-5}$
	85 +	$2.20 \times 10^{-4}$	$7.54 \times 10^{-4}$		$3.45 \times 10^{-3}$	$1.11 \times 10^{-3}$	$8.63 { imes} 10^{-4}$	$2.82 \times 10^{-4}$
3-4 ADLs	65 - 74	$1.24 \times 10^{-4}$	$5.49 \times 10^{-4}$	$1.05 \times 10^{-3}$		$1.34 \times 10^{-3}$	$1.66 \times 10^{-4}$	$7.73 \times 10^{-5}$
	75 - 84	$9.36 \times 10^{-5}$	$2.79 \times 10^{-4}$	$1.16 \times 10^{-3}$		$6.85 \times 10^{-4}$	$4.23 \times 10^{-4}$	$1.39 \times 10^{-4}$
	85 +	$4.96 \times 10^{-4}$	$8.75 \times 10^{-4}$	$3.25 \times 10^{-3}$		$2.49 \times 10^{-3}$	$1.70 \times 10^{-3}$	$7.06 \times 10^{-4}$
5-6 ADLs	65 - 74	$1.76 { imes} 10^{-4}$	$6.49 \times 10^{-4}$	$1.59 \times 10^{-3}$	$2.65 \times 10^{-3}$		$4.52{\times}10^{-4}$	$8.04 \times 10^{-5}$
	75 - 84	$1.53 \times 10^{-4}$	$2.29 \times 10^{-4}$	$8.35 \times 10^{-4}$	$6.70 \times 10^{-4}$		$3.41 \times 10^{-4}$	$1.99 \times 10^{-4}$
	85 +	$9.86 \times 10^{-5}$	$1.18 \times 10^{-4}$	$6.93 { imes} 10^{-4}$	$9.03 \times 10^{-4}$		$1.16 \times 10^{-3}$	$4.27 \times 10^{-4}$
Inst'd	65 - 74	$2.06 \times 10^{-5}$	$1.19 \times 10^{-5}$	$4.61 \times 10^{-5}$	$3.00 \times 10^{-5}$	$4.40 \times 10^{-5}$		$3.90 \times 10^{-5}$
	75 - 84	$8.82 \times 10^{-6}$	$1.44 \times 10^{-5}$	$2.60 \times 10^{-5}$	$7.53 \times 10^{-6}$	$2.44 \times 10^{-6}$		$2.98 \times 10^{-5}$
	85 +	$1.31 { imes} 10^{-5}$	$7.39{\times}10^{-6}$	$1.13{ imes}10^{-5}$	$7.57 \times 10^{-6}$	$1.44 \times 10^{-5}$		$7.40 \times 10^{-5}$

Table J.143: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for females using 10 year age groupings.

1989					1994 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$6.47 \times 10^{-6}$	$1.03 \times 10^{-5}$	$4.68 \times 10^{-6}$	$1.60 \times 10^{-6}$	$1.32 \times 10^{-6}$	$4.28 \times 10^{-7}$
	75 - 84		$3.49 \times 10^{-5}$	$5.51 \times 10^{-5}$	$2.83 \times 10^{-5}$	$1.02 \times 10^{-5}$	$7.80 \times 10^{-6}$	$2.23 \times 10^{-6}$
	85 +		$4.69 \times 10^{-4}$	$1.55 \times 10^{-3}$	$2.95 \times 10^{-3}$	$4.70 \times 10^{-4}$	$2.39{\times}10^{-4}$	$6.17 \times 10^{-5}$
IADL only	65 - 74	$5.72 \times 10^{-4}$		$2.08 \times 10^{-3}$	$1.07 \times 10^{-3}$	$2.89 \times 10^{-4}$	$1.19{\times}10^{-4}$	$4.33 \times 10^{-5}$
	75 - 84	$5.86  imes 10^{-4}$		$6.20 \times 10^{-3}$	$3.55{ imes}10^{-3}$	$1.05 \times 10^{-3}$	$4.54{ imes}10^{-4}$	$1.43 \times 10^{-4}$
	85 +	$2.15 \times 10^{-3}$		$3.71 \times 10^{-2}$	$7.98 \times 10^{-2}$	$8.18 \times 10^{-3}$	$3.39 \times 10^{-3}$	$1.10 \times 10^{-3}$
1-2 ADLs	65 - 74	$3.06 \times 10^{-4}$	$9.48 \times 10^{-4}$		$1.60 \times 10^{-3}$	$4.27 \times 10^{-4}$	$1.83 \times 10^{-4}$	$5.13 \times 10^{-5}$
	75 - 84	$2.34{\times}10^{-4}$	$1.20 \times 10^{-3}$		$3.11 \times 10^{-3}$	$9.51 { imes} 10^{-4}$	$3.58{\times}10^{-4}$	$8.38 \times 10^{-5}$
	85 +	$2.77{ imes}10^{-3}$	$9.51 \times 10^{-3}$		$1.40 \times 10^{-1}$	$1.40 \times 10^{-2}$	$4.42 \times 10^{-3}$	$1.64 \times 10^{-3}$
3–4 ADLs	65 - 74	$3.79 \times 10^{-4}$	$8.53 \times 10^{-4}$	$1.94 \times 10^{-3}$		$8.13 \times 10^{-4}$	$6.40 \times 10^{-4}$	$1.59 \times 10^{-4}$
	75 - 84	$2.58 \times 10^{-4}$	$1.34 \times 10^{-3}$	$3.85 \times 10^{-3}$		$2.08 \times 10^{-3}$	$8.36 \times 10^{-4}$	$1.60 \times 10^{-4}$
	85 +	$6.39 \times 10^{-3}$	$1.32{\times}10^{-2}$	$8.77 \times 10^{-2}$		$2.77{ imes}10^{-2}$	$8.44 \times 10^{-3}$	$3.49 \times 10^{-3}$
5-6 ADLs	65 - 74	$2.77 \times 10^{-4}$	$4.49 \times 10^{-4}$	$1.09 \times 10^{-3}$	$1.60 \times 10^{-3}$		$4.17{\times}10^{-4}$	$2.71 \times 10^{-4}$
	75 - 84	$2.43 \times 10^{-4}$	$6.17 \times 10^{-4}$	$1.56 \times 10^{-3}$	$2.79 \times 10^{-3}$		$9.47 \times 10^{-4}$	$3.17 \times 10^{-4}$
	85 +	$1.35 \times 10^{-3}$	$1.07 \times 10^{-3}$	$5.39 \times 10^{-3}$	$1.05 \times 10^{-2}$		$1.72 \times 10^{-3}$	$6.47 \times 10^{-4}$
Inst'd	65 - 74	$1.09 \times 10^{-4}$	$6.53 \times 10^{-5}$	$2.04 \times 10^{-4}$	$6.85 \times 10^{-5}$	$2.55 \times 10^{-5}$		$7.95 \times 10^{-5}$
	75 - 84	$3.32 \times 10^{-5}$	$4.74 \times 10^{-5}$	$8.04 \times 10^{-5}$	$6.32 \times 10^{-5}$	$3.89 \times 10^{-5}$		$6.97 \times 10^{-5}$
	85 +	$4.17{ imes}10^{-5}$	$2.26 \times 10^{-5}$	$4.91 { imes} 10^{-5}$	$6.01 \times 10^{-5}$	$1.16{ imes}10^{-5}$		$5.55 \times 10^{-5}$

Table J.144: Variance estimates of the MLEs of the annual transition intensities between disability states calculated from the 1982 and 1984 National Long-Term Care Surveys, for males using 10 year age groupings.

1982					1984 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$6.97 \times 10^{-6}$	$6.42 \times 10^{-6}$	$3.65 \times 10^{-6}$	$2.62 \times 10^{-6}$	$1.26 \times 10^{-6}$	$6.72 \times 10^{-6}$
	75 - 84		$5.25 \times 10^{-5}$	$4.65 \times 10^{-5}$	$1.91 \times 10^{-5}$	$2.01 \times 10^{-5}$	$1.16 \times 10^{-5}$	$3.80 \times 10^{-5}$
	85 +		$1.03 \times 10^{-3}$	$2.30 \times 10^{-3}$	$1.82 \times 10^{-3}$	$8.87 \times 10^{-4}$	$4.99 \times 10^{-4}$	$8.90 \times 10^{-4}$
IADL only	65 - 74	$9.45 \times 10^{-4}$		$1.50 \times 10^{-3}$	$7.25 \times 10^{-4}$	$5.52{\times}10^{-4}$	$1.22{\times}10^{-4}$	$4.59{\times}10^{-4}$
	75 - 84	$7.65 \times 10^{-4}$		$2.69 \times 10^{-3}$	$1.90 \times 10^{-3}$	$2.01 \times 10^{-3}$	$6.45 \times 10^{-4}$	$9.77 \times 10^{-4}$
	85 +	$6.48 \times 10^{-4}$		$1.14 \times 10^{-2}$	$1.21 \times 10^{-2}$	$6.16 \times 10^{-3}$	$2.56 \times 10^{-3}$	$4.32 \times 10^{-3}$
1-2 ADLs	65 - 74	$5.59{\times}10^{-4}$	$1.40 \times 10^{-3}$		$4.40 \times 10^{-3}$	$1.82 \times 10^{-3}$	$3.70 { imes} 10^{-4}$	$8.27{ imes}10^{-4}$
	75 - 84	$3.83{ imes}10^{-4}$	$1.98 \times 10^{-3}$		$3.56 \times 10^{-3}$	$2.36 \times 10^{-3}$	$3.81 \times 10^{-4}$	$1.61 \times 10^{-3}$
	85 +	$1.66 \times 10^{-3}$	$2.34 \times 10^{-3}$		$3.84 \times 10^{-2}$	$2.81 \times 10^{-2}$	$8.36 \times 10^{-3}$	$8.35 \times 10^{-3}$
3–4 ADLs	65 - 74	$1.41 \times 10^{-3}$	$1.86 \times 10^{-3}$	$1.06 \times 10^{-2}$		$9.35 \times 10^{-3}$	$6.67 \times 10^{-4}$	$2.94 \times 10^{-3}$
	75 - 84	$8.21 \times 10^{-4}$	$4.06 \times 10^{-3}$	$8.52 \times 10^{-3}$		$1.90 \times 10^{-2}$	$4.10 \times 10^{-3}$	$7.11 \times 10^{-3}$
	85 +	$7.73 \times 10^{-4}$	$6.76 \times 10^{-3}$	$2.36 \times 10^{-2}$		$1.34 \times 10^{-1}$	$2.48 \times 10^{-2}$	$3.85 \times 10^{-2}$
5–6 ADLs	65 - 74	$4.76 { imes} 10^{-4}$	$9.16 \times 10^{-4}$	$2.83{ imes}10^{-3}$	$4.04 \times 10^{-3}$		$5.99{\times}10^{-4}$	$1.75 { imes} 10^{-3}$
	75 - 84	$3.34 \times 10^{-4}$	$1.61 \times 10^{-3}$	$1.74 \times 10^{-3}$	$5.52 \times 10^{-3}$		$1.40 \times 10^{-3}$	$3.15 \times 10^{-3}$
	85 +	$3.09 \times 10^{-4}$	$7.82 \times 10^{-4}$	$7.61 \times 10^{-3}$	$2.31 \times 10^{-2}$		$5.10 \times 10^{-3}$	$1.53 \times 10^{-2}$
Inst'd	65 - 74	$2.29{\times}10^{-4}$	$1.19{\times}10^{-4}$	$3.72{ imes}10^{-4}$	$2.14{\times}10^{-4}$	$2.39{\times}10^{-4}$		$1.20{\times}10^{-3}$
	75 - 84	$6.74 \times 10^{-5}$	$1.40 \times 10^{-4}$	$8.81 \times 10^{-5}$	$2.30{\times}10^{-4}$	$1.03{\times}10^{-4}$		$1.40 \times 10^{-3}$
	85 +	$2.69{\times}10^{-5}$	$4.95 \times 10^{-6}$	$2.27{\times}10^{-5}$	$1.35{\times}10^{-5}$	$6.13 { imes} 10^{-6}$		$3.01{\times}10^{-3}$

Table J.145: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males using 10 year age groupings.

1984					1989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$5.44 \times 10^{-6}$	$7.17 \times 10^{-6}$	$3.35 \times 10^{-6}$	$8.16 \times 10^{-7}$	$6.48 \times 10^{-7}$	$5.69 \times 10^{-7}$
-	75 - 84		$3.18 \times 10^{-5}$	$2.18 \times 10^{-5}$	$8.39 \times 10^{-5}$	$4.64 \times 10^{-5}$	$1.10 \times 10^{-5}$	$5.33 \times 10^{-6}$
	85 +		$4.24 \times 10^{-3}$	$2.74 \times 10^{-3}$	$7.36 \times 10^{-4}$	$2.85 \times 10^{-4}$	$3.76 \times 10^{-4}$	$1.71 \times 10^{-4}$
IADL only	65 - 74	$3.86 \times 10^{-4}$		$3.41 \times 10^{-3}$	$1.76 \times 10^{-3}$	$3.12{ imes}10^{-4}$	$1.59 \times 10^{-4}$	$7.81 \times 10^{-5}$
Ŭ	75 - 84	$5.67 { imes} 10^{-4}$		$2.57{ imes}10^{-3}$	$2.12 \times 10^{-2}$	$1.27{ imes}10^{-2}$	$1.44 \times 10^{-3}$	$3.16 \times 10^{-4}$
	85 +	$3.85 \times 10^{-3}$		$1.26 \times 10^{-1}$	$3.21 \times 10^{-2}$	$1.23 \times 10^{-2}$	$1.37{ imes}10^{-2}$	$5.12 \times 10^{-3}$
1-2 ADLs	65 - 74	$3.65 \times 10^{-4}$	$2.33 \times 10^{-3}$		$5.87 \times 10^{-3}$	$8.43 \times 10^{-4}$	$2.21 \times 10^{-4}$	$8.39 \times 10^{-5}$
	75 - 84	$3.43 \times 10^{-4}$	$1.78 \times 10^{-3}$		$5.96 \times 10^{-3}$	$3.49 \times 10^{-3}$	$6.59 { imes} 10^{-4}$	$2.16 \times 10^{-4}$
	85 +	$1.62 \times 10^{-3}$	$8.99 \times 10^{-2}$		$1.99{ imes}10^{-2}$	$1.09 \times 10^{-2}$	$9.00 \times 10^{-3}$	$3.28 \times 10^{-3}$
3–4 ADLs	65 - 74	$5.53 \times 10^{-4}$	$3.75 \times 10^{-3}$	$1.77 \times 10^{-2}$		$2.41 \times 10^{-3}$	$7.38 \times 10^{-4}$	$2.47 \times 10^{-4}$
	75 - 84	$8.81 \times 10^{-4}$	$8.44 \times 10^{-3}$	$1.93 \times 10^{-2}$		$1.53 \times 10^{-1}$	$1.50 \times 10^{-2}$	$2.46 \times 10^{-3}$
	85 +	$8.74 \times 10^{-4}$	$1.45 \times 10^{-3}$	$1.12 \times 10^{-3}$		$3.19{\times}10^{-4}$	$1.14 \times 10^{-2}$	$5.32 \times 10^{-3}$
5-6 ADLs	65 - 74	$1.65 \times 10^{-4}$	$7.95 \times 10^{-4}$	$2.63 \times 10^{-3}$	$2.15 \times 10^{-3}$		$3.29 \times 10^{-4}$	$1.15 \times 10^{-4}$
	75 - 84	$5.10 \times 10^{-4}$	$4.51 \times 10^{-3}$	$1.09 \times 10^{-2}$	$1.08 \times 10^{-1}$		$6.77 \times 10^{-3}$	$9.95 \times 10^{-4}$
	85 +	$3.65 \times 10^{-4}$	$4.76 \times 10^{-4}$	$3.86 \times 10^{-4}$	$4.63 \times 10^{-3}$		$5.67 \times 10^{-3}$	$1.94 \times 10^{-3}$
Inst'd	65 - 74	$3.67 \times 10^{-5}$	$1.52 \times 10^{-4}$	$1.27{ imes}10^{-4}$	$1.12 \times 10^{-4}$	$1.72 \times 10^{-5}$		$1.02 \times 10^{-4}$
	75 - 84	$6.24 \times 10^{-5}$	$2.17{ imes}10^{-5}$	$3.92{ imes}10^{-5}$	$2.97{ imes}10^{-4}$	$1.44 \times 10^{-4}$		$4.25 \times 10^{-4}$
	85 +	$2.23{\times}10^{-4}$	$2.51{\times}10^{-4}$	$1.85{ imes}10^{-4}$	$6.37{ imes}10^{-5}$	$3.87{ imes}10^{-5}$		$3.72 \times 10^{-4}$

Table J.146: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males using 10 year age groupings.

1989					1994 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 74		$1.11 \times 10^{-5}$	$8.06 \times 10^{-6}$	$5.12 \times 10^{-6}$	$2.38 \times 10^{-6}$	$1.49 \times 10^{-6}$	$1.07 \times 10^{-6}$
-	75 - 84		$2.72 \times 10^{-5}$	$5.48 \times 10^{-5}$	$2.95 \times 10^{-5}$	$1.30 \times 10^{-5}$	$1.14 \times 10^{-5}$	$5.17 \times 10^{-6}$
	85 +		$1.29 \times 10^{-3}$	$8.83 \times 10^{-3}$	$1.24 \times 10^{-2}$	$1.38 \times 10^{-3}$	$7.32 \times 10^{-4}$	$2.78 \times 10^{-4}$
IADL only	65 - 74	$1.60 \times 10^{-3}$		$2.94 \times 10^{-3}$	$2.11 \times 10^{-3}$	$7.06 \times 10^{-4}$	$3.87{ imes}10^{-4}$	$1.88 \times 10^{-4}$
	75 - 84	$5.91{\times}10^{-4}$		$3.76{ imes}10^{-3}$	$2.20 \times 10^{-3}$	$7.94 { imes} 10^{-4}$	$6.67{ imes}10^{-4}$	$2.34 \times 10^{-4}$
	85 +	$6.19 \times 10^{-2}$		$1.76{ imes}10^1$	$2.55{ imes}10^1$	$1.24{ imes}10^0$	$2.18 \times 10^{-1}$	$2.05 \times 10^{-1}$
1-2 ADLs	65 - 74	$6.60 \times 10^{-4}$	$1.68 \times 10^{-3}$		$2.07 \times 10^{-3}$	$8.11 \times 10^{-4}$	$3.25 \times 10^{-4}$	$9.53 \times 10^{-5}$
	75 - 84	$6.05 { imes} 10^{-4}$	$1.14 \times 10^{-3}$		$4.81 \times 10^{-3}$	$2.34 \times 10^{-3}$	$1.78{ imes}10^{-3}$	$5.62 \times 10^{-4}$
	85 +	$6.20 \times 10^{-3}$	$5.67{ imes}10^{-2}$		$5.48 \times 10^{-1}$	$7.44 \times 10^{-2}$	$3.35{ imes}10^{-2}$	$1.04 \times 10^{-2}$
3–4 ADLs	65 - 74	$1.20 \times 10^{-3}$	$2.96 \times 10^{-3}$	$4.05 \times 10^{-3}$		$4.92 \times 10^{-3}$	$1.24 \times 10^{-3}$	$5.09 \times 10^{-4}$
	75 - 84	$1.00 \times 10^{-3}$	$1.00 \times 10^{-3}$	$2.21 \times 10^{-3}$		$5.34 \times 10^{-3}$	$3.42 \times 10^{-3}$	$1.37 \times 10^{-3}$
	85 +	$1.12 \times 10^{-1}$	$2.89 \times 10^{0}$	$2.48{ imes}10^1$		$2.05{ imes}10^0$	$4.47 \times 10^{-1}$	$3.08 \times 10^{-1}$
5–6 ADLs	65 - 74	$7.28 \times 10^{-4}$	$1.72 \times 10^{-3}$	$2.76 \times 10^{-3}$	$1.10 \times 10^{-3}$		$1.75 \times 10^{-3}$	$4.37 \times 10^{-4}$
	75 - 84	$8.86 \times 10^{-4}$	$2.98 \times 10^{-4}$	$1.17 \times 10^{-3}$	$2.36 \times 10^{-3}$		$1.54 \times 10^{-3}$	$6.14 \times 10^{-4}$
	85 +	$4.47 \times 10^{-3}$	$1.93 \times 10^{-2}$	$4.84 \times 10^{-2}$	$9.34 \times 10^{-2}$		$5.71 \times 10^{-2}$	$1.96 \times 10^{-2}$
Inst'd	65 - 74	$2.81{\times}10^{-4}$	$4.81 \times 10^{-4}$	$1.70 { imes} 10^{-4}$	$1.42 \times 10^{-4}$	$6.35 \times 10^{-5}$		$1.65 \times 10^{-4}$
	75 - 84	$1.08 { imes} 10^{-4}$	$6.80 \times 10^{-5}$	$3.62{\times}10^{-4}$	$3.09 \times 10^{-4}$	$9.23 \times 10^{-5}$		$1.77 \times 10^{-4}$
	85 +	$5.39{\times}10^{-4}$	$1.22 \times 10^{-3}$	$2.64 \times 10^{-3}$	$4.54 \times 10^{-3}$	$6.97{ imes}10^{-4}$		$1.12 \times 10^{-3}$

Table J.147: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1984					1989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		$4.63 \times 10^{-6}$	$4.90 \times 10^{-6}$	$2.10 \times 10^{-6}$	$1.19 \times 10^{-6}$	$4.17 \times 10^{-7}$	$3.22 \times 10^{-7}$
-	70 - 74		$4.92 \times 10^{-6}$	$3.95 \times 10^{-6}$	$1.92 \times 10^{-6}$	$1.16 \times 10^{-6}$	$8.40 \times 10^{-7}$	$4.37 \times 10^{-7}$
	75 - 79		$9.14 \times 10^{-6}$	$1.21 \times 10^{-5}$	$6.08 \times 10^{-6}$	$3.31 \times 10^{-6}$	$2.18 \times 10^{-6}$	$1.08 \times 10^{-6}$
	80 - 84		$4.50 \times 10^{-5}$	$7.82{ imes}10^{-5}$	$4.71 { imes} 10^{-5}$	$1.93{ imes}10^{-5}$	$1.66{ imes}10^{-5}$	$6.54 \times 10^{-6}$
IADL only	65 - 69	$4.16 \times 10^{-4}$		$2.98{ imes}10^{-3}$	$1.38 \times 10^{-3}$	$4.76 { imes} 10^{-4}$	$9.65 \times 10^{-5}$	$3.75 \times 10^{-5}$
	70 - 74	$1.84 \times 10^{-4}$		$6.75 \times 10^{-4}$	$3.09 \times 10^{-4}$	$1.78 { imes} 10^{-4}$	$8.65 \times 10^{-5}$	$3.84 \times 10^{-5}$
	75 - 79	$1.95 \times 10^{-4}$		$1.60 \times 10^{-3}$	$8.36 \times 10^{-4}$	$4.56 \times 10^{-4}$	$1.66 \times 10^{-4}$	$5.29 \times 10^{-5}$
	80 - 84	$2.19 \times 10^{-4}$		$2.61 \times 10^{-3}$	$1.94 \times 10^{-3}$	$6.46 \times 10^{-4}$	$4.13 \times 10^{-4}$	$1.20 \times 10^{-4}$
1-2 ADLs	65 - 69	$2.42 \times 10^{-4}$	$1.31 \times 10^{-3}$		$1.93 \times 10^{-3}$	$7.03 \times 10^{-4}$	$1.10 \times 10^{-4}$	$3.26 \times 10^{-5}$
	70 - 74	$6.46 \times 10^{-5}$	$1.92 \times 10^{-4}$		$3.66 \times 10^{-4}$	$1.65 { imes} 10^{-4}$	$5.21 \times 10^{-5}$	$2.14 \times 10^{-5}$
	75 - 79	$1.11 \times 10^{-4}$	$5.32 \times 10^{-4}$		$1.31 \times 10^{-3}$	$6.32 \times 10^{-4}$	$2.15 \times 10^{-4}$	$6.60 \times 10^{-5}$
	80 - 84	$1.88 \times 10^{-4}$	$7.86 \times 10^{-4}$		$1.74 \times 10^{-3}$	$4.70 \times 10^{-4}$	$4.52 \times 10^{-4}$	$8.50 \times 10^{-5}$
3-4 ADLs	65 - 69	$3.88 \times 10^{-4}$	$2.65 \times 10^{-3}$	$7.36 \times 10^{-3}$		$3.65 \times 10^{-3}$	$3.20 \times 10^{-4}$	$1.44 \times 10^{-4}$
	70 - 74	$1.27 \times 10^{-4}$	$3.08 \times 10^{-4}$	$9.22 \times 10^{-4}$		$6.74 \times 10^{-4}$	$2.13 \times 10^{-4}$	$9.41 \times 10^{-5}$
	75 - 79	$1.21 \times 10^{-4}$	$4.77 \times 10^{-4}$	$2.05 \times 10^{-3}$		$1.44 \times 10^{-3}$	$5.78 \times 10^{-4}$	$2.02 \times 10^{-4}$
	80 - 84	$2.35 \times 10^{-4}$	$6.09 \times 10^{-4}$	$2.58 \times 10^{-3}$		$1.98 \times 10^{-3}$	$1.10 \times 10^{-3}$	$3.39 \times 10^{-4}$
5-6 ADLs	65 - 69	$2.95 \times 10^{-4}$	$1.22 \times 10^{-3}$	$3.90 \times 10^{-3}$	$4.29 \times 10^{-3}$		$3.83 \times 10^{-4}$	$1.33 \times 10^{-4}$
	70 - 74	$9.54 \times 10^{-5}$	$5.20 \times 10^{-4}$	$9.12 \times 10^{-4}$	$1.27 \times 10^{-3}$		$3.97{ imes}10^{-4}$	$6.29 \times 10^{-5}$
	75 - 79	$8.84 \times 10^{-5}$	$6.56 \times 10^{-4}$	$1.82 \times 10^{-3}$	$1.20 \times 10^{-3}$		$4.86 \times 10^{-4}$	$2.32 \times 10^{-4}$
	80 - 84	$3.77 \times 10^{-4}$	$2.84 \times 10^{-4}$	$1.16 \times 10^{-3}$	$2.88 \times 10^{-3}$		$7.58 \times 10^{-4}$	$3.07 \times 10^{-4}$
Inst'd	65 - 69	$3.67 \times 10^{-5}$	$7.65 \times 10^{-5}$	$8.82 \times 10^{-5}$	$8.42 \times 10^{-5}$	$5.56 \times 10^{-5}$		$6.35 \times 10^{-5}$
	70 - 74	$1.93 \times 10^{-5}$	$3.53 \times 10^{-5}$	$4.42 \times 10^{-5}$	$2.97 \times 10^{-5}$	$2.25 \times 10^{-5}$		$5.24 \times 10^{-5}$
	75 - 79	$2.45 \times 10^{-5}$	$1.71 \times 10^{-5}$	$9.71 \times 10^{-5}$	$3.30 \times 10^{-5}$	$1.36 \times 10^{-5}$		$1.23 \times 10^{-4}$
	80 - 84	$1.93 \times 10^{-5}$	$2.96 \times 10^{-5}$	$2.02 \times 10^{-5}$	$1.94 \times 10^{-5}$	$5.65 \times 10^{-6}$		$6.41 \times 10^{-5}$

Table J.148: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males and females using 5 year age groupings.

1989					1994 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		$5.82 \times 10^{-6}$	$6.20 \times 10^{-6}$	$5.38 \times 10^{-6}$	$1.80 \times 10^{-6}$	$1.09 \times 10^{-6}$	$5.31 \times 10^{-7}$
Ū.	70 - 74		$1.07 \times 10^{-5}$	$1.44 \times 10^{-5}$	$6.12 \times 10^{-6}$	$1.96 \times 10^{-6}$	$1.61 \times 10^{-6}$	$7.77 \times 10^{-7}$
	75 - 79		$2.06 \times 10^{-5}$	$3.36 \times 10^{-5}$	$2.33 \times 10^{-5}$	$9.12 \times 10^{-6}$	$5.26 \times 10^{-6}$	$1.82 \times 10^{-6}$
	80 - 84		$7.96{ imes}10^{-5}$	$1.52{\times}10^{-4}$	$5.27{ imes}10^{-5}$	$1.84 { imes} 10^{-5}$	$1.89{ imes}10^{-5}$	$6.45 \times 10^{-6}$
IADL only	65 - 69	$1.07 \times 10^{-3}$		$1.02 \times 10^{-3}$	$1.78 \times 10^{-3}$	$4.96 { imes} 10^{-4}$	$1.66{ imes}10^{-4}$	$5.65 \times 10^{-5}$
	70 - 74	$7.40 \times 10^{-4}$		$4.85 \times 10^{-3}$	$2.11 \times 10^{-3}$	$4.03 \times 10^{-4}$	$2.49{ imes}10^{-4}$	$1.02 \times 10^{-4}$
	75 - 79	$6.02 \times 10^{-4}$		$2.99 \times 10^{-3}$	$2.33 \times 10^{-3}$	$8.45 \times 10^{-4}$	$2.78 { imes} 10^{-4}$	$1.23 \times 10^{-4}$
	80 - 84	$6.44 \times 10^{-4}$		$1.78 \times 10^{-2}$	$5.49 \times 10^{-3}$	$1.53 \times 10^{-3}$	$1.17 \times 10^{-3}$	$2.84 \times 10^{-4}$
1-2 ADLs	65 - 69	$6.18 \times 10^{-4}$	$8.98 \times 10^{-4}$		$2.47 \times 10^{-3}$	$5.62 \times 10^{-4}$	$2.27 \times 10^{-4}$	$7.59 \times 10^{-5}$
	70 - 74	$3.61 \times 10^{-4}$	$1.80 \times 10^{-3}$		$2.60 \times 10^{-3}$	$7.41 \times 10^{-4}$	$2.30 \times 10^{-4}$	$7.43 \times 10^{-5}$
	75 - 79	$3.48 \times 10^{-4}$	$8.66 \times 10^{-4}$		$4.00 \times 10^{-3}$	$1.34 \times 10^{-3}$	$4.15 \times 10^{-4}$	$9.98 \times 10^{-5}$
	80 - 84	$3.15 \times 10^{-4}$	$3.02 \times 10^{-3}$		$4.51 \times 10^{-3}$	$1.53 \times 10^{-3}$	$1.01 \times 10^{-3}$	$2.51 \times 10^{-4}$
3-4 ADLs	65 - 69	$1.08 \times 10^{-3}$	$2.53 \times 10^{-3}$	$2.53 \times 10^{-3}$		$2.10 \times 10^{-3}$	$1.09 \times 10^{-3}$	$2.29 \times 10^{-4}$
	70 - 74	$4.39 \times 10^{-4}$	$1.22 \times 10^{-3}$	$3.20 \times 10^{-3}$		$2.09 \times 10^{-3}$	$6.92 \times 10^{-4}$	$2.79 \times 10^{-4}$
	75 - 79	$4.79 \times 10^{-4}$	$1.43 \times 10^{-3}$	$3.58 \times 10^{-3}$		$3.12 \times 10^{-3}$	$1.11 \times 10^{-3}$	$2.49 \times 10^{-4}$
	80 - 84	$3.49 \times 10^{-4}$	$1.55 \times 10^{-3}$	$5.88 \times 10^{-3}$		$2.58 \times 10^{-3}$	$1.54 \times 10^{-3}$	$3.68 \times 10^{-4}$
5-6 ADLs	65 - 69	$4.04 \times 10^{-4}$	$5.32 \times 10^{-4}$	$1.36 \times 10^{-3}$	$1.81 \times 10^{-3}$		$1.21 \times 10^{-3}$	$3.87 \times 10^{-4}$
	70 - 74	$3.99 \times 10^{-4}$	$1.07 \times 10^{-3}$	$2.16 \times 10^{-3}$	$1.80 \times 10^{-3}$		$4.54 \times 10^{-4}$	$3.14 \times 10^{-4}$
	75 - 79	$6.58 \times 10^{-4}$	$5.61 \times 10^{-4}$	$1.71 \times 10^{-3}$	$3.34 \times 10^{-3}$		$1.30 \times 10^{-3}$	$4.46 \times 10^{-4}$
	80 - 84	$2.29 \times 10^{-4}$	$7.00 \times 10^{-4}$	$1.86 \times 10^{-3}$	$2.50 \times 10^{-3}$	_	$1.09 \times 10^{-3}$	$3.86 \times 10^{-4}$
Inst'd	65 - 69	$2.35 \times 10^{-4}$	$3.32 \times 10^{-4}$	$3.09 \times 10^{-4}$	$1.78 \times 10^{-4}$	$3.83 \times 10^{-5}$		$1.25 \times 10^{-4}$
	70 - 74	$1.21 \times 10^{-4}$	$8.26 \times 10^{-5}$	$1.80 \times 10^{-4}$	$7.42 \times 10^{-5}$	$3.20 \times 10^{-5}$		$8.74 \times 10^{-5}$
	75 - 79	$5.80 \times 10^{-5}$	$2.52 \times 10^{-5}$	$1.50 \times 10^{-4}$	$9.82 \times 10^{-5}$	$2.37 \times 10^{-5}$		$1.37 \times 10^{-4}$
	80 - 84	$4.51 \times 10^{-5}$	$9.52 \times 10^{-5}$	$1.47 \times 10^{-4}$	$1.21 \times 10^{-4}$	$7.28 \times 10^{-5}$		$7.21 \times 10^{-5}$

Table J.149: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for females using 5 year age groupings.

1984					1989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65–69		$9.51 \times 10^{-6}$	$7.31 \times 10^{-6}$	$3.24 \times 10^{-6}$	$3.06 \times 10^{-6}$	$1.48 \times 10^{-6}$	$6.16 \times 10^{-7}$
Ŭ	70 - 74		$1.54 \times 10^{-5}$	$2.42 \times 10^{-5}$	$9.31 \times 10^{-6}$	$5.99 \times 10^{-6}$	$3.93 \times 10^{-6}$	$1.46 \times 10^{-6}$
	75 - 79		$6.37 \times 10^{-5}$	$1.53 \times 10^{-4}$	$6.51 \times 10^{-5}$	$2.37 \times 10^{-5}$	$2.58 \times 10^{-5}$	$8.15 \times 10^{-6}$
	80 - 84		$2.28{ imes}10^{-4}$	$3.46{ imes}10^{-4}$	$2.63{ imes}10^{-4}$	$1.08 { imes} 10^{-4}$	$1.39{ imes}10^{-4}$	$3.61 \times 10^{-5}$
IADL only	65 - 69	$3.31 { imes} 10^{-4}$		$9.76{ imes}10^{-4}$	$3.68{ imes}10^{-4}$	$3.28{ imes}10^{-4}$	$1.11{\times}10^{-4}$	$4.89 \times 10^{-5}$
	70 - 74	$2.58 \times 10^{-4}$		$2.44 \times 10^{-3}$	$9.44 \times 10^{-4}$	$5.64 \times 10^{-4}$	$2.64 \times 10^{-4}$	$6.57 \times 10^{-5}$
	75 - 79	$2.86 \times 10^{-4}$		$4.76 \times 10^{-3}$	$2.08 \times 10^{-3}$	$5.09 \times 10^{-4}$	$5.19 \times 10^{-4}$	$1.34 \times 10^{-4}$
	80 - 84	$2.29 \times 10^{-4}$		$5.65 \times 10^{-3}$	$3.34 \times 10^{-3}$	$1.61 \times 10^{-3}$	$1.09 \times 10^{-3}$	$4.06 \times 10^{-4}$
1-2 ADLs	65 - 69	$6.42 \times 10^{-5}$	$2.30 \times 10^{-4}$		$3.68 \times 10^{-4}$	$2.32 \times 10^{-4}$	$6.41 \times 10^{-5}$	$2.37 \times 10^{-5}$
	70 - 74	$1.37{ imes}10^{-4}$	$5.89 \times 10^{-4}$		$1.33 \times 10^{-3}$	$8.13 \times 10^{-4}$	$3.03 \times 10^{-4}$	$7.69 \times 10^{-5}$
	75 - 79	$2.40 \times 10^{-4}$	$9.94 \times 10^{-4}$		$2.28 \times 10^{-3}$	$4.88 \times 10^{-4}$	$5.82 \times 10^{-4}$	$1.06 \times 10^{-4}$
	80 - 84	$2.20 \times 10^{-4}$	$7.54 \times 10^{-4}$		$3.45 \times 10^{-3}$	$1.11 \times 10^{-3}$	$8.63 \times 10^{-4}$	$2.82 \times 10^{-4}$
3-4 ADLs	65 - 69	$1.79 \times 10^{-4}$	$2.70 \times 10^{-4}$	$7.72 \times 10^{-4}$		$1.13 \times 10^{-3}$	$2.39 \times 10^{-4}$	$1.25 \times 10^{-4}$
	70 - 74	$1.64 \times 10^{-4}$	$4.46 \times 10^{-4}$	$1.57 \times 10^{-3}$		$1.16 \times 10^{-3}$	$6.29 \times 10^{-4}$	$1.81 \times 10^{-4}$
	75 - 79	$2.50 \times 10^{-4}$	$5.51 \times 10^{-4}$	$2.92 \times 10^{-3}$		$1.41 \times 10^{-3}$	$9.61 \times 10^{-4}$	$3.23 \times 10^{-4}$
	80 - 84	$4.96 \times 10^{-4}$	$8.75 \times 10^{-4}$	$3.25 \times 10^{-3}$		$2.49 \times 10^{-3}$	$1.70 \times 10^{-3}$	$7.06 \times 10^{-4}$
5-6 ADLs	65 - 69	$2.37 \times 10^{-4}$	$1.12 \times 10^{-3}$	$1.52 \times 10^{-3}$	$2.65 \times 10^{-3}$		$1.02 \times 10^{-3}$	$1.24 \times 10^{-4}$
	70 - 74	$1.62 \times 10^{-4}$	$7.90 \times 10^{-4}$	$2.63 \times 10^{-3}$	$1.32 \times 10^{-3}$		$7.20 \times 10^{-4}$	$4.04 \times 10^{-4}$
	75 - 79	$4.71 \times 10^{-4}$	$1.76 \times 10^{-4}$	$6.48 \times 10^{-4}$	$1.26 \times 10^{-3}$		$5.89 \times 10^{-4}$	$3.64 \times 10^{-4}$
	80 - 84	$9.86 \times 10^{-5}$	$1.18 \times 10^{-4}$	$6.93 \times 10^{-4}$	$9.03 \times 10^{-4}$		$1.16 \times 10^{-3}$	$4.27 \times 10^{-4}$
Inst'd	65 - 69	$2.13 \times 10^{-5}$	$1.39 \times 10^{-5}$	$5.51 \times 10^{-5}$	$1.98 \times 10^{-5}$	$3.84 \times 10^{-5}$		$5.44 \times 10^{-5}$
	70 - 74	$3.93 \times 10^{-5}$	$2.70 \times 10^{-5}$	$1.89 \times 10^{-4}$	$5.44 \times 10^{-5}$	$2.86 \times 10^{-5}$		$1.38 \times 10^{-4}$
	75 - 79	$1.88 \times 10^{-5}$	$3.63 \times 10^{-5}$	$2.89 \times 10^{-5}$	$9.59 \times 10^{-6}$	$3.19 \times 10^{-6}$		$5.60 \times 10^{-5}$
	80 - 84	$1.31 \times 10^{-5}$	$7.39 \times 10^{-6}$	$1.13 \times 10^{-5}$	$7.57 \times 10^{-6}$	$1.44 \times 10^{-5}$		$7.40 \times 10^{-5}$

Table J.150: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for females using 5 year age groupings.

1989					1994 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		$1.05 \times 10^{-5}$	$1.19 \times 10^{-5}$	$6.02 \times 10^{-6}$	$2.70 \times 10^{-6}$	$2.21 \times 10^{-6}$	$9.02 \times 10^{-7}$
Ū	70 - 74		$1.43 \times 10^{-5}$	$2.80 \times 10^{-5}$	$1.16 \times 10^{-5}$	$3.18 \times 10^{-6}$	$3.00 \times 10^{-6}$	$8.14 \times 10^{-7}$
	75 - 79		$2.89 \times 10^{-5}$	$5.04 \times 10^{-5}$	$3.10 \times 10^{-5}$	$1.07 \times 10^{-5}$	$9.02 \times 10^{-6}$	$2.51 \times 10^{-6}$
	80 - 84		$4.07{ imes}10^{-4}$	$5.80 \times 10^{-4}$	$1.38{ imes}10^{-4}$	$5.99 \times 10^{-5}$	$4.10 \times 10^{-5}$	$1.14 \times 10^{-5}$
IADL only	65 - 69	$1.44 \times 10^{-3}$		$1.16{ imes}10^{-3}$	$1.22 \times 10^{-3}$	$5.97{ imes}10^{-4}$	$2.49{ imes}10^{-4}$	$4.92 \times 10^{-5}$
	70 - 74	$7.57{ imes}10^{-4}$		$7.33 \times 10^{-3}$	$3.00 \times 10^{-3}$	$5.00 \times 10^{-4}$	$3.01 \times 10^{-4}$	$1.56 \times 10^{-4}$
	75 - 79	$8.74 \times 10^{-4}$		$4.53 \times 10^{-3}$	$3.08 \times 10^{-3}$	$1.10 \times 10^{-3}$	$3.75 \times 10^{-4}$	$1.58 \times 10^{-4}$
	80 - 84	$2.11 \times 10^{-3}$		$5.63 \times 10^{-2}$	$1.15 \times 10^{-2}$	$5.02 \times 10^{-3}$	$2.53 \times 10^{-3}$	$6.33 \times 10^{-4}$
1-2 ADLs	65 - 69	$8.52 \times 10^{-4}$	$1.44 \times 10^{-3}$		$1.99 \times 10^{-3}$	$5.91 \times 10^{-4}$	$3.78 \times 10^{-4}$	$7.65 \times 10^{-5}$
	70 - 74	$4.56 \times 10^{-4}$	$2.04 \times 10^{-3}$		$3.97{ imes}10^{-3}$	$8.84 \times 10^{-4}$	$3.32{ imes}10^{-4}$	$1.04 \times 10^{-4}$
	75 - 79	$4.35 \times 10^{-4}$	$1.16 \times 10^{-3}$		$4.13 \times 10^{-3}$	$1.11 \times 10^{-3}$	$3.61 \times 10^{-4}$	$1.21 \times 10^{-4}$
	80 - 84	$4.82 \times 10^{-4}$	$7.34 \times 10^{-3}$		$4.48 \times 10^{-3}$	$2.30 \times 10^{-3}$	$1.17 \times 10^{-3}$	$2.35 \times 10^{-4}$
3-4 ADLs	65 - 69	$1.06 \times 10^{-3}$	$3.20 \times 10^{-3}$	$1.34 \times 10^{-3}$		$1.10 \times 10^{-3}$	$1.74 \times 10^{-3}$	$2.89 \times 10^{-4}$
	70 - 74	$6.17 \times 10^{-4}$	$1.20 \times 10^{-3}$	$4.97 \times 10^{-3}$		$1.85 \times 10^{-3}$	$9.15 \times 10^{-4}$	$2.79 \times 10^{-4}$
	75 - 79	$4.71 \times 10^{-4}$	$1.64 \times 10^{-3}$	$4.22 \times 10^{-3}$		$2.56 \times 10^{-3}$	$9.84 \times 10^{-4}$	$2.68 \times 10^{-4}$
	80 - 84	$4.52 \times 10^{-4}$	$3.93 \times 10^{-3}$	$8.67 \times 10^{-3}$		$3.64 \times 10^{-3}$	$2.02 \times 10^{-3}$	$2.99 \times 10^{-4}$
5-6 ADLs	65 - 69	$4.51 \times 10^{-4}$	$6.55 \times 10^{-4}$	$4.29 \times 10^{-4}$	$2.17 \times 10^{-3}$		$1.47 \times 10^{-3}$	$3.98 \times 10^{-4}$
	70 - 74	$5.58 \times 10^{-4}$	$1.05 \times 10^{-3}$	$3.61 \times 10^{-3}$	$3.43 \times 10^{-3}$		$3.50 \times 10^{-4}$	$5.15 \times 10^{-4}$
	75 - 79	$5.87 \times 10^{-4}$	$6.81 \times 10^{-4}$	$2.74 \times 10^{-3}$	$3.69 \times 10^{-3}$		$1.48 \times 10^{-3}$	$4.25 \times 10^{-4}$
	80 - 84	$3.98 \times 10^{-4}$	$2.37 \times 10^{-3}$	$3.84 \times 10^{-3}$	$4.18 \times 10^{-3}$	_	$2.01 \times 10^{-3}$	$7.07 \times 10^{-4}$
Inst'd	65 - 69	$5.75 \times 10^{-4}$	$2.37 \times 10^{-4}$	$9.64 \times 10^{-4}$	$2.74 \times 10^{-4}$	$9.24 \times 10^{-5}$		$2.04 \times 10^{-4}$
	70 - 74	$1.28 \times 10^{-4}$	$8.12 \times 10^{-5}$	$2.53 \times 10^{-4}$	$8.83 \times 10^{-5}$	$3.37 \times 10^{-5}$		$1.20 \times 10^{-4}$
	75 - 79	$8.24 \times 10^{-5}$	$3.17 \times 10^{-5}$	$1.36 \times 10^{-4}$	$1.10 \times 10^{-4}$	$2.24 \times 10^{-5}$		$1.94 \times 10^{-4}$
	80 - 84	$5.53 \times 10^{-5}$	$2.32 \times 10^{-4}$	$2.29 \times 10^{-4}$	$1.07 \times 10^{-4}$	$1.35 \times 10^{-4}$		$9.43 \times 10^{-5}$

Table J.151: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1984 and 1989 National Long-Term Care Surveys, for males using 5 year age groupings.

1984					1989 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65 - 69		$1.27 \times 10^{-5}$	$1.62 \times 10^{-5}$	$6.82 \times 10^{-6}$	$2.03 \times 10^{-6}$	$7.00 \times 10^{-7}$	$9.90 \times 10^{-7}$
Ū.	70 - 74		$9.62 \times 10^{-6}$	$9.08 \times 10^{-6}$	$5.90 \times 10^{-6}$	$1.52 \times 10^{-6}$	$2.30 \times 10^{-6}$	$1.34 \times 10^{-6}$
	75 - 79		$2.43 \times 10^{-5}$	$1.92 \times 10^{-5}$	$3.07 \times 10^{-5}$	$1.43 \times 10^{-5}$	$5.71 \times 10^{-6}$	$4.02 \times 10^{-6}$
	80 - 84		$1.71 { imes} 10^{-4}$	$1.09{ imes}10^{-4}$	$1.29 \times 10^{-3}$	$7.45 \times 10^{-4}$	$1.67{ imes}10^{-4}$	$3.23 \times 10^{-5}$
IADL only	65 - 69	$1.45 \times 10^{-3}$		$1.12{ imes}10^{-2}$	$5.15 \times 10^{-3}$	$9.83 { imes} 10^{-4}$	$2.34{\times}10^{-4}$	$1.65{ imes}10^{-4}$
	70 - 74	$3.90 \times 10^{-4}$		$2.23{ imes}10^{-3}$	$1.67{ imes}10^{-3}$	$4.35 { imes} 10^{-4}$	$4.12 \times 10^{-4}$	$1.52{ imes}10^{-4}$
	75 - 79	$7.93 \times 10^{-4}$		$3.75 \times 10^{-3}$	$7.32 \times 10^{-3}$	$3.82 \times 10^{-3}$	$5.18 \times 10^{-4}$	$2.85 \times 10^{-4}$
	80 - 84	$1.23 \times 10^{-3}$		$6.74 \times 10^{-3}$	$2.58 \times 10^{-1}$	$1.41 \times 10^{-1}$	$2.54 \times 10^{-2}$	$1.39 \times 10^{-3}$
1-2 ADLs	65 - 69	$1.42 \times 10^{-3}$	$1.20 \times 10^{-2}$		$1.44 \times 10^{-2}$	$2.88 \times 10^{-3}$	$3.53 \times 10^{-4}$	$1.76 \times 10^{-4}$
	70 - 74	$4.59 \times 10^{-4}$	$1.01 \times 10^{-3}$		$3.17 \times 10^{-3}$	$7.22 \times 10^{-4}$	$3.24 \times 10^{-4}$	$1.20 \times 10^{-4}$
	75 - 79	$4.77 \times 10^{-4}$	$2.85 \times 10^{-3}$		$9.67 \times 10^{-3}$	$4.44 \times 10^{-3}$	$6.42 \times 10^{-4}$	$3.66 \times 10^{-4}$
	80 - 84	$9.56  imes 10^{-4}$	$3.51 \times 10^{-3}$		$9.15 \times 10^{-3}$	$6.67 \times 10^{-3}$	$2.46 \times 10^{-3}$	$5.74 \times 10^{-4}$
3-4 ADLs	65 - 69	$2.41 \times 10^{-3}$	$1.74 \times 10^{-2}$	$4.25 \times 10^{-2}$		$8.77 \times 10^{-3}$	$9.95 \times 10^{-4}$	$5.57 \times 10^{-4}$
	70 - 74	$6.29 \times 10^{-4}$	$2.21 \times 10^{-3}$	$1.05 \times 10^{-2}$		$2.18 \times 10^{-3}$	$1.52 \times 10^{-3}$	$3.78 \times 10^{-4}$
	75 - 79	$8.52 \times 10^{-4}$	$7.98 \times 10^{-3}$	$2.37 \times 10^{-2}$		$3.11 \times 10^{-2}$	$4.70 \times 10^{-3}$	$2.26 \times 10^{-3}$
	80 - 84	$2.41 \times 10^{-3}$	$1.15 \times 10^{-2}$	$1.96 \times 10^{-2}$		$4.00 \times 10^{-1}$	$8.61 \times 10^{-2}$	$6.03 \times 10^{-3}$
5-6 ADLs	65 - 69	$6.56 \times 10^{-4}$	$3.18 \times 10^{-3}$	$6.71 \times 10^{-3}$	$5.78 \times 10^{-3}$		$8.62 \times 10^{-4}$	$3.46 \times 10^{-4}$
	70 - 74	$2.09 \times 10^{-4}$	$8.08 \times 10^{-4}$	$2.90 \times 10^{-3}$	$2.48 \times 10^{-3}$		$5.66 \times 10^{-4}$	$1.65 \times 10^{-4}$
	75 - 79	$3.32 \times 10^{-4}$	$2.35 \times 10^{-3}$	$6.72 \times 10^{-3}$	$1.01 \times 10^{-2}$		$1.82 \times 10^{-3}$	$6.69 \times 10^{-4}$
	80 - 84	$4.71 \times 10^{-3}$	$2.74 \times 10^{-2}$	$4.87 \times 10^{-2}$	$2.03 \times 10^{0}$	_	$2.09 \times 10^{-1}$	$9.88 \times 10^{-3}$
Inst'd	65 - 69	$5.78 \times 10^{-5}$	$2.86 \times 10^{-4}$	$1.47 \times 10^{-4}$	$5.37 \times 10^{-5}$	$1.92 \times 10^{-5}$		$1.31 \times 10^{-4}$
	70 - 74	$1.08 \times 10^{-4}$	$3.82 \times 10^{-4}$	$2.81 \times 10^{-4}$	$4.46 \times 10^{-4}$	$5.58 \times 10^{-5}$		$3.38 \times 10^{-4}$
	75 - 79	$7.15 \times 10^{-5}$	$3.12 \times 10^{-5}$	$7.96 \times 10^{-5}$	$9.63 \times 10^{-5}$	$3.38 \times 10^{-5}$		$6.33 \times 10^{-4}$
	80 - 84	$2.77 \times 10^{-4}$	$1.04 \times 10^{-4}$	$1.42 \times 10^{-4}$	$2.11 \times 10^{-3}$	$9.31 \times 10^{-4}$		$1.37 \times 10^{-3}$

Table J.152: Variance estimates of the constrained (positive) MLEs of the annual transition intensities between disability states calculated from the 1989 and 1994 National Long-Term Care Surveys, for males using 5 year age groupings.

1989					1994 Status			
status	Age		IADL	1 - 2	3 - 4	5 - 6		
	group	Healthy	only	ADLs	ADLs	ADLs	Inst'd	Dead
Healthy	65–69		$1.11 \times 10^{-5}$	$9.41 \times 10^{-6}$	$7.12 \times 10^{-6}$	$3.72 \times 10^{-6}$	$2.07 \times 10^{-6}$	$1.36 \times 10^{-6}$
Ū	70 - 74		$3.86 \times 10^{-5}$	$2.49 \times 10^{-5}$	$1.12 \times 10^{-5}$	$5.76 \times 10^{-6}$	$3.97 \times 10^{-6}$	$2.96 \times 10^{-6}$
	75 - 79		$5.47 \times 10^{-5}$	$5.85 \times 10^{-5}$	$3.82 \times 10^{-5}$	$3.33 \times 10^{-5}$	$1.44 \times 10^{-5}$	$6.35 \times 10^{-6}$
	80 - 84		$6.95 \times 10^{-5}$	$3.50 \times 10^{-4}$	$1.82{ imes}10^{-4}$	$3.01 \times 10^{-5}$	$5.68 \times 10^{-5}$	$3.85 \times 10^{-5}$
IADL only	65 - 69	$3.17{ imes}10^{-3}$		$3.05 \times 10^{-3}$	$3.97{ imes}10^{-3}$	$1.15 \times 10^{-3}$	$4.35{\times}10^{-4}$	$5.97{ imes}10^{-4}$
	70 - 74	$3.49 \times 10^{-3}$		$8.58 \times 10^{-3}$	$3.50 \times 10^{-3}$	$1.54 \times 10^{-3}$	$9.86{ imes}10^{-4}$	$2.76 \times 10^{-4}$
	75 - 79	$1.69 \times 10^{-3}$		$4.41 \times 10^{-3}$	$2.86 \times 10^{-3}$	$2.43 \times 10^{-3}$	$9.86 \times 10^{-4}$	$4.62 \times 10^{-4}$
	80 - 84	$4.80 \times 10^{-4}$		$1.45 \times 10^{-2}$	$6.46 \times 10^{-3}$	$9.92 \times 10^{-4}$	$1.78 \times 10^{-3}$	$9.53 \times 10^{-4}$
1-2 ADLs	65 - 69	$1.55 \times 10^{-3}$	$1.09 \times 10^{-3}$		$3.32 \times 10^{-3}$	$9.72 \times 10^{-4}$	$6.83 \times 10^{-4}$	$2.87 \times 10^{-4}$
	70 - 74	$1.27 \times 10^{-3}$	$5.32 \times 10^{-3}$		$3.19 \times 10^{-3}$	$1.68 \times 10^{-3}$	$6.24 \times 10^{-4}$	$1.96 \times 10^{-4}$
	75 - 79	$1.08 \times 10^{-3}$	$1.30 \times 10^{-3}$		$4.65 \times 10^{-3}$	$5.61 \times 10^{-3}$	$2.30 \times 10^{-3}$	$4.11 \times 10^{-4}$
	80 - 84	$1.86 \times 10^{-3}$	$5.52 \times 10^{-3}$		$2.82 \times 10^{-2}$	$4.34 \times 10^{-3}$	$6.82 \times 10^{-3}$	$5.34 \times 10^{-3}$
3-4 ADLs	65 - 69	$2.97 \times 10^{-3}$	$3.53 \times 10^{-3}$	$9.01 \times 10^{-3}$		$7.80 \times 10^{-3}$	$1.45 \times 10^{-3}$	$5.01 \times 10^{-4}$
	70 - 74	$1.59 \times 10^{-3}$	$7.69 \times 10^{-3}$	$4.34 \times 10^{-3}$		$9.20 \times 10^{-3}$	$3.08 \times 10^{-3}$	$1.29 \times 10^{-3}$
	75 - 79	$2.72 \times 10^{-3}$	$4.10 \times 10^{-3}$	$3.89 \times 10^{-3}$		$1.76 \times 10^{-2}$	$8.45 \times 10^{-3}$	$1.44 \times 10^{-3}$
	80 - 84	$1.54 \times 10^{-3}$	$4.84 \times 10^{-4}$	$5.11 \times 10^{-3}$		$5.17 \times 10^{-3}$	$3.76 \times 10^{-3}$	$4.86 \times 10^{-3}$
5-6 ADLs	65 - 69	$1.72 \times 10^{-3}$	$1.50 \times 10^{-3}$	$9.30 \times 10^{-3}$	$3.05 \times 10^{-3}$		$3.49 \times 10^{-3}$	$1.22 \times 10^{-3}$
	70 - 74	$1.47 \times 10^{-3}$	$5.70 \times 10^{-3}$	$3.17 \times 10^{-3}$	$1.01 \times 10^{-3}$		$3.22 \times 10^{-3}$	$6.57 \times 10^{-4}$
	75 - 79	$3.91 \times 10^{-3}$	$1.68 \times 10^{-3}$	$2.24 \times 10^{-3}$	$6.98 \times 10^{-3}$		$5.55 \times 10^{-3}$	$2.23 \times 10^{-3}$
	80 - 84	$5.78 \times 10^{-4}$	$3.64 \times 10^{-4}$	$3.96 \times 10^{-3}$	$3.04 \times 10^{-3}$	-	$1.48 \times 10^{-3}$	$5.95 \times 10^{-4}$
Inst'd	65 - 69	$2.63 \times 10^{-4}$	$1.10 \times 10^{-3}$	$1.51 \times 10^{-4}$	$1.96 \times 10^{-4}$	$3.78 \times 10^{-5}$		$3.17 \times 10^{-4}$
	70 - 74	$7.10 \times 10^{-4}$	$5.08 \times 10^{-4}$	$4.95 \times 10^{-4}$	$3.27 \times 10^{-4}$	$1.80 \times 10^{-4}$		$3.16 \times 10^{-4}$
	75 - 79	$1.64 \times 10^{-4}$	$1.27 \times 10^{-4}$	$7.58 \times 10^{-4}$	$2.11 \times 10^{-4}$	$2.32 \times 10^{-4}$		$3.65 \times 10^{-4}$
	80 - 84	$2.67 \times 10^{-4}$	$1.09 \times 10^{-4}$	$4.16 \times 10^{-4}$	$8.83 \times 10^{-4}$	$1.23 \times 10^{-4}$		$3.41 \times 10^{-4}$

### Appendix K

# Tables of parameter values for the parametric transition intensities fitted to data grouped in 10-year age bands, calculated from the 1982, 1984, 1989 and 1994 NLTCS

The tables in this appendix give the parameter values for the parametric transition intensities  $(\overset{ij}{\mu}_{x+t}^{ij})$  fitted to the NLTCS in 10-year age bands, where, for  $65 \leq x+t \leq$  120:

$$\mathring{\mu}_{x+t}^{ij} = \begin{cases} A_{ij} + B_{ij} e^{C_{ij}((x-(70+M))+t)} & \text{if } (\overline{\mu}_{70+M}^{ij} - \overline{\mu}_{80+M}^{ij}) < (\overline{\mu}_{80+M}^{ij} - \overline{\mu}_{90+M}^{ij}) , \\ & \overline{\mu}_{70+M}^{ij} \neq \overline{\mu}_{80+M}^{ij} \neq \overline{\mu}_{90+M}^{ij} \text{ and } |C_{ij}| < 0.5 \\ & A_{ij} + D_{ij} (x+t) & \text{otherwise} \end{cases}$$

and with a lower bound of zero on all intensities at all ages and where M = 1 for the 1982–84 NLTCS and M = 2.5 for the 1984-89 and 1989–94 NLTCS (see Section 5.5 for more details). Tables K.153 and K.154 give the parameter values for the 1982–84 NLTCS for males and females—in aggregate the values are given in Table 5.65. The values for males, female and in aggregate for the 1984–89 and 1989–94 NLTCS are given in Tables K.155 to K.160.

Table K.153:	Parameter	values for	the par	ametric	${\rm transition}$	intensities	for	males,
calculated fro	om the 1982	and 1984	NLTCS,	grouped	d in 10-yea	r age band	s.	

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$1.95 \times 10^{-4}$	$2.03 \times 10^{-2}$	$8.63 \times 10^{-2}$	_
J	1–2 ADLs	$7.89 \times 10^{-3}$	$4.94 \times 10^{-3}$	$1.66 \times 10^{-1}$	-
	3–4 ADLs	$-1.82 \times 10^{-3}$	$5.45 \times 10^{-3}$	$-5.47 \times 10^{-2}$	_
	5–6 ADLs	$-7.80 \times 10^{-3}$	-	-	$1.70 \times 10^{-4}$
	Inst'd	$-4.66 \times 10^{-2}$	-	-	$7.20 \times 10^{-4}$
	Dead	$1.43 \times 10^{-2}$	$2.26 \times 10^{-2}$	$7.80 \times 10^{-2}$	-
IADL only	Healthy	$1.20 \times 10^{0}$	_	_	$-1.31 \times 10^{-2}$
J	1-2 $ADLs$	$-1.82 \times 10^{-1}$	-	-	$5.02 \times 10^{-3}$
	3–4 ADLs	$-3.24 \times 10^{-1}$	-	-	$4.58 \times 10^{-3}$
	5–6 ADLs	$-1.71 \times 10^{-2}$	-	-	$1.07 \times 10^{-3}$
	Inst'd	$-1.81 \times 10^{-1}$	-	-	$2.79 \times 10^{-3}$
	Dead	$8.06 \times 10^{-2}$	-	-	$1.77 \times 10^{-4}$
1-2 ADLs	Healthy	$-6.97 \times 10^{-3}$	_	-	$7.46 \times 10^{-4}$
	IADL only	$5.06 \times 10^{-1}$	-	-	$-4.80 \times 10^{-3}$
	3–4 ADLs	$4.14 \times 10^{-1}$	-	-	$-2.50 \times 10^{-3}$
	5–6 ADLs	$-8.67 \times 10^{-2}$	-	-	$2.00 \times 10^{-3}$
	Inst'd	$2.32 \times 10^{-1}$	-	-	$-2.62 \times 10^{-3}$
	Dead	$-3.17 \times 10^{-1}$	-	-	$6.19 \times 10^{-3}$
3–4 ADLs	Healthy	$-6.78 \times 10^{-3}$	$9.58 \times 10^{-2}$	$-1.32 \times 10^{-1}$	-
	IADL only	$-1.58 \times 10^{-1}$	$1.58 \times 10^{-1}$	$2.18 \times 10^{-2}$	-
	1–2 ADLs	$-2.31 \times 10^{1}$	$2.34{ imes}10^1$	$-5.85 \times 10^{-4}$	-
	5–6 ADLs	$2.91 \times 10^{-1}$	$2.00 \times 10^{-2}$	$1.53 \times 10^{-1}$	-
	Inst'd	$-1.16 \times 10^{0}$	$1.17{ imes}10^{0}$	$1.01 \times 10^{-2}$	-
	Dead	$1.13 \times 10^{-2}$	-	-	$2.38 \times 10^{-3}$
5–6 ADLs	Healthy	$1.57 \times 10^{-1}$	-	-	$-1.71 \times 10^{-3}$
	IADL only	$2.41 \times 10^{-1}$	-	-	$-2.55 \times 10^{-3}$
	1–2 ADLs	$3.20 \times 10^{-1}$	-	-	$-3.18 \times 10^{-3}$
	3–4 ADLs	$1.78 \times 10^{-1}$	-	-	$-5.49 \times 10^{-5}$
	Inst'd	$7.40 \times 10^{-3}$	-	-	$8.57 \times 10^{-4}$
	Dead	$1.84 \times 10^{-1}$	$6.38 \times 10^{-2}$	$7.88 \times 10^{-2}$	-
Inst'd	Healthy	$-1.91 \times 10^{-3}$	$3.73 \times 10^{-2}$	$-1.03 \times 10^{-1}$	-
	IADL only	$4.37 \times 10^{-2}$	-	-	$-4.79 \times 10^{-4}$
	1–2 ADLs	$6.59 \times 10^{-4}$	$2.58 \times 10^{-2}$	$-2.58{ imes}10^{-1}$	-
	3–4 ADLs	$4.32 \times 10^{-2}$	-	-	$-4.70 \times 10^{-4}$
	5-6 ADLs	$-7.43 \times 10^{-5}$	$1.72 \times 10^{-2}$	$-2.54{ imes}10^{-1}$	-
	Dead	$-1.25{ imes}10^{-1}$	$3.46 \times 10^{-1}$	$2.80 \times 10^{-2}$	-

Table K.154:	Parameter	values for	the para	metric tran	nsition i	ntensities f	or females,
calculated from	om the 1982	and 1984	NLTCS,	grouped in	n 10-yea	ar age band	s.

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-4.25 \times 10^{-2}$	$7.04 \times 10^{-2}$	$3.70 \times 10^{-2}$	-
J	1–2 ADLs	$2.18 \times 10^{-3}$	$1.04 \times 10^{-2}$	$1.11 \times 10^{-1}$	-
	3–4 ADLs	$-4.12 \times 10^{-2}$	-	-	$6.32 \times 10^{-4}$
	5–6 ADLs	$4.17 \times 10^{-3}$	$9.64 \times 10^{-5}$	$2.89 \times 10^{-1}$	-
	Inst'd	$-1.67 \times 10^{-3}$	$4.67 \times 10^{-3}$	$1.39 \times 10^{-1}$	-
	Dead	$-7.85 \times 10^{-2}$	$9.43 \times 10^{-2}$	$2.01 \times 10^{-2}$	-
IADL only	Healthy	$-3.04 \times 10^{-1}$	$5.25 \times 10^{-1}$	$-2.51 \times 10^{-2}$	-
v	1-2 $ADLs$	$-3.72 \times 10^{-1}$	-	-	$9.41 \times 10^{-3}$
	3–4 ADLs	$2.69 \times 10^{-1}$	-	-	$-3.18 \times 10^{-3}$
	5–6 ADLs	$8.29 \times 10^{-3}$	$7.20 \times 10^{-3}$	$1.24 \times 10^{-1}$	-
	Inst'd	$7.40 \times 10^{-3}$	$2.70 \times 10^{-2}$	$5.24 \times 10^{-2}$	-
	Dead	$-1.68 \times 10^{-1}$	-	-	$2.93 \times 10^{-3}$
1-2 ADLs	Healthy	$2.42 \times 10^{-1}$	-	-	$-2.26 \times 10^{-3}$
	IADL only	$6.08 \times 10^{-1}$	-	-	$-5.25 \times 10^{-3}$
	3–4 ADLs	$-3.69 \times 10^{-1}$	-	-	$7.30 \times 10^{-3}$
	5–6 ADLs	$1.66 \times 10^{-1}$	-	-	$-1.88 \times 10^{-3}$
	Inst'd	$-2.01 \times 10^{-1}$	-	-	$3.36 \times 10^{-3}$
	Dead	$-2.55 \times 10^{-3}$	-	-	$1.09 \times 10^{-3}$
3–4 ADLs	Healthy	$-9.79 \times 10^{-3}$	-	-	$1.60 \times 10^{-4}$
	IADL only	$-2.48 \times 10^{-1}$	-	-	$3.32 \times 10^{-3}$
	1-2 ADLs	$1.81 \times 10^{0}$	-	-	$-1.90 \times 10^{-2}$
	5-6 ADLs	$-6.34 \times 10^{-1}$	$8.96 \times 10^{-1}$	$1.43 \times 10^{-2}$	-
	Inst'd	$-1.64 \times 10^{-1}$	$1.91 \times 10^{-1}$	$2.80 \times 10^{-2}$	-
	Dead	$-1.24 \times 10^{-1}$	-	-	$2.11 \times 10^{-3}$
5–6 ADLs	Healthy	$9.16 \times 10^{-2}$	-	-	$-7.95 \times 10^{-4}$
	IADL only	$3.35 \times 10^{-1}$	-	-	$-3.67 \times 10^{-3}$
	1-2 ADLs	$-9.66{ imes}10^{-2}$	-	-	$1.86 \times 10^{-3}$
	3–4 ADLs	$1.44 \times 10^{-1}$	-	-	$2.24 \times 10^{-4}$
	Inst'd	$-1.88{ imes}10^{-1}$	-	-	$3.79 \times 10^{-3}$
	Dead	$-1.05 \times 10^{-1}$	-	-	$4.31 \times 10^{-3}$
Inst'd	Healthy	$4.29 \times 10^{-3}$	$2.43 \times 10^{-2}$	$-2.08{ imes}10^{-1}$	-
	IADL only	$2.69 \times 10^{-2}$	-	-	$-2.62 \times 10^{-4}$
	1-2 ADLs	$1.22 \times 10^{-2}$	-	-	$-1.33 \times 10^{-4}$
	3-4 ADLs	$1.71 \times 10^{-2}$	-	-	$-8.91 \times 10^{-5}$
	5-6 ADLs	$2.91 \times 10^{-2}$	-	-	$-2.70 \times 10^{-4}$
	Dead	$2.23 \times 10^{-2}$	$1.21 \times 10^{-1}$	$4.30 \times 10^{-2}$	-

Table K.155:	Parameter	values for	the par-	ametric	${\rm transition}$	intensities	for	males,
calculated fro	om the 1984	and 1989	NLTCS,	grouped	d in 10-yea	r age bands	5.	

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$1.16 \times 10^{-2}$	$3.90 \times 10^{-3}$	$1.54 \times 10^{-1}$	-
v	1-2 ADLs	$-5.02 \times 10^{-2}$	-	-	$8.23 \times 10^{-4}$
	3–4 ADLs	$-3.44 \times 10^{-2}$	-	-	$5.02 \times 10^{-4}$
	5–6 ADLs	$2.20 \times 10^{-3}$	-	-	$-1.47 \times 10^{-5}$
	Inst'd	$-9.83 \times 10^{-3}$	$1.30 \times 10^{-2}$	$5.57 \times 10^{-2}$	-
	Dead	$-2.99{ imes}10^{-1}$	$3.03 \times 10^{-1}$	$2.23 \times 10^{-3}$	-
IADL only	Healthy	$2.45 \times 10^{-1}$	-	-	$-2.06 \times 10^{-3}$
5	$1-2$ $\dot{ADLs}$	$6.75 \times 10^{-1}$	-	-	$-7.21 \times 10^{-3}$
	3–4 ADLs	$-7.18 \times 10^{-1}$	-	-	$9.97 \times 10^{-3}$
	5–6 ADLs	$-1.29 \times 10^{-1}$	-	-	$2.06 \times 10^{-3}$
	Inst'd	$1.86 \times 10^{-1}$	-	-	$-2.12 \times 10^{-3}$
	Dead	$1.93 \times 10^{-2}$	$7.96 \times 10^{-6}$	$4.17 \times 10^{-1}$	-
1-2 ADLs	Healthy	$1.66 \times 10^{-1}$	-	-	$-1.64 \times 10^{-3}$
	IADL only	$1.30 \times 10^{-1}$	-	-	$-2.83 \times 10^{-4}$
	3–4 ADLs	$4.87 \times 10^{-1}$	-	-	$-4.76 \times 10^{-3}$
	5–6 ADLs	$-1.69 \times 10^{-1}$	-	-	$2.81 \times 10^{-3}$
	Inst'd	$-1.87{ imes}10^{-1}$	-	-	$2.67 \times 10^{-3}$
	Dead	$-8.52 \times 10^{-2}$	-	-	$1.26 \times 10^{-3}$
3–4 ADLs	Healthy	$5.88 \times 10^{-2}$	-	-	$-6.60 \times 10^{-4}$
	IADL only	$-4.06 \times 10^{-8}$	$4.43 \times 10^{-8}$	$1.07 \times 10^{-1}$	-
	1-2 ADLs	$-1.07 \times 10^{-1}$	$3.56 \times 10^{-1}$	$-5.98 \times 10^{-2}$	-
	5–6 ADLs	$4.14 \times 10^{-1}$	-	-	$-4.45 \times 10^{-3}$
	Inst'd	$-5.22 \times 10^{-1}$	-	-	$7.90 \times 10^{-3}$
	Dead	$9.14 \times 10^{-3}$	$7.63 \times 10^{-3}$	$1.24 \times 10^{-1}$	-
5–6 ADLs	Healthy	$3.85 \times 10^{-2}$	-	-	$-4.29 \times 10^{-4}$
	IADL only	$9.41 \times 10^{-2}$	-	-	$-1.02 \times 10^{-3}$
	1-2 ADLs	$3.20 \times 10^{-1}$	-	-	$-3.44 \times 10^{-3}$
	3–4 ADLs	$4.53 \times 10^{-2}$	-	-	$2.49 \times 10^{-4}$
	Inst'd	$-1.82{ imes}10^{-1}$	-	-	$3.04 \times 10^{-3}$
	Dead	$-1.04 \times 10^{-2}$	-	-	$3.10 \times 10^{-4}$
Inst'd	Healthy	$-3.59 \times 10^{-2}$	$4.06 \times 10^{-2}$	$1.30 \times 10^{-2}$	-
	IADL only	$7.86 \times 10^{-2}$	-	-	$-9.36 \times 10^{-4}$
	1–2 ADLs	$-8.65 \times 10^{-5}$	$8.66 \times 10^{-5}$	$1.84 \times 10^{-1}$	-
	3–4 ADLs	$3.28 \times 10^{-2}$	-	-	$-3.39 \times 10^{-4}$
	5–6 ADLs	$-3.44 \times 10^{-3}$	-	-	$4.71 \times 10^{-5}$
	Dead	$-7.66{ imes}10^{-2}$	-	-	$1.49 \times 10^{-3}$

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$1.31 \times 10^{-2}$	$5.58 \times 10^{-3}$	$9.48 \times 10^{-2}$	-
· ·	1–2 ADLs	$-1.65{ imes}10^{-1}$	-	-	$2.40 \times 10^{-3}$
	3–4 ADLs	$3.16 \times 10^{-3}$	$1.39 \times 10^{-4}$	$2.68 \times 10^{-1}$	-
	5-6 ADLs	$-1.81 \times 10^{-2}$	-	-	$3.08 \times 10^{-4}$
	Inst'd	$2.86 \times 10^{-3}$	$1.45 \times 10^{-3}$	$1.77 \times 10^{-1}$	-
	Dead	$-2.13 \times 10^{-2}$	-	-	$3.26 \times 10^{-4}$
IADL only	Healthy	$-1.91 \times 10^{-2}$	$1.21 \times 10^{-1}$	$-7.33 \times 10^{-2}$	-
· ·	1–2 ADLs	$-9.61 \times 10^{-3}$	-	-	$2.65 \times 10^{-3}$
	3–4 ADLs	$-5.68 \times 10^{-2}$	-	-	$8.11 \times 10^{-4}$
	5–6 ADLs	$2.08 \times 10^{-2}$	$2.09 \times 10^{-4}$	$2.82 \times 10^{-1}$	-
	Inst'd	$2.59 \times 10^{-3}$	$1.42 \times 10^{-2}$	$5.99 \times 10^{-2}$	-
	Dead	$7.97{ imes}10^{-3}$	$1.06 \times 10^{-3}$	$1.63 \times 10^{-1}$	-
1-2 ADLs	Healthy	$1.24 \times 10^{-1}$	-	-	$-1.21 \times 10^{-3}$
	IADL only	$4.67 \times 10^{-2}$	-	-	$4.38 \times 10^{-5}$
	3–4 ADLs	$-2.52 \times 10^{-1}$	-	-	$4.69 \times 10^{-3}$
	5–6 ADLs	$-7.70 \times 10^{-2}$	-	-	$1.31 \times 10^{-3}$
	Inst'd	$-2.40 \times 10^{-1}$	-	-	$3.63 \times 10^{-3}$
	Dead	$6.94 \times 10^{-3}$	-	-	$-3.86 \times 10^{-5}$
3–4 ADLs	Healthy	$-3.49 \times 10^{-2}$	-	-	$5.57 \times 10^{-4}$
	IADL only	$1.34 \times 10^{-1}$	-	-	$-1.44 \times 10^{-3}$
	1–2 ADLs	$7.54 \times 10^{-2}$	$1.29 \times 10^{-2}$	$-1.01 \times 10^{-1}$	-
	5–6 ADLs	$1.56 \times 10^{-1}$	-	-	$-7.28 \times 10^{-4}$
	Inst'd	$-3.31 \times 10^{-1}$	-	-	$4.84 \times 10^{-3}$
	Dead	$2.08 \times 10^{-2}$	$4.38 \times 10^{-4}$	$2.08 \times 10^{-1}$	-
5–6 ADLs	Healthy	$4.69 \times 10^{-2}$	-	-	$-4.29 \times 10^{-4}$
	IADL only	$5.27 \times 10^{-2}$	-	-	$-5.66 \times 10^{-4}$
	1–2 ADLs	$-2.06 \times 10^{-2}$	$1.05 \times 10^{-1}$	$-4.08 \times 10^{-2}$	-
	3–4 ADLs	$3.14 \times 10^{-2}$	$1.12 \times 10^{-1}$	$-2.16 \times 10^{-1}$	-
	Inst'd	$-9.05 \times 10^{-2}$	-	-	$1.99 \times 10^{-3}$
	Dead	$-1.69 \times 10^{-1}$	-	-	$2.42 \times 10^{-3}$
Inst'd	Healthy	$1.34 \times 10^{-2}$	-	-	$-8.55 \times 10^{-5}$
	IADL only	$2.53 \times 10^{-5}$	-	-	$1.98 \times 10^{-5}$
	1-2 ADLs	$4.30 \times 10^{-2}$	-	-	$-4.50 \times 10^{-4}$
	3–4 ADLs	$2.42 \times 10^{-3}$	-	-	$-2.51 \times 10^{-5}$
	5–6 ADLs	$-9.74 \times 10^{-3}$	-	-	$1.30 \times 10^{-4}$
	Dead	$1.47 \times 10^{-2}$	$4.78 \times 10^{-3}$	$1.06 \times 10^{-1}$	-

Table K.156: Parameter values for the parametric transition intensities for females, calculated from the 1984 and 1989 NLTCS, grouped in 10-year age bands.

Table K.157: Parameter values for the parametric transition intensities for males and females, calculated from the 1984 and 1989 NLTCS, grouped in 10-year age bands.

From	То		Para	meter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$1.31 \times 10^{-2}$	$4.45 \times 10^{-3}$	$1.22 \times 10^{-1}$	-
, i i i i i i i i i i i i i i i i i i i	1–2 ADLs	$-1.23 \times 10^{-1}$	-	-	$1.82 \times 10^{-3}$
	3–4 ADLs	$2.52 \times 10^{-3}$	$5.37 \times 10^{-4}$	$1.91 \times 10^{-1}$	-
	5-6 ADLs	$-2.10 \times 10^{-2}$	-	-	$3.28 \times 10^{-4}$
	Inst'd	$1.50 \times 10^{-3}$	$2.35 \times 10^{-3}$	$1.46 \times 10^{-1}$	-
	Dead	$-3.18 \times 10^{-2}$	-	-	$4.83 \times 10^{-4}$
IADL only	Healthy	$-1.22 \times 10^{-2}$	$1.12 \times 10^{-1}$	$-6.38 \times 10^{-2}$	-
	1-2 ADLs	$1.76 \times 10^{-1}$	$2.79 \times 10^{-4}$	$2.85 \times 10^{-1}$	-
	3–4 ADLs	$-1.87{ imes}10^{-1}$	-	-	$2.62 \times 10^{-3}$
	5-6 ADLs	$-8.88 \times 10^{-3}$	$2.94 \times 10^{-2}$	$5.19 \times 10^{-2}$	-
	Inst'd	$2.23 \times 10^{-2}$	$1.87{ imes}10^{-4}$	$2.59{ imes}10^{-1}$	-
	Dead	$1.25 \times 10^{-2}$	$7.16 \times 10^{-5}$	$3.04 \times 10^{-1}$	-
1-2 ADLs	Healthy	$1.34 \times 10^{-1}$	-	-	$-1.31 \times 10^{-3}$
	IADL only	$5.24 \times 10^{-2}$	-	-	$2.08 \times 10^{-4}$
	3–4 ADLs	$6.46 \times 10^{-2}$	$3.92 \times 10^{-2}$	$5.92 \times 10^{-2}$	-
	5-6 ADLs	$2.25 \times 10^{-2}$	$1.16 \times 10^{-3}$	$1.59 \times 10^{-1}$	-
	Inst'd	$-2.25{ imes}10^{-1}$	-	-	$3.37 \times 10^{-3}$
	Dead	$1.60 \times 10^{-3}$	-	-	$4.54 \times 10^{-5}$
3–4 ADLs	Healthy	$-2.88 \times 10^{-3}$	-	-	$1.38 \times 10^{-4}$
	IADL only	$1.05 \times 10^{-1}$	-	-	$-1.18 \times 10^{-3}$
	1–2 ADLs	$6.65 \times 10^{-2}$	$6.66 \times 10^{-2}$	$-7.53{ imes}10^{-2}$	-
	5-6 ADLs	$1.24 \times 10^{-1}$	-	-	$-2.24 \times 10^{-4}$
	Inst'd	$-3.44{ imes}10^{-1}$	-	-	$5.13 \times 10^{-3}$
	Dead	$1.54 \times 10^{-2}$	$4.38 \times 10^{-3}$	$1.13 \times 10^{-1}$	-
5–6 ADLs	Healthy	$3.38 \times 10^{-2}$	-	-	$-3.13 \times 10^{-4}$
	IADL only	$8.72 \times 10^{-2}$	-	-	$-9.36 \times 10^{-4}$
	1–2 ADLs	$2.62 \times 10^{-1}$	-	-	$-2.58{ imes}10^{-3}$
	3–4 ADLs	$-2.86 \times 10^{-2}$	$1.37 \times 10^{-1}$	$-3.31 \times 10^{-2}$	-
	Inst'd	$5.39 \times 10^{-2}$	$3.95 \times 10^{-5}$	$3.75 \times 10^{-1}$	-
	Dead	$-1.26 \times 10^{-1}$	-	-	$1.86 \times 10^{-3}$
Inst'd	Healthy	$6.67 \times 10^{-3}$	$1.27 \times 10^{-5}$	$2.15 \times 10^{-1}$	-
	IADL only	$-1.58 \times 10^{-2}$	$2.21 \times 10^{-2}$	$-1.45 \times 10^{-2}$	-
	1–2 ADLs	$2.36 \times 10^{-2}$	-	-	$-2.30{ imes}10^{-4}$
	3–4 ADLs	$1.59 \times 10^{-4}$	$4.37 \times 10^{-3}$	$-2.44 \times 10^{-1}$	-
	5-6 ADLs	$-4.64 \times 10^{-3}$	-	-	$7.11 \times 10^{-5}$
	Dead	$-9.34 \times 10^{-2}$	-	-	$1.60 \times 10^{-3}$

Table K.158: Parameter values for the parametric transition intensities for males, calculated from the 1989 and 1994 NLTCS, grouped in 10-year age bands.

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-6.62 \times 10^{-2}$	_	_	$1.16 \times 10^{-3}$
v	1–2 ADLs	$-4.25 \times 10^{-2}$	$5.45 \times 10^{-2}$	$3.04 \times 10^{-2}$	-
	3–4 ADLs	$-9.58 \times 10^{-3}$	-	-	$1.83 \times 10^{-4}$
	5–6 ADLs	$3.35 \times 10^{-3}$	$2.12 \times 10^{-4}$	$2.22 \times 10^{-1}$	-
	Inst'd	$3.45 \times 10^{-3}$	$2.82 \times 10^{-4}$	$2.71 \times 10^{-1}$	-
	Dead	$3.47 \times 10^{-3}$	$2.14 \times 10^{-3}$	$1.02 \times 10^{-1}$	-
IADL only	Healthy	$-8.16 \times 10^{-2}$	$2.45 \times 10^{-1}$	$-5.50 \times 10^{-2}$	-
0	$1-2$ $\dot{ADLs}$	$-1.61 \times 10^{-2}$	-	-	$2.03 \times 10^{-3}$
	3–4 ADLs	$2.85 \times 10^{-2}$	-	-	$2.05 \times 10^{-4}$
	5–6 ADLs	$-1.05 \times 10^{-8}$	$1.36 \times 10^{-8}$	$1.62 \times 10^{-1}$	-
	Inst'd	$-2.65 \times 10^{-2}$	$5.43 \times 10^{-2}$	$-3.58 \times 10^{-2}$	-
	Dead	$8.38 \times 10^{-2}$	-	-	$-8.78 \times 10^{-4}$
1-2 ADLs	Healthy	$1.79 \times 10^{-1}$	_	_	$-1.74 \times 10^{-3}$
	IADL only	$-8.74 \times 10^{-2}$	$1.61 \times 10^{-1}$	$-1.82 \times 10^{-2}$	_
	3–4 ADLs	$-5.04 \times 10^{-2}$	-	-	$2.17 \times 10^{-3}$
	5–6 ADLs	$-6.43 \times 10^{-2}$	$6.43 \times 10^{-2}$	$7.14 \times 10^{-2}$	-
	Inst'd	$-3.52{ imes}10^{-1}$	-	-	$5.32 \times 10^{-3}$
	Dead	$-1.83 \times 10^{-1}$	-	-	$2.56 \times 10^{-3}$
3–4 ADLs	Healthy	$-1.54 \times 10^{-1}$	$1.96 \times 10^{-1}$	$-1.20 \times 10^{-2}$	_
	IADL only	$1.03 \times 10^{-1}$	_	_	$-1.01 \times 10^{-3}$
	1-2 ADLs	$4.97 \times 10^{-1}$	-	-	$-6.02 \times 10^{-3}$
	5–6 ADLs	$5.04 \times 10^{-1}$	-	-	$-4.44 \times 10^{-3}$
	Inst'd	$-8.14 \times 10^{-1}$	-	-	$1.14 \times 10^{-2}$
	Dead	$-2.38 \times 10^{-1}$	-	-	$3.70 \times 10^{-3}$
5–6 ADLs	Healthy	$-6.00 \times 10^{-2}$	_	_	$1.09 \times 10^{-3}$
	IADL only	$1.10 \times 10^{-1}$	_	_	$-1.33 \times 10^{-3}$
	1–2 ADLs	$-6.04 \times 10^{-3}$	$7.30 \times 10^{-2}$	$-1.25 \times 10^{-1}$	-
	3–4 ADLs	$-3.17 \times 10^{-1}$	-	-	$4.38 \times 10^{-3}$
	Inst'd	$4.43 \times 10^{-2}$	-	-	$4.60 \times 10^{-4}$
	Dead	$2.02 \times 10^{-2}$	$1.65 \times 10^{-3}$	$2.06 \times 10^{-1}$	-
Inst'd	Healthy	$3.66 \times 10^{-2}$	_	_	$-2.68 \times 10^{-4}$
	IADL only	$8.84 \times 10^{-2}$	_	_	$-1.02 \times 10^{-3}$
	1-2 ADLs	$-6.62 \times 10^{-2}$	_	_	$9.53 \times 10^{-4}$
	3–4 ADLs	$-4.52 \times 10^{-2}$	-	-	$6.79 \times 10^{-4}$
	5–6 ADLs	$2.46 \times 10^{-3}$	_	-	$-1.53 \times 10^{-5}$
	Dead	$2.05 \times 10^{-2}$	$1.22 \times 10^{-4}$	$3.04 \times 10^{-1}$	-

From	То		Para	meter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-1.42 \times 10^{-1}$	-	-	$2.16 \times 10^{-3}$
· ·	1–2 ADLs	$1.30 \times 10^{-2}$	$9.14 \times 10^{-3}$	$9.09 \times 10^{-2}$	-
	3–4 ADLs	$-3.99 \times 10^{-2}$	-	-	$5.50 \times 10^{-4}$
	5-6 ADLs	$2.45 \times 10^{-3}$	$3.17 \times 10^{-4}$	$2.52{ imes}10^{-1}$	-
	Inst'd	$1.26 \times 10^{-3}$	$6.09 \times 10^{-3}$	$1.13 \times 10^{-1}$	-
	Dead	$-1.41 \times 10^{-2}$	-	-	$2.26 \times 10^{-4}$
IADL only	Healthy	$4.49 \times 10^{-1}$	-	-	$-4.51 \times 10^{-3}$
-	1-2 ADLs	$-4.13 \times 10^{-1}$	-	-	$7.85 \times 10^{-3}$
	3–4 ADLs	$1.00 \times 10^{-1}$	-	-	$-9.93 \times 10^{-4}$
	5-6 ADLs	$-6.14 \times 10^{-2}$	-	-	$1.07 \times 10^{-3}$
	Inst'd	$-1.35 \times 10^{-1}$	-	-	$1.98 \times 10^{-3}$
	Dead	$-5.55 \times 10^{-2}$	-	-	$8.57 \times 10^{-4}$
1-2 ADLs	Healthy	$1.30 \times 10^{-1}$	-	-	$-1.09 \times 10^{-3}$
	IADL only	$1.19 \times 10^{-1}$	-	-	$-5.25 \times 10^{-4}$
	3–4 ADLs	$1.18 \times 10^{-1}$	$2.35 \times 10^{-2}$	$1.07 \times 10^{-1}$	-
	5-6 ADLs	$1.09 \times 10^{-1}$	-	-	$-9.19 \times 10^{-4}$
	Inst'd	$-6.54 \times 10^{-2}$	$6.54 \times 10^{-2}$	$4.70 \times 10^{-2}$	-
	Dead	$-7.16 \times 10^{-2}$	-	-	$1.05 \times 10^{-3}$
3–4 ADLs	Healthy	$9.54 \times 10^{-2}$	-	-	$-1.02 \times 10^{-3}$
	IADL only	$4.20 \times 10^{-2}$	-	-	$-2.03 \times 10^{-4}$
	1-2 ADLs	$-2.06 \times 10^{-3}$	$8.04 \times 10^{-2}$	$4.34 \times 10^{-2}$	-
	5-6 ADLs	$-1.88 \times 10^{-1}$	$2.50 \times 10^{-1}$	$2.22 \times 10^{-2}$	-
	Inst'd	$7.95  imes 10^{-2}$	$8.85 \times 10^{-4}$	$2.35{ imes}10^{-1}$	-
	Dead	$2.86 \times 10^{-2}$	-	-	$-2.20 \times 10^{-4}$
5–6 ADLs	Healthy	$1.72 \times 10^{-2}$	$3.66 \times 10^{-6}$	$4.50 \times 10^{-1}$	-
	IADL only	$3.44 \times 10^{-2}$	-	-	$-3.62 \times 10^{-4}$
	1-2 ADLs	$4.35 \times 10^{-2}$	-	-	$-2.62{ imes}10^{-4}$
	3-4 ADLs	$4.59 \times 10^{-2}$	-	-	$4.54 \times 10^{-4}$
	Inst'd	$-2.35{ imes}10^{-1}$	-	-	$3.75 \times 10^{-3}$
	Dead	$1.20 \times 10^{-2}$	-	-	$3.17 \times 10^{-4}$
Inst'd	Healthy	$3.99 \times 10^{-2}$	-	-	$-3.21 \times 10^{-4}$
	IADL only	$8.16 \times 10^{-3}$	-	-	$-6.29 \times 10^{-5}$
	1-2 ADLs	$3.90 \times 10^{-4}$	$1.92 \times 10^{-2}$	$-1.11 \times 10^{-1}$	-
	3–4 ADLs	$-1.21 \times 10^{-3}$	-	-	$3.98 \times 10^{-5}$
	5-6 ADLs	$9.81 \times 10^{-3}$	-	-	$-9.73 \times 10^{-5}$
	Dead	$-3.68 \times 10^{-3}$	_	_	$3.90 \times 10^{-4}$

Table K.159: Parameter values for the parametric transition intensities for females, calculated from the 1989 and 1994 NLTCS, grouped in 10-year age bands.
Table K.160: Parameter values for the parametric transition intensities for males and females, calculated from the 1989 and 1994 NLTCS, grouped in 10-year age bands.

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-1.11 \times 10^{-1}$	-	-	$1.75 \times 10^{-3}$
Č.	1–2 ADLs	$6.33 \times 10^{-3}$	$1.18 \times 10^{-2}$	$8.00 \times 10^{-2}$	-
	3–4 ADLs	$-3.43 \times 10^{-2}$	-	-	$4.83 \times 10^{-4}$
	5–6 ADLs	$3.40 \times 10^{-3}$	$2.36 \times 10^{-4}$	$2.54 \times 10^{-1}$	-
	Inst'd	$2.75 \times 10^{-3}$	$3.19 \times 10^{-3}$	$1.42 \times 10^{-1}$	-
	Dead	$1.73 \times 10^{-3}$	$2.06 \times 10^{-3}$	$8.86 \times 10^{-2}$	-
IADL only	Healthy	$5.82 \times 10^{-1}$	-	-	$-6.17 \times 10^{-3}$
	1-2 ADLs	$-4.35 \times 10^{-1}$	-	-	$8.02 \times 10^{-3}$
	3–4 ADLs	$2.10 \times 10^{-1}$	-	-	$-2.46 \times 10^{-3}$
	5–6 ADLs	$-1.04 \times 10^{-1}$	-	-	$1.57 \times 10^{-3}$
	Inst'd	$-5.72{ imes}10^{-2}$	-	-	$1.01 \times 10^{-3}$
	Dead	$-2.09 \times 10^{-2}$	-	-	$4.49 \times 10^{-4}$
1-2 ADLs	Healthy	$1.39 \times 10^{-1}$	-	-	$-1.21 \times 10^{-3}$
	IADL only	$1.82 \times 10^{-1}$	-	-	$-1.41 \times 10^{-3}$
	3–4 ADLs	$1.47{ imes}10^{-1}$	$8.04 \times 10^{-3}$	$1.57{ imes}10^{-1}$	-
	5-6 ADLs	$-1.10 \times 10^{-1}$	-	-	$1.76 \times 10^{-3}$
	Inst'd	$-1.39{ imes}10^{-1}$	$1.44 \times 10^{-1}$	$2.75 \times 10^{-2}$	-
	Dead	$-9.87{ imes}10^{-2}$	-	-	$1.41 \times 10^{-3}$
3–4 ADLs	Healthy	$1.48 \times 10^{-1}$	-	-	$-1.65 \times 10^{-3}$
	IADL only	$2.04 \times 10^{-2}$	-	-	$9.92 \times 10^{-5}$
	1–2 ADLs	$7.79 \times 10^{-2}$	$1.62 \times 10^{-3}$	$2.44 \times 10^{-1}$	-
	5-6 ADLs	$1.10 \times 10^{-1}$	$3.59 \times 10^{-3}$	$1.64 \times 10^{-1}$	-
	Inst'd	$-2.64{ imes}10^{-2}$	$9.28 \times 10^{-2}$	$2.82 \times 10^{-2}$	-
	Dead	$1.95 \times 10^{-2}$	-	-	$-1.37 \times 10^{-5}$
5–6 ADLs	Healthy	$-6.53{ imes}10^{-2}$	-	-	$1.11 \times 10^{-3}$
	IADL only	$3.99 \times 10^{-2}$	-	-	$-4.45 \times 10^{-4}$
	1-2 ADLs	$1.97{ imes}10^{-1}$	-	-	$-2.11 \times 10^{-3}$
	3–4 ADLs	$-1.28 \times 10^{-1}$	-	-	$2.47 \times 10^{-3}$
	Inst'd	$-1.62{ imes}10^{-1}$	-	-	$2.98 \times 10^{-3}$
	Dead	$-2.63 \times 10^{-2}$	-	-	$7.81 \times 10^{-4}$
Inst'd	Healthy	$3.98 \times 10^{-2}$	-	-	$-3.16 \times 10^{-4}$
	IADL only	$3.61 \times 10^{-3}$	$4.68 \times 10^{-3}$	$-2.82 \times 10^{-1}$	-
	1–2 ADLs	$5.60 \times 10^{-2}$	-	-	$-5.67{ imes}10^{-4}$
	3–4 ADLs	$-8.86 \times 10^{-3}$	-	-	$1.45 \times 10^{-4}$
	5-6 ADLs	$9.44 \times 10^{-3}$	-	-	$-9.00 \times 10^{-5}$
	Dead	$-2.84 \times 10^{-2}$	-	-	$7.03 \times 10^{-4}$

### Appendix L

# Tables of parameter values for the parametric transition intensities fitted to data grouped in 5-year age bands, calculated from the 1982, 1984, 1989 and 1994 NLTCS

The tables in this appendix give the parameter values for the parametric transition intensities  $(\mathring{\mu}_{x+t}^{ij})$  fitted to the NLTCS in 5-year age bands. The method used is, for each transition intensity, fit both a staight line and a Makeham curve using weighted least squares, the weights  $(w_{x+t}^{ij})$  being the inverse of the variance (i.e.  $w_{x+t}^{ij} = 1/\text{Var}[\overline{\mu}_{x+t}^{ij}]$ ). Then chose the parametric form which provides the best fit, in terms of the smallest sum of weighted squared residuals (see Section 5.6 for more details). Tables L.161 and L.162 give the parameter values for the 1982–84 NLTCS for males and females—in aggregate the values are given in Table 5.67. The values for males, female and in aggregate for the 1984–89 and 1989–94 NLTCS are given in Tables L.163 to L.168.

Table L.161:	Parameter	values for	• the par	$\operatorname{ametric}$	$\operatorname{transition}$	intensities	for	males,
calculated from	om the 1982	and 1984	NLTCS	, groupe	d in 5-year	age bands.	•	

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$1.06 \times 10^{-2}$	$7.88 \times 10^{-3}$	$1.18 \times 10^{-1}$	-
Ū	1-2 ADLs	$9.45 \times 10^{-4}$	$8.20 \times 10^{-3}$	$1.23 \times 10^{-1}$	-
	3–4 ADLs	$-2.01 \times 10^{-3}$	$6.66 \times 10^{-3}$	$-4.27 \times 10^{-2}$	-
	5–6 ADLs	$-2.66{ imes}10^{-2}$	-	-	$4.30 \times 10^{-4}$
	Inst'd	$-4.63 \times 10^{-3}$	$8.01 \times 10^{-3}$	$5.54 \times 10^{-2}$	-
	Dead	$-2.36{ imes}10^{-1}$	-	-	$3.83 \times 10^{-3}$
IADL only	Healthy	$1.14 \times 10^{0}$	-	-	$-1.24 \times 10^{-2}$
Ū.	1-2 ADLs	$7.33{ imes}10^{-2}$	$9.38 \times 10^{-2}$	$3.13 \times 10^{-2}$	-
	3–4 ADLs	$-3.11 \times 10^{-1}$	-	-	$4.42 \times 10^{-3}$
	5-6 ADLs	$1.72 \times 10^{-3}$	-	-	$5.71 \times 10^{-4}$
	Inst'd	$-1.73 \times 10^{-1}$	-	-	$2.67 \times 10^{-3}$
	Dead	$5.90 \times 10^{-3}$	-	-	$1.16 \times 10^{-3}$
1-2 ADLs	Healthy	$5.39 \times 10^{-2}$	-	-	$-4.02 \times 10^{-5}$
	IADL only	$4.94 \times 10^{-1}$	-	-	$-4.80 \times 10^{-3}$
	3–4 ADLs	$2.24 \times 10^{-1}$	-	-	$-4.91 \times 10^{-4}$
	5–6 ADLs	$1.47{ imes}10^{-1}$	-	-	$-1.08 \times 10^{-3}$
	Inst'd	$-9.68 \times 10^{-2}$	-	-	$1.71 \times 10^{-3}$
	Dead	$-2.96 \times 10^{-1}$	-	-	$5.86 \times 10^{-3}$
3–4 ADLs	Healthy	$-1.07 \times 10^{-1}$	$2.00 \times 10^{-1}$	$-2.91 \times 10^{-2}$	-
	IADL only	$-7.60 \times 10^{-2}$	$6.32 \times 10^{-2}$	$3.76 \times 10^{-2}$	-
	1–2 ADLs	$2.61 \times 10^{-2}$	$3.93 \times 10^{-1}$	$-1.25{ imes}10^{-1}$	-
	5–6 ADLs	$-5.53 \times 10^{-1}$	-	-	$1.13 \times 10^{-2}$
	Inst'd	$-1.73{ imes}10^{-1}$	$1.77 \times 10^{-1}$	$2.77{ imes}10^{-2}$	-
	Dead	$1.35 \times 10^{-1}$	-	-	$7.44 \times 10^{-4}$
5–6 ADLs	Healthy	$1.44 \times 10^{-1}$	-	-	$-1.56 \times 10^{-3}$
	IADL only	$2.39 \times 10^{-1}$	-	-	$-2.56 \times 10^{-3}$
	1–2 ADLs	$1.50 \times 10^{-1}$	-	-	$-1.16 \times 10^{-3}$
	3–4 ADLs	$1.71 \times 10^{-1}$	-	-	$-3.97 \times 10^{-4}$
	Inst'd	$1.38{ imes}10^{-2}$	-	-	$6.90 \times 10^{-4}$
	Dead	$1.86 \times 10^{-1}$	$4.90 \times 10^{-2}$	$7.21 \times 10^{-2}$	-
Inst'd	Healthy	$-8.33 \times 10^{-3}$	$3.35 \times 10^{-2}$	$-5.30 \times 10^{-2}$	-
	IADL only	$-8.54 \times 10^{-2}$	$9.50 \times 10^{-2}$	$-4.77 \times 10^{-3}$	-
	1-2 ADLs	$2.05 \times 10^{-2}$	-	-	$-2.17 \times 10^{-4}$
	3–4 ADLs	$5.91 \times 10^{-2}$	-	-	$-6.47 \times 10^{-4}$
	5-6 ADLs	$1.39 \times 10^{-2}$	-	-	$-1.52 \times 10^{-4}$
	Dead	$-9.25{ imes}10^{-1}$	-	-	$1.56 \times 10^{-2}$

Table L.16	2: Parame	eter values for the parametric transition intensities for females,
calculated	from the 1	1982 and 1984 NLTCS, grouped in 5-year age bands.
From	То	Parameter

I I UIII	10		I UI C		
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-2.11 \times 10^{-1}$	-	-	$3.38 \times 10^{-3}$
Č.	1–2 ADLs	$-5.58{ imes}10^{-2}$	$6.73 \times 10^{-2}$	$1.74 \times 10^{-2}$	-
	3–4 ADLs	$-5.27 \times 10^{-2}$	-	-	$7.94 \times 10^{-4}$
	5-6 ADLs	$1.26 \times 10^{-3}$	$2.07 \times 10^{-3}$	$8.71 \times 10^{-2}$	-
	Inst'd	$-2.66 \times 10^{-3}$	$4.02 \times 10^{-3}$	$1.29 \times 10^{-1}$	-
	Dead	$-7.96{ imes}10^{-3}$	$2.09 \times 10^{-2}$	$5.86 \times 10^{-2}$	-
IADL only	Healthy	$-1.87{ imes}10^{-1}$	$4.52 \times 10^{-1}$	$-3.59 \times 10^{-2}$	-
	1-2 ADLs	$-5.06{ imes}10^{-1}$	-	-	$1.11 \times 10^{-2}$
	3–4 ADLs	$-2.01 \times 10^{-3}$	$5.73 \times 10^{-2}$	$-1.83 \times 10^{-1}$	-
	5-6 ADLs	$7.89 \times 10^{-3}$	$4.83 \times 10^{-3}$	$1.28 \times 10^{-1}$	-
	Inst'd	$-1.58 \times 10^{-1}$	-	-	$2.63 \times 10^{-3}$
	Dead	$-1.37{ imes}10^{-1}$	-	-	$2.51 \times 10^{-3}$
1-2 ADLs	Healthy	$2.31 \times 10^{-1}$	-	-	$-2.14 \times 10^{-3}$
	IADL only	$5.75 \times 10^{-1}$	-	-	$-4.86 \times 10^{-3}$
	3–4 ADLs	$-3.29 \times 10^{-1}$	-	-	$6.73 \times 10^{-3}$
	5-6 ADLs	$-1.93{ imes}10^{-1}$	$2.22 \times 10^{-1}$	$-4.99 \times 10^{-3}$	-
	Inst'd	$-2.68{ imes}10^{-1}$	-	-	$4.21 \times 10^{-3}$
	Dead	$-1.35 \times 10^{-1}$	$2.03 \times 10^{-1}$	$6.05 \times 10^{-3}$	-
3–4 ADLs	Healthy	$2.40 \times 10^{-2}$	-	-	$-2.07{ imes}10^{-4}$
	IADL only	$-3.50 \times 10^{-2}$	$2.58 \times 10^{-2}$	$5.06 \times 10^{-2}$	-
	1-2 ADLs	$1.74{ imes}10^{0}$	-	-	$-1.83{ imes}10^{-2}$
	5-6 ADLs	$-1.07 \times 10^{0}$	-	-	$1.82 \times 10^{-2}$
	Inst'd	$-1.79{ imes}10^{-2}$	$4.31 \times 10^{-2}$	$6.04 \times 10^{-2}$	-
	Dead	$-1.25 \times 10^{-2}$	-	-	$6.87 \times 10^{-4}$
5-6 ADLs	Healthy	$7.75 \times 10^{-2}$	-	-	$-6.38{ imes}10^{-4}$
	IADL only	$3.04 \times 10^{-1}$	-	-	$-3.31 \times 10^{-3}$
	1-2 ADLs	$-2.84{ imes}10^{-2}$	$6.49 \times 10^{-2}$	$1.75 \times 10^{-2}$	-
	3-4 ADLs	$-3.95 \times 10^{-2}$	-	-	$2.40 \times 10^{-3}$
	Inst'd	$-2.92{ imes}10^{-1}$	-	-	$5.06 \times 10^{-3}$
	Dead	$1.12 \times 10^{-1}$	$6.61 \times 10^{-2}$	$4.35 \times 10^{-2}$	-
Inst'd	Healthy	$2.66 \times 10^{-3}$	$1.77 \times 10^{-2}$	$-9.62 \times 10^{-2}$	-
	IADL only	$1.63 \times 10^{-2}$	-	-	$-1.43 \times 10^{-4}$
	1–2 ADLs	$-1.22{ imes}10^{-1}$	$1.27{ imes}10^{-1}$	$-1.55{ imes}10^{-3}$	-
	3-4 ADLs	$6.59 \times 10^{-3}$	-	-	$3.19 \times 10^{-5}$
	5-6 ADLs	$5.13 \times 10^{-3}$	-	-	$-1.95 \times 10^{-5}$
	Dead	$-6.07 \times 10^{-1}$	-	-	$1.01 \times 10^{-2}$

Table L.163: Parameter values for the parametric transition intensities for males, calculated from the 1984 and 1989 NLTCS, grouped in 5-year age bands.

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$6.33 \times 10^{-3}$	$6.30 \times 10^{-3}$	$1.14 \times 10^{-1}$	-
v	1-2 ADLs	$-4.03 \times 10^{-2}$	$4.90 \times 10^{-2}$	$7.91 \times 10^{-3}$	-
	3–4 ADLs	$-3.01 \times 10^{-2}$	-	-	$4.38 \times 10^{-4}$
	5–6 ADLs	$-1.22 \times 10^{-3}$	$2.60 \times 10^{-3}$	$-3.87{ imes}10^{-2}$	-
	Inst'd	$-1.18 \times 10^{-2}$	$1.34 \times 10^{-2}$	$5.36 \times 10^{-2}$	-
	Dead	$1.30 \times 10^{-3}$	$2.73 \times 10^{-3}$	$8.11 \times 10^{-2}$	-
IADL only	Healthy	$-4.32 \times 10^{-1}$	$5.20 \times 10^{-1}$	$-4.69 \times 10^{-3}$	-
v	1-2 ADLs	$5.33 \times 10^{-2}$	$1.20 \times 10^{-1}$	$-1.80 \times 10^{-1}$	-
	3–4 ADLs	$-5.65 \times 10^{-1}$	-	-	$8.16 \times 10^{-3}$
	5–6 ADLs	$-1.90 \times 10^{-1}$	-	-	$2.87 \times 10^{-3}$
	Inst'd	$8.48 \times 10^{-2}$	-	-	$-8.27 \times 10^{-4}$
	Dead	$-4.88 \times 10^{-1}$	$5.06 \times 10^{-1}$	$3.97{ imes}10^{-4}$	-
1-2 ADLs	Healthy	$1.25 \times 10^{-1}$	-	-	$-1.10 \times 10^{-3}$
	IADL only	$4.22 \times 10^{-2}$	$8.65 \times 10^{-3}$	$1.20 \times 10^{-1}$	-
	3–4 ADLs	$-1.17{ imes}10^{-1}$	$2.60 \times 10^{-1}$	$-2.43 \times 10^{-2}$	-
	5–6 ADLs	$-5.91 \times 10^{-2}$	-	-	$1.29 \times 10^{-3}$
	Inst'd	$-2.36{ imes}10^{-1}$	$2.41 \times 10^{-1}$	$7.38 \times 10^{-3}$	-
	Dead	$-6.51 \times 10^{-2}$	-	-	$9.89 \times 10^{-4}$
3–4 ADLs	Healthy	$-1.26 \times 10^{-1}$	$1.36 \times 10^{-1}$	$-4.54 \times 10^{-3}$	-
	IADL only	$9.83 \times 10^{-2}$	-	-	$-1.07 \times 10^{-3}$
	1–2 ADLs	$9.77 \times 10^{-1}$	-	-	$-1.06{ imes}10^{-2}$
	5–6 ADLs	$4.06 \times 10^{-1}$	-	-	$-4.36 \times 10^{-3}$
	Inst'd	$-5.37{ imes}10^{-1}$	-	-	$7.95  imes 10^{-3}$
	Dead	$-1.20 \times 10^{-1}$	$1.31 \times 10^{-1}$	$1.44 \times 10^{-2}$	-
5–6 ADLs	Healthy	$-2.00 \times 10^{-2}$	$2.57 \times 10^{-2}$	$-1.50{ imes}10^{-2}$	-
	IADL only	$9.69 \times 10^{-2}$	-	-	$-1.05 \times 10^{-3}$
	1-2 ADLs	$3.56 \times 10^{-1}$	-	-	$-3.83 \times 10^{-3}$
	3-4 ADLs	$4.87 \times 10^{-2}$	-	-	$1.66 \times 10^{-4}$
	Inst'd	$-4.85 \times 10^{-2}$	$8.47 \times 10^{-2}$	$1.93 \times 10^{-2}$	-
	Dead	$-3.20 \times 10^{-1}$	$3.32 \times 10^{-1}$	$5.54 \times 10^{-4}$	-
Inst'd	Healthy	$-1.20 \times 10^{-1}$	$1.23 \times 10^{-1}$	$4.86 \times 10^{-3}$	-
	IADL only	$-1.90 \times 10^{-2}$	$2.75 \times 10^{-2}$	$-2.69 \times 10^{-2}$	-
	1-2 ADLs	$-7.31 \times 10^{-3}$	-	-	$1.11 \times 10^{-4}$
	3–4 ADLs	$3.69 \times 10^{-3}$	-	-	$-2.18 \times 10^{-5}$
	5-6 ADLs	$-2.11 \times 10^{-3}$	$2.34 \times 10^{-3}$	$7.11 \times 10^{-3}$	-
	Dead	$-8.51 \times 10^{-2}$	-	-	$1.59 \times 10^{-3}$

Table L.164:	Parameter	values for	the para	metric t	ransition	intensities	for	females,
calculated fr	5m the 1984	and 1989	NLTCS,	groupe	d in 5-yea	r age band	$\mathbf{s}.$	

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-4.52 \times 10^{-2}$	$6.04 \times 10^{-2}$	$1.77 \times 10^{-2}$	-
v	1–2 ADLs	$-1.64 \times 10^{-1}$	$1.68 \times 10^{-1}$	$1.22 \times 10^{-2}$	-
	3–4 ADLs	$1.92 \times 10^{-4}$	$2.15 \times 10^{-3}$	$9.47 \times 10^{-2}$	-
	5-6 ADLs	$-1.20 \times 10^{-2}$	$1.56 \times 10^{-2}$	$1.45 \times 10^{-2}$	-
	Inst'd	$1.48 \times 10^{-3}$	$1.98 \times 10^{-3}$	$1.35 \times 10^{-1}$	-
	Dead	$-1.20{ imes}10^{-2}$	$1.39 \times 10^{-2}$	$1.64 \times 10^{-2}$	-
IADL only	Healthy	$-8.35 \times 10^{-2}$	$1.93 \times 10^{-1}$	$-3.47 \times 10^{-2}$	-
v	1-2 $ADLs$	$7.88 \times 10^{-2}$	$8.64 \times 10^{-2}$	$2.13 \times 10^{-2}$	-
	3–4 ADLs	$-5.76 \times 10^{-2}$	-	-	$8.53 \times 10^{-4}$
	5-6 ADLs	$-6.87 \times 10^{-4}$	$1.62 \times 10^{-2}$	$4.83 \times 10^{-2}$	-
	Inst'd	$-8.76 \times 10^{-2}$	-	-	$1.44 \times 10^{-3}$
	Dead	$3.34 \times 10^{-4}$	$4.79 \times 10^{-3}$	$8.66 \times 10^{-2}$	-
1-2 ADLs	Healthy	$-1.90 \times 10^{-1}$	$2.17 \times 10^{-1}$	$-3.24 \times 10^{-3}$	-
	IADL only	$4.87 \times 10^{-2}$	-	-	$-9.45 \times 10^{-6}$
	3–4 ADLs	$-1.56 \times 10^{-2}$	$8.68 \times 10^{-2}$	$3.51 \times 10^{-2}$	-
	5-6 ADLs	$-1.65 \times 10^{-2}$	-	-	$5.16 \times 10^{-4}$
	Inst'd	$-6.79{ imes}10^{-2}$	$8.32 \times 10^{-2}$	$2.35 \times 10^{-2}$	-
	Dead	$-1.58{ imes}10^{-3}$	-	-	$8.26 \times 10^{-5}$
3–4 ADLs	Healthy	$6.89 \times 10^{-3}$	-	-	$9.45 \times 10^{-5}$
	IADL only	$2.46 \times 10^{-2}$	-	-	$-2.07 \times 10^{-4}$
	1–2 ADLs	$-1.38{ imes}10^{-1}$	$2.05 \times 10^{-1}$	$2.37 \times 10^{-3}$	-
	5-6 ADLs	$6.60 \times 10^{-2}$	-	-	$2.95 \times 10^{-4}$
	Inst'd	$-2.90{ imes}10^{-1}$	-	-	$4.30 \times 10^{-3}$
	Dead	$9.01 \times 10^{-3}$	$8.06 \times 10^{-3}$	$5.11 \times 10^{-2}$	-
5–6 ADLs	Healthy	$8.21 \times 10^{-3}$	-	-	$-1.49 \times 10^{-5}$
	IADL only	$8.97 \times 10^{-2}$	-	-	$-9.84 \times 10^{-4}$
	1–2 ADLs	$-1.68{ imes}10^{-1}$	$2.34 \times 10^{-1}$	$-1.20 \times 10^{-2}$	-
	3–4 ADLs	$-5.44 \times 10^{-2}$	$1.51 \times 10^{-1}$	$-2.57{ imes}10^{-2}$	-
	Inst'd	$3.82 \times 10^{-2}$	$4.68 \times 10^{-3}$	$1.22{ imes}10^{-1}$	-
	Dead	$-1.33 \times 10^{-1}$	-	-	$1.95 \times 10^{-3}$
Inst'd	Healthy	$-8.93 \times 10^{-2}$	$9.63 \times 10^{-2}$	$-5.60 \times 10^{-4}$	-
	IADL only	$-4.12 \times 10^{-3}$	-	-	$6.40 \times 10^{-5}$
	1–2 ADLs	$-2.22{ imes}10^{-2}$	$3.13 \times 10^{-2}$	$-1.41 \times 10^{-2}$	-
	3–4 ADLs	$-2.01 \times 10^{-1}$	$2.02 \times 10^{-1}$	$-1.70 \times 10^{-4}$	-
	5-6 ADLs	$-3.10 \times 10^{-2}$	$3.12 \times 10^{-2}$	$1.99 \times 10^{-3}$	-
	Dead	$1.57 \times 10^{-2}$	$2.08 \times 10^{-3}$	$1.26 \times 10^{-1}$	-

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$4.34 \times 10^{-3}$	$1.06 \times 10^{-2}$	$7.28 \times 10^{-2}$	-
v	1–2 ADLs	$-1.38{ imes}10^{-2}$	$2.02 \times 10^{-2}$	$5.07 \times 10^{-2}$	-
	3–4 ADLs	$-1.05 \times 10^{-4}$	$2.08 \times 10^{-3}$	$1.08 \times 10^{-1}$	-
	5-6 ADLs	$-1.55{ imes}10^{-2}$	$1.80 \times 10^{-2}$	$9.15 \times 10^{-3}$	-
	Inst'd	$-1.46 \times 10^{-3}$	$4.00 \times 10^{-3}$	$1.04 \times 10^{-1}$	-
	Dead	$-3.91 \times 10^{-3}$	$6.69 \times 10^{-3}$	$3.94 \times 10^{-2}$	-
IADL only	Healthy	$-6.90 \times 10^{-3}$	$1.27 \times 10^{-1}$	$-7.71 \times 10^{-2}$	-
v	1-2 $ADLs$	$1.30 \times 10^{-1}$	$1.46 \times 10^{-2}$	$7.45 \times 10^{-2}$	-
	3–4 ADLs	$-1.52 \times 10^{-1}$	-	-	$2.24 \times 10^{-3}$
	5–6 ADLs	$-2.33 \times 10^{-2}$	$3.66 \times 10^{-2}$	$4.25 \times 10^{-2}$	-
	Inst'd	$-5.52 \times 10^{-2}$	-	-	$1.05 \times 10^{-3}$
	Dead	$-3.27{ imes}10^{-2}$	$4.23 \times 10^{-2}$	$1.21 \times 10^{-2}$	-
1-2 ADLs	Healthy	$1.08 \times 10^{-1}$	-	-	$-1.02 \times 10^{-3}$
	IADL only	$1.29 \times 10^{-2}$	-	-	$5.82 \times 10^{-4}$
	3–4 ADLs	$6.14 \times 10^{-2}$	$2.28 \times 10^{-2}$	$7.14 \times 10^{-2}$	-
	5–6 ADLs	$1.39 \times 10^{-2}$	-	-	$1.81 \times 10^{-4}$
	Inst'd	$-6.77{ imes}10^{-2}$	$8.05 \times 10^{-2}$	$2.28 \times 10^{-2}$	-
	Dead	$-2.89 \times 10^{-3}$	-	-	$1.12 \times 10^{-4}$
3–4 ADLs	Healthy	$-5.47 \times 10^{-4}$	-	-	$1.55 \times 10^{-4}$
	IADL only	$-1.71 \times 10^{-2}$	$2.78 \times 10^{-2}$	$-2.15 \times 10^{-2}$	-
	1–2 ADLs	$7.92 \times 10^{-2}$	$4.22 \times 10^{-2}$	$-1.10 \times 10^{-1}$	-
	5–6 ADLs	$6.15 \times 10^{-2}$	-	-	$4.11 \times 10^{-4}$
	Inst'd	$-3.42 \times 10^{-1}$	-	-	$5.09 \times 10^{-3}$
	Dead	$1.28 \times 10^{-2}$	$4.62 \times 10^{-3}$	$1.00 \times 10^{-1}$	-
5–6 ADLs	Healthy	$1.39 \times 10^{-2}$	-	-	$-1.17{ imes}10^{-4}$
	IADL only	$1.08 \times 10^{-1}$	-	-	$-1.17 \times 10^{-3}$
	1–2 ADLs	$-1.06{ imes}10^{-1}$	$1.81 \times 10^{-1}$	$-1.76{ imes}10^{-2}$	-
	3–4 ADLs	$-3.48 \times 10^{-2}$	$1.34 \times 10^{-1}$	$-2.44 \times 10^{-2}$	-
	Inst'd	$4.28 \times 10^{-2}$	$3.80 \times 10^{-3}$	$1.31 \times 10^{-1}$	-
	Dead	$-3.52 \times 10^{-2}$	$3.96 \times 10^{-2}$	$2.76 \times 10^{-2}$	-
Inst'd	Healthy	$6.82 \times 10^{-4}$	$5.50 \times 10^{-3}$	$1.01 \times 10^{-2}$	-
	IADL only	$-3.12 \times 10^{-2}$	$3.63 \times 10^{-2}$	$-5.42 \times 10^{-3}$	-
	1–2 ADLs	$2.07 \times 10^{-2}$	-	-	$-2.12 \times 10^{-4}$
	3–4 ADLs	$-3.01 \times 10^{-3}$	$7.26 \times 10^{-3}$	$-3.56 \times 10^{-2}$	-
	5-6 ADLs	$-6.11 \times 10^{-2}$	$6.21 \times 10^{-2}$	$6.57{ imes}10^{-4}$	-
	Dead	$-5.45 \times 10^{-2}$	$7.38 \times 10^{-2}$	$1.67 \times 10^{-2}$	-

Table L.165: Parameter values for the parametric transition intensities for males and females, calculated from the 1984 and 1989 NLTCS, grouped in 5-year age bands.

Table L.166:	Parameter	values for	the para	ametric	transition	intensities	for	males,
calculated fro	om the 1989	and $1994$	NLTCS,	groupe	d in 5-year	age bands.		

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-6.39 \times 10^{-2}$	-	-	$1.11 \times 10^{-3}$
Ū	1-2 ADLs	$2.55 \times 10^{-3}$	$6.92 \times 10^{-3}$	$1.12 \times 10^{-1}$	-
	3–4 ADLs	$-2.37{ imes}10^{-2}$	-	-	$3.81 \times 10^{-4}$
	5–6 ADLs	$-6.26 \times 10^{-3}$	$9.59 \times 10^{-3}$	$1.25 \times 10^{-2}$	-
	Inst'd	$-6.57 \times 10^{-2}$	$6.86 \times 10^{-2}$	$4.87 \times 10^{-3}$	-
	Dead	$-4.56 \times 10^{-2}$	-	-	$6.92 \times 10^{-4}$
IADL only	Healthy	$9.98 \times 10^{-1}$	-	-	$-1.15 \times 10^{-2}$
v	$1-2$ $ {ADLs}$	$-4.21 \times 10^{-1}$	-	-	$7.05 \times 10^{-3}$
	3–4 ADLs	$1.12 \times 10^{-2}$	$4.43 \times 10^{-2}$	$-1.45 \times 10^{-1}$	-
	5-6 ADLs	$-8.01 \times 10^{-3}$	$8.38 \times 10^{-3}$	$6.95 \times 10^{-2}$	-
	Inst'd	$-7.84 \times 10^{-3}$	-	-	$3.51 \times 10^{-4}$
	Dead	$8.94 \times 10^{-2}$	-	-	$-9.11 \times 10^{-4}$
1-2 ADLs	Healthy	$2.06 \times 10^{-2}$	$3.97 \times 10^{-2}$	$-1.44 \times 10^{-1}$	_
	IADL only	$-1.30 \times 10^{-1}$	-	-	$2.17 \times 10^{-3}$
	3–4 ADLs	$5.79 \times 10^{-2}$	$2.36 \times 10^{-2}$	$8.29 \times 10^{-2}$	-
	5–6 ADLs	$-2.43 \times 10^{-1}$	-	-	$3.44 \times 10^{-3}$
	Inst'd	$-4.66 \times 10^{-1}$	$4.95 \times 10^{-1}$	$5.18 \times 10^{-3}$	-
	Dead	$-8.01 \times 10^{-3}$	$1.67 \times 10^{-2}$	$-6.88 \times 10^{-2}$	-
3–4 ADLs	Healthy	$-2.68 \times 10^{-1}$	$3.02 \times 10^{-1}$	$-4.11 \times 10^{-3}$	-
	IADL only	$2.61 \times 10^{-1}$	-	-	$-3.05 \times 10^{-3}$
	1-2 ADLs	$-2.01 \times 10^{-3}$	$6.93 \times 10^{-2}$	$-2.77 \times 10^{-1}$	-
	5-6 ADLs	$4.38 \times 10^{-1}$	-	-	$-3.79 \times 10^{-3}$
	Inst'd	$-4.59 \times 10^{-1}$	-	-	$6.60 \times 10^{-3}$
	Dead	$-4.02 \times 10^{-1}$	$4.04 \times 10^{-1}$	$1.00 \times 10^{-2}$	-
5–6 ADLs	Healthy	$4.00 \times 10^{-3}$	_	_	$1.45 \times 10^{-4}$
	IADL only	$3.93 \times 10^{-2}$	-	-	$-4.54 \times 10^{-4}$
	1-2 ADLs	$-2.64 \times 10^{-1}$	$3.09 \times 10^{-1}$	$-7.59 \times 10^{-3}$	-
	3–4 ADLs	$-1.86 \times 10^{-1}$	-	-	$2.56 \times 10^{-3}$
	Inst'd	$2.03 \times 10^{-1}$	-	-	$-1.72 \times 10^{-3}$
	Dead	$7.31 \times 10^{-2}$	-	-	$-6.86 \times 10^{-4}$
Inst'd	Healthy	$-4.99 \times 10^{-1}$	$5.05 \times 10^{-1}$	$1.11 \times 10^{-3}$	_
	IADL only	$-6.12 \times 10^{-2}$	$6.97 \times 10^{-2}$	$-5.71 \times 10^{-3}$	_
	1–2 ADLs	$-3.14 \times 10^{-2}$	_	_	$4.85 \times 10^{-4}$
	3–4 ADLs	$-1.10 \times 10^{-1}$	$1.10 \times 10^{-1}$	$5.33 \times 10^{-3}$	-
	5–6 ADLs	$-5.10 \times 10^{-2}$	$5.12 \times 10^{-2}$	$1.31 \times 10^{-3}$	-
	Dead	$-9.11 \times 10^{-2}$	$1.06 \times 10^{-1}$	$9.35 \times 10^{-3}$	-

From	То		Para	ameter	
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-1.08 \times 10^{-2}$	$2.33 \times 10^{-2}$	$4.58 \times 10^{-2}$	-
v	1–2 ADLs	$-6.11 \times 10^{-3}$	$2.51 \times 10^{-2}$	$4.19 \times 10^{-2}$	-
	3–4 ADLs	$-9.01 \times 10^{-2}$	$8.94 \times 10^{-2}$	$4.95 \times 10^{-3}$	-
	5-6 ADLs	$7.73 \times 10^{-4}$	$1.84 \times 10^{-3}$	$8.18 \times 10^{-2}$	-
	Inst'd	$-6.16 \times 10^{-3}$	$1.10 \times 10^{-2}$	$7.45 \times 10^{-2}$	-
	Dead	$1.43 \times 10^{-3}$	-	-	$1.34 \times 10^{-5}$
IADL only	Healthy	$-6.86 \times 10^{-3}$	$1.46 \times 10^{-1}$	$-6.44 \times 10^{-2}$	-
· ·	1-2 ADLs	$-8.04 \times 10^{-1}$	-	-	$1.25 \times 10^{-2}$
	3–4 ADLs	$-1.02 \times 10^{-2}$	$4.46 \times 10^{-2}$	$-1.31 \times 10^{-1}$	-
	5–6 ADLs	$8.92 \times 10^{-2}$	-	-	$-8.66 \times 10^{-4}$
	Inst'd	$-1.03 \times 10^{-1}$	-	-	$1.54 \times 10^{-3}$
	Dead	$-8.64 \times 10^{-2}$	-	-	$1.27 \times 10^{-3}$
1-2 ADLs	Healthy	$2.67 \times 10^{-1}$	-	-	$-2.82 \times 10^{-3}$
	IADL only	$-4.96 \times 10^{-2}$	-	-	$1.46 \times 10^{-3}$
	3–4 ADLs	$-1.22 \times 10^{-1}$	-	-	$3.47 \times 10^{-3}$
	5-6 ADLs	$1.15 \times 10^{-2}$	-	-	$2.88 \times 10^{-4}$
	Inst'd	$-9.01 \times 10^{-3}$	$7.19 \times 10^{-3}$	$1.22{ imes}10^{-1}$	-
	Dead	$-5.56 \times 10^{-2}$	-	-	$8.34 \times 10^{-4}$
3–4 ADLs	Healthy	$7.05 \times 10^{-3}$	-	-	$1.02 \times 10^{-4}$
	IADL only	$1.88 \times 10^{-1}$	-	-	$-2.09 \times 10^{-3}$
	1-2 ADLs	$-5.03 \times 10^{-1}$	-	-	$7.52 \times 10^{-3}$
	5–6 ADLs	$-3.63 \times 10^{-1}$	-	-	$5.81 \times 10^{-3}$
	Inst'd	$-9.55{ imes}10^{-2}$	$1.60 \times 10^{-1}$	$1.20 \times 10^{-2}$	-
	Dead	$1.94 \times 10^{-2}$	-	-	$-8.64 \times 10^{-5}$
5–6 ADLs	Healthy	$-9.46 \times 10^{-2}$	$1.07 \times 10^{-1}$	$5.51 \times 10^{-3}$	-
	IADL only	$2.42 \times 10^{-3}$	-	-	$8.68 \times 10^{-6}$
	1–2 ADLs	$-1.20 \times 10^{-1}$	-	-	$1.78 \times 10^{-3}$
	3–4 ADLs	$7.84 \times 10^{-3}$	-	-	$8.10 \times 10^{-4}$
	Inst'd	$-1.95{ imes}10^{-1}$	$2.04 \times 10^{-1}$	$1.68 \times 10^{-2}$	-
	Dead	$-4.04 \times 10^{-2}$	-	-	$9.13 \times 10^{-4}$
Inst'd	Healthy	$-1.22 \times 10^{-2}$	$2.98 \times 10^{-2}$	$-1.33 \times 10^{-2}$	-
	IADL only	$-2.67 \times 10^{-3}$	-	-	$5.17 \times 10^{-5}$
	1-2 ADLs	$2.51 \times 10^{-4}$	$2.31 \times 10^{-2}$	$-1.08 \times 10^{-1}$	-
	3–4 ADLs	$-1.95{ imes}10^{-4}$	-	-	$2.69 \times 10^{-5}$
	5–6 ADLs	$5.08 \times 10^{-3}$	-	-	$-4.82 \times 10^{-5}$
	Dead	$-4.52 \times 10^{-3}$	-	-	$3.80 \times 10^{-4}$

Table L.167: Parameter values for the parametric transition intensities for females, calculated from the 1989 and 1994 NLTCS, grouped in 5-year age bands.

From	То	Parameter			
State	State	А	В	$\mathbf{C}$	D
Healthy	IADL only	$-1.04 \times 10^{-1}$	-	-	$1.67 \times 10^{-3}$
v	1-2 ADLs	$1.07 \times 10^{-4}$	$1.51 \times 10^{-2}$	$6.17 \times 10^{-2}$	-
	3–4 ADLs	$-2.10 \times 10^{-2}$	$2.08 \times 10^{-2}$	$2.05 \times 10^{-2}$	-
	5–6 ADLs	$1.46 \times 10^{-3}$	$1.85 \times 10^{-3}$	$7.01 \times 10^{-2}$	-
	Inst'd	$3.29 \times 10^{-4}$	$3.98 \times 10^{-3}$	$1.15 \times 10^{-1}$	-
	Dead	$1.03 \times 10^{-3}$	$2.05 \times 10^{-3}$	$7.19 \times 10^{-2}$	-
IADL only	Healthy	$6.24 \times 10^{-1}$	-	-	$-6.70 \times 10^{-3}$
· ·	1–2 ADLs	$-8.99{ imes}10^{-1}$	-	-	$1.38 \times 10^{-2}$
	3–4 ADLs	$-4.31 \times 10^{-2}$	$9.06 \times 10^{-2}$	$-6.31 \times 10^{-2}$	-
	5–6 ADLs	$1.08 \times 10^{-2}$	-	-	$7.14 \times 10^{-5}$
	Inst'd	$-8.76 \times 10^{-2}$	-	-	$1.36 \times 10^{-3}$
	Dead	$-4.53 \times 10^{-2}$	-	-	$7.58 \times 10^{-4}$
1-2 ADLs	Healthy	$2.74 \times 10^{-1}$	-	-	$-2.91 \times 10^{-3}$
	IADL only	$-9.52 \times 10^{-2}$	-	-	$1.95 \times 10^{-3}$
	3–4 ADLs	$-6.42 \times 10^{-2}$	-	-	$2.92 \times 10^{-3}$
	5–6 ADLs	$-1.45 \times 10^{-1}$	$1.60 \times 10^{-1}$	$1.01 \times 10^{-2}$	-
	Inst'd	$-2.30{ imes}10^{-2}$	$2.28 \times 10^{-2}$	$8.01 \times 10^{-2}$	-
	Dead	$-5.40 \times 10^{-2}$	$5.75 \times 10^{-2}$	$1.47 \times 10^{-2}$	-
3–4 ADLs	Healthy	$-6.40 \times 10^{-2}$	$8.68 \times 10^{-2}$	$-8.23 \times 10^{-3}$	-
	IADL only	$2.09 \times 10^{-1}$	-	-	$-2.35 \times 10^{-3}$
	1–2 ADLs	$6.81 \times 10^{-2}$	$3.63 \times 10^{-3}$	$1.72{ imes}10^{-1}$	-
	5-6 ADLs	$-2.30 \times 10^{-2}$	-	-	$1.81 \times 10^{-3}$
	Inst'd	$3.61 \times 10^{-2}$	$2.44 \times 10^{-2}$	$6.58 \times 10^{-2}$	-
	Dead	$-1.95 \times 10^{-1}$	$2.07 \times 10^{-1}$	$3.16 \times 10^{-3}$	-
5–6 ADLs	Healthy	$-2.22{ imes}10^{-2}$	-	-	$5.21 \times 10^{-4}$
	IADL only	$-5.96 \times 10^{-3}$	-	-	$1.44 \times 10^{-4}$
	1–2 ADLs	$-2.94{ imes}10^{-2}$	$7.07 \times 10^{-2}$	$-2.76{ imes}10^{-2}$	-
	3–4 ADLs	$-1.30 \times 10^{-1}$	-	-	$2.43 \times 10^{-3}$
	Inst'd	$-4.54 \times 10^{-2}$	$8.60 \times 10^{-2}$	$2.50 \times 10^{-2}$	-
	Dead	$-2.93 \times 10^{-2}$	-	-	$8.14 \times 10^{-4}$
Inst'd	Healthy	$-3.60 \times 10^{-2}$	$5.31 \times 10^{-2}$	$-5.38{ imes}10^{-3}$	-
	IADL only	$-1.89{ imes}10^{-4}$	-	-	$3.71 \times 10^{-5}$
	1–2 ADLs	$6.36 \times 10^{-2}$	-	-	$-6.58{ imes}10^{-4}$
	3-4 ADLs	$-8.25 \times 10^{-3}$	-	-	$1.38 \times 10^{-4}$
	5-6 ADLs	$1.60 \times 10^{-3}$	-	-	$-3.20 \times 10^{-6}$
	Dead	$-2.55{ imes}10^{-2}$	-	-	$6.49 \times 10^{-4}$

Table L.168: Parameter values for the parametric transition intensities for males and females, calculated from the 1989 and 1994 NLTCS, grouped in 5-year age bands.

## Appendix M

Graphs of the constrained (positive) MLEs of the transition intensities, 95% confidence intervals and parametric fits, using data in 5-year age bands from the 1982 and 1984 NLTCS

Figures M.89 to M.94 and M.95 to M.100 give graphs of the transition intensities, for males and females respectively, out of states 1–6 in turn, for the 1982–84 NLTCS. The graphs for males and females together are given in Figures 5.49 to 5.54. They show the point estimates (constrained (positive) MLEs) for the data grouped in 5-year age bands, the confidence intervals calculated from the variance estimates (in Section 5.4) and the parametric form of the transition intensities (calculated in Section 5.6).



Figure M.89: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.90: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for males grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.91: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.92: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.93: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.94: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for males grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.95: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.96: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.97: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.98: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.99: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for females grouped in 5-year age bands in the 1982–84 NLTCS.



Figure M.100: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for females grouped in 5-year age bands in the 1982–84 NLTCS.

## Appendix N

Graphs of the constrained (positive) MLEs of the transition intensities, 95% confidence intervals and parametric fits, using data in 5-year age bands from the 1984 and 1989 NLTCS

Figures N.101 to N.106, N.107 to N.112 and N.113 to N.118 give graphs of the transition intensities, for males, females and in aggregate respectively, out of states 1–6 in turn, for the 1984–89 NLTCS. They show the point estimates (constrained (positive) MLEs) for the data grouped in 5-year age bands, the confidence intervals calculated from the variance estimates (in Section 5.4) and the parametric form of the transition intensities (calculated in Section 5.6).



Figure N.101: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.102: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for males grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.103: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.104: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.105: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.106: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for males grouped in 10-year age bands in the 1984–89 NLTCS.



Figure N.107: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.108: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.109: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.110: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.111: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.112: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.113: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males and females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.114: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for males and females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.115: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males and females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.116: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males and females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.117: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males and females grouped in 5-year age bands in the 1984–89 NLTCS.



Figure N.118: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for males and females grouped in 5-year age bands in the 1984–89 NLTCS.

## Appendix O

Graphs of the constrained (positive) MLEs of the transition intensities, 95% confidence intervals and parametric fits, using data in 5-year age bands from the 1989 and 1994 NLTCS

Figures O.119 to O.124, O.125 to O.130 and O.131 to O.136 give graphs of the transition intensities, for males, females and in aggregate respectively, out of states 1–6 in turn, for the 1989–94 NLTCS. They show the point estimates (constrained (positive) MLEs) for the data grouped in 5-year age bands, the confidence intervals calculated from the variance estimates (in Section 5.4) and the parametric form of the transition intensities (calculated in Section 5.6).



Figure O.119: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.120: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for males grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.121: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.122: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.123: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.124: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for males grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.125: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.126: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.127: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.128: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.129: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.130: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.131: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Healthy' state for males and females grouped in 10-year age bands in the 1989–94 NLTCS.



Figure O.132: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'IADL only' state for males and females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.133: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '1–2 ADLs' state for males and females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.134: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '3–4 ADLs' state for males and females grouped in 5-year age bands in the 1989–94 NLTCS.


Figure O.135: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the '5–6 ADLs' state for males and females grouped in 5-year age bands in the 1989–94 NLTCS.



Figure O.136: Graphs of point estimates (CMLE), approximate 95% confidence intervals and parametric fits of the transition intensities out of the 'Institutionalized' state for males and females grouped in 5-year age bands in the 1989–94 NLTCS.